# ECONOMIC DYNAMISM IN THE ASIA-PACIFIC

EDITED BY Grahame Thompson

Also available as a printed book see title verso for ISBN details

# **Economic Dynamism in the Asia-Pacific**

This book is part of a series produced in association with The Open University. While each book in the series is self-contained, there are also references to the other books. Readers should note that references to other books in the series appear in bold type. The list of other books in the series is:

Asia-Pacific in the New World Order, edited by Anthony McGrew and Christopher Brook

Culture and Society in the Asia-Pacific, edited by Richard Maidment and Colin Mackerras

*Governance in the Asia-Pacific*, edited by Richard Maidment, David Goldblatt and Jeremy Mitchell

## *The Asia-Pacific Profile*, edited by Bernard Eccleston, Michael Dawson and Deborah McNamara

The books form part of the Open University course DD302 *Pacific Studies*. If you would like to study this or any other Open University course, details can be obtained from the Course Reservations Centre, P.O. Box 724, The Open University, Milton Keynes, MK7 6ZS, United Kingdom. For availability of other course components, contact Open University Worldwide Ltd, The Berrill Building, Walton Hall, Milton Keynes, MK7 6AA, UK.

# Economic Dynamism in the Asia-Pacific:

The Growth of Integration and Competitiveness

Edited by Grahame Thompson



London and New York in association with



First published 1998 by Routledge 11 New Fetter Lane, London EC4P 4EE This edition published in the Taylor & Francis e-Library, 2005.

"To purchase your own copy of this or any of Taylor & Francis or Routledge's collection of thousands of eBooks please go to http://www.ebookstore.tandf.co.uk/."

Simultaneously published in the USA and Canada by Routledge 29 West 35th Street, New York, NY 10001

© The Open University 1998

Edited, designed and typeset by The Open University

All rights reserved. No part of this publication may be reprinted or reproduced or utilized in any form or by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying and recording, or in any information storage or retrieval system, without permission in writing from the publishers.

British Library Cataloguing in Publication Data A catalogue record for this book is available from The British Library

Library of Congress Cataloging in Publication Data A catalogue record for this book has been requested

ISBN 0-203-98012-3 Master e-book ISBN

ISBN 0-415-17273-X (hbk) ISBN 0-415-17274-8 (pbk)

1.1

## CONTENTS

	PREFACE	vii
Chapter 1	Introduction: contours of economic development in the Asia- Pacific	1
	Grahame Thompson	
PART 1 ECO	NOMIC DYNAMISM CONSIDERED	19
Chapter 2	<b>Patterns of trade, investment and migration in the Asia-Pacific</b> <b>region</b> <i>Partha Gangopadhyay</i>	20
Chapter 3	<b>Growth: its sources and consequences</b> <i>Ajit Singh</i>	53
Chapter 4	<b>Financial systems and monetary integration</b> <i>Grahame Thompson</i>	80
Chapter 5	<b>Engendering the 'economic miracle': the labour market in the</b> <b>Asia-Pacific</b> <i>Lucie Cheng and Ping-chun Hsuing</i>	109
Chapter 6	<b>The role of the state in economic development</b> Stephen W.K.Chiu and Tai-lok Lui	134
PART 2 THE	PROCESSES OF INTEGRATION	157
Chapter 7	<b>Economic integration in the Asia-Pacific region</b> <i>Ippei Yamazawa</i>	158
Chapter 8	Measuring the size of foreign multinationals in the Asia-Pacific Eric Ramstetter	179
Chapter 9	<b>East-Asian and Anglo-American business systems</b> <i>Richard Whitley</i>	206

Chapter10	<b>Chapter10 Technological systems, innovation and transfers</b> Brendan Barker and Akira Goto		
PART 3 CON	SEQUENCES AND IMPLICATIONS	262	
Chapter11	Winners and losers in the Asia-Pacific Paul Lubeck	263	
Chapter12	<b>The environment, traditional production and population</b> <i>Elspeth Young, Colin Hunt and R.Gerard Ward</i>	289	
Chapter13	The Pacific and beyond: APEC, ASEM and regional economic management Richard Higgott	319	
Chapter14	<b>Danger and opportunity in the Asia-Pacific</b> Jeffrey Henderson	338	
	ACKNOWLEDGEMENTS	364	
	LIST OF CONTRIBUTORS	366	
	INDEX	369	

### Series preface

The five volumes in this series are part of a new Open University course, *Pacific Studies*, which has been produced within the Faculty of Social Sciences. The appearance of *Pacific Studies* is due to the generous and enthusiastic support the course has received from the University and in particular from colleagues within the Faculty of Social Sciences. The support has been especially remarkable given that this course has ventured into relatively uncharted scholarly waters. The potential risks were readily apparent but the commitment always remained firm. I am very grateful.

There are too many people to thank individually, both within and outside of the Open University, but I must record my appreciation for some of them. Within the University, I would like to acknowledge my colleagues Anthony McGrew and Grahame Thompson. Pacific Studies could not have been made without them. Their role was central. They were present when the course was conceived and they lived with it through to the final stages. They also made the experience of making this course both very enjoyable and intellectually stimulating. Christopher Brook and Bernard Eccleston made an enormous contribution to the course far beyond their editorial roles in two of the books in the series. They read the successive drafts of all chapters with great care and their perceptive comments helped to improve these volumes considerably. David Goldblatt and Jeremy Mitchell, because of their other commitments, may have joined the Course Team relatively late in the production process, but their contributions, especially to Governance in the Asia-Pacific have been much appreciated. Michael Dawson played an especially important role in the production of The Asia-Pacific Profile and his calm and genial presence was valued as always. Jeremy Cooper and Eleanor Morris of the BBC were responsible for the excellent audio-visual component of Pacific Studies. Anne Carson, the Course Manager of Pacific Studies, was consistently cheerful and helpful. All of the volumes in this series have been greatly improved by the editorial craftsmanship of Stephen Clift, Tom Hunter and Kate Hunter, who have been under great pressure throughout the production of this course, but nevertheless delivered work of real quality. The striking cover designs of Richard Hoyle and Jonathan Davies speak for themselves and the artwork of Ray Munns in all five volumes has been most impressive. Paul Smith, whose recent retirement from the University will leave a very real gap, made his usual remarkable contribution in providing unusual and interesting illustrations. Giles Clark of the Copublishing Department was a constant source of encouragement and in addition his advice was always acute. Our colleagues in Project Control, especially Deborah Bywater,

and in the Operations Division of the University, were far more understanding and helpful than I had any right to expect. Anne Hunt and Mary Dicker, who have been responsible for so much of the work in this Faculty over the past several years, performed to their usual exacting standards by preparing the manuscripts in this series for publication with remarkable speed and accuracy. They were very ably assisted by Chris Meeks and Doreen Pendlebury.

Pacific Studies could not have been made without the help of academic colleagues based in the UK as well as in the Asia-Pacific region. This series of books has drawn on their scholarship and their expertise but above all on their generosity. I must record my appreciation to all of them for their participation in this project. The Course Team owes an especially large debt to Dr Gerry Segal, Senior Fellow at the International Institute of Strategic Studies, who was the External Assessor of Pacific Studies. He was both an enthusiastic supporter of this project as well as a very shrewd critic. His wise counsel and tough advice have greatly improved the volumes in this series. It has been a pleasure to work with Professor Colin Mackerras, Director of the Key Centre for Asian Studies and Languages at Griffith University in Australia. Griffith University and the Open University have collaborated over the production of Pacific Studies; an arrangement that has worked extremely well. The success of this collaboration has been due in no small part to Colin. Over the past three years I have come to appreciate his many qualities particularly his immense knowledge of the Asia-Pacific region as well as his patience and courtesy in dealing with those of us who know far less. I would also like to thank all of those colleagues at Griffith who have helped to make this collaboration so successful and worthwhile, especially Professor Tony Bennett, who played a key role during the initial discussions between the two universities. Frank Gibney, President of the Pacific Basin Institute, was always available with help, advice and encouragement. It was one of the real pleasures of this project to have met and worked with Frank and the PBI. This series has also benefited considerably from the enthusiasm and insight of Victoria Smith at Routledge.

The production of *Pacific Studies* was helped greatly through the assistance of several foundations. The Daiwa Anglo-Japanese Foundation awarded this project two grants and its Director General, Christopher Everett, was a model of generosity and support. He invited the Course Team to use the attractive facilities of the Foundation; an invitation which was accepted with enthusiasm. The grant from The Great Britain Sasakawa Foundation was also greatly appreciated as was the advice, encouragement and the shrewd counsel of Peter Hand, the Administrator of the Foundation. Mr Tomoyuki Sakurai the Director of the Japan Foundation in London was always interested in the development of *Pacific Studies* and I have no doubt that this resulted in a generous grant from the Foundation. Mr Haruhisa Takeuchi, formerly Director of the Japan Information and Cultural Centre, was most supportive during the early stages of this project and his successor at the Centre, Mr Masatoshi Muto has been no less helpful. Finally, I must record my thanks to the British Council in Australia for their assistance which was much appreciated.

Richard Maidment Chair, *Pacific Studies* Milton Keynes, November 1997

### Economic Dynamism in the Asia-Pacific: preface

Many individuals contributed to the production of this volume. In particular I would like to thank Jeffrey Henderson for his invaluable advice and assistance in structuring the book, suggesting chapter authors and commenting on the individual chapters. In addition Chris Brook, Bernie Eccleston, Tony McGrew, Richard Maidment and Gerald Segal offered their considerable advice on numerous drafts. The authors of individual chapters proved exceptionally efficient in working to a demanding schedule and taking into account the diverse comments from the Course Team and Editors. Without Doreen Pendlebury, Anne Hunt and Mary Dicker the final manuscript would not have been typed and presented so professionally. Tom Hunter and Kate Hunter did a marvellous job of editing individual chapters and to a very tight schedule.

> Grahame Thompson Walton Hall, January 1998

### CHAPTER 1 Introduction: contours of economic development in the Asia-Pacific

Grahame Thompson

### 1.1 The regional record considered

Reports from international agencies on the future of East Asia published in the 1950s and 1960s were pessimistic about the growth and developmental potential for the region's economies. The general sentiment was that only Japan and the Philippines were likely to reach the status of advanced countries. How wrong this assessment has proved to be! Whilst Japan has presumably passed all expectations, the Philippines has been conspicuous by its relatively poor performance when compared to its East Asian neighbours (see Table 1.2 below). Even in the 1970s and 1980s, the UN development reports were cautious as to the prospects for these countries as a whole.

In fact their record has been outstanding, and has far eclipsed the growth rates of advanced countries and other developing areas alike. Figure 1.1 demonstrates the comparative record of the East Asian economies over the period 1965–90, particularly the eight so called 'high-performing economies' (HPAEs; Japan, the Republic of Korea (hereafter South Korea), Malaysia, Indonesia, Taiwan, China, Hong Kong and Singapore). This record is confirmed by the trends identified in Figure 1.2, which shows an index of income per head for various developing areas plotted between 1970–90, the period of the rapid growth of interest in the economies of the Asia-Pacific region.

The consequences of these growth rates were that in 1995 the GDP per head in Hong Kong and Singapore were US\$ 23,900 and US\$ 22,600 respectively, compared to an average of just US\$ 19,400 for the rich industrial OECD economies as a whole. (Note that other countries were less well off at the time: Taiwan, US \$13,200; South Korea, US\$ 11,900; Malaysia, US\$ 10,400; Thailand, US\$ 8,000; Indonesia, US\$ 3,800; China, US\$ 3,100; Philippines, US\$ 2,800—all at 1995 purchasing power parity equivalents). By all accounts, these growth rates were as unanticipated by





Source: World Bank (1993, p. 2)

the countries involved as they were by the international organizations monitoring and analysing them. In retrospect, then, the period around the mid 1960s might be seen as one involving an unexpected *disjuncture* for the countries themselves, as well as for the international economic community beyond. It launched these countries onto a new and unanticipated growth trajectory.

These figures and trends provide the basic reason why the dynamic development of the countries in this region has become the object of so much analytical interest. Untangling the causes of this exceptional economic growth is the subject of a number of the chapters in this book, which examine the issue from different perspectives and approaches.

However, as soon as this is embarked upon another question immediately arises; How much should the countries in the region be considered as an integrating and interdependent regional grouping as a whole rather than as an isolated group of individual economies. Does it make up a genuine 'regional' economy? Broadly speaking, the chapters argue the positive case for this. The area at least demonstrates many of the main characteristics of a region as defined by **Buzan** (1998). Considering it as a region, however—particularly as an economic region—means that three further interconnected issues also arise.

First is the relationship between the East Asian side of the Pacific Ocean and the North American side. As the chapters in the book demonstrate, it would be impossible to treat the East Asian countries independently of their North American counterparts, since one of the main reasons for the growth of the East Asian economies is the fact of their access to the North American markets, particularly that of the US economy (see Chapter 2).



Source: World Bank (1997)

This provided the crucial demand for their products, and without it it is difficult to see how the economic advances they have made could have begun in the first place, let alone been sustained over the intermediate period. In addition there was the strategic and security relationships between the two sides of the Ocean that also fostered closer economic ties (McGrew and Brook, 1998).

Second, the regionalization of these economies has involved a public 'institutional moment' as well as being driven by the private decisions made by market agents. This latter is generally agreed to have been the major reason why a regional economy as such is emerging. As the chapters testify, however, the precise nature of the institutional moment—described as an 'open regionalism' in Chapter 7—is not one particularly familiar to a Western audience. The premier forum for this institutional moment in the construction of the Asia-Pacific as a region has been the APEC process, inaugurated in 1989. For reasons that are discussed in the book, this forum extended its embrace not only to the countries mentioned so far but also to the Anglo-American states to the South (Australia and New Zealand) and to Mexico in Central America, and to Chile in South America. The *diversity* of the economies linked together in this common purpose is the cause of both the strengths and the weaknesses of the integrative process, as later chapters demonstrate. But for reasons already mentioned it is still appropriate to treat the Asia-Pacific as a regional set of individual ones.

	1913	1925	1938	1955	1990
China	65	46	70	43	59
Indonesia	32	38	26	32	60
Taiwan			99	50	42
Japan	41	47	70	22	29
South Korea			100	35	40
Malaysia	44	39	35	30	37
Philippines	18	15	11	17	43
Thailand	62	71	65	52	51
Simple average	42	43	59	35	45
Excluding South Korea, Taiwan		43	46	33	47
Excluding South Korea, Taiwan and Japan	42	42	41	35	50

Table 1.1 East Asian trade as a share of totaltrade for different countries (exports plus importsas a percentage of total trade)

Note: Higher percentages indicate greater trade intensity Source: Petri (1994, p. 111)

An interesting aside concerns the historical significance of the contemporary integrationist moves being made in the Asia-Pacific region. As is pointed out in Chapter 7, there were precursors of this in the 1960s and 1970s. But in many ways there was a similar (or even greater) level of integration amongst the East Asian group achieved earlier this century, particularly during the period of Japanese imperial expansion of 1931–45, than had been achieved by the early 1990s (Petri, 1994). Table 1.1 shows the levels of East Asian trade for a number of countries over the period 1913–90. Thus even in 1913 the level of East Asian trade integration was higher than it was in 1955, and nearly as high as in 1990. The imperial bloc established by the Japanese—known as the 'Japanese Co-prosperity Sphere'—involved enormous amounts of Japanese public and private investment in the countries it invaded as well as trade with them. Since 1938 the trade intensity of the East Asian economies between themselves has declined as they have become more 'international' in terms of the spread of their trading partners, even as Asia-Pacific integration are not necessarily unprecedented.

However, there is a third aspect to the contemporary nature of regional development in the Asia-Pacific as a whole. It has also had something to do with the open nature of the global economy beyond the immediate shores of the Pacific Ocean. The growth of the region's economies has benefited from the liberal trade and overseas investment regimes adopted by the institutions set up to manage and govern the international economy after the ending of the Second World War. These organizations, particularly the World Bank and International Monetary Fund (IMF), the General Agreement on Tariffs and Trade (GATT), the World Trade Organization (WTO), and the Organization for Economic Cooperation (OECD), established the basic contours for a liberal open international trading system of the post-war years, one into which the economies of East Asia were adept at inserting themselves.

Thus there has been this further element involved in the overall picture of the development of this region and its economies; the structure of the international economy beyond the Pacific Rim itself. Of particular importance here is the engagement with the European economies and the integrative development of the European Union (EU) (see Chapters 2 and 13). One of the concerns expressed in this context has been the compatibility of the development of these regionalized economic blocks—considered in various permutations—with that open and liberal trading regime set up after the war (see Chapters 7 and 13). This regime came to be known as a 'multilateral' one in which all participants were treated as broadly on a par and formally at least negotiated and agreed amongst themselves on the nature of that trading system. The question this raises is whether the regionalism demonstrated by the EU, the North American Free Trade Agreement (NAFTA) and the APEC processes remain compatible with the continuation of a multilateral approach to international economic governance? This question is also explored in the chapters of this book where, broadly speaking, a positive answer to this is given, although with some qualification (see especially Chapter 14).

But herein lies another point of contention that figures in the chapters. What exactly has been the role of the World Bank and the GATT/WTO in the development of this region? Apart from these organizations emphasizing the maintenance of an open liberal international economy, they have also been instrumental in calling for policies of trade liberalization and internal de-regulation, directed at the East Asian economies in particular. Broadly speaking, this proposed policy package can be characterized as leading to a 'market friendly' approach to the economic policy pursued by the East Asian NIEs and ASEAN economies. The question is how far the actual practice of these economies followed this market-friendly advice, and what have been the implications for it emerging from the experience of their development strategies over the entire post-war period? Again this is the subject of considerable controversy, and it forms the core of one of the main themes for the book as a whole.

Whilst not an entirely new debate for the regional economies involved (see Chapter 4), the contemporary resonance of this controversy was stimulated by the World Bank's 1993 report, *The East Asian Miracle*. This report set many of the contours for the contemporary debate about the nature and form of the economic experiences in these countries. And the report has proved controversial, again as Chapters 3, 4, 6, 11 and 14 indicate. The story of how this particular report emerged and the intense diplomatic activity that went on around it is itself fascinating and revealing (Wade, 1996). The Japanese government financed the report since it wanted its particular development strategy properly assessed, and it thought that the other East Asian economies would demonstrate something novel and unique for the dominant Western interpretation of the intellectual line adopted by most Western economists and embodied in the World Bank's outlook and policy approach. The World Bank is often accused of being the bastion of a 'neo-classical' approach to the analysis of economic problems and to policy recommendations.

### **Box 1.1 Neo-classical economics**

Neo-classical economics is the dominant approach within the economics profession. Its analytical structure begins with the decisions and behaviours of individual autonomous economic agents, and asks how their activities, initially considered separately, combine to produce the best outcome for society as a whole. Economic actors are presumed to act 'rationally' in terms of calculating their own best interests and advantages. It is the operation of the market and competition between agents that is the mechanism securing the co-ordination of their individual decisions. As far as possible this mechanism should be left to itself to establish the best possible allocation and distribution of resources, via the operation of the price system. Government intervention only serves to 'distort' this automatically functioning system, and leads to a misallocation of resources. The same is true of any other impediment to the proper functioning of markets and prices, like the activity of monopolies, trade unions, artificial restrictions on the entry to markets, rules and regulations that restrict business, etc. This approach to economics, then, offers both a powerful analytical endorsement of the benefits to be had from the operation of a free market, and a ready made set of policy prescriptions as to how to most appropriately achieve this objective.

The question raised in the context of the East Asian economies in particular is how far their own developmental experiences conform to this model, or can be used to further justify its policy recommendations. Herein lies a major thematic issue that characterizes a number of the chapters (Chapters, 2, 3, 4, 6, 7, 11 and 14) in this book. For many scholars who have studied the East Asian region, the image of a free market system operating to provide the dynamic growth indicated above just does not adequately describe the characteristics of those economies. Indeed, quite the reverse is the case. These economies, they would argue, developed precisely because they did not either conform to the neo-classical model just outlined, nor follow its policy prescriptions. In particular, the role that the state played in the economy was crucial to their success, and the state initially operated to 'protect' these economies from both the full operation of internal markets and, perhaps even more importantly, the international market beyond.

The precise nature of the arguments over this issue can be left to unfold as the chapters are considered in turn. Although this theme is traditionally characterized as the debate between an approach stressing the importance of the 'market' on the one hand, and that stressing the importance of the 'state' on the other—in the form of the state *versus* the market—it will soon become apparent that this image is too simple. To begin with there have been significant *differences* in the role that the state and the market have played even within the East Asian sub-region of the Asia-Pacific, let alone for the region as a whole. As is demonstrated in the chapters, it is more the delicate *balance between* governments and business interests, which typify *all* the countries in the region, that accounts for their *diverse* development trajectories. And this relationship is not a static one, but a dynamic changing one, reacting and evolving with changes in internal and external pressures and circumstances, in elite outlooks and in policy fashions.

### 1.2 Characterizing the economies as a whole

The issues of the relationships between governments and states (Chapter 6), the nature of the financial systems (Chapter 4), the organization of business interests (which are themselves differently specified—Chapters 7 and 9), the characteristics of labour (Chapter 5) and of policy elites (Chapter 7), all combine not only to illustrate diversity and differences, but to raise an issue about how we can best characterize as a whole the countries and economics being dealt with in this book. What is the specification of their socio-economic characteristics and formations? Can any sensible generalizations be made into which we can 'place' particular countries?

Clearly, at one level the differences between 'capitalist' economies and 'command' (or 'communist') economies forms one such categorization, and this difference is manifest within the region. China, North Korea and Vietnam were probably the best illustrative examples of the communist type in the mid 1990s, although they demonstrated their own peculiarities at the same time. However, supposing we were to classify all the other countries of the Asia-Pacific simply as 'capitalist'-which at one level is quite reasonable—then we will not have advanced very far. As the chapters demonstrate, the interesting nature of this region is the vast organizational differences between those economies that can be included under the category of 'capitalist'. The economic performance of these different economies varies enormously. So we would need some other level or criteria for distinguishing between the members of the 'capitalist' group. Quite what these levels and criteria are, or could be, is itself highly disputed within the social sciences, and we are not about to solve these problems here let alone embark upon a long discussion of what they might entail. Suffice it to say that the chapters very much develop their own categorizations and criteria as suits their purpose and analytical style. Without wishing to pre-empt any of this there are some ways in which these categorizations share common ground and terms, which it is worth drawing attention to at this stage.

The first level is in terms of the distinction referred to already: that between 'capitalist' economies, broadly conceived, and 'communist' or command ones. A capitalist economy is typified by the private ownership of resources, the central mechanism of resource allocation being the market, with the operation of a price system determined by the 'free' interaction between supply and demand, and typified by a freedom of contractual arrangements between dispersed economic agents. A command economy, by contrast, is typified by the collective ownership of resources, the setting of prices by administrative edict, the planning of resource allocation, and the hierarchical determination of contractual arrangements by bureaucratic means mainly between giant productive organizations owned by the state. The 'plan' allocates rather than the 'market'.

One complication with this simple dichotomy is that forms of collective or public activity are operative within what are clearly capitalist economies, so the activity of the state is sometimes a crucial component in the overall operation of the capitalist economy, to varying degrees—the East Asian NIEs being obvious examples. On the other hand, in command economies, there has usually been at least some role and space for market transactions and operations, even if only a small one. For instance, some of the economic activity carried out in contemporary China could be argued to conform to a market system, with 'relatively' free price formation, whilst the ownership of resources remains more or less completely collective (in various forms and degrees—see below).

Thus we need a second level, which would be structured in terms of broad socioeconomic or socio-cultural differences between types of *capitalist* economies. Here a range of characterizations is possible.

An obvious sub-category is what are often described as Anglo-American type economies. Examples of these in the Asia-Pacific are the USA, Canada, Australia and New Zealand (even Chile might be included within this category). These economies often include strong commitments to a *laissez faire* ideology and practice, but historically combined with mixed-economy (public and private enterprise) elements. They rely upon a 'liberal' market economy, with free and open contractual arrangements, strong and independent industrial sectors and financial systems, an emphasis on macroeconomic management instead of 'industrial policy', and only moderate state intervention organized under a basic 'hands-off' philosophy.

In contrast to this would be the 'authoritarian capitalist' form, perhaps best exemplified by South Korea, Indonesia, Singapore, Burma/Myanmar, and perhaps Malaysia in the Asia-Pacific. This type of economy is typified by strong involvement of the state either directly in organizing economic activity and regulating it, or in setting the framework of encompassing rules and regulations for the operation of business and commerce. The 'guiding hand' of the state would be important here, often taking a paternalistic and even benign dictatorial form. Dissenting and autonomous sources of economic power are looked upon with great suspicion, and often overtly suppressed (e.g. organized labour). The economic elites are often small cliques involving extended families, ethnic groups and wealthy corporate interests.

A third discernible category could be classified as a 'negotiated economy'. This type of capitalist economy operates in a manner which stresses compromise between the various social strata, interest groups and corporate partners, negotiating to establish and maintain social order and peace. The form of the economic mechanism is to allow and encourage a formal autonomy in the operation of commercial life, though this may in fact be highly organized in terms of oligopolistic, cartel-like and conglomerate business groupings. But within this arrangement the emphasis is on the cautious formation of compromises to establish a consensus between the parties involved in key economic decisions about the allocation and the distribution of resources. Often this is done 'informally', through bargained negotiation (involving business and government agencies, which implicitly co-opt labour into the process). Any overt 'direction' in economic matters is avoided. State ownership of resources is minimal and often frowned upon. The classic example of this type of economy in the Asia-Pacific is represented by Japan, but there are strong elements of it to be found in Taiwan as well (though Taiwan also traditionally has higher levels of public ownership).

A final form of economy that can be considered at this intermediate level might be classified as 'manic' or 'chaotic' capitalism. There are two possible consequences of this form; a very successful variant and a very unsuccessful one. The main features of it are a relatively 'unorganized' form of economic order, which gives it a certain instability and 'transitional' nature. The 'manicness' is instilled into the economy by an emphasis on short-term entrepreneurialism, profiteering and marketeering (which are sometimes combined with corruption). All this can either act to prevent the stabilization of the

economy for long-term prosperity or, perversely, produce quite the opposite and lead to a particularly dynamic outcome. The economy is typified by an extreme form of *laissez-faire* ('hands-off') approach, both in terms of ideology and in the actual practice of the market system. Examples of this in the region would be Hong Kong for the 'dynamic' outcome, and the Philippines for the 'blocked' version. This type may also be emerging in other areas like Thailand, parts of mainland China and Papua New Guinea (PNG). Mexico also demonstrates some of its characteristics.

The point about these classifications is not necessarily to slot any single economy totally into any one or other of them, but to present them as typical forms of socioeconomic organization made 'available' to economic analysis. It may be that a particular country demonstrates predominant features that allow it to be classified accordingly, but very often (and perhaps most frequently) countries will demonstrate a mix or combination of features. They will thus be characterized by hybrid forms. The allocation of countries to examples given above is meant for illustrative purposes only, though it does give something of the flavour of each. Different analysts would no doubt allocate countries differently, or indeed challenge the broad classifications themselves (see Chapter 14 for a somewhat different way of presenting such a typology). Alternatively, they might not find these particular divisions useful and ignore them.

Indeed, to some extent that is what is done in the chapters that follow in the book. There is a variable use made of these kinds of categorizations. In addition, there is yet another level at which some of the chapters conduct their analyses. Chapters 9 and 10, for instance, also discuss specific *national systems* of business and innovation respectively. Whilst using some of the same analytical architecture as just sketched out, these cut across it in various ways according to their own particular objects of interest and analysis. Other chapters concentrate upon specific country examples not mentioned here which add additional, but different empirical weight to the categorizations made above (Chapter 12). Furthermore, Chapter 6 concentrates upon the very specific nature of *business-state* relations in some of the countries already mentioned, which its authors think better illustrates the diverse and perhaps unique pattern in each country. What is more, Chapter 7 argues that a key difference in the outlooks of the 'Asian' and the 'Western' members of APEC in their attitudes towards the process of regional economic integration may have something to do with broad socio-*cultural* differences and 'Asian values'.

But why should we be overly concerned with these issues? What difference does it make if there are very different socio-economic characteristics typifying particular countries? The response is to stress once again that differences matter. They matter to our understanding of the differential performance of the economies in the region, they matter to our understanding of the internal dynamics of its integration, and they matter for an understanding of what is happening to the economies contemporarily and what might happen to them in the future.

In some ways what Japan, the NIEs, the ASEAN countries and China have experienced, and indeed are continuing to experience, can be described as 'pathways from the periphery' (Haggard, 1990). Just as the advanced industrial economies of Europe and North America had done before, these economies—beginning with Japan in the 1950s—have been moving from a position of underdevelopment to one of advanced development. They have been moving from the 'periphery' of the industrial system towards its 'centre', or perhaps more accurately towards its 'centres'. Indeed this

dynamic movement has been creating often very new centres. What we have then is a dynamic evolution-involving different 'pathways'-but a movement not without its disjunctures and interruptions, as we have seen. The key point is that these different pathways are themselves importantly dependent upon the development and evolution of the different socio-economic formations already outlined, and these pathways and their socio-economic form are different ones. There is no single evolutionary path or model-a set of stages—through which all the countries have mechanically passed, or will pass, irrespective of their particular and potentially unique development strategy. Although, as the chapters demonstrate, it is possible to generalize about their experiences at a number of levels and in respect to a number of policy features, every country has 'played' those features or policy moves in a specific way according to their own history and circumstances. Thus whether it is the 'flying geese' model (Chapters 2, 10 and 14), the move from import substitution to export led growth (Chapters, 2, 3 and 6), the common emphasis on the equitable distribution of income (Chapter 11), the general role of the state and the market (Chapters 3, 4, 6 and 14), the growth of population and migration (Chapters, 3, 5 and 12), the commitment to 'open regionalism' (Chapters 7 and 13), all these have been inflected through the specific circumstances of each of the societies and their economies.

But neither should the emphasis placed upon the continued salience of the different and often unique socio-economic formations be exaggerated. These may themselves be undermined and change. Of particular pertinence here is the contemporary dramatic increase in a neo-liberal emphasis on deregulation, liberalization and privatization which seems to be affecting the economies of East Asia just as it has those of the Anglo-American type. Although this is not a completely new development, the present wave of fresh policy moves being adopted in these countries—which is even affecting the remaining communist states as well-is so sustained and widespread that it marks a potential radical turning point. To a large extent this policy shift is driven by a reinvigorated sense of the need for all economies to become 'internationally competitive' under a globalizing economic trend, not just an APEC regionalizing one. Quite how the different economies will react to this trend in the long term remains an open question, but it will undoubtedly affect them all in one way or another (Chapters, 4, 8 and 9). Although this is unlikely to result in a single common socio-economic formation being generated for all the countries in question-a kind of hybrid Anglo-American laissez faire capitalism—could it work to seriously undermine the pertinence of the above classifications? Here another issue arises.

#### **1.3 What of the future for the region?**

Figures 1.1 and 1.2 above provided evidence of the growth and significance of the economies in the Asia-Pacific region up to 1990, particularly those in East Asia. The period between the mid 1960s to 1990 may perhaps prove to be the high point for those economies involved. Since 1990, something of a reaction against viewing them as necessarily being able to unproblematically maintain their dynamic growth rates has set in. This can be judged against the data presented in Table 1.2 and Figure 1.3. The historical record on growth rates is compared in Table 1.2, confirming the picture already

### Introduction 11

presented up to 1990. Since then, over the period between 1990 and 1996, growth has faltered a little in some of the leading economies of the region. This is particularly true of the four 'little dragons'—Hong Kong, Singapore, Taiwan and South Korea. Of course, there still remains a continued growth potential in the following group of ASEAN countries and China (but see below), though they may also eventually experience the loss of dynamic being seen in the first group of industrializers. The possible reasons for this slowdown in growth rates are explored in a number of chapters, particularly Chapter 3 (though this chapter is confident of a continued growth dynamic for the economies).

	1970–79	1980–89	1990–96
Hong Kong	9.2	7.5	5.0
Singapore	9.4	7.2	8.3
Taiwan	10.2	8.1	6.3
South Korea	9.3	8.0	7.7
Malaysia	8.0	5.7	8.8
Thailand	7.3	7.2	8.6
Indonesia	7.8	5.7	7.2
China	7.5	9.3	10.1
Philippines	6.1	1.8	2.8
Rich industrial countries	3.4	2.6	2.0

## Table 1.2 GDP annual average percentagegrowth rates

Source: The Economist, 1 March 1997, p. 23

Another indicator of a possible change in dynamic for these economies can be seen from Figure 1.3, where the growth of merchandise export trade is given for 1995 and 1996. This slowed dramatically for the six main 'tiger' East Asian trading nations, from 14.5 per cent in 1995 to 3.5 per cent in 1996 (their average annual growth in trade between 1990 and 1996 was 10 per cent). This slowdown was one of the reasons behind the serious



\*Hong Kong, South Korea, Malaysia, Singapore, Taiwan, Thailand

## Figure 1.3 World merchandise export trade volume

Source: WTO (1997)

financial crisis that hit the East Asian economies in the second half of 1997 (see Chapter 4). This financial crisis threatened to seriously undermine the growth of some of the following ASEAN economies, and even the region as a whole. North American trade also slowed, though not so dramatically, and Japan experienced a negative growth in merchandise export volume in 1996. Note also how Latin American export trade remained relatively buoyant, perhaps indicating a new dynamic emerging from these countries, many of which also abut the Pacific Ocean. In general there was a world-wide slowdown in merchandise trade growth from the unusually high rate in 1995 of 8.5 per cent to a more normal 4 per cent in 1996.

Clearly, we should be very careful not to read too much into just a single year's figures or even those for a run of a few years. But the trends shown for the mid 1990s should make us cautious in expecting the dynamic identified from the 1960s to 1990 to necessarily continue into the future. It is possible that the early 1990s will be looked back upon as a period of another *disjuncture*—another turning point—for some of these 'early starter' economies. This is all further explored in the chapters below.

Of course, the dynamic of the region as a whole is hardly called into question or exhausted when there are a series of 'second-comer' economies moving up behind the 'early starters', and where the US economy itself seems at last to be emerging from a long period of adjustment and poor economic performances (Chapter 3). The 'optimistic' outcome here concerns the leading role that the Chinese economy may play in the future. Estimates of the world's leading economies in 2020 are shown in Figure 1.4.

It is mainland China that presents the most intriguing case study for the future of the region, as discussed in Chapters 2, 3 and 14. Will China become the largest and most important economy in the world by 2020 as the data in Figure 1.4 suggest? This clearly remains an open question, but the analyses contained in the chapters will undoubtedly contribute to a better understanding of the processes involved in China's growth prospects, as well as the obstacles that stand in the way of such a course (Chapter 14). An interesting feature here is the characterization of the Chinese economy itself, particularly in the context of the categorizations already discussed above. Has it now for all intents and purposes effectively become 'capitalist' or is it still a centrally planned command economy at heart? Both the 'commanding heights' of its economy and its newly emerging manufacturing sectors remain in either public ownership or collectively owned in one way or another (Chapter 14)—though large-scale privatization has increasingly become official policy there as well. Prices are still centrally set or heavily controlled. So how can such an 'administered' economy as that of the Chinese one achieve the startling growth results of the past and those predicted for it in the future? (See Table 1.2 for the historical record of Chinese growth rates.)

One thing that has clearly happened in China is a massive decentralization of economic administration: to regional governments, to municipal administrations and to local urban and rural collectives (Chapter 3). This is particularly so of the very dynamic rural and agricultural



Figure 1.4 Estimates of the world's largest economies 1995–2020 (in US\$billions)

### Source: Central Intelligence Agency (1995)

sector in China. With this enormous variety of arenas and levels of economic governance, the precise nature of the property rights regime in China remains unclear and obscure, but it is certainly not a classical capitalist one, at least as that has come to be understood in the West.

This Chinese experience clearly provides an uncomfortable one for those international organizations and analysts that have insisted on the operation of clear cut capitalist free markets for the successful economic growth in the region (e.g. the World Bank mentioned above). This is again an indicator of the *diversity* of economic organization seen within the region. Part of the fascination with this region thus rests with the possible lessons that can be learned from the variety and diversity of its experiences: for the way we approach not only economic analysis but also for the actual manner in which economic organization has evolved in such a richly varied part of the world. This is something explicitly returned to in the final chapter of this volume.

### 1.4 The chapters in outline

It remains now to outline the rationale and pattern of material covered by the rest of the chapters.

Chapter 2 begins by looking at the broad pattern of economic flows that characterize the relationships between the countries of the region. In addition it provides a preliminary classification of the countries according to their developmental status and resource availability. This serves to establish the grounds for understanding the dynamic of the region's economies and the key interlinkages between them. At the same time the chapter opens up the analysis of the large themes that have typified debate about these economies, some of which were noted above, and it indicates the different development trajectories of the major players in the region. The key economic linkage identified in the chapter is that between the USA and the East Asian economies, the nature and dynamics of which are explored in the chapter.

Chapter 3 follows the preliminary analysis of growth offered in Chapter 2 with a thorough review of the whole debate about why growth happened in this region and what the prospects are for its future. The chapter looks at the basic relationship between savings and investment, and then proceeds to examine how different economic models and approaches have handled the analysis of the growth process in this region. The chapter is optimistic about the continued growth potential in the region despite some approaches which stress the potential slow down in the region's growth as the possibility of generating further factor inputs shrink in the main economies.



The capitalist free market system at work in rural China? A street scene in Kunming

Chapters 4 and 5 build directly on Chapter 3. Chapter 4 examines the sources of one of the main building blocks for growth, that of capital investment. The chapter traces the developments of the individual country financial systems, to see how financial capital was generated during the development process. It looks at the important role that the public authorities played in both generating capital and deploying it in the East Asian economies, and how the state acted as a directing and regulatory body. The differences between financial and productive investment is highlighted and the role of macroeconomic management is examined. There is thus a discussion of monetary policy and exchange rate policy. Finally the chapter asks whether the financial markets in the region are integrating just as the real goods markets seem to be.

Chapter 5 turns to the other key building block for growth, namely labour, but the chapter does this in a particular way by concentrating on the gendered nature of the

growth dynamic of the region. What has been the nature of the relationship between male and female workers in the East Asian miracle? Here it is the inequalities in the labour market that are highlighted, and the subordinate, and often unrecognized role that females have played in economic growth that is focused upon.

Chapter 6 examines one of the most controversial aspects of the debate about the rise of the Asia-Pacific economies. What has been the role of the state in their development? The chapter explores this feature via a very close analysis of the different roles the state has played in the region, emphasizing the key relationships between the world of business and the public authorities. The basic argument is that the traditional state versus market formulations, however sophisticated these may be, are unable to capture the carefully crafted and often uniquely balanced modalities of the role ascribed to the state activity in these economies.

Chapter 7 inaugurates a change of direction for the book by looking at the process of integration within the Asia-Pacific region. This chapter represents a very personal view on the formation and success of the APEC process, written by one of its foremost advocates. The author is the Japanese representative on the 'eminent persons' group of policy advisers who operate to influence the APEC agendas and who guide the process through its negotiating phases (so called 'policy sherpas'). The analysis emphasizes a 'top down' approach to integration, where it is the role of government leaders to propose initiatives and negotiate between themselves over a strongly liberalizing agenda. The APEC process is termed an 'open regionalism', and its similarities and differences to the EU, NAFTA and genuine 'multilateralism' are highlighted.

The next chapter takes up the issue of the integration process driven by private decision-making processes, in particular that of overseas investment. But Chapter 8 is also something of a personal viewpoint. It looks at the role of multinational corporations (MNCs) in the development of the region, but in a very particular light. Other chapters emphasize the importance of foreign direct investment (FDI) in the process of integration and growing interdependence (Chapters 2, 4, 9, 10 and 14). This chapter questions the centrality given to 'overseas investment' in the Asia-Pacific region. Through a close and detailed analysis of the actual flows of foreign investment and foreign production, the chapter questions whether the process of integration has actually developed very far in the region. It is sceptical about the appropriateness of using FDI as a measure of overseas production, and proposes a series of other measures which, it is suggested, demonstrate the so far limited and very uneven extent of internationalized production in the region.

Chapter 9 is more straightforward, though equally challenging. Here it is the nature of the differential forms of business organization that is focused upon. A series of national systems of business are built up as explanatory devices. Contrasting the Anglo-American model of a 'business system' with that of the Japanese, South Korean and the Taiwanese 'Chinese family business', brings out the differences in the organization of production in these economies. The analysis also has important implications for our understanding of how these business systems are adapting and changing as the processes of international and regional integration gather pace.

Chapter 10 compliments Chapter 9 with an investigation into the specific role that technology plays in the process of economic development and regional integration. Rather like Chapter 9, it constructs the notion of a definite national system, but this time of innovation, and contrasts the different national and regional systems of innovation that

are current in the Asia-Pacific. The chapter also compliments the analysis of Chapter 3, since the growth process is one centrally involving technological advance. For Chapter 10 the MNC represents an important and influential conduit through which technological innovation is conducted and spread throughout the region, though the patterns of its operation are themselves dependent upon the 'efficiency' of the national and regional systems into which that technological innovation is placed or from where it arises.

Chapter 11 represents another move in the trajectory of the book, this time to look at the consequences of the growth and integration processes. Who have been the winners and who the losers from these processes? Although the East Asian countries have been marked by their relatively egalitarian policies and outcomes, there are always those who suffer and those who benefit from any process of development. This chapter analyses the nature of the equalities and inequalities to be found in the region, but concentrating on the losers. Other than in Chapter 5, the main focus and sentiment of the book has been to stress the positive nature of the growth and integration processes. The objective here is to rebalance this with an emphasis on the 'downside' of these, though still recognizing the relative equality within which these economies have progressed.

Chapter 12 follows this up with a look at the experience of the smaller territories and states in the region, in particular the Pacific Islands. These are often ignored in taking the wider picture of growth and development, but the pressures they display—population growth, the transformation of traditional production, the pressures on and reorganization of local resources, the migration of labour—all reverberate in the other larger countries as well. It is the impact of all this on the environment that is highlighted in this chapter.

Chapter 13 opens out the Asia-Pacific to beyond its own regional boundaries. It connects up with Chapter 7's analysis, but takes a different approach. The chapter is more sceptical as to the future prospects and past successes of the APEC process, but at the same time is sympathetic to the attempt to generate an adequate governance mechanism for the region as well as for the global economy beyond. The key relationships between the APEC process, the EU and the role of the USA is stressed.

Chapter 14 concludes the book with a review of some of its main themes, but also does some new things and brings new insights to bear on old themes. It stresses the opportunities and dangers inherent in the emergence of a new and potentially unstable regional economy. The chapter looks to the lessons to be learned from the experiences of this regional growth and integration, both analytical lessons and practical organizational ones; and stresses the integrative scope of the regional economy in a rapidly globalizing world.

### References

Buzan, B. (1998) 'The Asia-Pacific: what sort of region in what sort of world?' in McGrew, T. and Brook, C. (eds).

Central Intelligence Agency (1995) Factbook, Washington DC, Central Intelligence Agency.

Haggard, S. (1990) Pathways from the Periphery: The Politics of Growth in Newly Industrializing Countries, Ithaca, Cornell University Press.

McGrew, A. and Brook C. (eds) (1998) *Asia-Pacific in the New World Order*, London, Routledge in association with The Open University. Petri, P.A. (1994) 'The East Asian trading bloc: an analytical history' in Garnaut, R. and Drysdale, P. (eds) *Asia Pacific Regionalism*, Australia, Harper Educational.

Wade, R. (1996) 'Japan, the World Bank, and the art of paradigm maintenance: *The East Asian Miracle* in political perspective', *New Left Review*, no.217 May/June, pp. 3–36.

World Bank (1993) *The Asian Miracle: Economic Growth and Public Policy*, Oxford, Oxford University Press.

World Bank (1997) World Development Indicators, Oxford, Oxford University Press WTO (1997) Overseas Trade Statistics, Geneva, World Trade Organization Secretariat.

# PART 1 Economic Dynamism Considered

### CHAPTER 2 Patterns of trade, investment and migration in the Asia-Pacific region

Partha Gangopadhyay

### 2.1 Introduction

During the last two decades the Asia-Pacific region has outpaced all other regions in terms of economic growth and also assumed paramount importance in the new global economic order. This region registered significant increases in both intra-regional and inter-regional trade and investment flows: more than half of all the world's merchandise exports reach the regional markets. Significant changes have also taken place in the labour markets that have important ramifications in terms of migration. The Asia-Pacific region is also characterized by a great diversity: the region has major industrial giants like the USA and Japan. This region is also the home for rising economic powers like Taiwan and Korea and vibrant developing nations like Indonesia, Thailand, Mexico and Malaysia. Thus, it is important to categorize the nations of this region on the basis of some acceptable criteria. Traditionally there have been three criteria commonly used: level of economic development, rate of economic growth and the trade and payments position. Based on these three criteria we arrive at five groups of nations:

- 1 The industrial giants such as the USA and Japan.
- 2 The land-rich nations such as Australia, Canada and New Zealand.
- 3 The rapidly growing newly industrialized economies such as Hong Kong, South Korea, Singapore and Taiwan (NIEs).
- 4 The near-newly industrialized nations such as China, Thailand, Malaysia and Indonesia.
- 5 The traditional developing nations such as Burma/Myanmar, Vietnam, Cambodia, and Laos.

Different groups have confronted different economic problems during the last twenty years and have diverse resource bases. Thus their trade, investment and growth prospects differ considerably. It is critical to note that the trade, investment and growth prospects are heavily influenced by the present stage of development, the resource bases of a nation and also by national economic policies.

The plan of the chapter is as follows: Section 2 provides a brief summary of the above groups of nations. Section 3 introduces the trade strategies of some of these nations and analyses their government's successes and failures in this context. This will mean reclassifying the nations a little. Section 4 provides a detailed analysis of the major problems of the Asia-Pacific trade: the persistence, spread and consequences of trade imbalance between Japan and the USA, the impacts of such imbalances on external stability and the impacts of Japanese and American patterns of trade on the Asia-Pacific

region. Broadly speaking, this analysis underlies the main structural feature of the whole Asia-Pacific economy—the way Japanese trade surpluses are matched by US trade deficits, and whether this pattern can continue into the future. Section 5 examines the causes behind the recent changes in trade-flows in this region. It highlights three issues to explain such changes: the creation of strong trade intensity, in the emerging 'flying-geese' pattern of industrialization in the region, and the existing economic complementarity between nations. Section 6 provides the general model of development that most of these nations have been emulating: the statist development with an outward-oriented policy. The resultant trading structure and its sustainability is then analysed. A critical element of development in this region is the flow and distribution of investment. A detailed analysis of the sources and distribution of foreign direct investment (FDI) is presented. The impact of economic growth on the labour markets of the region is also examined. Finally, Section 7 concludes the chapter.

### 2.2 A brief summary of the groups

**Group 1:** The USA and Japan are the dominant economies and their economic performance, trade, investment and growth will be a critical factor in determining the overall economic progress of this region. Both these nations have almost exhausted the usual sources of economic growth: shifting resources from agriculture to industry, increasing labour force participation and increasing capital formation through increased domestic savings. At this juncture the major determinants of their economic performance are two-fold. First, a sound short-term macro-economic management is needed to boost the confidence of investors. Such macro-management calls for low interest rates, low inflation and small budget deficits. Secondly, a long term set of policies are needed to reduce structural rigidity and improve factor productivity. For the USA, such macro-economic management entails:

- 1 Change in the domestic savings-investment balance by boosting domestic savings.
- 2 Reduced fiscal deficit—a fiscal deficit is the (negative) gap between government revenue and expenditure.
- 3 Reduction in the demand for increased protectionism.

For Japan the key problems are to reduce structural rigidity in home goods industries and to rationalize the domestic capital market. (The structural rigidities imply restrictions on the flow of resources between industries that inhibit a smooth adjustment in production. Suppose the demand for motor cars goes up while the demand for rice goes down. If labour and capital cannot be easily transferred from the rice sector to the motor car sector, then the current increase in demand for motor cars cannot be met domestically despite the availability of labour and capital. Thus structural rigidity embodies inefficiency in allocating resources across industries.)

**Group 2:** Australia, New Zealand and Canada are industrialized nations but the size of their industrial sectors are smaller than that of Japan, or the USA. More importantly, these nations rely on their abundant natural resources for overall economic performance—they remain mainly primary producers of minerals and agricultural products. All these nations have embarked on deregulation and liberalization of their key

sectors to attain efficiency. These nations also have critical dependence on the international markets on which they trade their primary products, and thus growth of the international economy and openness of the international markets play a key role in affecting their economic performance.

**Group 3:** The third group of countries achieved unprecedented growth of their national incomes during the last three decades. Until the 1960s all these nations were labelled as the backward economies of the Asia-Pacific region. They had very low per capita incomes and low standards of living with a small industrial base. In the 1960s none were regarded as having a high growth potential. But quite contrary to expectations, these nations registered growth rates for nearly three decades in excess of growth in any other nations (see the relevant country profiles in **Eccleston** *et al.*, **1998**). It is this group of nations whose development path has been quite unique in recent history. These nations have not only questioned the existing models of growth but also provided important lessons for other parts of the world.

By 1990 these countries had per capita incomes higher than the average of the World's middle-income countries. As a result, they are often labelled as the 'newly industrialized economies'. The important thing to note is that all these four countries-Hong Kong, South Korea, Singapore and Taiwan, have had an outward-looking development strategy: export growth exceeded the growth of national income in each nation for more than three decades (Eccleston et al., 1998). Except for Singapore, the growth in exports was double that of national incomes. Also the export growths exceeded the growth in investment. In contrast, middle-income nations of other regions experienced a slower growth of exports until 1980, but since then their export growth exceeded the growth in their national incomes. The growth of the NIEs has been so rapid that their consumption growth also exceeded the average consumption growth in the global economy as a whole. Since their population growth rates declined, all these economic changes led to a substantial improvement in the standard of living and quality of life in the NIEs. Gradually, nonagricultural employment rose, which lifted the wage rate. For example, in South Korea, the income distribution became more equal after the 1970s (see Chapter 11). In addition, the growth of imports was as rapid as the growth of exports. Thus an outward orientation contributed to the unprecedented performance of these economies. It is important to note, however, that there are significant differences in these four economies: Singapore and Hong Kong have a very small agrarian sector while the other two have a vibrant agriculture sector. Secondly, private foreign investment was encouraged in Taiwan and Singapore whereas Hong Kong displayed a more neutral attitude towards it. In South Korea, there was an explicit bias against foreign investment until 1980. The bias in South Korea was in favour of the *Chaebol*, their domestic large business firms (see Chapters 6 and 9). The size of the firm in Taiwan and Hong Kong has been of a smaller scale, while Singapore's average firm size lies somewhere in between.

**Group 4:** This group of nations achieved and sustained above-average growth rates for three decades. Yet their achievement failed to reach the outstanding performance of the third group (though China has been expanding particularly rapidly since the mid 1980s). One may call this group a 'follower group' as they try to emulate the economic policy and development strategies of the Group 3 nations. In the mid 1990s these nations accelerated their pace of economic growth. They have now achieved the status of 'almost NIEs'.

**Group 5:** This group of nations still maintain a crucial dependence on agriculture and the industrial sector has yet to reach maturity. Growth rates of per capita incomes are close to the average of the developing world at large, though they could be the major sources of future growth and economic progress for the Asia-Pacific region. So far they have failed to achieve above-average growth rates.

### 2.3 Trade and national economic strategy

#### Government and development

In the 1960s the developing economies of the Asia-Pacific region were caught in the vicious cycle of poverty: they had low per capita income because of low labour productivity. Labour productivity was low due to a very small capital stock per head. Why was the capital stock so small? Because they had little savings to produce investment. Why were savings low? Because they had little income to save. Thus, the vicious cycle of poverty emerges. It was argued that market forces alone could not lift these nations from this vicious cycle of poverty. The 'visible hand' of government intervention was thought to be necessary to beat this cycle (Alam, 1989).

The second crucial idea in the literature on development economics was that agriculture is a backward sector and not capable of improving productivity and providing a surplus output. The industrial sector was given a primacy since it was believed to be an advanced sector. Consequently, the allocation of investment was drawn under the control of the government, and emphasis was placed on the allocation of capital in favour of the industrial sector. The notion was that industrial development, with stagnant agriculture, could bring about economic progress. In most of the developing nations such a bias against agriculture existed, except in the East Asian nations. Agriculture assumed a critical role in these nations (Balassa, 1988).

The third crucial idea of development economics was import-substituting industrialization: growth of domestic industry was encouraged to take place behind protective walls, far from the influence of competitive international markets. Imports were consciously curtailed and most of the developing nations adopted restrictive trade regimes. The East-Asian fast growing economies gradually departed from this inward-looking strategy and adopted an open economy model with critical dependence on both exports and imports (Balassa, 1988; Bhagwati, 1988).

The important point is that economic systems of all nations lie within a spectrum that ranges from pure bureaucratic command systems to atomistic market based competition. The Asia-Pacific nations are no exception. In this region the state wields different degrees of control over the economy. There are three levels of state involvement (Perkins and Roemer, 1991):

- 1 The state played a major role in capital formation—as in Taiwan and Japan—mainly to provide the infrastructure investments (Chapter 4).
- 2 The government also intervened in the production and distribution of certain key commodities and services.
- 3 Governments wielded significant control over the private sectors through:

- (a) Central bank control over commercial bank interest rates (see Chapter 4).
- (b) Quota restrictions on the import of key inputs.
- (c) Some controls existed on the sales and purchase of foreign exchange.
- (d) Discretionary taxes and administered prices were widely prevalent.

For example, in the socialist states of China, North Korea and Vietnam private enterprises enjoy very little autonomy from the visible hand of the government. It might be said that private enterprises are merely a component of the state bureaucracy in these nations. The degree of autonomy of large private enterprises is also low in South Korea, the Philippines, Taiwan and Indonesia. In some of these nations foreign exchange regulations and import restrictions were widely used. In South Korea, for example, in the 1970s, private banks were given directives to carry out investments deemed socially desirable. The government relied on the control of investments to induce private enterprises to take risks that would accelerate the growth of the economy (see Chapter 6).

The economic success of a nation in this kind of system depends on the nature and degree of controls and also on the extent that market forces are distorted by such interventions. In such countries well functioning markets did not already exist in many areas-they had to be stimulated or created, and many markets were regulated. These regulations included price regulation, quantity restrictions and input controls, which led markets to 'fail' in their task of allocating resources. This 'market failure' registered itself in the inability of sellers to supply the goods at internationally competitive, or justifiable, prices. Since the prices did not reflect the true costs, the allocation of resources based on these prices was also distorted. These endemic market failures caused a static misallocation of resources due to badly functioning price mechanisms, for example, in China, Vietnam, North Korea, the Philippines, and Mexico. Pessimism over the ability to export and market failures led to import-substituting industrialization in which the state tried to act as a benevolent allocator of resources to rectify the market failures. Similar economic strategies were chosen by Indonesia, Malaysia, Thailand and the Philippines to protect domestic industries from the forces of international competition (Higgott et al., 1993).

But in the recent years a serious concern has emerged about the interventionist nature of the state in economic matters, which has now been blamed for 'government failure'. Such government failure has been widespread amongst developing nations.

### **Box 2.1 Government failure**

Government failure occurs mainly because of *rent seeking, predatory behaviour* by the state itself and the emergence of *distributional coalitions*. Controls and regulations that accompany state intervention create monopoly profits and rents as entry to such industries is restricted. If there are interventions in markets then there are artificial impediments placed in the way of firms who want to enter these markets. The lack of competition in such markets would restrict outputs and increase prices, which will create the 'monopoly rent', or excess profits. Rent seeking activities embody the efforts of firms to capture these excess profits, or monopoly rent. Rent seeking activities thus entail direct and indirect use of seeking behaviour can be explained in the following way: suppose theft is a profitable opportunity which attracts many otherwise law-abiding people into this activity. As the incidence of theft goes up the government is forced to spend more resources employing more police to combat it. Such resources come from other economic activities which lower the outputs of these activities. First, the main economic effect of rent seeking is, therefore, the reduction in outputs in other sectors. Second, control and

regulations also create predatory behaviour by the state: the level of government intervention is more than the optimum level from the standpoint of the society since the state bureaucrats may get financial gains from restrictions on markets and consequent rent seeking activities among competing firms. Third, economic agents form 'distributional coalitions' to enhance their shares in the national income. Such distributional coalitions adopt pressure tactics and lobbying activities to influence government policy in their favour. All these activities lead to government failure.

Yet the success of the East Asia NIEs was mainly driven by state-led industrialization. Thus a basic question is: Why have governments not failed in the East Asia NIEs? The possible answer lies in the following.

- The economies of the NIEs were also characterized by sector-specific interventions.
- The states adopted export-oriented industrialization.
- The states maintained the importance of the public sector.
- The states avoided succumbing to the pressures of rent-seeking due to institutional arrangements. In North-East Asia the state enjoys relative freedom from predatory activities and coalitional pressures from interest groups.
- A dynamic and centrally controlled elite bureaucratic system manned by well-trained management talent came into existence.
- An authoritarian political system with the political exclusion of the labour movement has been in power.
- A close link between government and business was established.

### Relations between business and government and development

In some of the ASEAN nations and in much of Latin America the state machinery was fragile in the sense that it has been easily exposed to the ravages of rent-seeking, and was beset by an incompetent bureaucracy. But in a shift from the import-substituting industrialization to export-oriented development in the 1960s and 1970s Taiwan, Singapore and South Korea formulated policies without falling prey to rent-seeking activities. The corporate sector formed an alliance with the bureaucracy in which the 'state elites' had a leading role. Distributional coalitions were effectively controlled by authoritarian elite governments. In Taiwan the state intervened through direct investment and by establishing a network of state-owned enterprise. South Korea's approach was built on the Japanese model: planning officials sought to harness the energies of private corporations by guiding them into 'prioritized industries'. Among the incentive schemes were the provision of credit on favourable terms and the provision of infrastructural

support (Chapter 4). This strategy has been called 'industrial targeting strategy'. The states did not succumb to the pressure of interest groups while formulating the targets and operating the strategy (Dornbusch and Park, 1987). Rather the state 'disciplined' the private enterprises to make productive use of the subsidies and to use them for enhancing export performance (see Chapters 3, 4 and 6).

The role of foreign capital in North-East Asia was also restricted, and the emphasis placed upon encouraging domestic enterprises. Multinational companies were not allowed to enter in large numbers to set up direct operations. The multinationals were only granted entry on the condition that their activity would be in harmony with the national priorities. In Latin America and South-East Asia foreign direct investment played a dominant role while the national states exercised little control over them. In contrast, the corporate sector in North-East Asia was dominated by national firms who were under the influence of an elite bureaucracy (Chapter 6).

#### Government and outward-orientation in East Asia

An outward-oriented strategy creates a competitive environment that enhances the incentive to perform. The world economy is also a large market and thus the economies of East Asia could benefit from economies of scale. A typical firm of an East Asian nation can now produce a larger output for the world market and, thereby, reduces the unit cost of production. The forces of international competition also induced the private enterprises to reduce costs and thereby made domestic enterprises more internationally competitive.

While exports played a critical role in these nations, how central was the role of government in their economies? The 'governmental guidance' in Japan and South Korea had been instrumental in allocating resources and managing production decisions. On the other hand, Hong Kong has been more of a *laissez faire* economy where government took a 'hands-off approach. In Singapore significant interventions were applied in the labour and capital markets. In South Korea, however, the government had a very active role. Taiwan was less interventionist than South Korea.

Although in South Korea the government intervention was the greatest, the incentives for exports were uniform and did not lead to inefficiencies or too many sectoral distortions (Dornbusch and Park, 1987). The role of the public sector in overall investment was also limited; public expenditure as a percentage of national income was around 20 per cent, which was one of the lowest among the developing nations. During the 1960s a regulatory environment was introduced into South Korean industry and an active government hand was seen in promoting certain industries. In the early 1970s, the government promoted and guided investment into the heavy industries. But very soon the attempt was considered to be a failure and was reversed. It is necessary to understand, then, that the South Korean government has been quite flexible and since the 1980s liberalization has played an important role for the South Korean economy. Import liberalization started as early as 1960. During the subsequent years tariff rates were lowered and quantitative restrictions on trade were gradually removed. In the 1960s budgetary and financial reforms were introduced and in the early 1980s structural adjustments were undertaken. Thus, the South Korean economy displayed a continual
process of liberalization even though the role of government expenditure was quite limited in comparison with other developing nations.

Many have argued that the major lesson from the East-Asian economies is the contribution of on-going liberalization and removal of controls to their economic progress (Krause and Kim, 1991). At the outset, an agrarian transition was initiated by investing funds to raise agricultural productivity. As industrial growth starts to pick up, it was found essential to gradually lift controls over the industrial sector and expose it to the forces of international competition. Labour markets were largely left unregulated and the exchange rate was kept competitive to give a decisive boost to exports.

Similar patterns have been observed in the USA and Japan in the late 1980s. In Japan tax reforms and financial liberalization took place roughly at the same time. In the USA tax reforms and deregulation also gathered momentum in the 1980s. In the South-East Asian nations—such as Malaysia, Indonesia, Philippines, Thailand—natural resources played a significant role in promoting growth. But other than in Indonesia, these nations rely less and less on primary products and natural resources and more and more on the international economy.

The common theme of economic changes in the Asia-Pacific region is to identify and remove the constraints on efficient resource allocation and growth. Before examining that we look at the major features of trade in the Asia-Pacific region.

#### 2.4 Asia-Pacific trade in the current economic conjuncture

#### Major features of the Asia-Pacific trade

One of the most crucial aspects of the Pacific trade flows has been the persistent commercial imbalance between the USA and Japan (Higgott *et al.*, 1993; Kojima, 1973). For nearly two decades Japan has been the major *creditor* in the international economy. Japan consistently maintains a trade surplus with the USA and the world economy while the USA has had a huge trade deficit.

Second, the Asia-Pacific developing economies have traditionally 'looked East' towards Japan to emulate its economic strategy (Ariff, 1991). The key to this strategy was to maintain a trade surplus and there has been a lateral spread of the commercial imbalance in the region as more and more Asia-Pacific nations ran a trade surplus, especially against the US current account. In the late 1980s and early 1990s most of these economies undertook currency realignments and revaluations to create pronounced movements in their relative prices in favour of positive balances, but despite this the ASEAN countries in particular developed persistent trade



## Japan ties up the Asian market

deficits in the mid 1990s (see Chapter 4). However, the USA became increasingly indebted, which has led to economic conflict with Japan. This is because Japan is the 'immature creditor', with large trade surplus along with a capital surplus, and is also the world's second largest economy. One of the consequences of such huge imbalances is the emergence of an era of potential protectionism.

The GATT/WTO-based political regime for the governance of international trade relations is intended to ensure that domestic interventions by individual economies do not lead to any mutually destructive consequences. The major danger in this context is the fast-emerging 'new protectionism' which takes the form of administered and negotiated nontariff restrictions to trade (Schott, 1989). For example, since the 1950s the USA and other industrial nations have been utilizing voluntary exports restrictions (VER hereafter) for curtailing flows of exports of textiles, steel, and electronic products mainly from Japan and South Korea. Under this scheme an importing country encourages another country to reduce its exports voluntarily under the threat of higher all-round trade restrictions. Such growth of protectionism has been widespread which prompted the following comment: The growth of protectionism appears significant but its consequences do not' (Bhagwati, 1988, p.56). One possible explanation of this conundrum is the gradual emergence and establishment of 'managed trade': the new protectionism seeks to fix bilateral or overall market shares, and to slow down the increase in imports to give the domestic economy time and space to adjust to the changes in international markets. Under 'managed trade', governments aggressively bargain to establish market shares. In this context, VERs and anti-dumping moves have been part and parcel of the regime of 'managed trade'. What are the implications of these features

of trade? One of the major implications is in terms of the external stability of a nation. We address this issue in the next section.

#### External stability and growth in the region

It is a common observation that economies that are insolvent do not grow (Ariff, 1991). Thus, a cautious macroeconomic management is an essential component of the 'East Asian growth model'. Conversely, economies that are experiencing high levels of growth on a sustained basis find it easier to achieve external stability. In other words, these nations find it easier to repay international debts since they have high economic growth. Thus, the key to maintain external stability is to maintain economic growth.

To put this into perspective let us examine the South Korean experience: changes began in 1958 and gathered momentum by 1964. The exchange rate was rationalized and the won was depreciated in real terms. In 1964 the exchange rate was floated in an attempt to maintain its constant real value. The government achieved significant macroeconomic stability in 1965: inflation came under control and interest rate reforms were carried out. By the end of the 1960s quantitative restrictions on imports were withdrawn as exporters were allowed to import any amount of intermediate inputs. From 1967 the government gradually reduced tariffs and also liberalized imports (Krause and Sekiguchi, 1980). However, financial liberalization was slow, but it is continuing in the 1990s. These early measures of liberalization were targeting the trade and payments regime to dismantle the bias against international trade. South Korea's exceptional performance in terms of growth was mainly driven by the growth of exports. Yet in 1975 South Korea retreated back to a controlled environment and adopted the strategy of import-substituting industrialization. The purpose was to establish 'upstream' industries which produce inputs for other industries, such as heavy industry and chemicals. This led to a considerable slow-down in the growth of the economy. By 1981–82 the government reverted back to the outward-looking strategy and the growth rate accelerated. Thus liberalization in South Korea was not a 'once for all policy action'. It has been a disjointed but on-going process that encountered difficulties at times (Dornbusch and Park, 1987).

The second point is that the high ratios of exports to national incomes helped the Asia-Pacific nations avoid a debt crisis even though debt/output ratios for most of these nations were high in the 1990s. For example, South Korea and Indonesia had very high debtoutput ratios, yet their debt-servicing records are good (Sachs, 1985): most of these nations make regular payments of interests on debts and also pay back parts of the principal on borrowed capital. The relevant question therefore is: what determines the external stability of a nation?

There are three critical factors affecting their external stability (Ariff, 1991). First, manufactured export growth contributes to the diversification of exports away from the primary products, such as agricultural products and other natural resources. Such diversifications enhance external stability. Second, in almost all the nations external-orientation has reduced their domestic distortions progressively since the distorted domestic prices have been replaced by international prices. Such a progressive removal of distortions has a positive impact on external stability since the nation becomes more efficient in the use of its resources. Third, macroeconomic instability forces a nation to

adopt short-run stabilizing measures which inhibit market forces. Such impediments to market forces create distortions which adversely affect economic performance, as happened in the USA. Thus the macroeconomic stability and indirect intervention also helped these economies stabilize their external sector.

What is external stability? To put it simply, this can be seen as the gap between domestic savings and domestic investment of a nation such that the nation finds it easy to make necessary repayments without major distortions in resource allocation (see Chapters 3 and 4). To put it differently *imbalance* exists when the balance between the current account and long run capital flows can only be restored with an exchange rate adjustment. The primary question is whether such imbalances exist in this region. There are two possible sources of these: first, the role of Japan may cause such imbalances in this region. We take up this issue in the next section. Second, the USA-Japan trade imbalance may permeate the stability of the region. We examine this further below.

#### The Japan problem in the trans-Pacific trade

Japan may pose a serious problem for Pacific trade. In order to examine this let us look at some of the critical features of the trans-Pacific trade.

Between 1970 and 1990 the Pacific share of world trade went up from 42 per cent to 53 per cent. In the same period intra-regional trade went up from 17 per cent to 26 per cent of the world trade. It is also noteworthy that the APEC markets absorb nearly two-thirds of all exports of the region. The USA and Canada absorb one-third of all exports of the thirteen other APEC nations. The acceleration of economic integration of the region has been impressive, yet there are significant impediments such as restrictions on trade in agricultural products, voluntary export restraints and 'managed trade' by dominant players of the region (see Chapters 7 and 13). In the North-East Asian nations (Japan and South Korea) there exist significant barriers that protect domestically influential groups. Yet the major problem in the Pacific trade is Japan's unusual trading pattern (Higgott *et al*, 1993).

Is Japan a problem? Trading conflicts such as the ban of American-made aluminium baseball bats from the Japanese markets often hit the headlines. It is unequivocally true that there are significant trade barriers to accessing markets in Japan. The Japanese government also admits to these restrictive barriers and attributes them to the Japanese social customs and the labyrinth of internal administrative controls. The major problem is that the Japanese trade pattern has created a massive bilateral trade imbalance *since Japan has more exports than imports:* Japan is the major creditor of the Asia-Pacific region as well as in the global economy.

A major source of this feature is its very low import volume: it is estimated that Japan would import 40 per cent *more* manufactures if its markets were as open as other OECD economies. On the other hand, the USA's low savings ratio and high budget deficits have made the USA highly indebted—it has had to borrow to finance its trade and budget deficits. Thus the trade imbalance between these two super powers has created a significant problem for overall pan-Pacific trade. We return to these issues below. But despite some of its adverse impact on the Asia-Pacific region, Japan plays a positive role in shaping the economic future of this region, and this is addressed in the next section.

The positive contribution of Japan to the region

By the end of the 1980s, Japan turned out to be the world's largest source of foreign direct investment (FDI). Japan has further increased its FDI since the Plaza Agreement in 1985.

#### Box 2.2 The Plaza Accord

During the first half of the 1980s the Reagan Administration tightened fiscal expansion in the USA which resulted in high interest rates and consequent strengthening of the US dollar. This strengthening of the US dollar coincided with the build up of huge deficits in the USA's current account. By 1985 the USA became the largest debtor nation which created the possibility and fear of a dollar crisis. This fear induced the top five industrial nations to co-ordinate the devaluation of the US dollar against their currencies. The Ministers of Finance and the central bankers of these five nations met in the Plaza Hotel in New York in September 1985. It is interesting to note that the yen had been appreciating against the dollar since February 1985. The Plaza Accord in September 1985 further accelerated the pace of this appreciation of the ven. Such developments led to the emergence of Japan as the major creditor in the international economy. The appreciation of the yen reduced the competitiveness of Japanese exporters, and it put pressure on domestic producers to locate their activity abroad where production costs were lower. Thus began the large scale increase in Japanese FDI (Kwan, 1994). FDI represents capital for productive activity rather than for financial investment purposes—which is termed portfolio investment.

It is important to look at who are the beneficiaries of the Japanese FDI. The major destination for Japanese FDI has been North America, which takes a 40 per cent share. On the other hand, less than 20 per cent on average has gone to the developing nations of the Asia-Pacific. However, it is noteworthy that 80 per cent of all Japanese FDI to the developing world goes to the developing nations of the region. Two recent changes can be observed: (1) Japan has increasingly sent FDI to the ASEAN nations and to China; (2) Japan is still the most significant investor in South Korea, Thailand and Indonesia. In Taiwan, Hong Kong and the Philippines it is second to the USA, in Malaysia it is second to Singapore. In order to get a clear picture see Table 2.1.

Period	Hong Kong	Singapore	South Korea	Taiwan	China	Thailand	Malaysia	Indonesia	Philippines
1973– 76	64	146	292	111	0	71	154	550	78
1977– 80	85	467	295	134	1	120	251	843	143
1981– 82	30	323	59	96	8	99	77	476	55
1983– 84	19	342	69	130	22	118	227	268	20
1985– 86	66	198	178	385	46	112	97	93	57
1987	108	268	247	339	30	210	148	295	na
1988	85	179	254	303	203	625	346	298	na
1989	116	678	257	360	206	784	471	167	na
1990	114	270	147	513	161	714	592	536	na
1991	92	240	130	405	251	807	880	576	100
1992	73	210	105	292	500	657	704	610	60

Table 2.1 Japanese FDI in the Asia-Pacific (inUS\$m)

Source: Japanese Ministry of Finance (1992)

The major upsurge in FDI from Japan after 1985 is clear. But the distribution of the FDI provides interesting contrasts: the FDI to Singapore rose significantly until 1983–84, then it declined until 1988. FDI to South Korea and Hong Kong also fluctuated during the 1970s and 1980s but stabilized at a lower level in the late 1980s and early 1990s. The FDI to Taiwan remained quite stable after 1985, while the FDI to China has expanded rapidly. The major recipients of FDI were Thailand and Malaysia, during the early 1990s.

#### Regional implications of the USA's trade imbalance

In the 1980s the US economy emerged as the largest debtor nation. The US trade deficit reflects the inability of US products to compete in the global economy. It also reflects the massive imbalance between savings and investment in the USA—domestic savings being much lower than investment, leading to a potentially unsustainable current account deficit. Thus, the trade deficit has been a symptom of the ailing US economy. In part, this experience of a trade deficit by the USA is caused by restrictions on US products in the global economy.

What is the immediate implication of the USA's trade imbalance? A large US current account deficit allows other nations in the global economy, and particularly those in the

Asia-Pacific region, to follow a strategy of export-led growth and, thereby, export to US markets. The US trade deficit also allows other nations to reduce their own budget deficits since it provides them with buoyant economies. Thus the USA's plight is to the advantage of the other nations of the region. In recent years, however, most of the Asia-Pacific nations have started shifting away from US markets. Table 2.2 documents this shift. The crucial feature is that the trade imbalance between Japan and the NIEs has more than doubled during 1987–92 while the Japan-USA trade imbalance was reduced by 10 per cent in the same period. In addition, between 1987–92 the NIEs-USA trade imbalance declined by a massive 60 per cent.

	Trade volume	Trade imbalance
Japan-NIEs		
1980	26.6	11.8
1987	58.3	20.3
1992	98.8	46.5
Japan-USA		
1980	55.8	6.9
1987	115.1	52.1
1992	148.0	43.7
NIEs-USA		
1980	32.5	3.0
1987	81.2	34.1
1992	110.9	13.9

## Table 2.2 Trade volume and imbalances in theAsia-Pacific region (US\$bn)

Source: Kwan (1994, p. 109)

The linkage between Japan and the USA has weakened, mainly due to the growth in importance of Japan's trade balance with NIEs. The NIEs reduced their dependence on the USA for their export markets and increased their dependence on Japan for the supply of capital goods. Figure 2.1 indicates the shift in importance from the USA to the NIEs for Japan's trade balances.



= indicates the direction of trade surpluses and their relative size

#### Figure 2.1 Trade imbalances (\$USbn)

Source: Table 2.2

#### 2.5 Recent changes and future prospects of the Asia-Pacific trade

#### Changes in trade flows among the Asia-Pacific nations

During the 1980s world exports increased by 35 per cent while the export from the Asia-Pacific region went up by nearly 60 per cent. The centre of this growth included Japan, the NIEs, Thailand and the USA. South Korea, Taiwan and Hong Kong displayed high growth *rates* in trade with most of their partners. Thailand also recorded a good export performance which was very close to that of the NIEs in the late 1980s. There was also an impressive performance by Singapore and Malaysia. The less successful in the growth league were Indonesia and the Philippines, at least until the early 1990s.

Japan and the USA displayed contrasting growth rates: Japan experienced massive growth rates in exports while the USA did so in imports. Japan's exports more than doubled with most of its partners. But its imports were contained within a 7 per cent growth rate. The major sources of imports were from the NIEs, and Thailand. Australia and New Zealand had a more balanced growth in comparison with the USA. Very strong trade intensity indexes (the ratio of trade between partners and overall world trade) were recorded between the following partners: Australia-New Zealand; USA-Canada; Singapore-other ASEAN members; Malaysia-Thailand; Malaysia-Philippines; Singapore-Hong Kong; Hong Kong-Taiwan; Hong Kong-Philippines.

Such a strong trade intensity is mainly driven by the following factors: traditional trade ties, geographical proximity and entrepôt trade relations. In entrepôt trade one nation exports raw materials to another nation which exports back the finished products and may thereby benefit from the cheap labour of the second nation. Japan, South Korea and Taiwan also displayed very high trade intensity as a group. The ASEAN nations, Australia and New Zealand showed strong trade intensity. Canada has high trade

intensity with the USA, yet has very low trade intensity with the rest of the Asia-Pacific. South Korea and Taiwan increased their import intensity from Australia and New Zealand and increased their export intensity with the USA and Japan in the early 1990s. Interestingly, Japan reduced its trade intensity with all the Asia-Pacific nations, except the USA (see Chapter 4).

#### Evolving patterns of comparative advantage

There are two critical factors behind the changes in trade flows in the Asia-Pacific region. First, the trade flows were affected by changes in the comparative advantages of individual economies in this region. Second, the rapid economic growth driven by a sudden industrialization also changed the pattern of trade in this region.

#### **Box 2.3 Comparative advantage**

Consider two trading partners: Japonica and Ameritas. Both could produce two commodities, say plastic chopsticks and wooden toothpicks, which they trade between themselves. But Japonica has a *comparative advantage* in the production of plastic chopsticks while Ameritas has a *comparative advantage* in the production of wooden toothpicks. What does this mean? Basically it costs Japonica less to produce chopsticks *relative* to their cost of production in Ameritas, while it costs less to produce toothpicks in Ameritas *relative* to the cost of production of these in Japonica (costs are measured here in terms of the best alternative opportunity for production *foregone* by the decision to produce either of the commodities in the two economies).

Now, it may be that in *absolute* terms Japonica has a cost advantage over Ameritas in the production of *both* chopsticks and toothpicks, but the relative advantage is with Japonica in chopsticks and with Ameritas in toothpicks, so they specialize in that commodity for which they have the relative or comparative advantage. Thus every country has a relative advantage in some commodity or other, in which it would specialize and trade on the international market. This is the basis for the *mutual advantage* to both nations in trading. It is productivity levels that lie behind the relative differences in the costs of production in the two countries, and this is determined by different types of resources they have available ('factor endowments'), and technological differences between nations.

In simple trade theories, the comparative advantage of a nation is determined by its 'factor endowments'. The availability of cheap inputs determine, among other things, the comparative advantage. The specific stage of industrialization of the particular country also affects the comparative advantage by influencing the available technology. In the process of industrialization, the structure of comparative advantage shifts from simple, labour-intensive products to sophisticated capital and technology-intensive products. Japan has a clear structure of specialization with high comparative advantage of exports in a limited range of industrial products, for example, electronics, metal products, machinery and miscellaneous manufacturered products.

much more diversified. The structure of comparative advantage of Canada, Australia and New Zealand is centred on fewer products which are mainly foodstuffs and resourcebased goods. ASEAN nations have a structure of comparative advantages which is similar to the structure of the resourcerich developed nations of the region. South Korea and Taiwan have high comparative advantages in industrial products and low advantage in primary products.

The 'flying geese' model has sometimes been employed to analyse the changes in comparative advantage of different industries in the East Asia countries during the process of their economic development. Initially the model captured the development of specific industries in a single country. In recent years the model has been extended to incorporate shifts of industries from one nation to another.

#### Box 2.4 The 'flying geese' model

The development of a modern industry in 'latecomer industrialization' typically begins with the import of a product from the advanced nations followed by its import substitution. This subsequently leads to export initiatives, which forms the core of the 'catch up industrialization'. The idea is that each industry passes through five stages—introduction, import substitution, export, mature and reverseimport stages. These stages are represented by the humped shaped pattern for each industry shown in Figure 2.2(a). In the introduction stage domestic production starts with an imitation, or transfer of technology, yet the domestic market is mainly flooded with imports. In the second stage domestic production expands faster than increases in domestic consumption and imports start declining. The industry moves to the next stage when an export surplus is created. In the mature stage, production and exports start declining. Finally, the industry becomes a net importer again when the domestic costs increase. This accounts for change in the comparative advantage for each of the industries.

This pattern of first a low comparative advantage followed by an increase in the comparative advantage, and then a decline in the comparative advantage, is reproduced for each industry in turn (A, B, C, D, etc.) as the process of development progresses, hence the 'flying geese' analogy—each industrial profile looks like the flight formation of a flock of geese. The industries in which a country specializes are progressively more technically advanced and sophisticated, pointing to the nature of the developmental process.

The second part of the Figure, 2.2(b) shows what happens when this model is transferred from a single country to a sequence of countries who are thought to go through much the same cycle as just sketched above for each industry in turn. Another 'flying geese' pattern arises, this time pointing to the way different countries, or groups of countries, follow one another in a sequence of development (1, 2, 3, 4, etc.). This happens for each industry in turn.

Until the mid 1960s Japan's leading export sector was manufacturing industries of high labour intensity, like textiles, and Japan had the status of an almost NIE. Since then Japan's capital intensive machinery sector has been in steady ascendaracy By 1080 Japan reached the industrialized country stage. Starting from a developing country stage, Taiwan and South Korea both reached young NIE status by the late 1960s. In the early 1970s Taiwan reached the mature NIE stage and reached the industrialized country stage in the 1990s, but Hong Kong lagged behind. The ASEAN nations still have a trade structure similar to that of developing nations—though this is rapidly changing. The 1980s increase of Japanese FDI was concentrated on the manufacturing sectors of the ASEAN nations, which could change Malaysia and the Philippines into young NIEs by the turn of the century. It is expected that by the turn of the century Thailand will reach the mature NIE stage.

(This 'flying geese' model is further discussed (and criticized) in Chapter 10.)



#### Figure 2.2 Stages of trade structure

What is the rationale behind the flying geese model? It is due to a shifting competitiveness of an industry over time which is itself caused by changes in factor endowments of nations in the course of economic development. This process is further assisted by technology transfer from more developed to less developed nations.

A second feature of the regional trade patterns is the growing importance of intraindustry trade in the region. Intra-industry trade involves an interchange of products belonging to the same industry, for example cars are being exchanged between Japan and South Korea. Intraindustry trade between two nations is mainly driven by income levels (positively) and income gap (negatively). The index is very high between Japan and the NIEs, and ASEAN and the NIEs. An exception is the index between Hong Kong and Taiwan mainly because the trade here was driven by re-exports to and from China. Since the mid 1980s intra-industry trade between Japan and the East Asian nations has increased mainly due to the rapid industrialization of the East Asian nations.

In search of complementarity among the Asia-Pacific economies

Inter-industry trade between nations is driven by comparative advantage which is determined by the differences in technologies and resources or endowments. The more dissimilar nations are in terms of resources and technologies the more profitable will trade be between them. Thus if two nations have significant differences in terms of technologies and resources, the nations are said to have high complementarity. Thus we focus on the historical interdependencies, and not on the strategic ones (Higgott, Leaver and Ravenhill, 1993). High complimentarity has been observed in the following cases:

Japan vis-à-vis Australia, Canada, Indonesia and Malaysia South Korea and Taiwan vis-à-vis Australia and Malaysia Malaysia and Singapore South Korea and Taiwan and Hong Kong

Low complementarity has been observed between:

ASEAN *and* Canada, Australia and New Zealand *Among* the ASEAN countries Canada *and* Australia and New Zealand

Reductions in complementarities took place in the 1980s between Japan, and South Korea and Taiwan; and between South Korea and Taiwan themselves; and between Canada and Australia. A marked decline in complementarity has also been observed in the trade between Indonesia and Thailand, due mainly to Indonesia's self-sufficiency in rice cultivation.

#### 2.6 Trade, development and investment flows

The model of economic development in the region

The NIEs and nearly NIEs of Asia have tried to emulate the best features of the Japanese experience (Alam, 1989). The key elements of their development strategy were two-fold: first, all these nations adopted outward-oriented trade strategies. Secondly, in many of the East Asian nations the state was actively involved in guiding the process of industrialization by channelling resources into particular sectors. The state protected domestic and newly-emergent industries and also provided subsidies and assistance to

leading firms. Thus in Japan, followed by South Korea and Taiwan, the state provided a disciplined form of administrative guidance to pave the way for industrial development (White, 1989). Similar state intervention is also seen in South-East Asia, yet there the intervention lacked rigorous discipline. The intervention in Thailand has been much less pronounced. In Indonesia the state has been actively and extensively involved in the process of industrialization through regulations and direct investment. One crucial thing to note is that the quality of bureaucratic agency differed across nations. The quality and independence of the bureaucracy was high in North-East Asia, but was substantially less so in the South-East Asian nations.

#### The critical trading-structure of the Asia-Pacific region

As mentioned above, the key trading feature of the region is that Japan is the major supplier while the USA is the major absorber of commodity, and other flows. In turn, Japan supplies capital, intermediate products and technology to the NIEs and the ASEAN nations. These Japanese investments strengthen the supply side of their economies and also make them internationally competitive. The products these nations produce with Japanese capital and technology are then exported primarily to the USA.

As a result, Asia-Pacific developing economies achieved a strong exportled growth due to the supply side boost coming from Japanese export and due to the strong demand from the US market. In recent years the US demand has been declining mainly due to decline in its trade deficit. The US trade deficit is still very large, however, nearly US\$120 billion per year in the mid 1990s, which still provided a large market. In the coming years in order to stabilize the world capital market the US government will need to take measures to cut its budget deficit, increase domestic savings, adjust its exchange rates, and strengthen the supply sides of its economy (Haggard and Kaufman, 1992). This could result in the steady decline of international demand by the USA. In order to maintain the economic tempo of development in the region, Japan would have to absorb the products from the Asia-Pacific region.

Already there are some signs that Japanese imports of finished goods have been steadily rising due mainly to an upward revaluation of the yen, which makes imports cheaper relative to home production. The key issue is whether Japan can effectively replace the USA as the major absorber of exports from this region. The role of Japan as an export absorber of the region will depend on the following factors:

- Whether Japanese fiscal measures are expansionary.
- Whether the budget deficits can be expanded.
- Whether there is a strong growth in private investment to boost Japanese demand for imports.
- Whether its consumption patterns include more imports.
- The extent to which barriers to imports are reduced.

The critical questions are two-fold. First, how quickly will Japanese demand grow? Second, how much of this increased demand will be met from the Asia-Pacific region? Part of the answer will be found in the dismantling of non-tariff barriers to trade, government regulations, the introduction of domestic reforms and the adjustment of domestic price movements.

Another critical feature of the region is the persistent trade surplus of the NIEs: during the last ten years these nations have amassed more than US\$120 billion a year as trade surplus. This massive trade surplus from the region accounts for 50 per cent of the USA's trade deficit.

What are the sources of this trade surplus? First, the massive increase in export earnings by these NIEs. Second, these nations made significant gains due to terms of trade movement: their export prices registered larger increases than the increases in import prices. Third, the appreciation of their real exchange rates also boosted the surplus since the value of their surplus went up measured in terms of their domestic currency. The overall picture for this Asia-Pacific region can be summed up in terms of Figure 2.3.



## Figure 2.3 Trade patterns and payment surpluses

Notes: Deficits on surpluses are measured as SEA against other nations. Arrows show major flows of goods and services. Japan has maintained a consistent current account surplus even in the face of a strong yen. But it is expected that in the future Japanese exports will decrease and imports increase to reduce the current account surplus. In fact, Japan has managed to maintain a steady current account surplus, which implies that the demand for Japanese goods in the world economy remains strong. One possible explanation for this is the large and increasing US federal budget deficit which could have boosted domestic demand in the USA and thus demand for Japanese exports as well. One notable feature is that aggregate spending has been rising faster in Japan than in the USA. The emerging basic structure of trade is thus the following: The North-East Asian nations, due to their poor resource base, rely heavily on oil and raw material imports. In order to pay for these items these nations need to maintain a surplus on their trade in manufactures with other nations. So far the South-East Asian nations have proved to be complementary to the North-East Asian nations: the South-East Asian nations export resource-based primary goods and maintain a negative balance of trade in manufactures with the North-East Asian nations. In recent years, however, these South-East Asian nations have captured some of the labourintensive manufactures and begun to balance their overall trade in manufactures. This implies that in the long run this region must find a market for its manufactures outside this region to maintain a trade surplus in these items.

In order to catch up with the NIEs the South-East Asian nations have consistently imported capital, technology and components from Japan and thus maintained a chronic deficit with Japan. This dependence means that these South-East Asian nations have to earn a surplus from other nations. So far they have sent their products to the USA and Europe and used their surpluses earned there to finance their deficits with Japan. The durability of this pattern depends on the maintenance of their ability to export in the USA and European markets (see also Chapter 4).

Two other interesting cases are South Korea and Taiwan. From the early 1980s South Korea's current trade account began to improve. By 1986 the trade surplus reached 8.4 per cent of the South Korean national income. Exports grew at a rate of 18 per cent per annum, mainly driven by a strong US demand, while imports grew on an average 7.5 per cent per annum. During the 1980s the real exchange (nominal exchange rate minus the inflation rate) also depreciated by 5.6 per cent, while Japan's exchange rate had a massive appreciation of nearly 29 per cent in the 1980s. On the other hand, Taiwan failed to depreciate its exchange rate. As a result, South Korea was steadily gaining ground in the international economy at the expense of Japan and Taiwan. The slow growth of South Korean imports was mainly due to low oil prices in the 1980s. The savings-investment gap considerably improved due to a steep rise in savings, from 23 per cent of GDP in 1980 to 37.3 per cent in 1988, while investment remained stable at 29 per cent of GDP. Taiwan had been running a trade surplus for nearly two decades (over the 1970s and 80s). In 1986 this surplus rose to 22 per cent of the national income. In the mid 1980s its net exports declined while the domestic demand picked up. By 1987 the strong US demand led to a buoyant export market for Taiwan. Taiwan also managed to reduce its savingsinvestment gap, mainly due to a decline in its investment boom.

Thus, at the beginning of the 1990s the overall picture was as follows: For the East Asian NIEs and Japan, the USA had been the largest market for their exports. In the late 1980s-early 1990s, Japan, Taiwan and South Korea as a group accounted for more than 30 per cent of the USA's imports. It seems that the USA will continue to be the dominant

partner of the NIEs and Japan. Yet the budgetary changes in the USA may pose serious challenges for these economies. But there are a number of ways in which the Asia-Pacific region could cope with the deflationary effect of any US budget cuts. First, these nations could enhance their economic integration, and the diversity between them will make increased integration a possibility. Second, intra-industry trade amongst the NIEs will be an important contributor to their future growth. Third, Japan may act as the absorber of demand for this region. We see some signs of this as Japan increases its imports and consumer spending increases at a high rate.

The rise in regionalism is evident in the Asia-Pacific: intra-regional trade rose from 30.9 per cent in 1986 to 43.1 per cent in 1992 while the US market dropped from 34 per cent to 24 per cent over the same period. Intra-regional share of exports of NIEs increased from 31.9 per cent to 43.5 per cent while that of the USA slipped from 37 per cent to 24 per cent. Asia has also emerged as the major market for Japan. For the USA also the Asia-Pacific region has become a major export market. Thus there are signs of the necessary readjustments in regional trade flows. The USA is becoming less important as a destination for exports compared to Japan, the NIEs and the near NIEs.

What are the major determinants of trade flows? There are mainly three sources:

- The partner country's share of world imports which is mainly driven by the competitiveness of its exchange rate.
- The complementarity between nations.
- Bilateral trade barriers.

Trade and regional economic linkages in the Asia-Pacific Rim

In order to place the discussion on economic linkages in the proper perspective consider the data contained in Tables 2.3 and 2.4:

Country	<i>1983</i>	1984	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994
Indonesia	292	222	310	259	385	576	682	964	1482	1774	2004	2109
Thailand	348	400	162	261	182	1082	1650	1700	1847	1969	1493	640
Malaysia	1261	791	695	489	423	719	1846	2958	3998	5183	5206	4348
Philippines	105	9	12	127	307	936	563	530	544	228	763	1126

Table 2.3 Foreign direct investment in South-EastAsia (US\$m) 1983–94

Source: IMF, International Financial Statistics, Various Issues

Indonesia, Malaysia and Thailand in particular have been profoundly affected by the FDI from Japan and the NIEs of the Asia-Pacific region. On the other hand, the Philippines received a very small share of FDI.

The data in Table 2.4 clearly shows that the surge in FDI has come from the East-Asian region, where Japan took the leading role. Yet there is a clear

	Countries receiving investment				
Country making investment	Indonesia	Thailand	Malaysia	Philippines	
Japan	17	44	31	20	
Hong Kong	9	7	4	17	
South Korea	10	2	2	2	
Singapore	4	5	11	3	
Taiwan	3	11	25	19	
NIEs	25	25	42	40	
Japan+NIEs	42	69	73	60	

Table 2.4 Foreign direct investment by country oforigin (as % of total of receiving countries) in thenear NIEs: 1986–89

Source: Asian Development Bank (1991)

picture of uneven investment flow: between 1986–89 44 per cent of Japanese FDI in South-East Asia went to Thailand, while Indonesia received only 17 per cent. During the same period Thailand received 2 per cent of the South Korean FDI in South-East Asia while Indonesia received 10 per cent. The obvious question that follows concerns the reasons behind such a surge: first and foremost, all these South-East Asian nations had low labour costs and hence high profit margins. These high profit margins attracted capital. Secondly, all these economies undertook attractive and serious reforms that granted special deals to foreign investors. After the Plaza Accord, the Japanese Yen appreciated and exports from Japan became less attractive and Japanese investors looked for offshore markets to avoid the high export prices of Japanese goods. The investors of the Asia-Pacific region were also faced with rising protectionism in Europe and in the USA. Thus, all these factors meshed together to create a surge in FDI in the South-East Asian nations, which mainly came from North-East Asia. Box 2.5 indicates the Japanese firm Toyota's strategy of investment in SEA.

## Box 2.5 Integrated international production in the automobile industry: Toyota's network for auto parts in South-East Asia

Exports of motor vehicles from Japan amounted to nearly four million vehicles in 1995. About 32 per cent of these were made by the Toyota Motor Company, whose exports amounted to almost 38 per cent of total Japanese car production. In addition, Toyota's overseas production increased from 152,000 vehicles in 1985 to 1,253,300 in 1995—more than one-third of its total automobile production. In 1995, the number of vehicles produced overseas by Toyota exceeded, for the first time, its exports from Japan. At the end of 1995, Toyota had some 143,000 employees, more than 70,000 outside Japan.

Toyota has established integrated manufacturing systems in all three of its main markets—North America, Europe and Asia. At the end of 1995, Toyota had 35 overseas manufacturing affiliates in 25 countries, more than one-third of which were in Asia. Plants in China, Indonesia, Malaysia,

Philippines, Taiwan Province of China and Thailand produced 370,962 vehicles, nearly one-third of the company's overseas production in 1995.

Although Toyota's vehicle production in the region, as elsewhere, partly results from Asian countries' restrictions on imports of automobiles, the company has responded to the regional industrial co-operation policies of the ASEAN countries by establishing (in consultation with individual governments) a network of affiliates for parts supply to local and regional markets (including Japan). Toyota's intrafirm trade of parts and components in the region is co-ordinated by the Toyota Motor Management Company in Singapore (see Figure 2.4). Toyota exports diesel engines from Thailand, transmissions from the Philippines, steering gears from Malaysia and engines from Indonesia. In 1995, intra-firm exports among these affiliates accounted for about 20 per cent of exports of parts and components of the company's manufacturing affiliates worldwide. Exports of these affiliates geared to other destinations outside the ASEAN market accounted for another 5 per cent.

Toyota also plans to undertake specialized production of various models in its Asian affiliates, both for local sales and exports within and/or outside the region. These include Toyota's all-purpose vehicle (the Kijang) in Indonesia and a compact car in Thailand for possible exports in Asia but also to destinations in South America and the Middle East.



Figure 2.4 Transaction amount of parts and components in 1995: US\$100 million

Source: UNCTAD (1996, p. 100)

#### Lessons from the ASEAN trade

All the ASEAN economies had fluctuating fortunes during the 1980s. In the late 1980s, all these economies experienced an economic upturn. They witnessed an improvement in their current account and fiscal deficits, rationalization of their exchange rate regimes, falling debt service ratios, diversification and growth of exports and diversification of export markets, and an improving investment climate.

It is critical to note that the economic performance of these small open economies is driven mainly by external changes. Since most of them face common changes in the global economy, domestic policy plays a critical role in coping with these changes. Due to delayed responses of Indonesia and Thailand in reforming their real domestic economic sectors, their growth rates declined (MacIntyre and Jayasuriya, 1992). In the long run the policy makers must concentrate on investment and growth to pay back any debts they incur in the short run. In recent years, the ASEAN nations boosted short run growth by adopting counter cyclical policy: during the global recession these nations maintained domestic demand by creating budget deficits. Such budget deficits were mainly financed by borrowing from overseas (except Singapore) which led to a high debt burden for these nations in the early 1990s. Only when the domestic growth declined did these nations adopt short-run demand management policies to tackle the problem. Thus counter cyclical policies may turn out to be costly unless investment and growth pick up in the long run.

#### Resource gaps and investment flow in the Asia-Pacific region

In the 1980s when most of the ASEAN nations faced significant debt problems, the Asian NIEs became net international creditors (lenders). The developing nations of the Asia-Pacific region, by contrast, saw an increase in capital inflow (they became borrowers). Perhaps paradoxically, the developed countries of the region—the USA, Canada, Australia, New Zealand—also became net importers of capital, from Japan and the NIEs. In 1983–90 42 per cent of the Japanese FDI went to the USA and 17 per cent to Europe, 8 per cent to the NIEs, 4 per cent to ASEAN and 3 per cent to Australia.

A major strength for the Asia-Pacific economies has been the flow of FDI from within the region. The FDI involves the direct purchase of the assets of the private sector of a foreign nation by overseas firms who thereby stimulate the foreign economy. By contrast portfolio foreign investment takes place through the stock market of the recipient nation. The overseas investors allocate their capital across different sectors of different nations according to the indicators of the domestic stock markets (Chapter 4). The Japanese have also been major net portfolio investors. The USA was an important direct FDI and portfolio recipient, absorbing US\$50 billion in direct investment and US\$207 billion in portfolio investment between 1985–88. Over the same period the ASEAN nations received a total of US\$6.8 billion in FDI, while China received US\$6.5 billion. South Korea



The USA's FDI in China has been through private companies such as Pizza Hut

became an important investing country, especially in the South-East Asian region. The upshot is that some of the Asia-Pacific nations have become increasingly indebted relative to their national income, yet they have been successful in avoiding any serious debt crisis (Ariff, 1991). The secret is that the Asia-Pacific nations exhibited higher investment ratios over a longer period of time than other debtors—so they have generated enough growth and exports to finance their debt burden. Long term trends in the balance of payments are shown in Table 2.5.

The six stages are the usual pattern for nations to traverse. A country in the young debtor stage relies heavily on overseas funds to finance investment. All the accounts at this stage are negative. A country becomes

Stage	Balance on Trade Account	Current Account	Capital Account	Countries
1 Young debtor	_	-	_	Thailand, Malaysia, Philippines
2 Mature debtor	+	—	-	Indonesia
3 Debt reducer	+	+	-	China
4 Young creditor	+	+	+	Japan, Taiwan
5 Mature creditor	-	+	+	Singapore

Table 2.5 Six stages of balance of payments cycle

6	Asset	_	_	+	USA
	liquidation				

Note: – means deficit on this account. + means surplus on this account. Source: Kwan (1994, p. 19)

a mature debtor mainly driven by the export boom which leads to a surplus in trade. In the next stage further expansion of exports pushes the current account into surplus. A country enters the next stage when it invests significantly in offshore markets. In the next stage the nation becomes a mature creditor when the trade balance turns negative driven by expanding imports. The final stage embodies a huge trade deficit that calls for some asset liquidation (selling of assets). Thus most of East Asian nations borrowed to build the production side of their economies, which positively impacted on their export performance.

#### 2.7 Labour markets and migration

Economic growth in the region has created a relative labour scarcity and consequent improvement in standards of living of workers in some nations. Workers from less advanced nations now tend to migrate to the richer ones. Policy makers have also turned to limited migration as a means to overcome the labour shortage. Thus labour movement between nations has become an important issue. Labour movements, or migration, are mainly driven by two factors: push and pull factors. The push factor exists when migrants make a decision to migrate to avoid poverty and a poor standard of living at home. The pull factor works when the migrants make a move abroad to achieve an improvement in their quality of life. In any migration decision both these factors are present. The crucial message is that the relative standards of living and labour market conditions are the major determinants of the labour flows. In order to examine migration we focus on these twin issues here (see also Chapter 5).

#### Relative poverty and underdevelopment

Despite their experience of massive economic growth many of the Asia-Pacific nations are still beset with poverty and relative deprivation. Japan, the NIEs and two of the ASEAN countries (Thailand and Malaysia) have experienced a sharp reduction in poverty mainly through the indirect 'trickle down effect' of growth and also through the direct programmes of poverty alleviation (see Chapter 11). For example, in the NIEs the incidence of poverty declined by half during the 1970s and today less than 10 per cent of the population are affected by poverty. In the ASEAN nations the incidence of poverty also declined by half between 1960 and 1990, and in the late 1990s less than 20 per cent of the population are affected by poverty. In Indonesia, for example, in 1975 40 per cent of people lived in abject poverty, while this was less than 17 per cent in 1990. With the reduction in the incidence of poverty the standards of living in these nations have significantly gone up. Against this backdrop we still see endemic poverty in the Philippines, Vietnam, Laos, some parts of Indonesia, and in many parts of Latin America.

In these nations on average 30 per cent of the people still suffer from poverty and destitution. Thus, in the Asia-Pacific region there is an economic dualism: there are nations with high to medium standards of living such as the USA, Canada, Australia, New Zealand, Japan, the NIEs, Malaysia and Thailand. On the other hand, there are a large number of nations beset with the problem of low standards of living. Naturally, then, there is a tendency among workers of some of these nations to move internationally in search of a higher standard of living. Thus, there is a large supply of migrant workers in the region. In recent years the labour market conditions have tightened in many of the fast track economies leading to a massive increase in demand for migrant workers. The twin forces of demand and supply have resulted in limited migration in the region. In order to shed light on the actual flows and their ramifications, it is important to consider the labour markets of these nations. How important is this issue? The Asian Development Bank (ADB) recently reported, 'Some expatriate Philippine domestic servants are trained teachers. They earn more in Hong Kong than doctors in Manila' (ADB, 1994). Thus there is a great incentive for people to move across nations for better opportunities in the labour markets. What has caused such changes in the labour markets of the Asia-Pacific nations? We examine this now.

#### The labour markets

There are two polar cases. First, there are nations which have experienced significant tightening of their labour markets: that is, availability of labour relative to demands in such nations is low. Second, there are nations where there is still a surplus of labour. Thus, the migration pattern is from the second group of nations to the first group. Hong Kong's labour force increased in the 1990s at an annual rate of 2.5 per cent, yet unemployment remained low at 1.5 per cent per annum due to high economic growth and historically low women's participation rate. In Hong Kong the participation rate in the labour market has been steadily declining due to increased affluence. In South Korea due to an economic down turn in the mid 1990s, unemployment has climbed to 3 per cent. In Singapore the shortage of labour has become a major concern. The labour market has been tight with less that 3 per cent unemployment. In 1993 the retirement age was increased from 55 to 60 to ease the pressure on the labour market in Singapore. The financial sector and the construction sector have been under enormous pressure from labour shortages. In order to plug the gap, Singapore relies heavily on foreign workers: on average, foreign workers constitute 20 per cent of the total labour force. In Taiwan the unemployment rate was less than 2 per cent during 1990-96, but there is a severe shortage of labour in low paid jobs. In Malaysia the unemployment rate was around 3 per cent despite a healthy growth in the labour force. In order to cope with increased demand for labour Malaysia critically relies on foreign workers. At least half a million out of the 8 million workers are from overseas. In 1993 Malaysia lifted a ban on the entry of foreign workers: the manufacturing sector is free to recruit skilled workers and to hire unskilled workers from the pool of foreign workers already in Malaysia. Despite such a relaxation of immigration rules, there is a large body of illegal immigrants, mainly from Indonesia and Philippines, in Malaysia, which forced Malaysia and Indonesia to come to an agreement in 1993 to control such labour flow. In Thailand unemployment was about 3.2 per cent in the early 1990s which made the labour market very tight. The Labour force

growth there has significantly slowed down due to demographic trends and extension of compulsory education. Despite high unemployment rates in Australia, Canada, New Zealand and the USA, these nations still attract migrants who move there in search of a better quality of life.

By contrast, in Indonesia the unemployment rate is 10 per cent while the incidence of poverty is 17 per cent. The official unemployment rate in China is 5 per cent, yet there is a serious disguised unemployment rate in the state owned enterprises where productivity and wage rates are low. Cambodia is plagued by unemployment and poverty. For Vietnam there is not much published information. There is a significant growth in labour intensive services and small scale and medium scale industries that expand job opportunities. But such an expansion has been offset by shedding of jobs in the civil service and the military. Every year there are about one million new entrants in the labour market. The conservative estimate puts the unemployment rate in excess of 10 per cent in Vietnam. Similar situations can be seen in the Latin American nations. Despite a lot of inertia in the movement, there is a large scale, often illegal, flow of labour from the second group of nations into the first group of scarce labour economies of the Asia-Pacific region.

#### Rationale behind labour shortage

We have already highlighted that NIEs and near NIEs of the Asia-Pacific region experienced tightening of their labour markets. The factors responsible are two-fold:

- 1 Sound labour market policies.
- 2 Spectacular economic growth.

#### Sound labour market policies

The major policy was to restrain the real wage growth rate and allow growth in employment until full employment is reached. Once the economy reached full employment, real wages were allowed to grow. In 1960 the unemployment rate fell to 1.7 per cent in Hong Kong due to restraint on real wages: since real wages were held down, an increase in the inflation rate increased profitability, and hence investment, which boosted the economy and created employment. Due to similar policies, Taiwan's unemployment rate fell to 4.3 per cent in 1963. Until then real wages were sluggish as they grew by only 0.5 per cent per annum. Once full employment was almost achieved, real wages started soaring: in Hong Kong the real wages were increased four-fold within 30 years, seven times in South Korea, eight times in Taiwan, ten times in Japan. The labour market tightening was policy-induced which led to massive increases in real wages and consequent improvements in earning opportunities and higher standards of living. This prompted Bhagwati to write:

These countries appear to have used authoritarian methods to keep tradeunion wage demand under control and to build on this base a successful macro policy of low inflation without which they would likely lapse into repeated over-valuation, or occasional exchange controls, and the attendant inefficiencies of implied import substitution.

(Bhagwati, 1987, p. 100)

The other type of migration derives from the flows of Japanese FDI. The Japanese multinationals have spread quickly in Asia and their primary goal is to transplant Japanese production organization onto Asian soil, mainly by fostering Japanese labour management regimes. The top management comprises mostly Japanese ex-patriots. Thus, there is a spread of migration of high profile managerial talents from Japan to the rest of the Asia-Pacific region (see also Chapter 10).

#### What is the implication?

The major role of migration is to achieve a better mix of labour and capital which normally improves the general welfare of a country. But such migration comes at a potential cost: the most obvious cost is that migration is upsetting the ethnic homogeneity of these nations. This has created unfortunate tensions. Second, in most of these countries the migrants are forced into segmented labour markets, which are characterized by wage discrimination. Finally, migration of professionals has led to a better spread of human capital, albeit with a resentment from many locals.

#### 2.8 Conclusion

Conventional wisdom places a high emphasis on trade and investment policies to improve the standards of living in the Asia-Pacific region. The perception is that the complementarity of resources, highly skilled labour markets and outward looking commercial sectors would act as an engine of economic progress in this region. Most of the East Asian economies have been heavily reliant on the external sector.

The further notable feature of their economies has been the rising degree of economic interdependence during the last few decades. There are two possible explanations. First, the formation and growth of trading blocks, such as ASEAN, APEC, NAFTA and AFTA, have provided the formal effort in integrating these economies (see Chapters 7 and 13). Second, and possibly more importantly, the 'flying geese' paradigm argues that the neighbouring nations are at different levels of economic development and hence have diverse comparative advantages. The diversity of comparative advantages has enhanced the trade and investment flows among them.

In this light, the issues concerning trade and investment have four distinct foci. First, we examined the trade patterns among these nations and unravelled the strengths and weaknesses of the trans-Pacific trade. Second, we addressed the question of sustainability of the trade patterns. Third, we examined the flows of investment to highlight how the comparative advantages are created and exploited by the investment flows in this region. Fourth, we examined the labour flows.

The crux of the trading issues is that both Japan and the NIEs have traditionally maintained a trade surplus with the USA to finance their deficits with the primary commodity exporting nations. These nations in turn rely on Japan and the NIEs to maintain the investment boom and growth, and to export their marketable surplus to the USA and the Western European nations. Such a trading pattern has been characterized as the triangular pattern of trade in which Japan acts as the 'supplier' and the USA acts as the 'absorber', while the NIEs maintain a positive trade surplus with the USA. The persistence of growth and progress of the region depends on the sustainability of such a trading pattern. This in turn depends on the external stability of these economies: whether the USA will continue its large trade deficits which are a product of domestic imbalance between savings and investment. If the USA reduces its budget deficit, most of the economies of the region will face external instability and will need to diversify their export markets. Since the 1985 Plaza Accord, these economies have shown signs of dynamic changes and the ability to diversify: the growth rates of the Asian nations no longer follow that of the USA, while inter-regional trade and investment have assumed critical importance. Exchange rate fluctuations and induced changes in trade and FDI have been the major determinants of growth (see Kwan, 1994). A shift has appeared in East Asian exports from the US market to the other Asia-Pacific markets: the share of intra-regional trade rose from 30.9 per cent in 1986 to 43.1 per cent in 1992. In contrast, the share of exports to the US market declined from 34.1 per cent to 24 per cent in the same period. The FDI from the East Asian nations also played a key role in fostering growth in this region: for example, the FDI from Japan has become the most important determinant of the growth rate in the ASEAN countries.

Thus a critical question is: can the USA continue to act as the major absorber of the region until the other economies can take up the slack? There are signs that the USA is reducing its external deficits, which will have severe repercussions on the growth prospects of the region. Unless Japan, the NIEs and the near-NIEs expand their imports and thereby expand their demands, the region could face a marked slow down in growth.

Quite naturally, there is a significant flow of labour from the less developed nations to more developed nations of the Asia-Pacific region. Such migration is mainly driven by the tightening of the labour markets in the more developed countries in contrast to a surplus of labour in less developed ones. Though such migration is welfare improving, it has caused ethnic tension. Migration in some of the Asia-Pacific nations has also led to segmented labour markets in which foreign workers are consistently discriminated against. The upshot is that the potential of the Asia-Pacific nations is yet to be fully realized, due mainly to expanding opportunities through increased interdependence. Capital is quite mobile in this region leading to increased economic growth for most. The critical problem is that of movement of labour across borders. For capitalists the Asia-Pacific region is a borderless economy, while labour still faces social and ethnic divisions. The success of the region will thus also critically depend on how fast the labour markets will integrate. This remains at best an open question.

#### References

- Alam, M. (1989) Governments and Markets in Economic Development Strategies: Lessons from Korea, Taiwan and Japan, New York, Prager.
- Ariff, M. (1991) *The Pacific Economy: Growth and External Stability*, London, Allen and Unwin. Asian Development Bank (1991) *Asian Development Outlook*, Manila, ADB.
- Asian Development Bank (1994) Asian Outlook, Report, Manila, ADB.

Balassa, B. (1988) 'The lessons of East Asian development: an overview', *Economic Development* and Cultural Change.

- Bhagwati, J. (1987) 'Protectionism: old wine in new bottles' in Salvatore, D. (ed.) *The New Protectionist Threat to World Welfare*, New York, North Holland.
- Bhagwati, J. (1988) 'Export-promoting trade strategy: issues and evidence', *World Bank Research Observer*, vol.3, no.1.
- Dornbusch, R. and Park, Y.C. (1987) 'Korean growth policy', *Brookings Papers on Economic Activity*, Washington DC, Brookings Institute.
- Ecclestone, B., Dawson, M. and McNamara, D. (eds) (1998) *The Asia-Pacific Profile*, London, Routledge in association with The Open University.
- Haggard, S. and Kaufman, R. (1992) The Politics of Economic Adjustment, Princeton, Princeton University Press.
- Higgott, R., Leaver, R. and Ravenhill, J. (1993) *Pacific Economic Relations in the 1990s, Cooperation or Conflict?*, London, Allen and Unwin.
- Japanese Ministry of Finance (1992) Exes Cain Take Gepps, Tokyo, Japanese Government.
- Kojima, K. (1973) 'Reorganisation of North-South trade: Japan's foreign economic policy for the 1970s', *Hitotsubashi Journal of Economics*, vol.13, no.2, pp. 1–28.
- Krause, L. and Sekiguchi, S. (1980) *Economic Interactions in the Pacific*, Washington DC, Brookings Institute.
- Krause, L. and Kim, K. (1991) *Liberalization in the Process of Economic Development*, Berkeley CA, University of California Press.
- Kwan, C.H. (1994) Economic Interdependence in the Asia-Pacific Region, London, Routledge.
- MacIntyre, A. and Jayasuriya, K. (1992) *The Dynamics of Economic Policy Reform in South-East Asia and the Southwest Pacific*, Singapore, Oxford University Press.
- McKibbin, W. and Sachs, J. (1989) 'Implications of policy rules for the world economy' in Bryant, R. et al. (eds) Macroeconomic Policy in an Interdependent World, Washington DC, Brookings Institute.
- Perkins, D. and Roemer, M. (1991) *Economic Systems Reforms in Developing Countries*, HIID Press.
- Sachs, J.D. (1985) 'External debt and macroeconomic management in Latin America and East Asia', *Brookings Paper on Economic Activity, no.2, pp. 523–64, Washington DC, Brookings* Institute.
- Schott, J. (ed.) (1989) Free Trade Areas and US Trade Policy, Washington DC, Institute for International Economics.
- UNCTAD (1996) World Investment Report 1996: Investment, Trade and International Policy Arrangements, New York, United Nations.
- White, G. (1989) Development States in East Asia, London, Macmillan.

### CHAPTER 3 Growth: its sources and consequences

#### Ajit Singh

#### 3.1 Introduction

This chapter, which is in three parts, reviews economic growth, its causes and consequences in the Asia-Pacific countries. The first part (Sections 3.2-3.7) is concerned with the question of fast economic growth in a group of East Asian developing countries. It reviews the extraordinary economic progress of these nations during the last three or four decades and examines alternative explanations for this phenomenon. This analysis is based on my previously published papers (Singh, 1994, 1995a, and 1997b) to which the reader is referred for a further discussion of the issues and for the sources of the data presented.

In reflecting on East Asian economic expansion, the Chinese economy, because of the size of the country's population, deserves special consideration. The second part of this chapter (Section 3.8) is therefore concerned with the fast economic growth in China over the last fifteen years. The Chinese economy expanded at a rate of nearly 10 per cent per annum in the 1980s, a shade above the pace of South Korea. In the 1990s, the Chinese average annual growth rate has been even faster, nearly 2.5 percentage points greater than South Korea's. The important point is that when South Korea grew at nearly 8 per cent a year for fifteen years, it was an extraordinary achievement of which the world took note. But when China, with a billion people, achieved an even faster growth rate, it was not just extraordinary, but a potentially epoch-making event.

The third part of this chapter (Section 3.9) is concerned with a rather different kind of economy on the other side of the Pacific, i.e. that of the world's most advanced industrial country, the USA. During the last two decades, relative to other industrial countries, the US economy has, as we shall see, performed quite well in certain important dimensions. Specifically, its employment record is much better than that of Western Europe—the average annual unemployment rate in the USA in the 1990s has been roughly a third less than that of the European Union countries. On the other hand, productivity growth, and hence growth of real wages,



The new economic zones like Shenzen, close to Hong Kong, have helped fuel China's rapid economic growth

in the USA has been much slower, both with respect to that country's own previous record and in relation to Western Europe and Japan. Real wages of manual workers have not increased in the US economy for the last twenty years. This part of the chapter will briefly examine some of the analytical issues raised by these strengths and weaknesses of the USA's economic record.

For reasons which will become clear in the following discussion, the bulk of this chapter is concerned with the extraordinary drama of East Asian economic growth. The story of the USA, although important in its own right, is not of such historic significance as is the East Asian drama and therefore gets less attention here.

#### 3.2 Fast economic growth in East Asia: an international perspective

It is no exaggeration to say that the economic expansion of a number of countries in East Asia over the last three or four decades is the most successful story of sustained industrialization and economic development in the history of mankind, at least in the narrow sense of rising per capita incomes.

Japan set the example. Recall that Japan in 1950 produced less than 5 million tonnes of crude steel per annum and a little over 30 thousand motor vehicles of all types. The USA's output of steel at that time was nearly 90 million tonnes and it produced about 7 million automobiles per year. By the mid 1970s the Japanese had caught up with the USA in the production of steel and replaced West Germany as the world's largest exporter of cars. By 1980 Japan overtook the USA to become the largest producer of automobiles in the world.

The Japanese experience has by no means been unique. It was self-consciously emulated by countries like South Korea and Taiwan, with results that are perhaps even more spectacular. In 1955, South Korea was unequivocally industrially backward. Its net value of manufacturing output per head was US\$8 compared with US\$7 in India and US\$60 in Mexico. Since then, South Korea has managed to transform itself from being largely an agricultural society to the point where it is the second most important country in the world in electronic memory chip (DRAM) technology (through its firm Samsung). By the year 2000, the country is expected to become the fourth largest car producer in the world. Nothing could be more symptomatic of the changing map of world industry when, reversing historic roles, a hitherto developing country like South Korea becomes a chief foreign direct investor in the heart of the industrial West, i.e., the UK. In 1996 the South Korean giant LG Group decided to install a factory in Wales and invest US\$2.6 billion. This was said to be the largest single investment in the European Union from outside the member states.

To put East Asian economic development in a comparative international perspective, Table 3.1 provides information on overall economic growth over the last three decades for different regions of the world economy, as well as for selected developing and developed countries. In this table the World Bank's definition of 'developing countries' is used, i.e. it is: 'all low and middle income economies'.

The following points which emerge from Table 3.1 deserve particular attention:

1 Over the fifteen years 1980–94, East Asia has been by far the most dynamic region in the world economy. Although the East Asian economies were growing very fast even in the previous fifteen years (1965–80), the gap between their growth rates and those of other developing regions, for example Latin America, was relatively small (7.3 per cent *vs* 6 per cent for Latin America). However, in the 1980s, economic growth collapsed in Latin America (from 6 per cent per annum to 1.7 per cent per annum) while there was an increase in East Asian economic growth to 7.9 per cent per annum.

2 The regional data in Table 3.1 is at a high level of aggregation. For example, the East Asia region of the Pacific comprises many countries, including large countries like China and Indonesia as well as small Pacific island countries such as the Solomon Islands and Tonga. There are, however, only a small number of these countries (but including some of the most populous ones), which have recorded sustained fast growth over the last three to four decades. These are the countries which have captured the world's imagination by virtue of their extraordinary economic success and are, therefore, the main object of analysis of this chapter. Table 3.2 reports the basic data on economic

# Table 3.1 Trends in GDP growth: selecteddeveloping regions, China and India, selectedindustrial countries and the world: 1965–94(average annual percentage growth)

	1965–80	1980–89	1990–94
Low income economies	4.8	2.9	1.4
(excluding China and India)			
China	6.8	10.2	12.9
India	3.6	5.8	3.8
Middle income economies	6.3	2.2	0.2
Latin America	6.0	1.7	3.6
Sub-Saharan Africa	4.2	1.7	0.9
South Asia	3.6	5.7	3.9
East Asia and Pacific	7.3	7.9	9.4
All low and middle income economies	5.9	3.1	1.9
High income economies	3.8	3.2	1.7
USA	2.7	3.0	2.5
Japan	6.6	4.1	1.2
Germany	3.3	2.2	1.1
World	4.1	3.1	1.8

Note: The World Bank defined income groups according to GNP per capita in 1994 as follows: 1 Low income US\$725 or less.

2 Middle income US\$726-US\$8,955.

3 High income US\$8,956 or more.

Source: World Bank (1992, 1996); IMF (1996).

growth for these high performance economies—Hong Kong, Singapore, South Korea, Malaysia, Thailand, Indonesia, and China.

3 It is customary and analytically useful to distinguish between two groups of countries within East Asia—specifically between North-East Asia (Japan, China, North Korea, South Korea, Hong Kong and Taiwan) and South-East Asia (Brunei, Burma/Myanmar, Cambodia, Indonesia, Laos, Malaysia, Philippines, Singapore, Thailand and Vietnam). South Korea and Taiwan, together with Singapore and Hong Kong, are also referred to as the first-tier 'newly industrializing economies' (NIEs). Malaysia, Thailand and Indonesia, where fast economic growth began a little later, are referred to as the second-tier NIEs; in the terminology of Chapter 2, the 'almost NIEs'. As we shall see below, in the discussion in the following sections, the second-tier NIEs have followed different economic policies in some important respects from those

of the firsttier countries. It is significant, as may be observed from Table 3.2, that despite this, during the period 1990–96, the second-tier countries have been just as successful as the first-tier ones (though see Chapter 1).

However, it is also important to note that although the recent economic record of these two groups is indeed similar, detailed data compiled over a longer period reveals that there is an appreciable performance gap between the groups. The exact result will depend on which periods and which countries have been considered, but over the last three decades or so the annual per capita GDP growth rates of Japan and the first-tier NIEs have, on average, been roughly 2 percentage points higher than those of the second-tier NIEs. The cumulative impact of this growth gap over 30 years is significant. For example, Malaysia's per capita income in 1961 was almost three times that of South Korea and almost twice that of Taiwan (Malaysia then included Singapore, so purely 'Malaysian' income would have been somewhat lower). It remained higher than South Korea, and about one-third that of Taiwan (UNCTAD, 1996).

4 The North-East and the South-East Asian NIEs have not only had an excellent record of long-term economic growth, but have also had much lower inflation rates than countries in other developing regions, particularly Latin America. World Bank (1996) statistics show that the average annual inflation rate of Asian-Pacific countries during 1980–93 was 7.1 per cent compared with the developing country average of 72.8 per cent, and 245 per cent for Latin American and Caribbean countries.

	1970-79	1980-89	1990-96			
Hong Kong	9.2	7.5	5.0			
Singapore	9.4	7.2	8.3			
Taiwan	10.2	8.1	6.3			
South Korea	9.3	8.0	7.7			
Malaysia	8.0	5.7	8.8			
Thailand	7.3	7.2	8.6			
Indonesia	7.8	5.7	7.2			
China	7.5	9.3	10.1			

## Table 3.2 Fast growing East Asian NIEs. GDPgrowth: annual average percentage, 1970–96

Source: The Economist (1 March 1997, p. 23)

'The East Asian countries' excellent record of economic growth and relative price stability has certainly translated into impressive increases in the average standards of living, reductions in poverty, increasing real wages and rising employment:

• The International Labour Organisation (ILO) provides evidence that during the 1980s, in fast-growing East Asia, the former labour surplus economies such as Taiwan, South

Korea, Singapore and Malaysia found themselves confronted with labour shortages. As a result, there was a significant immigration of labour into these countries from neighbouring low-income nations. In these dynamic economies manufacturing employment rose at a rate of over 6 per cent per annum during this decade whilst, at the same time, real earnings increased at an average rate of 5 per cent per annum.

- With respect to poverty, available evidence for the 1980s for individual fast-growing NIEs suggests sizeable reductions in its incidence. Thus in China the incidence of absolute poverty fell from 28 per cent of the population in 1980 to 10 per cent in 1990; in Indonesia the corresponding reduction was from 29 per cent to 15 per cent; in the Republic of South Korea from 10 per cent to 5 per cent, and in Malaysia from 9 per cent to 2 per cent (see Chapter 11).
- A remarkable feature of the development of the East Asian NIEs during the relevant period has been that not only has the rate of growth been very high, but income distribution has become more—rather than less—equal. However, this conclusion that inequality has declined may hold in relation to incomes, but not necessarily for wealth (see Chapter 11).

The above highly positive record of the East Asian NIEs stands in striking contrast to that of Latin America and Sub-Saharan Africa in recent years. For example, with regard to Latin America, the ILO reports that during the 1980s, there was a steady fall in modern sector employment, with paid employment falling at a rate of 0.1 per cent per annum. This reversed the trend of the previous three decades when steady economic growth had led to a significant expansion of modern-sector employment. In most Latin American countries, the average real wage fell during the 1980s, recovering in only a few countries towards the end of the decade. Minimum wages fell on average by 24 per cent in real terms across the region, while average earnings in the informal sector declined by 42 per cent.

#### 3.3 Why did North-East Asian NIEs grow so fast?

The central analytical and policy question raised by the East Asian NIEs economic experience is of course, what are the causes of the fast economic growth in these countries?

There is no agreement on this question. Indeed, there is a continuing controversy in which the main protagonists are the World Bank with some orthodox economists on one side, and a number of academic economists, not all of whom are heterodox, on the other. This debate is important for two reasons. First, the World Bank professes to base its policy recommendations for countries around the globe on what it regards as the lessons to be drawn from the experience of these highly successful East Asian countries. Second, from an analytical point of view, the debate is clearly of central importance, precisely because of the fast growth of these economies over a sustained period. Thus, the resolution of this debate would inevitably have an important bearing on the economists' general ideas on growth and development.

With the publication of the World Bank's important study, *The East Asian Miracle*, in 1993 there has been some useful narrowing of differences between the two sides.

However, there remains a wide gulf on a range of significant issues as will become clear from Box 3.1.

#### Box 3.1 Two seminal World Bank studies:

#### 'The Challenge of Development' and 'The East Asian Miracle'

The World Bank's studies, The Challenge of Development (1991) and The East Asian Miracle (1993) are seminal works which provide a comprehensive account of the bank's economists' thinking on development problems and their conclusions on public policy. The Challenge of Development is important because, in the words of the then President of the World Bank, Barber Conable, it 'synthesizes and interprets the lessons of 40 years of development experience' by the bank's economists. The significance of the 1993 study, The East Asian Miracle, lies in the fact that the Bank economists invariably justify their policy advice to developing countries around the world by reference to the experience of the sustained fast growth of the East Asian economies. However, the two studies complement each other and need to be studied together. The first provides the Bank's general analytical framework and its broad market-oriented approach to development issues. The second, whilst acknowledging heavy government intervention in East Asia, argues nevertheless that the experience of these countries is still compatible with the 1991 Report's recommendation of a market-friendly approach, and therefore does not necessitate any significant departures in the Bank's policy advice. This argument is, however, highly controversial as will become evident in the course of this chapter.

#### The total factor productivity (TFP) approach to economic growth

In view of the fact, as suggested above, that the World Bank's views on the subject have wide practical policy significance for developing countries, it will be useful to start this discussion with a careful examination of the bank's economists' analyses of East Asian economic growth. The theoretical foundation of these analyses is the so-called TFP approach to economic growth. In this approach, which is based on the growth accounting framework of conventional economics, economic growth is decomposed into three components: the first due to the growth of labour input, the second due to the growth of capital input, and the third due to an increase in the productivity of both capital and labour. The latter is referred to as the growth in the TFP of labour and capital. This can, in theory, arise from a variety of factors, including, importantly, technical progress, economies of scale, and an improvement in the quality of the inputs.

This framework was initially used by economists for estimating empirically the sources of economic growth for advanced economies. It was typically found that only a small part of a country's economic growth can be explained in terms of increases in the supply of labour or capital inputs, and that most of the remainder (for example, as much as 80 per cent for the US economy) is due to an improvement in the productivity per unit of labour and capital. It is, however, important to note for our later discussion that in such empirical exercises, the total factor productivity of capital and labour is not measured

directly. It is instead estimated indirectly as the 'residual' growth; it is the difference between the actual growth rate, and that due to the expansion of the labour and the capital inputs separately. This residual is then broadly attributed (often by qualitative analysis) to the factors of the kind mentioned earlier. Since it is difficult to measure technical progress, economists have great difficulty in saying precisely, in quantitative terms, what proportion of TFP growth is, for example, due to technical progress, and what proportion is due to an improvement in the quality of labour and capital. In view of this, the TFP or the 'residual' growth rate is sometimes referred to as a 'coefficient of ignorance', in that economists have not yet found a way of ascribing it exactly to the various factors which may be involved. The precise way the TFP approach works in a technical sense is shown in Box 3.2.

Box 3.2 Sources of economic growth, the production function approach

The production function links production (Y) to inputs of capital (K), labour (L) and technical progress (A).

Thus

Y=Af(K, L)

This specific production function, called the Cobb-Douglas production function after the names of Professors Cobb and Douglas who widely used it in the 1930s, takes the following form:

 $Y = AK^{\alpha} - L^{(1-\alpha)}$ 

(2)

(1)

where, the exponent a lies between 0 and 1.

The Cobb-Douglas production function has the property of constant returns to scale, that is, a given proportionate increase in capital and labour respectively leads to the same proportionate increase in output. From the point of view of economic theory, this function also has the property that if the factors K and L are paid their marginal products as they would be under perfect competition,  $\alpha$  will be the share of capital in total output and  $(1-\alpha)$  that of labour. Moreover, the payment of factors according to this scheme will be equal to the total product. This result follows from manipulation of the production function as (2).

Further manipulation of equation (2) involves taking logs on both sides and deriving the following expression;

 $Log Y = Log A + \alpha Log K + (1-\alpha) Log L$ 

(3)

differentiating with respect to time, (denoted by *t*) we obtain,  $1/Y \cdot dY/dt = 1/A \cdot dA/dt + \alpha \cdot 1/K \cdot dK/dt + (1-\alpha) 1/L \cdot dL/dt$ 

(4)

This may be expressed as;

$$\dot{Y} = \dot{A} + \alpha \dot{K} + (1 - \alpha) \dot{L}$$
<sup>(5)</sup>

Where,  $\dot{Y}$ ,  $\dot{A}$ ,  $\dot{K}$ ,  $\dot{L}$  denote the instantaneous proportionate growth rates of the respective variables.

Equation (5) is the basis of the decomposition of the growth of output into that of the growth of capital, that due to the growth of the labour force and that due to technical progress, given here by,

$$A = dA/A \cdot 1/dt$$

The equation implies that output will increase, on account of technical progress even if there is no increase in the inputs K and L.

The total factor productivity of growth is given by,

 $\dot{A} = \dot{Y} - \alpha \dot{K} - (1 - \alpha) \dot{L}$ <sup>(7)</sup>

(6)

i.e. it is a measure of the total growth of output which cannot be accounted for by growth due to the combined effect of the growth of either the labour input or the capital input. The combined effect is measured as a weighted average of the inputs of labour and capital together where the weight (a) of the capital input is the share of capital in total output, and the corresponding weight of labour  $(1-\alpha)$  is the share of labour in total output.

This approach to sources of economic growth has serious limitations. Briefly, first, it is an entirely 'supply side' analysis of economic growth—it does not take any account of 'demand'. Second, it assumes full employment and full utilization of resources, so that, equation (5) suggests that output will increase if there is an increase in labour cost. The proposition is however only realistic if the increased labour force were all able to find jobs and were not unemployed, hence the demand side is important. This theory also ignores the fact that capital and labour may not contribute to economic growth in the simple additive fashion employed by equation (5). There may be interactions between capital, labour and technical progress which may suggest, for example, that employment can only increase if there is an increase in the labour force, or that technical progress can only take place if there is an increase in the supply of capital goods. The TFP approach ignores all such interactions. For a lucid discussion of the subject, the reader should consult Nelson (1981).

Be that as it may, the World Bank economists' basic thesis is that economic growth is determined essentially by the growth of TFP. Those countries which have a higher TFP growth will also have a higher overall economic growth and vice versa. Bank economists further assert that changes in TFP are determined mainly by economic policy—the degree

of openness of an economy, the extent of competition in the domestic product and factor markets, and investment in physical and human capital (education), particularly the latter. The underlying chain of causation is that competition and education promote technical progress, and therefore TFP growth, and hence economic expansion. 'Free mobility of people, capital, and technology' and 'free entry and exit of firms' are regarded as being particularly conducive to the spread of knowledge and technical change.

Comparing East Asia with other regions shows that, unadjusted for quality, its rate of growth of labour input has not been greater than that elsewhere. It has however had a much faster growth of capital input. The latter is reflected in the comparative savings-investment records of a group of nine Asian and nine Latin American countries given in Tables 3.3 and 3.4. The comparison with Latin America is particularly interesting because in the 1950s and 1960s, it was the Latin American countries which saved more than the Asian countries. However, in the 1990s, the median Asian economy saved and invested nearly 30 per cent of GDP compared with a figure of about 20 per cent for the Latin American countries. It is notable that between 1955 and 1965, the average domestic savings in South Korea were only 3.3 per cent of GDP; this compares with a figure of over 35 per cent for the 1990s. The data show that of the six East Asian economies listed in Tables 3.3 and 3.4, five of them (the Philippines being an exception) have been able to attain exceptionally high savings and investment rates over the last fifteen years, and some (for example, South Korea) over a longer period.

surings us a percentage of ODT )								
	1955–65	1965–73	1973–80	1980–89	1990–94			
Asia								
China	_	$28.7^{1}$	_	35.0 <sup>2</sup>	39.3			
India	_	17.9	22.3	21.5	20.8			
Indonesia	_	13.7	24.6	27.6	31.2			
South Korea	3.3	21.5	26.4	32.8	35.1			
Malaysia	23.3	21.6	29.4	29.3	29.5			
Pakistan	11.8	_	10.9	14.8	17.6			
Philippines	10.9	20.6	24.3	17.6	18.1			
Sri Lanka	13.5	14.6	13.5	15.6	12.7			
Thailand	16.7	22.6	21.5	22.6	34.1			
Median	12.7	20.6	23.3	22.6	29.5			
Latin America								
Argentina	18.9	19.7	21.2	11.2	17.1			
Bolivia	8.1	29.2	18.2	2.0	6.3			

Table 3.3 Domestic savings in Asian and Latin American countries 1955–94 (gross national savings as a percentage of GDP)
Brazil	16.0	24.3	21.7	19.7	20.4
Chile	10.8	12.9	12.2	9.7	24.4
Colombia	18.8	17.2	19.2	17.4	21.3
Ecuador	15.4	16.3	21.2	16.6	19.4
Mexico	14.6	19.9	21.3	21.3	17.2
Peru	20.9	27.2	24.9	22.0	18.9
Venezuela	32.0	30.0	34.5	23.6	19.8
Median	16.0	19.9	21.3	17.4	19.8

Notes: <sup>1</sup>Figure refers only to 1970

<sup>2</sup> Figure refers only to 1980

Sources: World Bank (1991); United Nations (1966); Asian Development Bank (1995); ECLA (1995).

Parenthetically, some readers may be puzzled by the fact that there is a discrepancy between saving and investment figures in Tables 3.3 and 3.4. Why, for example, was the South Korean domestic savings rate between 1955–65 only 3.3 per cent of GDP (Table 3.3), while the corresponding investment rate was 14.3 per cent (Table 3.4). How does one explain the difference since *ex-post* savings should be equal to investment by the national income accounting identity? Strictly speaking, such an identity holds only for a closed economy. In an open economy the accounting identity takes the following form:

Y = C + I + (X - M)

(8)

(9)

where *Y* is national income

*C* is consumption expenditure *I* is investment expenditure *X* and *M* are expenditures on exports and imports respectively. Readjusting equation (8) gives: Y-C=I+(X-M)

Given that Y–C=savings (S), substituting in equation (9) gives: S=I+(X-M)

(10)

Therefore:

$$I-S=M-X$$

(11)

From equation (11), if investment is greater than savings (I>S), this implies that the country is running a current account deficit on its trade account (M>X), which is being financed by net capital inflows on the

	$\mathbf{ODI}$ )				
	1955–65	1965–73	1973–80	1980–89	1990–94
Asia					
China	_	$28.5^{1}$	_	35.0 <sup>2</sup>	36.8
India	_	18.4	22.6	23.9	23.9
Indonesia	_	15.8	24.5	30.4	30.2
South Korea	14.3	25.1	31.8	31.2	36.7
Malaysia	14.2	22.3	28.7	32.2	34.6
Pakistan	-	16.0	16.5	18.8	19.7
Philippines	11.4	20.6	29.1	21.7	23.1
Sri Lanka	14.0	15.8	20.6	25.8	23.0
Thailand	17.6	23.8	26.6	26.7	40.7
Median	14.2	19.5	25.5	26.7	29.0
Latin America					
Argentina	19.7	19.8	21.8	15.5	16.7
Bolivia	16.7	25.4	24.9	12.2	15.4
Brazil	16.4	26.1	26.2	21.5	20.5
Chile	11.3	14.4	17.4	18.1	25.3
Colombia	18.9	18.9	18.8	20.4	18.3
Ecuador	14.8	19.0	26.7	23.2	20.6
Mexico	15.1	21.4	25.2	23.1	22.4
Peru	22.4	27.7	28.9	26.2	21.5
Venezuela	22.0	29.3	32.6	22.0	16.3
Median	16.7	21.4	25.2	21.5	20.5

Table 3.4 Investment performance of Asian andLatin American economies between 1955 and1994 (gross domestic investment as percentage ofGDP)

Notes: <sup>1</sup> Figure refers only to 1970

<sup>2</sup> Figure refers only to 1980

Source: World Bank (1996)

capital account through, for example, foreign aid, private investment or other capital inflows.

The answer to the South Korean puzzle above is therefore straightforward: during the earlier period 1955–65 South Korean domestic savings were being supplemented by a

large injection of foreign capital inflows to bridge the current account deficit implied by the excess of investment over savings in that period. These inflows came mainly from the USA in the form of economic and military aid. As the South Korean economy became more developed and richer, more and more of the country's total investment was financed by domestic savings. Indeed, by the 1980s South Korean investment as a proportion of GDP was smaller than domestic savings (see the South Korean investment and savings figures in Tables 3.3 and 3.4). What this suggests is that in the period 1980–89, South Korea was running a current account surplus and was therefore most likely a net investor abroad.

The empirical analysis reported by the World Bank economists in *The East Asian Miracle* shows that the high rates of investment in East Asian countries have made an important contribution to their overall fast economic growth. However, this analysis also suggests that investments in these countries were more efficiently utilized, and hence were more productive than elsewhere. The study's estimates of the TFP growth rates indicate that these were considerably higher in the 'Miracle' countries than in other developing economies.

### Market-friendly strategy of development and economic openness

As pointed out earlier, the next step in the World Bank economists' analysis is to suggest that these comparatively high growth rates of TFP of East Asian countries were largely due to superior economic policies which have been followed in these countries. Specifically, the economists call attention to two related aspects of economic policy. First, they argue that these countries implemented a so-called 'market-friendly' strategy of development. 'Market-friendly' is a vague term which can mean all things to all people and can also be a mere tautology. However, to their credit, the bank's economists defined the concept fairly precisely in *The Challenge of Development* (see Box 3.1). Government interventions are regarded as being market-friendly if they meet the following criteria:

- 1 *Intervene reluctantly*. Let markets work unless it is demonstrably better to step in...[It] is usually a mistake for the state to carry out physical production, or to protect the domestic production of a good that can be imported more cheaply and whose local production offers few spillover benefits. (Spillover benefits are indirect benefits to the economy as a whole which arise from the local production of the goods whose imports are being restricted. Such benefit may include the stimulation of production of related goods, training of work force, etc.)
- 2 *Apply checks and balances*. Put interventions continually to the discipline of international and domestic markets.
- 3 *Intervene openly*. Make interventions simple, transparent and subject to rules rather than official discretion.

Overall, the state's role in economic development in this 'market-friendly' strategy is regarded as being at best limited to providing the social, legal and economic infrastructure, and to creating a suitable climate for private enterprise to flourish. In other words, the role of a 'market-friendly' government in this conception is essentially that of

'a night watchman' (i.e. a government which only provides the broad framework for the private sector to operate in, but does not actually interfere directly in its activities).

The second related set of policies to which the World Bank economists ascribe the superiority of the TFP growth of the East Asian economies is their greater openness and close integration with the world economy. Together with vigorous competition in the domestic markets, it is suggested that openness to international competition and foreign direct investment have forced East Asian corporations to be efficient; it has also enabled them to reap the full benefits of foreign technology.

### 3.4 The TFP approach: alternative estimates and interpretations.

The foregoing analysis of the World Bank's economists has been subject to serious criticisms both empirically and theoretically. A number of academic economists have produced empirical estimates of TFP growth in East Asia which contradict the World Bank's findings. In an important contribution, the US economist Alwyn Young (1994) has presented extremely interesting international evidence on TFP growth for a sample of over one hundred countries. Young's study shows that over the period 1960-85 the four leading East Asian NIEs (Taiwan, Hong-Kong, Singapore and South Korea) were among the top five in the world league in terms of growth of output per capita. However, as Table 3.5 indicates, in terms of TFP growth, which Young calculated for the same group of over 100 countries but over a somewhat shorter time period 1970-85, the NIEs performance was no longer so spectacular. (Table 3.5, for reasons of space, presents data only for the top 66 countries rather than the full sample of 118.) The table indicates that in the TFP league, Hong Kong ranked sixth, Taiwan twenty-first, South Korea twentyfourth and Singapore sixty-third. The table shows that Bangladesh, a poor developing country with a large population and low growth prospects, ranked higher than either Taiwan or South Korea in terms of TFP growth. Bangladesh was ninety-fifth on Young's list of 118 countries in relation to per capita growth of GDP.

Other studies suggest that, for somewhat different periods than above, South Korea and Taiwan had almost zero TFP growth. In other words, in terms of the TFP methodology, most, if not all, of the extraordinarily high economic growth of many East Asian countries can simply be explained by the fast expansion of factor inputs, including *inter alia* capital inputs arising from very high rates of capital accumulation (Rodrik, 1995). In that sense, it is suggested that there is no miracle about East Asia—it is basically a question of high rates of investment.

However, an interesting interpretation of these new empirical findings on TFP growth in East Asia is provided by another US economist Paul Krugman (1994). He argues, on the basis of these results, that the high growth rates of the 'East Asian Miracle' economies are not sustainable. This, in his view, is for the following reasons. It is unrealistic to expect that countries which are already investing 35–40 per cent of their GDP will be able to raise their rate of investment much higher still. Krugman goes on to point out that these countries similarly already have highly educated and high quality labour forces, which limits the scope for further improvement in these spheres as well. In these circumstances, without technical progress, eventual decreasing returns to investment will set in and limit the growth potential of these economies.

1         Egypt         0.035         23         Guinea         0.014         45         Turkey         0.008           2         Pakistan         0.030         24         South Korea         0.014         46         Netherlands         0.008           3         Botswana         0.029         25         Iran         0.014         47         Ethiopia         0.007           4         Congo         0.028         26         Burma         0.014         48         Austria         0.007           5         Malta         0.025         27         Mauritius         0.013         49         Australia         0.006           7         Syria         0.025         29         Denmark         0.013         51         Kenya         0.006           8         Zimbabwe         0.024         30         Israel         0.012         52         France         0.004           10         Tunisia         0.024         31         Greece         0.012         55         Honduras         0.004           11         Cameroon         0.024         33         Luxembourg         0.011         56         Portugal         0.004           13         U			-		•				
3Botswana0.02925Iran0.01447Ethiopia0.0074Congo0.02826Burma0.01448Austria0.0075Malta0.02627Mauritius0.01349Australia0.0076Hong Kong0.02528China0.01350Spain0.0067Syria0.02529Denmark0.01351Kenya0.0068Zimbabwe0.02430Israel0.01252France0.0039Gabon0.02431Greece0.01253Liberia0.00410Tunisia0.02433Luxembourg0.01154Paraguay0.00411Cameroon0.02433Luxembourg0.01155Honduras0.00412Lesotho0.02135Tanzania0.01157USA0.00413Uganda0.01937Sweden0.01059Canada0.00314Cyprus0.01937Sweden0.01059Canada0.00315Ihailand0.01839Malawi0.01061Algeria0.00316Bangladesh0.01839Malawi0.01061Ichia, f.Rep.0.00317Iceland0.01840Brazil0.01061Ichia, f.Rep.0.00318Italy0.01840Brazil<	1	Egypt	0.035	23	Guinea	0.014	45	Turkey	0.008
4         Congo         0.028         26         Burma         0.014         48         Austria         0.007           5         Malta         0.026         27         Mauritius         0.013         49         Australia         0.007           6         Hong Kong         0.025         28         China         0.013         50         Spain         0.006           7         Syria         0.025         29         Denmark         0.013         51         Kenya         0.006           8         Zimbabwe         0.024         30         Israel         0.012         52         France         0.004           10         Tunisia         0.024         31         Greece         0.012         53         Liberia         0.004           11         Cameroon         0.024         32         Japan         0.012         55         Honduras         0.004           12         Lesotho         0.021         34         Yugoslavia         0.011         56         Portugal         0.004           13         Uganda         0.021         35         Tanzania         0.011         58         Belgium         0.004           14         Cyprus<	2	Pakistan	0.030	24	South Korea	0.014	46	Netherlands	0.008
5         Malta         0.026         27         Mauritius         0.013         49         Australia         0.007           6         Hong Kong         0.025         28         China         0.013         50         Spain         0.006           7         Syria         0.025         29         Denmark         0.013         51         Kenya         0.006           8         Zimbabwe         0.024         30         Israel         0.012         52         France         0.005           9         Gabon         0.024         31         Greece         0.012         53         Liberia         0.004           10         Tunisia         0.024         32         Japan         0.012         54         Paraguay         0.004           11         Cameroon         0.024         33         Luxembourg         0.012         55         Honduras         0.004           12         Lesotho         0.021         36         Colombia         0.011         56         Portugal         0.004           13         Uganda         0.021         36         Colombia         0.011         58         Belgium         0.004           14         Cyp	3	Botswana	0.029	25	Iran	0.014	47	Ethiopia	0.007
6Hong Kong0.02528China0.01350Spain0.0067Syria0.02529Denmark0.01351Kenya0.0068Zimbabwe0.02430Israel0.01252France0.0059Gabon0.02431Greece0.01253Liberia0.00410Tunisia0.02432Japan0.01254Paraguay0.00411Cameroon0.02433Luxembourg0.01255Honduras0.00412Lesotho0.02134Yugoslavia0.01156Portugal0.00413Uganda0.02135Tanzania0.01157USA0.00414Cyprus0.02136Colombia0.01158Belgium0.00315Hailand0.01937Sweden0.01059Canada0.00316Bangladesh0.01839Malawia0.01061Cent. Af. Rep.0.00218Italy0.01840Brazil0.01062India0.00119Norway0.01741Panama0.00963Singapore0.00110Finland0.01542UK0.00964Sri Lanka0.011	4	Congo	0.028	26	Burma	0.014	48	Austria	0.007
7         Syria         0.025         29         Denmark         0.013         51         Kenya         0.006           8         Zimbabwe         0.024         30         Israel         0.012         52         France         0.005           9         Gabon         0.024         31         Greece         0.012         53         Liberia         0.004           10         Tunisia         0.024         32         Japan         0.012         54         Paraguay         0.004           11         Cameroon         0.024         33         Luxembourg         0.012         55         Honduras         0.004           12         Lesotho         0.021         34         Yugoslavia         0.011         56         Portugal         0.004           13         Uganda         0.021         35         Tanzania         0.011         57         USA         0.004           14         Cyprus         0.021         36         Colombia         0.011         58         Belgium         0.004           15         Thailand         0.019         37         Sweden         0.010         59         Canada         0.003           16         Bang	5	Malta	0.026	27	Mauritius	0.013	49	Australia	0.007
8         Zimbabwe         0.024         30         Israel         0.012         52         France         0.005           9         Gabon         0.024         31         Greece         0.012         53         Liberia         0.004           10         Tunisia         0.024         32         Japan         0.012         54         Paraguay         0.004           11         Cameroon         0.024         33         Luxembourg         0.012         55         Honduras         0.004           12         Lesotho         0.022         34         Yugoslavia         0.011         56         Portugal         0.004           13         Uganda         0.021         35         Tanzania         0.011         57         USA         0.004           14         Cyprus         0.021         36         Colombia         0.011         58         Belgium         0.004           15         Thailand         0.019         37         Sweden         0.010         59         Canada         0.003           16         Bangladesh         0.018         39         Malawi         0.010         61         Cent. Af. Rep.         0.002           18	6	Hong Kong	0.025	28	China	0.013	50	Spain	0.006
9Gabon0.02431Greece0.01253Liberia0.00410Tunisia0.02432Japan0.01254Paraguay0.00411Cameroon0.02433Luxembourg0.01255Honduras0.00412Lesotho0.02234Yugoslavia0.01156Portugal0.00413Uganda0.02135Tanzania0.01157USA0.00414Cyprus0.02136Colombia0.01158Belgium0.00415Thailand0.01937Sweden0.01059Canada0.00316Bangladesh0.01938Malaysia0.01060Algeria0.00117Iceland0.01840Brazil0.01062India0.00118Italy0.01741Panama0.00963Singapore0.00119Finland0.01542UK0.00964Sri Lanka0.001	7	Syria	0.025	29	Denmark	0.013	51	Kenya	0.006
10Tunisia0.02432Japan0.01254Paraguay0.00411Cameroon0.02433Luxembourg0.01255Honduras0.00412Lesotho0.02234Yugoslavia0.01156Portugal0.00413Uganda0.02135Tanzania0.01157USA0.00414Cyprus0.02136Colombia0.01158Belgium0.00415Thailand0.01937Sweden0.01059Canada0.00316Bangladesh0.01839Malawia0.01060Algeria0.00218Italy0.01840Brazil0.01063Singapore0.00119Norway0.01741Panama0.00964Sri Lanka0.001	8	Zimbabwe	0.024	30	Israel	0.012	52	France	0.005
11       Cameroon       0.024       33       Luxembourg       0.012       55       Honduras       0.004         12       Lesotho       0.022       34       Yugoslavia       0.011       56       Portugal       0.004         13       Uganda       0.021       35       Tanzania       0.011       57       USA       0.004         14       Cyprus       0.021       36       Colombia       0.011       58       Belgium       0.004         15       Thailand       0.019       37       Sweden       0.010       59       Canada       0.003         16       Bangladesh       0.019       38       Malaysia       0.010       60       Algeria       0.002         18       Italy       0.018       40       Brazil       0.010       62       India       0.001         19       Norway       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	9	Gabon	0.024	31	Greece	0.012	53	Liberia	0.004
12       Lesotho       0.022       34       Yugoslavia       0.011       56       Portugal       0.004         13       Uganda       0.021       35       Tanzania       0.011       57       USA       0.004         14       Cyprus       0.021       36       Colombia       0.011       58       Belgium       0.004         15       Thailand       0.019       37       Sweden       0.010       59       Canada       0.003         16       Bangladesh       0.019       38       Malaysia       0.010       60       Algeria       0.003         17       Iceland       0.018       39       Malawi       0.010       61       Cent. Af. Rep.       0.002         18       Italy       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	10	Tunisia	0.024	32	Japan	0.012	54	Paraguay	0.004
13       Uganda       0.021       35       Tanzania       0.011       57       USA       0.004         14       Cyprus       0.021       36       Colombia       0.011       58       Belgium       0.004         15       Thailand       0.019       37       Sweden       0.010       59       Canada       0.003         16       Bangladesh       0.019       38       Malaysia       0.010       60       Algeria       0.003         17       Iceland       0.018       39       Malawi       0.010       61       Cent. Af. Rep.       0.002         18       Italy       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	11	Cameroon	0.024	33	Luxembourg	0.012	55	Honduras	0.004
14Cyprus0.02136Colombia0.01158Belgium0.00415Thailand0.01937Sweden0.01059Canada0.00316Bangladesh0.01938Malaysia0.01060Algeria0.00317Iceland0.01839Malawi0.01061Cent. Af. Rep.0.00218Italy0.01840Brazil0.01062India0.00119Norway0.01741Panama0.00963Singapore0.00120Finland0.01542UK0.00964Sri Lanka0.001	12	Lesotho	0.022	34	Yugoslavia	0.011	56	Portugal	0.004
15       Thailand       0.019       37       Sweden       0.010       59       Canada       0.003         16       Bangladesh       0.019       38       Malaysia       0.010       60       Algeria       0.003         17       Iceland       0.018       39       Malawi       0.010       61       Cent. Af. Rep.       0.002         18       Italy       0.018       40       Brazil       0.010       62       India       0.001         19       Norway       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	13	Uganda	0.021	35	Tanzania	0.011	57	USA	0.004
16       Bangladesh       0.019       38       Malaysia       0.010       60       Algeria       0.003         17       Iceland       0.018       39       Malawi       0.010       61       Cent. Af. Rep.       0.002         18       Italy       0.018       40       Brazil       0.010       62       India       0.001         19       Norway       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	14	Cyprus	0.021	36	Colombia	0.011	58	Belgium	0.004
17       Iceland       0.018       39       Malawi       0.010       61       Cent. Af. Rep.       0.002         18       Italy       0.018       40       Brazil       0.010       62       India       0.001         19       Norway       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	15	Thailand	0.019	37	Sweden	0.010	59	Canada	0.003
18       Italy       0.018       40       Brazil       0.010       62       India       0.001         19       Norway       0.017       41       Panama       0.009       63       Singapore       0.001         20       Finland       0.015       42       UK       0.009       64       Sri Lanka       0.001	16	Bangladesh	0.019	38	Malaysia	0.010	60	Algeria	0.003
19         Norway         0.017         41         Panama         0.009         63         Singapore         0.001           20         Finland         0.015         42         UK         0.009         64         Sri Lanka         0.001	17	Iceland	0.018	39	Malawi	0.010	61	Cent. Af. Rep.	0.002
20 Finland         0.015         42         UK         0.009         64         Sri Lanka         0.001	18	Italy	0.018	40	Brazil	0.010	62	India	0.001
	19	Norway	0.017	41	Panama	0.009	63	Singapore	0.001
<b>21 Taiwan 0.015</b> 43 W.Germany 0.009 <b>65 Fiji 0.001</b>	20	Finland	0.015	42	UK	0.009	64	Sri Lanka	0.001
	21	Taiwan	0.015	43	W.Germany	0.009	65	Fiji	0.001
<b>22 Ecuador 0.014</b> 44 Mali 0.008 66 Switzerland 0.000	22	Ecuador	0.014	44	Mali	0.008	66	Switzerland	0.000

## Table 3.5 Annual growth of 'total factorproductivity' (1970–85)

Note: Asia-Pacific countries are printed in bold type Source: Young (1994, p 970)

Krugman's interpretation can however be challenged on a number of grounds. First, it has been pointed out that even within the confines of the growth accounting framework (employed by Krugman himself), there is considerable scope for improving the quality of labour input in East Asia. Educational levels in many East Asian countries are still considerably lower than those in advanced countries. Therefore, it will be a long time before decreasing returns of the kind Krugman is referring to set in.

Second, however, a more powerful critique of the Krugman interpretation is provided by those economists who do not accept the growth accounting framework and instead put forward an alternative non-neo-classical approach to economic growth. These economists suggest that the effects of technical progress cannot be separated from those of the expansion of capital input. This is because it is argued that technical progress can only take place through the introduction of new machines, i.e. through an increase in capital inputs. Even replacement investment is associated, in this view, with technical progress, because when an old machine is replaced by a new one the latter is likely to be technologically more advanced and not simply a new copy of the old one. Therefore, in this analysis, there is no reason for decreasing returns to occur since the higher the rate of investment, the greater would be the turnover of machines and the greater would be the technical progress. This in turn would also lead to greater learning by doing and, through a process of cumulative causation, should result in a virtuous circle of greater competitiveness and faster economic growth. According to non-neo-classical analysis, therefore, the high growth rates of the exemplar East Asian countries were mostly, if not entirely, due to their very high rates of capital accumulation, which embodied technological progress.

In conclusion, returning to the TFP analysis, it is important to observe that even if the measured TFP for a country like South Korea is zero, it certainly does not mean that the country has literally made no 'technical progress' in the common usage of the term. This would clearly be an absurd conclusion for a country which has, within a short period of 30 years, progressed from largely exporting agricultural and textiles products to exporting motor cars and advanced computer chips. All that the zero TFP result tells us is that the country's economic growth is entirely due to the increased use of inputs, rather than due to a greater efficiency in their use of inputs. Equating greater efficiency in this sense with 'technical progress' is significant and meaningful only within the framework of the TFP model. To re-iterate, one could argue that the TFP approach does not properly measure the contribution of technology to economic growth since much technology is embodied in capital goods, and this fact is not reflected in the simple model given in equation 5 in Box 3.2. Moreover, as noted in Box 3.2 on TFP earlier, critics are also right to suggest that the TFP approach overstates the contribution of labour relative to capital by ignoring the fact that an increased supply of labour by itself does not lead to more growth of output, unless there was also growth of capital stock. As the economist Joan Robinson pointed out earlier this century, capital is not malleable in the real world and therefore a given amount of capital cannot be stretched to provide productive employment to every size of the labour force.

### 3.5 Did East Asian NIEs follow a 'market-friendly' strategy?

Apart from the TFP analysis, other aspects of the World Bank's theses concerning East Asian development have also been subject to stringent criticism. The bank's economists' claim, that East Asian countries (including Japan during 1950 to 1973, when it was more like a developing country but also enjoyed rapid growth) followed a market-friendly strategy, has met widespread scepticism. The World Bank's critics have raised a number of questions: Did these countries follow the Bank's 'market-friendly' prescriptions in the precise sense outlined earlier? Did their governments intervene in the markets 'reluctantly'? Did they, for example, leave the prices and production priorities to be determined by the market forces and simply provide the necessary infrastructure for private enterprise to flourish? How 'transparent' was the government intervention in industry? To achieve their colossal economic success, how closely did these countries integrate with the world economy?

Unless otherwise specified, the following discussion, for reasons of space, is confined to the two leading East Asian 'tiger' economies of South Korea and Taiwan, as well as Japan during the period 1950 to 1973. There is overwhelming evidence to show that their governments did not intervene either (a) reluctantly or (b) transparently in any of these economies (see Chapters 4 and 6). Specifically, during their periods of fast economic growth, the governments in all three reference countries used a wide array of interventionist instruments including many of the following:

- import controls;
- control over foreign exchange allocations;
- provision of subsidized credit, often at negative real interest rates, to favoured firms and industries;
- control over multinational investment and foreign equity ownership;
- heavy subsidization and 'coercion' of exports, particularly in South Korea;
- a highly active state technology policy;
- restrictions on domestic competition and government encouragement of a variety of cartel arrangements in the product markets;
- promotion of conglomerate enterprises through mergers and other government measures (again particularly in South Korea);
- wide use of 'administrative guidance', indicating non-transparency of government interventions (see, for example, Amsden, 1989 and Wade, 1990).

In other words, the governments in Japan, South Korea and Taiwan did all the things which the 'market-friendly' strategy for development is not supposed to do. Above all, all three countries followed an 'industrial strategy'—a set of policies to deliberately change the market prices and production priorities—which is explicitly ruled out by this approach. The World Bank economists acknowledge that there was significant state intervention in each of these countries, but argue that 'these economies refute the case for thorough-going *dirigisme* (state intervention in the economy) as convincingly as they refute the case for *laissez-faire*'. Critical economists agree that the experience of these countries is certainly an argument against *laissez-faire;* nor does it provide any support for 'command' planning of production of the Soviet-type, which in effect supplants the market altogether. However, for mixed economy developing countries with effective governments, these economists suggest that the post-war East Asian economic history is unequivocally an argument for adopting an industrial strategy, for guiding the market, and not following the hands-off 'market-friendly' approach recommended by the World Bank.

### 3.6 'Openness' and East Asian economic development

There is a great deal of evidence which also does not support the World Bank claim that these exemplar East Asian countries either sought or implemented a close integration with the world economy. Consider the following:

- As a result of its explosive economic growth at a rate of nearly 10 per cent per annum during the 1950s and 60s, by the end of that period Japan had graduated to the status of an OECD country, i.e. it had become a member of the developed country club. Unlike developing countries which under GATT were able to provide infant industry protection, OECD member countries were obliged to abolish such restrictions against freer trade. (Infant industry protection refers to restrictions on imports of products which a developing country has just begun to produce domestically. Since, compared with the older and larger competing firms from abroad, the domestic industry is likely to be at a lower or an 'infant' level of development, it will be at a competitive disadvantage in relation to those foreign firms. In these circumstances orthodox theory acknowledges the case for protection for a relatively limited period of time. This is to permit the infant industry to 'grow up' so as to be able to compete on more equal terms with foreign rivals.) Nevertheless, as late as 1979, manufactured imports amounted to only 2.4 per cent of the Japanese GDP; the corresponding percentages in the UK and other countries of the EEC was five to six times larger. Even in the USA, which traditionally because of its continental size has a relatively closed economy, the volume of imported manufactured goods in the late 1970s was proportionally almost twice as large as in Japan. Clearly, during the 1960s and 1970s (and even more so in the 1950s) the Japanese economy operated under a regime of draconian import controls, whether practised formally or informally.
- South Korea, during the last decade, has become a major producer and exporter of cars. It is expected to become the fourth largest car producer in the world by the year 2000. Even now, it has sizeable exports to the USA and Western Europe. And yet, in 1995, South Korea still imported only 4,000 cars from abroad. Essentially, the South Korean government has heavily protected its car industry for the last 30 years.
- Instead of welcoming foreign direct investment, the fact is that both Japan and South Korea (but not Taiwan) discouraged it, particularly during their respective periods of fast industrialization. It is not that the Japanese and South Korean governments were averse to obtaining technology from abroad, quite the contrary. Rather, these governments evidently took the view that it was cheaper and more conducive for national development to 'import' foreign technology through means other than FDI, e.g. licensing (Chapter 10).
- A useful measure of 'price distortion' for an economy is the extent to which its relative domestic prices differ from the international relative prices for the same products (see Chapter 2). On this measure, the estimates of which are presented by the World Bank economists in *The East Asian Miracle* itself, it turns out that Japan, South Korea and Taiwan were among the most distorted economies. Relative domestic prices conformed less to international relative prices in these countries than in Brazil, India, Mexico, and Venezuela. Most of the last mentioned countries are often held up by the

Bretton Woods institutions as prime examples of countries which do not 'get the prices right'.

### 3.7 East Asian experience: alternative perspectives

The analysis of Sections 3.4 to 3.6 has indicated that the various World Bank theses on fast economic growth in East Asia can be argued to be deeply flawed, both empirically, as well as analytically. How can this phenomenon then be explained?

### Government-business interactions, the financial system and successful coordination

There is a fair degree of agreement that the key to fast economic growth in East Asia lies in the very high rates of savings and investments attained by these economies. Savings and investments are the subjects of a separate chapter in this book (Chapter 4), where these issues will be discussed in detail. However, for the sake of completeness of the argument, it is necessary here to note that:

- 1 these high savings and investments were carried out largely by the private sector, particularly in Japan and South Korea;
- 2 they were not a spontaneous outcome of the working of the free market, but were policy induced.

Research shows that the close relationship between the government and business and their interactions, together with the 'long-termist' qualities of the financial system existing in these countries, played a key role in this process (see Chapter 4). For example, in Japan the government not only provided fiscal incentives to the corporate sector, but also implemented protection, had lax enforcement of anti-trust laws (designed to prevent excessive concentration of industry by a small number of firms) as well as practised financial repression (interest rates kept deliberately low, particularly for firms and industries favoured by government industrial policy). All this led to greater rents and profits for the private corporate sector than would otherwise have been the case. However, the government also ensured through other policies that these greater corporate profits were not simply to be consumed or paid as dividends to shareholders, but were in fact reinvested. The government's role was also crucial in raising and maintaining at a high level the corporate propensity to invest, primarily by addressing the problem of coordination failures which are ubiquitous in the real world of incomplete and imperfect markets. The US economist, Dani Rodrick (1994), particularly emphasizes the role of the government in this sphere in ensuring high rates of investment in the East Asian economies. The co-ordination problem was extremely important during Japan's high growth phase, as what in effect the Ministry of International Trade and Industry (MITI) did in that period was to orchestrate investment and technological races among oligopolistic firms in favoured industries. Such races were carefully controlled as, otherwise, excess capacity might have been created which would have adversely affected the future corporate inducement to invest.

In South Korea, the role of the government was even more pronounced in all these spheres. Here the government was not just a co-ordinator of investment decisions, but in fact a co-partner with the private sector. Research shows how government policies and government business interactions resulted in extraordinarily fast upgrading of the industrial structure of that country. The government's complete control over the financial system for much of this period was pivotal in this process.

### The optimal degree of openness

Turning to the question of openness, the Indian economist, the late Professor Chakravarty, and the present author have suggested in Chakravarty and Singh (1988) that the fast growing East Asian countries did not seek close integration with the world economy (as the World Bank economists would have us believe) but implemented, rather, what may be called 'strategic integration'. In other words, Japan, South Korea and Taiwan were open to the international economy only up to the point that it was in their interest to be so in order to maximize national economic growth.

Chakravarty and Singh (1988) argue that 'openness' is a multidimensional concept: apart from trade, a country can be 'open' or not so open with respect to financial and capital markets, in relation to technology, science, culture, education, and inward and outward migration. Moreover, a country can choose to be open in some directions (say trade) but not so open in others (such as foreign direct investment or financial markets). Their analysis suggests that there is no unique optimum form or degree of openness which holds true for all countries at all times.

Indeed, as orthodox economic analysis now recognizes, in the real world of economies of scale, learning by doing and imperfect competition, even free trade is not necessarily optimal for a country. In the Chakravarty and Singh analysis, a number of factors affect the desirable degree and nature of openness: the world economic configuration, the past history of the economy, its state of development, among others. The timing and sequence of opening are also critical. They point out that there may be serious irreversible losses if the wrong kind of openness is attempted or the timing and sequence are incorrect.

The East Asian experience of 'strategic' rather than 'close' integration with the world economy is only comprehensible within this kind of theoretical framework. Thus, in terms of these concepts, countries like Japan and South Korea chose to be open with respect to exports and closed in relation to imports. Similarly, they were open with respect to the interchange of scientific and technical knowledge, but not so open with respect to foreign direct investment. They were also, for much of their fast growth periods, not open to free international capital flows.

It is also useful to consider the experience of the second-tier NIEs—Indonesia, Malaysia, Thailand—within the Chakravarty and Singh framework. As noted earlier, in these South-East Asian economies, foreign direct investment has played a far more important role than it did in Japan or South Korea. One interpretation of this phenomenon is that as a consequence of the fast development of the East Asian countries, the secondtier NIEs are faced with a different historical situation, which makes the optimal degree of openness different for these countries. In this new situation, it is advantageous for the South-East Asian NIEs to attract industries which are no longer economic in the first-tier countries because of the growth of their real wages—as suggested by the so-called 'flying geese' model of Asian economic development (see Chapters 2 and 10).

### 3.8 China: the plan and the market and fast economic growth

As indicated earlier, the extraordinary economic growth achieved by China during the last fifteen years is an epoch-making event. How has such fast growth come about? (The following section is based on Singh, 1996b.)

It is tempting to say, and is often asserted by orthodox economists, that the Chinese experience shows the virtues of the free market in unleashing entrepreneurship and rapid wealth creation. Such a story may meet the ideological predilections of the orthodox economists, as well as the international financial institutions, but is unfortunately greatly at variance with the facts. Although the Chinese Communist Party over the last two decades has progressively introduced markets and allowed a modest degree of private enterprise, China is very far from being a free market economy:

- 1 China does not have nationally integrated product markets compared, for example, with a country like India. Apart from the relatively poorer transport structure, Chinese provinces and municipal authorities are prone to impose restrictions on free movement of goods to protect local industry. This leads to imperfections and to the fragmentation of product markets.
- 2 Although China is attempting to establish capital markets, at the present stage of development these can only be described as embryonic. Investment allocation is essentially done by government controlled banks and the planning authorities, rather than by the free market.
- 3 China does not yet have free or competitive labour markets. People are still, by and large, allocated to jobs by government departments rather than by the market.
- 4 Although since 1978 China has been following a so-called 'open door' policy in its economic dealings with the rest of the world, and has benefited enormously from this, its economy is very far from being closely integrated with the world economy. The country maintains a whole plethora of restrictions on imports and on the free movements of capital. China in recent years has been a major recipient of foreign direct investment, but such investment is nevertheless subject to government controls and restrictions at the national, the provincial, and the local levels.

All this raises the question that if competitive markets are crucial to economic growth, as the Bretton Woods' institutions continually assert (IMF and World Bank), then how come the Chinese economy has had this extraordinary economic growth for a long sustained period with markets which are highly imperfect, segmented, or do not exist at all? Equally anomalously for conventional economics, a clear definition of property rights and private ownership of productive assets is thought to be critical for economic efficiency, for technical progress and for wealth creation. Yet in China 90 per cent of industrial capital is in state hands, and most land is still collectively owned. As *The Economist* (28 November, 1992, p. 16) ruefully remarked, the story would be much simpler to tell if only the highly successful Chinese small scale enterprises—the so-called

TVEs (township and village enterprises)—were privately owned. However, by and large, they are not; their assets being owned by the towns and the villages.

It could of course always be argued that the Chinese growth rate would have been even faster had they had free and competitive markets and private ownership of the means of production. This is, however, highly unlikely in view of the fact that for a large country the Chinese economy has recorded historically unprecedented growth rates in the reference period. What the Chinese story tells us so far is that a country can perfectly well grow extremely fast without having free and competitive markets and private ownership of productive assets.



Decentralization has breathed new life into country districts of China. Here new buildings rise in Kunming city centre in Yunnan Province

What then accounts for China's extraordinary economic performance during the last fifteen years? Research suggests that an important factor has been the very high degree of decentralization which the Chinese have carried out as a part of the economic reform programme. This has transferred most production decisions from the central ministries in Beijing to the provincial and local levels. Such decentralization may have some unfavourable side-effects, such as market fragmentation, mentioned earlier, but on the whole it has been a powerful force in promoting economic efficiency, in motivating people and releasing their creative energies.

Moreover, it must be emphasized that although the Chinese economic performance is anomalous from the perspective of the World Bank's competitive markets paradigm, it accords very well with the actual experience of the other East Asian countries as outlined in the previous sections. Starting from an extreme position where the 'plan' totally dominated the market, the Chinese are attempting to achieve, like the other highly successful East Asian NIEs, that desirable combination of the 'plan' and the market which is most conducive to rapid industrialization. Similarly, like these other economies, the Chinese are seeking strategic, rather than close, integration with the world economy.

### 3.9 The US economy: achieving full employment with rising real wages

As noted in Section 3.1, the USA, the world's most advanced economy, has performed very well in important dimensions during the last ten years. The rate of unemployment is low, as is inflation, and the economy is expanding at a trend rate of growth of 3 per cent per annum. This may appear small beer as compared with the contemporary East Asian rates of economic growth, but this pace is nevertheless faster than that of either Western Europe or Japan. A priori, economic growth in the world's most advanced economy may be expected to be slower than in others, mainly because it is at the technological frontier. Therefore, unlike the less advanced countries which can copy the leader's technology, the leading country is obliged to carry out the more difficult task of further technological development by itself. In that sense, the slower growth in the USA compared to China or South Korea does not necessarily reflect any economic inefficiency. By the same token, the faster US growth, compared to that of Western Europe or Japan during the last decade, is highly creditable. (See Singh 1995b, 1997a and 1997c for a fuller analysis.)

Nevertheless, there are chinks in the US armour. First, although the rate of unemployment is low, there is still considerable involuntary unemployvqment. As the US economist, Lester Thurow, notes:

There are 8.1 million American workers in temporary jobs, 2 million who work 'on call' and 8.3 million self-employed 'independent contractors' (many of whom are down-sized professionals who have very few clients but call themselves self-employed consultants because they are too proud to admit that they are unemployed). Most of these more than 18 million people are looking for more work and better jobs. Together these contingent workers account for 14 per cent of the workforce.

(Thurow, 1992)

Second, and more importantly, the US economy has not been delivering the increases in real wages (money wages adjusted for inflation), which American workers have traditionally come to expect. Until recently real wages of each generation of American workers have historically been twice as high as those of the previous generation. This process seems to have stopped in 1973, since when real wages, particularly of manual workers, have fallen rather than increased. The average real wage for non-supervisory workers in the USA declined at a rate of 0.3% per annum in the 1970s, 1.0 per cent per annum in the 1980s and 0.3 per cent per annum between 1990 and 1994. In the 1960s, in contrast, the corresponding average real wage rose at a rate of 1.4 per cent per annum.

Third, in addition to the huge under-employment and stationary or declining real wages, another unfavourable aspect of the labour market experience in the USA in the recent period has been the growing inequality of wages. Wage dispersion had decreased

in the USA during the 1950s and 1960s, but wages have become much more unequal during the last two decades.

It is customary to suggest that the superior US performance, in terms of employment, is due to the greater wage price flexibility of the US economy, compared to European economies. However, the story must surely be more complicated than that in view of the fact that the US labour market is not only more flexible now in relation to the European economies, but it was also more flexible in the 1950s and 1960s. Yet Europe had more or less full employment in the earlier period, and that record was far superior to that of the USA. The essential point here is that Europe outperformed the USA at the time because the European economies were growing at twice their current rate, and at nearly one and a half times the then US rate. That enabled these economies not only to have fast employment growth, but also rising real wages.

Although the US growth rate in the 1950s and 1960s was slower than that of Western Europe, it was faster than its current rate. The higher growth rate of the earlier period enabled the country to have not only as good as, if not a better, employment record than it does today, but more significantly it also enabled real wages to rise appreciably in that period (as was noted earlier).

Clearly, to meet the historic aspirations of the American people, it is not enough to have high overall employment, but it is also essential to have growing real wages. This would require a trend increase in the rate of growth of the US economy to levels such as those attained in the 50s and 60s, when, as seen above, the economy was able to deliver on both these dimensions.

An important question which arises from the above analysis is whether the market forces left to themselves will be able to generate the required increase in the trend rate of economic growth. Since such fast growth has not materialized in the last two decades, the answer to this question may be presumed to be no. That then raises the issue of what kind of public interventions will help to achieve the desired objective, and whether or not such interventions would be acceptable in the USA's political culture.

#### 3.10 Conclusion

This chapter has examined economic growth in East Asia, which has emerged as by far the most dynamic region of the world economy. Over the last four decades several East Asian countries have expanded for sustained periods at rates which are historically unprecedented. Alternative theories have been reviewed concerning the long-term growth processes in these countries, including China and Japan. The last part of the chapter has commented briefly on employment, real wages and economic growth in the world's most advanced industrial country on the other side of the Pacific—the USA.

What lessons can be drawn from the analysis of sources of economic growth in the capitalist East Asian NIEs, as well as China and the USA? The first point which emerges is that the East Asian countries have achieved extraordinarily high rates of overall growth, despite their comparatively poor performance in terms of growth of total factor productivity. Second, regarding proximate causes, we have seen that these high growth rates can be explained in terms of high rates of savings, investment and human capital formation. In that sense, it is right to suggest that the East Asian economic development

is not a miracle—it is compatible with standard economic theory. Third, however, the previous analysis indicated that although these high savings and investment rates were by and large implemented by the private sector in most East Asian countries (other than China, see below), nonetheless government played a critical role. The government did not simply follow a hands off, market friendly approach in the World Bank sense, but rather played a crucial co-ordinating, as well as stimulating, role in this process. Further, contrary to the World Bank report, the integration of the East Asian countries with the world economy has been strategic, rather than close.

With respect to these proximate sources of economic growth, the Chinese story is similar, despite its not being a free enterprise economy. Most Chinese land and industrial capital is not privately owned but is under collective ownership. China has introduced markets in the last twenty years, but most product, capital and labour markets are highly imperfect, segmented and incomplete. Moreover, notwithstanding the 'open door' policy launched in 1978, the Chinese have also implemented a strategic rather than a close integration with the world economy. Thus in all these respects, the Chinese experience is not compatible with the World Bank paradigm of competitive markets being a necessary condition for fast economic growth.

The Chinese experience is also similar to that of other East Asian NIEs in another sense. In China also, the growth of factor inputs (specifically, high rates of human and physical capital formation) have been much more important than the growth of total factor productivity in accounting for its overall economic growth. Before Chairman Deng's reforms in 1978, China had a Stalinist command economy with a high degree of centralization of economic decision making. Under Deng's reform, the Chinese introduced markets and decentralization, but they are attempting to find an optimal combination of the plan and the market which will best promote the country's economic growth.

There are two further conclusions which emerge from the analysis of TFP growth in the East Asian NIEs which deserve to be highlighted. First, low TFP growth does not imply that a country has had no or slow technical progress in the ordinary meaning of these terms. That would be a silly conclusion for a country like South Korea in the light of its extraordinary success in continually upgrading its export structure. Thirty years ago it was exporting mainly textiles. Today much the larger part of its exports come from cars, computer chips and other technologically advanced products. The second and related conclusion is that Krugman's inference that these countries will not be able to sustain these high growth rates is only valid within his own neo-classical theoretical framework. In a non-neo-classical analysis where technical progress is embodied in new capital goods as well as replacement capital, there is no reason to expect that high rates of investment will necessarily lead to decreasing returns.

Before we conclude the discussion of East Asia and China, it is necessary to address the following puzzle which may have troubled many readers. The puzzle is why did the state succeed so spectacularly in East Asia when it failed elsewhere, for example in Latin America and Africa? This is a large question, but very briefly, the answer is that apart from the close relationship between the government and business in these countries, and the particular characteristics of their financial systems, there were two other factors which were extremely important. First, unlike many countries in Latin America and South Asia, although the East Asian countries implemented import controls, they also had export orientation. Indeed, export promotion and import controls in East Asia were organically linked. Import controls provided the Japanese and the South Korean corporations, for instance, with high profits which enabled them to raise their rates of investment and to increase their share in the world markets. The corporations in these countries were obliged to promote exports by government policies which made it clear to any corporation that to get ahead, to get help from the state or to be able to get a licence to import foreign technology, it had to export. The second factor which was extremely important is implicit in the above discussion. Unlike developing countries elsewhere which also provided subsidies or imposed restrictions on imports, the East Asian governments imposed strict performance standards on the recipients of the government largesse. These standards often took the form of specific targets for exports or for technological upgrading.

Finally, we turn to the case of the USA which differs from that of East Asia in two important dimensions which have been the focus of this chapter. First, unlike in East Asia, total factor productivity, in terms of conventional growth accounting, has been an important determinant of US economic growth. Second, the role of the state has been different, rather than being necessarily less extensive. The US government intervenes in the country's industrial development in a wide range of ways, for example, health and safety standards, antitrust laws, environmental control measures, government procurement, especially of military goods. Nevertheless, there is a qualitative difference between US interventions and, say, Japanese government interventions in relation to their respective economies. The latter are co-ordinated, primarily by MITI (see Chapter 6), and constitute an overall industrial policy, whereas in the USA there is no such strategic coordination. During Japan's high growth phase, MITI could use its arsenal of coercive laws, including, specifically, control over the allocation of foreign exchange to bring about the desired co-ordination of industrial investments. However, since Japan's graduation to membership of the OECD (the rich country's club) around 1970, MITI has to rely much more on its persuasive powers, but nevertheless it continues to carry out its tasks of strategic co-ordination of Japanese industrial development in the light of evolving international competition.

This chapter has argued that although the US economy has performed well along some dimensions in the last decade compared with western Europe and Japan, it has been unable to meet the historic aspirations of its people. In order to provide both full employment and rising real wages to satisfy these aspirations, it is necessary to have a trend increase in the rate of growth of the US economy. The chapter suggests that market forces left to themselves are unlikely to generate the required rates of growth.

#### References

Amsden, A. (1989) Asia's Next Giant, New York, Oxford University Press.
Asian Development Bank (1995) Asian Outlook, Manila, ADB.
Chakravarty, S. and Singh, A. (1988) The Desirable forms of Economic Openness in the South, mimeo, Helsinki, WIDER.
ECLA (1995) Statistical Yearbook for Latin America and the Caribbean, Santiago, ECLAC.
IMF (1996) World Economic Outlook, Washington DC, International Monetary Fund.
Krugman, P. (1994) 'The myth of Asia's miracle', Foreign Affairs, vol. 73, no. 6, pp. 63–75.

- Nelson, R.R. (1981) 'Research on productivity growth and productivity differences: dead ends and new departures', *Journal of Economic Literature*, vol. XIX, pp. 1029–64, September.
- Rodrik, D. (1995) 'Getting intervention right: how South Korea and Taiwan grew rich', *Economic Policy*, no. 20, April; pp. 55–97.
- Singh, A. (1994) 'Openness and the market friendly approach to development: learning the right lessons from development experience', World Development, vol. 22, no. 12, pp. 1811–24.
- Singh, A. (1995a) 'Institutional requirements for full employment in advanced economies', *International Labour Review*, vol. 134, no. 4–5.
- Singh, A. (1995b) 'The causes of fast economic growth in East Asia', UNCTAD Review, Geneva, UNCTAD.
- Singh, A. (1996a) 'Profits, savings, investment and fast economic growth: a perspective on Asian catch up and implications for Latin America', *Department of Applied Economics Discussion Papers in Accounting and Finance, no.AF33*, Cambridge, University of Cambridge.
- Singh, A. (1996b) 'The plan the market and evolutionary economic reform in China' in Abdullah, A. and Khan, A.R. (eds) *State, Market and Development: Essays in Honour of Rehman Sobhan,* Dhaka, The University Press Limited.
- Singh, A. (1997a) 'Liberalisation and globalisation: an unhealthy euphoria' in Michie, J. and Grieve Smith, J. (eds) *Employment and Economic Performance*, Oxford, Oxford University Press.
- Singh, A. (1997b) 'Catching up with the West: a perspective on Asian economic development and lessons for Latin America' in Emmerij, L. (ed.) *Economic and Social Development into the XXIst century*, Washington DC, Johns Hopkins Press for the Inter-American Development Bank.
- Singh, A. (1997c) 'Expanding employment in the global economy: the high road or the low road?' in Arestis, P., Palma, G. and Sawyer, M. (eds) *Markets, Unemployment and Economic Policy, Essays in Honour of Geoff. Harcourt, vol.2,* London, Routledge.
- Thurow, L. (1992) Head to Head: The Coming Economic Battle Among Japan, Europe and America's Business, New York, Morrow.
- UNCTAD (1996) Trade and Development Review, Geneva, UNCTAD.
- Wade, R. (1990) Governing the Market, Princeton, Princeton University Press.
- World Bank (1991) *The Challenge of Development, The World Development Report,* Washington DC, The World Bank.
- World Bank (1992) World Development Report, Washington DC, The World Bank.
- World Bank (1993) The East Asian Miracle, Oxford, Oxford University Press.
- World Bank (1996) Data Disk, Washington DC, The World Bank.
- Young, A. (1994) 'Lessons from the East Asian NICs: a contrarian view', *European Economic Review*, vol. 38, no. 3/4.

### CHAPTER 4 Financial systems and monetary integration

Grahame Thompson

### 4.1 Introduction

A key element in the overall explanation of the growth record for the Asia-Pacific is the role played by the financial systems in the countries of the region. Financial systems serve to organize the flows of financial resources that are so important for investment and for the general operation of an economy. A modern economy is typified by the degree and sophistication of its financial system: how effectively it facilitates the exchange of goods and services by providing a measure of prices and means of payment function, and how it helps in the organization and allocation of economic resources so as to maximize investment and national income growth.

The way the financial system assists in these processes is often discussed under the heading of 'financial development'. 'By the process of financial development is meant the gradual evolution, in the course of economic development, of financial institutions— money, banks and other financial intermediaries, and organized securities markets' (Arndt, 1983, p.86). Financial development places command over the disposal of real resources in the hands of those agents that control flows of money, financial institutions and financial markets. Thus the holding of financial resources is in many ways just as important to the overall development process as is the control over land resources, or physical infrastructures such as transport, power and water resources, or the development of human capital and an educated workforce.

The idea of financial development allows an important distinction to be drawn between, on the one hand, *physical*, or 'real', resources and capital, and, on the other, *financial*, or 'money' resources and capital. Financial capital and real capital, whilst different and possibly subject to different processes of development and with different consequences for economic growth, may also be viewed as complementary to one another. The precise relationship between these two is the subject of a great deal of economic debate and is not settled. Some argue that 'real capital' is the key, and greater attention should be paid to developments in the real economy (the formation of physical capital, education and employment levels, the volume and composition of real output, etc.), while others argue that just as important as these are 'monetary capital' and the development of monetary variables and aggregates (interest rates, money supply, financial flows, and the asset and liability structures of various agents). What is happening to the financial dynamic of an economy is just as important as its real dynamic, if not more so, argue the latter group, because financial incentives are the key to the mobilization of these other real resources anyway. It was in the context of reviewing

the development process in many of the East Asian economies in the 1960s and 1970s particularly Taiwan and Korea—that the importance attributed to the purely financial side of the economy for development was stressed (McKinnon, 1973; Shaw, 1973). As we shall see, the debate originated by these analyses in the 1970s continues today.

For the most part this chapter concentrates upon financial developments and financial flows. This process is typified by the increased *monetarization* of the region's economies. But there is still a significant proportion of *subsistence* economic activity in many of the countries of the region, and barter remains an important means of exchange (see Chapter 12). This does not pass through a market exchange so is not registered in a financial form. However, for the most part, the monetarized sector has expanded in the regional economies to such an extent that concentrating upon this will not distort the analysis—though the extent of the non-monetarized sector was still perhaps surprisingly large even in the 1980s (Arndt, 1983, Table 2, p. 92; Cole, 1988, p. 28). The non-monetarized sector tends to be the greatest in those economies designated by Gangopadhy (Chapter 2) as the 'traditional developing nations' and 'almost-NIE' types, such as Indonesia, the Philippines, Burma and Laos.

The process of financial development has also involved the *deepening* of financial relationships in these economies. The development of the financial system leads not only to an extension of monetary relations—so that more and more economic activity is drawn in under the monetary net—but to an enrichment in levels of financial intermediation as well. Retail commercial banks are an obvious first institutional level in a monetarized economy. But they are then supplemented by the development of a set of secondary financial institutions such as specialist savings banks, mortgage lenders, merchant banks, insurance companies, discount houses, etc., and then further by the emergence of various types of security markets. These latter deal with the issue of commercial company shares and bonds and the secondary trading of these, and with the government bond market as the role of government expenditure expands. In addition, as the economy internationalizes another level can be added, which deals with all the international financial transactions that typify a sophisticated and increasingly open economy.

The central role of the financial system is to undertake the task of *financial intermediation*. Financial intermediation refers to the process of gathering financial surpluses from a large number of small savers, aggregating these, and then lending them on to borrowers who wish to use the surpluses for financial or real investment. The financial institutions and markets thus act as intermediaries between lenders and borrowers. They thereby help constitute aggregate savings and investments in an economy (see Chapter 3). But financial institutions and markets are not simply passive conduits for this process. They always establish terms and conditions by which they will borrow and lend. So a key element in the allocation of financial resources in any economy is the organizational structure of its financial system, and the terms and conditions under which its financial institutions and markets operate. There is a diversity in these as between the regional economies and they produce different outcomes as a result. Part of the debate is over how efficient these different systems are in terms of the growth performances they have contributed towards.

An additional complication here is that the process of financial development also implicates what has come to be known as *financial repression*. In some ways this is seen as the opposite of financial development. It refers to various controls put on the operation

of the financial system, usually in the form of government action. Thus, various types of liquidity controls and capital adequacy constraints are placed upon banks, and legal ceilings on interest rates and credit access conditions have been implemented by governments. Governments have also subsidized credit, or physically directed credit and investment funds to favour different industrial sectors or product lines in the economy. They have used taxation to stimulate or repress certain activities and markets. Regulations on who can and who cannot operate in various parts of the financial system and markets have been operative. Foreign capital and exchange controls have been prevalent at times. All these go towards establishing restrictions and direct and indirect controls over the free operation of the financial system. Those advocating a 'market friendly' approach towards economic policy (such as McKinnon and Shaw mentioned above) have designated these as 'impediments' to the efficient operation of the financial system, with a concomitant loss of performative outcomes. On the other hand, those less wedded to the market-friendly approach have designated these practices as precisely the ones that have helped the regional economies to achieve their spectacular successes. There are some advantages in the 'rigidities' that these practices have installed in the economies, and these have positively worked to enhance economic outcomes (Wade, 1985; Harris, 1988).

A final introductory point is that this chapter concentrates on the financial systems in what might be termed the mainstream market economies of the region. There are two exceptions to these which are not discussed at any great length here.

The first is the remaining communist states. Here the state directly allocates credit through its control of the financial system and banking organizations. It is the state banking system that gathers surpluses from its productive organizations and citizens, and then passes these on to those organizations responsible for investment as and where the plan targets dictate. In fact, there are always informal lines of credit operating in even the most closely controlled and state directed economies of this kind. For instance, productive organizations can extend trade credit between themselves within the various chains of raw materials, component and goods supply. Also, although in theory the state should exercise central control over the money supply in these countries, in fact, this is often not the practice and rampant inflation can result from lax controls. In addition, the state banking system is often over-burdened with 'non-performing assets' which are debts that state enterprises will never be able to repay.

The second exception to the main thrust of this chapter involves Islamic banking. This is important in Malaysia and to a lesser extent in Indonesia, the main Islamic countries of the Asia-Pacific region. Islamic banking forbids the operation of interest payments. As a result, the Islamic banking sector of these economies places the emphasis on the participation of lenders in the normal profits that emerge from enterprise. Fixed interest rate returns are replaced by variable returns dependent upon profitable successes (or losses) of the borrowing enterprise.

The plan of the rest of the chapter is as follows. The next section looks at the general trends in financial development in the regional economies, concentrating upon the East Asian nations. Then the chapter reviews issues associated with financial liberalization. This is followed by an analysis of international financial developments. The chapter then moves on to look at the role of the financial system in various macroeconomic management issues. The final main part of the chapter raises issues of financial

integration across the Pacific Rim and beyond. This is followed by a short concluding section.

### 4.2 Financial development in the Asia-Pacific

The economies of East Asia have had a long history of financial development, the origins of which lay in the historic importance of rice and other grain production to their economies. Combined with a monsoonal climate that sharply delineated the growing season from the dormant one, and which often resulted in crop failure, this led East Asian societies to develop a sophisticated grain surplus and reserve system. The legitimacy and survival of early political formations in these countries were highly dependent upon how well they administered and protected their grain reserve and redistribution systems. This resulted in an early and easy move from a grain based credit system to first a metallic coinage based financial one, and then into a fiat or paper system (see, for instance, the case of China discussed in Yang, 1952).

In addition, this early encounter between proto-state political formations and the protection of the credit systems—which eventually evolved into genuinely financial systems—led to the strong predisposition in these countries towards modern governments' involvement in the raising and allocation of financial resources. This is something deeply embedded in these societies, a legacy of a particular historical trajectory, and one which marks them off from the more privately commercial evolution of Anglo-American type financial systems. The role of government has always been ambiguously placed *vis-à-vis* the development of the financial system in the Anglo-American historical experience. The consequences of these different historical experiences linger even today.

Herein, in part at least, lies the origin of one of the enduring points of debate about the form and organization of the financial system in the economies of East Asia and those of the Anglo-American type; the proper role and consequences of governmental regulation or guidance of the financial system—what in modern parlance is argued to lead to 'financial repression' as mentioned above. We return to this issue in a moment.

### Explaining high levels of savings and investment

One of the most significant features of the growth experience of the East Asian nations is the high levels of investment in their economies. This is argued to be a key element in their economic success (Chapter 2). Supporting these levels of investment has been the phenomenal growth in domestic savings, particularly household savings. Aggregate Taiwanese savings, for instance, increased from about 5 per cent of GDP in 1950 to over 30 per cent in the late 1970s (Wade, 1985, p. 107); Korean savings increased from 3 per cent in the 1950s to 35 per cent in the early 1990s (Chapter 3, Table 3.3). Why have these countries been able to save so much?

To begin with, the underdeveloped nature of the state sector and government expenditure may have had something to do with it. A low ratio of government expenditure to GDP means that even if the government sector displays net dissavings, that is, if government expenditure is higher than government revenues (which was often the case in the 1950s and early 1960s in the NIEs and almost-NIEs), then its impact on the economy's savings rate overall is still small. In fact, particularly since the mid 1960s, the government sector had mainly been in surplus in these countries, as Table 4.1 indicates, which added to the positive savings rate. Government savings represent the vigorous use of fiscal policy to mobilize surpluses, via means of taxation, and any surpluses generated by publicly owned enterprises. But the main bulk of the savings is undertaken by households (and to a lesser extent by firms—the other contributor to private savings as shown in Table 4.1). Here a number of considerations enter the picture.

One element involves another aspect of a relatively low state involvement in the economy. The absence of a developed welfare and social security system means that households have had to save to provide for this themselves. Similarly, with the absence of a large public housing stock the pressure to save for private housing is greater. And with the

Economy/year	Public savings	Private savings
East Asia		
Indonesia 1981–88	7.7	14.0
Japan		
1945–54	5.3	12.0
1955–70	6.2	17.2
1971-80	4.6	20.1
1981–88	5.1	15.8
Malaysia		
1961–80	3.2	18.7
1981–90	10.3	19.1
Philippines		
1980–83	10.4	11.5
1984–87	1.4	14.1
Singapore		
1974–80	5.5	22.6
1981–90	18.5	24.0
Thailand		
1980–85	14.3	4.7
1986–87	8.6	14.6
Latin Amorica		

### Table 4.1 Public and private savings in a selectionof Asia-Pacific developing economies

Latin America

Chile		
1980–84	3.9	2.3
1985–87	7.9	0.6
Mexico 1980–87	4.3	17.5

### Financial systems and monetary integration 85

Source: adapted from World Bank (1993, Table 5.6, p. 210)

underdeveloped nature of the financial system the ability to raise finance for house purchase—and additionally, consumer durable purchase—is restricted. Combined with the practices of demanding very large down payments for housing and consumer credit, all these have contributed to a high savings propensity.

In addition, the actual form of the financial system can make a significant impact. The East Asian economies established means of mobilizing small savings through postal savings institutions very early in their development processes. These proved to be a low cost and secure means of raising the savings ratio (World Bank, 1993, pp. 218–20). In Japan, Post Office savings accounts paid no tax on interest earned, and they received favourable rates compared to commercial banks.

Another feature concerns the access to finance by small and medium sized businesses. By and large, the banking sectors in the East Asian economies have either been government owned or highly concentrated in private hands (cartelized and obligopolistic). As discussed further in a moment, governments have used their powers here to direct credit in various ways, and this has traditionally favoured their own publicly owned enterprises or large private companies (for example, the *chaebol* in Korea), or those companies attached to the large manufacturing and trading groups (for example, the Japanese *keiretsu*, which each include a bank as a central part of their conglomerate structure). Small independent companies have been disadvantaged as a result, and they had to resort to their own savings to invest. Given the importance of the 'family firm' in these countries, this often takes the form of personal and household savings. (For a full explanation of these different business arrangements amongst the East Asian countries see Chapters 6 and 9.)

A further aspect of the way the financial systems have operated relates to their regulation and control. Governments have been loath to see any part of their financial system collapse. They have supervised them closely in terms of their solvency, risk profiles and capital adequacy ratios, regulating spreads and non-performing assets. They have exercised tight controls over licensing arrangements and ownership structures. Prudential regulation has been the order of the day. In addition, during periods of financial distress—when an actual crisis strikes—governments have quickly come to the rescue to bail out the banks in difficulty, or to press other stronger financial institutions to lend their support or take them over. This has produced a general climate of confidence and security for depositors. Coupled with a commitment to a stable macroeconomic environment and low levels of inflation, lenders have been well served by their governments, with positive long-term results in terms of their attitudes towards savings. This is not to say that there have been no financial crises or bouts of inflation—there have been many serious incidents of these—but it is to say that they have occurred less often, and been generally better controlled events, than in other comparable countries.

A final issue in terms of the characteristics of the financial system is the question of 'forced' savings. Some East Asian governments (for example, in Japan, Malaysia and Singapore) have in effect made saving mandatory, in terms of pension plans, welfare provision, and for the purchase of consumption goods. The most spectacular example of this is Singapore's Central Provident Fund (CPF), into which citizens are forced to invest their savings for all manner of welfare benefits (see **White, 1998**). (The CPF, which is an off-balance sheet item in the government accounts and thus not part of government revenues or expenditures, has become an attractive model for many in Europe and North America who are seeking to 'reform' their own welfare systems.) It is argue that the CPF boosted savings in Singapore by four per cent of GDP per year during the 1970s and 1980s (World Bank, 1993, p.219).

### What role for interest rates?

One of the most controversial aspects of East Asian financial development in the post-Second World War period concerns the impact of the regulated, guided and directly controlled financial systems found there. A key element in this form of government intervention has been control of the interest rate and, in particular, the artificial repression of the rate to below what might have been its market rate. The imposition of such 'interest rate ceilings', often implemented along with the deliberate direction of credit into certain favoured areas of the economy, is generally recognized to have been prevalent in most of the economies at least at some time in the post-war period. This experience is contrasted to the much more 'open' and market friendly nature of US interest rate determination, but also of the other Anglo-American economies in the region (along with Hong Kong, which took a more *laissez faire* attitude towards the development of its financial system under UK guidance). In a moment we come to the whole debate about the liberalization and deregulation of these economies, but it is generally recognized that such controls have existed (and continue to exist in many cases).

To put this into a slightly more formal framework, consider Figure 4.1 which sketches the relationship between real deposit interest rates and real growth rates ('real' rates are money rates adjusted for inflation).

A simple traditional view is illustrated by Figure 4.1(a), where there is a linear relationship between the interest rate and growth rate. As the real deposit interest rate increases we might expect the real growth rate to increase since there is an incentive for both savers to save and borrowers to borrow, resulting in investment and growth. Both savers and borrowers receive real returns to their activity as the economy moves into the positive real-valued right hand quadrant of the diagram. Even with negative real interest rates, however (though positive money rates), some real growth is likely since 'money illusion' is a commonly observed phenomenon (savers and investors do not totally discount for inflationary conditions).



### Figure 4.1 *Real interest rate and real growth rate*

But probably a more adequate picture is illustrated in Figure 1.4(b). Here the effect of real interest rates on growth is a quadratic one. First the real growth rate increases as the real interest rates increase. But then it peaks and falls away as the real interest rates expand further. At low real interest rates, the growth rate is low because of the lack of incentives to invest. This incentive expands as the interest rates increase, but after a while the growth rate increase falls away as fewer and fewer attractive investment opportunities present themselves. Eventually the schedule turns down and the growth rate declines. Why might this be the case? At very high real interest rate levels too many risky and adventurous investments are undertaken, which, as they prove unsuccessful, produce lower growth rates. It could even be the case that the schedule moves into a negative real growth rate phase if the interest rate were so high as to lead to purely speculative activity and investment in very risky ventures.

In both cases (a) and (b) discussed above the suppression of interest rates (or higher inflation) will lead to lower growth rates than would otherwise have been the case. In fact, attempts to test a model such as that shown in Figure 4.1(b) on the empirical conditions found in the Asia-Pacific economies have not proved very successful, mainly because the real deposit rates of interest have not varied over a wide enough range to register sufficient observations on the horizontal axis (Fry, 1995, p.43). But the figure still provides us with a convenient framework for discussing further the possible impacts of the interest rate on savings and investments.

### Looking towards investment

Other things remaining equal, lower interest rates would normally lead us to expect lower savings and lower investment in an economy. But despite the restrictions on interest rates and credit mentioned above, the East Asian economies have displayed very high comparative savings and investment ratios, and historically spectacular GNP growth rates. This is the apparent paradox that lies behind the big debate about the actual effects

of government interventions in the financial system of the economies in question. Might these interventions be the actual stimulus for that growth rather than a constraint upon it?

Before we pursue this directly, it is important to recognize another endemic feature of these economies' financial systems. This involves an 'unorganized' part of that system, what is termed the 'curb market'. Most of the economies have experienced a two-tier financial system: an organized and formal part and an unorganized and informal part. Many of the restrictions put on the formal market may have led to the stimulation of the informal, 'curb-side' part. Alternatively they may have operated as complementary to each other. Curb-markets are usually viewed as inefficient, since they may have to cater for only those borrowers rejected by the formal sector, who are thereby thought to be more risky to lend to, and curb market interest rates are therefore much higher than they need be. The overall result is a less than optimal outcome for everyone.

However, curb markets seem to have played an important part in the financing of the small and medium sized family business enterprises disadvantaged by the 'big firm' bias of the formal financial sector, in providing consumer credit, and in the crucial area of export finance (particularly before the governments took this up with a vengeance as they turned to an export orientated growth strategy in the late 1960s and 1970s—see Chapter 2). As financial development proceeded, the curb market seems to have become gradually absorbed into the formal sector, though it remains important in some of the almost-NIEs and traditional developing countries of the region. Export credit became a key provision of the formal systems, along with priority finance for the small and medium sized businesses. Both of these were once again guided and organized by the governments themselves, however. So we have yet other instances of governments intervening strongly to promote their particular policy goals as their economic priorities changed.

The traditional way that finance is thought to be raised for investment in the Anglo-American systems is via the stock market. Companies can float their shares on the market and use the proceeds to expand their business. In fact this is of less importance than is often recognized, even in the Anglo-American systems, since it is internal savings that contribute the greatest proportion of resources for company investment, via depreciation allowances and retained earnings. Only a minor proportion is generated from external sources such as the stock market or as loans from banks.

As Figure 4.2 demonstrates, the stock markets in the Asia-Pacific region vary considerably in size. Of the East Asian economies, only Japan had a sizeable stock market in terms of capitalization (the worth of the shares traded) in the mid 1990s. In addition, the number of companies traded on the East Asian markets (and in Australia) remained modest, as shown in Figure 4.3. Hong Kong, the largest of the region's emerging markets, is only about 5 per cent of New York's and 9 per cent of Tokyo's.

Thus the role of the equity market in these economies remained small, at least until the mid 1990s. Their main activity had been in the secondary market of buying and selling already listed shares. To some extent this is the result of government restrictions put upon the operation of stock markets in the countries, which have not favoured their development. Similarly with the corporate bond market. But more recently these markets have expanded, particularly in Korea, Taiwan and China, and with further liberalization their role in mobilizing financial capital might be expected to expand.

All the East Asian developing countries have established specialist financial institutions explicitly designed to provide long-term development finance at favourable

rates. These development banks have been a central part of the governments' overall attempts to mobilize domestic savings for investment, and to attract foreign capital. They have been substantial long-term lenders in Indonesia, Japan, Korea, Taiwan and China in particular. There remains a debate about the effectiveness of these institutions, however. Some have argued that they have made a significant contribution to rapid industrialization (World Bank, 1993, pp. 226–8), while others have



## Figure 4.2 Stock market capitalization (end of 1993)

Source: Pacific Economic Cooperation Council (1994, Figure 19.5, p. 117)



Figure 4.3 Listed companies in Asia-Pacific stock exchanges (end of 1993)

Source: Pacific Economic Cooperation Council (1994, Figure 19.5, p. 116)



Tokyo stock exchange

suggested that their record is a poor one, that they have not become self-supporting, autonomous institutions capable of mobilizing resources on entirely commercial terms (Fry, 1995, pp. 50–1).

An additional reason for the high levels of domestic investment, and one of the most controversial, is controls on capital outflows. Japan, Korea, Taiwan, China, Indonesia, Thailand and Malaysia have all employed these kinds of restrictions in the formative period of their economic take-offs. Put alongside the repression of interest rates discussed above (which encourages investment), and the important role of direct government investment in infrastructure and in the enterprises owned by the public sector in various countries, the basic explanation for investment becomes clearer. What is impressive about these countries is the wide range of institutions and policies all designed to meet the single target of increasing investment, and the determination of the authorities to activate these at any time. Thus, in many ways, the direct and indirect involvement of the government in these economies would seem to have stimulated the high levels and rates of investment, rather than to have restricted them.

#### 4.3 Pressures for liberalization and deregulation

As mentioned above, the trajectory of financial development is towards a wider and wider embrace of the financial system, and for its deepening in terms of levels and markets. This is not a smooth evolutionary process, however, nor may it quite capture the full complexities of the historical experience of all the countries of East Asia. An added element involves the relationship between government controls and financial liberalization.

A simple evolutionary approach might suggest that at first the systems were characterized by heavy government involvement, which was then followed by successive rounds of decontrol and liberalization. In fact the relationship between control and decontrol has also been complex and non-linear. There have been rounds of liberalization in parts of the economies' financial systems throughout their development processes, with often quite rapid and radical changes in policy direction. Nor have these changes been all in the single direction of a more market-friendly approach. They have oscillated between more liberalization and less. The 1960s saw some early liberalization in Korea and Taiwan in particular, which stimulated the work of Shaw and McKinnon referred to in Section 4.1. This was followed by the almost reinvention of regulation in the late 1960s and into the 1970s. Then there was another round of deregulation in the very late 1970s and into the 1980s as the economies reacted to the adjustments pressed upon them by the turbulence in the international financial system. Once again, however, the effects of deregulation seem to have melted away as the recovery took place in the 1990s.

Thus perhaps the best way of viewing the question of regulation and deregulation is not along a single dimension but as involving multidimensional adjustments and manoeuvrings, in which the governments have been most concerned to maintain their capacities to 'supervise' their financial systems in the interest of their own domestic economic development. Often this has frustrated those pressing for greater overt 'disengagement' from intervention, but the records seem to show that this has not too adversely affected their growth rates, at least during the period of the main growth and maturity of their economies. Thus the governments have shown a very pragmatic attitude towards liberalization, bending to pressures and circumstances when that has seemed necessary, and then regaining the initiative as the pressures subsided.

However, a number of more recent developments may be conspiring to alter quite radically the possibilities of this pragmatic and careful approach. There has been a much more sustained change in the sentiment of economic policy making in the world at large, which is deeply affecting the economies of East Asia, just as it affected the Anglo-American ones beforehand. The momentum of a neo-liberal, *laissez faire* deregulatory programme, which is much more sustained and thoroughgoing than any before, is having a major impact on the economies generally, not just in respect to their financial systems. Pressures to sell off public assets and to denationalize the banking systems, to remove any remaining government restrictions on the operations of financial markets, to use competition policy to push private markets to reduce cartels and widen ownership, and

generally to withdraw from all but the most minimal of intervention, are more intense than ever before.

This programme is aided by two further developments: the slowdown in the growth of many of the economies (Chapter 1 and Figure 4.7 below), which was led by the Japanese economy in the early 1990s, and the increasing internationalization of the financial markets throughout the region. Both of these have added to the calls, from domestic as well as from international sources, for the further opening up of the economies to the freer operation of market forces. And these calls will be increasingly difficult to resist, so that a radical change in approach may be under way. These calls are particularly intense in the case of Japan, whose financial system is generally agreed not to have developed sufficient flexibility as its economy has matured into an advanced-country status.

Whether these liberalization moves will be for the good of all the economies in the region is another matter. For those wedded to the market-friendly approach it goes without saying that it will be. They have led the calls for this kind of deeper deregulatory programme ever since the 1960s. But, as we have seen, this may not have been in the best interests of the economies concerned. It will affect the almost-NIEs and traditional development economies of the region in particular, since they will not be able to develop behind the protective barriers that were operative for Japan and the first tier of NIEs in the 1960s and 1970s.

What this brief analysis of the pressures for further and rapid deregulation reveals is the need for further analysis of where these economies stand in an international context. Up until now we have been dealing with what is an essentially closed economy model. It is time to reconsider this, now set within the terms of the international economy.

### 4.4 The internationalization of the financial systems

There are three main initial aspects to discuss, which serve to introduce the international relationships in which the economies are located. As indicated in both Chapters 2 and 3, savings and investment need not be confined to the domestic economy if there is some international transmission mechanism that redistributes financial resources from one country to another. Here we refer to official and unofficial development assistance, to FDI flows and to portfolio investment as separate mechanisms for the international flows of capital into and out of a country. These make it possible for domestic investment not to be matched by domestic savings.

### Forms of international financial investment

Because of the strategic importance of the Asia-Pacific region, particularly its East Asian part, financial assistance from the USA was an important early source of capital for development. Post-war reconstruction in Japan became a US priority. In addition, strategic considerations led to a good deal of military aid to Japan, Korea and Taiwan in particular. This aid was important in launching these economies onto a growth trajectory, but the direct significance of non-military aid soon fell away as their internal growth dynamic began to emerge in the 1960s (for example, as their internal savings rates escalated—see Section 4.2). However, military aid has remained indirectly important in

that it has released these countries from some of the expenditure involved in providing for their own defence. The USA carried much of the burden from its own expenditures. Other official development assistance has been in decline in the region, though that originating from Japan remains an important source, particularly for the almost-NIEs and traditional developing countries. Indonesia remains the most important recipient of official financial assistance in the region.

Probably of much greater long-term significance are FDI flows, which add directly to the potential real productive investment in an economy. The trajectory of these flows was extensively discussed in Chapter 2. Inflows played only a very small part in the Japanese economy, and they were similarly low in the early years of Korea's take-off. Only after 1965 did FDI flows into Korea begin to expand. Taiwan followed soon after. Singapore has been the country most dependent upon FDI flows for its investment (responsible for 37 per cent of capital formation between 1991 and 1993), followed by Malaysia (25 per cent over the same period). In 1970 FDI inflows within East Asia and the Western Pacific were US\$270 million. By 1994 these had expanded to US\$42.7 billion. As pointed out in Chapter 2, the striking feature of the region's FDI flows is the growing intra-regional nature of the investments involved. In the mid 1990s the five main capital exporting countries were Japan, Taiwan, Korea, Singapore and Thailand. Japan is by far the largest intra-regional investor, followed by Korea and Taiwan, and more recently by Singapore and Thailand. The major recipients were Thailand, Malaysia, Indonesia and China. China became the single largest developing economy recipient of FDI in the world in the mid 1990s. While it remains the most alluring prospect to international investors, overheating of its economy and infrastructural problems could make China increasingly less attractive. Vietnam (along with India) looks to be the next hottest target for international investors.

In recent years the East Asian countries have also attracted substantial volumes of portfolio capital. Unlike FDI, portfolio investment represents the purchase of stocks and bonds. It is financial investment as opposed to the real investment associated with FDI. A number of developments have aided the trend towards portfolio investment. As the economies have matured, their stock markets and bond markets have grown, though as pointed out above their stock markets in particular remain thin in terms of the number of participants and operators and the size of investments. But the rate of growth of these markets in the 1990s has been particularly sharp, especially those involving the participation of foreign investors. Foreign portfolio investment in East Asia and the Western Pacific rose from US\$2.3 billion in 1990 to over US\$18 billion in 1995. However, there remain some restrictions on the short-term inflow of capital in many of the countries and also, perhaps more importantly, on the outflow of capital. But, in general terms, it is the marked liberalization of the economies in the region since the early 1980s that has opened them up to portfolio and other financial investments. It is Hong Kong and Singapore that carved themselves out a strong regional position as financial centres, largely because of their perceived stability and open nature, depth of participation, and low risks.

Another development that has made the East Asian economies attractive for portfolio investment is the growth of the bond issue made by governments. Governments have found this a convenient method of raising finance, and the growth in the credibility of governments in the region has added to their ability to attract foreign funds. The general macroeconomic stability in the region has also enhanced this kind of finance, since investors can be more certain of their returns, particularly as the exchange rates were stabilized. Following from this has been a growth in derivatives trading, particularly in Japan and Singapore. Derivatives are pure monetary instruments—options taken on the movement in prices of the underlying financial assets or commodities.

But portfolio capital is very volatile capital. It can represent 'hot money', that moves in an out of an economy quickly according to only slight changes in market sentiment. Thus the rise in importance of this kind of investment can increase the instability in an economy. Given the rapidity of its build up, the potential for instability is magnified as it increases very quickly relative to GDP growth. When the size of foreign inflows is large relative to the size of market capitalization, the potential for uncontrollable financial bubbles increases with higher foreign participation.

### Some macroeconomic considerations

Hot money of this kind not only flows into countries because of the underlying attractiveness of the conditions there, but also because of changes in the condition in the traditional alternative sites of investment, notably the older advanced industrial economies. Thus the recession and slow down in growth in the USA and Europe may have stimulated the move into the East Asian market. As real interest rates increase in the traditional countries, and as they come out of recession, that money can quickly flow out of the East Asian economies if it is allowed to. In relation to this, however, studies have shown that securitized capital flows to the East Asian economies are much less sensitive to changes in US interest rates and economic conditions than are comparable flows into Latin American countries (Das, 1996, pp. 329–30). So hot money has been more stable in East Asia than it has been in Latin America.

Large capital inflows into the South-East Asian economies indicate to a deterioration of their current accounts. Given that the balance of payments overall has to balance, an increase in one part implies a decrease in another; as current flows decrease, capital flows increase. This means that the trading account has moved into deficit, which is exactly what tended to happen as these countries built up large deficits with Japan in particular. They had been importing capital equipment from Japan to feed their development processes. Additionally, capital inflows mean interest payment outflows on the current account, which can also lead to a further deterioration in the current account.

Another consequence is that capital inflows will put pressures for an appreciation of the exchange rates of the countries in receipt of them. The need for capital inflows can also affect domestic interest rates and the money supply, which means that macroeconomic conditions change. The reaction of the monetary authorities to financial inflows is important for the overall consequences and impact on the domestic economy. In particular, the domestic authorities can try to 'sterilize' the inflows in various ways, so as to minimize their impact on the domestic economy (Dean, 1996).

For instance, as money moves in, one reaction would be to try to 'mop it up' by buying up the foreign currency and storing it as reserves. This kind of 'open market' sterilization is the traditional one used by the East Asian economies, but it was in decline in the mid 1990s because it was costly for governments and proved not to be too effective. Instead they were moving towards sterilization by requiring commercial banks to hold much higher reserves, in effect by putting added restrictions on the ability of those banks to create their own liquidity and hence the domestic money supply. Finally, several of the Asia-Pacific countries, such as Indonesia, Malaysia, Singapore, Taiwan and Thailand, have moved pension fund money from commercial banks into the central bank, again to try to restrict the growth in the domestic money supply. These measures are just another manifestation of the guided and controlled nature of the financial systems in these countries. Of course they have also resorted to fiscal policy to try to mop up the added liquidity that capital flows induce.

All these responses point to the difficulty of conducting an independent macroeconomic policy in an increasingly interdependent and integrated international economy. This is the issue taken up more explicitly in the next two sections.

### 4.5 Exchange rate regimes

An important aspect of exchange rate stability is the relationship of the domestic currency to the main currencies used in international trade and finance. All international trade is invoiced in a particular currency, and investments are denominated in the main currencies as well. Thus what is happening to the values of these main currencies relative to one another, and to the currency of the domestic economy, can have a considerable impact on economic performance. It affects the competitiveness of economies, the returns they gain from trade and the worth of their underlying assets.

Table 4.2 puts this into perspective for a number of different indicators of international trading and financial activities. The US dollar remains the dominant currency used in international trade (b). This is also true of official currency holdings (a), and in terms of transactions on the foreign exchange markets (c), and finally in terms of the currency in which assets and liabilities are denominated (d). Thus the US dollar continues as the *lingua franca* of the international economic system (Frankel, 1995). But, as the table indicates, the dollar has lost ground, particularly to the German mark (DM) and the Japanese yen. This is significantly so in the case of the currency of denomination of financial assets and liabilities (d). Here it was the ASEAN economies and China that had been most rapidly diversifying their official reserves by reducing their commitments to the dollar and increasing those of the yen and the DM. However, overall, it is important to note the continued commitment to the US dollar by the East Asian countries, particularly the ASEAN members. There is as yet no strong indication, for instance, that the yen is poised to replace the dollar as the currency of choice for these transactions. More on this in a moment.

# Table 4.2 The international role of the maincurrencies

### (a) Official role

		Share of total official currency holdings (per cent)				
		end 1973		end 1983	end 1995	
US dollar		7	6.1	71.1	61.5	
European currencies <sup>1</sup>		1	4.3	15.8	20.1	
of which:	German mark		7.1	11.7	14.2	
	Yen		0.1	4.9	7.4	
<sup>1</sup> Pound sterlin	ng, German mark, French	franc, Dutcl	h guildei	r.		
Number of currencies linked to:		1983	1994	1994 (percentage	e of world GNP)	
the dollar		34	25	1.5	53	
European currencies (including the ecu)		) 18	19	0.2	25	

### (b) Currency use in international trade

	Share of the main currencies as regards use in international trade						
		1980	1992				
	Percentage of world exports	Internationalization ratio <sup>1</sup>	Percentage of world exports	Internationalization ratio <sup>1</sup>			
US dollar	56.4	4.5	47.6	3.6			
German mark	13.6	1.4	15.5	1.4			
Yen	2.1	0.3	4.8	0.6			

<sup>1</sup> Ratio of world exports denominated in currency relative to that country's exports.

	Breakdown	Breakdown of transactions by currency <sup>1</sup>						
	April 1989	April 1992	April 1995					
US dollar	90	82	83					
German mark	27	40	37					
Yen	27	23	24					
Other	56	55	56					

(c) Transactions on foreign exchange markets

Total as % <sup>2</sup> 200 200	200
---------------------------------	-----

<sup>1</sup> Gross turnover. Daily averages.

 $^2$  Since any transaction on the foreign exchange market involves two currencies, the total of the proportions of transactions involving a given currency is 200 per cent.

(d) Currency in which financial assets and liabilities are denominated

		Share of outstanding international bonds			
		end 1981	end 1992	end 1995	
Dollar		52.6	40.3	34.2	
European currencies		20.2	33.0	37.1	
of which:	German mark	NA	10.0	12.3	
	Yen	6.9	12.4	15.7	

NA=not available

		Share of world private portfolio				
		end 1981	end 1992	end 1995		
Dollar		67.3	46.0	39.8		
European currencies		13.2	35.2	36.9		
of which:	German mark	NA	14.7	15.6		
	Yen	2.2	6.9	11.5		

NA=not available

Source: European Commission (1997, Annex 2, p. 18)

This continued commitment to the US dollar is confirmed if we look at the actual way these economies conduct their exchange rate operations. Table 4.3 specifies the official exchange rate regimes of a selection of East Asian economies as notified to the IMF. But this in fact disguises how they actually conduct their exchange rate operations, since most of the managed/free floats actually involve surrogate alignments to the US dollar (this is measured by calculating the degree of variation of domestic exchange rates to the US dollar and to other currencies; the lower the degree of variation, the closer the effective alignment—see Bénassy-Quéré, 1996, pp. 19–38). Most of the East Asian economies thus anchor their currencies to the dollar and not to the yen. The questions are, why is this the case and what might its consequences be?

To begin with, let us look at the broad pattern of intra-Asia-Pacific trade, as indicated by the data plotted in Figure 4.4. This shows the aggregated direction of intra-Pacific exports: East Asia⇒East Asia compared with East Asia⇒USA. On the basis of this comparison there looks to be a clear move towards the development of an East Asian trading
Hong Kong	Pegged to the US\$
Korea	Managed float <sup>1</sup>
Singapore	Managed float
Taiwan	Not IMF member
Indonesia	Managed float with US\$ reference
Malaysia	Managed float
Philippines	Free float
Thailand	Pegged to a basket <sup>2</sup>
China	Managed float
Myanmar	Pegged to the SDR <sup>3</sup>

## Table 4.3 Official exchange rate regimes (end of1994)

<sup>1</sup>Managed float=currency allowed to float, but managed by the authorities to smooth out fluctuations in value.

<sup>2</sup>Basket=group of currencies weighted by their importance in country's trade.

<sup>3</sup>Special drawing rights (SDR)=an international money equivalent managed by the IMF.

Source: adapted from Bénassy-Quéré (1996, Table 2.2, p. 18)



Figure 4.4 Intra-East Asian trade compared to East Asia-US trade

bloc, at the expense of East Asia-USA trading relationships. It looks, therefore, as though there might be an incentive for the East Asian countries to form a single currency bloc as well, perhaps organized around the yen (Kwan, 1996). However, this is misleading if one does not look more closely at the *disaggregated* trading relationships between the country groups involved, and at their trading relationships beyond the purely Asia-Pacific arena.

Such a richer pattern is demonstrated by the data included in Table 4.4. Here we see the trade relationships between the NIEs, the ASEAN countries, Japan, the USA, Europe and elsewhere for a selection of the East Asian countries separately, 1973 compared to 1993. Within just the intra-Asia-Pacific trade some important trends and changes can be identified:

- 1 intra-NIEs trade is developing at the expense of exports to the US and Japan;
- 2 the NIEs have also become major suppliers for ASEAN countries, at the expense of Japan;
- 3 Japan's trade intensity with East Asia declined, while it increased a little with the USA (see also Chapter 1, Table 1.1).

In addition, the NIEs and ASEAN countries continue to build very important trade relationships with Europe and elsewhere. Thus, broadly speaking, the data in Table 4.4 indicate to the continued *diversity* of the trading relationships of the main East Asian countries, and particularly no

## Table 4.4 Overall patterns of East Asian externaltrade, 1973 and 1993

Exporting country	To the USA		To Japan		To NIEs		To ASEAN		To the EU15		Elsewhere	
	1973	1993	1973	1993	1973	1993	1973	1993	1973	1993	1973	1993
Japan	27.7	29.4	-	-	13.5	19.1	7.4	9.2	14.3	16.2	37.1	26.1
Hong Kong	35.3	22.5	5.7	4.0	5.0	8.9	2.7	3.7	32.6	21.6	18.7	39.3
Korea	33.6	21.3	37.8	14.3	5.7	11.1	2.0	7.7	10.7	12.1	10.2	33.5
Singapore	16.6	21.9	10.3	7.0	9.2	15.6	22.4	23.4	16.2	14.5	25.3	17.6
Taiwan	42.1	28.3	14.8	11.2	9.4	9.9	4.1	7.1	13.0	15.3	16.6	28.2
Indonesia	12.1	13.0	56.3	31.7	14.9	21.1	1.1	3.9	11.5	14.8	4.1	15.5
Malaysia	13.3	21.0	29.7	15.5	16.1	29.4	1.4	5.5	23.0	14.9	16.5	13.7
Philippines	35.2	38.2	40.4	18.9	4.7	12.2	1.2	3.5	13.0	16.4	5.5	10.8
Thailand	10.7	22.2	28.3	17.9	14.8	15.5	12.1	4.3	19.4	18.9	14.7	21.2
China	1.4	29.0	20.1	19.8	19.3	9.0	1.1	3.2	13.8	20.5	44.3	18.5

Orientation of exports by selected East Asian countries (% of total exports of each country)

Origin of imports of selected East Asian countries (% of total imports of each country)

Importing country	From the From USA Japan		From NIEs		From ASEAN		From the EU15		Elsewhere			
	1973	1993	1973	1993	1973	1993	1973	1993	1973	1993	1973	1993
Japan	24.6	22.1	-	-	6.5	11.8	12.1	12.3	9.2	13.8	47.6	40.0
Hong Kong	13.4	9.1	21.1	18.7	10.3	23.3	3.3	6.0	18.7	22.0	33.2	20.9
Korea	27.2	19.3	13.0	26.0	1.8	4.4	8.1	6.6	7.2	13.7	42.7	30.0
Singapore	15.5	14.3	20.6	22.6	5.9	9.6	17.1	21.5	15.9	13.1	25.0	18.9
Taiwan	22.4	20.5	38.8	32.8	4.4	7.3	4.5	6.3	13.9	14.5	16.0	18.6
Indonesia	17.4	10.8	36.5	23.6	9.4	20.5	2.6	3.0	20.5	21.6	13.6	20.5
Malaysia	8.2	16.0	22.1	26.7	13.9	26.1	7.7	5.3	22.0	14.1	26.1	11.8
Philippines	26.9	19.3	33.7	27.2	4.2	17.3	1.5	5.0	13.3	12.6	21.5	18.6
Thailand	13.1	9.1	38.3	31.2	8.1	17.2	1.3	5.7	20.4	16.6	18.8	20.2
China	13.8	11.6	20.3	26.7	3.7	27.7	1.7	3.1	16.3	14.5	44.2	16.4

Source: adapted from Bénassy-Quéré (1996, Box 3.2, p. 39)

clear trend towards a greater reliance on trade just with Japan (indeed, to some extent the opposite). So there was no obvious incentive for these countries to change the currency in which they had traditionally invoiced their trade, and hence to which they anchored their domestic currencies.

What is more, this incentive not to change was all the greater given what had been happening to the relative exchange rates of the three largest international currencies discussed in relationship to Table 4.3 above, at least up until the mid 1990s. As Figure 4.5 shows, the Japanese ven was appreciating against the German mark and particularly against the US dollar after 1975, and most rapidly since the Plaza Accord of 1985 (see Chapter 2). Given that the East Asian currencies were mainly linked to the US dollar, which was depreciating in value, this meant that their currencies had been depreciating as well, compared to the DM and the yen. A depreciating currency makes an economy's exports more attractive and imports into it less so. Thus, in part, one of the reasons for a better than otherwise competitiveness of the East Asian economies has been this downwards pressure on their currency values produced by their anchor to the US dollar. This has made them more competitive in relationship to Japan, and partly explains the changing pattern of their trade relationships referred to in connection to the discussion of the data in Table 4.4. There has clearly been no incentive to switch to a yen anchor whilst that currency had been appreciating in value. From the point of view of the ASEAN countries in particular, macroeconomic stability is more likely to have emerged from a continuation of the existing policy. There was no obvious



#### Figure 4.5 Effective yen, Deutschmark and US dollar exchange rates (1975–94)

Source: Kwan (1996, Figure 1, p. 4)

yen bloc forming in East Asia. However, things did begin to change in the mid 1990s. The dollar began to *appreciate* against the yen after 1995, so the yen began to *depreciate* in value relatively. This could have longer-term consequences in changing the patterns so far described if the depreciation of the yen continues.

Of course, these potential favourable exchange rate movements are not the only determinant of the underlying competitiveness of an economy or its overall balance of payments position. They only represent pressures and tendencies, which can be overwhelmed by other trends and underlying structural features. As mentioned above, capital flows as well as trade flows determine the overall balance of payments, and a number of the ASEAN economies experienced growing trade deficits as they stocked up with capital goods, despite the favourable pressures on their exchange rates (see Table 4.5 in Section 4.6). In an increasingly open and internationalized economy, competitiveness also attracts capital inflows, which puts upward pressures on a currency. In terms of assets, holding these denominated in a currency that is appreciating in value will mean they are worth more in terms of the home currency, which could account for the move by the East Asian countries away from dollar denominated assets and into those denominated in yen, as mentioned above. So there are always trade-offs, conflicting pressures and counter-dilemmas involved with international financial activity. The next section focuses upon the issue of the possible growing financial interdependence and integration of the Asia-Pacific economies.

#### 4.6 An Asia-Pacific financial integration?

The economies of the Asia-Pacific region are increasingly becoming interlinked as trade and financial activity expands across national borders. This interlinking is enhanced by the growth of the real integration of the economies as production networks stretch out across the Pacific Ocean (see Chapters 2, 8, 10 and 14), but this section concentrates upon the growth of purely financial integration. In part this is a truly global phenomenon as the financial markets of the international economy expand their range and integration. Thus in some ways it is difficult to discuss the nature of just trans-Asia-Pacific financial integration, since this is itself set within the growth of what is for many a global phenomenon. There are as many links between the Asia-Pacific economies and those financial markets that lie *beyond* the region as there are between the markets of those economies *within* the region. Thus, in part, the discussion must be conducted at two levels: the first in respect to the international economy in general and the second in respect to just the Asia-Pacific region.

As already indicated in this chapter there are plenty of examples of financial interlinkages between the economies of the region. But *financial interlinkages* are not quite the same as *financial integration*. The first has been a feature of economies ever since national borders were established. International financial interlinkages express the way two or more economies are linked as a financial transaction takes place between them. It is a fairly simple economic relationship: a discrete one-off exchange, and one that does not necessarily imply a change in the nature of the agents or economies involved with that transaction. However, financial integration implies the establishment of a much deeper and more continuous relationship that somehow binds the agents or economies together and changes the way they operate. It would establish a newly integrated entity, the parts of which operate in conjunction with one another. The question is, then, how to measure whether this process of integration has actually happened in the international financial system.

Traditionally there are a number of ways this is done or approached. The first is to look at the relationship between aggregate national savings and domestic investments in an economy, to see how closely they are correlated. If they are closely correlated (as one changes so does the other), then the argument is that the international financial system is *not* that integrated. Financial integration allows for flows between countries, so that domestic economies remain relatively closed to these cross border flows. The same applies with another measure, consumption patterns between countries. These are the flip-side of savings and investment. A high degree of financial integration would imply that consumers have access to finance on the international markets and can thus vary their consumption patterns accordingly so as to match lifetime earnings with lifetime consumption for different countries. The continued differences in lifetime consumption patterns would indicate to a continued non-integration of the financial systems.

In fact, attempts to establish whether these measures operate for the APEC countries have been largely unsuccessful (Kearney, 1996, p. 352). In part this is because the testing techniques are not robust enough to sort out the issue properly, but at another level it indicates to the continued non-integrations of the economies in question. Indeed, this

result is one confirmed at the global level by looking at a wider set of OECD countries (Thompson, 1997, Table 2, p. 164).

Other measures can also be used to test the integration hypothesis. One is to investigate the degree of real interest rate convergence between economies. If the financial systems were closely integrated, then real interest rates would also be at a similar level. This implies that the return on assets will be converging amongst the economies, as capital moves freely between them. Another related measure would be the way stock market prices move in similar directions in the different markets, or the way adjustments in the exchange rates take place so as to equalize the real convergence between different countries.

In fact, all of these measures are extremely difficult to properly operationalize because of a number of technical difficulties in making





Source: The Observer, 31 August 1997

valid comparisons between different countries' systems (Kearney, 1996). This is nowhere more so than as between the Asia-Pacific countries as a whole, or even between a subset of the East Asian/ASEAN ones. All the diverse mechanisms of 'financial repression' discussed above serve to undermine a legitimate comparison.

Although not as rigorous, there is, however, some other rather more superficial evidence that integration is happening between the East Asian financial systems. In July and August 1997, for instance, a serious financial crisis seemed to hit a set of the ASEAN economies more or less simultaneously, and overflowed into Singapore and Korea as well. In technical parlance, their markets suffered from 'contagion' (as from a contagious

disease), where the crisis in one market quickly spreads to others. The way this affected the markets is shown in Figure 4.6. Stock market prices and currency values fell as capital tried to move out of the economies simultaneously. They seemed to suffer a genuine co-ordinated attack from disillusioned foreign investors.

But what were the reasons for this attack? An answer to this could tell us something more general about the state of these economies in the late 1990s. The authorities in some countries tried to blame it on 'foreign speculators', but the real sources of the crisis lay elsewhere, and mainly on the domestic front. Financial panics of this kind happen quickly and dramatically after pressure has built up over a longer period of time. The first point to recall is that the countries in question experienced an unexpected downturn in their growth rates in 1995–96 (with the exception of the Philippines), and an added sudden fall in their export growth rates (Chapter 1). Figure 4.7 reminds us of the trajectory of growth rates over the 1990s, clearly indicating to the fall in 1996. Second, as mentioned above, most of the countries in question were running a current account deficit, indicating to their underlying 'uncompetitiveness'. This is shown in Table 4.5. What is more, countries such as Malaysia and Thailand were suffering from a considerable over-heating of their economies, given the large inflows of portfolio capital and FDI investment, much of which had found its way into construction and real estate developments and not into directly productive activity. Finally, the consequences of the dollar appreciation after 1995 were also important, since these economies had attached their currencies to a now increasingly *uncompetitive* currency. Thus all the ingredients were present for a crisis of this kind to emerge, most of which were home-grown and to do with the state of the real economies.





Source: The Observer, 31 August 1997

But what does this say about financial integration? First, the coincidental downturn in these financial systems need not *necessarily* indicate to a high degree of such integration *amongst themselves* since the immediate stimulus for it came from a common 'outside', amongst an amorphous mass of foreign investors. Second, although Indonesia, Thailand, the Philippines and Malaysia did demonstrate a large current account deficit, Korea (not shown in these figures, but which also suffered some contagion) did not, and Singapore shows a massive current account surplus. (Note also

	1993	1994	1995	1996 (estimate)
USA	-1.5	-2.2	-2.1	-2.0
Japan	+3.1	+2.8	+2.2	+1.9
Australia	-3.7	-4.8	-5.3	-4.6
Indonesia	-1.6	-1.7	-3.7	-4.1
Korea	+0.1	-1.2	-2.0	-1.5
Malaysia	-4.6	-5.9	-8.5	-7.9
Mexico	-6.4	-7.7	-0.1	-0.9
Philippines	-5.5	-4.4	-3.3	-2.7
Singapore	+9.1	+17.3	+18.3	+14.7
Taiwan	+3.0	+2.6	+1.6	+1.8
Thailand	-5.4	-5.7	-7.1	-7.0

## Table 4.5 Selected Asia-Pacific countries' current accounts 1993–96 (% of GDP)

Note: -in deficit; +in surplus

Source: derived from International Monetary Fund (1996, Table 7, p. 37)

that Taiwan had such a surplus and was not so affected by contagion.) What is more, much larger current account deficits were demonstrated by Australia, but it was not involved. Thus if there were any genuine integration, it probably involved only a few countries in the East Asian region, was not strongly embedded, and certainly did not involve the entire Asia-Pacific economy.

Finally, what might have been the consequences of these events in the summer of 1997? In part a response to this depends upon an issue raised at the beginning of this chapter.

Those who look upon the financial aspects of an economy as being the most important would view these events as very significant in their own terms. They demonstrated a lack of confidence by the international financial community in the future of these economies. They were also used to press for further deregulation on those economies, particularly in respect to the freedom of capital movement *out* of the countries concerned. One of the reasons advanced for the growing lack of confidence of international investors was that

financial capital could not be easily enough withdrawn for the ASEAN economies, although it was always welcomed as an inward investment.

On the other hand, those who remain wedded to the real side of an economy as providing the key indicators of success would probably remain a little more ambiguous in their judgements. The causes of the financial crisis looked to be real ones, associated with a downturn in real growth rates (see Figure 4.7). That was bad news. But growth rates were still running at around 6 per cent, an exceptionally buoyant position compared to the growth rates being experienced in the advanced economies of North America and Western Europe. The question was whether the financial crisis would further exacerbate the real downturn, or whether a rapid recovery was possible as the effects of the financial downturn were quickly reversed (see also Chapter 3).

#### 4.7 Conclusion

This chapter has reviewed the role of the financial systems in the economies of the Asia-Pacific region, concentrating upon the high performing economies of East Asia. The initial problem set for the chapter was to explain the very high investment rates found in these economies, and to assess their possible contribution to overall economic performance. In particular, here was a debate about the relative merits of a 'marketfriendly' as opposed to a 'government-friendly' approach to economic policy making. The conclusion was that systematic government involvement in the financial systems did not seem to have hindered these economies in their development processes as much as the market-friendly theorists might have expected.

However, times are changing. The possibility of the governments maintaining a tight control over their financial systems in the traditional way looks unlikely as the world's economies internationalize further and the tide of neo-liberal policy advice and pressures for deregulation continue to gather force. The economies of Asia-Pacific are probably in a unique period of transformation in this respect.

But we should not expect radical structural change quickly or precipitously. The economies and their financial systems are tied into a complex set of domestic and international economic relationships, not all of which are driving towards disruption of the existing patterns of financial organization. There remains plenty of scope for a consolidation around existing patterns and practices. In addition, the extent of genuine financial integration (as opposed to financial interlinkaging) still looks to be marginal, again leaving opportunities for the re-establishment of effective national responses to the changing international environment.

#### References

- Arndt, H.W. (1983) 'Financial development in Asia', Asian Development Review, vol. 1, no. 1, pp. 86–100.
- Bénassy-Quéré, A. (1996) 'Potentialities and opportunities of the Euro as an international currency', *Economic Papers No.115*, Brussels, European Commission.
- Cole, B.C. (1988) 'Financial development in Asia', Asian-Pacific Economic Literature, vol. 2, no. 2, pp. 26–47.

- Das, D.K. (1996) 'Emerging markets and macroeconomic stabilization: with special reference to Asia-Pacific economies', *Journal of the Asia-Pacific Economy*, vol. 1, no. 3, pp. 319–46.
- Dean, J.W. (1996) 'Recent capital flows to Asia-Pacific countries: trade offs and dilemmas', *Journal of the Asia-Pacific Economy*, vol. 1, no. 3, pp. 287–317.
- European Commission (1997) 'External aspects of economic and monetary union', *Euro-Paper No 1*, July, Brussels, European Commission.
- Frankel, J.A. (1995) 'Still the lingua franca: the exaggerated death of the dollar', *Foreign Affairs*, vol. 74, no. 4, pp. 9–16.
- Fry, M.J. (1995) 'Financial development in Asia: some analytical issues', Asian-Pacific Economic Literature, vol. 9, no. 1, pp. 40–57.
- Harris, L. (1988) 'Financial reform and economic growth: a new interpretation of South Korea's experience' in Harris, L., Coakley, J., Croasdale, M. and Evans, T. (eds) *New Perspectives on the Financial System*, London, Groom Helm.

International Monetary Fund (IMF) (1996) World Economic Outlook, May, Washington, DC, IMF.

- Kearney, C. (1996) 'International financial integration: measures and policy implications', *Journal of the Asia-Pacific Economy*, vol. 1, no. 3, pp. 347–64.
- Kwan, C.H. (1996) 'A yen bloc in Asia: an integrative approach', *Journal of the Asia-Pacific Economy*', vol. 1, no. 1, pp. 1–21.
- McKinnon, R.I. (1973) *Money and Capital in Economic Development*, Washington, DC, The Brookings Institution.
- Pacific Economic Cooperation Council (1994) Pacific Economic Development Report 1995: Advancing Regional Integration, Pacific Economic Cooperation Council, Singapore.
- Shaw, E.S. (1973) *Financial Deepening in Economic Development*, Oxford, Oxford University Press.
- Thompson, G.F. (1997) 'Globalization and the possibilities for domestic economic policy', *Internationale Politik und Gesellschaft*, no. 2, pp. 161–71.
- Wade, R. (1985) 'East Asian financial systems as a challenge to economics: lessons from Taiwan', *California Management Review*, vol. XXVII, no. 4, pp. 106–27.
- White, G. (1998) 'Politics of welfare' in Maidment, R., Goldblatt, D. and Mitchell, J. (eds) *Governance in the Pacific*, London, Routledge in association with The Open University.
- World Bank (1993) The Asian Miracle, Oxford, Oxford University Press.
- Yang, L.S. (1952) Money and Credit in China: A Short History, Cambridge, Mass, Harvard University Press.

## CHAPTER 5 Engendering the 'economic miracle': the labour market in the Asia-Pacific

Lucie Cheng and Ping-chun Hsiung

#### **5.1 Introduction**

Since the late 1960s, following the phenomenal recovery of the Japanese economy after the Second World War, a number of East-Asian countries have seen unprecedented economic growth. The great diversity in area and population size, culture, social organization and historical legacy among these countries continues to frustrate scholars who struggle to find 'grand theories' or linear explanations for these 'economic miracles'. The rise of Asia-Pacific and the corresponding decline of the hegemonic 'West' provide a fertile ground for competing theories of capitalist economic development that generally focus on one or more of the following: cultural values, social institutions, a strong state, geopolitical factors and the world economic system. While the role of labour figures prominently in these theories, it is generally more taken for granted than given serious analysis. When it is examined, labour is either genderless or male. It is not until very recently, with the increasing awareness of the gendered nature of relationships between economic policy and labour use and the penetration of feminist thought into Asia-Pacific studies, that we began to see discussions of the different roles of male and female labour in the East-Asian 'economic miracle' and of how this 'miracle' has impacted on male and female labour differently.

This chapter will summarize these discussions, focusing especially on women's labour and the strategies pursued by selected Asia-Pacific nations and the consequences of those strategies for gender equality in terms of women's access to jobs, entrepreneurship, mobility and security. We will show that while women are absorbed by global capitalism into the labour force, their reproductive roles have not diminished. Women are also resisting global capitalism in a variety of forms. We argue that only by 'engendering' the 'economic miracle' can we reach an adequate understanding of the rise of the Asia-Pacific.



Workers in a surgical glove factory, Penang, Malaysia

#### 5.2 Economic development and gender equality

Discussions of the relationship between economic development and gender equality the world over have been controversial. In general they revolve around two opposing arguments. One side contends that notwithstanding the fact that there is still some way to go, economic development has created jobs for women which have led to independent incomes and self-confidence. The integration of women into the labour force has been advantageous in achieving equality with men (Jaquette, 1982). The general growth of the economy has 'trickled down' more or less automatically, so that those previously disadvantaged have also received some reward.

The opposing view sees economic development as somewhat detrimental to gender equality because it either marginalizes women, pushing them into secondary, peripheral or informal jobs that earn a fraction of men's wages (Boserup, 1970), or leads to a double burden on women by proletarianizing their productive roles while insisting on the maintenance of their conventional reproductive functions (Mies, 1986). Thus there is no automatic advantage accruing to women in particular, but simply a continuation, or even a deterioration, of their relatively disadvantaged status.

This chapter evaluates these three positions  $vis-\dot{a}-vis$  women's labour in selected countries of the Asia-Pacific. Since other chapters discuss women's subsistence and reproductive work, the focus here is on a gender analysis of women's income-generating activities in both traditional and non-traditional economic sectors (see also Chapters 11 and 12). Through a systematic examination of available statistics, the chapter discusses whether or not women have achieved, or are closer to achieving, parity in the labour market in terms of gender division of paid work, comparable worth and access to

promotion and mobility. It also considers intra- and inter-country differences and the Asia-Pacific as a region in comparison with the rest of the world.

#### Women and men in paid work: a narrowing divide?

Over the last three and a half decades, as the economy of the Asia-Pacific region has grown, an increasing number of women have been drawn into the labour force. This pattern of increase is similar to the history of the industrialized countries in the West. Worldwide, more than 40 per cent of women over fifteen years of age are now in the labour force, the percentage having grown almost twice as fast as that of the male labour force. Of the ten Asia-Pacific countries examined, only two, the Philippines and Indonesia, show a lower female labour force participation rate than the worldwide average.

Despite the overall increase in women's labour force participation, Figure 5.1 shows that the percentage of women in the labour force consistently lags far behind that of men. Except for Thailand and the People's Republic of China (PRC), where differences between the two rates are less than 20 percentage points, men's labour force participation rates exceed women's by a large margin. In Singapore, for example, men's rate was higher than women's by 52 percentage points in 1970, and even in 1995 the difference was still more than 30 points.

Figure 5.1 also indicates that while the overall rates for men and women have generally increased, the patterns are by no means uniform. For the more industrially advanced countries such as Japan, Singapore and Hong Kong, men's rates show a downward trend until 1995, while women's rates increased continuously during the same period. Taiwan and Korea show some slight variations, where women's rates first increased from 1970 to 1990 and then declined in 1995. For most late-comer countries, such as the Philippines and Indonesia, men's rates increased continuously throughout the period whereas women's rates show a marked decline from 1990 to 1995. For Malaysia both men and women's labour force participation increased sharply, while for Thailand both declined during the last five-year period. The PRC has the highest labour force participation rate for women (80 per cent) and men (96 per cent), partly due to a difference in its measurement system. Unlike other countries in the region that report labour force participation rates based on the population of 15 years and older, the PRC uses 16 to 59 years old for males and 16 to 54 for females as bases for their calculation.

Additional statistical data also reveal a persistent gender division of labour by industry and inequalities in terms of occupational distribution and remuneration. As Table 5.1 indicates, for all countries, women are concentrated in manufacturing and trade and service industries, but are almost absent from the construction, mining and utilities industries.



Figure 5.1 Labour force participation rates for those fifteen years and older by gender (per cent)

Sources: International Labour Organization (various years); 1995 data from The World Bank (1996, p. 3); Taiwan data from Republic of China (various years; 1993a)

### Table 5.1 Distribution of women's employment by industry (women as a percentage of total employment in that sector)

Country	Year	Economic sector									
		А	В	С	D	Е	F	G	Н	Ι	J
Japan	1970	50.85	15.00	37.62	10.71	13.45	45.95	12.96	46.21	44.30	28.57
	1980	49.05	9.09	38.55	13.33	14.03	46.15	11.71	42.45	45.95	30.77
	1990	47.78	16.67	39.53	13.33	16.33	48.69	14.63	44.96	47.93	36.67
Singapore	1970	21.36	9.46	33.63	7.00	6.53	18.96	4.99	33.24	16.54	44.65
	1980	25.17	18.75	46.10	11.34	9.09	35.23	16.98	46.96	33.96	5.26
	1990	15.38	12.50	46.78	14.93	6.88	39.59	21.43	48.06	51.15	0.00
Taiwan	1970	32.31	11.55	29.37	9.43	3.27	20.78	10.57	NA	15.84	18.62
	1979	36.35	14.54	44.21	9.55	9.00	14.22	13.81	40.97	25.28	17.39
	1986	NA	15.51	49.58	12.12	11.69	46.06	17.48	40.71	55.03	NA
Hong Kong	1970	23.18	11.65	12.70	6.41	2.93	21.96	8.22	NA	12.55	28.22
	1980	25.09	NA	50.21	8.60	4.31	30.37	20.83	38.70	44.82	23.79
	1990	29.96	0.00	41.87	8.99	4.42	39.39	14.71	42.77	48.47	0.00
Republic of	1971	12.57	3.04	23.55	0.85	0.94	27.42	1.93	NA	24.40	46.67
Korea	1980	43.77	8.87	39.28	11.63	8.56	48.61	9.05	31.02	34.49	0.00
	1990	46.18	10.13	42.44	12.86	10.10	52.96	8.67	37.84	43.01	0.00
Philippines	1970	19.18	5.77	54.14	3.03	0.68	54.71	2.21	NA	55.78	25.68
	1977	18.57	5.56	43.88	16.70	1.83	61.56	4.89	NA	50.04	24.80
	1990	25.18	9.02	45.75	17.58	1.85	63.69	3.96	39.41	55.59	26.67
Indonesia	1971	18.06	4.77	56.48	14.08	0.79	43.45	1.51	17.67	23.92	NA
	1980	31.28	17.74	42.40	12.45	3.15	47.52	1.68	15.46	31.77	43.31
	1990	30.06	16.22	45.28	8.09	2.88	52.20	2.30	26.07	35.36	55.00
Malaysia	1970	38.00	12.60	28.13	5.35	6.78	18.23	4.36	NA	28.90	47.83
	1980	38.96	10.36	40.07	6.98	7.12	29.27	6.26	29.54	29.42	32.83
	1990	34.52	11.38	47.68	9.45	4.65	37.50	11.46	36.60	39.49	0.00

Thailand	1970	14.50	NA	42.06	NA	10.00	51.70	1.28	NA	44.92	23.53
	1980	50.44	30.47	46.61	14.90	18.54	58.71	7.84	35.32	41.25	41.58
	1990	47.38	14.84	49.92	20.26	16.64	53.77	10.62	NA	51.87	40.93
China	1980	36.17	22.69	42.21	41.53	24.79	40.03	21.43	20.63	54.13	30.26
	1990	37.44	25.28	44.34	35.94	13.28	47.06	21.45	35.15	34.73	32.39

Note: A=agriculture, forestry and fishing; B=mining and quarrying; C=manufacturing;

D=electricity, gas and water; E=construction; F=trade and commerce; G=transport, storage and communication; H=financing, insurance, real estate and business services; I=community, social and personal services; J=non-classified activities.

NA=not available.

Sources: International Labour Office (various years); Taiwan data from Republic of China (various years; 1993a); China data from People's Republic of China (various years)

In 1990, for example, women comprised more than 40 per cent of all manufacturing workers in all countries except Japan (column C); and less than 20 per cent in mining except China (column B). When Asia-Pacific countries are compared with each other, we find that as the economy of the country shifts away from labour-intensive manufacturing women's pattern of employment also changes. Their concentration in manufacturing decreases while that in trade and services increases. This is clear with Japan and the four 'little dragons': Singapore, Hong Kong, Taiwan and South Korea.

Economic development has not yet brought gender equality in paid work. Similar to the worldwide position, women in the Asia-Pacific continue to be under-represented in the more prestigious and well-paid occupations and over-represented in the lower status and less well-paid ones. The extent of inequality varies across countries in the region. Take managers, for example. The most blatant case is represented by South Korea, where women made up only 4.2 per cent of all managers in 1993. Even in the best case, Singapore, male outnumber female managers by almost three to one. A similar discrepancy between men and women in the managerial category is observed in 1990 for all countries (Figure 5.2). In contrast, proportionately more women than men are found among clerical and service workers in most of the countries.

The professional category indicates an interesting though somewhat misleading situation. As Figure 5.2 shows, for almost all countries proportionately more women than men are employed in this category. This is especially glaring for the Philippines and Malaysia. There are two reasons for this: the rising levels of education for women in all Asia-Pacific countries and the increasing demand for educated women to fill occupations that are extensions of traditional female roles, such as nursing and elementary and kindergarten teaching. These occupations are included in the professional category shown in Figure 5.2, and they are at the lowest rung of the professional ladder. Both the similarities and diversities in gender distribution of occupations are clearly demonstrated by the varying heights of the bars.

As in Western industrialized countries, women's wages in Asia-Pacific are only a fraction of those of their male counterparts. Generally speaking though, women's access to jobs in the modern sectors tends to narrow the gender gap in earnings. This is indicated by data in Table 5.2. In the USA, while women's wages were only 64 per cent of their

male counterparts in 1980, by 1990 they rose to 70 per cent. In the Asia-Pacific the most dramatic case of wage gap reduction is found in Malaysia, where the female/male wage ratio was reduced by 12 percentage points in ten years. In some countries, and for some industries, the gap between male and female earnings has not only persisted but has widened over the past two decades. An example is provided by Taiwan. In five out of nine industries—agriculture, mining, manufacturing, construction and commerce—the gains in wage equality observed in the 1970s had been largely eroded by the 1980s.



Figure 5.2 Gender distribution of employment by occupation, 1990 (shown as a percentage of either total male or female employment)

Sources: International Labour Organization (1992); Taiwan data from Republic of China (1993a); China data from All China Women Federation and Shaanxi Provincial Women Federation (1991); Hong Kong data from Hong Kong Census and Statistics Department (1991)

	/			
Country	Year	Female/male wage	Year	Female/male wage
Japan	1982	43.1	1991	42.9
Taiwan	1988	64.5	1992	67.1
Hong Kong	1987	74.0	1996	80.0
Republic of Korea	1984	41.9	1988	51.0
Philippines	1978	70.9	1988	80.0
Indonesia	1986	55.6	1992	60.0
Malaysia	1973	56.8	1984	69.3
Thailand	1980	73.5	1990	79.8
China	1978	82.3	1988	85.6
USA	1980	64.0	1990	70.0
Australia			1987	87.0

# Table 5.2 Ratio of female to male earnings inselected Asia-Pacific countries, various years (percent)

Sources: China data from All China Women Federation and Shaanxi Provincial Women Federation (1991, pp. 318–19); Hong Kong data (excluding unpaid family workers) from Hong Kong Census and Statistics Department (1997, p. 95); Taiwan data from Republic of China (1993a, p. 32); others from The World Bank (1996, p. 21)

While part of the wage gap can be explained by differences in educational attainment and work experience between men and women, studies failed to account for this persistent wage inequality by human capital factors alone. The overall gender gap in wages has been attributed to differential access to education and jobs, interruptions in women's employment due to child rearing and other family responsibilities, and to the seniority system in industries. Korean women are paid less than men with the same education and in the same industry. Moreover, the differential increases rather than declines as women gain more education. Scholars maintain that this phenomenon is largely a function of the under-employment of women caused by pervasive discrimination.

The overall diversity in labour force participation among countries in the Asia-Pacific is clearly reflected in the hierarchical nature of the wage structure. Figure 5.3 shows the huge difference in earnings between an engineer in Hong Kong and Tokyo on the one hand, and one in Manila and Jakarta on the other. The most dramatic is the wage differential between a female unskilled industrial worker in Jakarta and an engineer in Hong Kong. The latter's earnings are more than 30 times the former. The huge variance

in earnings is a major driving force towards international labour migration in the region (see Chapter 2).

In summary, despite the increasing participation of women in the labour force, gender division of labour has persisted and women are still concentrated in less prestigious, less responsible and less well-paid jobs. When women and men both work full time in the same industry, women still receive lower wages than men. Beyond these similarities among countries, there are significant variations. No single theory can explain the complexities. Commonly accepted explanations include the different timing and trajectory of economic growth in different countries, different



# Figure 5.3 Wage hierarchy in the Asia-Pacific, 1994 (US dollars per year)

Source: The World Bank (1996, p. 4)

institutional arrangements and different cultural and ideological systems. Our data do not reflect women's work in general, since they seriously undercount those in the nontraditional sectors or in the informal economy and disregard completely those engaged in unpaid work, the two categories where women are over-represented. In terms of gender differences in paid work, the general pattern observed for this region indicates a persistent but narrowing, rather than widening, divide.

Education: women's capital resources or entrapment?

Scholars and other observers have attributed the 'economic miracle' of Japan and the newly industrialized countries in the Asia-Pacific to a well-educated, hard-working and disciplined labour force. While women's educational attainment has improved during the last three decades, for all countries, as Table 5.3 shows, women still lag behind men in terms of access to college education. However, again, there are important differences among the countries. For comparable years, Korea has the best educated labour force

regardless of gender, while the People's Republic of China has the least. In terms of gender, the largest difference in educational attainment between men and women is observed in Indonesia. Similar to what has been reported in the West, gender continues to be a significant variable in tracking students into specific subjects and fields of study. While male students are counselled towards careers in science and engineering, women students are channelled towards the humanities.

Country	Year		Women		Men			
		1st level	2nd level	3rd level	1st level	2nd level	3rd level	
Japan	1980	51.40	41.54	7.06	48.17	38.92	12.91	
Singapore	1980	58.46	37.75	3.79	59.18	35.28	5.54	
Taiwan	1992	40.64	49.32	10.04	33.60	51.59	14.81	
Hong Kong	1996	26.20	58.36	15.44	23.91	57.99	18.10	
Republic of Korea	1990	46.63	42.78	10.58	41.35	39.49	19.16	
Indonesia	1992	73.34	25.23	1.43	66.73	25.91	7.36	
Malaysia	1990	61.14	36.08	2.78	61.72	34.99	3.29	
China	1992	70.69	28.38	0.92	66.87	31.56	1.57	

# Table 5.3 Percentage distribution of educationallevel by gender

Note: 1st level=completion of compulsory education, 6 to 9 years of schooling; 2nd level= completion of middle school or vocational school, 12–14 years; 3rd level=15 years and above Sources: Taiwan data from Republic of China (1993b, pp. 90–5); Hong Kong data from Hong Kong Census and Statistics Department (1997, p. 67); other data from The World Bank (1995)

As more and more women become college educated, they often face serious barriers in comparable employment and advancement. Newspapers report that many bright young women in East Asia, notably in Japan, are trying to escape corporate sexism and bleak job prospects by finding work overseas. At the same time, an increasing number of East-Asian women have successfully broken into the corporate and political worlds. The implication of education for gender equality in employment and promotion remains controversial.

Two important aspects of the relationship between gender, education and economic development in the Asia-Pacific experience bear specific scrutiny. On the one hand, gender inequality is perpetuated in the process and content of schooling. At the same time, however, the educational system has been transformed to facilitate a capitalist production that relies heavily on female labour. Let us examine these two aspects one by one.

Similar to the West, schooling in the newly industrialized economies (NIEs) is an engendered socialization process where male and female students are taught to develop gender-proper roles, behaviours and career aspirations. Textbooks at all levels continue to portray stereotypical images of women as being caring and loving and men as strong and

independent. Nurses, mothers and school teachers are represented by female figures while men are the doctors, soldiers, pilots and engineers. We therefore caution against an overly optimistic interpretation of numerical increases in women's educational access, since access alone does not automatically or necessarily bring about a more gender equal society, unless the process and content of schooling are altered.

An even more pertinent aspect of education in relation to the region's economic growth is the emergence of a partnership between school and factory systems that tends to entrap students, especially rural females, into early labour force participation. This plays out in two distinctive ways. First, the school is converted from an educational institution into a brokerage agent which prepares its students for the employment demands of the newly flourishing manufacturing sector. In April or May each year, factories send representatives out to junior middle schools in rural areas to recruit potential workers, usually 15 or 16 year olds.



Small industry car-seat cover manufacturing in Yen Bai, North Vietnam

The representatives, armed with official endorsements from school principals, placement counsellors and teachers, meet with the local elites. As a result of such networking, parents of school graduates who give consent to have their children work in the factory often believe they are doing something in accordance with the wishes of the schools and the local authorities. Sometimes, the new factory pays a placement counsellor from the

school to accompany the new recruits on their first trip from home to work. At other times, the principal or a teacher is brought in to visit and comfort the youngsters after they have worked in the factory for a month or so. The visit often coincides with the period when many of the young workers are most inclined to quit, either because of homesickness or the hardship of factory work.

The transformation of the school into a recruiting agency for the factory suggests that capitalist production has tapped into the pre-existing hierarchical structure of the school system to meet its need for a stable supply of malleable labour. In other words, the school system has been integrated into the NIEs developmental process not only on the basis of its formal, educational function, but also as an institution that is capable of extending its influence and control over its students even after their graduation.

Furthermore, since the mid 1960s, under state support, co-operative programmes have been widely set up at the junior and senior middle school levels through which students engage in full-time factory work during the day and take part-time courses for educational credits at night. Under these programmes, factory employers pay part of the tuition fees for their student workers. In exchange, a portion of the student workers' monthly pay is kept by the employer. The student worker can retrieve the money only upon completion of his/her three-year tenure at the factory. That is, a student worker automatically loses part of his/her wages for changing jobs within a three-year period.

These co-operative programmes serve multiple functions. First, the prospect of earning a high school diploma while working in the factory is enormously appealing to student workers, especially female junior middle school graduates in rural areas whose parents were mostly unwilling to support them for further schooling. Second, the programme helps to stabilize and sometimes discipline a capitalist labour force through various mechanisms. In particular, the three-year contractual arrangement with an automatic wage forfeiture proviso ties the worker to the employer. This is especially favourable to the factory employer, since student workers are quickly worn out by the repetitive and monotonous routine and without the proviso may decide to leave the factory. Many female student workers cite the aspiration of getting a high school diploma as their main reason for enduring the type of factory employment which otherwise would be unbearable. The co-operative programme is advantageous to the employer in yet another way. Whenever conflict arises on the shop floor, the school authority is called upon to resolve it. This ranges from negotiating wages and work schedules to settling infights among student workers. It is not uncommon for teachers to side with the employer, rather than the student workers, when it comes to capitalist discipline. The co-operative programme has been widely adopted by private vocational schools in the region which otherwise would have difficulty in retaining high levels of student enrolment.

The above discussion suggests that capitalist development in the East-Asian NIEs has been nourished by cultural norms that value education. Such symbiotic relationships are embedded in the way that educational and manufacturing systems sustain each other.

#### The role of the state: perpetuating the double burden?

In studies of the origins of the economic development of the NIEs, the state is often recognized for its strong, active involvement in financial policies, market management and labour regulation. A less articulated area has been the implication of various regulations and policies on gender equality. Here we focus our discussion on the gender implications of the export-oriented growth strategy of development pursued by Japan and the NIEs.

In the 1960s, all the first-tier NIEs shifted from import substitution to export-led industrialization (see Chapter 2). The main objectives were to draw their under-employed and unemployed population into the export sectors, while at the same time overcoming limitations that had prevented them from further pursuing an import substitution strategy. Policy makers at the time engaged heavily in debates about the merits of such policy shifts, because a labour-intensive, export-led industrialization entails suppressing labour's wages for the enrichment of foreign investors and subjecting workers in the Export Processing Zones (EPZs) to abuses and exploitation. (EPZs are special areas giving tax and other advantage to companies operating there.) Taiwan's experience provides an example.

The idea of EPZs emerged in Taiwan in 1956. It took ten years before a proposal was approved and the actual construction of the first EPZ began. Li, an economist heavily involved in Taiwan's economic policy formation process, recalls debates at the time. According to Li, supporters of the EPZ downplayed the prospect of foreign investors' exploitation by arguing that 'it was better to provide a job to someone who would otherwise be unemployed than to be concerned with the profits of the employer' (Li, 1985). In retrospect, Li states, 'it was largely an investor's market in the 1960s and 1970s, whether one likes the fact or not' (ibid.).

It is beyond the scope of this chapter to discuss whether or not a labour-intensive, export-oriented strategy was the government's best choice at the time. We do want to point out, however, that the consequences of such a choice have not been gender neutral. The manufacturing sector, which has been the motor of the NIEs export-led growth, has relied heavily on female labour force participation; women have been over-represented in the manufacturing sector (see Table 5.1). This has been the case for every listed country. For the four 'little dragons'—Singapore, Hong Kong, Tawain and South Korea—between one-fifth and two-thirds of the entire female labour force participated in manufacturing production in 1980.

Since the state's own political legitimacy is founded on continuing economic growth and social stability, it is not surprising to find that labour regulation and developmental programmes often have these twin goals in mind. Two programmes of the ruling Nationalist Party in Taiwan aimed at incorporating married women into its manufacturing production are illustrative.

In distinction to South Korea and Singapore, where large-scale factories employing mainly young single girls have dominated the manufacturing sectors, more than 85 per cent of factories in Taiwan's export sector have fewer than 30 workers, with married women as their main source of labour. The Living Rooms as Factories programme was part of the government's effort to target an under-utilized labour reserve—married women and their children—for informal homework or to turn their own living rooms into factory production sites. Its parallel programme, Mothers' Workshops, on the other hand, emphasized the importance of married women's traditional familial responsibilities as mothers, wives and daughters-in-law. Taken together, these two programmes constitute the state's intentional effort to facilitate Taiwan's export-led industrialization on the one hand, and to reconcile the potential conflict between female labour force participation

and women's roles in the family on the other. Married women are incorporated into the labour force not as regular workers but as homeworkers, and the subsidiary and supplementary character of their employment reinforces their subordinate and dependent status at work and in the family.

#### 5.3 Gender, entrepreneurship and the family firm

In Asia-Pacific, entrepreneurship, the key to capitalist development, is often linked to the family firm (see Chapter 9). Scholars have considered Asia-Pacific familism as both a detriment and as an advantage to economic development. Many have pointed out that Asia-Pacific businesses rely on kinship networks and familial ties for financial and labour needs (Redding, 1993). Commitment to the well-being and prosperity of the family is said to be the driving force behind numerous men and women whose extraordinary work behaviour is being held up as a model for the rest of the world. Although these scholars have captured the significance of familism in East-Asian societies, such an approach assumes that family members constitute a homogeneous group with identical interests and equal power in decision making. The approach fails to recognize individual member's positional relationship with each other and with the collective unit. Nor does it explore how the supposedly 'collective' interest is constructed. While Chapter 9 in this book discusses the role of 'the Chinese family firm', here we examine the dynamics within family enterprises characteristic of several Asia-Pacific countries. Specifically, we discuss women's role in household-based entrepreneurship and the impact of householdbased entrepreneurship on gender relationships.

#### Women's roles in household-based entrepreneurship

Although East-Asian family businesses are typically owned by men, women play a very central role. The owner's wife is indispensable to the establishment and success of a small-scale family business: her dowry is pulled in as start-up capital; her women-centred social network is activated for labour recruitment; the technological know-how she acquired from working before marriage gives the family business a competitive edge; and her bookkeeping skills are employed to keep transaction records straight. For factories that provide room and board for out-of-town workers, the owner's wife becomes the natural and unpaid cook and housekeeper.

In addition, the labour of an owner's wife is indispensable because she can always be counted on for overtime work. She is also the most loyal, reliable supervisor that a factory owner can ever find. By providing close supervision on the shop floor, a wife serves to maximize productivity, minimize possible waste of materials and, consequently, increase business profitability. The following description of the work process in a Taiwanese wooden jewellery-box factory illustrates how the owner's wife acts as a supervisor and rate-of-work setter:

Today I [Hsiung] was assigned to wrap every wooden jewellery box with a piece of paper. Show-Li [fellow worker] stood to my right putting each wrapped jewellery box into a coloured paper box. Ching-may, the owner's wife, performed the next steps in the process by putting four Styrofoam protectors on the bottom of each jewellery box, closing the bottom of the paper box and then passing it to the next person, Ya-ling. Ya-ling in turn had to right the paper box, put another four protectors on the top of the jewellery box and then close the cover of the paper box before passing it to the next person for final packing. Across the table, there were four workers doing the same job we were doing. I had to keep up the pace so that Show-Li and the rest of the workers on my side would not be standing there without anything to do. Ching-may kept yelling to the workers who did the final packing to speed up. 'Hey you, don't stand there. Hurry up!' she said. She yelled at San-mi, who stood across from me, 'You are too slow. Go and switch with Jin-pao.' San-mi went over to take Jin-pao's place to do the final packing. Jin-pao came and took over San-mi's job. Ching-may's yelling and giving orders were quite intimidating. While I was trying to push myself to go faster, I heard her say to Jin-pao, 'Smart guys should avoid standing next to me. They get exhausted easily.'

(Hsiung, 1996, p. 105)

Although the growth of small-scale, family-centred factories make experienced, hardworking female workers attractive marital partners for the male owners, their indispensable role and contribution have not allowed them to achieve equal partnership with their husbands. Besides, their contribution has been invisible and unrewarded in monetary terms. Women who marry factory owners are more likely to work as unwaged family workers than the workers' wives. Although the owners' wives as a group enjoy higher social status and more luxurious lifestyles than their female worker counterparts, they have not been able to surmount gender disparities in relation to their husbands. More importantly, household-based entrepreneurship has made it possible for many skilled male workers to experience upward mobility by setting up their own businesses, which in turn strengthens their status as head of the household. Unfortunately, such rewards and opportunities are not equally available to their wives, or to women in general.

In summary, the family firm, touted by scholars and writers as the engine of Asian economic success, is rooted in gender ideology and held together by the waged and unwaged labour of women. Gender and class intersect in the complex relations between owner and his wife and between owner's wife and female employees. The family firm works to facilitate upward mobility for some while keeping others down.

#### 5.4 The feminization of intra-regional migrant labour

Until the 1980s, most research on international migration and work focused on men and the role of women was largely unrecognized. Women migrants' economic and social contributions were generally considered trivial or irrelevant, because when women migrated they were routinely viewed as 'tied-migrants', either dependants of male migrants or passive participants in migration. However, the migration of women has become an important issue due to the increase of independent women migrants. This is especially true for the Asia-Pacific (Chapters 2, 10 and 11).

During the last three decades the intra-regional flow of women in Asia has been higher than in any other region. In 1976 women constituted less than 15 per cent of the 146,400 Asians who left their countries to work overseas; by 1987 they comprised some 27 per cent of Asians who left that year to work temporarily abroad. In the early 1990s, approximately 1.5 million Asian women were legally and illegally working abroad, with an outflow of 800,000 a year from Asian countries of origin. Major sending countries have been the Philippines, Indonesia, Sri Lanka and Thailand; and those receiving have been the Middle East—particularly Saudi Arabia and Kuwait—Hong Kong, Japan, Taiwan, Singapore, Malaysia and Brunei.

There is a clear gender difference in the migration process, perhaps related to geographic proximity between countries of origin and destination, specific labour needs in the receiving country, the employment situation in the sending country and cultural and social factors. For example, Filipino women emigrants outnumbered their male counterparts by twelve to one for Asian destinations, although they accounted for less than a third of Filipinos going on contract work to the Middle East. Indonesian female migrant workers outnumber their male counterparts by nearly three to one, the proportion having increased sharply during the last decade. Though total numbers are lower for other Asian countries, they nevertheless have high proportions of women overseas, working mainly as domestic workers.

We attribute the feminization of migrant labour to several factors. First is the generalization of gender stereotypes in the division of labour across national borders, which matches women with 'feminine jobs' overseas. Second, it is the outcome of mature labour markets, with their high rate of participation of women in the local labour force and the consequent demand for women in traditional reproductive jobs such as housekeeping and care-giving. Third, the development of tourism and new service industries has created certain employment opportunities for women overseas. Fourth, uneven economic development in the region has increased the gap between rich and poor countries, as well as polarized intra-country rural-urban and class differences, creating a pool of potential migrant workers, including women. Last, but not least, is the penetration of global capitalism into the region, incorporating all countries into a world market of commodities and services which makes it mandatory for these countries to obtain as much foreign exchange as possible to help balance international trade. Labour supply and demand, facilitated by global capitalism, result in a transnational network of organizations that profit from human trafficking.

Concern for national image, reports of physical abuse, and pressure from humanitarian groups and women's organizations have led several countries to establish specific policies regarding female labour flows. Indonesian women must be at least 22 years old to be employed overseas. In addition, when women are recruited through authorized agents there are various restrictions regarding place of employment for household workers and the male/female ratio, although these restrictions may be lifted under certain conditions. In Thailand there is a ban on the recruitment of women for overseas work in general except in the case of selected countries of employment. The Thai government, under domestic and international pressure to curb its notorious image as a haven to the

sex trade, now requires entertainers to hold a diploma from a school of arts and a license, and forbids them to perform in night clubs abroad.

An extreme contrast in government policies on women emigrant workers is provided by Malaysia and the Philippines. While the former does not seem to have any restrictions, the latter has developed a set of explicit and detailed rules according to the type of work and country of destination. Filipino women domestic workers must be at least 25 years old and they are banned from employment in certain countries where serious abuses have been reported. The government requires a minimum age of 23 for women entertainers and has a similar ban on selected countries. Women entertainers must also pass the required academic and skills tests, possess an Artist Record Book and undergo a predeparture showcase preview. Nurses employed abroad must be at least 23 years old, possess a Bachelor of Science degree in nursing and have one year's work experience in the Philippines.

Perhaps as might be expected, women migrant workers in the receiving countries do not seem to comply with these elaborate regulations. Female migration for overseas employment shows an almost exclusive concentration in the services sector, namely domestic workers and entertainers, or sex service providers. A significant number of women working overseas are nurses, although their destinations are not concentrated in other Asian countries as they are with the other two occupations. The social construction of gender and racial stereotypes within the labour recruitment process help to channel women migrants into these jobs.

#### Domestic workers

The increasing flow of female domestic workers is evidence of a new transnational division of labour between middle class women in the receiving countries and working class women in the sending countries. The use of domestic workers enables middle class women to join the labour force, leaving childcare and other household chores to the migrant women workers. This is certainly true in major East-Asian cities such as Singapore, Hong Kong and Taiwan. These three cities have emerged as the main destinations of employment for women migrating for domestic services in the Asia-Pacific since the mid 1980s. By the end of 1995, there were 152,000 migrant domestic workers in Hong Kong, more than 80,000 in Singapore and approximately 10,000 in Taiwan. In Hong Kong, migrant women come mainly from the Philippines and Indonesia. In Taiwan, women from the Philippines make up more than 90 per cent of the domestic workers on the island. In recent years, women domestic workers from mainland China are increasingly found in Chinese-speaking Taiwan and Hong Kong.

The Philippines is by far the most important source of Asian domestic workers, and their major destinations are the Middle East, Singapore and Hong Kong. In 1992, Hong Kong reported 66,000 Filipino domestic workers, Singapore 50,000 (1993) and Malaysia 10,000. The number of Filipino domestic workers in Hong Kong had almost doubled to 125,000. Women migrating from the Philippines are mostly from 20 to 34 years old, although there are 5.5 per cent between the ages of 15 and 19. These domestic workers are well-educated women, between 30 to 43 per cent having completed college.

In response to reports of serious abuse, the Philippines imposed a ban on the exit of domestic workers to Singapore in 1987, but this did not mean a drop in the flow as work

permits approval would be issued in Singapore to those arriving with tourist visas. In fact, these attitudes in the two countries led to the growth of private, often disguised recruitment agents and the informalization of migrant domestic work.

The execution of a Filipino domestic worker convicted of the murder of her employer by the Singaporean government, in spite of diplomatic and social protests in the Philippines, brought another ban on the deployment of domestic workers to Singapore in 1995. However, it only affected the medium-sized agencies, leaving the large and small agencies untouched. The latter still offer domestic workers to interested employers in Singapore. This ban was further circumvented by the re-routing of Filipinos via Kuala Lumpur, Brunei and Hong Kong. Any reducation in the number of Filipino workers is now made up by increases in Sri Lankan (from 12,000 to 16,000) and Indonesian (from 11,000 to 23,000) domestic workers.

Migrant domestic workers are usually paid substandard wages which are sometimes delayed or withheld. Days off are not observed, food is inadequate, accommodation is poor and unsafe, medical benefits are nil and mobility is limited. Furthermore, they are subject to maltreatment, sexual harassment and other abuses such as excessive workloads and working for additional households. Workers unwilling to endure these abusive conditions often seek refuge in their embassies. Some run away from their employers and may become super-exploited illegal workers, liable to be arrested by the police and deported or returned to their employers.

Workers of different nationality and ethnicity are subject to different discriminatory treatment. For example, Filipinos earn 50 per cent more than women from Indonesia and Sri Lanka because of their command of the English language. Domestic workers are not considered as 'labour' in many receiving countries, which helps to explain their abusive treatment by employers. In Japan, migrant women domestic workers are seen as housewives' and mothers' helpers, filling voids created by the occupational advancement of Japanese women.

In comparison with other Asia-Pacific countries, Hong Kong seems to have the most progressive legal system for the protection of domestic service workers. Both foreign and local domestic workers are protected by the labour law with a minimum wage system. This is probably the consequence of the strong presence of non-government organizations (NGOs) in the territory. Migrant women domestic workers are more vulnerable both in Taiwan and Singapore due to the lack of legal protection, the denial of access to the redress system, the inadequacy of the legal infrastructure, the transfer of responsibility from the state to individual providers, intrusive immigration regulations and the limited function of the NGO community. They are excluded from the protection of labour laws.

Employers have the power to terminate the contracts of migrant workers without due notice, cancel their work permits and repatriate the workers as long as no legal action has been initiated. In both Singapore and Taiwan, migrant women domestic workers even have to go through pregnancy tests every six months, and if the result is positive they are subject to immediate deportation. Singapore has the additional requirement of a security bond of US\$5,000 for each migrant domestic worker from her employer, higher than any other category of foreign workers. The bond is forfeited if the maid becomes pregnant.

#### Entertainers and sex service providers

As is the case with domestic workers, the outflow of entertainers is dominated by the Philippines, although Thailand and mainland China are now playing an increasingly important part. The main destinations for entertainers are Japan, Hong Kong and Taiwan, through both legal and illegal channels. Although some women are indeed professional entertainers or artists, the name is also a euphemism for sex service providers. We discuss the sex industry in Section 5.5; here we focus on the situation of foreign migrant sex service providers.

The export of entertainers to Japan represents a reverse of 'sex tourism' which has escalated in Asia-Pacific since the 'economic miracle' first appeared in Japan in the 1960s. Organized by tour operators and tacitly, sometimes even blatantly, promoted by governments eager to gain foreign exchange, 'sex tourism' has brought millions of men, mostly Japanese, to other Asia-Pacific countries. As 'sex tourism' became less overt and blatant following strong protests from origin and destination countries, there was a corresponding rise in the intra-regional movement of female tourists who often ended up working illegally as 'entertainers'.

Murata Noriko estimates that there are more than 100,000 foreign women working in the sex industry in Japan, a lucrative business which brings in trillions of yen a year (AMPO, 1996, p. 115). Filipinos were the first large group to come in the early 1980s, followed by Thai women. Entertainers from the Philippines could earn US\$1,500 per month but, due to exorbitant brokerage fees and transportation debts, they would usually receive only about one-third of this amount.

Compared with Filipinos, Thai women are easily victimized since they tend to come from rural communities and cannot speak Japanese or English.

Many find it impossible to escape from forced prostitution at small bars (called snacks) or regular bars in city red light districts or at local hot spring resorts. An especially large number of Thai women are tricked into coming to Japan by agents working for local and international human smuggling gangs. They are lured by promises of jobs in factories and told that they will be able to pay their debts off in a very short time, send considerable amounts of money home to their families and make their parents happy.

(AMPO, 1996, p. 116)

Just like the Filipino domestic workers in Singapore who rebelled against oppressive conditions of employment and were deported, imprisoned or even executed, Thai migrant women workers trying to escape from sex slavery by killing their owners or managers were prosecuted for their 'heinous' crime. Murata quotes a letter written by one of the accused to the Thai embassy in Tokyo:

There are no factories where we can work, there are only bars and pubs and men who only think about drinking and having sex. I really suffered a lot. We had to go to bed with dirty men and strangers. If you don't do whatever you are told to do, you are beaten by the boss or the proprietress (mama-san).

For them we Thai women are mere animals. They have the power of life and death over us. Japan is not heaven but hell for us Thai women. It's a barefaced lie that the cherry blossoms are waiting for you. What is waiting for you are men whose only concern is having sex.

No matter how hard you work, putting your life on the line, there are no rewards, not even a single baht (Thai denomination of money). If you don't obey them, they'll hit you, beat you and do whatever they want. To them, we have no value other than that of a tool they use to make money by entertaining drunks and providing sex.

Japan is not a place for Thai women who are eager to work. Please stay in Thailand. In our country people still have sympathy. No matter how much you have to suffer, no matter how hard you have to work, it's far better than being trapped in hell like me.

(AMPO, 1996, p. 118)

#### 5.5 The sex industry and child prostitution

The exploitation of sex, and especially child prostitution, is linked to the globalization of the sex industry, tourism and the uneven economic development in the region. In the last section we focused on the situation of foreign sex workers; here we discuss the sex industry and efforts to curb child prostitution.

Japan's phenomenal economic growth was quickly followed by a tremendous change in the lifestyles of the Japanese. Instead of preaching frugality and the delay of gratification, the Japanese government, together with the media, promoted mass consumption, even calling it a virtue in 1961. Japanese scholars and writers have attributed the commodification of sex, the rise of pornographic culture, the development of sex tourism and child prostitution to the sudden prosperity of Japan *vis-à-vis* its neighbours. What Japanese men could not easily obtain at home, they sought elsewhere in the Asia-Pacific. Taiwan and South Korea, both once colonized by Japan, were the favourite destinations in the 1960s. But as these two countries achieved newly industralized economy (NIE) status, concerns of international image and the growth of a protest movement tipped the cost-benefit calculus of sex tourism. The *nouveaux riches* Taiwanese and Korean men joined their Japanese brothers in seeking sexual gratification in their less fortunate neighbouring countries.

Airline magazines, tourist guide books and even the government tourist bureau publications promoted the 'buying of spring'. Travel agencies organized package tours on the principle of 'sightseeing by day and devouring women at night'. As recently as September 1994, four Japanese writers calling themselves the Asian Sexual Customs Study Group published a book entitled *The Thailand Prostitution Guide*, which became a best seller.



#### Prostitutes in a bar in Bangkok, Thailand

Thailand and Taiwan are notorious for the availability of child prostitution even though both countries have laws against it. Girls as young as eleven are found as prostitutes. The traditional ideology of self-sacrifice and submissiveness plays an important role in child prostitution, as girls, mostly from rural areas, 'volunteer' or are forced by their real or adoptive parents to trade sexual services for family betterment or for the education of their male siblings. Wiping out child prostitution has been a very prominent goal of women's movements in the Asia-Pacific. Activities include pressuring the government to pass and enforce laws against human trafficking and the buying and selling of children's sexual services, building networks of rescue, providing health care and shelter and establishing rehabilitation centres to help former child prostitutes to return to school and the community.

#### 5.6 Women's resistance, agency and compliance

With the exception of South Korea and, to a lesser extent Taiwan, workers' struggle against capitalist exploitation in NIEs rarely takes the form of collective, organized and large-scale union movements. This has to do with the suppressive nature of state legislation on labour and the lack of formal, institutional protection for workers. However, this does not imply that Asia-Pacific workers are mindless victims. Recently, with capitalists relocating their factories and firms to countries with cheaper labour within and outside of the region, workers' class consciousness seems to be on the rise in Taiwan. In this section we focus on women workers' resistance to capitalist exploitation and their struggles for a better life.

The range of issues subjected to on-going negotiation and resistance is rather wide. It includes standard, contractual issues such as product quality, production quotas, overtime and overtime payment, as well as non-contractual issues such as eating and talking at work, dress codes, etc. This suggests that Asia-Pacific capitalists seek to extend their control beyond workers' physical labour to include also workers' attitudes and social behaviour, perceived as having an effect on production and the image of the firm. The employers' profit-driven and production-oriented managerial focus is in direct conflict with the multiple objectives that workers intend to accomplish through employment.

In addition to conventional means of resistance such as verbal negotiation and absenteeism, workers in the NIEs' manufacturing sector employ other ways of defining their own boundaries and guarding their integrity. For example, in order to avoid undue exploitation of their skills by the owners, male workers in Hong Kong's carpentry industry and in Taiwan's wooden jewellery box factories deliberately terminate regular daily exchange with others by displaying moodiness or 'bad temper' attitudes. In contrast, female workers in Taiwan and Singapore deliberately engage in wrangling and teasing with their supervisors and/or owners to skilfully set a boundary of acceptable labour extraction and a limit to the relationship.

The fact that negotiation and resistance have been carried out on an on-going basis suggests that managerial effort to control is constantly challenged. While none of the negotiation tactics and resistant strategies are aimed at challenging the production regime as a whole, or the workplace hierarchy in particular, women workers do pursue their daily struggle with concrete, subjective objectives. For example, despite legislative regulations concerning overtime and overtime payment in many of the NIEs, in practice frequent violations and inefficient law enforcement compel the workers to negotiate individual arrangements with their employers. This piecemeal method allows the workers to accomplish their individual, immediate goals without calling upon any abstract notion of justice or law enforcement. The relative absence of an organized union movement in the

Asia-Pacific has fostered the development of a form of workers' resistance that is individualistic and clandestine. Our analysis leads us to an appreciation of the workers' own perception and definition of the situation, and their corresponding tactics and strategies of struggle.

Asia-Pacific NIEs are noted as having some of the most notorious and oppressive labour regimes, while their workers are portrayed as suppressed, mindless victims. This fails to include the workers' subjective understanding of their experiences. A comparison of the experiences of married female workers in Hong Kong with those of single female migrant workers in Canton, China, is instructive. According to our research, married women in Hong Kong are unwilling to endure on-the-job stress, or to invest a substantial amount of time and effort in acquiring additional skills or qualifications. They perceive factory work as a job, not a career, and therefore refuse to let it override their real priority, their familial responsibilities. They are confident of being capable of becoming the foreladies, technicians, or supervisors if they were willing to put aside their familial role. In this context, married women in Hong Kong easily accept their subordination to their male superiors. Nevertheless, such acceptance does not render them mere passive followers of ruthless capitalist disciplines. From time to time, married female workers in Hong Kong strategically call upon their familial responsibilities to put off unwarranted demands from their supervisors.

In contrast, employment opportunities in China's newly established coastal factories provide single, migrant female workers not only with precious cash earning opportunities and an escape from the agricultural backwater, but also the chance to meet potential marriage partners. For a supposedly promising future, these girls choose to leave family lives behind, take the risk of looking for jobs in strange cities and endure hardship in a factory job. Within this political, economic context, single, migrant female workers in Canton are more likely to comply, rather than challenge capitalist control. Again, to appreciate the engendered agency behind such compliance, one needs to go beyond the simple, rigid notion of passivity and self-victimization.

## 5.7 Summary: towards a feminist perspective on the 'economic miracle'

This chapter has sought to uncover the gendered nature of Asia-Pacific economic growth by examining the labour roles that women have played in the development process and the consequences this economic growth has had for them. We argue that, in many ways, women receive the short end of the stick. They have been incorporated into the labour force, but their significant contribution to the 'economic miracle' has not yet won them gender equality. Although more and more women are entering the labour market, they are still concentrated in just a few industries, unlike their male counterparts. Women workers are also concentrated in the lower end of the occupational structure and wage hierarchy. Even within the same occupation, women are paid less than men. These general patterns are similar to the experience of women worldwide. Our discussion does not include work in the reproductive area, where women are disproportionately concentrated. Suffice to point out that women everywhere have taken on waged work in addition to their unwaged family labour. While economic development has opened up many opportunities for waged work for both men and women, women have been incorporated into capitalist production in a quite different way. Global capitalism, intra-regional economic disparities and the social construction of gender roles operate together to track and sometimes trap women into conventional jobs, and to thwart their advancement in education, work and pay.

Our analysis points not only to the general pattern of women's experience in Asia-Pacific development, but also to experiences related to class and to particular countries. As more middle-class women enter the labour force in economically more developed countries such as Japan and the four 'little dragons', migrant women from poorer neighbouring countries fill their domestic labour needs. Increased labour force participation of women in one country has consequences for the socialization and commodification of housework as well as the engendering of intra-regional migration. The differences we have observed among the countries are a function of their position in the world economy, the development strategy of the state and their cultural and ideological systems.

Furthermore, we argue that the subjective meaning of work is not the same for women as it is for men, and that women in the Asia-Pacific have developed their own particular forms of resistance to gender discrimination and class exploitation. Any approach that seeks to understand women workers' compliance with, and resistance to, global capitalism must begin with an appreciation of their own definitions.

#### References

- All China Women Federation and Shaanxi Provincial Women Federation (eds) (1991) *Statistics on Chinese Women (1949–1989)*, Beijing, China Statistical Publishing House.
- AMPO-Japan Asia Quarterly Review (ed.) (1996) Voices from the Japanese Women's Movement, Armonk and London, M.E.Sharpe.
- Boserup, A. (1970) Women's Role in Economic Development, London, Allen and Unwin.
- Cheng, L. and Hsiung, P-C. (1992) 'Women, export-oriented growth, and the state: the case of Taiwan' in Appelbaum, R.P. and Henderson, J. (eds) *States and Development in the Asian Pacific Rim*, London and New Delhi, Sage Publications.
- Gelb, J. and Palley, M.L. (eds) (1994) *Women of Japan and Korea: Continuity and Change*, Philadelphia, Temple University Press.
- Hong Kong Census and Statistics Department (1991) *Population Census Main Report 1991*, Hong Kong.
- Hong Kong Census and Statistics Department (1996) Hong Kong Annual Digest of Statistics, Hong Kong.
- Hong Kong Census and Statistics Department (1997) 1996 Population By-Census Main Report, Hong Kong.
- Hsiung, P-C. (1996) *Living Rooms as Factories: Class, Gender, and the Satellite Factory System in Taiwan, Philadelphia, Temple University Press.*
- International Labour Organization (ILO) (1992) Yearbook of Labor Statistics, ILO, Geneva.
- International Labour Organization (ILO) (various years) Yearbook of Labor Statistics, ILO, Geneva.
- Jaquette, J. (1982) 'Women and modernization theory: a decade of feminist criticism', *World Politics*, vol. 34, no. 2, pp. 267–84.
- Li, K.T. (1985) 'Contributions of women in the labor force to economic development in Taiwan, the Republic of China', *Industry of Free China*, vol. 64, pp. 1–8.

Mies, M. (1986) *Patriarchy and Accumulation on a World Scale*, London, Zed Press. People's Republic of China (PRC) (various years) *Yearbook of Labour Statistics*, Beijing, PRC. Redding, S.G. (1993) *The Spirit of Chinese Capitalism*, Berlin and New York, de Gruyter. Republic of China (RC) (1993a) *Report on the Manpower Utilization Survey Taiwan Area*,

*Republic of China, 1993,* Taipei, Directorate-General of Budget, Accounting and Statistics. Republic of China (RC) (1993b) *Statistical Yearbook of Taiwan Province,* Taipei, RC.

Republic of China (RC) (various years) Yearbook of Labour Statistics, Taipei, RC.

The World Bank (1995) *Toward Gender Equality: The Role of Public Policy*, Washington, DC, The World Bank.

The World Bank (1996) *Involving Workers in East Asia's Growth*, Washington, DC, The World Bank.

## CHAPTER 6 The role of the state in economic development

Stephen W.K.Chiu and Tai-lok Lui

#### 6.1 Introduction

By now it is customary to attribute the economic success of East Asia to the special role played by the state in the development process. Here we follow Rueschemeyer and Evans' conception of the state as 'a set of organizations invested with the authority to make binding decisions for people and organizations juridically located in a particular territory and to implement these decisions using, if necessary, force' (1985, p. 47). It is argued that in contrast to the Anglo-American model of regulatory states which play a refereeing role only in the economy, East Asian states are directly involved in the economy and have a significant influence on private decisions. Even the World Bank, long a bastion of free market philosophy, has recently admitted that the state could play a larger role than neo-classical economics would call for. Its 1993 report on the Asia-Pacific observes that: 'More selective interventions—forced savings, tax policies to promote (sometimes very specific) investments, sharing risks, restricting capital outflow, and repressing interest rates—also appear to have succeeded in some HPAEs (High Performing Asian Economies) especially Japan, Korea, Singapore, and Taiwan, China' (World Bank, 1993, p. 242).

Yet this in fact represents a sea change in the assessment of the contribution made by the state in East Asian development. In the 1970s, when world-wide attention first focused on the economic success of the East Asian newly industrialized economies (NIEs), these states were hailed as models *par excellence* of a growth strategy based on an unfettered market mechanism. Against the contemporary Keynesian orthodoxy in development economics, some neo-classical economists argued that the East Asian NIEs' high speed growth came about only after the state dismantled restrictive, protective industries policies and set prices (interest rates, wages and exchange rates) correctly (see also Chapter 3). Into the 1980s, however, a new orthodoxy in development economics and a different interpretation of the East Asian experience emerged. Studies began emphasizing the idea of the 'developmental state' itself as responsible for East Asia's success. Chalmers Johnson's (1982) study of Japan pushed this perspective to the forefront of academic discourse, while the IDS Bulletin's special 1984 issue on the East Asian developmental states reflected the growing importance of this 'revisionist' perspective, as opposed to the neo-classical orthodoxy (Wade and White, 1984). Country studies of the East Asian NIEs, such as Amsden's on Korea (1989) and Wade's on Taiwan (1988), developed a state-centric account of development in the region. Comparative analyses, be they monographs (Haggard, 1990), or anthologies (Deyo, 1987;
Appelbaum and Henderson, 1992), also began to anchor in the statist line in their interpretation of the East Asian experience. The concept of the developmental state also became increasingly current in East Asian studies (Castells, 1992). Both the neo-classical and statist approaches are often analytical and prescriptive in nature. They are based on a particular conception of the development process and its requirements and offer a specific description and empirical generalization of the role of the state. More importantly, these accounts also propose a prescription about the best mix of public policies to promote economic development. Focusing on the East Asian NIEs and Japan, this chapter will trace the changing perception of the role of the state in East Asian development, with an emphasis on the diversity of the relationship between public policy and the economy.

#### 6.2 The market-led model

The magnitude of change in the mainstream view of the role of the state in development in general, or East Asian development in particular, can perhaps be gauged by the view of the World Bank itself. Wade (1996) provides a fascinating account of this change in the World Bank's view of the East Asian miracle. Until the 1980s the World Bank was the ultimate institutional anchor of a market-led model of development and East Asia was taken as the epitome of this development strategy. In the *World Development Report 1987*, it offers a direct statement consistent with the mainstream neo-classical view in economics (the stream of economics originating from Alfred Marshall which stresses the positive effect of an unrestricted market in the optimal allocation of resources) in support of the free trade and free market model of development. In its lending policy, the World Bank strives to coax developing states to adopt a more 'market-friendly' programme of development, as do other multilateral financial bodies such as the International Monetary Fund.

Accounts of East Asian development that emphasize the importance of market forces are best represented in the writings on the East Asian NIEs of economists associated with the World Bank in the 1970s and 1980s, such as Ian Little and Bela Balassa (Little, 1981; Balassa *et al*, 1982; Balassa, 1988). In this model, the state's role is relatively limited to catalyst and corrector of market failures. Occupying the centre stage are private entrepreneurs who respond eagerly to market stimuli, capitalizing on cheap, plentiful supplies of labour. This interpretation starts with a free trade regime, whereby national policy allocates resources in accordance with the country's existing comparative advantage (see Chapter 2). Thus, Little (1981) stresses the positive effects of 'almost free trade conditions for exports' in East Asian NIEs' success. By 'getting the price right' through trade liberalization and exchange rate reform, East Asian NIE states provide the optimal environment for the growth of private enterprise.

During the East Asian NIEs' transition to industrial capitalism, the states in Korea and Taiwan removed fetters surrounding private industry, by abolishing restrictions on trade and devaluing currencies. In the case of Hong Kong, the maintenance of a free trade regime accounted for the city's successful industrialization. Freed from the fetters of government inter vention, capitalizing on their comparative advantage, the East Asian NIEs therefore embarked on export-oriented, labour-intensive industrialization. Government stability is also important because it provides a stable, long term horizon for private business calculations. A regulatory framework and infrastructural capacity is seen as beneficial, but any interference in private decision making is not. Hence, as Balassa remarks in his study of the 'lessons' of East Asian development:

the principal contribution of government in the Far Eastern NIEs has been to create a modern infrastructure, to provide a stable incentive system, and to ensure that government bureaucracy will help rather than hinder exports... More generally, less use has been made of government regulation and bureaucratic controls in East Asia than elsewhere in the developing world. Finally, there have been fewer policy-imposed distortions in labour and capital markets, and greater reliance has been placed on private enterprise.

(Balassa, 1988, pp. 286-8)

It is interesting to note, however, that in the same issue of the journal *Economic Development and Cultural Change* in which Balassa's article appeared, another eminent development economist, Paul W.Kuznets, also compares Japan, Taiwan and South Korea and concludes with a diametrically different view of the state from that of Balassa. As Kuznets states, in these three countries, '[g]overnment intervention, though restricted by the need to keep exports competitive, has been pervasive' (Kuznets, 1988, p. S36).

According to the mainstream World Bank view, however, the protagonists of the East Asian industrialization narrative are the private entrepreneurs, and the state's role is best conceived as catalytic rather than 'pervasive'.

# Latin America vs. East Asia: the secret of export-led growth

Focusing on the East Asian NIEs, Balassa (1988) observes that these countries (apart from Hong Kong) have passed through the first stage of import-substituting industrialization (in which the local market is primary and imports of labour-intensive consumer products are 'substituted' by local production). However, instead of following the path of the Latin American NIEs in adopting the second stage of import-substituting industrialization (local production of capital-intensive producer goods substituting for imports), the East Asian NIEs shifted to a new develop-mental strategy called exportoriented industrialization, with export manufacturing as the engine of economic growth (see Chapters 2 and 3).

For Balassa it was the growth of exports in the East Asian NIEs that accounted for their GDP growth rates being the highest amongst developing countries. First, Balassa states that 'exports contribute to resource allocation according to comparative advantage. At the same time, these gains cumulate over time as the efficiency of new investment is enhanced through its orientation toward industries that correspond to the comparative advantage of the countries concerned' (Balassa, 1988, pp. 280–1). Second, exports make it possible for the East Asian NIEs to overcome the limitations of their small domestic markets by ensuring full utilization of resources and reaping the advantages of large-scale production. Third, while import substitution often leads to protectionism and monopolies, export-oriented industrialization provides the 'carrot and the stick' of competition,

inducing technological change in export industries that must keep up with modern technology in order to improve their position in the world market.

According to Balassa, the four determinants of the East Asian NIEs' favourable economic performance are: stability of the incentive system, limited government intervention, well-functioning labour and capital markets, and reliance on private capital (1988, pp. 286–8).

First, the East Asian NIE states have a long tradition of encouraging exports by establishing incentives, eliminating administrative obstacles, and creating a favourable environment for exporters. In addition, the incentive system for exporters is relatively stable. For instance, the East Asian NIEs avoid the appreciation or fluctuation of the exchange rate, and exporters can usually expect that the incentives they receive will be maintained in the future. This is in contrast with the Latin American NIE states, which allow the export exchange rate to fluctuate and wages to rise, thereby greatly reducing the profitability of exporters.

Second, although the states in East Asia actively work to create a positive environment for economic growth, they rarely stray outside this limit and they do not interfere with the free workings of markets. As a result, the scope of administrative control is much more limited in East Asia than in Latin America. In the latter case, not only are there pervasive controls over investment, prices, and imports, but decisions are also generally made on a case-by-case basis, thereby creating uncertainty from a business perspective.

Third, states in the East Asian NIEs have instituted fewer policy-imposed distortions over labour and capital markets. While labour markets are generally free in the East Asian NIEs, they are highly regulated in the Latin American NIEs. Balassa complains that prohibitions on discharging labour and high severance payments in the Latin American NIEs have increased their cost of labour. Moreover, capital markets are also freer in the East Asian NIEs than in the Latin American NIEs. In the East Asian NIEs, interest rates are tied to market rates to provide incentives for domestic savings and to discourage the outflow of capital. In the Latin American NIEs, by contrast, artificially low interest rates have reinforced the effects of overvalued currency rates in encouraging the outflow of capital.

Finally, greater reliance has been placed on the private sector in the East Asian NIEs than those in Latin America. In the East Asian NIEs, private enterprise takes the lead in making the necessary investments, and, through exposure to international competition, in becoming efficient and profitable. On the other hand, public enterprise tends to play a more important role in the Latin American NIEs than in the East Asian NIEs. In the early 1980s, the outlays of public enterprise accounted for 26 per cent of GDP in Mexico compared to only 4 per cent in South Korea. Balassa argues that economic growth is negatively correlated with the size of the public sector, the share of government expenditures in GDP, and the tax burden (see also Chapter 3).

#### Problems of the neo-classical imagery

As a set of empirical generalizations, neo-classical accounts of Asian industrialization have been under heavy fire lately, especially in terms of the applicability of the model to all of East Asia. Even if Hong Kong and Singapore were indeed perfect examples of the free market theory (we shall return to this question later), the same does not apply to Japan, Taiwan and Korea. In emphasizing the efficacy of the market, free marketeers are liable to overlook the fact that Japan, Korea and Taiwan's economic structures and policies are a far cry from the image of the unfettered market. Neo-classical theories argue that the establishment and maintenance of a free trade regime has been critical to the industrial take-off of East Asia. Hence they trace the explosive industrial growth of Taiwan and Korea in the 1960s to policy reforms in the late 1950s and early 1960s liberalizing international trade. Nonetheless, subsequent scholars such as Luedde-Neurath (1986) and Wade (1988; 1993) question the extent of such liberalizing measures.

First, they argue that studies conducted by neo-classical theorists such as Balassa are methodologically flawed, and that the picture of Taiwan and Korea as free trade regimes compared to other industrializing countries is highly misleading. These neo-classical theorists argue that Taiwan, Korea and Singapore have much lower levels of protection over manufacturing than such countries as Israel, Colombia and Argentina. Balassa's calculation of effective protection was based on a comparison of the prices of domestically produced commodities and foreign produced versions of the same item. Given the existence of tariff and trade restrictions, the domestic price is expected to be higher than the international free trade price. Balassa and his associates reject the use of legal tariffs and quantitative restrictions on imports for their analysis, relying instead on direct price comparisons. However, while direct price comparisons of domestic and international markets yield low rates of protection for the East Asian NIEs, we cannot ignore the fact that both Korean and Taiwanese legal tariff rates for imports averaged over 60 per cent in the 1960s. Furthermore, in Korea in 1968, some 294 (74 per cent) out of 396 commodities were subject to quantitative import controls (Wade, 1988). In a word, the trade regimes in Korea and Taiwan, with their full array of both quantitative and qualitative controls, cannot be characterized as *laissez-faire* (Wade, 1993).

Second, market-led approaches also assume a supply of production inputs free from administrative encumbrances, another pillar of the East Asian NIEs' market mechanism. One prominent fact disproving this account is that the East Asian financial system, a key source of funds, is never as free as the economists expect (see Chapter 4). Almost all the commercial banks in Taiwan and Korea were owned by the state until the late 1980s. Woo (1991), furthermore, contends that Korea's financial reforms of the early 1960s in fact strengthened, not loosened, state control over the financial system. Given these circumstances, it is difficult to sustain the view that fund allocation is unaffected by state actions. While Hong Kong and Singapore's financial systems are relatively 'undistorted' by government intervention (in the neo-classical sense), the Korean and Taiwanese states' allocation of funds is a critical component of their development strategies.

Third, few can deny that the Korean and Taiwanese states implemented an extensive industrial policy that shaped the course of development. In Korea, the state funnelled a large amount of capital into large-scale conglomerates (or *chaebol*), enabling them to invest in new ventures and expand their existing production lines. Again, this contradicts the theories of neo-classical economics and *laissez-faire*, which would emphasize the importance of small-scale competitive firms as the ideal 'market structure' for successful capitalist development. In Taiwan, as late as the 1970s, the public enterprises' share in production and investment was among the highest in the capitalist world. Taiwan possessed more the characteristics of a 'mixed economy' during its early stage of industrialization than that of a free market economy.

#### Chaebol

*Chaebol* are the Korean equivalent of the Japanese *keiretsu* (described in Section 6.4). They can be defined as business groups consisting of large companies in highly diversified business areas, owned and managed by family members or relatives (Steers *et al.*, 1989). It is a tricky business to count which ones are *chaebol* and which are not, but it is common for observers to include at least 30 such groups in the rank of the *chaebol* (Jones, 1994; *The Economist*, 1996). All of the *chaebol* are highly diversified. The largest of them, Samsung, has products ranging from aeroplanes and semiconductors to flour and printing paper. It also operates a wide range of service firms such as insurance, hotels and department stores.

The economic might of the *chaebol* is formidable, although estimations of their size vary. One estimate puts the combined sales of the four biggest *chaebol*— Hyundai, Samsung, LG and Daewoo—as equivalent to 80 per cent of GDP and 60 per cent of exports (*The Economist*, 1996, p. 58). These four are also among the top Fortune 100 multinationals outside the USA. Their presence is no longer confined to South Korea, and a few of them have, since the 1980s, become household names in the USA and Europe

(automobiles from Hyundai and Daewoo and electrical and electronic appliances from Samsung).

The rise of the *chaebol* to economic dominance has always been attributed to the heavy-handed intervention of the South Korean state in the economy. By allocating loans made by state-owned banks to the *chaebol* and guaranteeing their foreign borrowing, the state funnelled massive amounts of capital to these groups and helped their expansion and diversification. For example, during the 1980s the *chaebol* received a large amount of such 'policy loans' so that they could enter the high-tech electronics sector. With state assistance, Samsung Electronics has become a major player in the mass production of memory chips.

(See Chapter 9 for further elaboration of the nature and importance of the *chaebol.*)

The cases of Singapore and Hong Kong, as well as those of Taiwan and Korea, seem at odds with the neo-classical description. Singapore, for instance, fits the *laissez-faire* image poorly. The state distributes extensive incentives to foreign investors in strategic sectors, and also provides infrastructural assistance selectively to firms that the state considers beneficial to industrialization. The state also exercises heavy-handed control over the labour movement and the determination of wages. Even in Hong Kong, the state is intricately linked to economic development in various ways. For example, the state is heavily involved in the reproduction of labour power through provisions in public housing and education (see Castells, 1992; Chiu *et al.*, 1995).

### 6.3 The state-led models

Revisionist interpretations of the East Asian NIEs provide a diametrically different picture from the neo-classical perspective, epitomized by the concept of the developmental state (see Johnson, 1987; Castells, 1992). This intellectual paradigm draws historical sustenance from the argument that successful 'late development' takes a very different form from that of earlier industrializers (Gerschenkron, 1962); the former's developmental process becomes less 'spontaneous', with the state assuming the role of the major agent of social transformation. Partly in reaction to the neo-classical onslaught, two major studies on East Asian industrialization are devoted to this theme (Wade and White, 1984; Deyo, 1987). As Wade and White observe:

If we turn to Japan, South Korea, and Taiwan, among the most dramatic and equitable cases in the history of capitalist development, industrialization has in each case been accompanied by aggressive government intervention. The authorities have acted to guide markets and moderate the competitive process in a way that neo-classical economics says public officials cannot get right.

(Wade and White, 1984, p. 1)

Deyo also concludes his volume on the 'new East Asian industrialism' by proposing a 'strategic capacity' model, emphasizing that:

[the] state's commitment to economic expansion and, more important, its *capacity* to implement well-chosen development strategies differentiates these NIEs from other developing countries better endowed in natural resources, scale of domestic markets, and other economic assets.

(Deyo, 1987, p. 228: emphasis original)

The statist perspective emerged as a critique of the free market interpretation of developing countries. Instead of emphasizing free markets, trade liberalization, private enterprise, and the restricted role of the state, the statist perspective contends that states have a strategic role to play in taming domestic and international market forces and harnessing them to national ends. Instead of focusing on maximum profitability on the basis of current comparative advantage, the statist perspective focuses on the phenomenon of 'late industrialization'. In this respect, the statist perspective is still sharing the Gerschenkronian view of late development, arguing for the importance of a strong state to overcome market imperfections and the various bottlenecks of industrialization (Gerschenkron, 1962; Rueschemeyer and Evans, 1985).

According to the statist view, state intervention is necessary for successful late industrialization. Following Gerschenkron, Amsden (1989) contends that East Asian industrialization is characterized by its 'lateness' rather than by its 'newness' (as in 'newly' industrialized economies). As latecomers, East Asian firms must compete with established Western firms that can introduce new technologies quickly and thereby earn

higher profits. This does, of course, allow the East Asian firms to acquire, 'learn,' or 'borrow,' the more codified elements of a given technology from the West without having to develop them for themselves. Nevertheless, there is generally a great gap between buying, borrowing, or stealing the codified elements on one hand and mastering the technology in production on the other. In other words, the magnitude of the problems facing latecomers called for the developmental states to offset the disadvantages faced by East Asian firms in international competition and to move the NIEs' industrial structure toward more technologically dynamic activities (Wade, 1992).

# The capitalist developmental state

In the spirit of this line of argument, Johnson first popularized the idea of the capitalist developmental state. Contrasting it to the states in socialist and Western capitalist economies, he contends that the socialist state seeks to substitute administrative commands for private entrepreneurship and market mechanism, while in Western models the state is primarily regulatory and is expected to play the role of a neutral umpire in the marketplace. The capitalist developmental state does not attempt to replace the market mechanism and private decision making, but neither does it abdicate to private profitseeking behaviours in the development process. It will strive to *influence* private business decisions by persuasion, coercion, and by manipulating the parameters of private decision making. In Johnson's words, the 'logic of such a system derives from the *interaction* of two sub-systems, one public and geared to developmental goals and the other private and geared to profit maximization' (Johnson, 1987, pp. 141–2, emphasis original).

The developmental state, according to Johnson's (1987) conception, possesses the following features:

- 1 As the name implies, economic development (in terms of growth, productivity, and competitiveness) is the foremost priority of state action. The state single-mindedly adheres to economic development even at the expense of other objectives, such as equality and social welfare. 'Developmental elites are generated and come to the fore because of the desire to break out of the stagnation of dependency and underdevelopment; the truly successful ones understand that they need the market to maintain efficiency, motivate the people over the long term, and serve as a check on institutionalized corruption while they are battling against underdevelopment' (Johnson, 1987, p. 140).
- 2 Since the developmental state is not a socialist state, it has a firm commitment to private property and the market. The market, however, is closely governed by state managers who formulate strategic industrial policy to promote development. In other words, the developmental state elite actively intervene in the economy, but in a 'market-conforming' manner.
- 3 Within the state bureaucracy, a pilot agency (such as MITI in Japan) plays a key role in strategic policy formulation and implementation. This agency is given sufficient scope to take initiatives and operate effectively, and is staffed by the best managerial talent available to the state bureaucracy. Johnson suggests that the strictly meritocratic recruitment into the bureaucracy not only ensures a high degree of bureaucratic capability, but also generates a sense of unity and common identity on the part of the bureaucratic elite. Their long-term, often lifetime, service in the same bureaucratic

agency also reinforces this *esprit de corps* among the bureaucrats and enables them to accumulate experience and expertise in economic matters.

#### MITI

The Japanese Ministry of International Trade and Industry (MITI) was established in 1949 and is generally recognized to have been a key institution in the emergence of Japan as a major economic power. MITI took responsibility for Japanese 'industrial policy' (*sangyo seisku*—a term reportedly invented by the Japanese) which, particularly in the 1960s and 1970s, guided both the private sector firms and the public sector in a direction of rapid economic growth. It represents a 'pilot agency' designed not so much to direct investment and trade into particular sectors or branches of industry, but to persuade, through consensus building measures, the big Japanese manufacturing and trading companies to enter new areas of economic activity and exporting. It has never espoused an

ideology of public ownership. Rather, MITI is run by highly trained 'pragmatic bureaucrats', whose role has been to spot the new trends and product areas, to assess the strengths and weaknesses of Japanese producers in these fields, and to propose policies to strengthen the ability of the private sector to compete in these areas. On occasions this has meant that MITI has tried to organize and re-organize the domestic productive structure to meet its objectives: to encourage the merging of domestic production, to help establish new lines of production, to organize cooperative activity amongst firms, etc.

But MITI has not been without significant powerful policy instruments designed to push companies in particular directions. Technological transfers, joint venturing, patent rights, royalty conditions, licensing agreements: all were subject to its jurisdiction during the rapid phase of industrialization from the 1950s to the 1970s. In addition, MITI has had responsibility for everything from the setting of domestic electricity prices through to the regulation of bicycle racing. Through a series of high-powered advisory committees it has scrutinized all aspects of Japanese industrial performance and actively encouraged 'high speed growth'.

MITI was credited with being the main instrument of the 'developmental state' idea first popularized by Chalmers Johnson (1982). Ever since then, this thesis has itself been the subject of refinement and criticism. MITI is often credited with failing to spot developments in key areas such as information technology. Its success in encouraging private companies to act together has been questioned, for example, in the case of the development of the petro-chemical industry. Whether MITI was quite as central to the industrial development process of the 1950s to 1960s, compared to the importance of ordinary commercial decisions made by Japanese companies without its involvement, is the subject of fierce controversy. Also, MITI's role may have declined in importance as the Japanese economy matured and its companies transformed themselves into world class operations with outstanding international competitive success.

With respect to how the state promotes late industrialization, Amsden's (1989) study on South Korea emphasizes the dual policies of 'subsidies' and 'discipline'. If the first Industrial Revolution was built on *laissez-faire*, and the second on infant industry protection (see Chapter 2), then Amsden argues that South Korea's late industrialization was founded on subsidies. The allocation of subsidies has rendered the South Korean government not merely a banker but an entrepreneur, using subsidies to decide what, when, and how much to produce, and which strategic industries to favour. Subsidies are necessary because South Korean firms cannot initially compete against Japanese products, even in such highly labour-intensive industries as cotton spinning and weaving. Needless to say, the long gestation periods and relatively low profitability (through industries' adolescent period) make capital-intensive industries rather unattractive investments to South Korean firms. According to Amsden, South Korea's entry into heavy industries, and the emergence of cotton textiles as its leading export industry, provide graphic evidence of the need for state intervention under conditions of late industrialization.

Nevertheless, while the South Korean government subsidizes strategic industrial firms, it also imposes 'discipline' on them. For example, the government specifies stringent performance requirements (notably in the field of exports) in return for the subsidies it provides. Such discipline over private firms involves both rewarding good performers and penalizing poor ones. 'Carrots and sticks' take the form of granting or withholding industrial licensing, government bank loans, advanced technology acquired through the government's investment in foreign licensing and technical assistance, etc. Since the South Korean government deliberately refrains from bailing out firms which are badly managed in otherwise profitable industries, government subsidies do not lead to a waste of resources, as they do in the case of socialist countries and many developing countries in Latin America. Thus Amsden asserts that '[w]here Korea differs from most other late industrializing countries is in the discipline its state exercises over private firms' (1989, p. 14). This interpretation of the development process is also argued to apply to Taiwan (Wade, 1992).

Finally, what is the relationship between the developmental state and other social institutions in the society? Onis (1991) points out that East Asian industrializing states are unusual because they experience both bureaucratic autonomy and public-private cooperation. On the one hand, there is a high degree of bureaucratic autonomy and capacity facilitated by meritocratic recruitment and a sense of unity and mission among state managers. This allows the state and bureaucratic elites to avoid being 'captured' by their private-sector clients and to develop national strategic developmental policies independent of the other powerful groups in society. Bureaucratic autonomy is also safeguarded by the depoliticization of major economic decisions, or what Johnson calls the separation between 'reigning' and 'ruling': 'the politicians set broad goals, protect the technocratic bureaucracy from political pressures, perform "safety valve" functions when the bureaucracy makes mistakes, and take the heat when corruption scandals are uncovered...the official bureaucracy does the actual planning, intervening, and guiding of the economy' (Johnson, 1987, p. 152).

On the other hand, there are close institutional links between the developmental state and private sector conglomerates, banks, and trading companies that dominate strategic sectors of the economy. The state attempts to implement its industrial policy through the various sector 'peak associations', that is, the largest business or industrial associations. In Johnson's (1987) words, 'administrative guidance' often works without the backing of legislative coercion because of the long-term informal relationship between bureaucrats and businessmen. Of course, the omnipresent threat of the withdrawal of state support and incentives is also an instrument to 'discipline' the private sector. As a result, the private sector is highly co-operative with the state policy of subsidies and discipline. Onis argues that this unusual mixture of bureaucratic autonomy and public-private sector co-operation brings about the emergence of a strong autonomous state which is not only capable of formulating strategic developmental goals, but is also able to translate these broad national goals into effective policy action to promote late industrialization in East Asia.

#### The limitations of the strong state model

In spite of the popularity of these stylized 'models' of East Asian development in academic and non-academic discourses, their limited validity and applicability are easily discernible. In invoking the 'East Asian model of development', we must consider Colin Bradford's early warning: The real story of what constitutes successful development [in the East Asian NIEs] is more subtle and indeed quite different from what the stereotypes [of development models] suggest' (Bradford, 1986, p. 122). Subscribing uncritically to either of the two dominant interpretations of East Asian industrialization causes a real problem: a lack of depth in the comparisons. There are two symptoms of this problem. First, researchers are prone to generalize across East Asian NIE cases without systematically comparing relevant phenomena. Second, they sometimes overuse the single case that most proves their theoretical position at the expense of paying attention to others. By positing a 'unique' East Asian model of development generalizable to the whole region, discussions of the region's developmental experiences assume too much uniformity within the 'East Asian phenomeno'.

If not the free market, is a strong, developmentalist state the motor behind the 'East Asian miracle'? Here we argue that statist theorists' imagery is also overdrawn and cannot capture the nuances of Japan and the East Asian NIEs' actual developmental experiences. Once again, deeper comparative analysis reveals a more complex picture: only Japan and Korea among the East Asian economies seem to conform to the expectations of the developmental state model; the other three NIEs deviate from this norm to varying degrees and towards different directions.

The case most clearly deviating from the statist model is Hong Kong. While Hong Kong is hardly a neo-classical utopia, it is also a far cry from the developmental state. The public sector accounts for only a modest share of the economy and public ownership of productive enterprises is almost negligible. As much as possible, the state refrains from intervening in the allocation of resources as well as in their management. It performs no industrial targeting, and encourages labour and management to determine conditions of employment 'voluntarily' by negotiation. The state elite even actively espouses a *laissez-faire* philosophy to justify its non-intervention in the economy.

In Singapore, though the state appears omnipotent and omnipresent, its role in industrialization is rather complex. The state does take an active role in indicative planning and invests in productive enterprises. Nevertheless, foreign enterprises remain

the most important agents in the industrialization process. The state's role is actually more indirect than in the other NIEs and not as great as the statist literature leads us to believe. We argue that the Singaporean state's interventions centre mainly on fiscal incentives, infrastructural provisions, and the labour market; the state is neither capable of, nor willing to, offer financial assistance to the manufacturing sector.

Taiwan also seems enigmatic judging from the statist point of view. The Taiwanese state was one of the heaviest investors in production among the capitalist economies, but most of its export industries were privately owned, and the state's relationship to these operations was at best arm's length. The multitude of small and medium-sized export enterprises basically thrive on their own, competing on managerial ability and cheap labour (Amsden, 1985; Ho, 1980). The linkages between the public enterprise sector and the private sector are also far from organic, as the state is more concerned with maintaining a monopoly of key sectors through state enterprises. Based on these observations, we agree with Evans' exhortation: 'it is more important to ask "what kind" of state involvement rather than "how much" (Evans, 1995, p. 11). To understand the role of the state in development, we need to depart from the dichotomous duplex of free markets and strong states and move towards a more nuanced analysis of the diverse roles assumed by the state in East Asia.

## 6.4 Towards state-business interdependence?

Statist theory emphasizes the state's autonomy in making decisions and its dominating capacity to bring about results in the marketplace. The state-led view of East Asian development, however, quickly gives rise to a theoretical reaction which emphasizes the interactions and institutional linkages between the state elite and societal actors. This new, broadened focus on state-society relations echoes Gilbert and Howe's call for the need to examine the convergence of state and class capacities in bringing about a particular outcome:

We argue that state-centred theorists disregard the *interrelation* of state and society; in viewing the state as an independent entity, they fail to see how it is related to the wider society. Further, they oversimplify societal forces and ignore class conflict within and beyond the state. State and society are interdependent, and must be analyzed as such.

(Gilbert and Howe, 1991, p. 205; emphasis original)

In the discussion of East Asian development, there is a similar view stressing the *interdependence* of state and business. Weiss calls this the 'governed interdependence theory', premised on the proposition that 'the ability of East Asian firms and industry more generally to adapt quickly to economic change is based on a system that socializes risk and thereby co-ordinates change across a broad array of organizations—both public and private' (Weiss, 1995, p. 594). Governed interdependence refers to a system of central co-ordination in which the government and industry co-operate and communicate to bring about innovation and realize competitive potentials. Again, this model has both analytical and policy implications, since it simultaneously generalizes about state-

business co-ordination in East Asia and how this relationship has contributed to the economic success of the region.

This theory departs from the state-led model in its insistence that development policies cannot simply be *imposed* upon the private sector without compromising the effectiveness of such policies. The really effective policies are those formulated in consultation with the private sector and implemented with the willing co-operation of firms. 'Co-ordination' is the key to this system. Why is industrial co-ordination possible? Apart from the autonomy emphasized so much in the state-led models, Weiss attributes the institutional capacities for co-ordination to the proper kind of state-industry linkages. She argues that in Korea, Taiwan and Japan, elaborate matrices of institutional linkages have been established between state agencies and the private sector. Such 'policy networks' provide a vital mechanism for acquiring information and for co-ordinating agreement with the private sector' (Weiss, 1995, p. 600). In Japan, for example, MITI benefits from the work of over 250 deliberative councils which enable the state to consult the private sector and to collect valuable information. Only by so doing can a strong ministry avoid formulating policies in isolation from the private sector and act in concert with industry. These policy networks are themselves vitally aided by the highly 'organized' nature of the Japanese enterprise system, based upon the keiretsu form of business group.

# Keiretsu

*Keiretsu* is the name for the modern Japanese enterprise group. They comprise a collection of firms in various industries (themselves known as *kaisha*) which hold shares in one another's businesses, borrow from the same financial institution which serves all group members, and tend to behave strategically as a group. Some of these groups evolved from pre-Second World War industrial conglomerates knows as *zaibatsu*, which were broken up after the War. The six largest contemporary *keiretsu* are Mitsui, Mitsubishi, Sumitomo, Funo, Sanwa and Ikkan. The first three are former *zaibatsu*, while some well-known former *zaibatsu* joined together to form the Funo group, which includes companies such as the Nissan motor corporation. Other motor companies such as Daihatsu and Isuzu are members of the Sanwa and Ikkan groups respectively; Toyota is a member of the Mitsui group.

These groups are noted for their cross holding of shares, both between companies within the same group and between the groups themselves. The groups are often considered to be led by an important banking company. These relationships form the basis for the strong degree of 'group loyalty' and, to a lesser extent, industry loyalty, which typifies Japanese business activity. In addition to the 'bank-led' groups mentioned above, there are a further series of smaller non-bank, or 'independent', groups. A well-known example of this category is the Sony Corporation, which itself has over eighty subsidiary and affiliate companies in which it owns shares. Unlike the pre-war *zaibatsu*, there are no holding companies that dictate actions

and determine strategy associated with any of these groups. They are groups of 'independent' decision-making companies who simply co-operate with each other. company (*Sogo Shosha*) as part of its overall activity, and there are others attached to some of the other groups. These companies specialize in exporting and importing in a wide range of commodities, and help to organize domestic wholesale and retail activity. The top nine of these general trading companies accounted for about 50 per cent of Japanese exports and 65 per cent of imports in the later 1980s, giving them enormous power over foreign trade.

(Source: Ito, 1992)

(See Chapter 9 for further elaboration on the nature of these Japanese business groups.)

To facilitate public-private co-ordination, it is essential for industry to be organized in an *encompassing* fashion. Fragmentation in interest representation of the private sector will be inimical to co-ordination, while a highly centralized and active peak association among producers is essential. In the Japanese case, the *Keidanren* (federation of economic organizations) and industry-specific manufacturing trade associations provide a basis for co-ordination. In Korea, the highly centralized industrial structure dominated by the *chaebol* invariably facilitates public-private co-operation, and the state also fosters the growth of encompassing peak associations in the private sector. Taiwan is also characterized by the presence of such encompassing institutions as the Taiwan Textile Federation and the Taiwan Electrical Appliances Manufacturer Association (Weiss, 1995, pp. 602–4).

Weiss's theory is in fact a more sophisticated formulation of earlier studies of Japanese state-business relations, and her contribution is also distinctive in attempting to generalize beyond the Japanese case. For example, in his study of energy policies in Japan, Samuels (1987) talks of 'the politics of reciprocal consent' to highlight the political interdependence of state and market. 'Reciprocal consent' is the mutual accommodation of state and market. It is an iterative process of reassurance among market players and public officials, one that works better where the parties to these negotiations are stable and where the institutions that guarantee their compacts are enduring' (Samuels, 1987, p. 8). He points out that as a result of such 'reciprocal consent', the Japanese state pursues an energy policy that largely seeks to guarantee the stability of the private market rather than to compete or displace private entrepreneurship.

Similarly, Okimoto seeks to qualify the record of MITI's industrial policy as a history of unmixed state-led success. Rather than a simple imposition of public priorities over private impulses, he attributed the peculiar effectiveness of Japan's industrial policy to its particular role in consensus building:

It has served as the main instrument for consensus building, the vehicle for information exchange and public-private communication. Close government-business relations would be hard to imagine in its absence. Indeed the whole system of consensus, on which Japan's political economy relies, would be hard to maintain without industrial policy as an integrative mechanism.

(Okimoto, 1989, p. 231)

Calder's study of industrial finance in Japan and his formulation of 'strategic capitalism' also stresses the existence of a hybrid public-private system, 'driven pre-eminently by market-oriented private-sector calculations, but with active public-sector involvement to encourage public spiritedness and long-range vision' (Calder, 1993, p. 16).

While not dealing exclusively with East Asia, Evans also highlights the fact that successful development states could not simply be autonomous, they are also 'embedded in a concrete set of social ties that binds the state to society and provides institutionalized channels for the continual negotiation and re-negotiation of goals and policies' (Evans, 1995, p. 12).

#### Criticisms of governed interdependence

As a set of policy prescriptions, the view outlined above has the promise of correcting the excesses of the state-led models. The simple moral of the story is that either too much or too little state, and too much or too little market, is no good for the long-term growth of the economy. Effective state co-ordination of economic change comes only when there is a proper balance of state and market. The focus on the institutional underpinning of effective policy is valuable. However, while we are sympathetic to the policy prescriptions of this new approach, we find Weiss's model, as an analytical approach, problematic on three counts.

First, by proposing that the governed interdependence model explains why some states are more efficient than others in co-ordinating economic growth, Weiss is, at the empirical level, simply trying to replace one general homogeneous imagery of East Asian growth with another one. Are state-business relations in Japan, Korea and Taiwan equally co-operative, and is the balance of state and business power the same across the three countries? If Japan can be characterized by a high level of public-private co-operation and a balance of bureaucratic autonomy and linkages with the private sector, can the same be said about the Korean case? Amsden has provided ample evidence of how the Korean state sought to change business behaviours in the light of public objectives. The variety of measures by the Korean state to solicit compliance from the private sector is also legendary, such as the use of tax investigations to rein in recalcitrant firms. Furthermore, how much can we attribute Taiwan's export success to public-private coordination? For one thing, Taiwan's export manufacturing is characterized by a multitude of small and medium-sized firms, which does not seem to be too conducive to the kind of centralized co-ordination that Weiss envisages. The political schism for most of the postwar years between the Mainlander Kuomintang (KMT) regime and the primarily Taiwanese small entrepreneurs also makes co-operation difficult.

Second, Weiss chides previous research as being static in scope, typically basing their conclusions on a snapshot view of government-industry relations frozen in time (Weiss, 1996, p.606). She maintains, quite rightly, that it is essential to capture the dynamic picture of state-business relations. It is unclear, however, what time frames her model is deemed applicable to. If governed interdependence is the primary factor explaining effective state 'intervention' in East Asian development, we should expect it to be able to help us make sense of the post-war industrial take-off. Yet Weiss also points out that government-business relations have been changed 'from a situation in which government (the economic bureaucracy) was largely making the big decisions—usually out of

necessity, due to a destabilized or underdeveloped business sector—to a situation in which the private sector is being increasingly invited to take more initiatives' (Weiss, 1995, p.607). If this is true (and we basically agree that it is), the problem becomes whether Weiss thinks that past economic growth in Taiwan and Korea can be explained by public-private co-operation or not, since she acknowledges that state-business relations are undergoing a process of transformation in these two economies. Is she also guilty of over-generalizing from current experiences? For example, if the relationship between the KMT-controlled state and the Taiwanese bourgeoisie can be viewed as co-operative since the 1980s, the same certainly cannot be said for earlier decades.

Third, in generalizing the idea of governed interdependence to the case of Taiwan, Korea and Japan, Weiss is liable to mix explanations with prescriptions. Even if coordination, communication and co-operation between the public and private sector is the ideal pattern of relations between the two, Weiss's analysis still poses the question of whether this model adequately explains the economic success of the three economies. Prescribing public-private co-ordination is one thing; arguing this has actually happened and attributing past success to it are entirely different issues. If we could conceptualize, for example, a continuum of different degrees of state-business co-operation, an interesting question would be: what are the conditions responsible for co-operation, or, conversely, for non-co-operation? We would then see state-business relations as a result of the incessant negotiation, bargaining, and even conflicts and struggles between state agencies and private firms. Their interests may converge in certain countries, but diverge in others. Or public-private co-operation may be more possible in certain sectors within certain countries but not others. Seen in this light, we need to ascertain whether cooperation is really the case in a certain economy over a certain period, and then try to deduce the possible impact on economic performance. In other words, we need to separate prescription from explanation in the analysis of the role of the state in East Asian development, or even developmental processes in general. For example, while Evans (1995) also prescribes state-society co-ordination, he is careful to examine the conditions both for its presence and absence in different developing countries.

We are more inclined towards the view of the relationship between 'polities' and development expounded by Friedman (1988). He argues that statists such as Johnson often subscribe to a view of development which is inherently neo-classical. It is based on an apolitical conception of the relationship between state action and industrial change. The role of the state, in this view, is to implement rules or allocate resources which prompt private sector actors to adopt the most efficient 'best practices' of production. Friedman adds:

Comparative research in political economy was often refined to the coding of different forms of state regulation and the attempt to show which practices best enhanced efficiency. The problem, however, and one that the Japanese case highlights, is that politics may be much more significant than the conventional research scheme admits. The Japanese case demonstrates that significant political events affecting industrial development may not involve the state at all, but rather are shaped by worker ideologies, interfirm co-operation, and the like. The political resolution of conflicts in these areas may often be of importance equal to or greater than that of state actions.

(Friedman, 1988, p. 209)

As we have delineated elsewhere, it is necessary to consider the organizational framework and wider institutional context in which industrial development occurs, rather than simply focusing on the role of the state (Chiu *et al.*, 1997). By placing 'polities' at the centre of the analysis, we are steadfastly committed to putting the empirical analysis of the complex roles of the state in development rather than simply generalizing on the best practice of state intervention. In some cases, it is undeniable that the role of the state is critical in orchestrating the institutional configurations of development. Even in these cases, however, it is important to recognize the wider political origins of state actions. Furthermore, as our comparison of Hong Kong and Singapore indicates, there may be no one single best practice for development. Different roles of the state, coupled with different institutional frameworks, may produce equally impressive economic growth but with divergent patterns and distributional consequences.

#### 6.5 The many faces of the state in East Asia

A good way to re-orient our conceptualization of the role of the state in East Asian development is to re-examine the divergent experiences of state interventions in the postwar era. Without going into the merits of different sets of policy prescriptions, we believe that the market-led, state-led, and governed interdependence models are all unable to capture the empirical and historical complexity of the roles of the state in East Asian development. Simply put, we have seen not one single model, but *models* of East Asian development. Such a typology helps us address the region's diversity without lapsing into empiricism. Coming to grips with the different strategic roles played by the state in industrialization in Korea, Taiwan, Singapore and Hong Kong will help us decipher a major source of diversity in the region. Different courses of state action constitute the most marked difference among the industrialization patterns of the East Asian economies, and these, in their turn, exacerbate other differences.

If we set the case of Japan aside, there are indeed some outstanding similarities in the East Asian NIEs' developmental patterns. As discussed earlier, the East Asian NIEs shared remarkably similar timing in their industrial take-off and rates of economic growth. Also commonly acknowledged as important are export and industrial expansion for generating economic growth (Balassa, 1988; Chen, 1979). However, recent studies show growing sensitivity to the diversity in developmental experience within the East Asian region. For example, Haggard (1990) identifies two development trajectories among the East Asian NIEs: Korea and Taiwan share the same export-led growth strategy, while Hong Kong and Singapore share an entrepôt (transhipment) growth strategy. Cheng, on the other hand, seeks to account for the East Asian NIEs' industrial restructuring process in the 1970s. As he states:

The four East Asian NI[E]s diverged in their responses to the identical challenges [of industrial restructuring] specified above. In rhetoric, all of

the four NI[E]s in 1971 simultaneously announced their intent to restructure industry in order to adjust to new parameters in international economic systems. In reality, the four NIEs diverged in their approaches to industrial and economic changes, as manifested in the variation of their industrial policies, mechanisms that implement policies, and policy outcomes.

#### (Cheng, 1987, p. 15)

These studies represent a significant advance over previous studies that lumped the East Asian NIEs into a single model of development. Nevertheless, they still do not give sufficient attention to the diversities in the development trajectories and strategies of East Asian NIEs. While recognizing the differences among the NIEs in his concrete analysis, Haggard overlooks such differences in conceptualizing a typology of development strategies. According to his typology, all East Asian NIEs pursue an export-led growth strategy. The differences between Singapore and Hong Kong are also not captured by the so-called common entrepôt (transhipment) growth strategy. Cheng, on the other hand, recognizes the East Asian NIEs' divergent responses to industrial restructuring, but assumes a common developmental strategy among the four countries *before* their restructuring. 'While the four East Asian NIEs converged to initiate the ELI [export-led industrialization] and became the same crop of small and open economies, they part company with each other in the advanced stage of ELI' (Cheng, 1987, p. 46). Building on these analyses, we will now briefly discuss the salient differences in the state's role among the East Asian economies.

If there are controversies over whether Japan is really a developmental state or whether the state had a major impact on the post-war 'miracle', there is no doubt about the archetypical status of the South Korean state as the strong developmental state in East Asia. If Japan could claim to have a relatively strong bourgeoisie after the Second World War, South Korea had only a minuscule business community at tFhe aftermath of independence. The economy was in a real shambles in the early 1950s after the Korean War. The Korean state, especially after the assumption of power by Park Chung Hee, did indeed have a major impact on the course of development from the 1960s. We do not want to go into the hypothetical question of what would have happened had the state acted differently; this would not have been possible given the economic and political circumstances of the time. What we want to say is that the actions of the Korean state are inseparable from what we consider to be an economic success story, or at least inseparable from the specific ways in which the South Korean economy attained its current level of development. The state channelled large amount of foreign aid and credit into a selected group of large conglomerates, the chaebol, and helped them to grow into Fortune 500 calibre. It opened up the economy to foreign trade and pursued aggressively an active industrial policy to promote export-oriented production. Compared with Japan, the Korean state did all a developmental state is supposed to do, and more.

The case of Taiwan is different. While we agree that the Taiwanese state under the KMT has largely been active, playing a significant role in the course of post-war industrialization, we think that most proponents of a developmental state in Taiwan ignore an important fact about Taiwan's post-war development: the role of small and medium-sized enterprises (SMEs). These SMEs have been the backbone of Taiwan's

manufacturing sector and have generated much of the export earnings. They are deeply inserted into the international division of labour and are important producers of a large variety of consumer and capital products, for example, auto parts, shoes and garments. Since the 1980s a new breed of high technology SMEs has emerged to produce electronics components and computer hardware.

Taiwanese SMEs have of course benefited from the relatively sound and stable monetary and fiscal policies, the export credit and tax rebates system, as well as the impressive investments in infrastructure and education. Nevertheless, the existing record suggests that the multitude of Taiwanese SMEs have been relatively untouched by government industrial policies (except perhaps the export credit system), at least during the hyper-growth period of the 1960s and '70s. There have always been complaints about government policy not benefiting the SMEs in Taiwan; for example, the state-owned commercial banks have been regarded as acting more like pawn shops (imposing strict collateral requirements) by many SME entrepreneurs, and it has been difficult for the latter to get credit from them. Instead, a bustling informal financial market has met most of the SMEs' credit needs. In contrast, the public enterprises benefit disproportionately from bank credits and foreign loans. The SMEs in Taiwan thrive largely on their flexibility and a dense network of collaborations among themselves.

The cases of the two East Asian city economies also pose interesting contrasts with the larger East Asian economies, and also diverge between themselves. They shared a similar colonial experience, were 'underdevel-oped' and faced a similar set of 'external' pressures and constraints. We documented in an earlier work how Hong Kong and Singapore parted ways in the early post-war era from a common entrepôt and British colonial legacy (Chiu *et al.*, 1997).

In Singapore, the state played an active role in shaping post-war industrialization, but in ways markedly different from Taiwan, South Korea and Japan. All three of the large East Asian states have jealously guarded their domestic markets from foreign imports with a protectionist tariff and quota system. They have also restricted the inflow of foreign direct investments, although with substantial variations among the three in terms of the rigidity and scope of restrictions (with Taiwan less restrictive than the others). Singapore, on the other hand, has kept its economy open to foreign trade and investment. In fact, the basic tenor of the state's developmental strategy is to actively promote and attract foreign direct investment in Singapore's manufacturing sector. Since the 1960s, the Singaporean state under the ruling People's Action Party has installed a fiscal incentive system to various 'pioneer' industries and firms which substantially benefits foreign firms producing locally. The state tamed the once militant labour movement and built an industrial relations system based on tripartite co-operation between the state, management and trade unions. Government infrastructure constructions served an important role in facilitating the coming of foreign firms. State-invested enterprises also seek to induce foreign investments by forming strategic alliances with foreign partners, thus reducing significantly the risk of such undertakings.

Hong Kong, by contrast, has not pursued a selective industrial policy. Throughout the course of its post-war industrial development it has repeatedly rejected suggestions from its manufacturers for assistance schemes or incentives directed specifically towards manufacturing industries. In the 1950s, for example, it refused to establish an industrial bank, commonplace in other East Asian economies, to provide long-term credits to

industries. The Industry Department plays only a supportive role for manufacturers and the most salient areas of government assistance to industries have been in infrastructure and overseas promotions of domestic exports. The Hong Kong state has striven to maintain a universalist and non-selective approach towards economic development, rather than giving special favours to specific sectors. It has, however, played a major role in the provision of means of collective consumption which help keep the lid on wage increases. It has been the largest landlord in Hong Kong through its massive public housing programme; the education system, from primary to tertiary institutions, has largely been state funded; and the public health system provides the population with cheap medical services. The most visible role of the state in Hong Kong's development is, of course, in its renowned low and simple tax system, as well as the British legal framework that stresses the enforcement of contracts. While lately the state has moved to give more selective incentives to high technology industries, such efforts have been widely regarded as too little, too late.

#### 6.6 Conclusion

The focus of this brief survey of the role of the state in East Asian development has been analytical. We have concentrated on the analytical value of different models; that is, have they been able to unearth the empirical and historical patterns of state actions in the postwar development of East Asia? Are their generalizations accurate in view of the available information, especially from the comparative vantage point? With these considerations in mind, we raised analytical doubts about the market-led, state-led and interdependence models of East Asian development, arguing that these models have not been able to capture adequately the empirical and historical diversities among Japan and the East Asian NIEs in terms of the role of the state in development. Neither the imageries of the free market, the strong development state, nor state-business co-operation have been able to describe the experiences of the East Asian economies.

We have not gone into great detail regarding the prescriptive side of these models, although our doubts about their analytical value should guide any interpretations of wholesale generalizations regarding the proper role of the state. Indeed, the East Asian economies followed very different kinds of prescriptions. Among those states commonly assumed to be strong, the South Korean state dominated business, in Japan the state and private business excel in co-ordination and communication, and in Taiwan the state combined a heavy-handed intervention through state enterprises with a benign neglect of small export enterprises, at least until the 1980s. Between the two alleged weaker states, the Singapore state developed strength from weakness and has become increasingly visible in economic affairs since independence, while the Hong Kong colonial state steadfastly resisted the temptation of active industrial policy, but involved itself deeply in other areas.

It is still possible, however, to make some more modest generalizations. From a broader comparative point of view, all five states exhibit what Evans (1995) call 'embedded autonomy', that is, combining an autonomous and strong state bureaucracy with an embeddedness in a network of ties with the private sector. In this view, the internal organization of these states approach a Weberian bureaucracy: meritocratic,

coherent, and with a clear chain of command. At the same time, they also take into account inputs from the society when formulating policies and are not simply an insulated Leviathan sitting above the civil society (see Rueschemeyer and Evans, 1985). To (over)simplify Evans' argument, successful state intervention depends on a competent civil service and the willingness of the state elite to listen to opinions from the society. The opposite of these are the many African states, where the state is 'soft', personalistic, particularistic and patrimonial (Hyden, 1983).

Still, that does not mean that a strong state has to be an interventionist state, or that the intervention has to be of a particular kind. On this our position is more circumspect. The proper mix of public policies to promote economic development, as the East Asian experiences suggest, is probably contingent and hinged upon the concrete situation. In Hong Kong, where private entrepreneurship is dynamic, a more restrained state suffices. In post-war Japan, MITI gave the already well-developed private sector a push and offered administrative guidance regarding the direction of development. In South Korea and Singapore, the state intervened actively in the absence of a dynamic local bourgeoisie. In Taiwan, the Nationalists erected a huge edifice of state enterprises and kept a distance from indigenous Taiwanese enterprises, but local entrepreneurs sprang up to take advantage of the infrastructure and political stability generated by the state. The remarkable success of the East Asian economies suggests that these policies, ranging from *laissez-faire* to omnipresent intervention, are all effective to a certain extent, relative to their particular national circumstances. The relationship, however, between the concrete historical situations and the need for a specific kind of state action is clearly beyond the scope of the present discussion. More important, we argue that even if there is a need for a certain policy, or a policy is clearly counterproductive, there is no guarantee that the good policy will be implemented or the bad policy avoided. The political determinants of state development policies are, therefore, analytically as important as, if not prior to, the prescriptive question.

#### References

- Amsden, A. (1985) 'The state and Taiwan's economic development'in Evans, P., Rueschemeyer, D. and Skocpol, T. (eds) *Bringing the State Back In*, Cambridge, Cambridge University Press.
- Amsden, A. (1989) *Asia's Next Giant: South Korea and Late Industrialization*, New York, Oxford University Press.
- Appelbaum, R. and Henderson, J. (eds) (1992) States and Development in the Asian Pacific Rim, Newbury Park, Sage.

Balassa, B. (1988) 'The lessons of East Asian development: an overview', *Economic Development and Cultural Change*, vol. 36, no. 3 (supplement), S273–90.

Balassa, B. et al. (1982) Development Strategies in Semi-Industrial Economies, Baltimore, Johns Hopkins University Press.

Bradford, C. (1986) 'East Asian "models": myths and lessons' in Lewis, J.P. and Kallab, V. (eds) *Development Strategies Reconsidered*, New Brunswick, Transaction Books.

- Calder, K. (1993) Strategic Capitalism: Private Business and Public Purpose in Japanese Industrial Finance, Princeton, Princeton University Press.
- Castells, M. (1992) 'Four Asian tigers with a dragon head: a comparative analysis of the state, economy and society in the Asian Pacific rim' in Appelbaum, R. and Henderson, J. (eds) *States and Development in the Asian Pacific Rim*, Newbury Park, Sage.

- Chen, E.K.Y. (1979) Hyper-Growth in Asian Economies: A Comparative Study of Hong Kong, Japan, Korea, Singapore and Taiwan, London, Macmillan.
- Cheng, Tun-Jen (1987) 'The politics of industrial transformation: the case of the East Asian NICs', PhD Dissertation, University of California, Berkeley.
- Chiu, S.W.K., Ho, K-C. and Tai-lok Lui (1995) 'A tale of two cities rekindled: Hong Kong and Singapore's divergent paths to industrialism', *Journal of Developing Societies*, vol. 11, pp. 98–122.
- Chiu, S.W.K., Ho, K-C. and Tai-Lok Lui (1997) *City-States in the Global Economy: Industrial Restructuring in Hong Kong and Singapore*, Boulder, CO, Westview.
- Deyo, F. (ed.) (1987) *The Political Economy of East Asian Industrialism*, Ithaca, Cornell University Press.
- The Economist (1996) 'In the shadow of the chaebol', 6 July, pp. 57-8.
- Evans, P. (1995) *Embedded Autonomy: States and Industrial Transformation*, Princeton, Princeton University Press.
- Friedman, D. (1988) *The Misunderstood Miracle: Industrial Development and Political Change in Japan*, Ithaca, Cornell University Press.
- Gerschenkron, A. (1962) *Economic Backwardness in Historical Perspective*, Cambridge, Mass, Harvard University Press.
- Gilbert, J. and Howe, C. (1991) 'Beyond "state vs society": theories of the state and new deal agricultural policies', *American Sociological Review*, vol. 56, no. 2, pp. 204–20.
- Gold, T.B. (1986) State and Society in the Taiwan Miracle, Armonk, New York, M.E. Sharpe.
- Haggard, S. (1990) Pathways from the Periphery: The Politics of Growth in the Newly Industrializing Countries, Ithaca, Cornell University Press.
- Hyden, G. (1983) No Shortcuts to Progress: African Development Management in Perspective, Berkeley, University of California Press.
- Ho, S.P.S. (1980) 'Small-scale enterprises in Korea and Taiwan', World Bank Staff Working Paper No.384, Washington, World Bank.
- Ito, T. (1992) The Japanese Economy, Cambridge, Mass, MIT Press.
- Johnson, C. (1982) MITI and the Japanese Miracle, Stanford, Stanford University Press.
- Johnson, C. (1987) 'Political institutions and economic performance: the government-business relationship in Japan, South Korea and Taiwan' in Deyo, F. (ed.) *The Political Economy of the New Asian Industrialism*, Ithaca, Cornell University Press.
- Jones, L. (1994) 'Big business groups in South Korea: causation, growth and policies' in Lee-Jay Cho and Yoon Hyung Kim (eds) Korea's Political Economy: An Institutional Perspective, Boulder, CO, Westview.
- Kuznets, P.W. (1988) 'An East Asian model of economic development: Japan, Taiwan, and South Korea', *Economic Development and Cultural Change*, vol. 36, no. 3 (supplement), pp. 11–43.
- Little, I.M.D. (1981) The experiences and causes of rapid labour-intensive development in Korea, Taiwan Province, Hong Kong and Singapore and the possibility of emulation' in Lee, E. (ed.) *Export-led Industrialization and Development*, Bangkok, ILO-ARTEP.
- Luedde-Neurath, R. (1986) Import Control and Export-Oriented Development, Boulder, CO, Westview.
- Okimoto, D.I. (1989) Between MITI and the Market, Stanford, Stanford University Press.
- Onis, Z. (1991) The logic of the developmental state', *Comparative Politics*, review article, vol. 24, no. 1, pp. 109–26.
- Rueschemeyer, D. and Evans, P. (1985) 'The state and economic transformation: toward an analysis of the conditions underlying effective intervention' in Evans, P., Rueschemeyer, D. and Skocpol, T. (eds) *Bringing the State Back In*, Cambridge, Cambridge University Press.
- Samuels, R. (1987) *The Business of the Japanese State: Energy Markets in Comparative and Historical Perspective*, Ithaca, Cornell University Press.
- Steers, R., Yoo Keun Shin and Ungson, G. (1989) *The Chaebol: Korea's New Industrial Might*, New York, Harper and Row.

- Wade, R. (1988) 'State intervention in "outward-looking" development: neo-classical theory and Taiwanese practice' in White, G. (ed.) Developmental States in East Asia, London, Macmillan.
- Wade, R. (1990) Governing the Market: Economic Theory and the Role of Government in East Asian Industrialization, Princeton, Princeton University Press.
- Wade, R. (1992) 'East Asian economic success: conflicting perspectives, partial insight, shaky evidence', World Politics, review article, vol. 44, no. 2, pp. 270–320.
- Wade, R. (1993) 'Managing trade: Taiwan and South Korea as challenges to economics and political science', *Comparative Politics*, vol. 25, no. 2, pp. 147–67.
- Wade, R. (1996) 'Japan, the World Bank and the art of paradigm maintenance: *The East Asian Miracle* in political perspective', *New Left Review*, no. 217, pp. 3–36.
- Wade, R. and White, G. (eds) (1984) 'Developmentalist states in East Asia', *Institute of Developmental Studies Bulletin*, special issue, Brighton, University of Sussex.
- Weiss, L. (1995) 'Governed interdependence: rethinking the government-business relationship in East Asia', *The Pacific Review*, vol. 8, no. 4, pp. 589–616.
- Woo, Jun-en (1991) *Race to the Swift: State and Finance in Korean Industrialisation*, New York, Columbia University Press.
- World Bank (1993) *The East Asian Miracle: Economic Growth and Public Policy*, New York, Oxford University Press.

# PART 2 The Processes of Integration

# CHAPTER 7 Economic integration in the Asia-Pacific region

Ippei Yamazawa

### 7.1 Introduction

It seems to be a universal phenomenon that, while private firms expand their activities beyond national borders and become multinational companies, individual national governments-of both developing and developed countries alike-try to attract multinational firms to start up on their home territories in order to take advantage of their technology and capital for stimulating new industries and generating production and employment. This phenomenon has not taken place uniformly across the world, but more intensively in some geographical areas than in others. It is a process called 'regional economic integration'. The form regional integration takes ranges from the elimination of tariffs and non-tariff barriers to the harmonization of domestic regulations and policies, and even to the creation of a single currency area. Its most advanced form can be found in the European Union (EU), which has undertaken extensive integration for the past forty years, and now aims for monetary union and possibly political union sometime in the future. The North American Free Trade Agreement (NAFTA) is at a lower level of integration as it moves towards becoming a free trade area (FTA). These FTAs not only focus on the elimination of cross border measures that limit trade, but also introduce the harmonization of domestic commercial conditions to varying degrees.

In the Asia-Pacific region, a much looser form of integration has been going on since 1989 known as the Asia Pacific Economic Co-operation (APEC) forum. Although the extent of its integration remains at a lower level, its membership coverage and possible leadership role in the world economy attracts attention to its current developments. Why do the APEC countries prefer a loose form of integration? Can they achieve real integration of practical significance, which is different from that of the European Union or NAFTA? Can they prevent discrimination against non-members, or prevent co-operation with non-members? After analysing the major characteristics of APEC, I will examine two questions: how will APEC develop in the future? and second, will APEC liberalization be implemented



in a manner consistent with the multilateral regime that works under the auspices of the World Trade Organization (WTO)?

#### 7.2 Emerging regional integration in the Asia-Pacific

The Asia-Pacific Economic Co-operation forum is an emerging regional group of eighteen member economies in the Asia-Pacific. In 1997 it comprised eleven Asian countries; six ASEAN members plus China, Hong Kong, Japan, South Korea, and Chinese Taipei; the three Oceania countries of Australia, New Zealand, and Papua New Guinea (PNG); and the four American countries of Canada, the USA, Mexico, and Chile. In the run up to its Subic meeting in the Philippines in November 1996, APEC member governments prepared detailed concrete programmes for regional co-operation. APEC has now become a major regional group, second only to the European Union in terms of its membership coverage. The combined share of its eighteen members in the world total GDP and export trade amounted to 47.8 per cent (in 1989) and 39.1 per cent (in 1990) respectively, which should be compared with 23.7 per cent and 40.9 per cent for EU12 in the same years.

APEC began in 1989 as a series of annual ministerial meetings. However, since 1993 the national leaders have met each year, and these meetings have been setting its future direction. In Seattle in November 1993, the APEC leaders envisioned APEC as 'a community of Asia Pacific economies', a flexible forum for promoting economic growth in the region, which is quite different from such tightly structured organizations as the European Union and the North American Free Trade Agreement.

In Bogor in November 1994, the leaders declared their political commitment to the idea that developed members should achieve free and open trade in the region by 2010 and the rest should achieve the same goal by 2020. In addition, trade 'facilitation' (enhancement) and development co-operation would be promoted as well. Trade liberalization in the region for the next 25 years will be slower than liberalization schemes elsewhere but it is supplemented by the facilitation and development co-operation processes (APEC, 1995a; APEC/EPG, 1994).



Figure 7.1 Map of members of APEC

In 1995 the Osaka Action Agenda provided a guideline for implementing the policy measures designed to reach these targets. In 1996 the APEC leaders adopted the Manila Action Plans for APEC (MAPA) in which all members submitted their individual action plans (IPAs) to be implemented from 1997 onwards.

It should be remembered that economic co-operation in the Pacific region was not something that arose just in the past few years. The movement for economic co-operation started in the middle of the 1960s and experienced three surges of growth. These developments in the Pacific had a close and observable connection to moves in the development of the European Community. The first proposal for Pacific economic co-operation was made by economists and businessmen in the 1960s, stimulated by the successful development of the European Common Market. Annual gatherings of businessmen, known as the Pacific Basin Economic Council (PBEC), started in 1967 and a gathering of economists, the Pacific Trade and Development conference (PAFTAD), began in 1968. Table 7.1 presents a history of those initiatives as well as of APEC.

Pacific co-operation activity decelerated during the oil crisis in the middle of the 1970s but revived toward the end of that decade, partly benefiting from the oil and resources boom in the ASEAN countries. The Japanese Prime Minister Masayoshi Ohira made a proposal for a 'Pacific community' in his inaugural speech in 1978. Coming from an influential politician, this proposal extended the discussion of Pacific co-operation

Date	Initiative
1967	Association of South-East Asian Nations (ASEAN) started with five members
Late 1960s	Proposals for Pacific economic co-operation; PBEC (1967) and PAFTAD (1968)
1978	Proposal for Pacific Economic Community by Japanese PM (Ohira). Report of PEC Study Group (1979)
1980	The first Pacific Economic Co-operation Council (PECC) meeting (Canberra)
1989	Canada-USA Free Trade Agreement (CUSFTA)
1989	Asia Pacific Economic Co-operation (APEC) Ministerial meeting started in Canberra with 12 members
1990	APEC II (Singapore)
1991	APEC III (Seoul); three Chinas joined, Seoul Declaration
1992	APEC IV (Bangkok); Eminent Persons Group established
1993	CUSFTA expanded to NAFTA by including Mexico
1993	APEC V (Seattle); Informal Economic Leaders Meeting, Mexico joined
1994	APEC VI (Jakarta/Bogor); Bogor Declaration, PNG and Chile joined
1995	Vietnam joined ASEAN
1995	APEC VII (Osaka); Osaka Action Agenda adopted with Initial Actions announced
1996 March	Asia-Europe Summit Meeting (AESM) in Bangkok
1996 November	APEC VIII (Subic); action plans announced and implemented by individual members
1996 December	WTO Ministerial Meeting in Singapore
1997	APEC IX (Canada)
1998	APEC X (Malaysia)
1999	APEX XI (New Zealand)

# Table 7.1 History of economic co-operation inAsia-Pacific

beyond economists and businessmen to a wider circle of politicians, diplomats, academics in international relations, and the mass media. Subsequently, Prime Minister Ohira established a study group which prepared a report on Pacific co-operation in 1979. During his visit to Australia and New Zealand, Ohira showed his proposal to the prime ministers of the two countries who welcomed it. Australian Prime Minister Malcolm Fraser organized a seminar in Canberra in 1980 on economic co-operation which has continued as the Pacific Economic Co-operation Council (PECC) forum. It started with thirteen members, five developed countries (the liberal market economies of Australia, Canada, Japan, New Zealand and the USA), six ASEAN members, South Korea, and a

Pacific island nations group; it added China and Chinese Taipei in 1986. It now comprises 21 members following the addition of Hong Kong, Mexico, Chile, Peru, Columbia and Russia, giving it a larger membership than the current APEC. The PECC has the unique tripartite participation of business, academia, and government, all acting in their private capacities. It has been providing a regular free exchange of views and extensive research in order to form a region wide consegsus on various issues of economic of economic co-operation in the Asia-Pacific.

In Europe during this period (the late 1970's) momentum toward integration slowed down, and European economies stagnated, an event that became known as the 'European crists'. The stagnation continues into the early 1980's, when the European Community countries introduced their 'single European Market' programme. They were pushed along in this direction by the rapid industrialization of the Asia-Pacific countries and the increased export of Asian products to the European market. The EU also added three new members, Greece, spain, and Portugal, bringing its membership to twelve. Deeper integration and an enlarged market stimulated European firms to increase their businees activity and engage in an active mergers and acquisitons policy acroos national borders. This brought on a boom, often termed the 'European fever', in the latter half of the 1980s. The proposal for APEC, the first official inter-governmental forum on economic co-operation in the Asia-Pacific, came during this third surge in European integration in the late 1980s.

Reflecting its Asian members' preference, APEC started with a loose, informal structure, with no treaty like the Treaty of Rome or the Masstricht Treaty. But this informal structure has tended to mislead some writers encouraging them to underrestimate its possible impacts. However, its membership coverage and current action agenda should assure any sceptics that APEC can take a joint initiative in managing the world economy together with the European Union (see Chapter 13).

# 7.3 Co-Operation from diversity

The following three key terms describes APEC's major characteristics:

- diversity among its members
- high growth potential
- informal structure.

A large diversity among members is the main features of Asia-Pacific economics (see Chapter 2). This is more so than for any other regional economy in the world.

1 There is a difference in natural in the reasources endowmnet and in size of geographical area.

- 2 There is a great difference in the stage economic development; some have already matured into advantage economies while others have only begun to take off over the past two decades, and still have a high growth potentional.
- 3 They are divided into several groups on religious grounds, their cultural heritage, and in terms of their philosophical values and outlooks. huntingtion (1993) contended that in the post-Cold War era the conflict between different ideologies would be replaced by a

'clash of civilizations' between different groups countries. However, the Asia-Pacific has so far avoided a severe clash of civilization.

- 4 They were divided into market and socialist economics in the Cold War era. Although some socialist economies are being transformed into market economies, it will take them many decaded to complete their transformation.
- 5 Three sub-groups of free trade areas (FTAs) exist in the region promoting their own liberalization programmes: the North American Free Trade Agreement (NAFTA), the ASEAN Free Trade Area (AFTA), and the Australia-New Zealand Closer Economic Relations Agreement (ANZCER), all these having predated the formation of APEC. They have been active since the late 1980s, partly interaction with the APEC process (see Table 7.1).
- 6 In addintion, several sub-regional informal economic zones (SREZs) have emerged. These are the 'Growth triangle', the Greater South China economic zone, the Yellow Sea economic zone, the Baht zone, and the Japan Sea economic zone, all shown in Figure 7.2 and defined in Table 7.2. Each of these is composed of neighbouring provinces in different



# Figure 7.2 Map of the sub-regional economic zones in East Asia

countries closely interlinked through trade, investment, and personnel movement across national borders, and forming a unique base for rapid development in the Asia-Pacific.

Table 7.2 Constituent members of the sub-regional economic zones in East Asia

- 'Growth triangle': Singapore, Johore State of Malaysia, and Batam and Liau Islands
- Bant zone: Thailand, Laos, and Cambodia
- *Greater South China economic zone:* Hong Kong, China's Guangdong and Fujian provinces, and Taiwan
- *Yellow Sea economic zone:* coastal areas facing the Yellow Sea of North and North-East China, North and South Korea, and Japan
- Japan Sea economic zone: coastal areas of North-East China, Russian Far East, South and North Korea, and Japan.

#### Box 7.1 Sub-regional informal economic zones

Before we go on to analyse the APEC process further it is worthwhile examining the mechanism underlying the development of the East Asian SREZs (see also Brook, 1998). They are characterized as follows: first, each forms a natural economic territory across a national border, in which neighbourhood trade has been activated at some time or another. Second, there are both a centre and a periphery in each SREZ, where the growth of the centre tends to be constrained by its limited area and labour force. These pressures at the centre are transmitted to the periphery through trade, investment, and labour movement. Third, no formal agreements are made between the governments concerned. In some SREZs, free trade zones are established by local governments in the periphery in order to accelerate the transfer of industries from the centre. In the 'Growth triangle', Singapore and the adjacent Johore State of Malaysia have had long developed border trade, which was accelerated partly by the inflow of foreign direct investment in this area since the late 1980s, and partly by the limit on further development of Singapore's territory and its human resources (see Chapter 2). The development of Batam Island followed a tacit Chok Tong, the respective leaders of Indonesia and Singapore, but it has agreement in 1988 between President Suharto and Prime Minister Goh been stimulated by the same market mechanism, as in the case of Johore, and its management has been left to the private sectors of the two countries.

Fourth, all SREZs except for the 'growth triangle' are located along the borders between market and socialist economies. Formerly, trade and investment exchanges between both sides were severely restricted. With the end of the Cold War and the adoption of more open economic policies by the socialist countries, these border zones have begun to show signs of economic vitality, similar to other areas of the region, and cross-border movements of merchandise, investment, and even people have begun to thrive. In the case of the Greater South China zone and the Yellow Sea zone, it goes without saying that the development of cross-border economic

activity in these regions owes much to the political acquiescence, and even encouragement, of the governments concerned. This has taken the form of, for example, Chinese leaders outlining the country's new open economic policy and a tacit acceptance of the situation by the governments of Taiwan and South Korea. As for the South Chine zone, confidence in its future is very much dependent on the general perception of conditions now that Hong Kong has been returned to China. However, these kinds of informal relations are different from the formal agreements governing relations among countries of the EU and NAFTA, since they involve no officially established mechanism for the management of the ongoing economic integration.

Finally, the region has never attempted any region-wide *formal* integration and thus it is often called a 'market-driven integration' of the Asia-Pacific *vis-à-vis* an 'institutional integration based on a treaty' as in the EU and NAFTA.

Returning to the diversity of the APEC countries, this has two aspects, one positive and the other negative. On the positive side, Asia-Pacific economies have taken advantage of their diversity and the resulting economic complementarity between them (see Chapter 2). A wide ranging disparity and different resource endowments have generated high economic complementarity within the region, which in turn has stimulated active trade and investment and enabled many developing members to achieve high growth rates—the 'East Asian miracle' (see Chapter 3). The developed economies could maintain their growth through export of resource products, technology products, and high value-added services and invest in the manufacturing activities of the developing economies. Thus Asia Pacific economies have achieved a strong interdependence and rapid development of the region. Intra-APEC trade increased from 56 per cent in 1980 to 66 per cent in 1990, which compares with an increase from 53 per cent to 63 per cent within the EC12. The main motivation for promoting APEC is to sustain high growth in the region which is shared by all members.

On the negative side, a result of this diversity is insufficient understanding of each member's position and interests, it requires more time to form a region-wide consensus and to take joint action by members. This diversity has made it difficult to start with any formal structures. Nevertheless, its market driven integration is now affected by persistent imbalances and frequent trade disputes between members (see Chapter 2). Bottlenecks caused by poor infrastructures in the member states with a developing economy impede further realization of their high growth potential. Asia-Pacific economies need to strengthen their co-operation in order to mitigate these impediments to further growth in the region.

Thus the three key terms of diversity, high growth potential and informal structures are interrelated and are all reflected in the Osaka Action Agenda. I will examine this now in greater detail.

## 7.4 Osaka Action Agenda and its implementation

The Osaka Action Agenda consists of two parts. Part One covers trade liberalization and facilitation (the reduction of the costs of doing business) and Part Two covers economic and technical co-operation (APEC, 1995b).

Trade liberalization and facilitation

The action agenda for trade liberalization and facilitation starts with nine general principles; comprehensiveness, WTO-consistency, comparability, non-discrimination, transparency, standstill, simultaneous start/continuous process/differentiated timetables, flexibility, and co-operation.

It has an extensive coverage of fifteen areas: tariffs, non-tariff measures, services, investment, standard and conformance, customs procedures, intellectual property rights, competition policy, government procurement, deregulation, rules of origin, dispute mediation, mobility of business people, implementation of the Uruguay Round outcomes, information gathering and analysis. The action agenda suggests a menu of actions by individual member governments and concerted actions by all members in individual areas.

Its new and particular modality—the way to implement liberalization and facilitation programmes—is the 'concerted unilateral liberalization'. That is, individual member governments announce unilaterally their own liberalization and facilitation programmes and implement them in accordance with their own domestic rules. However, individual APEC members also closely watch each other's liberalization programme and its implementation. Pressures then build up on each member so that they feel obliged to submit a liberalization programme as extensive as those of their neighbours. They are also encouraged to implement the programmes they have committed themselves to so as not to lose face. The whole process relies upon a 'peer pressure' among APEC members to urge all members to join the liberalization programme. This is the essence of concerted unilateral liberalization.

This modality is often criticized as non-assertive in comparison with the Western approaches to negotiating, as demonstrated by the GATT and WTO liberalization agreements. These are legally binding on the signatories. This means they can be punished and sanctioned if they fail to implement their commitments. At such an initial stage this legalistic approach to negotiations cannot be accepted by the Asian members of APEC in particular. However, this should not be understood as Asian members' hesitance to commit to liberalization. Asian members have so far implemented trade and investment liberalization unilaterally, and have realized that their recent high growth has been based on their open economic policy and continued efforts to liberalize their trade and investment regimes. This the leaders recognize as indispensable for further growth, which is reflected in their commitment to the Bogor Declaration. This new modality is based on such past experience and calls for the unilateral liberalization in a concerted manner within the Osaka Action Agenda framework. This is a practical way of promoting liberalization without losing momentum for the liberalization process enhanced by the Bogor Declaration. It would take several more years of negotiation if there was an attempt to try to change APEC into a negotiating body like the GATT or WTO.

## Economic and technical co-operation

Part two of the Osaka Action Agenda covers thirteen areas of economic and technical cooperation: human resource development, industrial science and technology, small and medium sized enterprises, economic infrastructure, energy, transportation, telecommunication and information, tourism, trade and investment data, trade promotion, marine resource conservation, fisheries, and agricultural technology (see also Chapter 10). Environmental protection is mentioned in relation to several areas but is not included as an independent area.

Co-operation programmes are needed within APEC to fill gaps in the technology levels, managerial and administrative capabilities, public infrastructure, and so on, resulting from the diversity of its members. Both bilateral and multilateral development co-operation programmes have so far been implemented to resolve these deficiencies and they are likely to continue. However, new elements have evolved in these areas and new types of co-operation programmes are now being developed.

The Osaka Action Agenda sets a new modality for APEC co-operation. It emphasizes the departure from the conventional modality of distinct donor-recipient relationships. Member governments contribute the resources they think they have available on a voluntary basis, such as financial funds, technology, and human skills, and they feel all will gain from this type of co-operation programme. Consistency with market mechanism is emphasized and the participation of the private sector is encouraged. Economic cooperation has so far attracted less attention but it will provide a variety of business opportunities in the region.

It still needs to be explained how co-operation programmes organized along these guidelines are to be implemented. Three hundred and twenty projects have been proposed under the APEC Work Projects initiative. But these remain to be studied and discussed, so have yet to be implemented. Up to 1997 there had been no visible achievements. A Japanese proposal in 1997 for Partners for Progress (PFP) aimed to make a breakthrough in the hesitation associated with the actual implementation of projects, pushing beyond mere studies and seminars. It made a cautious start with technical co-operation in training customs officials, improving administrative capability of intellectual property rights (IPR), standard and conformance (S&C), and competition policy. These are very necessary for successful implementation of liberalization and facilitation and could be easily agreed upon. Japan has also taken the initiative to establish a new task force to tackle such medium- and long-term issues as population, environment, energy and foodstuffs.

### 7.5 Liberalization and facilitation in progress

In 1996, the leaders adopted the Manila Action Plans for APEC in which individual APEC members announced their Individual Action Plans (IAPs) of liberalization and facilitation to be implemented from 1997 onwords (APEC, 1996). The crucial question is, are they big enough to achive their Bogor target?

First, all APEC members submitted their IAPs in spite of prior speculation to the contrary. It is thus something that they will start to implement just two years after the Bogor Declaration.

Second, however, the commitments differ greatly between members, partly because of their nature as unilateral announcements. The chair country provided a common format of IAPs in a matrix of fifteen areas times three different time frames (short, intermediate, and long term). Three types of responces were identified: (1) some members commited concrete measures within a clear time frame; (2) some stated an intent that they would make efforts to achieve the Bogor target; (3) others said they would examine their current

measures for possible amendment. By and large many members committed to (1) for their short-term measures, while (2) and (3) tended to prevail for intermediate and long-term measures. Indeed detailed implementation plans covering 15 to 25 years cannot be expected immediately.

Third, the comparison of the IAPs between countries is never easy because of different levels of current impediaments and different target dates. while they are assessed aginst the common formaty, the positive list of liberlazation announced in IAPs needs to be compared with existing imopediments in order to find out how much still remains to be done (Yamazawa, 1997).

How many impediaments still remian to trade and investment in the Aisa-pacific region? APEC has commissioned studies by the PECC's trade Policy Forum to undertake the first region-wide survey of these impediments (APEC, 1995c). The studies covered tariffs and non-tariff barriers on commodity trade, and regulations on services, trades, foreign investment, and intellectual property rights. The level and structure of impdiaments to reflecting their different resource endowments and stages of industrial development.

However, the studies found that numerous impediaments are common to members throughout the region:

- 1 Tariff levels differ greatly between members. Developed economy members have tariffs of around 5 per cent on many items. Developing economy members have tariffs of 10–20 per cent, while a few have levels of over 30 per cent on many itmes. In both developed and developing members, higher tariffs still remain on textiles,leather goods, and wood products (15–20 per cent in developed economy members and 25–60 per cent in developoing economy members). Figure 7.3 shows the average tariff for all APEC countries in 1993 arranged on an industry by industry basis.
- 2 NTBs (non-tariff barriers) are imposed on agriculture, labour-intensive manufactures, steel and automobiles by many members. Including the NTB elements, tariff equivalent rates (excess of domestic prices over import prices) reach very high levels, especially for agricultural products.
- 3 The services trade accounts for a third of APEC's commodity trade. However, many trade sectors are regulated and some are completely closed. Figure 7.4 shows the frequency of impediments to full liberalization of service sectors amongst the APEC countries. Moreover, among members there is considerable variation in the regulations they still maintain. Developed economy members have a score of 100 in three to seven areas, while developing economy members have the same score in more than half of the areas.



# Figure 7.3 Average tariffs of APEC members (1993)

Source: APEC (1995)

- 4 FDI (foreign direct investment) is still restricted in market access and national treatment and is subject to fiscal incentives (subsidies and tax exemptions) and performance requirements (local content requirements, export requirements, and foreign exchange balancing conditions).
- 5 Most members have made changes in existing domestic legal structures to put in place substantive protection of intellectual property rights. Despite these initiatives there still remains a substantial variance in this protection, and many developing members have some distance to travel to meet their Trade Related Intellectual Property (TRIP) obligations under the WTO agreement.

These remaining impediments, as clarified by the PECC's studies, suggest that we should anticipate strong resistance to liberalization and deregulation from vested interest groups

in those sectors. Generally speaking, two broad types of arguments can be anticipated. One is the argument by developing country members for infant industry protection. They are in the process of 'catching-up industrialization' and may claim infant industry protection for new technology industries and high value-added services (see Chapter 2).

The second argument concerns the protection of such older industries as agriculture and labour-intensive manufactures. At present the developed members are calling for this protection, but it will also start coming from developing members as they reach higher income levels. These industries cause another type of adjustment difficulty. APEC senior officials discussed until the last minutes before the end of the Osaka meeting whether agriculture could be excluded from the liberalization agenda. At those discussions the exclusion was supported by only four members: Japan, South Korea, Chinese Taipei, and China. However, agricultural protection will likely become a concern of the ASEAN countries and Mexico in the near future as their economies develop further and income disparity widens between agriculture and other industries. Agricultural protection can be easily politicized which frequently prevents rational economic solutions. Increased budgetary burdens at home and international commitments to the WTO and APEC for liberalization are likely to be the only means for breaking through these arguments.

Labour-intensive industries such as textiles and footwear share the same difficulty as agriculture, although with less extensive protection and political distortion. Currently only developed countries complain of increasing imports from low-income economies, but the developing members of APEC currently exporting these products also maintain high tariffs on their import of these products. The latter members will soon be importers of these products as their income levels increase, and they can restrict imports from lower-income countries. It is important for all of APEC's members not to exclude these difficult sectors from their liberalization agendas.

The best test of the commitments of APEC to liberalization is to look at what has been proposed for *tariff reduction*. Most members indicated their plans for tariff reduction over the next few years. Some members attach time schedules for reducing these to zero or in terms of sectoral details. Generally speaking, while developed members committed themselves to a tariff reduction of 'UR plus a small alpha' (Uruguay Round plus a small difference) but still short of achieving the Bogor target (a zero tariff by 2010), some developing members committed to a 'UR plus a large alpha' (Uruguay Round plus a larger difference) and sufficient for achieving their Bogor target (a zero tariff by 2020). This can be illustrated in the schematic diagram Figure 7.5. The dotted lines show linear tariff reduction over 15 and 25 years, while the solid lines show 'actual reduction (up to 1995) plus IAP commitment (from 1996 to 2000)'.
### Economic integration in the Asia-Pacific region 171

Computer services					
Tourism					
Construction				7	
Value-added telecommunication Rental services					
Banking					
Distribution					
Insurance					
Other business services					
Professional services					
Environmental services				)	1
Courier services					
Real estate services					7
Auxiliary transport services					
Maritime transport					7
Audio-visual services					7
Recreational services					
Road transport					
Rail transport					
Education					
Air transport					
Pipeline transport					
R&D services					
Health and social services					
Basic telecom unications					
Internal water transport					
Postal services					
		20		60	80 100
	0	20	40 frequency (	60 96)	oo 100

### Figure 7.4 Average impediments to service trade by sectors for APEC (indicators of the absence of commitments as a percentage)

Source: APEC (1995)

sectors





I will now clarify the relationship between Uruguay Round commitments, Bogor targets and APEC/IAP (as well as unilateral reduction and Initial Actions) in both their time scale and tariff levels. This will help assess individual member's IAP commitments within this complicated context of tariff reductions. It also shows the two track approach by APEC.

As regards *non-tariff measures* (NTMs), developed members, plus Chile, Hong Kong, and Singapore claimed that they had no NTMs inconsistent with WTO guidelines. Nevertheless, Canada and the USA stated that they would phase out the multifibre agreement (MFA—a specific international tariff and quota agreement dealing with textile and other fibre trade) quota restrictions by 2004. Japan, South Korea, and Philippines excepted rice from their liberalization package. South Korea committed itself to eliminate most of the current NTMs by 2001. Some developing members did not refer to any specific NTMs but stated that they would study and review their NTMs and gradually reduce the number.

As regards *services*, all members seem to be more cautious here partly because the services trade is more regulated than is the commodity trade by many members, and partly because some services trade is now being negotiated at the WTO. Some members outlined a clear commitment to the liberalization in many services, while many members referred to selected sectors such as telecommunications, transport, tourism, financial services, and business services. However, the submission of IAPs urges APEC members to get more committed to liberalization in this area, which helps the WTO with its liberalization process.

As far as *investment* is concerned, developed members have already achieved liberalization beyond APEC's non-binding investment principles and they are ready to adopt the Multilateral Agreement on Investment established by the Organization for Economic Co-operation and Development (OECD). However, developing members are still cautious about improving their investment regimes. Since FDI has been a prime motivator of the continued growth of the Asia-Pacific economies, the investment liberalization needs to proceed liberalization in other areas. Slow liberalization over 25 years, as suggested by some developing members, may mean a failure on their part to attract sufficient FDI needed for their development programmes.

A further eleven areas are grouped under the heading of *facilitation* measures, which are concerned with 'reducing the cost of doing business' in the Asia-Pacific, an equally important task in the APEC process. These measures are similar to the Single European Market programme. However, while EU members have already implemented over 90 per cent of the harmonization of nearly 300 physical, technical and fiscal measures among themselves, APEC has just started in this direction. Facilitation measures are implemented jointly as Collective Action Plans (CAPs) as well as individually by member countries. However, as yet no agreed check list of progress has been established and a straightforward comparison between different countries' efforts is difficult. Nevertheless the 'nugget' of the APEC liberalization process may be found in these areas. Member governments have to: (1) establish domestic rules or legal frameworks; (2) give sufficient information and make them transparent to both local and foreign firms; and (3) either mutually recognize each other's rules and frameworks or adjust their own to international rules and frameworks. Developed members have already completed phase (1) and are committed to phase (2), and express their intention to proceed to (3). However, many developing members have yet to achieve phase (1) and will take time to proceed to (3). Individual IAPs in these areas reflect this difference in their current preparedness.

Prospects differ between different facilitation measures. Clear progress has been made in establishing common customs procedures, in clarifying standards and conformance criteria, and in securing agreement on the mobility of business persons. These are to be followed by protocols on intellectual property rights, rules of origin, and government purchases. Competition policy and deregulation cannot proceed beyond level (2). The availability of facilitation and technical co-operation measures supporting liberalization measures organized along those lines is certainly an advantage that APEC has over the WTO.

To conclude, APEC's unilateral liberalization process has made a good start. It could induce tariff reduction of 'UR plus' from most members and help to maintain the current momentum for liberalization in NTMs, services, and investment. However, the IAPs of most members have so far only assured progress over the next few years. This needs to be encouraged so that its momentum will be maintained to enable all members to reach their established goals.

APEC has already introduced two mechanisms to encourage this. One is the 'rolling plan formula' and the other is to incorporate the business sectors into the APEC process. The current IAPs are not the final versions but they will continue to be improved every year. Ministers have invited business leaders to jointly monitor and review the progress of APEC liberalization. As the third mechanism we might add 'independent analysis by academia', which will help to publicize APEC progress and attract support from the private sector.

With its basic philosophy of consensus and voluntarism the APEC process cannot be fast track. But both the Bogor Declaration and the Osaka Action Agenda have shown the future direction of changes in Asia-Pacific economic order.

### 7.6 A new model of regional integration

APEC's informal style often leads observers into levelling criticism at it. Some argue that APEC is ineffective because of the diversity of its members. Others argue that East Asia will continue its rapid economic development even without an institution like APEC. Some critics also complain that nothing concrete was decided at either Bogor or Osaka.

All of these criticisms are based on applying an EU—or NAFTA- type model to APEC. A much better model is that of the Open Economic Association (OEA), a much looser regional integration than either the EU or NAFTA. An OEA is:

- open in that its structure and its policies do not lead to discrimination against trade and investment with the rest of the world;
- economic in its primary policy focus;
- a voluntary association in that its members do not cede sovereignty to any supranational regional institution (Yamazawa, 1992 and 1996).

Recognizing the increasingly sophisticated nature of international economic transactions, the scope of an OEA goes well beyond traditional FTAs. The tempo of trade liberalization may be less rapid than in traditional FTAs, but liberalization will be applied both to members and non-members alike on a most favoured nation (MFN) basis. (MFN refers to the principle that whatever favourable conditions are agreed for trade between any two nations should be extended to others on a non-discriminatory and similar basis.) In addition, the gradual liberalization of trade in goods and services will be supplemented by facilitation to dismantle all impediments to international economic transactions as well as to development co-operation. This balanced programme reflects the vast differences in stages of development, current level of impediments to trade and investment, and preparedness for reform among members.

An OEA is a departure from the textbook typology of regional integration, which is derived from the five stages proposed by Bela Balassa (1961). The five stages comprise of a 'free trade area (FTA)', a 'customs union', a 'common market', an 'economic union', and a 'complete economic union', with development taking place in roughly this order. Balassa's typology was developed by 'focusing on the elimination of discriminatory practices within a particular region'. Two problems arise when we apply this concept to the Asia-Pacific. First, the FTA as the first stage tends to be accompanied by discrimination against non members. Second, we need a concept with a lower degree of integration, incorporating non-discrimination against non members and supplementing partial liberalization with facilitation and co-operation programmes. This is closer to the APEC model.

Balassa himself acknowledged the potential for international economic co-operation of various kinds, extending beyond the simple elimination of discriminatory practices. The

EU Common Agricultural Policy (CAP) has been in place since 1967, while co-operation in the steel and energy industries has been a feature of European life since before the founding of the EC itself. The momentum for regional integration calls not only for the abolition of discriminatory practices, but also for a large measure of policy and institutional co-operation and co-ordination. There is a fair degree of success in policy co-ordination and economic co-operation among the members of any regional integration.

The highly inter-dependent, private-sector driven growth of the Asia-Pacific economies has been generated by active trade and investment in spite of the remaining tariffs and NTMs. It is not necessary to further strengthen this market driven integration but sufficient to have a gradual trade liberalization, facilitating trade and investment, and economic co-operation in order to create a steady trade and investment expansion in the region.

With this realistic concept of an OEA to hand, we can address the criticisms of APEC noted above. OEA is feasible within the vast diversity of the Asia-Pacific. OEA is needed in order to continue the current high growth rates of the region. At Bogor, the leaders committed themselves to an Asia-Pacific OEA. The Osaka Agenda made it clear that they aimed at an Asia-Pacific OEA.

This is APEC's vision of 'a community of Asia-Pacific economies'. If countries share an objective of sustaining the regional growth and co-operate with each other, the region will be worthy of the title 'community'. There is an evolving sense of community in the Asia-Pacific region. The feeling of community will enhance the certainty and stability of the region, thereby attracting the private enterprises of members and non members alike.

### 7.7 APEC's consistency with multilateral trading regime

Regional integration has often been criticized in the past as inconsistent with multilateral liberalization. But it has recently become widely agreed that regional integration and multilateral liberalization can be consistent and have actually evolved together in many ways. A recent report by the OECD (1995) surveyed the existing regional integrations such as the EU, EFTA, CUSFTA, and NAFTA and came to the conclusion that some aspects of regional integration, such as preferential tariff reductions and strict rules of origin, can by nature discriminate against non members and indeed be contradictory to multilateral liberalization. However, the regional integration groups have also implemented measures such as harmonization of rules and standards, investment principles, services trade policy, intellectual property rights, and environmental protection and industrial co-operation. These measures do not discriminate against others but serve as a laboratory, a halfway house, in the effort to move from national standards to a multilateral standard.

This new concept of regional integration is consistent with the 'open regionalism' often mentioned in regard to APEC. 'Open regionalism' or 'open regional co-operation' implies the promotion of regional co-operation in a way consistent with multilateral rules. This two pronged thrust of 'open regionalism' has become an important aspect of the APEC process. It conveys a positive image of 'good regional integration' in distinction to

the notion of a 'Fortress Europe' often mentioned in respect to the European Community in the 1980s.

Because of the long-term interdependence of trade and investment links beyond the region, Asia-Pacific economies have shown a great interest in global trade liberalization and have participated actively in the Uruguay Round negotiations. The Osaka Action Agenda confirmed 'consistency with multilateral liberalization' as one of its general principles and suggested accelerated implementation of the liberalization commitments of the Uruguay Round. Like other GATT/WTO commitments, it is likely to be ruled by the most favoured nation's (MFN) treatment of GATT Article 1. Many APEC members will extend their APEC liberalization to members and non members alike on a MFN basis. The implementation of these liberalization packages will be monitored and reviewed jointly for the continued attention of all members, and the mutual encouragement between them.

But APEC members have not yet agreed upon whether to apply the MFN treatment to non-APEC members. A majority of APEC members are supportive of non-discrimination to non-members, while a minority have objections to it on the ground that it will allow the EU unfair benefits, and insists that the APEC liberalization should be applied to the EU only if the EU implements a matching liberalization on the MFN basis. This reciprocal application of the APEC liberalization departs from the OEA model, but can still be referred to as 'open regionalism'. Of course this causes some ambiguity in the term, which invites criticism by non-APEC economists. This difference in philosophy has yet to be resolved.

However, it is unlikely that this difference will lead immediately to the APEC liberalization programmes discriminating against non members. The facilitation part of the programme, as well as the investment liberalization, is unlikely to lead to discrimination against non-members. Indeed there are likely to be some sensitive areas for which unilateral liberalization is difficult to achieve, which, however, will be left to multilateral negotiation at a later stage.

Multilateral liberalization is the best but we cannot be optimistic about its speedy progress. All cautious observers realize that multilateral liberalization will not move forward unless certain key players work together. The APEC group as well as the EU are supposed to be its prime movers. APEC governments have started to talk to the EU to encourage it to join in a similar accelerated implementation of the Uruguay Round outcome (see Chapter 13). At a later stage, APEC could call for a joint initiative for a new WTO round of global liberalization.

APEC and the EU share the same adjustment difficulty in agriculture and textiles and they will need a wider stage for co-ordination and negotiation for these difficult sectors than their own regional groups. If APEC and the EU take a joint initiative in launching the new WTO round of multilateral liberalization, the possibility of a 'free-rider' problem arising (one side taking advantage of the other) will be dissolved, and the momentum for multilateral liberalization will increase significantly.

#### 7.8 APEC and Europe

Let me summarize this analysis of APEC by reference to its relationship to the European Union.

Regional approaches to trade and investment liberalization has been widely adopted in recent years. This represents a 'pragmatic approach by individual nation-states' in response to the 'globalization of business firm activity'. Nation-states responsible for maintaining economic growth and employment have to encourage both domestic and foreign firms to operate actively within their territories. To this end the elimination of the impediments to cross-border transactions and deregulation of various restrictive domestic measures are the ones being resorted to instead of direct tax incentives to firms. The recently inaugurated World Trade Organization (WTO) also aims in this same direction, but takes a longer time to reach agreement because it has 127 members. Many members of the WTO jump ahead of this with a quicker and more manageable solution by combining with their like-minded neighbours.

Both the EU and APEC have adopted pragmatic regional approaches but they represent different models for regional co-operation. The EU is a successful 'role model' of regional integration. For the past 40 years it has achieved liberalization of trade, investment and personnel movement, elimination of a substantial part of physical, technical, and fiscal impediments, and is now attempting a monetary union by 2000. APEC only started in 1989 and has now launched an action programme characterized as the Manila Action Plan. Its facilitation programmes aim at a similar economic outcome as the Single European Market programme, but this is still at its initial stage. APEC will remain similar to a Pacific OECD (the discussion forum of the older advanced countries) just sufficient to promote liberalization, facilitation, and economic and technical co-operation, but will never aim at a political union. APEC is a different 'role model' of regional co-operation. The EU model may be applicable only to Europe, though the APEC model might be applicable to regional groups with diverse members in other parts of the world (Shand and Kalirajan, 1997).

Its open regionalism is a function of its wide range of trading partners and its diverse membership, but it is consistent with the WTO regime. The APEC liberalization will be made more effective through linking it with the WTO liberalization. APEC has good reasons for co-operating closely with the EU, another major actor in the WTO regime in promoting global liberalization.

The Asia-Europe Summit Meeting (AESM) becomes important in this context. The AESM began in Bangkok in March 1996 in order to strengthen the weakest link in the triad of Asia, America and Europe. It represents another attempt of inter-regional cooperation (see Chapter 13). Unlike APEC, AESM was started with a meeting of 26 leaders, ten Asian (seven ASEAN plus Japan, China and South Korea) and 16 EU leaders (including the Chairman of the European Commission) but it has now extended its activity so as to promote the inter-regional business at ministerial meetings, and collaborative business fora (see also **Smith, 1998**).

The AESM is another APEC type model for inter-regional co-operation. It contains diverse members, Asians and Europeans. Concrete programmes for liberalization,

facilitation, and economic and technical co-operation could be modelled after those in APEC, and voluntarism and flexibility need to be observed in implementing these programmes.

#### References

- APEC/EPG (1994) Achieving the APEC Vision: Free and Open Trade in the Asia Pacific, Singapore, APEC Secretariat.
- APEC (1995a) Selected APEC Documents: 1989–1994, Singapore, APEC Secretariat.
- APEC (1995b) *The Osaka Action Agenda: Implementation of the Bogor Declaration*, Osaka, APEC Secretariat.
- APEC (1995c) Survey of Impediments to Trade and Investment in the APEC Region, Singapore, APEC Secretariat.
- APEC (1996) Manila Action Plans for APEC, Manila, APEC Secretariat.
- Balassa, B. (1961) The Theory of Economic Integration, Homewood, Richard D.Irwin.
- Brook, C. (1998) 'Regionalism and globalism' in McGrew, A. and Brook, C. (eds).
- Huntington, S.P. (1993) 'The clash of civilizations?', Foreign Affairs, vol. 72, no. 3.
- McGrew, A. and Brook, C. (eds) (1998) *Asia-Pacific in the New World Order*, London, Routledge in association with The Open University.
- Organization for Economic Co-operation and Development (1995) *Regional Integration Agreements and the Multilateral Trading System: Are They Compatible*? TD/ TC (93) 15, Paris, OECD.
- Shand, R. and Kalirajan, K.P. (1997) 'Yamazawa's open economic association: an Indian Ocean grouping for economic co-operation', *The Developing Economies*, vol. XXXVI, no. 1, pp. 3–27.
- Smith, M. (1998) 'The European Union and the Asia-Pacific' in McGrew, A. and Brook, C. (eds).
- Yamazawa, I. (1992) 'On Pacific economic integration', *The Economic Journal*, vol. 102, no. 415, pp. 1519–29.
- Yamazawa, I. (1996) 'APEC's new development and its implications for non member developing countries', *The Developing Economies*, vol. XXXIV, no. 2, pp. 113–37.
- Yamazawa, I. (1997) 'APEC's liberalization and WTO', *The Australian Economic Review*, vol. 30, no. 1.

### CHAPTER 8 Measuring the size of foreign multinationals in the Asia-Pacific

Eric Ramstetter

#### 8.1 The problem

The primary focus of this chapter is a very simple question: have foreign multinational corporations (MNCs) become more important in host economies in the Asia-Pacific region in recent years? This issue, though important, is somewhat narrower in scope than those taken up by other chapters in this book. A key question to ask of the Asia-Pacific region is whether MNCs have become increasingly important agents of regional integration. However, this chapter argues that because of the lack of data it is impossible to provide an accurate answer to that question. For example, there is no information from home country sources on MNCs from Hong Kong, only limited information on MNCs from Korea, Singapore and Taiwan, and while there are a lot of data on Japanese MNCs, much of the data are highly unreliable in some respect or another (Ramstetter, 1996a). It is also impossible to assemble consistent and comprehensive information on foreign MNCs classified by country of origin from host economy sources. Thus, there are literally no comprehensive and reliable data on the economic activity of MNCs (for example, foreign direct investment flows or stocks, employment, production, trade) classified by source economy and recipient economy in the Asia-Pacific region (and most other regions). This makes it difficult to show how MNCs have contributed to regional integration.

Of course there are empirical studies that attempt to analyse the contribution of MNCs to regional integration. But these studies are highly speculative, due in large part to the lack of meaningful data on which to base their analyses (see, for example, Bora, 1996). Other studies have a sounder analytical basis, but typically focus heavily on Japanese and/or US MNCs because the data for these MNCs are relatively abundant (Kreinin *et al.*, 1997; Petri and Plummer, 1995). This chapter attempts to widen the focus somewhat to the activities of foreign MNCs in eleven host economies in the Asia-Pacific. Although only indirectly related to issues of economic integration, this focus has the advantage of facilitating a simple, yet reasonably sound analysis of trends in MNCs in a large number of the major host economies in the region.

Economists often face measurement problems, but these problems can be particularly severe when it comes to analysing MNCs. For example, it is much easier to analyse aspects of economic integration related to international trade because there are data on trade flows classified by source economy and recipient economy (as well as by industry). As indicated above, corresponding data do not exist for the economic activities of MNCs. Indeed, only one economic activity of MNCs is measured in a manner that is even vaguely standardized internationally: foreign direct investment (FDI) flows as reported in balance of payments data. However, even here there are important holes in the data as Hong Kong, a major investing and host economy, does not publish data on outward FDI, and no consistent breakdowns by host/recipient country and/ or industry are available.

As will be detailed below, in the aggregate, both inward FDI flows and stocks (that is, cumulative flows) have increased very rapidly in the Asia-Pacific region since the mid 1980s. Correspondingly, it is commonly believed that MNCs have become more important. However, there are two important reasons why even this seemingly straightforward conclusion may not be correct.

First, FDI is a financial flow, the stock of FDI being one item, and often a relatively small item, on the liability side of a corporate balance sheet. More precisely, FDI refers to international investment from one economy to another, where the foreign ownership share exceeds a certain threshold. These thresholds differ across reporting economies, the most common threshold in recent years being ten per cent. FDI consists of equity, loans from related companies, and reinvested earnings, though a number of economies do not report data on reinvested earnings. In addition, official publications from some economies (for example, Japan and Taiwan) often use data on reported or approved FDI, instead of data on FDI actually remitted and reported in the balance of payments. However, even when measured correctly, changes in FDI may simply reflect a restructuring of a firm's liabilities and have little or no relationship to changes in production or other related activities in MNCs, such as employment and trade. Moreover, except in cases where issues related to the balance of payments or corporate finance are the primary concern, production and related variables are usually thought to provide more meaningful measures of MNC activities.

Second, there is no counterpart to FDI for non-MNCs and it is thus very difficult to measure the size of MNCs relative to all firms in an economy in terms of FDI. Two kinds of measures, ratios of FDI to gross domestic product (GDP) or ratios of FDI to total investment or fixed investment are often used as proxies in this regard. However, they both have distinct drawbacks. On the other hand, production (for example, GDP) and related indicators (for example, employment, trade) can be measured in the same way for both MNCs and non-MNCs. Thus it is a rather straightforward exercise to measure the shares of MNCs in terms of production or related indicators.

This chapter examines changes in the level of foreign MNC presence in Asia-Pacific economies since 1970 by comparing trends in ratios of inward FDI *flows* to GDP, and inward FDI *stocks* to GDP, and with foreign MNC *shares* of production (=GDP). Due to the lack of production data for multinational parents in key investing economies (Hong Kong, Korea, Singapore and Taiwan), this comparison is limited to host economies, that is to the comparison of measures based on inward FDI and measures based on the production of foreign MNCs. In order to put this comparison in context, the chapter first looks at why economists are interested in measuring the activities of MNCs and the various uses of such measures (Section 8.2). Section 8.3 describes indicators used to measure foreign MNC presence and the statistical tools used to examine foreign MNC presence in eleven Asia-Pacific economies, and some of the major factors thought to underlie the trends observed are indicated. Finally, in Section 8.5, the major conclusions

of the analysis are summarized. The chapter is based heavily on Lipsey *et al.* (1995) and Ramstetter (1996b).

#### 8.2 Why and how to measure the activities of MNCs

The first step here is to clearly define what is meant by a multinational corporation. For the purposes of this chapter an MNC is defined as a firm with operations in two or more countries. A foreign MNC is then a firm that has a foreign ownership share above a certain cut-off. This definition can be ambiguous in important respects. For example, as with FDI flows, ownership cut-offs differ among reporting economies and even among different data sources for one reporting economy. More importantly, when joint ventures (firms with more than one owner) are involved, especially when no one owner has majority control, any ownership criterion can become ambiguous, especially when trying to distinguish among groups of foreign owners. In addition, some firms classified as foreign MNCs in this chapter may actually be better classified as local firms, as their only operations may be in the local economy (for example, a Taiwanese factory transplanted to Thailand where the owner remains in Taiwan).

Moreover, this definition of an MNC does not distinguish between types of MNCs, nor does it encompass the vast array of non-ownership based international relationships (for example, international subcontracting of various types) that exist among firms (see Chapters 10 and 14). There may also be important distinctions between affiliates of MNCs established through mergers and acquisitions and affiliates arising from the formation of new affiliates, between investments in expanding the operation of previously existing affiliates and investments in the establish-ment of new affiliates, or between wholly foreign owned affiliates and joint ventures. It is important to note that the analysis here says nothing about these differences. In short, what is of most interest here, and what is of most fundamental importance to the economic analysis of MNCs, is the distinction between whether a firm has operations in one economy or in two or more economies.

Correspondingly, the economic theory of the multinational corporation focuses first and foremost on the question of why a firm chooses to become a multinational and incur costs of cross-border operations not incurred by non-MNCs. Very simply put, this question is commonly answered by identifying the advantages that MNCs have that allow them to overcome the disadvantages of the additional costs of operating across international borders. The interested reader is encouraged to see more comprehensive surveys of this literature (Caves, 1996; Dunning, 1993), but here I summarize the three sets of advantages that are often hypothesized in this regard.

The first set consists of advantages accruing from exploitation of assets that belong to a given firm. These assets are often called *firm-specific assets* and advantages accruing from the possession of such assets are called ownership advantages. Important examples of such firm-specific assets are patents, in-house research capability, and exclusive marketing networks. The intangible nature of such assets is often emphasized in comparison to the tangible nature of fixed assets (for example, buildings, machinery).

The second set of advantages are those accruing from the internalization of economic transactions within a single firm unit. These advantages are called *internalization* 

advantages. For example, when uncertainty makes inter-firm transactions risky and thus costly, a firm can often reduce costs by arranging to carry out that transaction or an equivalent activity within the firm. Suppose, for example, a firm develops a very advanced semiconductor that can greatly improve the performance of a personal computer. In such a case there is likely to be a problem of asymmetric information—namely, that the firm developing the semiconductor will know far more about its capabilities than any prospective buyer. This will lead to a tendency for the buyer to undervalue the semiconductor (from the perspective of the developer) and will thus create a motive for the developer to produce its own personal computers, using the semiconductor, in an attempt to extract (what the developer perceives as) the full value from its development efforts.

Another example is when quality is very important to a final goods producer (for example, a personal computer company). Such a producer will be very reluctant to purchase from an intermediate goods supplier (for example, a subcontractor), unless they are assured that concerns about quality control will be addressed by the supplier. Moreover, the firm producing the intermediate goods may have a smaller incentive to emphasize quality than the final goods producer. If this is the case, the final goods producer may have to produce the intermediate goods itself rather than buying it from a supplier. Both of these are examples where transactions that would take place in arm's length markets may end up internalized within a single firm. The existence of such internalization advantages is often thought to be a major reason for the existence of MNCs.

The third set consists of advantages accrued from operating in a specific location or *locational advantages*. Traditional examples of locational advantages are reductions in the costs of serving markets when firms are faced with high levels of trade protection in the target market, and reductions in production costs afforded by increased access to lower cost factors of production (for example, labour and natural resources).

Together, these three elements comprise Cunning's OLI (ownershiplocationinternalization) paradigm (see Dunning, 1993). There is an extensive theoretical debate over whether all of these advantages are a necessary condition for a firm to become a multinational (and thus for FDI to occur), with some arguing that internalization advantages alone are sufficient to explain the existence of the multinational firm (see Buckley and Casson, 1991; Rugman, 1985). However, from an empirical point of view the general agreement that MNCs tend to possess a distinctive set of firm-specific, intangible assets is important, whether or not such assets are necessary for a firm to become a multinational. The possession of distinctive firm-specific, intangible assets is important because it implies that the behaviour of MNCs differs systematically from the behaviour of non-MNCs. More specifically, there are at least three interrelated sets of firm-specific, intangible assets that MNCs are thought to possess in relatively large amounts: production technology, marketing networks, and management know-how.

The possession of superior production technology (and superior management knowhow) implies that MNCs tend to be more efficient than non-MNCs. One simple example lending support to this proposition is the casual observation that foreign MNC shares of host country production tend to be larger than corresponding shares of employment. In other words, the average product of labour (=production per worker) often tends to be relatively high in foreign MNCs. However, it is also important to note that there is often very large *variation* in productivity differentials between MNCs and non-MNCs (see Ramstetter, 1994, 1995a, 1996c). As a result, there is a relatively high statistical probability (greater than five per cent) that one would be wrong if one concluded that there are productivity differentials between MNCs and non-MNCs in many cases.

In addition, the relatively sophisticated marketing networks of MNCs, particularly those related to international trade, also lead to the expectation that MNCs will be more dependent on trade than non-MNCs. This proposition is again supported by the casual observation that foreign MNC shares of host country exports tend to be much larger than shares of production and employment, or, in other words, that trade propensities (for example, export-sales ratios or ratios of imports to total inputs) are higher in MNCs than non-MNCs. It is also important that the variation of differentials between MNCs and non-MNCs tends to be much smaller in terms of trade propensities than in terms of average labour productivity (Ramstetter, 1994, 1995a, 1996c). Thus, there is a relatively low statistical probability (less than five per cent) of being wrong if one concludes that differentials between MNCs and non-MNCs exist in terms of trade propensities.

In some contrast to the focus on differences between MNCs and non-MNCs discussed above, this chapter is concerned with the more basic questions of:

1 How important are foreign MNCs in Asia-Pacific production?

2 How has the importance of foreign MNCs changed over time in the Asia-Pacific?

The answers to these questions are, however, closely related to the discussion above, because the relative size of MNCs differs dramatically depending on the activity measured. For example, if one were to focus on the relatively low shares of foreign MNCs in employment, one would likely conclude that foreign MNCs are generally unimportant in the Asia-Pacific. At the other extreme, if one were to focus on the large foreign MNC shares of exports, one would likely conclude that MNCs are very important. Furthermore, a focus on production would lead one to the conclusion that MNCs are of intermediate importance.

As explained in the introduction, FDI is the only indicator of economic activity in MNCs that is even vaguely standardized internationally—and measures of this have been used extensively in other chapters. It should also be emphasized that data on FDI are available far more often and far more quickly than data on production and related variables. As a result, FDI flows or stocks are often used as a general proxy for the level of economic activities in MNCs. For example, to quote from the United Nations:

World-wide foreign-direct-investment (FDI) flows quadrupled between 1980 and 1990... This rapid growth is an indication of the growing importance of FDI as an instrument of international economic integration. (United Nations Transnational Corporations and Management Division, 1993, p. 1)

The principal measure of annual changes in the cross-border investment activities of TNCs [transnational corporations] is FDI flows. As long as such flows remain positive, even if they decline from year to year, they mark the expansion of TNC activities.

(United Nations Conference on Trade and Development, 1994, p. 9)

These statements are somewhat vague on a very important point, that is whether 'the growing importance of FDI' and 'the expansion of TNC activities' refer to financial activities of MNCs only, or whether they also refer to production and related activities in these firms. Nonetheless, these works, and dozens more like them, imply that FDI is a proxy for the activities of MNCs in general, not just for activities related to international finance.

### 8.3 A simple methodology for evaluating foreign MNC presence

As indicated above, increases in FDI are often cited as evidence that MNCs have been much more important in the Asia-Pacific region since the 1970s. In contrast, I believe that this assertion is so often incorrect that it constitutes a myth. However, it is important to recognize that those who make this claim are wrong not because they lack empirical evidence to support the assertion. Rather, my argument is that they err in relying on FDI-based indicators that are not robust for evaluating the relative size of foreign MNCs or foreign MNC presence. One may think that this is just nit-picking because FDI flows and FDI stocks, though not definitionally correlated with the production of MNCs and related indicators, are in fact correlated with them empirically. To some extent this viewpoint is probably valid. On the other hand, the evidence below suggests that trends in FDI flows and FDI stocks, even when averaged over relatively long periods of time, are, in a number of cases, poor proxies for trends in production.

More specifically, FDI flows and stocks often display trends that differ markedly from trends in MNCs, and trends in FDI flows and FDI stocks are often subject to far more variation than trends in MNC production. These points are examined in some detail below. To the extent that data are available, trends in the following four indicators are compared for the 1970–95 period:

- 1 ratios of inward FDI flows to gross domestic product (GDP);
- 2 ratios of inward FDI stocks (defined as cumulative FDI flows from the first year for which data are available, usually 1970) to GDP;
- 3 shares of foreign MNCs in production (usually measured as GDP=value added) in all industries;
- 4 shares of foreign MNCs in manufacturing production (again, usually measured as GDP=value added).

These indicators are examined in two steps. First, the trends in each indicator are plotted against time so that trends can be examined visually. Second, two basic statistical properties of each measure, mean ratios and a measure of the variation in mean ratios, the coefficients of variation, are calculated for each indicator for three sub-periods, 1970–78, 1979–86 and 1987–95. The mean ratio is simply the average of the ratio in question for the period in question. The coefficient of variation is the ratio of the standard deviation of the mean ratio itself for the period in question, where the standard deviation is a measure of the average difference between annual ratios and the mean ratio for the period in question. For example, in a sample of nine observations (the largest sample used here) the mean ratio can be said to display a very large degree of variation if the associated coefficient of variation is greater than 0.54, because if one concludes that

the mean ratio is greater than zero in such a case, there is a relatively large statistical probability (greater than five per cent) that even this very weak conclusion would be incorrect.

A large number of statistical difficulties and caveats arise in any comparison of economic activity across countries. As already mentioned, definitions differ, coverage is often inconsistent and patchy, and methodologies for data collection and presentation are far from uniform. All these remain very important in the case of the FDI and production indicators which are being dealt with here. However, what follows is the most consistent set of comparative statistics I could construct given the limitations of the data. (The sources of these data are given in the appendix to this chapter.)

### 8.4 The relative size of multinationals in selected Asia-Pacific economies

This section will attempt to answer two questions. Do FDI-based measures and MNC shares of production:

1 display similar trends over time?

2 display similar levels of variation or volatility over time?

For simplicity, the data will be presented country by country. In addition to examining the patterns observed from the data, an attempt will be made to describe some of the more important factors underlying the patterns observed. It should be noted, however, that discussions of underlying factors are not comprehensive, but only suggestive in nature.

#### Canada

In Canada, ratios of inward FDI flows or stocks to GDP were rather low in the 1970s, decreased markedly in 1980 to 1982, and then rose to a rather higher level thereafter (see Figure 8.1). Correspondingly, mean ratios of FDI flows to GDP decreased between 1970–78 and 1979–86, but then increased rather dramatically in 1987–95 (see Table 8.1).

In marked contrast, foreign MNC shares of production (that is, ratios of production by foreign MNCs to GDP), both in all industries and in manufacturing, rose between 1970 and 1974 but tended to decline thereafter. Moreover, foreign MNC shares moved in a much narrower range than did FDI-based measures, as indicated by relatively low coefficients of variation. Here, it is important to note that the low frequency of data on foreign MNC shares of production means that comparisons of variation are not as meaningful statistically as corresponding comparisons for countries with more observations. Nonetheless, the contrast between the two types of measures is stark, with FDI-based measures indicating that MNCs have become much more important in Canada since the mid 1980s, while foreign MNC shares of production tended to decline and moved in a much narrower range.



### Figure 8.1 Foreign MNC presence in Canada

Table 8.1 Foreign MNC presence in Canada,mean and variation

	Mean	ratios (per	r cent)	Coefficients of variation			
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95	
FDI flows/GDP, all	0.44	0.22	1.21	0.91	3.69	0.41	
FDI stocks/GDP, all	2.06	1.66	7.67	0.24	0.65	0.33	
MNC production/GDP, all	16.97	16.20	14.95	0.05	NA	0.04	
MNC production/GDP, mfg	13.77	NA	11.10	0.05	NA	NA	

Note: all calculations exclude years for which data are not available; NA=no data are available for the period

These trends are probably closely related to the fact that the United States is by far the largest investor in Canada, and that US MNCs have had a large presence in Canada for the entire period under study. In other words, assuming that trends in US MNCs dominate trends in foreign MNCs in Canada, these data indicate that US investors have greatly increased the foreign liabilities of their Canadian affiliates, but that increases in production by US affiliates in Canada have been much more moderate.

### Japan

Japan contrasts with other developed economies (for example, Canada and the United States) in that foreign MNCs are apparently relatively unimportant by any measure (see Figure 8.2 and Table 8.2). Moreover, the importance of foreign MNCs has apparently declined over time, with ratios of FDI flows to GDP and foreign MNC shares of production all showing downward trends. However, although ratios of FDI flows to GDP

and foreign MNC shares of production show similar downward trends, foreign MNC shares declined much more than FDI-based measures.



### Figure 8.2 Foreign MNC presence in Japan

unu v						
Measure, industry	Mean	ratios (per	cent)	Coefficients of variation		
	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.03	0.02	0.02	0.97	0.64	1.75
FDI stocks/GDP, all	0.13	0.18	0.18	0.32	0.22	0.23
MNC production/GDP, all	1.76	1.29	1.15	0.07	0.24	0.09
MNC production/GDP, mfg	5.14	3.64	3.01	0.00	0.24	0.08

Table 8.2 Foreign MNC presence in Japan, meanand variation

Note: all calculations exclude years for which data are not available Additional notes and sources: see Appendix A

Nonetheless, as in the Canadian case, the FDI-based measures generally fluctuate in a much wider range, as indicated by relatively large coefficients of variation shown in Table 8.2.

The low level of foreign MNC activity in Japan is often cited as evidence of the closed nature of the Japanese market (see Encarnation, 1992). The downward trend in foreign MNC presence in Japan is significant because there is general agreement that Japan's policies toward foreign MNCs were liberalized greatly in the mid 1970s and then almost completely in 1980. The continued downward trend after this liberalization is an important piece of evidence suggesting that Japanese policies toward foreign MNCs are not the cause of the low and declining level of foreign MNC activity in Japan, at least since 1980 (see Ramstetter and James, 1993).

### United States

The United States is a very important example of a country where all indicators reveal relatively large increases in foreign MNC presence during the period studied (see Figure 8.3 and Table 8.3). However, here again there



## Figure 8.3 Foreign MNC presence in the United States

Table 8.3 Foreign MNC presence in the U	nited
States, mean and variation	

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.16	0.56	0.83	0.35	0.37	0.44
FDI stocks/GDP, all	0.57	2.76	7.08	0.58	0.33	0.13
MNC production/GDP, all	1.78	3.09	4.20	0.08	0.14	0.08
MNC production/GDP, mfg	3.85	7.21	11.55	0.06	0.18	0.13

Note: all calculations exclude years for which data are not available Additional notes and sources: see Appendix A

are important differences in the trends of FDI-based measures and of foreign MNC shares. First, comparing 1987–95 with previous periods, FDI-based measures indicate a much larger increase in foreign presence (more that fivefold) than do foreign MNC shares (two to threefold). Second, the fluctuations in FDI-based measures tend to be much larger than fluctuations in foreign MNC shares, as indicated by larger coefficients of variation.

Trends in the United States have very important global implications because the United States is the largest single economy and the largest recipient of inward FDI in the world. Indeed, the increases in foreign MNC presence in the United States are so large in

absolute terms that they were a major element in the post-1980s increase of MNC shares of production worldwide (Lipsey *et al.*, 1995). The large increase in foreign MNC presence in the United States has many causes, all of which cannot be detailed here. However, among the potentially important causes are: the liberalization of financial markets in developed economies in the late 1970s and early 1980s; the relatively rapid growth of the US economy with relatively low inflation in the 1980s and 1990s (compared to the 1970s); the increased competitiveness of European and Japanese investors; and the large decline in the value of the US dollar in 1978 and again in 1985 (which in turn led to large decreases in the prices of US assets and production costs when measured in foreign currency).

### Hong Kong

The data on Hong Kong refer only to FDI from Organization for Economic Co-operation and Development (OECD) economies as this is all that could be reasonably estimated. Substantial FDI from China and other non-OECD economies in Asia (for example, Korea, Taiwan, Singapore) is excluded. Moreover, foreign MNC shares of production can be calculated for manufacturing only and for the post 1983 period only. Since the vast majority of inward FDI in Hong Kong is in non-manufacturing, this means that there are large definitional differences between FDI-based measures and foreign MNC shares of production. Nonetheless, both the FDI-based measures and foreign MNC shares indicate a substantial increase in foreign presence between 1979–86 and 1987–95 (see Figure 8.4 and Table 8.4). However, here again, the FDI-based measures tend to fluctuate in a much wider range as indicated by relatively large coefficients of variation for these measures.



Figure 8.4 Foreign MNC presence in Hong Kong

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	1.17	1.83	2.70	0.20	0.62	0.72
FDI stocks/GDP, all	2.05	9.47	17.24	0.45	0.39	0.07
MNC production/GDP, mfg	NA	12.30	16.64	NA	0.09	0.16

### Table 8.4 Foreign MNC presence in Hong Kong,mean and variation

Note: all calculations exclude years for which data are not available; NA=no data are available for the period

Additional notes and sources: see Appendix A

With regard to investment in Hong Kong, two factors are particularly important. First, Hong Kong has always been very open to foreign trade and FDI, and has maintained very low corporate tax rates. It has always provided an attractive policy environment for foreign MNCs seeking to invest in East Asia, and it is very unlikely that the recent increase in foreign presence in Hong Kong is due to changes in Hong Kong's economic policies. Second, Hong Kong is an entrepôt and a large portion of foreign MNC presence there is focused on providing goods and services for third markets—for example, China, Japan, the United States, and South-East Asia. Moreover, Hong Kong is a key financial and co-ordinating centre for MNCs in the region, and a very large portion (about half or more) of its large outward flows of FDI originate from affiliates of foreign MNCs operating there (Low *et al.*, 1996). Assuming that China itself does not experience substantial economic or political difficulties, these characteristics are also powerful reasons to expect that Hong Kong's reversion to Chinese sovereignty may have little effect on foreign MNC presence.

#### Korea

Korea, like Japan, has very low ratios of FDI flows and stocks to GDP (see Figure 8.5 and Table 8.5). These ratios declined between 1970–78 and 1979–86 and then increased again in 1987–95. Average foreign MNC shares of production also declined between 1974–78 and 1984–86, the only years for which such ratios can be calculated. As in the Canadian case, it is not very meaningful to compare variation in FDI-based measures of foreign MNC shares of production due to the lack of data. However, the comparisons that can be made reveal that variation of foreign MNC shares is much larger relative to the variation in FDI-based measures than in most of the other economies studied in this chapter. In other words, the limited data that exists in the Korean case differs from most of the other cases studied here in revealing relatively little difference in the variation of trends of FDI-based measures and the variation of trends in foreign shares of production.

As in Japan, foreign MNC presence is very limited in Korea. Moreover, like Japan in the 1960s, the low level of foreign MNC presence can be attributed to restrictions on foreign MNCs, several of which remain today. In Korea, restrictions on Japanese MNCs are particularly strict. For example, sales of Japanese cars under Japanese labels are forbidden in Korea, though the Japanese do manufacture and sell a large number of cars in joint ventures with Korean firms. Moreover, despite these restrictions, Japanese MNCs are among the leading investors in Korea and are likely to continue to be important in this respect in the future. It is also of some interest that, despite the fact that Korean restrictions on foreign MNCs are probably far stricter than corresponding Japanese restrictions have ever been, foreign MNC shares of production are apparently substantially larger in Korea than in Japan.



# Figure 8.5 Foreign MNC presence in Korea

# Table 8.5 Foreign MNC presence in Korea, meanand variation

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.46	0.15	0.36	0.48	0.90	0.37
FDI stocks/GDP, all	1.61	1.28	2.14	0.28	0.13	0.10
MNC production/GDP, all	4.42	NA	NA	0.26	NA	NA
MNC production/GDP, mfg	14.92	11.41	NA	0.23	0.06	NA

Note: all calculations exclude years for which data are not available; NA=no data are available for the period

Additional notes and sources: see Appendix A

### Singapore

Relative to the size of the host economy, foreign MNCs are far larger in Singapore than in any other economy in East Asia, and perhaps in the world. An indication of this is very high ratios of FDI stocks to GDP in recent years (70 per cent in 1987–95, see Figure 8.6 and Table 8.6). Moreover, there have been very large increases in FDI flows and stocks in recent years and commensurate increases in ratios of these variables to GDP. As in Hong Kong, foreign MNC shares of production are only available for manufacturing, and because foreign FDI non-manufacturing is extremely important, comparisons of FDI-based measures and foreign MNC shares are imprecise in this case as well. However, it is also important that, as in the United States, the increase in foreign MNC shares of manufacturing production is much smaller than increases in FDI-based measures. Part of the reason for this is the extremely high level—over 60 per cent—of foreign MNC shares in manufacturing. Here again, fluctuations in foreign MNC shares are much smaller than corresponding fluctuations in FDI-based measures.



### Figure 8.6 Foreign MNC presence in Singapore

Table 8.6 Foreign MNC presence in Singapore,mean and variation

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	5.34	8.84	10.21	0.27	0.25	0.35
FDI stocks/GDP, all	19.13	49.40	70.30	0.45	0.26	0.08
MNC production/GDP, mfg	63.86	65.75	71.73	0.02	0.03	0.02

Note: all calculations exclude years for which data are not available Additional notes and sources: see Appendix A

The contrast between the very large foreign presence of MNCs in Singapore and the much more moderate presence in Hong Kong is of some interest, given that these economies share several important characteristics. Both economies are very open to foreign MNCs and international trade, are entrepôts for their respective regions, and are important outward investors with large portions of that outward investment coming from foreign MNCs in each respective economy (Low *et al.*, 1996). Given these important similarities, it is somewhat puzzling that foreign MNC shares of manufacturing

production are four to five times larger in Singapore than in Hong Kong. Three possible explanations for this are that:

- 1 Hong Kong is a much larger economy (about twofold);
- 2 high wages in the public sector in Singapore have tended to encourage the best workers to seek employment in the public sector;
- 3 policies in Singapore have tended to favour foreign MNCs and public corporations compared to local entrepreneurs.

However, there are also important reasons to expect that foreign MNC presence should actually be larger in Hong Kong than in Singapore. Namely, the policy environment is probably less friendly to foreign MNCs in Singapore because:

- 1 the government tends to intervene more in the economy, especially in labour and capital markets, than in Hong Kong;
- 2 corporate tax rates are higher in Singapore;
- 3 restrictions related to social policy tend to be stricter in Singapore.

In short, the much larger foreign MNC presence in Singapore seems paradoxical in many respects. Furthermore, the low foreign MNC shares of manufacturing production in Hong Kong also seems paradoxical when it is realized that corresponding shares tended to be somewhat higher in Taiwan and Indonesia (see below), two economies which have restricted the access of foreign MNCs to a much greater extent than Hong Kong or Singapore.

### Taiwan

Taiwan is another economy where ratios of FDI flows and stocks to GDP have generally been very low (see Figure 8.7 and Table 8.7). As in Korea, ratios of FDI flows to GDP fell between 1970-78 and 1979-86 before rising in 1987-95. However, in contrast to Korea, ratios of FDI stocks to GDP rose more continuously over these three periods, with the largest increase between the second and third periods. In all industries, foreign MNC shares of production changed very little between the first two periods but then rose substantially in the later one. In manufacturing, foreign MNC shares fell slightly between the first and second periods, rising somewhat in the later period. However, mean ratios were relatively constant in the 17 to 20 per cent range. In the first two periods, fluctuations in ratios of FDI flows to GDP were much larger than fluctuations in foreign MNC shares of production, but of similar magnitude in the third period, and fluctuations in ratios of FDI stocks to GDP were the smallest of any indicator in the later two periods. Thus, Taiwan is an example where FDI-related indicators and foreign MNC shares display similar trends and, in the later period, similarly high levels of variation. However, it should be noted that the surveys from which MNC shares are taken are not mandatory and are subject to wide variation in sample coverage. In short, variations in survey coverage may be one reason why relatively large variation is observed for foreign MNC shares in Taiwan.

Data for Taiwan are suggestive in at least two other important respects. First, as in Korea, foreign MNC shares of production are larger than might be expected given the low level of FDI in the economy relative to GDP. Indeed, the fact that foreign MNCs

have accounted for an average of about one fifth of manufacturing production in Taiwan would seem to suggest that low levels of FDI may not necessarily imply low levels of MNC presence. Second, on the other hand, it also seems clear that recent policy reforms, notably the liberalization of capital markets and restrictions on foreign MNCs in service industries in the late 1980s, have resulted in increases in foreign MNC shares of production, as well as increases in FDI-related indicators. This fact suggests that a similar process may have occurred in Korea, where policy restrictions were also relaxed in the 1980s and FDI-GDP ratios have been rising thereafter.



# Figure 8.7 Foreign MNC presence in Taiwan

Table 8.7 Foreign MNC presence in Taiwan,mean and variation

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.52	0.36	0.68	0.55	0.51	0.18
FDI stocks/GDP, all	1.86	2.37	4.03	0.17	0.05	0.02
MNC production/GDP, all	6.84	6.91	12.34	0.11	0.13	0.07
MNC production/GDP, mfg	19.10	17.24	20.05	0.07	0.10	0.10

Note: all calculations exclude years for which data are not available

Additional notes and sources: see Appendix A

### Indonesia

The data for Indonesia display patterns that are similar to those observed in the Taiwanese data in two important respects (see Figure 8.8 and Table 8.8). First, FDI flow-GDP ratios fell between 1970–78 and 1979–86 but then rose again in 1987–95. Second, ratios of FDI stocks to GDP rose continuously over these three periods, with by far the

largest increase between the second and third periods. However, the data on foreign MNC shares of manufacturing production contrast with Taiwanese data and FDI-related indicators for Indonesia in suggesting that foreign MNC presence tended to fall over the three periods. Moreover, the Indonesian data follow a general pattern observed in this sample of countries, with foreign MNC shares of production fluctuating less that FDI-related indicators.

Indonesia has always been rather receptive to FDI by foreign MNCs, but it also tended to severely restrict the activities of those foreign MNCs, as well as local private firms, until the late 1980s. In this respect, the fact that Indonesia had among the highest levels of import protection in the region prior to 1986 is particularly important, and the liberalization of trade begun



## Figure 8.8 Foreign MNC presence in Indonesia

	Mean	ratios (per	cent)	Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.86	0.29	1.14	0.81	0.31	0.43
FDI stocks/GDP, all	2.68	3.35	6.69	0.35	0.20	0.19
MNC production/GDP, mfg	26.00	23.25	20.07	0.14	0.18	0.13

Table 8.8 Foreign MNC presence in Indonesia,mean and variation

Note: all calculations exclude years for which data are not available Additional notes and sources: see Appendix A

in 1986 has been the major cause of the extremely large changes in the Indonesian economy in subsequent years. Increases in inward FDI are often pointed to as one of the favourable effects resulting from the liberalization of trade and the relaxation of other restrictions on private business. However, as in Canada, it is interesting that the large increase of FDI-related measures were not accompanied by increases in foreign MNC

shares, despite the fact that manufacturing apparently accounted for a large portion of the increase in FDI after 1992. (These data are from unpublished mimeos of the Investment Co-ordinating Board and refer to realized FDI stocks outside of oil and banking.)

#### Malaysia

Like Singapore, Malaysia is distinguished by an extremely large presence of foreign MNCs compared to other countries in this sample, with the ratio of FDI stocks to GDP approaching 50 per cent in recent years (see Figure 8.9 and Table 8.9). Moreover, again like Singapore, FDI-based indicators have increased over the three periods identified, with the largest increases coming between 1979–86 and 1987–95. In marked contrast to Singapore, however, foreign MNC shares of production fell between 1970–78 and



### Figure 8.9 Foreign MNC presence in Malaysia

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	3.05	3.41	5.52	0.38	0.39	0.49
FDI stocks/GDP, all	10.14	25.05	39.74	0.55	0.28	0.15
MNC production/GDP, all	17.77	17.17	18.97	0.11	0.07	0.17
MNC production/GDP, mfg	36.05	27.91	36.34	0.15	0.04	0.18

# Table 8.9 Foreign MNC presence in Malaysia,mean and variation

Note: all calculations exclude years for which data are not available Additional notes and sources: see Appendix A 1979–86 but then rose afterwards. In all industries, mean shares changed very little, staying in the 17 to 19 per cent range. This is, in a sense, the opposite of patterns observed in Korea and Taiwan in that foreign MNC shares of production are below what they might be expected to be given the high level of FDI stock to GDP ratios. In manufacturing, the decline in the middle period was more pronounced than in all industries, but the pattern was similar in that foreign MNC shares of production were of similar magnitude in the earlier and later periods. Moreover, as in most other economies in this sample, foreign MNC shares of production tended to fluctuate less than FDI-related measures, the exception being the FDI stock to GDP ratio in the later period.

In contrast to Hong Kong, where the level of foreign MNC presence may be considered paradoxically low, it may be considered paradoxically high in both Malaysia and Singapore, particularly in manufacturing. Part of the reason for this is the importance of the electrical and electronic machinery industry in these economies and the almost total domination of this industry by foreign MNCs. For example, data from the Malaysian Industrial Development Authority (1996) indicate that electrical and electronic machinery accounted for 18 per cent of the fixed assets in all manufacturing in 1994 and that foreign MNCs account for 88 per cent of all the fixed assets in electrical and electronic machinery. Data from the Singapore Economic Development Board (1996) indicate that these shares were also large—28 per cent and 86 per cent respectively—in that country in 1994. As illustrated by Plummer and Ramstetter (1991), in Asia MNCs tend to be present in industries where knowledge-based intangible assets such as technologies generated by research and development and marketing networks (terminology from Markusen, 1991) are relatively important.

Although significant, this is certainly only one of the factors underlying the large foreign MNC presence in these economies. For example, it also seems to be no coincidence that both Singapore and Malaysia have a similar geography and history, though it is very difficult to interpret the role of these factors in economic terms. Indeed, it may be a failure to appreciate the importance of these factors that leads me to consider the high level of foreign multinational presence in Malaysia and Singapore as somewhat paradoxical.

#### Thailand

In Thailand, ratios of FDI flows to GDP remained rather constant throughout the 1970– 78 and 1979–86 periods, but then increased a good deal in 1987–95, while ratios of FDI stocks to GDP increased steadily over the three periods (see Figure 8.10 and Table 8.10). Here again, however, fluctuations in ratios of FDI flows to GDP are quite pronounced, with coefficients of variation being rather large (0.4 to 0.5). Data on foreign MNC shares of production are very sparse and the figures used here refer only to foreign MNCs that are accorded promoted status and responded to surveys by the Board of Investment. These figures indicate that foreign MNC shares were rather steady in the 13 to 16 per cent range in 1974, 1986 and 1990. More comprehensive estimates for 1990 (from Ramstetter, 1994, 1996b) indicate that the share of all foreign MNCs, including nonpromoted MNCs and promoted MNCs that did not respond to the aforementioned survey, in manufacturing value added was roughly twice this level (about 31 per cent). The Thai case combines elements observed in other South-East Asian economies. First, there has been very rapid growth in manufacturing in recent years, especially in electrical and electronic machinery as well as the car industries. As described above, foreign MNCs dominate the former industry and have a very large presence in the latter industry as well. Second, like Indonesia, Thailand previously had relatively high levels of import protection but these barriers have gradually liberalized, especially in recent years. Third, another important factor not yet mentioned with respect to Asian economies is the role of macroeconomic fluctuations.



### Figure 8.10 Foreign MNC presence in Thailand

Table 8.10 Foreign MNC presence in Thailand,mean and variation

	Mean ratios (per cent)			Coefficients of variation		
Measure, industry	1970–78	1979–86	1987–95	1970–78	1979–86	1987–95
FDI flows/GDP, all	0.65	0.63	1.71	0.49	0.41	0.41
FDI stocks/GDP, all	2.45	4.47	9.45	0.44	0.29	0.22
MNC production/GDP, mfg	15.53	13.32	14.76	NA	NA	NA

Note: all calculations exclude years for which data are not available; NA=no data are available for the period

Additional notes and sources: see Appendix A

Thailand, like Indonesia, Malaysia, and Singapore, experienced a marked economic slowdown in the early to mid 1980s. Moreover, due in large part to austere macroeconomic policies during these difficult years, a robust recovery and economic boom followed in the late 1980s. The fact that this recovery coincided with the large appreciations of the yen and the New Taiwan dollar is also thought to be significant by some. Other studies have shown that variation in macroeconomic variables can explain a very large portion of FDI fluctuations in Thailand and in Malaysia (Ramstetter, 1995b,c).

In short, it is no coincidence that the rapid growth of FDI followed closely behind the macroeconomic boom of the late 1980s in these economies.

#### China

Like the United States, China is a large economy that has, by any measure, experienced significant increases in foreign MNC presence in the 1980s and 1990s (see Figure 8.11 and Table 8.11).

China differs a great deal from the other economies studied here, both developed and developing, in that it was a centrally planned economy until 1978 and only began the transition to a market-based economy thereafter. There are two important characteristics of this transition. First, although



### Figure 8.11 Indicators of foreign MNC presence in China

	Mean ratios (per cent)		Coefficients of variation	
Measure, industry	1981–86	1987–95	1981–86	1987–95
FDI flows/GDP, all	0.39	2.50	0.52	0.88
FDI stocks/GDP, all	1.00	8.20	0.75	0.75
Private prod./gross output, industry	1.79	14.27	0.83	0.59
MNC production/gross output, industry	NA	9.07	NA	0.17

**Table 8.11** Indicators of foreign MNC presence inChina, mean and variation

Note: all calculations exclude years for which data are not available; NA=no data are available for the period

Additional notes and sources: see Appendix A

there were virtually no foreign MNCs active in China before this transition, China quickly opened up to foreign MNCs. Moreover, although China instituted severe restrictions on the activities of foreign MNCs, the lure of what is potentially the largest national economy in the world led to a large increase in foreign MNC presence. Indeed, China became among the largest Asian recipients of FDI flows by the late 1980s and became the second largest recipient worldwide, behind the United States, in the mid 1990s (International Monetary Fund, 1997). Accordingly, ratios of FDI flows and stocks to GDP rose rapidly throughout the post-1978 period. Moreover, due in large part to the fact that these ratios increased extremely rapidly, the variation in these ratios within subperiods was also quite large.

Second, there were virtually no private firms in China before the 1978 transition, and the growth of foreign MNCs was a major element of the rapid growth of the private business sector. On a national level it is impossible to measure foreign MNC shares of production directly for 1992 and earlier. However, because the growth of MNCs and private business in general have been closely related in China, it is instructive to look at trends in the shares of private enterprises (that is, firms that are not state owned or collective owned) in the gross output of all industry (that is, value added plus intermediate inputs in manufacturing, plus mining, construction, and electricity, gas and water). This share represents an upper limit on the shares of foreign MNCs in gross output. In a manner very similar to FDI flows and stocks, the private share of industrial production grew rapidly, with large variations within sub-periods that are, again, a direct result of the rapid rates of growth. Since 1993 it is possible to get estimates of gross output for foreign MNCs with independent accounting systems. These MNCs accounted for 7 to 10 per cent of all gross output of all industrial enterprises in 1993-95, compared to 19 to 29 per cent for all private firms. In short, as China has liberalized, foreign MNCs have apparently seized a substantial portion of the opportunities that China's economic reforms have allowed private business to pursue.

### 8.5 Conclusion

The evidence presented above suggests at least three very simple conclusions. First, trends in foreign MNC shares of production in the eleven economies studied suggest that foreign MNCs have become more important in recent years in a number of economies in the region, and probably the region as a whole. For example, these ratios rose very rapidly in China, rather rapidly in Hong Kong and the United States, and more modestly in Singapore, Taiwan and Malaysia. Moreover, it is fairly clear that these increases far outweigh the moderate decreases in Canada, Japan and Indonesia, mainly due to the relatively big increases in the large US and Chinese economies. A similar conclusion can be reached about foreign MNC shares worldwide (see Lipsey *et al.*, 1995). However, it is also important to emphasize that the region-wide increase was heavily concentrated in two economies, China and the United States.

Second, if one interprets FDI-based measures as a proxy for foreign MNC presence, one gets the impression that the increase in foreign MNC presence was much greater than if one focuses on foreign MNC shares of production. For example, FDI-based measures were higher in 1987–95 than in 1970–78 or 1979–86 in a wide range of Asia-Pacific

economies—Canada, the United States, Hong Kong, Singapore, Taiwan, Indonesia, Malaysia and Thailand. However, foreign MNC shares of production increased very little, or actually declined, in most of the economies studied—Canada, Singapore, Taiwan, Indonesia, Malaysia and Thailand. Furthermore, although both FDI-based measures and foreign MNCs shares of production increased appreciably in Hong Kong and the United States, the rate at which foreign MNC shares of production increased was much lower than the rate at which FDI-related indicators increased. In short, only one economy, China, experienced similarly high rates of growth in foreign MNC shares of production and FDI-related indicators.

Third, if one evaluates foreign MNC presence from FDI-based indicators, one would conclude from the large variation over time that it is extremely volatile. However, if one evaluates on the basis of foreign MNC shares of production, foreign MNC presence appears to be much less volatile, with variation in these shares tending to be a good deal lower than variation in FDI-based indicators in Canada, Japan, the United States, Hong Kong, Singapore, Indonesia and Malaysia. Only in Korea, Taiwan in the later period, and China is variation of a similar magnitude for both measures of foreign MNC presence.

In addition to describing the evidence leading to these three simple conclusions, the chapter also tried to point out some of the major causes for the trends observed. These causes are numerous, and the relationships involved are very complex, making it impossible for this chapter to draw clear and comprehensive conclusions. Indeed, it would take at least a book, including a lot of statistical analysis that has not yet been performed, to accomplish that task. However, it is possible to indicate several factors that such an analysis would probably reveal to be important.

First, macroeconomic trends, for example interrelated trends in economic growth rates, inflation rates, interest rates, and exchange rates, as well as related policies, were thought to be important. Second, policies towards foreign MNCs are of obvious importance in some cases. Third, other policies, for example those towards international trade, international capital flows, taxation, and private business in general were also thought to be significant. Fourth, industrial structure was also thought to be important, as MNCs tend to be concentrated in industries with specific characteristics.

To return to the three simple conclusions outlined above, it is worth emphasizing two important implications that follow from the second and third of them. The second finding, namely that FDI-based indicators and foreign MNC shares of production often display very different trends, strongly suggests that FDI-based indicators are rather poor indicators of foreign MNC presence. More specifically, since foreign MNC shares of production are clearly a more accurate measure of foreign MNC presence, focusing on FDI-related measures apparently leads to significant over estimation of the extent to which foreign MNC presence has grown in the Asia-Pacific region since the 1970s. In this respect, however, it is also important to acknowledge that this chapter has presented estimates of foreign shares in terms of one indicator only, that is production. Similar calculations of foreign MNC shares of production. Thus, the reader should understand that this chapter analyses limited aspects of foreign MNC presence only and that more comprehensive descriptions remain an item on the agenda for future research. The third conclusion described above, that foreign MNC shares of production tend to vary less over time than FDI-related indicators, has perhaps an even more important implication. Namely, economic activities of foreign MNCs have often been viewed as highly volatile by host economy policymakers, a view that is reinforced if one focuses on the large variation of FDI-related indicators. On the other hand, the relatively low variation in foreign MNC shares of production suggests that this view may be incorrect. In other words, although FDI may be a relatively volatile and hence unreliable source of balance of payments' finance, the production of foreign MNCs is generally much less volatile and hence more reliable.

Finally, if this chapter has one message to communicate to its readers, it is to reiterate the need for close attention to data issues when analysing MNCs in the Asia-Pacific region (or elsewhere). Correspondingly, it is wise to be highly sceptical of any analyst, economist or otherwise, pretending to know more than the data can tell one.

#### References

- Asian Development Bank (various years a) *Key Indicators of Developing Member Countries*, vol. 15–16 (April issues, 1984–85), vol. 17–21 (July issues, 1986–90), Manila, Asian Development Bank.
- Asian Development Bank (various years b) Key Indicators of Developing Asian and Pacific Countries, vol. 22–27, (1991–96 issues), Manila, Oxford University Press.
- Bora, B. (1996) 'Foreign direct investment' in Bora, B. and Findlay, C. (eds) *Regional Integration and the Asia-Pacific*, Melbourne, Oxford University Press.
- Buckley, P.J. and Casson, M. (1991) *The Future of the Multinational Enterprise*, 2nd edn, London, Macmillan.
- Caves, R.E. (1996) *Multinational Enterprise and Economic Analysis*, 2nd edn, Cambridge, Cambridge University Press.
- Dunning, J.H. (1993) Multinational Enterprises and the Global Economy, Workingham, Addison-Wesley.
- Encarnation, D. (1992) *Rivals Beyond Trade: America versus Japan in Global Competition*, Ithaca, Cornell University Press.
- Fahim-Nader, M. and Zeile, W.J. (1997) 'Foreign direct investment in the United States: new investment in 1996, affiliate operations in 1995', Survey of Current Business, June, pp. 42–69.
- Hong Kong, Census and Statistics Department (various years) *Survey of Industrial Production*, 1992–94 issues, Hong Kong, Census and Statistics Department.
- Indonesia, Biro Pusat Statistik (various years) diskettes with data from *Statistik Industri*, 1990–94, Jakarta, Biro Pusat Statistik.
- International Monetary Fund (1997) *International Financial Statistics*, March, CD-ROM, Washington, DC, International Monetary Fund.
- Japan, Ministry of Finance (various years) *Hojin Kigyou Toukei Shiho (Corporation Statistics Quarterly)*, April-June issues 1994–96.
- Japan, Ministry of International Trade and Industry (1995) Gaishikei Kigyou no Doukou (Trends in Foreign-Owned Firms), no. 27, Tokyo, Ministry of Finance Printing Bureau.
- Kreinin, M.E., Plummer, M.G. and Abe, S. (1997) 'Export and direct foreign investment links: a three country comparison', paper presented at International
- Conference on 'International Economic Links and Policy Formations: Where do we go from here?', Kobe University, Kobe, Japan, June 12–13.
- Lipsey, R.E., Blomstrom, M. and Ramstetter, E.D. (1995) 'Internationalized production in world output', NBER Working Paper 5385, Cambridge, Mass.

- Low, L., Ramstetter, E.D. and Yeung, H.W-C. (1996) 'Accounting for outward direct investment from Hong Kong and Singapore: who controls what?', NBER Working Paper 5858, Cambridge, Mass.
- Malaysia, Department of Statistics (various years) *Report on the Financial Survey of Limited Companies*, 1992–94 issues, Kuala Lumpur, Department of Statistics.
- Malaysia Industrial Development Authority (1996) Statistics on the Manufacturing Sector 1991– 1995, Kuala Lumpur, Malaysia Industrial Development Authority.
- Markusen, J.R. (1991) 'The theory of the multinational enterprise: a common analytical framework' in Ramstetter, E.D. (ed.) Direct Foreign Investment in Asia's Developing Economies and Structural Change in the Asia-Pacific Region, Boulder, CO, Westview Press.
- Organisation for Economic Co-operation and Development (OECD) (various years) *Geographical Distribution of Financial Flows to Developing Countries*, 1990–93, 1991–94 issues, Paris, OECD.
- People's Republic of China, State Statistical Bureau (various years) *China Statistical Yearbook*, 1994–96 issues, Beijing, State Statistical Bureau.
- Petri, P.A. and Plummer, M.G. (1995) 'The determinants of US direct investment abroad: evidence of trade-investment linkages', paper presented at International Conference on Economic Development and Co-operation in the Pacific Basin, University of California at Berkeley, Berkeley, CA, 30 June–1 July.
- Plummer, M.G. and Ramstetter, E.D. (1991) 'Multinational affiliates and the changing division of labor in the Asia-Pacific region' in Ramstetter, E.D. (ed.) *Direct Foreign Investment in Asia's Developing Economies and Structural Change in the Asia-Pacific Region*, Boulder, CO, Westview Press.
- Ramstetter, E.D. (1994) 'Comparisons of Japanese multinationals and other firms in Thailand's non-oil manufacturing industries', *ASEAN Economic Bulletin*, vol. 11, no. 1, pp. 36–58.
- Ramstetter, E.D. (1995a) 'Characteristics of multinational firms in Malaysia: a time series perspective' in Toida, M. and Hiratsuka, D. (eds) Ajia Kogyoken no Keizai Bunseki to Yosoku (IV), (Projections for Asian Industrializing Region (IV)), Tokyo, Institute of Developing Economies.
- Ramstetter, E.D. (1995b) 'Sectoral flows of foreign direct investment in Malaysia' in Economic and Social Commission for Asia and the Pacific, Sectoral Studies of Foreign Direct Investment in Asia and the Pacific, ST/ESCAP/1501, New York, United Nations.
- Ramstetter, E.D. (1995c) 'Sectoral flows of foreign direct investment in Thailand' in Economic and Social Commission for Asia and the Pacific, Sectoral Studies of Foreign Direct Investment in Asia and the Pacific, ST/ESCAP/1501, New York, United Nations.
- Ramstetter, E.D. (1996a) 'Estimating economic activities by Japanese transnational corporations: how to make sense of the data?', *Transnational Corporations*, vol. 5, no. 2, pp. 107–43.
- Ramstetter, E.D. (1996b) 'Trends in production in foreign multinational firms in Asian economies: a note on an economic myth related to poor measurement', *Kansai University Review of Economics and Business*, vol. 24, no. 1–2, pp. 49–107.
- Ramstetter, E.D. (1996c) 'Characteristics of Singapore's manufacturing establishments by nationality of ownership' in Toida, M. and Hiratsuka, D. (eds) *Ajia Kogyoken no Keizai Bunseki* to Yosoku (V) (Projections for Asian Industrializing Region (V)), Tokyo, Institute of Developing Economies.
- Ramstetter, E.D. and James, W.E. (1993) 'Multinationals, Japan-US economic relations, and economic policy: the uncomfortable reality', *Transnational Corporations*, vol. 2, no. 3, pp. 68– 96.
- Republic of China, Investment Commission (various years) A Survey of Overseas Chinese and Foreign Firms and Their Effects on National Economic Development, 1992–94 issues, Taipei, Investment Commission (in Chinese).
- Republic of China, Central Bank of China (1997) 'Balance of payments, Taiwan District, The Republic of China', March issue, Taipei, The Central Bank of China.

Rugman, A.M. (1985) 'Internalization is still a general theory of foreign direct investment', Weltwirtschaftliches Archiv, Band 121, Heft 3, pp. 570–5.

Singapore Economic Development Board (1996) mimeos of unpublished data from the Census on Industrial Production, September 1996, December 1996.

STAT-USA (1997) Internet-based collection of US government statistics, accessed June 1997.

- United Nations Conference on Trade and Development (1994) *World Investment Report 1994, Transnational Corporations, Employment and the Workplace,* New York and Geneva, United Nations (Sales No.E94.II.A.14).
- United Nations Transnational Corporations and Management Division (1993) World Investment Directory, Volume III: Developed Countries, New York, United Nations (Sales NO.E.93.II.A.9).

World Bank (1995) World Data 1995, CD-ROM, Washington, DC, World Bank.

World Bank (1997) World Development Indicators 1997, CD-ROM, Washington, DC, World Bank.

#### Appendix A Notes on the data and their sources

The following notes and sources are summarized by indicator and country.

### 1 Foreign direct investment (FDI)

*Notes:* Reinvested earnings are not included for Japan, Singapore, Indonesia, Malaysia, Thailand and China. For Hong Kong, FDI data refer to FDI from OECD economies only, as reported by OECD economies. For all other economies data are taken from balance of payments statistics. Original data are reported in US dollars or special drawing rights (SDKs) and translated to US dollars using annual average exchange rates. Stocks are cumulative flows calculated from US dollar figures.

*Sources:* International Monetary Fund (1997); Ramstetter (1996b); Republic of China, Central Bank of China (1997).

### 2 Gross domestic product (GDP)

*Notes:* Data reported in local currency and translated to US dollars using annual average exchange rates.

*Sources:* Asian Development Bank (various years a, b), International Monetary Fund (1997), Ramstetter (1996b); World Bank (1995, 1997).

#### 3 MNC production and foreign MNC shares

For detailed notes on these data sources see Lipsey *et al.* (1995) and Ramstetter (1996b). Data for 1991, 1992, or 1993 (depending on the country) are generally taken from these sources. Updates are from the additional sources summarized below.

Japan: data for 1993 from Japan, Ministry of Finance (various years) and Japan, Ministry of International Trade and Industry (1995).

United States: foreign MNC production for 1990–95 are from Fahim-Nader and Zeile (1997) and shares are calculated with GDP data from the sources noted above and STAT-USA (1997).

Hong Kong: data for 1993–94 from Hong Kong, Census and Statistics Department (various issues). All data refer to gross value added of firms with majority- or wholly-foreign plants as a share of all gross value added in manufacturing.

Singapore: data for 1993–1994 from Singapore Economic Development Board (1996). All data refer to gross value added of firms with majority- or wholly-foreign plants as a share of all gross value added in manufacturing.

Taiwan: data for 1992–94 are taken from Republic of China, Investment Commission (various years). All data refer to published MNC shares of production.

Indonesia: data for 1991–94 are extrapolations based on the series in Ramstetter (1996b) and Indonesia, Biro Pusat Statistik (various years). Extrapolations are used because the 1975–90 estimates are from the backcast series which is adjusted for plants not responding to the industrial survey, while the 1991–94 data are from raw data not adjusted in this manner.

Malaysia: data for 1992–94 are from Malaysia, Department of Statistics (various years). All data refer to production of foreign limited companies as shares of GDP.

### CHAPTER 9 East-Asian and Anglo-American business systems

Richard Whitley

### 9.1 Introduction

The post-war success of Japanese firms in Western markets has been accompanied by the widespread recognition that they behave in quite different ways from many Western ones, especially US and British companies. The more recent growth and export success of firms in South Korea (henceforth Korea), Taiwan, Hong Kong and Singapore—the four 'little dragons'—has also highlighted the considerable success and variety of economic systems in East Asia. Firms and markets in these economies not only differ greatly from those in Anglo-American capitalism—including Australia—but also vary considerably among themselves. The post-war Japanese *kaisha* (corporation) is quite different in many important respects from the Korean *chaebol* (conglomerate), and from the Chinese family business (CFB) that dominates the export industries of Taiwan and Hong Kong (see Chapter 6). Additionally, the ways in which these firms deal with each other also differ such that they constitute distinctive business systems, or forms of capitalism (Whitley, 1992a).

These different types of capitalism reflect variations in processes of industrialization—which in turn partly reflect differences in the nature of pre-industrial Japan, Korea and China—and continuing differences in the nature of dominant institutions in these societies. The ways in which the political, financial and labour systems have developed during the twentieth century vary considerably across East Asia and have led to contrasting kinds of firms and market relations becoming established. Continuing differences between these societies mean that these contrasts will remain significant despite the growing interdependence of their economies and firms.

This chapter describes the main characteristics of these varieties of capitalism as distinct business systems. First, the key components and characteristics of business systems as particular forms of economic organization are described. Then a brief section outlines the main features of the Anglo-American business system, which typifies to varying degrees the USA, Australia and New Zealand in the Asia-Pacific region. The main body of the chapter summarizes in turn the central characteristics of the Japanese, Korean and Taiwanese business systems as they have developed over the past four decades or so. Comparisons with the dominant features of Anglo-American business systems are drawn where appropriate. These business systems are treated in a very generalized way in this chapter. The Anglo-American one, in particular, displays considerable variation between the countries and regions in which it operates, and between different sectors. The picture presented here is something of a stereotype which
can act as a benchmark against which the distinguishing features of the other systems are compared.

# 9.2 Components of business systems

In comparing and contrasting the organization of market economies, it is useful to summarize their major characteristics as aspects of particular *business systems*. Business systems are distinctive configurations of firms and markets which have become established in particular institutional contexts as the dominant ways of structuring economic activities (Whitley, 1992b). These contexts are usually national because of the critical role of the nation state in establishing and maintaining property rights and managing economic development and other key institutional arenas such as the labour system. There are three basic components of business systems that vary in particular ways to constitute distinct forms of economic organization. They are:

- the nature of economic agents;
- the nature of market relations;
- the nature of the work co-ordination and control system.

# The nature of economic agents

The first basic component of a business system is the nature of the economic agent, or 'firm'. Is it, for instance, the diversified, divisionalized Anglo-American corporation, or the small family business of the Italian industrial districts and overseas Chinese communities (Redding, 1990)? It is important to note here that the privately owned resource-controlling and allocating unit is the object of analysis, not necessarily the individual 'firm' which may be only part of a larger grouping.

Four critical aspects of economic agents that vary between business systems are:

- 1 The diversity of activities controlled and the capabilities developed, from being narrowly specialized in a single production process, through to some integration of complementary activities in a production chain, to a high degree of vertical and horizontal diversification.
- 2 The degree and rate of change in central, or 'core', activities, from Western holding companies buying and selling businesses as items in an investment portfolio to firms committed to particular skills and industries.
- 3 The extent of owner involvement in, and control over, the management of firms. Portfolio investors in Western, especially Anglo-American, capital markets, for example, typically have little interest in the affairs of any individual firm, as they focus on financial returns from holding particular assets. Banks and other large owners who are locked into the destinies of specific clients, on the other hand, have to take a much greater interest in their success and often share risks with them in undertaking new developments. Personal and family owners, of course, are usually highly involved in the management of their firms.

4 The nature of risk management and growth strategies differ significantly across countries in terms of the degree of internalization of risks and restriction of commitments, and the focus on internal growth as distinct from external acquisitions.

# Market organization

The second component of business systems concerns the nature of market relationships between firms. Are they, for example, mostly short term, *ad hoc* and impersonal, as in many commodity markets and Anglo-American industries, or are they more particularistic with longer-term commitments being common between specific firms? And are relations between firms co-ordinated, vertically and/or horizontally, by powerful intermediaries, such as the Japanese general trading companies, or are they highly competitive and predatory with outcomes largely determined by short-term price movements? In general terms, market economies vary considerably in the importance and stability of networks and commitments, and obligations between firms, as is exemplified by the analysis of business groups and networks in East Asia in comparison with Western economies (Hamilton and Biggart, 1988; Orru *et al.*, 1991). Two key aspects of market organization concern:

- 1 The extent of vertical co-ordination, which refers to the degree to which exchange relationships between suppliers and customers within an industry are repeated and based upon mutual commitments with some knowledge and risk sharing, as well as involving third parties such as banks or trading companies.
- 2 Horizontal collaboration, which refers to the extent to which agents co-operate in setting standards, lobbying state agencies, negotiating with labour organizations, establishing and monitoring training programmes, etc. within industries, and share information, resources and developments across industries.

# Work co-ordination and control

The third component of business systems is the dominant pattern of work co-ordination and control, employment relations and management. Economies clearly differ in the ways in which work is divided, allocated and co-ordinated in major economic agents, and these variations are closely connected to differences in dominant institutions, especially those governing the generation and availability of skilled labour power. This component is not just a matter of bureaucratization and centralization, but rather includes broader considerations of authority relations and personnel policies as part of the general pattern of work organization which has become established in particular market economies. It is the overall way in which economic agents manage their resources and activities that is the focus here, not so much the particular variations in formal organizational structures. Key dimensions here are:

1 the centralization of decision making and control, which concerns the delegation of operational and strategic control to non-owning managers and to lower levels of the hierarchy within organizations;

- 2 employer-employee interdependence, which refers to the extent to which both employers and employees are locked into each other's destinies and cannot easily change these commitments;
- 3 manager-worker distance, which concerns the separation between managerial staff and manual workers in terms of their education and skills, conditions of employment and reward systems;
- 4 discretion over how work is performed varies greatly between business systems with some institutionalizing strong supervisor control while others delegate substantial control to workers, either in groups or individually;
- 5 specialization, which likewise varies considerably between the rigid, inflexible specification of tasks, jobs and roles characteristic of 'scientific management' principles and the more fluid and diffuse approaches of many Asian firms.

These dimensions of the three basic components of business systems are listed in summary form in Box 9.1 and will be applied to the post-war business systems of Japan, Korea and Taiwan after a brief discussion of the Anglo-American business system.

# Box 9.1 Business system components and their key characteristics

# The nature of economic agents

- 1 Diversity of economic activities and capabilities co-ordinated by owners and/or managers.
- 2 Degree and rate of change of economic activities and capabilities.
- 3 Extent of owner involvement in, and control over, firm management.
- 4 Prevalent growth and risk management strategies.

# Market organization

- 1 Extent of vertical co-ordination of inputs and outputs and risk sharing between economic agents in production chains.
- 2 Extent of horizontal collaboration between firms within an industry and between firms across industries.

# Work co-ordination and control

- 1 Centralization of decision making and control.
- 2 Extent of employer-employee mutual dependence and commitment.
- 3 Manager-worker differentiation and separation.
- 4 Supervisor control of work processes and work organization.
- 5 Task and role specialization

# 9.3 The Anglo-American business system in summary

Bearing in mind the general characteristics of a business system outlined in the previous section, how can we characterize the system that typifies the Anglo-American countries? Using the headings in Box 9.1 as a guide, Box 9.2 summarizes the main features of this particular system (Chandler, 1990; Hollingsworth, 1991).

# Box 9.2 Characteristics of the Anglo-American business system

Nature of economic agents

- 1 Diversified and divisionalized firms. Conglomerate operations common.
- 2 High rate of company formation and death. Take-over and mergers the common means for company expansion.
- 3 Diversified and arm's length share ownership. Owners see their share holding as a portfolio, with little interest in the direct management of firms. Financial returns stressed. Low levels of small and medium sized firms with family ownership.
- 4 Low degree of organic internal growth; external acquisition levels high. Risks are internalized rather than shared.

# Market organization

- 1 *Ad hoc* and impersonal market relations; little close networking with suppliers; any vertical integration is 'internal' to firms rather than through external relations.
- 2 Low levels of horizontal collaboration. Little cross company share holding. Low commitment to industry or sector-wide information exchange, training or lobbying.

# Work co-ordination and control

- 1 Highly centralized strategic decision making and control by managers. Financial targets and control tight and paramount.
- 2 Frequent job changes for both shop floor and managerial personnel. Reliance on external labour market. Frequent lay-offs.
- **3** High separation of managers' and workers' reward structures and benefits systems. Inflexible and job-specific compensation schemes.
- 4 Strong supervision and control of workers, little worker autonomy, low levels of team work and 'trust'.
- 5 High individual specification of tasks. Little generalized skill competencies. Worker organizations reinforce demarcation structures.

As mentioned in the introductory remarks, the summary in Box 9.2 represents something of a stereotypical view of the Anglo-American system. In fact, there are significant variations between countries and sectors, and the system is subject to dynamic changes that have altered its typical characteristics over the years. But the summary can act as a convenient benchmark for comparisons. Before that is done, however, it is worth going through the features of the system in a little more detail.

#### The nature of economic agents

As far as the nature of economic agents in the Anglo-American system is concerned, the typical leading firm has been a large diversified corporation, particularly in the 1970s and 1980s. These operate in a range of product markets where there is no necessary connection between the different branches of the business. Firms thus tend to be operated as divisionalized units, with a relatively separated structure of functions within them (marketing, finance, R&D, production units). One feature of this system is the high rate of company formation and extinction. This is partly because of the turbulent nature of the economies in which the firms exist, booms and slumps leading to high birth and death rates for companies as the business cycle takes its course. But it is also due to an endemic feature of the financial systems in these countries, which encourages take-over and merger activity by companies. Company expansion tends to be through the absorption of other companies as much as through internally-generated growth.

The form of the financial systems in Anglo-American economies is a crucial element in the way their business systems work. 'Outside' ownership is prevalent, where individuals and, more importantly, specialist financial institutions own and manage a share portfolio of company assets. There is little direct cross-holding of shares between companies, or ownership by commercial banks. The direct management of companies is left to the managerial 'insiders' who run them, whilst the owners tend to look upon their share holding as a financial asset from which they derive an income. This is often argued to lead to a short-term outlook by the managers of companies, whose main objective is to provide a steady income stream from dividends for their shareholders and to enhance shareholder value. Thus they can often neglect the longer-term productive and innovative development of their companies. One consequence of this has been the relatively low proportions of small and medium sized firms in these systems.

#### Market organization

The market structure in these systems tends to be organized on an arm's length basis. Separate firms contract between themselves for specific inputs and outputs as discrete organizations. Supply chains are a discrete set of stages, organized via price relationships on a one-to-one competitive basis. Long-term close relationships between suppliers and main firms, seen as part of an extended network of co-operative operations, are relatively rare. Traditionally, the organization of production has relied upon an internal vertical relationship between different stages in a single production process within the firm, though this may be breaking down somewhat as more and more functions are externalized into specialized units. Managerial integration of relatively distinct but related aspects of business activity within the same firm is rather weak. The separation

and independent nature of the business units has meant that there is traditionally a low level of commitment to industry-wide bodies which provide technological intelligence, common training and skill development programmes, information on market developments, and the like, and which represent the branch or industry politically at the level of lobbying and bargaining.

#### Work co-ordination and control

It is when we move to the area of work organization that the main differences amongst the Anglo-American business systems in the Asia-Pacific region emerge. These have traditionally arisen because of different degrees of worker organization. The US economy, for instance, has usually displayed low levels of organized labour, which has had to operate in a generally hostile commercial environment. Australia, on the other hand, has traditionally had high levels of unionized labour, which for long periods was explicitly incorporated, as a kind of 'partner', into many economic and business management decision-making processes. However, bearing these differences in mind, there are some broad common features in the way that work has tended to be organized.

Managerial prerogatives are strongly represented and defended. Strategic decision making is centralized within firms, particularly in the case of financial controls and targets. This is seen as the main way that diversified and divisionalized firms can be monitored and controlled. A second feature is that changing jobs can be common, particularly for top managers, who regularly move in and out of different companies. Further down the job hierarchy, security of tenure is low as lay-offs and redundancies are common. There tends to be a two-tiered labour market, where first resort is to an internal, sometimes informal, recruitment procedure, followed by a resort to the external labour market for outside recruitment. The labour force and the labour market are seen as highly differentiated and discriminating, however. Job demarcation is strong, payment systems discriminate between job grades, and benefit systems within the firm are often highly differentiated and inflexible. 'Social systems' within the firm are hierarchical, with little autonomy granted to workers. Team work is not common, so supervision is strong and ubiquitous.

Where there is strong independent worker organization, the demarcation of work tasks is often high. Them and us' attitudes have prevailed, with conflictual relations between management and worker organizations. Generalized skill competencies and flexibility in work organization tend to be low.

This description sums up the main features of the Anglo-American business system. Of course it has developed over the twentieth century and is not static, and, as suggested earlier, there is a lot of variation. An important feature of the system is the way in which it is incorporating elements of the East-Asian, and particularly the Japanese, system as these countries set up businesses within the USA and other Anglo-American economies (especially the UK).

#### 9.4 The Japanese business system

#### The nature of economic agents

The major locus of authoritative resources control and co-ordination in post-war Japan is the large, industrially specialized corporation, or *kaisha*, run by internally promoted university educated managers with considerable autonomy from shareholders (Abegglen and Stalk, 1985, p. 177; Clark, 1979, pp. 55–64; Odagiri, 1992; see also Chapter 6). These corporations are enmeshed in extensive networks of mutual obligation and trust which both provide support and constrain strategic choices (Gerlach, 1992).

Considering first their diversity, Japanese companies are relatively specialized in two ways. First, they tend not to incorporate all the production and allied processes required to manufacture particular products into the managerial hierarchy in the way that large corporations do in North America. Second, they tend to restrict their major fields of economic activity to a single industrial sector and diversify within it. Thus, unrelated diversification is lower in Japan than in North America (Kagono *et al.*, 1985, pp. 25–49).

Japanese companies concentrate much more on the particular skills and competencies which distinguish them as successful competitors than is common in Anglo-American firms. While the latter typically co-ordinate a considerable variety of activities and functional skills through the authority system, Japanese firms prefer to subcontract many activities which, although essential to their products and services, are basically complementary to their main activity. According to Friedman, 'about 70 per cent of all companies with more than thirty employees subcontract for a proportion of their production needs' (1988, p. 146). Subcontracting is often focused on activities which can be relatively easily measured and assessed, such as the production of standardized parts and services. As Clark suggests, while production in America and Britain is a matter of organizing people within companies, in Japan it is a matter of organizing companies and their interconnections (1979, p. 64).

This focus on specialized, similar activities relying on the same set of capabilities is echoed by a preference for specializing in a single industry and sector. While diversification into new fields of activity does occur quite often in Japan, this tends to be related to current capabilities, such as Honda moving from motor cycles to cars and Komatsu from construction equipment to forklift vehicles. Furthermore, once established as successful growing companies, these subsidiaries are usually hived off as separate, independent companies. Thus Toyota set up a specialized steel factory in 1935 because they could not find a supplier in Japan who could produce steel of the required quality, but then spun it off as an independent firm, Aichi Steel, in 1940 (Cusumano, 1985, p. 63). Similarly, Fujitsu separated itself from its fast growing numerical controlled machine tool and robotics division, Fanuc, in 1972 (Abegglen and Stalk, 1985, p. 180). This separation tends to be more thoroughgoing in Japan than in the West, with former subsidiaries developing their own networks and alliances and establishing their own enterprise union (Clark, 1979, p. 61; Odagiri, 1992, pp. 144–51).

The extensive reliance of large Japanese *kaisha* on subcontracting is closely connected to another significant feature of the Japanese business system: the continued importance

of small and medium sized firms. The significance of small firms in Japan, especially in manufacturing, has often been seen as the key to large firm success there because many are tied to large assembly firms who can squeeze them when demand declines. However, this dual structure appears less widespread in the 1960s and 1970s than it was in the 1950s, especially in the machinery industry, and many small firms are less dependent on a single large customer than was once thought (Aoki, 1988, pp. 219–23).

A further important characteristic of large firms in Japan is the considerable autonomy their managers have from property rights' holders. Although family share holdings were significant sources of co-ordination and power in the pre-war *zaibatsu*—the diversified holding companies which dominated the Japanese economy up to the Second World War—salaried, college educated managers were relied upon to run them throughout the twentieth century. Since the US occupation, which broke up the *zaibatsu* and prohibited large holding companies, the owner controlled business has largely disappeared from the ranks of large Japanese firms and the separation of ownership from control has become considerable. As Abegglen and Stalk put it, 'the common stock shareholder in the Japanese company is more in the position of a preferred shareholder in a Western company' and has little, if any, voice in corporate affairs (1985, p. 184; see also Aoki, 1988, pp. 120–7).

These characteristics of large firms in Japan have major implications for the ways in which strategic choices are made and firms develop. The close relationships with suppliers and customers enable them to share risks with business partners but also restrict firms' choices and limit their ability to take major breaks from previous activities. Radical changes in the nature of business activities undertaken by firms are also restricted by the long-term commitment to the current 'core' workforce and its skills that is characteristic of post-war *kaisha*. Kagono *et al.* found that top managers in Japanese firms felt much more constrained in seeking new markets and changing direction by their relations with distributors, customers, suppliers and competitors than did those in US firms in the same industries (1985, p. 26).

Additionally, specialization plus extensive subcontracting plus internal promotions to the board of directors and autonomy from outside shareholders encourages incremental decision making rather than long-term grand designs. Change in large Japanese firms is more a process of continual improvement to current operations and developing internal resources than major shifts in direction involving the acquisition of new and different resources. Pressures for short-term financial results are weaker than in the USA and growth goals dominate (Abegglen and Stalk, 1985).

# Market organization

The Japanese business system is characterized by high levels of market organization and dominated by strong inter-firm mutual obligation networks, some of which constitute large business groups, such as those formed by the Mitsui, Mitsubishi and Sumitomo associated companies (Hamilton and Biggart, 1988; see also Chapter 6). The relatively specialized nature of many *kaisha* means that transactions which would be co-ordinated internally in most Western societies occur across firms' boundaries in Japan. As a result, the volume of wholesale trade in Japan is four times the volume of retail trade, compared to a ratio of less than two in many Western countries (Dore, 1986, p. 80). These

transactions are typically organized around relatively long-term commitments between particular firms which range over a variety of exchanges. Networks of such commitments link firms within and across industrial sectors to a much greater extent than is common among Western companies (Clark, 1979, pp. 221–2) and facilitate information exchange, joint technical development and risk sharing. While some of these networks are based on share swaps and joint ownership of other companies' shares, firms are also connected through joint financing of new ventures, joint production agreements and the development of common distribution channels.

Because firms tend to trade with a relatively small number of specific partners on a long-term basis, they compete for partnerships covering many transactions with customers rather than for individual, one-off sales. Thus, market exchanges in Japan are more specific to, and reciprocal with, particular economic actors than in many Western countries. They are more organized in the sense that they typically occur repeatedly between specific firms and incorporate a number of transactions. They also tend to be linked to broad obligation and trust relations (Sako, 1992).

A particularly important feature of these quasi-contractual networks is the emergence of stable business groups which collectively dominate the Japanese economy. These groups consist of quite large numbers of independent firms which are linked by a number of means, such as mutual share holdings, sharing information, managers and sometimes investments, on a long-term basis (Gerlach, 1992; Orru *et al.*, 1991; Westney, 1996). Two major kinds of business groups can be distinguished: vertically organized *keiretsu* and horizontally connected inter-market groups, or *kigyo shudan* (Odagiri, 1992, pp. 167–88). The former, such as those co-ordinated by the major car assemblers Toyota and Nissan, co-ordinate flows within a particular sector or industry, while the latter, such as Mitsui and Mitsubishi, link activities in different sectors, including financial services and international trade. Whereas the former tend to be dominated by a single large prime contractor, the latter groups are less hierarchically organized and there is not usually a central co-ordinating agency.

The sixteen largest business groups of both kinds contained 65 of the largest 100 firms in Japan in 1980, controlled 26 per cent of the total paid up capital of all non-financial enterprises, sold 33 per cent of all manufactured goods by value and employed a fifth of the total manufacturing workforce (Orru *et al.*, 1991). Collectively they dominate the financial services and manufacturing sectors, especially in transportation equipment, chemicals, electronics, basic metals and petroleum. While not integrated through formal authority structures, these extensive and cohesive networks of reciprocal share holdings, joint ventures and information sharing co-ordinate a wide variety of economic activities in Japan.

The largest vertically linked business groups grew out of the networks of subcontractors already described. Thus among the ten largest 'independent' and 'semiindependent' groups are those organized by Nissan, Hitachi, Matsushita and Toshiba-IHI. They include their major subcontractors as well as successful subsidiaries that have been set up as separate companies. Additionally, two large, more diversified groups have formed around the Tokai Bank and the Industrial Bank of Japan (Orru *et al.*, 1991). These latter groups, and others based upon department stores and railway companies, tend to be less integrated and cohesive than the vertically linked networks dominated by a single large parent company, or the six largest inter-market groups.

#### Work co-ordination and control

Turning now to consider how work is organized and controlled inside large Japanese firms in the post-war period, the most marked contrasts with US corporations concern the division of labour, together with differences in the work group discretion and performance assessment, and employment policies. Essentially, jobs and responsibilities are less individually specific and separate in Japanese firms and roles often overlap (Kagono *et al.*, 1985, pp. 112–21; Koike, 1987). Such overlap is easier to manage when employers and core employees are highly dependent upon each other and expect to be working together for a long time, as in the Japanese organization-based employment system (Dore, 1986). The dominant unit of work performance and assessment is the work group rather than the individual and, despite elaborate work manuals, these groups have considerable latitude in deciding how tasks are to be carried out (Lincoln and Kalleberg, 1990, pp. 85–9).

Responsibilities for making decisions are also shared and collective to a much greater extent in Japan than in many Western firms. The *ringi* system of circulating proposals and suggestions for approval by all interested parties before final approval by top management ensures widespread commitment before decisions are made and diffuses responsibility for it. As Clark puts it: 'In Japan [decision making] is presented as collective until it is worth someone's while to claim a decision as his own' (1979, p. 130). In general, individual authority to commit resources is highly constrained and restricted so that managers usually have to obtain the support of colleagues for any substantial project and cannot make decisions independently. Thus, departments and sections are not regarded as 'belonging' to their heads in the way some Western managers consider they have the right to control 'their' organizational unit. Because of overlapping responsibilities and limited individual authority, ownership is shared and collective rather than specific to the individual manager (Clark, 1979, pp. 126–34).

The considerable reliance on 'bottom up' ways of initiating change in Japanese firms, exemplified by the *ringi* system, suggests that decision making is less centralized in practice than purely formal measures of the source of ultimate authorization would suggest (Lincoln and Kalleberg, 1990, pp. 206–10). While top management often has to agree to proposals, in many cases this is largely a formality once the circulation of the *ringi* has produced widespread support. Furthermore, where top management suggests changes which are not wholeheartedly supported by middle management groups, they may not be carried through.

This strong role of middle managers in large Japanese firms is linked to the distinctive nature of their employment policies. The institutionalization of long-term employment commitments to the 'core' workforce—which has been extended to most male manual workers in large companies since the war (Koike, 1987; 1994)—together with a correspondingly high dependence of employees on their particular employer, has facilitated the development of trust and common commitments among different groups of managers and workers. Delegation of significant influence to middle managers is obviously easier when nearly all male employees have spent their working lives in the same firm and are dependent on its success.

The long-term employment system in large Japanese companies excludes most women, who typically leave upon marriage or the birth of their first child, as well as 'temporary' workers (Clark, 1979). It thus applies to male high school and university graduates who work until they are 55 or 60 for the same employer and are effectively locked into its fate through seniority-based reward systems, bonus payments, corporate welfare programmes and the high social status associated with membership of a large and successful company. Movement to another firm is usually regarded as a sign of failure and disloyalty, and can often result in large salary reductions as well as loss of ancillary benefits.

This high level of dependence upon the 'permanent' core workforce of large Japanese firms means that managers rely on skill flexibility and intra-firm mobility between tasks and roles to deal with economic change and have to generate considerable commitment to corporate objectives among employees (Aoki, 1988, pp. 50-71). Since enforced redundancy of core employees is often taken as a sign of managerial incompetence among large employers, and sometimes a cause for top managers to resign (Dore, 1986, pp. 88-119), competitive pressures require extensive on-the-job retraining and skill development, together with frequent job changes and, sometimes, mobility between work sites. The high rate of successful technological change in Japanese factories reflects, and requires, worker acceptance of the need for role flexibility which, in turn, is based on relatively high levels of mutual trust (Abegglen and Stalk, 1985, pp. 130-2), in addition to increasing competition for promotion (Koike, 1994; Odagiri, 1992, pp. 50-76). Because employment for the core workforce is a long-term commitment on both sides of the labour 'contract', high levels of flexibility and willingness to change tasks are easier to develop than in firms where the dominant reaction to reductions in demand is enforced redundancies. Essentially, the rigidity of external labour markets for large Japanese firms is counterbalanced by the considerable internal flexibility of employees, while the reverse tends to be the case in the Anglo-American economies.

An important characteristic of the post-war Japanese employment system which encourages employee commitment is the organization of unions on an enterprise basis. Although craft-based and industry-based unions had developed during the process of industrialization in Japan (Dore, 1973, pp. 388–403), the dominant pattern in many industries today is one of enterprise unions. Supported by the state and banks, many large firms established these 'second' unions in the early 1950s as a replacement for the more radical industry-wide unions which had developed during the US occupation (Cusumano, 1985, pp. 149–74; Dore, 1973, pp. 327–9), sometimes after a bitter strike such as that at Nissan in 1953.

In general, the level of unionization of the workforce increases with firm size and unions form an integral part of the managerial system in larger firms. Senior union officials sometimes come from the ranks of middle management and, in some firms, taking on such roles is seen as a useful career move. As Cusumano puts it: 'the union's labour affairs department and the company's personnel department [at Nissan] coordinated their activities to the point where, to an outside observer, they seemed almost indistinguishable' (1985, p. 171). Union leaders had considerable influence on managerial promotions and managers, in turn, ensured that the union was supported by the workforce. However, Koike (1987) points out that not all enterprise unions are simply creatures of top management in Japan, and strikes and disputes are quite common. Overall, although many unions are members of industry federations which negotiate on annual basic wage increases, there seems little doubt that the enterprise unions are the most important representative unit in the regulation of work and the total reward system and their interests are more tied to the growth and success of the individual firm than are craft- or industry-based unions elsewhere.

The characteristics of the post-war Japanese business system are summarized in Box 9.3.

# Box 9.3 Characteristics of the post-war Japanese business system

# The nature of economic agents

- 1 Diversification limited to technologically and/or market related activities.
- 2 Low rate of change of core activities and capabilities.
- 3 Low involvement of shareholders in management, but substantial corporate cross share holdings to cement business relationships and limit take-over threats.
- 4 Growth preferred to short-term profits, risks shared with business partners.

# Market organization

- 1 High degree of vertical co-ordination through customer-supplier networks, trading companies and banks.
- 2 Strong intra- and inter-industry collaboration through trade associations, intermarket groups, etc.

Work co-ordination and control

- 1 Strong role of middle management limits centralization in practice.
- 2 High employer-male employee interdependence in large firms.
- 3 Low manager-male employee differentiation.
- 4 Considerable work group discretion over work processes with supervisor involvement.
- 5 Low individual specialization of tasks and roles.

# 9.5 The Korean business system

# The nature of economic agents

The dominant economic actors in post-Second World War Korea are the fast growing, diversified 'financial cliques', or *chaebol* (see Chapter 6). They dominate the heavy manufacturing and chemical industries sector, with most fields of economic activity being divided between three or four *chaebol*. Even more striking is the domination of the largest five and ten *chaebol*. In 1991, the sales of the top five accounted for just under 50 per cent of Korea's GNP (Fields, 1995, p. 7). These also exhibited the greatest growth and diversification rates in the 1970s. As well as dominating many manufacturing industries, the *chaebol* also dominate significant parts of the service sector. In addition, the construction industry has become a favoured route to diversification and many *chaebol* are also active in transport services, insurance and related financial services.

Finally, seven large general trading companies which are members of the largest ten *chaebol* have come to dominate Korea's export trade.

These large conglomerates are recent creations. Seventeen of the largest 20 *chaebol* in 1986 were founded after the end of the Japanese colonial period in 1945, and seven were established between 1955 and 1967 (Fields, 1995, p. 33). Furthermore, even when the original company was a pre-1945 product, most *chaebol* only developed into diversified firms in the 1960s and 1970s (Kim, 1991). This rapid expansion and diversification was largely directed and underwritten by the state, especially the massive shift into the heavy manufacturing and chemical industries in the 1970s which transformed the *chaebol* into widely diversified conglomerates (Woo, 1991, pp. 128–46). As a result, by 1983 'the average *chaebol* had firms operating in five different manufacturing industries' and a 'quarter of its manufacturing workforce employed in industries altogether unrelated to its primary manufacturing activity' (Zeile, 1991, p. 307). Furthermore, the top *chaebol* were even more diversified, with their average diversification indices being 'one and a half to two times as large as the averages for the top fifty *chaebol*' (ibid.).

The dominant firms in the Korean economy, then, are fast growing and highly diversified, with an ability to move into totally new fields of activity quickly and successfully under state direction and patronage. It is important to note, though, that the *chaebol* are usually quite vertically integrated, and horizontal diversification typically came after such integration, so that they are not identical to the unrelatedly diversified conglomerates which developed in many Anglo-American economies in the 1980s.

The high level of diversity of economic resources and activities controlled by the *chaebol* since the 1960s provides considerable self-sufficiency and, in general, they perform many more activities 'in-house' rather than relying on subcontractors like the Japanese. For example, the Hyundai Motor Company subcontracts only 40 per cent of the value added in its cars, and most of that is to firms within the Hyundai group (Amsden, 1989, p. 184). However, this self-sufficiency refers only to relations with other firms. As already pointed out, the *chaebol* became established under the aegis of the Korean state and remain highly dependent on state supplied and subsidized credit (Woo, 1991, pp. 148–75). They have demonstrated this dependence by responding rapidly to state policies promoting new industries and, in particular, exports (Kim, 1991). Thus the *chaebol* are not as isolated from state agencies as the large diversified Anglo-American corporations and are able to rely on the state to share investment risks.

The *chaebol* remain largely family owned and controlled, despite their rapid growth and state pressure to sell shares on the stock market (Kim, 1991). Most of their expansion was funded by state subsidized debt, and so did not dilute family share holdings, and where shares have been publicly issued many were bought back by family-controlled subsidies (Orru *et al.*, 1991; Woo, 1991, p. 175). Family ownership here continues to mean largely family control, with most of the leading posts held by family members and/or trusted colleagues from the same region or high school as the founding entrepreneur (Janelli, 1993; Kim, 1991). Ownership and control are thus closely linked in the Korean *chaebol*.

As already indicated, the *chaebol* have grown extremely quickly, and usually at the expense of profitability. Although there are considerable difficulties in relying on the public financial statements of Korean companies (Janelli, 1993, pp. 124–9), detailed analyses of the *chaebol* suggest that 'the objective of the firms of the large *chaebol* is not

to maximize profits but to maximize sales' (quoted in Janelli, 1993, p. 91). Janelli's own analysis of the *chaebol* he studied suggested that 'since 1970, its profits have barely kept pace with its dividends' (1993, p. 94). This is because ownership rights are held for control purposes more than for income, and growth has been financed by state provided and subsidized credit rather than out of retained profits. Also critical, of course, is the extra influence with the state that large size can provide and the fierce competition between the *chaebol* for being the largest enterprise group (Kim, 1991).

# Market organization

The large size and self-sufficiency of the Korean *chaebol* mean that they exhibit low interdependence with suppliers and customers and are able to dominate small and medium sized firms. Because their dependence on suppliers is lower than in Japan, managers do not have to be so concerned about linkages with them. However, transactions between enterprises in Korea are more personal and particularistic than they are in the USA, and formal details of contracts appear to be less critical in determining a firm's behaviour when unforeseen circumstances arise. According to Janelli's respondents, reputations for fair dealing and showing sensitivity to the needs and circumstances of exchange partners were important aspects of market relations, although it is not clear how much this attitude affected dealings with small firms as opposed to those with other large enterprises (1993, pp. 187–92). In general, legal details are less significant in governing inter-firm transactions in Korea than in the USA, but probably more so than in Chinese communities (Kao, 1991; Silin, 1976).

Relations between the *chaebol* tend to be fiercely competitive, with considerable reluctance to co-operate over joint projects such as complementary R&D programmes (Wade, 1990, pp. 315–16). New industries are often the site of intense competition for dominance, and the major driving force behind many new investments often appears to be corporate rivalry for the leading position, as in the recent expansion of the petrochemical industry. This high degree of competition between the leading *chaebol*, which has been fuelled by the state's policy of selecting entrants to new industries and opportunities on the basis of competitive success, has severely limited the development of sector-based organizations in Korea (Amsden, 1989, pp. 64–76; Wade, 1990). With the exception of the cotton spinning industry in the 1950s and 1960s, there have been few if any industry-wide trade associations or similar bodies promoting co-operation between firms and collectively lobbying the state. Additionally, since the dominant economic actors are highly diversified, their interests in any one sector are less important than they would be in economies where firms are more focused on particular industries, and so the incentives to develop strong sector-based collectivities are correspondingly lower.

# Work co-ordination and control

The significance of family ownership and control of the *chaebol* is reflected in the importance of personal authority and the patriarchal management style of the chief executive. In the largest *chaebol*, especially those concentrated in heavy manufacturing industries, there is greater emphasis on formal rules and procedures, but personal discretion over how these are interpreted and applied remains high and it is clear that

authority is much more personally focused than in large Japanese corporations (Bae, 1987; Chung *et al.*, 1988). An example of how formal procedures are often less significant than they might at first appear is given by Bae's account of recruitment at Hyundai's car factory in Ulsan (1987, pp. 44–9). Although there was a formal requirement for new workers to pass an entrance examination, Bae concludes by saying: 'Despite everything, recruitment of workers at Hyundai is based more on recommendation than on any other consideration' (1987, p. 48). Similar practices were observed by Kim (1991) and Janelli (1993) in their studies of large firms, and the use of personal recommendations seems to have become even more marked since the severe labour unrest of the late 1980s, when university students and others joined the *chaebol* as manual workers and became involved in organizing protests and strikes.

The importance of personal authority and avowal of a paternalistic ideology are accompanied by a largely authoritarian, not to say militaristic, management style. Generally, the Korean management system is Ocharacterized by top-down decision making, enforcement of vertical hierarchical relationships, low levels of consultation with subordinates, and low levels of trust, both horizontally and vertically (Bae, 1987; Chung *et al.*, 1988). Superiors tend to be seen as remote and less interested in subordinates' concerns or their ability to contribute than in their obedience. As Janelli puts it: 'subordinates advanced the view that the company was like the army' (1993, p. 223) and 'in many ways my military experience served as a better guide to behaviour in the office than my understanding of American bureaucracies or South Korean villages and universities' (ibid., p. 226). Similarly, Kim describes relations between supervisors and subordinates as 'formal, distant, and authoritarian' (1991, p. 150) and cites some of the workers in Poongsan Corporation as viewing hierarchical relations as being worse than in the military.

This remoteness and emphasis on personal authority are echoed by the highly centralized nature of decision making and co-ordination of diverse activities in the Korean *chaebol*. Major decisions are taken personally by the chairman of each group and the top managers of subsidiary companies are clearly subordinated to him. In particular, all senior management positions in the entire group of firms are controlled by the chairman and he often moves managers between subsidiaries. As Amsden puts it: The personnel function in general, and the labour relations function in particular, are almost non-existent as a staff responsibility, whether at the group or subsidiary level' (1989, p. 325). Managerial careers and rewards are thus directly controlled by the owner and his family, and depend greatly upon personal relationships with them.

The high level of centralization encourages considerable integration of economic activities, as capital, technology and personnel can be centrally allocated and moved between subsidiaries (Kim, 1991). An important role in this central direction and coordination of the *chaebol* is played by the secretariat, or planning group, which is the key corporate staff unit under the direct control of the chairman. This group allocates and analyses information from subsidiaries, and elsewhere, evaluates performance and prepares proposals for new ventures and shifting resources. It is largely responsible for co-ordinating plans and mobilizing resources across the whole *chaebol*, so that these diversified conglomerates are in fact managed as cohesive economic entities with a unified group culture focused on the *chaebol* owner. In this respect they are quite different from those Anglo-American conglomerate holding companies in which the

central office functions more like a bank than as an integrated planning and control agency, and subsidiaries are simply set annual financial targets as separate, discrete entities.

Turning to consider employment policies, the level of employer-employee commitment is limited for most manual workers in the *chaebol*. Although seniority does appear to be important in affecting wage rates, and employers do provide accommodation and other fringe benefits in the newer industries, Korean firms are reluctant to make the sorts of long-term commitments to their workforce that many large Japanese ones do (Amsden, 1989; Bae, 1987). Mobility between firms, both enforced and voluntary, is considerably greater for manual workers—and some non-manual—than is common in Japan. Annual labour turnover rates of between 52 per cent and 72 per cent were quite usual in the 1970s in Korea, and were especially high in manufacturing industries (Michell, 1988, p. 109).

Additionally, leading firms in Korea sometimes poach skilled workers from competitors rather than invest in training programmes (Amsden, 1989, pp. 275–87; Janelli, 1993, p. 139). Even where workers do not leave very often, this is more because they are locked into their current employer through high levels of overtime pay than because they feel committed to the firm (Bae, 1987), as was demonstrated by the wave of strikes and protests in 1987 and 1988 (Wilkinson, 1994). White collar employees are more favoured and tend to remain with large employers, not least because their pay and conditions are usually substantially better than they could obtain by moving (Janelli, 1993, p. 153).

Personnel and reward policies are highly segmented in Korea, especially by gender, education and sector of employment. Women are paid around half of the average male wage and typically work in light, labour-intensive industries such as textiles, apparel, rubber and word processing (Amsden, 1989; Deyo, 1989; Michell, 1988). Men are more concentrated in heavy industry, where they tend to be more highly skilled and where employers take more pains to keep them. Relatedly, the level of formal education strongly affects wage rates, with college graduates being paid around three times the wages of primary school graduates and one and a half times as much as high school graduates. It is very difficult to become a manager in the leading *chaebol* without a college degree, and the differential between managers and production workers is over twice as great in Korea as in the USA (Amsden, 1989, pp. 230–1). Employers have thus been able to follow quite different labour management policies according to market conditions, especially in the export-oriented manufacturing sector (Deyo, 1989).

Finally, the organization and control of tasks in the *chaebol* varies between sections of the workforce. The specialization of roles, skills and authority, in particular, is greater for unskilled manual workers than for other employees. According to Bae, 90 per cent of the manual workforce at Hyundai Motors remained at their initial job level and were not systematically upgraded (1987, p. 60). Unskilled workers continued to carry out relatively narrow tasks without much movement between jobs and skill categories. Similarly, Kim found a considerable degree of worker specialization at Poongsan Corporation, with only 20 per cent having changed jobs more than once, and then nearly always in the same speciality (1991, p. 209). However, more skilled and non-manual workers do appear to be moved between tasks and sections, and sometimes develop more varied skills, in the larger and more diversified *chaebol*. Because of the importance of

personal authority in the Korean *chaebol*, jobs and responsibilities are more determined by superiors' wishes than by formal rules. Supervisory discretion means that the division of labour is less formally prescribed in the *chaebol* than in more rule-governed work systems. Overall, then, the degree of specialization is mixed, combining task flexibility and variety for some employees with considerable rigidity and narrowness for unskilled manual workers.

This emphasis on personal authority and control, together with the prevalent ideology of paternalism in most Korean companies, encourages close supervision of task performance. Frequently, the physical layout of office furniture and work space in general is carefully arranged so as to maximize supervisor surveillance of work processes (Janelli, 1993, p. 164). Section chiefs and department heads also ensure control by narrowly circumscribing subordinates' tasks and carefully monitoring all messages coming into the work unit and work outputs. Subordinates in general were seen as children, needing firm guidance and direction. As mentioned above, these aspects of the *chaebol* were also related to considerable social distance between superiors and subordinates and permitted considerable personal discretion to supervisors in evaluating the performance and worth of their juniors.

The characteristics of the Korean business system are summarized in Box 9.4.

# Box 9.4 Characteristics of the post-war Korean business system

# The nature of economic agents

1 High diversification of activities within and across industries.

- 2 Considerable change in core activities, from light industry in the 1950s to heavy industry in the 1980s, as well as to service sector activities.
- 3 Strong owner involvement in, and control of, management.
- 4 Growth goals dominate, with risks shared with the state and state-controlled banks.

# Market organization

- 1 Weak co-ordination of inputs and outputs outside chaebol boundaries.
- 2 Low horizontal collaboration outside *chaebol* boundaries.

# Work co-ordination and control

- 1 High degree of centralization.
- 2 Low employer-manual worker interdependence and commitment.
- 3 High manager-manual worker differentiation and distance.
- 4 Low work-group autonomy and discretion; strong managerial control of work processes.
- 5 Specialized division of labour for unskilled and semi-skilled manual workers; flexibility expected of skilled and non-manual workers.

#### 9.6 The Taiwanese business system

#### The nature of economic agents

In Taiwan there are two major types of enterprise: the large state owned firms, which are mostly in the capital-intensive sector, and the small to medium sized family owned and controlled firms in the export sector. As Wade points out: 'From the early 1950s onward Taiwan has had one of the biggest public enterprise sectors outside the communist bloc and Sub-Saharan Africa' (1990, p. 176), and public enterprises contributed about twice as much to GDP as their equivalents in Korea in the 1970s. Similarly, their share of gross fixed capital formation was well over 30 per cent in that decade when Korea's was under a quarter and Japan's under an eighth. Indeed, the only Asian countries with a comparable public sector contribution to capital investment were India and Burma. In 1980 the Taiwanese Ministry of Economic Affairs owned firms in the power, petroleum, mining, aluminium, phosphates, alkali, sugar, chemicals, fertilizers, petrochemicals, steel, shipbuilding, engineering and machinery industries, while the Ministry of Finance owned four banks and eight insurance companies (Wade, 1990, p. 178). These public enterprises were very large by comparison with privately owned ones and often dominated, if not monopolized, their sectors. Thus the state has retained ownership and control of the 'commanding heights' of the economy in Taiwan, especially the upstream capital intensive sectors.

The private sector does have some large enterprises, often established with state support, such as Nanya Plastics, but it is much less concentrated than its Korean equivalent. In 1990 those in the private sector employing under 300 people employed 70 per cent of the labour force and produced 60 per cent of Taiwan's exports (Fields, 1995, p. 64). Furthermore, while the number of manufacturing firms increased 250 per cent between 1966 and 1976, the number of employees per firm grew by only 29 per cent, in strong contrast to Korea where the employment total per firm doubled in the same period. These smaller companies dominate the export trade; those with under 300 employees accounted for 65 per cent of manufactured exports in 1985 (Wade, 1990, p. 70). It should, however, be borne in mind that many of these smaller firms are members of socially linked business groups, of which the largest 100 generated sales amounting to 34 per cent of Taiwan's GNP, and in 1991 the ten largest employed 12.6 per cent of the total workforce (Fields, 1995, pp. 64-5; Hamilton and Biggart, 1988; Hamilton and Kao, 1990). These groups are not, though, nearly as tightly integrated and controlled as the Korean chaebol, but are more akin to personally connected networks. Furthermore, many larger private manufacturing enterprises remain single unit firms and do not form business groups. Overall, then, the Taiwanese economy is dualistic, with the large state enterprises dominating the inputs of the many smaller firms in the export sector.

Vertical integration is weak in privately-owned Taiwanese firms and they are rarely self-sufficient in terms of combining the management of key processes and activities in one organization. Instead, they are usually highly interdependent with other enterprises for inputs and for distributing their outputs, and form fluid sub-contracting networks (Hamilton and Kao, 1990; Redding, 1990). However, this interdependence is not usually accompanied by a willingness to share long-term risks with suppliers and buyers. Instead, more restricted and limited connections are preferred.

Diversification of a horizontal nature—'opportunistic' in Hamilton and Kao's (1990) terms—is, though, more widespread in private firms, especially those forming business groups of associated companies. While by no means all successful firms develop into highly diversified business groups, including some of the largest, those that do diversify tend to move into a variety of sectors in a seemingly *ad hoc* and idiosyncratic way, often as the result of personal requests or obligations. Additionally, they appear able to switch products and markets quite readily in response to changing patterns of demand (Redding, 1990).

An important point about such diversification and growth is that it is not usually associated with managerial integration, but rather is achieved by setting up legally separate firms linked through common ownership and family management. Diversification is thus informal and, typically, personal. Each firm, in the sense of a formal management structure, tends to be quite specialized in its resources and spheres of economic activities, but families may well invest in and control a variety of businesses. While leading managers of each firm in a business group may well be members of the same family, they are not usually structured into a distinct hierarchy of authority relations. Overall, then, managerial diversification is limited in Chinese family businesses (CFBs), although entrepreneurial diversification can be much greater.

These firms are nearly all owned and controlled by families, as indeed are most Chinese businesses throughout South-East Asia (Redding, 1990). Owners are highly involved in the running of their firms and there are strong connections between ownership and direction of economic activities. Wong (1988, pp. 170–2) has characterized the pervasive economic ethos generating this strong drive towards autonomy and proprietorship in Chinese communities as 'entrepreneurial familism' which has led to high rates of new firm formation in Taiwan, Hong Kong and elsewhere as families seek to be in control of their own businesses (see Gates, 1987; Greenhalgh, 1984). This emphasis on family ownership and control means that the dominant goal is the acquisition and growth of family wealth rather than the growth of the firm as a separate entity. The pursuit of large size irrespective of profitability is not the dominant objective in these firms, especially if it could lead to the loss of personal control or to being considered a threat to the interests of the family firm. It also discourages risk sharing and long-term mutual commitments between firms.

# Market organization

The specialization and interdependence of Taiwanese family businesses mean that they have to rely on each other to obtain inputs for their products and services and to distribute and market them. Thus multiple market connections between firms are crucial to their operation. However, these are not necessarily long term or based on mutual obligations. Rather, inter-firm links are often managed in such a way as to reduce risks, and so commitments to other economic actors are restricted. Exchange partners may, then, be numerous and selected on the basis of their personal reputations for competence and reliability, but do not usually form networks of long-term trust and reciprocal loyalty (Redding, 1990). Subcontracting, for instance, may involve a number of firms without any long-term commitment to continued orders being implied. Similarly, trading companies may select a particular supplier to fulfil an order without any obligation to do

so for successive orders. Market relations can thus change rapidly and are quite fluid, flexibility being emphasized over long-term risk sharing (Hamilton and Kao, 1990).

Business partnerships, on the other hand, often do involve long-term reciprocal commitments and can lead to the development of elaborate networks of personal obligation which structure strategic decisions and new ventures (Kao, 1991). Where significant resources are involved and firms need to undertake activities jointly, connections are highly personal and dependent on trust between the owners. Without high levels of personal trust, such partnerships cannot be formed successfully in Taiwan and, as a result, many medium sized firms do not grow into large enterprises because they are unable to find partners they can rely on. While straightforward trading relationships, then, are quite limited in their mutual commitment, more substantial alliances and joint activities involve considerable personal obligations, often on a long-term basis. Even here, though, flexibility is valued, and families will often prefer to establish new ventures with a number of different partners rather than expand existing ones with their current associates (Hamilton and Kao, 1990).

Inter-firm connections are, then, highly personal, and form extensive networks of mutual, albeit often asymmetric, obligations, without developing into the sorts of long-term, wide-ranging inter-organizational linkages found in the Japanese inter-market groups (Hamilton and Biggart, 1988; Orru *et al.*, 1991). Indeed, when the Taiwanese state did try to encourage the formation of Japanese style sub-contracting arrangements, they failed (Hamilton and Biggart, 1988). Equally, attempts to establish trading companies as long-term co-ordinating agencies in Taiwan have been less successful than in Korea, and there are few major intermediaries performing similar functions to the Japanese *sogo shosha* or German banks (Wade, 1990, pp. 160–5).

Sectoral organizations are also limited by this concern with personal control, as well as being restricted by the state's intolerance of independent intermediary organizations, particularly those dominated by Taiwanese (Wade, 1990). Additionally, given the flexibility of most CFBs and their unwillingness to commit major resources to any single industry or activity for a long time, stable associations of industry-specific enterprises are difficult to maintain. However, the high population density in Taiwan—and of course in Hong Kong too—combined with considerable cultural homogeneity among the Taiwanese, has encouraged a strong reliance on reputation as the primary means of ensuring compliance with exchange commitments in the absence of an independent and reliable legal system (Gates, 1987; Greenhalgh, 1984; Silin, 1976).

#### Work co-ordination and control

One of the most significant characteristics of CFBs is the highly personal nature of authority relationships and the concentration of authority in the paternalistic owner (Deyo, 1989; Kao, 1991; Redding, 1990; Silin, 1976). Formal rules and procedures do exist in the larger firms, but these are less important in controlling activities than the personal decisions of owners, and formal status in the managerial hierarchy is often less significant than a manager's personal ties to the owner. As Silin suggests, 'the primary goal of the executive is then to secure and increase the respect of the boss, that is, to increase personal interaction with him' (1976, p. 67). Despite the growing tendency for owner-managers to have been educated in Western universities and business schools, and

increasing claims that they follow 'modern' management methods, there is little evidence to suggest that the highly personal nature of authority in Chinese firms has diminished (Redding, 1990).

As well as being personal, authority is also quite remote and distant in Chinese businesses. According to Silin, leadership is typically seen as a moral quality of individuals rather than as a technical skill which can be modified or taught (1976, pp. 57–66). This emphasis on moral superiority means that the owner is sharply distinguished from employees, including managers, and is the source of the business 'wisdom' which guides decisions. As a result, the morally inferior cannot legitimately question his choices and are rarely called upon to contribute to the decision-making process. Centralization is thus high in CFBs, especially for financial, marketing and personnel decisions, which are usually made by the owner (Redding, 1990, pp. 174–5). Even when there is a formal hierarchy, middle managers are often bypassed because of a lack of trust in formal accounting systems and subordinates and a strong preference for personal contact (Silin, 1976, pp. 74–85).

This centralization of control within firms ensures that activities are tightly integrated, but the diverse and often unconnected nature of activities in many business groups in Taiwan mean that their formal co-ordination is weak. However, the strong personal ties between subsidiary heads and the owning family—often the same people—enable close collaboration and control to be maintained. This is especially true for entrepreneurial as distinct from operational management issues. While, then, the actual operations of each sub-unit may not be very closely integrated—and certainly not as much as in the Korean *chaebol* or Japanese corporations—they are by no means run as discrete entities under the umbrella of a relatively remote holding company. Rather, key resources may be switched between them and strategic choices are made by the group owner through direct personal links.

The personal nature of authority relations in Chinese family firms and the strong ideology of paternalism encourage considerable expectations of reciprocity in employment policies and practices. However, these are mitigated when there is little prior basis for personal commitments and employees were previously strangers. In practice, obligations become more attenuated as connections become more distant from the basic family unit. Thus, the strongest ties and sense of commitment occur between family members, somewhat weaker ones between schoolmates, neighbours and more distant kin, and the weakest ones between those who were strangers before employment. Long-term commitments and seniority-based promotion practices tend to be reserved for those workers with strong personal ties to the owner, while previously unknown staff hired through impersonal channels neither expect nor receive such commitments. In particular, young, female, semi-skilled, non-family workers in the light manufacturing export sector are expected to stay for a short time only and are rarely trained for more demanding posts (Deyo, 1989; Gates, 1987). As in Korea, women in general tend to be worse off than men, and white-collar staff are better treated than many manual workers, though these differences do not seem so rigid and sharply institutionalized in Taiwanese firms (Gates, 1987; Silin, 1976). In general, the segmentation of labour markets by gender and qualification is less strong here than in Korea and Japan (see Chapter 5).

Because senior managerial posts are restricted to family members or those who have family-like connections to the owner, many skilled workers and managers prefer to leave and start up their own businesses once they have acquired business skills and some capital (Greenhalgh, 1984; Wong, 1988). Both the general cultural preference for personal business ownership over employment, and the unwillingness to trust non-family subordinates on the part of employers, limit the scope and length of employer-employee commitments in the Chinese family business.

With regard to task structure and control, the importance of personal relationships and authority in CFBs means that formal specification of roles and positions is less important than in most Western societies. Equally, jobs and skills are not rigidly defined and separated by formal procedures, but are fairly broad and flexible. In the large Taiwanese firm studied by Silin (1976), roles were fluidly defined, many managers held multiple positions and their responsibilities were liable to be changed suddenly at the behest of the owner. Similarly, many managers in Taiwanese business groups hold a considerable number of posts and are rarely restricted to a single specialized role (Hamilton and Kao, 1990). Overall, then, role and authority specialization are low in these firms.

The strong commitment to patriarchal relationships in the workplace, and in society as a whole, means that superior-subordinate relations are quite remote and distant, particularly those between the owner-manager and employees (Redding, 1990; Silin, 1976). Similarly, as in Korea, paternalism implies a lack of confidence in the abilities and commitment of staff, so that close supervision of work performance is a feature of Taiwanese firms, as is considerable personal discretion in how authority is exercised, especially at the top of the enterprise.

The characteristics of private Taiwanese businesses are summarized in Box 9.5.

# Box 9.5 Characteristics of the post-war Taiwanese business system

# The nature of economic agents

- 1 Specialized organizations combined with diversified family-owned businesses.
- 2 Considerable changes in family business activities.
- 3 Strong owner involvement in, and control of, management.
- 4 Profitability and growth important; opportunistic diversification in family businesses; risks managed through limiting commitments and high degree of flexibility.

# Market organization

- 1 Low vertical co-ordination.
- 2 Low horizontal collaboration except in personally linked partnerships and business groups.

# Work co-ordination and control

- 1 High degree of centralization.
- 2 Low employer-manual worker interdependence and commitment beyond familylike personal attachments.
- 3 High manager-manual worker distance and owning family-manager separation.

# 4 Low task and role specialization except for female semi-skilled workers.5 Specialized division of labour for unskilled and semi-skilled manual workers; flexibility expected of skilled and non-manual workers.

# 9.7 Overall comparisons

As Box 9.6 shows, many of the characteristics of the business systems in Japan, Korea and Taiwan vary quite sharply, as well as being distinct from those of Anglo-American business systems as summarized in Box 9.2. In particular, the Japanese business system differs from those of Korea and Taiwan in a number of dimensions. The latter differ most significantly in the nature of dominant firms, while sharing many characteristics in the work co-ordination and control area. The distinctiveness of the Japanese

Characteristics	Business systems Business system				
	Japan	Korea	Taiwan	Anglo-American	
The nature of economic agents		·			
1 Diversity of activities	limited	high	low except for opportunistic	high	
2 Rate of change of activities	low	high	considerable	considerable	
3 Owner involvement and control	low	high	high	low	
4 Risk and growth strategies	risk-sharing with partners	diversification and state	flexibility, opportunism	diversification, growth through	
Market organization					
1 Extent of vertical co- ordination	high	low	low	low	
2 Extent of horizontal co-operation between	high	low	low except for personal	low	
Work co-ordination and control					
1 Centralization of decision making and	low in practice	high	high	medium	

**Box 9.6** Summary of characteristics of post-war East-Asian and Anglo-American business systems

decision making and control	practice			
2 Extent and scope of employer-employee interdependence and commitment	high	low	low	low
3 Manager-worker distance and differentiation	limited	high	high	medium
4 Work-group task discretion	considerable	low	limited	limited
5 Specialization of tasks, roles and skills	low	low except for non-skilled	low	high

system results partly from its pre-industrial legacy and the associated pattern of industrialization, and partly from the institutional changes made during the US occupation of 1945–52. Many of the common characteristics found in Korea and Taiwan similarly stem from particular features of preindustrial China and Korea, together with the consequences of Japanese colonial rule over Korea and Taiwan during the first half of the twentieth century and the authoritarian states that directed industrialization in both economies (Whitley, 1992a).

# 9.8 Changes in East-Asian business systems: globalization and institutional developments

In the past decade or so, a number of significant changes have occurred in the Asia-Pacific which could modify some of these characteristics of East-Asian business systems. Three are especially important. First, the interdependence of national economies and firms has grown dramatically as Japanese, and then Korean and Taiwanese, outward investment to other economies, particularly China, increased in the 1980s and 1990s. Second, as these economies have industrialized and developed, their complexity and internal differentiation have increased considerably and growth rates, particularly in Japan, have levelled off. Third, domestic institutions, especially political systems, have changed and this has affected state-business relations. After a brief discussion of the first of these three, this section considers the impact of these phenomena on the Japanese, Korean and Taiwanese business systems.

# International economic interdependencies

The recent and current increase in intra-regional trade and investment in the Asia-Pacific is a major component of what many observers see as the globalization of economic activity at the end of the twentieth century. Not only has trade between industrialized economies grown considerably, but more of it is being managed within corporate boundaries and co-ordinated across national borders. As national firms internationalize their operations, we might expect them to become more detached from domestic institutions and agencies, especially the state, and develop more 'international' characteristics which then modify domestic business systems. In this view, the relatively homogenous and distinctive business systems of Japan, Korea and Taiwan should change towards more standardized forms of economic organization.

However, it is important to note that this growth in outward investment from these three economies has not been matched by a correspondingly large inflow of foreign direct investment, or of large firms seeking to raise the bulk of their new investment from international capital markets. Thus, the impact of foreign firms and external ways of organizing economic activities on Japan, Korea and Taiwan has been much less than the impact of firms from these three countries on the other economies of the Asia-Pacific. Indeed, the reproduction of many aspects of inter-firm relations by Japanese companies in foreign locations, as well as some labour management practices, suggests that internationalization here means more the externalization and generalization of Japanese ways of doing business to other societies than the transformation of their domestic economies by foreign patterns of economic organization. Similarly, the restructuring of 'global commodity chains' (Gereffi, 1996; see also Whitley, 1996) has not yet resulted in major changes to the nature and organization of economic agents in Korea and Taiwan (see also Chapter 14).

Additionally, the increasing complexity of domestic economies, and slowing of their growth rates, might be expected to modify the ability of the state to manage economic development and, in particular, to direct the activities of large companies. In the cases of Korea and Taiwan, the role of the state in the economy may become more like the post-war Japanese co-ordinating agency than the more directive and authoritarian agency of most of the post-war period. This tendency is likely to be enhanced by the recent democratization and liberalization of the state in these countries, although the geopolitical context and the traditionally high prestige of the state in these societies will limit the reduction of the state's role in managing economic and social development. Insofar as the state continues to become more democratic and open to political competition in Korea and Taiwan, business dependence on the state will probably decline (Kim, 1991), but this does not mean that state-business relations are becoming more Anglo-American. The state remains very powerful in these societies, and firms are unlikely to be able to ignore state policies or agents in their strategic choices.

#### Changes in Japan

Despite the growth in Japan of foreign portfolio investment and, in some cases, direct foreign acquisition of substantial share holdings in Japanese firms, the extensive networks of reciprocal share holdings and broad business relationships between banks, trust companies, insurance companies and inter-market group members remain important features of the Japanese economy, and share ownership between firms is more often an expression of long-term business commitments than a narrowly focused financial connection. Moreover, substantial foreign share ownership does not confer control, and senior managers continue to enjoy high levels of autonomy from beneficial owners and are not subject to a market for corporate control. Despite low capital gains and dividend

payouts in the recession, insurance companies and other large shareholders do not seem to have required radical changes in financial policies or managerial personnel.

Equally, the growth of outward foreign direct investment (FDI) does not seem to have greatly weakened inter-firm ties within Japan. While the inter-market inheritors of the zaibatsu may have declined in significance relative to the vertically quasi-integrated keiretsu as new industries have developed (see Chapter 6), and the role of the general trading companies has changed to include third party transactions, the overall importance of business groups and particularistic, obligational links between firms does not appear to have declined (Gerlach, 1992; Sako, 1992; Westney, 1996). As large companies have relocated some of their plants to China and elsewhere in the Asia-Pacific, many suppliers have followed to maintain established ties. The comparatively long recession in the 1990s and rise of the yen since 1985 have likewise encouraged considerable restructuring of many large firms and relocation of many plants to China and elsewhere. Most of the labour surplus resulting from the 1980s expansion and its reverse has, however, been managed by established procedures such as 'lending' staff to subsidiaries and subcontractors, reducing bonus payments and 'temporary' workers, encouraging early retirements, ceasing graduate recruitment and changing jobs rather than engaging in widespread compulsory redundancies. Long-term interdependencies between large employers and their core workforce seem to have survived the early 1990s, and established patterns of work co-ordination and control also do not appear to have changed greatly (Koike, 1994). Given the systemic nature of the post-war Japanese business system, and the limited degree of institutional change that has occurred in the past 20 or so years, this is perhaps not too surprising.

Overall, then, the diversity of institutional contexts of Japanese FDI, the dominant position of many firms in their sectors, often on a worldwide basis, continued commitment to, and dependence on, the Japanese economy and institutions, together with incremental and limited institutional change in Japan, have limited the extent of change in the nature of firms and markets there. A few characteristics of the post-war Japanese business system have been modified, such as the close interdependence between main banks and their major clients and the co-ordination role of the sogo shosha. But as successful firms became cash rich and integrated forward into distribution and marketing, substantial changes of the key dimensions discussed above have not occurred. Furthermore, they are unlikely to as long as leading firms remain tied to the Japanese economy and its institutions—assuming these latter do not change radically—and to regional economies that they can dominate. Incremental change seems much more probable than discontinuous shifts in firm type and priorities, and certainly a move to Anglo-American forms of economic organization is most unlikely given the limited influence of international capital markets in Japan, the low rate of US and UK FDI into Japan and the success of many Japanese firms in exporting many aspects of their form of capitalism.

#### Changes in Korea

In the case of Korea, both the extent of internationalization of the *chaebol* activities and of foreign influence on the Korean economy are less than in Japan. However, the move to a civilian presidential regime, greater toleration of opposition political parties and

independent trade unions, as well as growing international pressure against overt state financial support of the *chaebol* and bureaucratic regulation of economic activities, represent considerable changes in the institutional context of the Korean business system which are likely to affect the ways in which the *chaebol* develop. As Woo (1991) and Kim (1989) among others have noted, the largest *chaebol* are not so dependent on the state as they were in the 1960s and 1970s, and manifestations of disobedience of state decisions and wishes have become more noticeable in the 1990s, particularly concerning entry into sectors such as cars, petrochemicals and steel and the divestment of owner control. Economic growth, educational expansion and the decline of demographic growth have also resulted in the increase of an educated middle class and a reluctance of many workers to undertake dirty and dangerous jobs at low wages.

The relative liberalization of state direction of the economy and the development of some political competition and of genuine labour representation, together with its greater complexity, seem likely to encourage some decentralization of control within the *chaebol*, as well as the development of horizontal links between them, as the degree of vertical dependence on the state, and the associated overwhelming dominance of political risk, declines. Insofar as diversification was a response to this dependence, it may also decline, as of course the state has requested. However, since overall size remains a crucial feature of *chaebol* success and power, and their growing ability to generate cash reduces their dependence on the state-governed banks, it seems unlikely that they will dispose of substantial subsidiaries, particularly if they think their rivals will acquire them. Relatedly, the strong emphasis on owner control—rather than passive portfolio management—on the part of the owning families in Korea (Janelli, 1993), seems likely to limit internal decentralization, at least as long as trust in formal procedures and institutions remains low.

The rapid expansion of overseas facilities and attempts to compete more on the basis of advanced technologies and innovative products may, of course, encourage less directive and personal managerial practices as organizational complexity grows. However, the *chaebol* remain highly dependent on their home economy and institutions so that the influence of foreign investments remains limited. Additionally, they too are investing quite heavily in China and other parts of the Asia-Pacific, so that they are not dependent on a single type of foreign location with its possibly sharply different institutional context. So far there seems little indication that family owners are willing to hand over the reins of power to 'professional' managers, despite some restructuring of top management and the reduction in size of the chairman's secretariat in some *chaebol*. Similarly, the extent of risk sharing between *chaebol* in new developments remains limited, despite state encouragement in some cases, as does their development of long-term commitments to smaller suppliers and small and medium sized enterprises in general.

Perhaps the most likely area of change in the Korean business system concerns labour management and work structure. Both institutional changes and the increasing complexity of operations and tasks might be expected to encourage some *chaebol* to develop 'Japanese' forms of labour relations and flexibility—as Amsden (1989) claims to have found at the Pohang Iron and Steel Company. As labour markets tighten and workers' skills become more important in improving products and productivity, eliciting commitment from at least the core manual workforce to contribute to corporate goals and

use their initiative in a positive manner could gain priority over earlier managerial objectives (You, 1994). Investment in training and internal career ladders for manual workers can thus be expected to grow, as can the use of bonuses and 'fringe' benefits to retain skilled staff. Whether authoritarian supervision practices and tight control over task performance will change dramatically, given recent Korean history and the continued military threat from the North, remains questionable, but outright repression seems less likely to be effective as a labour management strategy.

An important factor affecting the development of new employment policies and supervisory practices is the role of the unions. In particular, the relative strength of independent unions, and whether they are organized on an enterprise, sector or national basis, will obviously influence owners' and managers' investments in the 'white collarization' of manual workers. The highly political nature of union organization and competition in Korea, especially the state insistence on a company-union structure and unwillingness to recognize the independent Korea Confederation of Trade Unions, suggests that Japanese levels of employer-employee interdependence and commitment are unlikely to become institutionalized in the near future. As long as the major employers rely on the state to manage labour disputes and labour organizations are political as well as bargaining entities, together with the continuing distrust between employers and employees resulting from authoritarian management practices and *chaebol* ties to the military-backed regime, it is difficult to see how radically new forms of labour management are going to develop in Korea.

A further constraint on the incorporation of the male manual workforce into full membership of the corporate 'family' is the traditional disdain of manual labour in Korea and the high prestige of, and returns to, formal education, especially in non-technical subjects (Amsden, 1989; Michell, 1988). The labour force remains highly stratified by the formal educational system and manual skills in manufacturing continue to be regarded as less valuable than white-collar skills certified by higher education institutions. As Cho puts it, 'the wide wage differentials between production and clerical types of work do not necessarily reflect differences in their respective productivity' (1994, pp. 93-4; see also You, 1994). Since 1987-89, despite the substantial increases in production workers' real wages and some narrowing of differentials, there seems little evidence that these distinctions and prestige rankings have changed significantly in Korea, and so the establishment of Japanese forms of employment commitment to manual workers remains difficult there. Indeed, the intensification of the relocation of many chaebol production facilities to China and other cheap labour sites after the strikes and sit-ins of the late 1980s suggests that large employers in Korea find it easier to search for new workers than to implement quite different ways of managing current ones (Wilkinson, 1994, pp. 106–7).

Overall, then, the owner-controlled *chaebol* remain quite strongly tied to the Korean economy and the state, despite their growing overseas investment, strong cash flow and the weakening of the state's control over the economy as it has become more complex and political liberalization has grown. The highly diversified, centralized and risk-taking nature of the *chaebol* does not appear to have been greatly modified over the past decade, although some changes in managerial structures and practices have occurred in some of them. Despite government exhortations to support small and medium sized enterprises (SMEs), some policies to assist them, and periodic efforts to restrict the growth and range

of activities of the *chaebol*, the Korean economy remains dominated by the *chaebol*, who continue to behave in the 1990s as they did before (Cho, 1994; You, 1994). Significant changes to the Korean business system, then, have not yet taken place, and probably would require much more radical changes in the role of the state than have occurred so far.

## Changes in Taiwan

In the case of Taiwan, the democratization of the state and the election of a native Taiwanese as President, coupled with the extensive outflow of investment to China and ASEAN countries, might be thought to reduce the differentiation of Kuomintang (KMT) controlled upstream enterprises from Taiwanese CFBs in export industries and, perhaps, encourage a move away from original equipment manufacturing strategies to more integrated manufacturing and marketing organizations. As the mainlander-Taiwanese divide reduces, and the state becomes more responsive to Taiwanese economic interests, the growth of more Taiwanese firms into large companies with state assistance, especially in research and development activities, could be anticipated, especially if the banking system is reformed and encouraged to invest in smaller firms. Relatedly, the internationalization of operations and investment in distribution and marketing channels in North America and Europe by some CFBs can be seen as leading to the development of more integrated organizations which, perhaps, are becoming less personally controlled by owning families. The liberalization of labour legislation and some loosening of KMT control over unions (Wilkinson, 1994) might also herald less autocratic management practices and greater tolerance of worker autonomy and bargaining rights. Increases in wage rates and tight labour markets could also encourage paternalist employers to develop labour retention and development strategies to upgrade skills and products.

As in Korea, though, the relocation of many plants overseas and some investment in downstream activities have been insufficient to 'internationalize' most Taiwanese companies in the sense of transferring their key activities and personnel out of Taiwan, and have not been accompanied by major foreign investment in the domestic economyas distinct from the export processing zones. Similarly, although some firms such as Acer and Mitac have developed brand names and tried to move away from a concentration on own equipment manufacture (OEM) activities, their success has been mixed and their impact certainly not as marked as the Korean *chaebol* in North America and Europe. While state support for Taiwanese owned firms has increased in the 1980s and 1990s, especially in electronics (Wade, 1990), the state and the KMT still seem to dominate the heavy industry sectors of the economy, and firms employing less than ten people, including the self-employed, still employed 51.6 per cent of all employees in 1990 (Wilkinson, 1994, p. 139). In contrast, those with over 500 employees accounted for 3.9 per cent of all employees then, although this figure rose to 9.5 per cent for manufacturing firms. There appears to be little evidence of a substantial Taiwanese Mittelstand (the medium sized company sector in Germany) becoming established, at least up to the early 1990s.

Similarly, the high degree of organizational specialization and considerable ownership diversification in the larger CFBs and business groups seem to have continued into the 1990s in Taiwan. The importance of entrepreneurial familism (Wong, 1988), and its

associated low level of trust between owners and non-family employees—or employees with whom family-like relationships have not been established—remains considerable and discourages organizational integration of complementary economic activities. Relatedly, strong owner control, close supervision of work processes and a reliance on authoritarian paternalism as the prevalent way of managing labour relations seem to continue as distinctive features of Taiwanese enterprises which have encouraged the outflow of capital to China and elsewhere as Taiwanese workers became more expensive and demanding (Wilkinson, 1994, pp. 142–3). As in Korea, many CFBs in Taiwan have preferred to export their dominant management style and work control practices to locations where constraints are less onerous than to develop new ways of managing. Additionally, when they have established manufacturing operations in Europe and North America, anecdotal evidence suggests they have been much less successful than Japanese firms, partly because of difficulties in adjusting to different managerial and workforce expectations.

Overall, then, the political changes in Taiwan and the increasing internationalization of some CFBs' operations do not yet appear to have been so substantial as to modify radically established patterns of management and development. Even where ownermanagers have been educated abroad—as many Taiwanese engineers have been—and state support for industrial development has been considerable, as in the electronics and information technology industries, prevalent ways of managing risks and growth, and the labour force, remain dominant, and the traditional characteristics of CFBs continue to be reproduced in most Taiwanese firms (Deyo, 1989; Wilkinson, 1994). They seem unlikely to change significantly unless the nature of the Taiwanese family alters and the business environment becomes more formally regulated, such that owner-managers develop more trust in formal institutions and processes, as well as in long-term investment in key employees. While political risks in Taiwan have diminished, risks in both product and labour markets remain high for most firms, and risk-sharing institutions are only weakly developed. The pursuit of flexibility is likely therefore to remain a priority for most Taiwanese enterprises.

#### Conclusion

In sum, then, international and domestic changes in firms' environments have not resulted in significant alterations in the nature of these East-Asian business systems, and seem unlikely to do so without more radical transformations of their domestic institutions and agencies. Because of the distinctive nature of these institutions, incremental changes in firms and markets in Japan, Korea and Taiwan are most unlikely to result in them resembling Anglo-American ones, and so East Asia will continue to exhibit distinctive, but highly competitive, forms of economic organization which challenge Anglo-American ones.

#### References

Abegglen, J.C. and Stalk, G. (1985) *Kaisha, The Japanese Corporation*, New York, Basic Books. Amsden, A.M. (1989) *Asia's Next Giant*, Oxford, Oxford University Press.

- Aoki, M. (1988) Information, Incentives, and Bargaining in the Japanese Economy, Cambridge, Cambridge University Press.
- Bae, K. (1987) Automobile Workers in Korea, Seoul, Seoul National University Press.
- Chandler, A.D. (1990) Scale and Scope, Cambridge, MA, Harvard University Press.
- Cho, S. (1994) *The Dynamics of Korean Economic Development*, Washington DC, Institute for International Economics.
- Chung, K.H., Lee, H.C. and Okumura, A. (1988) 'The managerial practices of Korean, American and Japanese firms', *Journal of East and West Studies*, vol. 17 pp. 45–74.
- Clark, R. (1979) The Japanese Company, New Haven, CT, Yale University Press.
- Cusumano, M.A. (1985) *The Japanese Automobile Industry: Technology and Management at Nissan and Toyota*, Cambridge, MA, Harvard University Press.
- Deyo, F.C. (1989) Beneath the Miracle: Labour Subordination in the New Asian Industrialism, Berkeley, University of California Press.
- Dore, R.P. (1973) British Factory-Japanese Factory, London, Allen and Unwin.
- Dore, R.P. (1986) Flexible Rigidities, Stanford, CA, Stanford University Press.
- Fields, K.J. (1995) *Enterprise and the State in Korea and Taiwan*, Ithaca, NY, Cornell University Press.
- Friedman, D. (1988) The Misunderstood Miracle, Ithaca, NY, Cornell University Press.
- Gates, H. (1987) *Chinese Working Class Lives: Getting By in Taiwan*, Ithaca, NY, Cornell University Press.
- Gereffi, G. (1996) 'Commodity chains and regional divisions of labour in East Asia', *Journal of Asian Business*, vol. 12, pp. 75–112.
- Gerlach, M. (1992) Alliance Capitalism, Berkeley, CA, University of California.
- Greenhalgh, S. (1984) 'Networks and their nodes: urban society in Taiwan', *The China Quarterly*, vol. 99, pp. 529–52.
- Hamilton, G. and Biggart, N.W. (1988) 'Market, culture and authority: a comparative analysis of management and organization in the Far East', *American Journal of Sociology*, vol. 94, supplement, pp. 552–94.
- Hamilton, G. and Kao, C.S. (1990) 'The institutional foundation of Chinese business: the family firm in Taiwan', *Comparative Social Research*, vol. 12, pp. 95–112.
- Hollingsworth, R. (1991) 'The logic of co-ordinating American manufacturing sectors' in Campbell, J.L. *et al.* (eds) *Governance of the American Economy*, Cambridge, Cambridge University Press.
- Janelli, R.L. (1993) Making, Capitalism: The Social and Cultural Construction of a South Korean Conglomerate, Stanford, CA, Stanford University Press.
- Kagono, T., Alonaka, I., Sakakibara, K. and Okumara, A. (1985) *Strategic vs. Evolutionary Management*, Amsterdam, North Holland.
- Kao, C-S. (1991) 'Personal trust in the large businesses in Taiwan' in G.Hamilton (ed.) Business Networks and Economic Development in East and Southeast Asia, Centre of Asian Studies, University of Hong Kong.
- Kim, C.S. (1992) The Culture of Korean Industry, Tucson, University of Arizona Press.
- Kim, E.M. (1989) 'From dominance to symbiosis: state and *chaebol* in Korea', *Pacific Focus*, vol. 3, pp. 105–21.
- Kim, E.M. (1991) 'The industrial organization and growth of the Korean *chaebol*' in Hamilton, G. (ed.) *Business Networks and Economic Development in East and Southeast Asia*, Centre of Asian Studies, University of Hong Kong.
- Koike, K. (1987) 'Human resource development and labour-management relations' in Yamamura, K. and Yasuba, Y. (eds) *The Political Economy of Japan, Volume I*, Stanford, CA, Stanford University Press.
- Koike, K. (1994) 'Learning and incentive systems in Japanese industry' in Aoki, M. and Dore, R. (eds) *The Japanese Firm*, Oxford, Oxford University Press.

- Lincoln, J.R. and Kalleberg, A.L. (1990) Culture, Control and Commitment, Cambridge, Cambridge University Press.
- Michell, T. (1988) From a Developing to a Newly Industrialized Country: The Republic of Korea, 1961–82, Geneva, International Labour Organization.
- Odagiri, H. (1992) *Growth through Competition, Competition through Growth,* Oxford, Oxford University Press.
- Orru, M., Biggart, N.W. and Hamilton, G. (1991) 'Organizational isomorphism in East Asia' in Powell, W.W. and DiMaggio, P.J. (eds) *The New Institutionalism in Organizational Analysis*, Chicago, IL, University of Chicago Press.
- Redding, S.G. (1990) The Spirit of Chinese Capitalism, Berlin, de Gruyter.
- Sako, M. (1992) Prices, Quality and Trust, Cambridge, Cambridge University Press.
- Silin, R.H. (1976) *Leadership and Values: The Organization of Large Scale Taiwanese Enterprises*, Cambridge, MA, Harvard University Press.
- Wade, R. (1990) Governing the Market, Princeton, NJ, Princeton University Press.
- Westney, E. (1996) 'The Japanese business system: key features and prospects for changes', *Journal of Asian Business*, vol. 12, pp. 21–50.
- Whitley, R. (1992a) Business Systems in East Asia: Firms, Markets and Societies, London, Sage.
- Whitley, R. (ed.) (1992b) European Business Systems: Firms and Markets in their National Contexts, London, Sage.
- Whitley, R. (1996) 'Business systems and global commodity chains: competing or complementary forms of economic organization?', *Competition and Change*, vol. 1, pp. 411–25.
- Wilkinson, B. (1994) Labour and Industry in the Asia-Pacific, Berlin, de Gruyter.
- Wong, S-L. (1988) 'The applicability of Asian family values to other sociocultural settings' in Berger, P.L. and Hsiao, H-H.M. (eds) *In Search of an East Asian Development Model*, New Brunswick, NJ, Transaction Books.
- Woo, Yung-En (1991) Race to the Swift, New York, Columbia University Press.
- You, Jong-il (1994) 'Labour institutions and economic development in the Republic of Korea' in Rodgers, G. (ed.) *Workers, Institutions and Economic Growth in Asia,* Geneva, International Labour Organization.
- Zeile, W. (1991) 'Industrial policy and organizational efficiency: the Korean *chaebol* examined' in Hamilton, G. (ed.) *Business Networks and Economic Development in East and Southeast Asia*, Centre of Asian Studies, University of Hong Kong.

# CHAPTER 10 Technological systems, innovation and transfers

Brendan Barker and Akira Goto

## **10.1 Introduction**

Throughout the Asia-Pacific region technology is seen as the driving force behind national economic competitiveness. Led by the USA and Japan and more recently by Taiwan, South Korea, and Singapore—countries in the region have invested heavily to enhance their scientific and technological (S&T) capability. In this chapter we first look at the relationship between technology and economic growth (see also Chapter 3). We argue that the technological capability of the Asia-Pacific region countries has been enhanced both by public sector policies designed to promote S&T, and as a result of the activities of firms undertaking research and development (R&D) and technology acquisition in pursuit of their own competitive goals.

We introduce the concept of a *national system of innovation* which has proved useful in comparing the key characteristics of the systems of national institutions and policies supporting technology development and innovation in different countries. In our analysis of these systems we make a distinction between Asian-Pacific systems (i.e. Japan, the NIEs and the ASEAN countries) the Euro-Pacific systems (i.e. the USA, Australia, Mexico, etc.), and China as a special case.

In the following section we look at flows of technology and knowledge between countries in the region which leads us on to discuss the emergence of a regional system of innovation. In the final section, we briefly look at the relationship between the Pacific region and the global economy.

#### 10.2 Technology and economic growth

The major part of technological progress today occurs as a result of the intense *research and development* (R&D) activity undertaken by industry. Most of this effort is guided by market needs. R&D has two facets; one is to create new knowledge, the other is to create *technological capability* which is of critical importance when firms try to learn technology developed elsewhere.

As the product of R&D is knowledge, there is an inherent problem of *externality*.

#### **Box 10.1 Externality**

Externality refers to the way that the activity of one economic agent has effects on another that should not, or cannot, be ignored by the first. The most important of these in market systems is when the 'external effects' of the actions do not pass through the price system and register as a market transaction. There are two major types of externality: negative and positive.

A *negative externality* arises when one action has negative effects on other agents, though the agent that causes these *negative costs* does not have to pay for the damage done. As an example we can point to the way that pollution arising from car exhaust fumes affects pedestrians, but the car driver is not responsible for the added 'cost' incurred by those pedestrians.

A *positive externality* arises when the activity has external affects which produce a benefit to others, but when the agent responsible for those benefits cannot gain financial reward for them. This is the case with R&D activity and technological development. These can produce *spillover benefits* to society at large, which are not taken into account as compensation to that company or agent that produces the benefit. Sometimes this is expressed as an *external social benefit* which is more than the purely *private benefits* accruing to the firm.

While it takes a large amount of time and money to create new knowledge, it can easily be transmitted and copied at little extra cost. In one sense, it is socially desirable that the new knowledge, once created, is used as widely as possible. However, if individual firms are unable to *appropriate* (gain the benefit of) the fruits of their costly R&D, they will have less incentive to invest in R&D. The intellectual property system (e.g. patents) is designed to solve this problem by providing some protection to innovators.

Even with such protection, there are many ways for technology to 'spill-over'. Firms and countries can 'reverse engineer' advanced products (dismantling a competitors' products to learn about the technology embodied in them); they can send their own researchers and engineers to other countries or invite foreign researchers from other countries; they can subscribe to journals or attend conferences; they can learn through licensing agreements; they can learn from the plants established by developed country companies in their countries, etc. Most of these activities are informal and not reflected in R&D and technology trade statistics, which cover only a small part of innovative activity.

The inherent externality of technological knowledge makes it a very difficult 'good' to trade. Usually, the best way for a firm to profit from its superior technology is to manufacture and sell a product (or to employ a process) using that technology. For this reason, the successful exploitation of an innovation requires investment in complementary assets, such as a manufacturing facility or sales network, in addition to the initial investment in the R&D itself. It follows that a stable and innovation-friendly macroeconomic environment is a necessary condition for successful innovation to both occur and be exploited.

With this theoretical framework in mind, we now look at the issues surrounding technological progress and economic growth in the Asia-Pacific region. First, East Asian

countries have been, and remain, among the most rapidly growing in the world. The reasons for this extraordinary growth have been the subject of much study and often heated debate (see Chapters 3, 4 and 6 in particular). A reasonable conclusion would appear to be that growth in the region has been driven both by increased inputs (e.g. larger, more highly skilled workforces, greater capital investment) and by the increasing use of new technology which has allowed those inputs to be used more effectively (Chapter 3). The relative importance of these sources of growth may differ among countries and over time.

The second issue relates to the question of who are the major players in this process of technological progress and economic growth. Again, there are two camps. The first contends that the leading role was played by government. The second asserts that market mechanism and competition drove technological and economic development. Once again, these two are not substitutes (Chapter 6). The market mechanism functions in the institutional, social, political and cultural context of any country. In most North-East Asian countries the government played a major role by establishing the institutional framework (such as property rights) and a healthy macro environment (low inflation and government deficits, and a reasonable exchange rate) that allowed innovative activities to be rewarded. More directly, governments also played an important role in defining and implementing industrial policy—particularly with regard to promoting technology-intensive industries. The success of such policies in supporting economic growth has been questioned (we return to this later).

The third issue relates to the nature of technology mentioned above. In spite of attempts to protect new technology by patenting etc., it is still relatively easily transferred. Economic studies indicate the significance of such R&D spillovers. Coe, Helpman and Hoffmaister have shown that an addition of US\$100 to either the US or Japanese domestic R&D stock raises total GDP in the 77 developing countries as a group by almost US\$25 (Coe, Helpman and Hoffmaister, 1995).

Latecomer firms and countries have an advantage in that they can learn and use technology already developed elsewhere. We have already noted that there are many ways in which this learning can occur. For countries that are latecomers to industrialization, learning by borrowing and improving on technologies already developed by firms in more advanced economies has proved to be the most important path to rapid economic growth (Amsden, 1989). To learn and become capable of using advanced technology, the receiving country has to have the capability to do so. In the East Asian countries, well developed education systems and a degree of openness were the background to their capability to learn (Chapter 2). These characteristics have often been absent in developing countries in other regions and are one reason for the difficulty in replicating the East Asian success story.

It is not surprising therefore that developed countries prefer strong intellectual protection while developing countries prefer a less stringent regime. In recent years this has been a source of friction between developed and developing countries in the region (the disputes in the early 1990s between the USA and China over intellectual property rights being a case in point).

#### 10.3 National systems of innovation

The rate and direction of technical change both shapes and is shaped by the institutional and policy environment in a country. Different national technological and institutional environments have been investigated in an effort to understand why technological change appears to occur more rapidly and efficiently in some countries than in others. One fruitful avenue of research recognizes that innovation takes place in 'systems' of public and private institutions and the linkages between those institutions.

It is through such systems that new knowledge is generated and, together with existing knowledge, comes to be applied. This perspective led to the concept of a *national system of innovation* (NSI) (Freeman, 1988; Lundvall, 1992; Nelson, 1994). A 'national system of innovation' consists of those *organizations, institutions and linkages in a specific country which generate, diffuse and apply scientific and technological knowledge.* The NSI concept begins with an assumption that national competitive performance depends on the technological capabilities of a nation's firms. The ability of these firms to innovate depends on their own capabilities and on their interactions with various external sources of knowledge, facilities and financing. Countries which facilitate these linkages are more likely to flourish than those which do not. (This NSI concept to some extent parallels the concept of a national business system discussed in Chapter 9.)

The NSI concept is not without its critics: indeed, the use of the terms 'national', 'innovation' and 'system' have all been questioned. Nelson addresses these concerns in a review of the seminal 1994 study on NSI he co-ordinated (Nelson, 1996). He accepts that, on the one hand the concept of 'national' may be too broad and on the other too limited. Many innovation systems appear to be *sectorally specific* with different patterns of interaction between sectors in a national context. At the same time many of these innovation systems spill over national boundaries reflecting the fact that scientific and technological activity is increasingly international in its creation and dissemination—in part because of the role of multinational corporations and in part because of the growing importance of international collaborative links between researchers (Chapter 8). Nevertheless, there remain significant differences in the way each country achieves innovation and there is enough commonality across sectors within a country to make the national focus a reasonable one. Furthermore, it is still at the national level that science and technology policy is primarily defined and implemented.

Nelson also addresses definitional problems with the terms 'innovation' and 'system'. Innovation is often taken to mean the first to do something. In the NSI context the term is usually used in a much broader sense to 'encompass the processes by which firms master and set into practice product designs and manufacturing processes that are *new to them*'. Thus innovation in this context is taken to mean both the creation of new knowledge and the exploitation of existing knowledge from other sources inside and outside of the country. Similarly, the term 'system' is used in a more general sense than when it is taken to mean something that is consciously designed and built. In the context of a NSI, the concept is used to refer to the set of institutions whose interactions determine the innovative performance of national firms—including the educational system, the attitudes of firms, the form of the financial system, role of governmental organizations. In this
context that is no presumption that the system has been consciously designed or even that it works effectively or coherently.

Comparative studies have attempted to identify structural differences between NSI that promote or hamper innovation in different countries. Work, such as the 1994 Nelson study mentioned above, attempts to describe, analyse and compare the key processes and institutional actors involved—industry, university, government, etc. identifying the similarities and differences of these national systems, and the extent to which such patterns can explain the variation of national economic performance.

In most economies the private sector is the main source of technology development and transfer (see Table 10.1, column 3). The public sector however, plays an important role in monitoring, and participating with, the private sector. Governments support innovation directly, by conducting and paying for R&D, and indirectly, through measures that encourage R&D and commercialization by private firms. National systems of innovation are distinguished in part by the mix of these policies at a given point in time, and in part by persisting patterns of industrial and institutional structures and relationships that form the context for particular cases of innovation and set limits to the policies governments can pursue in the short and medium terms. Each country's technological progress is conditioned by the institutional settings and past historical developments of the country, and these factors are affected by the pattern of innovation of the country at the same time. In other words, technology and institution co-evolve over time and form a unique national innovation system.

National innovation systems can be categorized in several different ways. For convenience we divide systems in the Asia-Pacific region up into those nations with a European heritage (e.g. the USA, Australia, New Zealand); Japan and the NIEs (South Korea, Taiwan, Singapore, Hong Kong); and the remaining 'ASEAN' countries (Brunei, Thailand, Malaysia, Indonesia, the Philippines and Vietnam) and China. Several caveats are in order regarding this classification. Despite some similarities, there exist significant differences between the countries within each of these categories. The European Asia-Pacific grouping for instance includes some of the richest countries in the world as well as some of the poorest, some of these countries have political and legal traditions rooted in England and France, others a blend of Spanish and indigenous civilizations. Similarly, in spite of much talk of 'shared Asian values', differences between the NIEs and Japan and between the NIEs themselves are no less apparent. This diversity is also true of the ASEAN countries—particularly Vietnam which still maintains a communist government. China is in a class of its own—both because of its size and its unique features.

#### 'European' Asia-Pacific nations

In the category of European Asia-Pacific nations we include the USA, Canada, Australia and New Zealand—together, more problematically, with the Pacific Central and South American countries such as Mexico and Chile. Typically these countries each have a strong, higher education system and a well developed market economy. Traditionally these countries have looked towards Europe for cultural, scientific, and economic links. However, with the growth of East Asia as a major economic region they have become increasingly integrated with the rest of the Asia-Pacific region. All European Asia-Pacific countries are fostering closer S&T and industry ties with the rest of the Asia-Pacific.

Of the European Asia-Pacific countries, the USA is unique in several respects. The sheer size of its economy and R&D expenditure (see Table 10.1) are unparalleled. Large military and medical sectors have driven the development of technologically advanced products. Much research activity is centred on the higher education sector which has also been a particularly successful incubator of highly innovative research oriented firms. This success in commercializing academic research has been supported by an effective venture capital sector which encourages start-up firms. However, US industry has not always been successful in exploiting this strong research base.

In common with the USA, Canada, Australia and New Zealand also have strong basic research traditions. However, in recent years there has also been a growing feeling that high public investment in science has not produced an adequate return in terms of enhanced national competitiveness. New Zealand, in particular, has responded with a comprehensive restructuring of public sector research to foster greater competitiveness and has prioritized its relatively small resource base towards research on agriculture based manufacturing. There have been similar (although less radical) changes in the structure of the Australian science and technology system in an attempt to make both academic and public sector research more competitive and targeted towards commercial outcomes whilst at the same time building business participation in R&D.

	expenditure a. (million ECU) G		<i>R&amp;D</i> % <i>R&amp;D</i>		Number	Share	Share	High tech			
			as %	Expen diture by Sector of Perfor mance		of scient	of	of US	nology trade		
			of GDP (1992)			ists and engineers (1993)	world scien tific	patents (1993)	Imports (1992)	Exports (1992)	
							papers (1981– 84)		(million	ı ECU)	
				B	usines	s H.Ea	lucation Gov	vernment			
USA	112,503	(1993)	2.67	69.6	15.8	10.8	887,600	34.6	50.1	192,248	172,066
Canada	5,658	(1993)	1.45	55.0	26.0	17.0	66,502	4.5	2.2	49,029	41,360
Mexico	776	(1989)	0.31	52.6	14.4	11.6	8,595	0.3	0.1	17,850	7,209
Chile	434	(1992)	0.78	12.0	44.6	42.4	5,956	0.2	0.0	3,143	285
Australia	2,811	(1990)	1.56	44.2	26.9	27.6	78,538	2.1	0.4	14,361	2,565
New Zealand	280	(1992)	0.98	28.0	31.0	41.0	NA	0.4	0.0	3,231	387
Japan	44,237	(1993)	2.92	66.0	20.1	9.3	753,870	7.3	24.2	37,269	180,778
South Korea	5,176	(1991)	2.33	71.5	7.2	4.4	90,328	0.2	0.9	22,857	22,760
Taiwan	2,331	(1992)	1.82	52.6	14.4	11.6	33,179	0.4	1.4	NA	NA

## Table 10.1 R&D statistics for selected countries in the Pacific region (year?)

Singapore	401	(1993)	1.12	62.0	15.8	22.2	9,725	0.1	0.0	26,387	27,139
Hong Kong		NA	0.08	47.6	19.6	32.7	NA	0.2	0.1	36,711	8,282
Malaysia	73	(1989)	0.37	45.8	9.0	46.0	NA	0.1	0.0	15,7311	11,0951
Thailand	342	(1991)	0.16	5.0	20.2	69.6	NA	0.1	0.0	13,492	5,420 <sup>1</sup>
The Philip pines	110	(1984)	0.21	21.8	14.7	58.8	5,344	0.0	0.0	3,907	1,408
Indonesia	980	(1993)	0.26	13.0	4.0	81.0	66,668	0.0	0.0	8,566	1,296
China	16,917	(1993)	0.6	22.7	17.7	49.9	418,500	0.9	0.1	29,158	12,064

Notes:<sup>1</sup>=1991

NA=Not available

Sources: Turpin and Spence (1996 various pages); European Report on Science and Technology Indicators 1994, EUR 15897 ZN NSF; Human Resources for Science and Technology: The Asian Region, NSF93–303

Central and South American countries face many problems as they attempt to emulate the rapid growth seen in East Asia. With the recent exception of Chile these countries have not been able to create a supportive macroeconomic environment, have little or no active technology policy, have little money to invest in R&D and lack the comprehensive policies for human capital development, all of which were characteristics of the development policies adopted by the East Asian NIEs during the 1950s and 1960s.

#### 'Asian' Asia-Pacific nations

Japan experienced a very high rate of economic growth in the 1950s and 1960s followed by the NIEs (South Korea, Taiwan, Singapore, and Hong Kong) in the 1970s and 1980s. A very active private sector and, except for Hong Kong, active government policy to promote the private sector were the twin engines of this rapid growth. The East Asian-Pacific nations were particularly successful in learning, adapting and exploiting the advanced technology of the developed countries. This effective learning process was supported by the national emphasis on education in these countries. Learning was a prerequisite for firms in these countries to compete effectively in their domestic and export markets.

The governments in these countries maintained a basically healthy macroeconomic policy, i.e. low rate of inflation, low government deficit, and realistic exchange rate, which created an environment favourable for economic growth (Chapters 2 and 4). The governments also used industrial policy to promote specific targeted industries such as steel, auto, and electronics employing such means as protecting the domestic market, providing finance on favourable terms, providing subsidies and other financial incentives, and giving assistance to domestic firms to acquire technology from foreign firms. At the same time, the savings rate in these countries rose rapidly along with the high economic growth, which in turn allowed investment in new advanced plant and equipment (Chapter 3).

The effectiveness of the industrial policy followed by these countries during their development phase remains an open and hotly debated question. It is true that most of the targeted industries grew rapidly and became very competitive in world markets. However, there were also industries which became internationally successful without government help, such as the manufacture of VCRs and facsimile machines in Japan. There are also other industries which, in spite of massive government support have failed to become internationally competitive. For instance, Japan's aircraft manufacturing industry has been a future 'key industry' for more than two decades, with little to show for high levels of public investment. Similarly, the World Bank contends that South Korea's promotion of 'heavy and chemical industries' in the 1970s was associated with a substantial cost to the economy as a whole and wasted precious resources that could have been better used elsewhere (Chapter 6). Nevertheless it remains true that economies in the region were remarkably successful in creating the conditions for rapid and sustained growth. In many ways the emphasis on R&D and technological development in those countries was driven by strategic reasons. The governments wanted to secure their economic health as a bulwark against military and possibly economic competition from mainly Communist powers.

Japan is the dominant science and technology power in East Asia (see Table 10.1). The nation has traditionally depended on its industrial sector to take the leading role in R&D, while the university sector has played less of a role in generating new knowledge than is the case in the USA. Nevertheless, Japanese universities played a significant (although often overlooked) role in aiding industry to acquire and exploit new technology. The university sector in Japan continues to be an important source of high quality talent and links into both the national and international scientific communities. Now that Japan has caught up with, and in many areas of technology surpassed, the other industrialized nations, there is a recognition that Japan must focus more on the creation of new fundamental knowledge. In the 1990s attention was being paid to significantly upgrading the research infrastructure and quality of research carried out in the universities and national research institutes.

Like Japan, South Korea also has a strongly industry-based science infrastructure: half of the approximately 1,000 research institutions in the country are in private industry and half of these are heavily concentrated in the ten largest *chaebol* (industrial conglomerates), particularly within the electronics and chemical industries (Table 10.1). Like Japan before it, a traditional reliance on reverse engineering of imported technologies is now being replaced by more indigenous knowledge creation—with an increasing emphasis on more fundamental long-term research.

Central to current innovation strategies in both Japan and South Korea is the internationalization of the R&D base. South Korean industry is now following similar internationalizing strategies to that in Japan in establishing laboratories in the West. Domestic and international co-operative R&D ventures in both countries are being encouraged.

Compared to the breadth of technological development supported in Japan and South Korea, Taiwan and Singapore have been successful in particular niches. Although, for strategic reasons, Taiwan has a number of large, state-owned industries which are not internationally competitive, it also has a very dynamic sector of small and medium-sized companies which act primarily as suppliers to US and Japanese multinational electronics companies. A different strategy is being followed by the Singaporean government which has given particular encouragement to multinational corporations to locate their higher value-added activities in Singapore.

#### The ASEAN countries

In the 1980s, Japan and South Korea started to actively invest overseas—to be joined subsequently in these investments by the largely ethnic Chinese entrepreneurs of Hong Kong, Singapore and Taiwan who brought much needed technology and management expertise (see next section). The major destination of this foreign direct investment was other parts of East Asia such as the ASEAN countries and China where wage levels were lower and markets expanding rapidly as the economies grew. This investment supported domestic activities designed to enhance the indigenous knowledge base.

For instance, although Malaysia had a well established public research institute base, this was primarily oriented towards agriculture—a legacy of the prior British colonial administration. A current priority is to strengthen institutional capacity and to enhancing industry-public sector linkages. Industry is seen as the source of future growth in national R&D activity. Similarly, Thailand places a high priority in further stimulating growth in private sector R&D involvement as well as in expanding international S&T collaboration as a way of strengthening national capability. Strong support is being given to generic technologies—microelectronics, information technology and biotechnology.

Human resource development is a high priority in both Malaysia and Thailand. Similarly, both the Philippines and Indonesia recognize the importance of education and training if they are to effectively capture international flows of technical and economic capital. However, in the Philippines, where the culture is more Westernized through prior American influence and literacy rates are very high, the brain drain is a significant problem. The Philippines planned to produce an additional 3,000 scientists and engineers by 1998, while at the same time strengthening and upgrading the S&T infrastructure.

Indonesia is still at a comparatively early stage of technological transformation of the economy. Nearly three quarters of total manufacturing output is in industries featuring products with a low-technology intensity. Consequently, small scale industry and the informal sector are prioritized for support. In parallel however, development of sophisticated technological capabilities within 'strategic' industries such as aeronautics, energy and electronics, is being supported as a means to 'pull' the economy out of its developing country status.

Finally, although Vietnam joined ASEAN in 1995, it has a political system that is very different from, and is at a stage of development that is well behind, that of the other ASEAN countries.

#### China

China is unique in several respects. First and foremost the country is huge. Thus while the proportion of scientists and engineers to the rest of the working population is relatively low, the absolute number is large (Table 10.1). In addition, although per capita GDP is low, a well developed technological infrastructure has been created to underpin an advanced military sector. Because of this technologically sophisticated sector, the absorptive capacity of the country is high in a way which is not true for other countries at a comparable developmental stage.

To sustain growth China must address a number of critical problems. Decentralization will liberate the wealth-creating potential of those provinces favourably located on the Asia-Pacific seaboard. However, this has led to large disparities in wealth which is causing significant tensions between regions in China and the central government. The continued move toward market-based institutions will require reform of the financial sector and state enterprises—which increasingly act as a major drag on the rest of the economy. The recent handover of Hong Kong has created yet more pressures.

#### Convergence in systems

In recent years there has been growing convergence between countries in the region. For instance, many aspects of Japanese manufacturing systems are becoming globalized (e.g. 'just in time' inventory control, total quality management, extensive subcontracting and more besides). These tend to embody a particular process technology. At the same time many traditional Japanese practices are breaking down in the face of global competition—for instance, lifetime employment. The East Asian-Pacific countries—particularly the more advanced ones—are beginning to place greater emphasis on supporting more fundamental research. At the same time, a number of the European Asia-Pacific countries have begun to prioritize their research in terms of national requirements in a way that has been explicitly modelled on what is believed to happen in the East Asian-Pacific countries.

A major reason for this convergence is the increasing regionalization of innovation systems. National systems of innovation are embedded within the international economy, linked with one another through flows of capital and labour, goods and services (e.g. high technology equipment, consultancy), information and knowledge (e.g. collaborative scientific research) (see Chapters 8 and 14). The R&D process itself is becoming more internationalized. Large multinationals are finding it to their advantage to undertake various aspects of the innovation process in different countries. Furthermore, the costs and risks of undertaking large, expensive R&D projects are increasingly beyond the scope of individual enterprises, leading to a range of international collaborative links between different organizations.

In a later section we look at how these regional systems of innovation are developing. First we examine the flows of technology in the region which are underpinning the emergence of these systems.

#### **10.4 Technology flows**

A triangular relationship of technology transfer exists among the USA, Japan, and the Asian NIEs and ASEAN countries (see also Chapter 2). The USA has created various types of new scientific and technological knowledge, which has been diffused to other countries including Japan, the Asian NIEs, and ASEAN countries. Japan and the East Asian countries have developed their technological innovations and industrial growth based upon this knowledge. According to statistical data, Japan and the USA are the two

major suppliers of industrial technologies to South Korea, Taiwan, and to other Asian countries. Japan still continues to import advanced technologies from the USA, but also supplies the USA and East Asia with industrial technology. This system is summed up in Figure 10.1.





The vast expansion of international trade and the associated increases in the presence of multinational firms played an important role in the diffusion of knowledge across borders (Chapter 8). In particular, Japan and the NIEs benefited greatly from access to the US market (Chapter 2). Nelson and Pack argue that this gave companies the opportunity and the challenge of competing on world markets, and forced them to judge their competencies against those of leading companies in the USA (Nelson and Pack, 1996). In addition, the possibility of exporting to the USA made large-scale production economic.

Immediately after the Second World War, and even more so during the Cold War, the USA became a key economic player in the Pacific region. The USA provided aid and military security arrangements bilaterally, and American business was the source of much investment throughout the region. However, American manufacturer participation in the Japanese market was limited primarily to technology sales. This strategy allowed technology to flow into the Japanese economy while investment restrictions excluded foreigners and foreign control. The vast majority of USA-Japan corporate ventures from the 1950s until the mid 1980s involved licensing agreements that entailed the straight-out sale of US patents, directly transferring basic research and development knowledge to Japanese manufacturers. From 1951 through to 1983, Japanese companies entered into some 42,000 contracts for technology imports, paying a cumulative price of US\$17 billion. The cost of acquiring this advanced technology from the USA and other advanced countries was far lower than it would have been to develop it domestically.

The regionalization of production in East Asia has its foundations in the Japanese colonial period. Japanese production in Taiwan, South Korea and Manchuria was an integral part of the pre-1945 Japanese empire (Chapter 1). The post-war roots of interfirm alliances in East Asia date back to the direct investment in Taiwan by Japanese firms in such industries as electronics and machinery manufacturing in the late 1950s. These investments were initially aimed at establishing a presence in the local market. They were followed by a series of equity and non-equity alliances between Japanese and local enterprises in response to local ownership and content regulations promulgated by the Taiwanese government. The same pattern was replicated, to a lesser extent, in South Korea in the aftermath of the signing of the Japan-South Korea normalization treaty in 1965. In the last decade such networks have undergone a significant expansion as Malaysia, Thailand and coastal China have all become linked to production in North-East Asia. Much of this latest investment derives from the transnational relationships linking the Chinese entrepreneurs in these countries.

Ravenhill argues that three sets of interacting factors underlay the surge in foreign investment. The first related to the soaring costs of land, labour and other factors in Japan, South Korea and Taiwan in the middle of the 1980s. The currency appreciations brought about by the Plaza Agreement exacerbated the problems, and were sufficient to overcome inertia and prompt companies to relocate some aspects of their manufacturing in lower cost countries, and to change patterns of trade (Ravenhill, 1994—see also Chapters 2 and 4).

The second group of factors was political. Tensions over trade imbalances and market access between Japan and the USA were extended to South Korea and Taiwan as the last two quickly exploited the advantages they gained in the immediate post-Plaza period. These tensions in turn generated domestic political forces that interacted with and reinforced underlying economic changes. For instance, the Japanese government increasingly pressured corporations to increase both their foreign direct investment and their sourcing from their overseas subsidiaries.

The third set of factors relates to changes in the production process and the advent of information technology which facilitated flexible production techniques and transnational communication. The diffusion of just-in-time production and value-added networks has facilitated greater inter-firm co-ordination (Chapter 14). This trend toward networks has also been strengthened by the rise of component suppliers as technological innovators in their own right, and by the way microelectronic technologies lend themselves to R&D based on inter-sectoral collaboration and producer-user interaction.

The consequence of these interacting forces has been a dramatic surge in foreign investment from Japan, South Korea, and Taiwan. Japan's investment in manufacturing in other East Asian countries in the years 1986–89 exceeded the cumulative total for the 1951–85 period. In 1988 Japanese manufacturing investment in ASEAN exceeded that in the Asian NIEs for the first time as Japanese offshore production of consumer goods for the global market shifted from the NIEs to the ASEAN countries. Meanwhile, the growth in Taiwanese and South Korean investment in ASEAN was even more impressive; total investment from the NIEs (including Singapore) in Malaysia and Thailand exceeded inflows from Japan by the end of the 1980s. Starting in the late 1980s, capital from ethnic Chinese in East Asia also became an important source of investment in the region. In the

1990s, this has been joined by capital from China. Foreign direct investment from East Asia is spreading to Vietnam, India, and other countries, seeking new frontiers of growth.

To these three sets of factors we add a fourth which is the growing importance of South-East Asia as an indigenous source of capital. Increasingly East Asian firms have begun to re-invest their profits in the region. Such patterns of investment have now built up their own momentum with currently, about 40 per cent of the FDI inflows into the developing countries of the region originating from other developing countries in the region. As a result, companies from these countries have become the single most important investors in East Asia, ahead of firms from Japan, the USA and Europe (Sauvant, 1996).

#### The role of multinational corporations

The generation and transfer of technology, innovation capabilities and skills within multinational corporations (MNCs) and their effects on other firms through linkages and spillovers has important implications for country performance (but see also Chapter 8). Given the dominant role of MNCs in innovation and technology development, inward foreign direct investment (FDI) and other relationships with MNCs are an important means for host countries to advance their technological capabilities. At the same time, outward FDI can strengthen the technological capacities of home countries by allowing their firms to access technology otherwise difficult to obtain and to spread R&D costs.

The R&D activities of MNCs are becoming more dispersed as MNCs utilize more cost-effective sources of new knowledge. Despite the 'stickiness' resulting from an established pattern of locating R&D facilities in home countries and the costs of organizing and co-ordinating geographically dispersed R&D activities, this trend is likely to increase.

In recent years technology alliances in the region have proliferated. For instance, Hitachi of Japan has formed an alliance with Goldstar of South Korea, under which Hitachi provides DRAM technology to Goldstar. Toshiba, also of Japan has transferred all prototyping, development and production of some video-cassette recorder models to Samsung Electronics of South Korea in order to concentrate on other models. Multinational corporations also establish links with local research centres and research institutions in home and/or host countries. For instance, in China, Ford has established a \$1.6 million R&D foundation in collaboration with the Chinese Government, aimed at funding projects in universities and research institutions.

Firms are typically careful to retain control over their vital know-how when investing in foreign countries. MNCs have been criticized for typically licensing older technologies to firms in developing countries while keeping newer technologies at home, with the result that local firms have been prevented from acquiring and mastering the technology necessary to compete with firms in developed countries. However, there is some dispute as to whether MNCs are really transferring 'R' to the less developed Asia-Pacific nations, though they have tended to disperse 'D' more readily. Nevertheless, MNCs are the primary conduits for the transfer of technology and related skills from developed to less developed countries.



Transferred technology: a factory in Bataan making training shoes for an MNC

While member firms within a MNC system have privileged access to technologies and skills, MNCs also establish collaborative relationships with firms outside their production systems for generating or transferring technology. Such collaborative arrangements include technology alliances among MNCs of similar strength and between MNCs and smaller-scale local enterprises; and co-operative arrangements between MNCs and universities or research institutions. These arrangements allow information sharing and joint problem solving. Most importantly they also support what has been called 'learning by interaction' by which local organizations learn international best practice from the MNCs they are working with. Box 10.2 indicates how outsourcing by MNCs is increasingly integrating technological activity across the Asia-Pacific region.

#### Box 10.2 Outsourcing and technological integration

[The US sportswear company Nike] subcontracts the manufacture of its athletic shoes and clothing to 40 separate locations, mostly in South and South-East Asia. Design and marketing are done by the parent firm, with new designs relayed by satellite to a computer-aided design/computeraided-manufacturing (CAD/CAM) facility run by a subcontractor in Taiwan Province of China. Prototypes are constructed and modified, and final plans are sent by facsimile to subcontractors throughout the region, where some Nike employees are present to ensure quality control. The output is sold world-wide. Since individual subcontractors are dependent upon the parent for design, standards and marketing, they are not capable of standing alone.

In recent voors Nike has been ernending in China Indonesia and Thailand

which offer substantial savings on labour costs. As a result—and despite big improvements in manufacturing productivity—most subcontractors in the Republic of Korea have been unable to retain Nike's business. Nike is constantly introducing new products and operates in many price-sensitive markets. It therefore has both the opportunity and the need continually to seek out lower-cost suppliers. For some products, however, as in the electronics industry, the technological aspects of competitiveness are beginning to outweigh the cost considerations [...]

[For instance] United States-based merchant semiconductor companies expanded their output in the 1960s and 1970s in an increasingly competitive environment, so took to outsourcing in various Asian countries. They did so largely through whollyowned affiliates, as parent MNCs sought to retain close control over the underlying technology which, in this industry, is a major source of firm-specific advantages. Most corporate functions remained within the parent firm. Wafers manufactured in the United States were air-freighted to Asia, where they were assembled into circuits. The circuits were then air-freighted back to the parent firm for testing and distribution. In some cases, local firms developed the manufacturing skills and acted as subcontractors. Over time, and especially when Japanese semiconductor MNCs expanded into Asia, the Asian industry attracted more sophisticated parts of the process, including research and development and wafer fabrication. Some local firms, such as those from the Republic of Korea, have themselves become competitors in world markets. [...]

Source: United Nations (1993, pp. 120–1)

Customer-supplier linkages between firms is a particularly important channel for the transfer of technology and know-how. The crucial contribution of these linkages to a host economy lies not only through the business generated for local firms, but in the knowledge and technology flow from MNCs to local firms. Such flows may be in the form of designs, drawings, specifications, manufacturing knowledge/process know-how, quality control, productivity enhancing techniques, manage-ment know-how, training and the like. Often, personnel from foreign affiliates train personnel in supplier firms in the production of required components or parts, leading to the upgrading of skills in local firms. Increasingly suppliers and subcontractors are involved at an early stage in product or process changes so that they can gain the know-how to effect changes in their products. Japanese MNCs, especially in the automotive sector, are well known for this kind of relationship with their subcontractors (see Chapter 2). Such technology and skill flows contribute to a continuous upgrading of domestic firms and thus enhance their competitiveness.

Closely related to the transfer of technology and innovatory capabilities within MNCs is the transfer of skills necessary for utilizing technologies or participating in technological development. The principal modes of skills transfer are the use of expatriate employees and, most importantly, training of local employees.

#### Flows of highly skilled labour

Accompanying the high growth of foreign direct investment in the Asia-Pacific region has been the striking increase in the flow of professional, managerial and other skilled workers who have been accompanying investments by multinational companies throughout the region. Such flows are another important route by which technical knowhow and best practice are disseminated. It is significant that flows of people across national borders in the Asia-Pacific has grown twice as fast as the flows of people worldwide.

In earlier years most multinationals used their own expatriates to manage their operations but most countries increasingly rely on managerial and professional staff from within the region. As foreign investment flows from within the NIEs has gradually overtaken investment from Japan, a large number of professionals and highly skilled workers from Hong Kong, Republic of Korea, Taiwan/China and Singapore are working abroad. Companies from these countries are also more willing than Japan to employ third-country nationals (Gamier, 1996). Also as job opportunities due to economic recession decreased in the developed countries in the region especially Australia and New Zealand, highly skilled labour from these countries has moved into the fast growing NIEs and other economies—Malaysia, Indonesia, Thailand and more recently the Philippines.

Of the 87,000 professionals and managers from Hong Kong working abroad almost 50,000 are daily commuters to China's Southern Provinces. In early 1996, there were 110,000 expatriates working in Hong Kong coming from the USA (36,000), Canada (29,000), Japan (24,200), Australia (21,100) and the United Kingdom (34,500) (Gamier, 1996). Recent estimates from Indonesia reported that there were in early 1995 almost 57,000 expatriate workers in the country with a wage bill of US\$2.5 billion (Amjad, 1996). The example of Japan is also revealing. In 1993, about 273,000 Japanese workers resided abroad (with 25 per cent in the Asia-Pacific alone) working as the expatriate staff of the subsidiaries of Japanese MNCs. In addition to this number, between 15,000 to 20,000 people are thought to have travelled abroad each year in the 1990s for short periods (of a few months at most) to provide some type of service.

Another interesting development is the steady flow of East Asians who are returning from North America, Europe and Australia as new and equally well-paid opportunities open up in their own or neighbouring countries within the region (and as opportunities in these developed countries have dried up). Taiwan, for example, has witnessed a dramatic reversal of the 'brain-drain' as more and more people who were born in Taiwan, but were trained in the USA, return to become high-tech entrepreneurs. Similar developments are taking place with respect to South Korea and Hong Kong as well.

Many of these individuals have well-established professional networks in the USA or other countries, and can use these networks to obtain information and finance. The networks between researchers in the north-east and south-east regions of the Asia-Pacific are still strongly biased towards the more advanced industrial countries of Europe and the USA rather than with other nations of East Asia.

However, there are signs of change emerging. Many researchers are now returning home from the USA to Taiwan, Singapore and Hong Kong because of better opportunities, and are maintaining their cross-Asian-Pacific networks from their new positions.

#### 10.5 Regional systems of innovation

Although cultural, economic and political differences are likely to remain a feature across the economies in the region, there is growing evidence that many of the nations of Asia are paying increasing attention to internationalizing their R&D systems in order to capture the 'flows' of knowledge that an increasingly globalized knowledge economy represents. This is part of a wider process of regionalization which is being driven by several forces, all of which have served to foster greater intraregional trade and technology exchange.

'Flying geese', air displays and changing industrial structure

The spreading of industries from country to country has been likened to that of 'flying geese'. The 'flying geese' model was first used to describe the life-cycles of industries in the course of economic development (Akamatsu, 1964). The focus was on specific industries in specific countries. This was extended to study the dynamic changes in the industrial structure in specific countries, and subsequently to the shift of industries from one country to another (see Chapter 2).

In the original form of the model, the life cycle of a particular industry is represented by the trends in the value (or volume) of imports, production and exports. For a particular industry, the level of imports first rises and then declines. The same rise-and-fall pattern is later repeated by domestic production and by exports. When plotted against time, imports, domestic production and exports form a pattern of overlapping inverted V-shape curves like wild-geese flying in orderly ranks. Figure 10.2 is designed to remind us of the flying geese analysis presented in Chapter 2. Part (a) shows the basic pattern for a single industry in a single country.

This pattern is followed by one industry after another. If, to simplify things, we replace the earlier import, production and export curves by ones representing the changing competitiveness of industries over time it is possible to map changes in the individual structure of a country over time (Figure 10.2b). A typical sequence is the shift from the textile industry to the chemical industry and then further to the steel and automobile industries.



(c) For a particular industry across different countries

### Figure 10.2 The 'flying geese' metaphor

The 'internationalized' version of the metaphor describes the movement of industries from more advanced countries to less advanced ones. A typical arrangement sees Japan developing first, gaining a strong technological base, then as wages and other costs rise, production facilities are relocated to the NIEs and then to other lower-cost economies (Figure 10.2c). According to proponents of the model, a sophisticated division of labour emerges among Asian-Pacific countries.

Chen argues that the 'flying geese' hypothesis is no longer adequate to describe the existing and future pattern of industrialization in the region (Chen, 1995). He suggests that it is no longer always true that an industry is first established in an advanced country and then passed on to the next tier of countries in the next phase of the product cycle—in a linear-like pattern. Rather, it is a specialization of different countries in the different *sub-sectors* of a particular industry that is affected by a newly developed technology soon after the technology is commercialized. Each time a new technology becomes commercialized, a cluster of economies engaged in a new pattern of production and a different division of labour will occur. To describe this vision of industrial development and specialization, Chen uses the analogy of an air display or aerobatics: aeroplanes taking part in an air display form into different patterns, each aeroplane has a specific role to play depending on the particular pattern. During the course of the air display planes are

continuously changing position and roles—there is no clearly defined tiers or groups of planes.

This specialization does not exhibit any 'flying geese' pattern in which specialization is in accordance with a pecking order of economies. Instead, each developing country takes up a *specific role* and engages in a *particular* application of the technology to a certain sector all at the same time. With the possibility of technological leapfrogging, the area of specialization in the application of a particular technology is not necessarily related to the level of development of that economy. With a new technology emerging, another clustering will take place with different countries responding somewhat differently.

The 'flying geese' model has also been criticized as an explanation of past developments—at least in its emphasis on Japan leading the rest of East Asia along the path to development (Hobday, 1996). First, some versions overstress the importance of Japanese FDI—Japanese FDI was only a minor source of growth and capital formation in each of the NIEs. In particular, the 'flying geese' idea grossly understates (or ignores) the importance of the US economy as an export market for East Asian goods and tends, conversely to neglect the weakness of Japan as an importer of goods. Exports to Japan by Malaysia, Thailand, and Indonesia have mostly been of primary commodities, while their imports from Japan have almost all been manufactured goods. The manufactured exports that do make it to Japan rarely come from purely South-East Asian firms; they are usually produced by Japanese subsidiaries or joint ventures affiliated with a Japanese company. The NIEs owe their growth to independent efforts, investments and rapid technological learning linked to a variety of sources, not only Japan.

Finally, rather like Chen, Bernard and Ravenhill (1995) have argued that the nature of production in the Asia-Pacific region is becoming *interdependent*. It is organized in the form of coincidental hierarchical networks stretching across different countries, driven by the strategies of firms who look to the comparative and technological advantage of *particular* territorial sites for *particular* parts of the production processes. Production structures are thus increasingly linked across countries. Parts of the same product are produced (within the same 'industry') coincidentally in different countries rather than in one country in turn (see also Chapter 14).

#### Emerging regional structures

A prominent feature of the East Asian political economy is that, in contrast to North America or Europe, regional economic integration has been both firm- and productionled, as opposed to being initiated by governments (Ravenhill, 1994). Nevertheless, the 1980s saw the growing importance of regional co-operative mechanisms (Chapters 7 and 13). Three levels of officially-sponsored integration may be distinguished in East Asia: the sub-regional, most notably the emergence of transborder growth areas (SREZs, Chapter 7); the regional, as for instance the ASEAN Free Trade Area (AFTA); and the supra-regional in the form of the Asia-Pacific Economic Co-operation (APEC) grouping.

At the regional level, ASEAN's Committee on Science and Technology (COST) (supported in part by the Fund for Science and Technology) has been established to focus on projects of regional significance in mutually beneficial areas such as environment, energy, and food and agricultural technology.

There have also been developments at the supra-regional level. In 1989, the Pacific Economic Co-operation Council (PECC) inaugurated a Science and Technology Task Force in order 'to provide a forum where consensus can be reached and actions taken on programmes and policy recommendations that foster Asia-Pacific science and technology co-operation. This in turn will foster greater and more equitable economic and social development.'

Task Force members include representatives of twenty Pacific Rim economies. Each member economy of PECC's S&T Task Force has a science and technology advisory committee, thus creating a Pacific Rim network of business, academic, and government experts in S&T. Science and technology subcommittees, led by various member countries, are responsible for developing programmes in technology transfer, human resource development, and co-operative R&D.

In October 1995, APEC ministers responsible for S&T met for the first time in Beijing to discuss how to advance the process of co-operation in science and technology amongst the member economies. Ministers agreed that support for science and technology should advance the goals of: building robust and competitive economies in the region; promoting environmental quality and sustainable use of natural resources; improving the quality of life, human health and development, and knowledge; and fostering improved private/business sector exchanges and interaction in technology co-operation.

Ministers entrusted the Working Group on Industrial Science and Technology (WGIST) to continue its work on science and technology policy and programme discussion and information sharing, acknowledging that this is an essential element of furthering understanding in the region.

#### The Asia-Pacific region in the global system of innovation

Finally, it is worth saying a few words about the Asia-Pacific region in the global system of innovation. In recent years there has been much talk of the growing relationship between the three regions based on East Asia, North America, and Europe (Chapter 13). In terms of investment and knowledge flows the pattern of this relationship is somewhat imbalanced: North America has strong reciprocal links with both the rest of the Asia-Pacific region and Europe; the link between Europe and the Asia-Pacific region is less strong, although growing.

Whereas European countries have a significant number of firms that invest abroad, only some 3 per cent of the total FDI stock and flows of the EU are directed towards developing East Asia. By contrast, developing East Asia countries' FDI in Europe already accounts for about 4 per cent of total developing Asia's FDI—a figure likely to increase as East Asian firms build up their outward investment stocks (Sauvant, 1996).

R&D activities, not just of US or European firms, but also of Japanese and other East Asian firms have been increasingly located in one or both of the other 'triad' areas. Another more recent feature of this internationalization has been the growth in so-called strategic technology alliances. In both these cases of 'knowledge transfer'—R&D and strategic alliances—the pattern is similar to that for FDI: i.e. strong reciprocal links between North America and both the rest of the Asia-Pacific region and Europe; with less strong links between Europe and the East Asian region (European Commission, 1994).

#### **10.6 Conclusion**

Though expressed in different ways in different countries at different times science and technology (S&T) is seen to be a central feature of any development strategy. Governments have largely looked to the business sector to drive S&T growth, whilst also directing public sector science towards the marketplace. The USA and Japan are the most technologically advanced countries in the world and a major source of new technology to other countries in the Asia-Pacific region. Flows of investment and knowledge have been important factors underpinning the success of the region as a whole.

The ability to learn and exploit foreign technology requires a domestic capability. The East Asian countries, with their investment in infrastructure and human capital had this capability to learn. The governments of these countries also provided sufficiently stable macroeconomic environments and industrial policies that were, on the whole, supportive of technologyled development. It is no accident that few if any of these conditions are found in the poorly-performing central and south American countries of the Asia-Pacific.

The countries of the Asia-Pacific region, starting from diverse cultural and organizational practices have largely been successful in creating frameworks to support the exploitation of new technology from both domestic and foreign sources. Countries in the region continue to invest and plan for S&T in the long term, with the more advanced countries supporting the long-term fundamental research needed to underpin the industries of the next century.

#### References

- Akamatsu, K (1962) 'A historical pattern of economic growth in developing countries', *The Developing Economies*, March-September.
- Amjad, R. (1996) 'International labour migration and the implications in the APEC region', mimeo, APEC Study Center Conference Papers, Singapore, APEC Secretariat.
- Amsden, A.H. (1989) *Asia's Next Giant: South Korea and Late Industrialization*, New York, Oxford University Press.
- Bernard, M. and Ravenhill, J. (1995) 'Beyond product cycles and flying geese: rationalization, hierarchy, and the industrialization of East Asia', *World Politics*, vol. 47, January, pp. 171–209.
- Coe, D.T., Helpman, E. and Hoffmaister, W. (1995) 'North-South R&D spillovers', *NBER Working Paper Series*, no. 5048.
- European Commission (1994) *The European Report on Science and Technology Indicators*, Brussels, European Commission.
- Freeman, C. (1988) 'Japan: a new national system of innovation' in Dosi, G. *et al.*, *Technical Change and Economic Theory*, London, Pinter.
- Gamier, P. (1996) 'Service providers: a growing dimension of highly skilled labour in Asia', paper presented at *Conference on 'The Dynamics of Labour Migration in Asia*, 6–8 March, Tokyo, Nihon University.

Hobday, M. (1996) Innovation in East Asia: The Challenge to Japan, Aldershot, Edward Elgar.

- Lundvall, B.A. (ed.) (1992) National Systems of Innovations: Towards a Theory of Innovation and Interactive Learning, London, Pinter.
- National Science Board (1996) *Science and Engineering Indicators—1996*, Washington DC, US Government Printing Office.
- National Science Foundation (1995) *Asia's New High Tech Competitors*, Arlington, VA, National Science Foundation.

- Nelson, R.R. (1996) 'National innovation systems: a retrospective on a study', *The Sources of Economic Growth*, Cambridge MA, Harvard University Press.
- Nelson, R.R. (ed.) (1994) National Innovation Systems: A Comparative Analysis. New York, Oxford University Press.
- Petri, P.A. (1993) 'Government and the East Asian miracle: one model or many?' paper prepared for the *World Bank Conference on Government and East Asian Economic Success*. Honolulu. The World Bank.
- Ravenhill, J. (1994) *The Regionalization of Production and Competitiveness in East Asia*, mimeo, Canberra, Department of International Relations, Australian National University.
- Sauvant, K.P. (1996) 'International trade and investment trends', keynote address, *First Annual Australian Conference on International Trade, Education and Research*, Melbourne, University of Melbourne.

The World Bank (1993) The East-Asia Miracle, New York, Oxford University Press.

Turpin, T. and Spence, H. (1996) *Science and Technology, Collaboration and Development Among Asia-Pacific Economies, mimeo, Melbourne, APEC Centre, University of Melbourne.* 

United Nations (1993) World Investment Report 1993, New York, United Nations.

United Nations (1995) World Investment Report 1995: Transnational Corporations and Competitiveness, New York, United Nations.

## PART 3 Consequences and Implications

### CHAPTER 11 Winners and losers in the Asia-Pacific

Paul Lubeck

#### **11.1 Introduction**

Earlier chapters of this book emphasized the historical uniqueness and extraordinary performance of the Asia-Pacific economy. For many, but certainly not all, countries on the East Asian side of the Asia-Pacific, the accepted statistical measures of rapid industrialization presented below support this generalization. Yet, it is also true that many countries increasingly integrated into the Asia-Pacific economy, i.e. the Philippines, Vietnam, Mexico and Peru, have not enjoyed the performance characteristic of what are called North-East and South-East Asia's *newly industrializing economies* (NIEs). Before exploring the way in which historical, social structural and cultural factors—class, gender, ethnicity, regional identities and political networks—acted *within* states to determine 'winners and losers', let us first review differences among nation-states so as to underscore how state development policies sort out winners and losers in the Asia-Pacific.

#### 11.2 Why states and industrial development policies matter

The centrality of citizenship in a nation-state means that state's organizational capacity—legitimacy, discipline, efficiency, social cohesion—acting in alliance with national enterprises competing for world market opportunities, in effect, combine to designate winners and losers at the aggregate level of nation-states. That is to say, for the majority of people in the developing countries of the Asia-Pacific, who aspire to raise their living standards to the levels enjoyed by advanced industrial states (i.e. the USA, Japan, the EU), *citizenship* in a nation-state determines in large part whether individuals or social groups enjoy the benefits of growth that earlier chapters celebrate. Indeed, if one examines the evidence by national state as presented in Table 11.1, there is no doubt at the level of the nation-state who, in fact, are winners and who losers: winners live in South Korea, Taiwan and Singapore and losers tend to live in the Philippines, Indonesia, Mexico and Peru.

What development policies explain success among Asia-Pacific states?

If citizenship and state capacity are necessary, if not sufficient conditions for designating winners and losers, it's not surprising that state strategies and state industrial policies are hotly debated by both development theorists and policy makers in powerful multilateral agencies like the World Bank (Chapters 3, 4 and 6). Typically, development theorists

debate explanations for rapid economic and social development. In the Asia-Pacific, the debate focuses on development policy and exactly how these policies produced winners and losers. In order to frame our analysis of factors and policies responsible for creating winners and losers in the Asia-Pacific, it will help to review the three main explanations for the East Asian NIE's astonishing success: (1) geopolitical, (2) market-conforming and (3) developmental statist (**McGrew and Brook, 1998**, and Chapter 6).

- 1 Geopolitical explanations argue that the strategic and military interests of hegemonic states (the USA) forced them to construct a system that distributed rich financial and military aid to their clients, thus nurturing East Asian elites as a bulwark against Communist expansion. Geopolitical considerations thereby provided emerging NIEs with abundant investment capital, advanced technology and access to international export markets under extremely favourable conditions, i.e. free military and economic aid. Geopolitical approaches are world systemic in scope, emphasizing historically unique or 'conjunctural' factors that intersect at one historical phase. That is, the conjuncture of the Cold War facilitated the emergence of the NIEs. It follows, however, that a conjunctural outcome is difficult to replicate because of different historical conditions, global structures, timing and social structures.
- 2 *Market-conforming* explanations argue that East Asian states achieved 'growth with equity' by implementing market friendly, realistic and pragmatic policies. According to this view, NIEs implemented sound macroeconomic fundamentals (i.e. low inflation, managed debt, real interest rates, market-determined currency values, etc.); promoted private and public sector efficiencies; and fostered competitive discipline among firms through the use of undistorted, world market 'price signals'. Essentially liberal, market-conforming explanations conform to the policy prescriptions of Anglo-American theorists and correlate with the World Bank's policy recommendations.
- 3 *Developmental statist* explanations argue that technocratic state elites—deeply 'embedded' socially and culturally within East Asian societies—planned and implemented long-term industrial planning strategies in

Country		Po	pulatior	1		Edu	cation	Income			
	Popul ation (milli ons); mid- 1994	Popul ation; avg annual growth rate (%) 1990– 94	Life expe ctancy at birth (yrs) 1992	Life expe ctancy at birth (yrs) 1994	Urban popul ation (as % of total) 1992	Adult lite racy rate (%) 1992	Comb ined first, second and third level gross educa tion enrol ment ratio (%) 1992	Real GDP per capita (PPPUSS) 1992	GNP per capita (US\$) 1994	GNP per cap ita avg annual growth 1985– 94	
Mature NIEs											
Korea	44.50 <sup>1</sup>	$0.90^{1}$	71.10	$71^{1}$	77	97.40	79	9,250	8,2601	$7.80^{1}$	
Taiwan	$21.30^2$	$1.00^{2}$	$74.80^{2}$	75 <sup>2</sup>	86 <sup>2</sup>	91.00 <sup>2</sup>	85 <sup>2</sup>	NA	11,597 <sup>2</sup>	$14.20^2$	
Sing apore	2.90 <sup>1</sup>	2.00 <sup>1</sup>	74.80	75 <sup>1</sup>	100	89.90	68	18,330	22,5001	6.10 <sup>1</sup>	
Emerging NIEs											
Malay sia	19.70 <sup>1</sup>	2.40 <sup>1</sup>	70.80	71 <sup>1</sup>	51	81.50	60	7,790	3,4801	5.60 <sup>1</sup>	
Thail and	58.00 <sup>1</sup>	1.00 <sup>1</sup>	69.00	69 <sup>1</sup>	19	93.50	53	5,950	2.410 <sup>1</sup>	8.60 <sup>1</sup>	
Indon esia	190.40 <sup>1</sup>	1.60 <sup>1</sup>	62.70	63 <sup>1</sup>	33	82.50	60	2,950	880 <sup>1</sup>	$6.00^{1}$	
Philip pines	67.00 <sup>1</sup>	2.20 <sup>1</sup>	66.30	65 <sup>1</sup>	51	94.00	77	2,550	950 <sup>1</sup>	1.70 <sup>1</sup>	
China	1,190.90 <sup>1</sup>	$1.20^{1}$	68.50	69 <sup>1</sup>	28	79.30	55	1,950	530 <sup>1</sup>	$7.80^{1}$	
Pacific Latin- America NIEs											
Chile	14.00 <sup>1</sup>	$1.50^{1}$	73.80	72 <sup>1</sup>	84	94.50	71	8,410	3,5201	6.50 <sup>1</sup>	
Mexico	88.50 <sup>1</sup>	$2.00^{1}$	70.80	$71^{1}$	74	88.60	65	7,300	4, 1801	$0.90^{1}$	

Table 11.1 Basic indicators

Peru	$23.20^{1}$	1.90 <sup>1</sup>	66.00	65 <sup>1</sup>	71	87.30	79	3,300	2410 <sup>1</sup>	$-2.00^{1}$
------	-------------	-------------------	-------	-----------------	----	-------	----	-------	-------------------	-------------

<sup>1</sup> Data from *World Development Report 1996*, World Bank, Oxford University Press, New York, 1996.
 <sup>2</sup> Data from *Statistical Yearbook of the Republic of China*, Executive Yuan, Directorate-General of Budget, Accounting, and Statistics. Republic of China, 1996.
 Source: United Nations (1995)

close alliance with national business classes. Together, they shifted their global comparative advantage by export-oriented industrialization. In practice, this means bureaucratic elites, typically entrenched within authoritarian states, restructured, protected and, above all, financed firms who pursued national industrial planning targets. Simultaneously, state elites constructed complementary educational, health and social infrastructure. In effect, this approach argues that East Asian developmental states triumphed because they successfully competed in the world export market by 'governing their internal market, rather than by internalizing the competitive logic of the international market (Wade, 1990).

Despite differences over the relative weight assigned to markets and political institutions, it is reassuring that virtually all analysts now agree that successful NIEs were invariably directed by strong and competent state bureaucracies. Bureaucrats implemented more or less interventionist industrial strategies, co-ordinated investment among firms and state sponsored industries, assessed world market opportunities and, in general, guided firms and citizens toward the fulfilment of national economic 'targets' as well as social developmental goals. Yet, because interventionist, authoritarian bureaucracies are so common elsewhere, developmental theorists have had difficulty answering the question how East Asian NIEs achieved such remarkably *high* rates of growth with such unusually *low* rates of social inequality.

#### How did East Asian NIEs achieve high growth with social equity?

In the early 1990s conflicting answers to the growth with equity question drove Japanese technocrats to lock horns with the free marketeers at the American dominated World Bank. It seems that Japanese policy makers objected when the World Bank tried to force orthodox, deflationary, 'structural adjustment' policies (privatization, currency devaluation, withdrawal of public credit, and elimination of export subsidies) on South-East Asian states as a condition for loans. As Asia's first developmentalist state, Japanese foreign policy makers favoured what worked for Japan: that is, industrial planning, export targets, domestic cartels and, most importantly, subsidized credit to industries targeted by long-term industrial planners. In order to resolve this dispute, Japan agreed to fund a World Bank managed study to resolve which of the two contending policies—developmentalist state planning or market-oriented liberalism—explained the success of the East Asian model of growth with equity. Once published and reviewed in journals, the report produced little consensus. Instead, it generated a high temperature debate about methodology and evidence and the role of states and markets (Fishlow, 1994).

The goal of the World Bank report, *The East Asian Miracle* (EAM) (World Bank, 1993), was to demonstrate unequivocally which policies explained the growth with equity outcome in eight high performing Asian economies (HPAEs) between 1960 and 1990,

i.e. *mature* and *emerging* Asian NIEs (Chapter 1). Interestingly, in spite of theoretical differences, both the authors and critics of the EAM did agree on several issues. First they agreed that, in contrast to the predictions of orthodox growth theory, social inequality did not rise in HPAEs during early industrialization as it had in other world regions (Western Europe, North America, Latin America). In a journal reviewing the EAM, one of its authors (John Page) summarized his interpretation of the unique features of the East Asian experience:

Where the HPAEs are unique is in combining this rapid sustained growth with highly equal income distributions. The HPAEs have been unusually successful at sharing the fruits of growth... The HPAEs have enjoyed much higher per capita income growth while income distribution has been more equal than in other developing economies. Moreover, the fastest growing East Asian economies, Japan and the 'four tigers' [South Korea, Taiwan, Singapore and Hong Kong] are the most equal.

(Page, 1994, p. 617)

Applied to our topic of winners and losers, therefore, the evidence confirms a correlation between developmentalist statist industrial strategies and achieving high growth with equity. The EAM confirms that the greatest income equality was found along those independent states (excluding colonial Hong Kong) which pursued the greatest degree of state planning and developmentalist industrial policy i.e. Japan, South Korea, Taiwan and Singapore, i.e. *mature* NIEs. Yet variation and complexity was recognized also. Despite the claim of orthodox economic theory for universal explanations, the EAM concludes that no single abstract model explains the diversity of outcomes, nor was any universal explanation for high growth with social equity discovered. Instead, the authors emphasized that specific social structural features of each country and the timing and sequence (i.e. the conjuncture) of industrialization shaped the outcome in each national state.

We found that the diversity of experience, the variety of institutions, and the great variation in policies among the high-performing Asian economies means that there is no [single] East Asian model of rapid growth with equity. Rather each of eight economies we studied used various combinations of policies at different times to perform the functions needed for rapid growth,

(World Bank, 1993, p. 366)

Recognition by the World Bank, both of the power of developmental statist policies as well as the diversity of institutional approaches used to achieve growth with equity, is a truly remarkable admission from an agency known to be hostile to developmental statist strategies. Figure 11.1 presents the correlation between income inequality and growth of gross domestic product from 1965–89 from the EAM report. The upper left hand quadrant marks the spot for 'winners' as well as the site for most of the developmentalist states in the region. Active state intervention, therefore, is correlated with the highest rates of growth and widest participation in the fruits of that growth.



Note: Income inequality is measured by the ratio of the income shares of the richest 20 per cent and the poorest 20 per cent of the population. A higher figure indicates greater inequality.

# Figure 11.1 Income inequality and growth of GDP, 1965–89

Source: World Bank (1993, p. 31)

#### 11.3 The distribution of Asia-Pacific success: key questions

Though the evidence confirms that state development policy forms the matrix from which national winners and losers emerge, the aggregate statistics in the tables and charts presented above, tell us very little about the internal distribution of resources, opportunities and the 'quality of social life' (personal security, rule of law and

environmental justice) within Asia-Pacific societies. To clarify the experience of groups and individuals within these societies, three central questions guide our analysis:

- 1 What is the internal distribution of income, opportunity and poverty and how has this distribution created winners and losers?
- 2 How do these distributions of income, poverty and socio-political access to opportunity overlap with, and thus cut across genders, ethnic groups and regions within national states?
- 3 When confronted by distributional effects, arbitrary state power, breakdowns in the rule of law and environmental degradation, to what degree are citizens able to petition and redress their grievances?

Given the population, educational achievements and income levels presented in Table 11.1, answering these questions suggests that it will be helpful to construct four 'clusters' of countries: (1) mature NIEs—South Korea, Taiwan and Singapore; (2) almost NIEs—Malaysia, Thailand, Indonesia and the Philippines; (3) China—mainland China and Hong Kong; and (4) Pacific Latin America—Chile, Mexico and Peru. Latin American countries are included mostly for comparison but also to raise the question whether the East Asian NIE model can be replicated elsewhere. Moreover, because the timing of each cluster's shift to *export-oriented industrialization* is sequential over time—beginning with the mature NIEs to Pacific Latin America—one can assess how winners and losers emerged from the combined effects of the historical conjuncture, state development policy and market forces.

#### 11.4 Social class issues: income distribution and poverty reduction

#### Mature NIEs: South Korea and Taiwan

Let us begin with South Korea and Taiwan, two 'iconic' NIEs, which share a common history of Confucian bureaucratic authority, the socio-economic restructuring induced by Japanese colonialism, a dependent geopolitical relationship to the USA, comparative ethnic and national homogeneity and an extremely thorough land reform programme. The conjuncture was indeed, unique. After the Second World War, Japanese state agencies and private firms withdrew, yet left intact a rationalized and commercial agricultural system, a transport and communication infrastructure, a war-induced import-substitution industrial sector and a fierce sense of anti-imperialist (later anti-Communist) nationalism. Note that while US military agencies distributed economic and military resources, few American international firms invested. Hence, in contrast to Latin America and South-East Asia, South Korea and Taiwan's political and military elites possessed unrivaled autonomy from the pressures exerted by international firms as well as from an antiindustrial agrarian landlord



A winner? Even in the street a young Kowloon office worker is constantly in touch by mobile phone and pager

class. This autonomy allowed these elites relative freedom to implement an export oriented industrialization strategy.

Because both South Korea and Taiwan emerged from Japanese colonialism (1945) as overwhelmingly peasant societies, land reform was the foundation for growth with equity, largely because it benefited both upper and lower classes. In South Korea, the land reform programme embraced a 'land to the tiller' policy which outlawed tenancy, encouraged landlords to sell land to tenants at very favourable prices and set an upper limit of approximately 3 hectares per household. Between 1945 and 1965 the proportion of farm households which wholly owned their land rose from 14 to 70 per cent, and the proportion of pure tenants fell from 49 per cent to 7 per cent.

In Taiwan, both pressure form US advisors and an auto-critique of the KMT's failed agrarian policy on mainland China transformed the peasantry from tenant farmers to small property owners with highly commercialized linkages to export markets and small scale industries. In sum, rents were capped; land holdings were limited to 3 hectares; and state agricultural services provided subsidized fertilizers and modern inputs. 'By 1973, almost 80 per cent of the agricultural population consisted of owner-cultivators and another tenth of part owners. Only 6 per cent of farm income accrued to landlords and money lenders' (Amsden, 1985, p. 85).

While the land reform's benefit to the peasantry is obvious, the landlord class also gained accumulating opportunities. In exchange for their loss of landed property and tenant rents, the South Korean landlords either received payments from the peasantry or bonds from the government which could be exchanged for the purchase of industrial facilities recently vacated by the Japanese colonialists. Bonds were also sold for a portion of their face value and, in general, became an asset for rising entrepreneurial classes, of both landlord-origin and commercial-origin. In Taiwan, landlords were compensated by directed payment, land bonds and/ or stock in public enterprises, often based on former Japanese assets. In both South Korea and Taiwan, therefore, the land reform stabilized the countryside, provided cheap food to the emerging urban manufacturing sector and contributed to the formation of accumulating entrepreneurial classes. Hence, the geopolitical conjuncture was a foundation for land reform programmes that contributed to growth with equity.

Tables 11.2 and 11.3 illustrate how land reform together with other state-led redistribution policies, i.e. universal primary education and health services, constructed a comparatively equal income distribution, low poverty rates and low infant mortality rates. This provided South Korea and Taiwan with comparatively egalitarian social structures prior to export-oriented industrialization. Note also the difference between them. The highly authoritarian, centralized state and highly concentrated industrial firm structure of South Korea, i.e. the chaebol and the more centralized state financial and industrial policies, correlate with a trend toward rising income *inequality* over time reflected in higher 'Gini' coefficients (i.e. from 34 in 1964, to 40 in 1988). In contrast, Taiwan's reliance on small-scale industries and more decentralized state financial and industrial intervention correlates with rising income equality and lower 'Gini' coefficients over time (i.e. 36 to 31) ('Gini' coefficients are discussed in Box 11.1). Finally, the urban bias of South Korea's centralized and concentrated industrial model and the comparative absence of rural industries and infrastructural funding since the 1960s explains a general decline in the condition of the peasantry. In fact, rural incomes lagged behind urban incomes, indebtedness increased as land prices rose and migration to cities spiralled. More alarmingly for inequality, by 1984 63.2 per cent of South Korea's farmers were tenants and by 1986, 30 per cent of farmland was in tenancy as compared to 5 per cent for Taiwan.

### Box 11.1 Measuring the equitable distribution of income: the Gini coefficient

The Gini coefficient is used to measure the distribution of income according to the correlation between the amount of income received by each quintile (one-fifth or 20 per cent) of a population. If each fifth of the population from the top fifth to the bottom fifth received exactly one-fifth of a national income then the coefficient would be zero, i.e. perfect equality of income distribution. In so far as the Gini coefficient departs from zero, income inequality is said to increase. Typically, the Gini coefficient varies from about 0.25 to 0.70. The Gini coefficient is based upon a curve called the Lorenz curve.

	Pov	verty	Life cl	hances	Safe Water						
	% Urban poverty 1990	% Rural poverty 1990	Infant mortality rate (per 1,000 live births) 1960	Infant mortality rate (per 100 live births) 1992	% of total population with access to safe water 1980	% of total population with access to safe water 1993	Rural 1988– 93				
Mature NIE	Ēs.										
Taiwan	NA	NA	32 <sup>3</sup>	62	67 <sup>2</sup>	86 <sup>2</sup>	NA	NA			
Singapore	NA	NA	36	6	$100^{1}$	$100^{1}$	NA	100			
Emerging N	/IEs										
Malaysia	8	23	73	13	$80^{1}$	$78^{1}$	66	96			
Thailand	7	29	103	37	66 <sup>1</sup>	NA	72	87			
Indonesia	20	16	139	58	32 <sup>1</sup>	42 <sup>1</sup>	43	68			
Philippines	40	54	80	44	54 <sup>1</sup>	81 <sup>1</sup>	79	85			
China	NA	12	150	44	NA	71 <sup>1</sup>	60	99			
Pacific Lati	in-America	ın NIEs									
Chile	NA	NA	114	16	85 <sup>1</sup>	86 <sup>1</sup>	75	98			
Mexico	23	43	92	36	74 <sup>1</sup>	$78^{1}$	66	94			
Peru	52	72	142	64	54 <sup>1</sup>	58 <sup>1</sup>	18	75			

 Table 11.2 Poverty and life chances

<sup>1</sup> Data *from World Development Report 1996*, World Bank, Oxford University Press, New York, 1996.

<sup>2</sup> Data from *Statistical Yearbook of the Republic of China 1996*, Executive Yuan, Directorate-General of Budget, Accounting, and Statistics, Republic of China, 1996.

<sup>3</sup> Data from *Statistical Yearbook of the Republic of China 1977*, Executive Yuan, Directorate-General of Budget, Accounting, and Statistics, Republic of China, 1977.

NA=Data not available

Source: United Nations (1995)

		Distr	ibution	n of inc	ome or	consun	nption						
	Sur vey year	Low est 10%	Lo west 20%	Sec ond quin tile	Third qui ntile	Fou rth qui ntile	High est 20%	Hig hest 10%	GINI index— year	GINI index	GINI index— year	GINI index	Chan ge in GINI index
Mature	NIEs												
South Korea	1988	NA	7.40	12.30	16.30	21.80	42.20	27.60	1964 <sup>3</sup>	34 <sup>3</sup>	1988 <sup>3</sup>	40.00 <sup>3</sup>	6.00
Taiwan	1995 <sup>2</sup>	NA	7.30 <sup>2</sup>	13.00 <sup>2</sup>	$17. 40^2$	23.40 <sup>2</sup>	39.00 <sup>2</sup>	NA	1964 <sup>3</sup>	36 <sup>3</sup>	1990 <sup>3</sup>	31.00 <sup>3</sup>	-5.00
Singa pore	1982– 3	NA	5.10	9.90	14. 60	21.40	48.90	33. 50	1966 <sup>3</sup>	SO <sup>3</sup>	1989 <sup>3</sup>	$49.00^{3}$	-1.00
Emergin	ng NIEs												
Mala ysia	1989	1.90	4.60	8.30	13.00	20.40	53.70	37.90	1968 <sup>3</sup>	SO <sup>3</sup>	1989 <sup>1</sup>	48.40 <sup>1</sup>	-1.60
Thai zland	1992	2.50	5.60	8.70	13.00	20.00	52.70	37.10	1965 <sup>3</sup>	41 <sup>3</sup>	1992 <sup>1</sup>	46.20 <sup>1</sup>	5.20
Indon esia	1993	3.90	8.70	12.30	16.30	22.10	40.70	25. 60	1976 <sup>3</sup>	34 <sup>3</sup>	1993 <sup>1</sup>	31.70 <sup>1</sup>	-2.30
Phili ppines	1988	2.80	6.50	10.10	14. 40	21.20	47.80	32.10	1961 <sup>3</sup>	49 <sup>3</sup>	1988 <sup>1</sup>	40.70 <sup>1</sup>	-8.30
China	1992	2.60	6.20	10.50	15.80	23.60	43.90	26.80	NA	NA	1992 <sup>1</sup>	37.60 <sup>1</sup>	NA
Pacific I America													
Chile	1994	1.40	3.50	6.60	10.90	18.10	61.00	46.10	1969 <sup>4</sup>	$SO^4$	1994 <sup>1</sup>	56.50	6.50
Mexico	1992	1.60	4.10	7.80	12.50	20.20	55.30	39.20	1969 <sup>4</sup>	57 <sup>4</sup>	1992 <sup>1</sup>	50.30	-6.70
Peru	1994	1.90	4.90	9.20	14.10	21.40	50.40	34.30	1961 <sup>4</sup>	59 <sup>4</sup>	1994 <sup>1</sup>	44.90	-14.10

Table 11.3 Income distribution and GINImeasures

<sup>1</sup> Data from World *Development Report 1996*, World Bank, Oxford University Press, New York, 1996.

<sup>2</sup> Data from *Statistical Yearbook of the Republic of China 1996*, Executive Yuan, Directorate-General of Budget, Accounting and Statistics, Republic of China, 1996.

<sup>3</sup> Data from Asia-Pacific Economies, Islam, lyanatul, and Anis Chowdbury, Routledge, New York, 1997.

<sup>4</sup> Data from Kamisaruk, C. and Ortega, J.G. (1996) *Statistical Abstract of Latin America.*, Los Angeles, UCLA Latin-American Publications.

NA=Data not available

Source: United Nations (1996)

Mature NIEs: winners and losers under export-oriented industrialization

Let us now turn to the winners and losers during the period of rapid industrialization, beginning in the late 1950s for Taiwan and early 60s for South Korea, a time when rising productivity and economic growth raised the income of virtually all citizens. In South Korea, close collaboration among bureaucrats and firms meant that political access became an important source of finance, technology and market access, mostly because the South Korean state effectively controlled all financial resources and technology choices. Corruption was widespread as a bevy of prosecutions confirm, but did not block rapid growth and accumulation. Military officers, bureaucrats and chaebol owners were the clear winners during the 60s, 70s and early 80s. Engineers in export-oriented manufacturing were paid well but their income share only averaged four times that of workers, a comparatively low ratio when compared to NIEs like Brazil. At the same time, industrialization has transformed the class structure of South Korea. A survey undertaken in 1991 shows that 34 per cent identify themselves as lower middle class, 33 per cent as middle class and only 26 per cent as lower class (Cotton and Hyung-a van Leest, 1996, p.188). Since the shift to democracy, politics have been transformed and the position of the middle and skilled working classes strengthened. And a rentier, consumption-oriented urban elite, specializing in real estate speculation and the underground economy has also expanded significantly.

Unlike Taiwan, South Korea's transition to electoral democracy (1987) was driven by endemic and turbulent social conflict: urban demonstrations, national strikes, military interventions, confrontations led by militant students and workers and regionally-based opposition. Because these movements served as the midwife for electoral democracy, their grievances superbly illuminate the perspective of the losers who paid the bill for South Korea's miraculous growth. Under the dictatorship (1961–87) South Korean workers were forced to work the longest hours recorded by any nation in the world, typically 12 hour days averaging about 54 hours per week. Free democratic trade unions were banned; church affiliated support groups were repressed as well. Security agents infiltrated worker's organizations and enforced state corporatist ideology within large factories. Bello and Rosenfeld summarize the condition of South Korean workers succinctly:

At the beginning of 1987, before aggressive unionism brought about the big wage gains of the three years, the hourly pay of Korean manufacturing workers was 11 per cent of United States workers, 14 per cent of Japan's, 75 per cent of Taiwan's, and 80 per cent of Hong Kong's... Korea has the world's highest rate of industrial accidents with an average of 5 workers killed daily and another 390 injured. The country also leads the world in the rate of occupation-related illnesses...

(Bellow and Rosenfeld, 1990, pp. 24–5)

Not surprisingly, from such conditions there erupted a volcanic wave of strikes and violent confrontations against both state and *chaebol* auth-orities. Mobilized by a populist, nationalist, multi-class social and cultural movement—the *Minjung* movement—South Korean workers joined with students and middle-class oppositional groups to demand democratic reform. In doing so they constructed social solidarity, a culture of opposition and even a South Korean version of liberation theology. Between

1986–89, the Ministry of Labour recorded over 900,000 new union members. In 1987 alone, recorded strikes numbered 3,617 and person days lost reached 6,964,900. Real industrial wages rose especially among unionized workers. And, quite unexpectedly, the Minjung movement's alliance with labour has endured. For, despite the efforts of an alliance of *chaebol* leaders and politicians to reduce the security and social benefits of South Korean workers in 1996 (passed during a secret late night session of parliament), strikes led by militant trade unionists successfully forced the government to back down, thereby limiting the power of South Korea's winners.

Let us examine why Taiwan's income distribution is so stellar that it has become a middle-class democratic society effectively eliminating poverty (see Tables 11.2 and 11.3). Taiwan's post-war social and political structure is one where the Mandarinspeaking mainlanders dominated the political apparatus and pivotal state industries (petroleum, fertilizer, steel and industrial innovation) and Hokkien and Haka-speaking Taiwanese controlled agriculture and small-scale industry. Representing two-thirds of industrial output at its height, the small and medium industrial sector (SMI) operates in hierarchical networks of subcontracting firms geared to exports, located in specialized industrial districts geared to maximize flexibility by organizing themselves through family relationships (see Chapter 9). In practice, central factories develop networks of satellite subcontracting networks in specialized industrial districts integrated by networks of family and even household level firms. Simultaneously, the central government targets export markets, creates incentives for upgrading and builds technical infrastructures or semi-public enterprises to support the central and satellite factory networks. All of which acts to redistribute income and opportunity, reduces intrusive administrative cost while reinforcing Chinese notions of familial authority and generates a foreign exchange surplus estimated at US\$90 billion (1995).

To briefly review the winners, Taiwan's industrial structure and redistribution policies created a diverse middle and skilled working class. State employment, affiliation with the KMT and officialdom constituted the older middle and upper class. As the industrialization process unfolded, a new middle class demanding an end to authoritarianism, democracy and new styles of consumption arose. According to Chu:

the new middle class is made up of the professions and the owners of small and medium sized businesses. They were the main initiators of the opposition movement and the key catalysts for the great 1987 political liberalization. With the application of labour-intensive technology under the export-led industrialization policy, numerous affluent workers who experienced an improved standard of living with overtime pay were spawned.

(Chu, 1996, p. 208)

The statistics in the tables confirm that Singapore qualifies as a mature NIE because of its growth rate, income distribution, health and education levels and technical achievements, all of which were promoted by a highly rationalized authoritarian state. Nonetheless, because it is a multiethnic city with a population of under 3 million, is a former British colony and was temporarily a member of the Malaysian federation, depends largely on investment by international firms rather than national capitalists, it shares more with the

emerging NIEs of South-East Asia. Currently Singapore serves as a regional growth pole, high tech hub and financial service centre for that region.

#### 11.5 Emerging NIEs: Malaysia and Indonesia and the South-East Asian model

In contrast to South Korea and Taiwan, South-East Asia's almost NIEs experienced no severe rupture with their former imperial overlords after 1945, nor was a thorough land reform implemented. Social structural continuity between the winners and losers is the norm. Aristocratic classes and their commercialized urban allies dominate the political parties and societies of Malaysia and Thailand; a military corporate group (ABRI) leads a similar coalition in Indonesia; and an alliance of agrarian landlords and urban oligarchs rules the Philippines regardless of whether dictatorship or democracy is in vogue. True, the power and competency of technocratic developmentalist state bureaucrats has increased, but fails to reach the discipline and competency witnessed in South Korea or Taiwan. All South-East Asian emerging NIEs have abundant natural resources so that plantation crops and raw material exports (petroleum, rubber, rice, palm products, lumber, tin) constitute a powerful export interest which constrains the ability of state-technocratic planners to pursue industrialization.

Nationalism is an important resource in the cases of Japan, South Korea and Taiwan. The last mentioned country's near ethnic homogeneity became a valuable resource for political elites who mobilized nationalism to reach developmentalist state goals. Most importantly, common ethnonationalist and cultural and background enabled an 'embedded' alliance between political-bureaucratic and entrepreneurial-business elites to emerge, one based on a common culture of trust, marriage and mutual interest. In contrast, the social structure of South-East Asia's emerging NIEs is much more fragmented and therefore nationalist mobilization and elite integration is far more difficult to actualize. Indeed, ethnic heterogeneity is very high, especially in Indonesia, so that regional, ethnic and religious identities (especially Islam) must be recognized in order to construct even a fragile developmental coalition in pursuit of industrialization.

History matters here as well. The historical legacy of large-scale Chinese immigration from highly commercialized centres into South-East Asia, especially since the eighteenth century, combined with the 'racialized' nature of European colonial polices which divided communities, means that an ethnic division of labour arose: Chinese dominated urban commerce and capital accumulation in general and the indigenous groups controlled the state. To be sure, variation is great. In the Philippines and Thailand, common religious institutions, i.e. Catholicism and Buddhism respectively, allowed much social integration especially among the urban elite and professional classes. Elsewhere like Malaysia and Indonesia, the communal requirements of Islam and Chinese culture do not allow for social integration, so that alliances, typically based on exchanging political access for commercial expertise and capital, are not based on trust nor ethno-national cohesion.

Hence, unlike South Korea and Taiwan where the pivotal alliance is between the state and national firms, in South-East Asia the dominant alliance is between multinational firms and state based politicians and technocrats. Foreign direct investment (FDI) originally came into the region from the USA and Western Europe. However, with the rise of Japan, South Korea and Taiwan, and the subsequent revaluing of their currencies upwards, especially the Japanese yen, the mature NIEs have shifted labour intensive and increasingly medium-level, high technology industries (i.e. auto-assembly, chip assembly, low level design and increasingly wafer fabrication) to the emerging NIEs of South-East Asia (Chapters 2, 10 and 14). Singapore led the way by creating state-sponsored innovation alliances with international electronics firms. Malaysia and Thailand now follow in Singapore's wake. Local accumulating groups are either members of the immigrant Chinese community, fledgling indigenous entrepreneurs or officials investing the 'rents' they have gleaned from public contracts or from selling political access. Winners, therefore, are aristocrats, military officers, politicians, high public servants and an emerging professional and entrepreneurial class mostly located in services (finance, housing, hotels, tourism, commerce) rather than in innovative industrial production.

Yoshihara, presenting a structural-dependency viewpoint, sums up the limits of South-East Asian capitalism by emphasizing its 'intractable problems':

One [problem] arises from a low level of technology. Even though past industrialization looks impressive on the surface, the level it has reached is not commensurate with the region's own level of technology. Essentially, South-East Asia has relied on foreign companies to make up for its technological deficiency. This has produced such technology dependent capitalists as compradore capitalists, and has also allowed foreign capital to come in, in order to generate new exports... Technological backwardness, the low quality of government intervention, and Chinese discrimination are the three most difficult problems afflicting the capitalism of South-East Asia... Despite industrial progress, its role has been far from dynamic. Rather, the capitalism that has emerged in South-East Asia is a new brand: ersatz capitalism.

(Yoshihara, 1988, pp. 130–1)

Malaysia: poverty, redistribution and ethnic alliances of accumulation

If one examines state industrial policy since Prime Minister Mahathir bin Mohamad assumed office in 1981 and initiated his 'Look East policy', three developmental policy outcomes are evident:

- 1 a highly protected, state-sponsored auto, steel and cement industry controlled by the Malays on the South Korean model;
- 2 a recent push to develop technology, linkages and human capital so as to relieve labour shortages and to drive manufacturing and information technology up the 'value chain'; and
- 3 a long-term reliance on foreign manufacturing and R&D investments from international electronics firms located mostly in Free Trade Zones in virtual imitation of the Singaporean strategy.

Since 1987, after surviving a debt driven crisis and over-extension of the state sector, Malaysian industrial policy has focused on attracting high technology investments and developing linkages between international firms and small and medium sized enterprises (SMEs). Unemployment, estimated at 17 per cent in 1970, disappeared in the 1990s with the result that between 2 to 2.5 million foreign workers now work on the plantations, construction sites and low wage occupations.

In terms of political accountability, a limited parliamentary democracy is dominated by a coalition of ethnically based political parties, dominated by the United Malay Nationalist Organization (UMNO). The Chinese and Indian ethnic groups are also part of this national front. UMNO's organizational network reaches from the capital to the village level. It is a source of patronage, contracts, rents and upward mobility for the Malays who were formerly the losers. Within UMNO competition for office is intense, but in the national elections the system is rigged in favour of rural communities, the UMNO coalition and the elite ethnic alliance for accumulation. Despite being a semiparliamentary democracy, elections are clean, office holders can be voted out and policies rejected. In theory, courts are independent. In practice, though sometimes asserting their independence and favouring the opposition, they are props for the UMNO-dominated coalition. Because Malays are not an overwhelming majority, and because Malay oppositional parties also compete for votes, UMNO must rely on Indian and Chinese parties to hold the two-thirds majority they need to amend the constitution. Hence, bargaining and negotiation among ethnic elites is the norm. The outcome is a mixture of authoritarianism and controlled democratic participation.

As indicated earlier, Malaysia's ethnic diversity separates the political and economic elites and thus poses a challenge to economic policy makers. Malaysia's solution is worth reflection for resolving ethnic cleavages in multi-ethnic states. After ethnic rioting in May 1969, an affirmative action policy called the New Economic Policy (NEP) was introduced with the objective of abolishing absolute poverty, redistributing equities among Malaysian citizens and restructuring the ethnic division of labour by creating a Malay (Bumiputera or Bumi) professional, managerial, and industrial class as well as an urban middle class from a largely peasant constituency. If defined as interventionist state policy, designed to restore social peace and co-ordinate historical imbalances in opportunity, then the NEP has to be seen as a successful redistribution policy for the losers. It has preserved social peace while sustaining foreign and local investment and a high rate of economic growth, i.e. since 1987 exceeding 8.5 per cent per year.

The NEP legitimated state intervention into the economy by reducing absolute levels of poverty from 49.3 per cent in 1971 to 15 per cent by 1990. Among the Bumis (i.e. Malays) poverty was the highest at nearly two-thirds in 1970, but was reduced to 20.8 per cent in Peninsular Malaysia by 1990, and to about 13 per cent today. Hence, analogous to land reform in the two NIEs, the NEP was a successful income redistribution mechanism which, together with educational and other investments, facilitated the urbanization of the Malay peasantry from agrarian to urban-industrial life. This is no mean feat, if Malaysia is compared to Sri Lanka, Pakistan or Indonesia.

A second objective of the NEP was to restructure the ethnic division of labour such that ethnicity was no longer identified with occupation or sector. The government's evaluation of the NEP indicates that substantial progress was made through state agencies that provided credit, quotas, contracting opportunities and innumerable other
opportunities to the Malays and other 'indigenous' ethnic groups in the East Malaysian states of Borneo and Sarawak as well as the indigenous tribal groups (i.e. excluding the Chinese and Indians). Given the unequal access to political opportunity, the distribution of income within the Malay community has become more unequal but, on balance, the overall economic situation of the impoverished Malay peasantry has improved immensely. State expenditure on education, housing and health have created a large middle class, a relatively affluent working class and very little overt poverty. More surprising is the rise of a new Malay professional and capital accumulating class, originally closely linked to UMNO, but not technically competent and therefore much less dependent on patronage and rents.

A third objective of the NEP, the restructuring of corporate equity, dealt more directly with the heart of Malaysian capitalism and problems posed by the ethnic division of labour. By 1990, ownership of foreign equity shifted from 63.3 per cent in 1970 to 25.1 per cent; an outcome achieved through the intervention and management of the Malaysian state elite. Chinese and Indian shares of equity actually increased during this period from 32.2 per cent to 46.2 per cent; an increase of 43 percentage points and 6.2 per cent above their target. Income data exhibits the same pattern. For despite privileging the Malay and redistributing opportunity and rents to them, Chinese income rose faster and higher than did the Malays, largely due to demand for their skill, capital and business acumen during a period of 30 per cent, but registered an enormous increase from the 2.3 per cent registered in 1970. A controversy exists regarding the ownership of the nominee shares, for critics believe that the nominees are most likely owned by Bumis who wish to hide their rentier accumulations.

To summarize, while Yoshihara's evaluation applies to Malaysia, recent developments suggest that technology is deepening in the electronics sector, the internationalization of production does offer technology transfer opportunities to local firms and state support for SMEs is making some headway. The income distribution policies have not reduced income inequality, but absolute poverty, a labour shortage and the expansion of education for both men and women have transformed opportunity. The real losers, however, are the peasants who are unable to take advantage of urban opportunity, political patronage and modern education. Similarly, the indigenous ethnic groups, often displaced by dams, logging and plantation agriculture have also paid a very high price for Malaysia's qualified success.

### Indonesia: diversity, patrimonialism and paradox

The sheer geographical scale and large population of Indonesia challenge one to make meaningful generalizations about winners and losers or growth with equity. Over 13,000 islands stretch for more than 5,000 kilometers East to West upon which reside a population approaching 200 million citizens who identify with at least 300 distinct ethnic groups and speak even more languages. About 60 per cent of the population lives on the central island of Java. Divided into 27 provinces, the high degree of ethnic diversity, religious pluralism (Muslim, Christian, Buddhist, Hindu, indigenous) and radically different cultures pose a governance problem of truly Herculean dimensions. Since the bloody overthrow of the populist regime of Sukarno in which Chinese and leftist were

slaughtered (1965–67), General Suharto has ruled through an alliance of mostly Javanese military officers, civilian technocrats and loyal clients—termed the 'New Order'. The military (ABRI) is a corporatist, technocratic and administrative institution which not only has 100 parliamentary seats reserved for officers, but also administers government monopolies, holds a large share of directorships of state industries and even holds provincial and lower level administrative posts. National security and the fear of chaos arising from a bubbling cauldron of class, ethnic and religious tensions are trotted out to justify these privileges. Patronage, corruption, military repression of dissent and the appropriation of public national resources to shore up support for the regime has been normal for over three decades.

In Indonesia, patrimonialism overrides any developmental theory informed by developmental statism or liberal market deregulation. Policies are inconsistent, capricious, often overturned or not enforced, depending on the status and connections of the parties. The rule of law and public accountability are exceedingly weak. Instead, 'cronyism' measured by the intimacy of ones' connection to the Suharto family govern access to monopolies, tariff protection and shares of profitable state-controlled industries. The rape of Indonesia's natural resources, inefficiencies flowing from state control over the economy and the social tensions generated by the resentment of the beneficiaries of this kleptocratic regime define the parameters of Indonesia's winners and losers. Examine Box 11.2 where the media documents the absurdity of two highly protected auto assembly companies, each connected to a different son of President Suharto, which only raises the cost of automobiles for Indonesian consumers.

### Box 11.2 2 Suharto Sons In Car Business

JAKARTA, Indonesia, June 5 (AP)—The Hyundai Motor Company of South Korea signed an agreement today to build cars here with two companies controlled by one of President Suharto's sons.

Hyundai has a half stake in the venture, PT Bimantara Indonesia. The other half is held by PT Cakra Nusa and PT Asriland, both headed by Mr. Suharto's son Bambang Trihatmojo. The new company will produce cars in competition with the Timor, Indonesia's so-called national car being built by another Suharto son, Hutomo Mandala Putra, known as Tommy, in partnership with Kia Motors of South Korea. On Wednesday, the Suharto Government announced that all Government agencies would buy the national car.

Bimantara Hyundai Indonesia said it would build a factory to produce sedan body and engine components.

The Timor has been at the center of a trade dispute with American, Japanese and European makers. It is exempt from duties and taxes that can add 50 percent to the price of a car bought in Indonesia.

**Associated Press.** 

#### Poverty, inequality and income distribution

Given this consensus regarding the legitimacy of the regime, patterns of endemic corruption and the competency of the public service, it is paradoxical that Indonesia's official statistics, usually supported by the World Bank's researchers, register an astonishingly equal income distribution, and low poverty rate, i.e. a Gini index of 31.7 in 1993 (Table 11.3). Hence, according to the World Bank, Indonesia is tied with Taiwan. Sceptics note that other measures of poverty and equity such as access to safe water, infant mortality and maternal mortality do not correlate with the rosy numbers put out by the government nor the World Bank. Others argue that the bottom-level measure for poverty is excessively low and understates rural poverty. Observers note far more poverty than official statistics register, but there are reasons why earlier policies may have reduced poverty and lowered inequality especially in rural areas.

Despite the rise of export-oriented manufacturing, nearly two-thirds of Indonesians still live in rural areas. Hence government agrarian policies that disseminated the fruits of the 'green revolution' among small scale rice farmers probably raised living standards and reduced poverty. Land ownership has become more concentrated in Java, but economies of scale do not operate in rice production. Together with rural infrastructure, new irrigation systems, a competent family planning programme and market reforms for rural producers in the 1970s, new technologies of rice production made a real difference in poverty reduction. Yet note the consequences of the industrial transformation of Indonesia. By 1990, the contribution of agriculture had declined to a 20 per cent share from 45 per cent in 1970, while manufacturing had risen from 8 per cent to 19 per cent. Unfortunately, the positive effect of agricultural reform on poverty reduction and income inequality diminishes as urban-industrialization rises for it is the urban population who feels excluded and restive for change.

The other paradoxical success of Indonesia arises from the fact that among OPEC states, it adjusted to the price collapse of the 80s far better than any other oil exporting state. The liberalization programme begun in the 80s, to some degree reformed the monopolies, the irrational taxation systems and the financial system especially foreign exchange. However, between 1988 and 1990, Indonesia's foreign debt rose to US\$80 billion, raising its debt service ratio to 32 per cent of its exports. In response the technocrats deregulated financial practices and liberalized opportunities for business. If one recalls that approximately 70 to 80 per cent of the population identifying themselves as Chinese, one can grasp the limits on bureaucratic reform in Indonesia. Unlike Malaysia, no publicly sanctioned ethnic alliance of accumulation nor redistribution programmes exists to buffer the racial tension generated by deregulation of the economy. Space does not permit a discussion of how minority groups, like the Timorese's human rights or the indigenous shifting agriculturists and hunter-gatherers have been excluded and declared losers by the Javanese dominated regime.

Whatever the World Bank's statistics may suggest, the process of deregulation has combined with resentment of Suharto's 30 year reign of patronage and repression. The excluded—the urban poor, the new Islamist reformers and the Westernized middle class—are now clamouring for accountability and electoral reform. The excluded were enraged by the government's exclusion of Sukarno's daughter from contesting the 1997 election, by their repression of labour and democratic activists and by the violence that erupted from the rigging of elections in Java and Kalimantan. Both Indonesia's winners and losers await the succession crisis that looms after the passing of Suharto, who in 1997 was over 75 years old.

## 11.6 China: the triumph of 'market Leninism' and new social inequalities

Few reversals in modern history rival the shift, in governmental approach, which began in 1978 with the de-collectivization of agriculture, than the economic liberalization of China. The Communist party remains intact, shorn of its revolutionary vigour, yet actively co-ordinating the many new and older segments of the economy. In agriculture peasant resistance to centralized control, comparatively low output and a preference for family controlled plots eventually gave birth to the responsibility system. Peasants now exercise *de facto* ownership of land, production and marketing are carried out by the household and taxes are paid to local authorities to cover social services and welfare needs of the indigenous population. Output doubled in the decade after decollectivization of agriculture; production diversified into specialized crops; and nonfarming activities increased significantly. Few would argue that the responsibility system was not popular with the peasantry, nor that increased output and control raised their living standards. The question remains whether the new system is endurable as an environmentally sustainable quasi-market system and whether the income equality will be sustained over time.

The large state-managed industrial sector is modelled after that of the Soviet Union with little incentive to respond to new market demand, innovate with new technologies or increase its efficiency. Debt, losses and demand for their products is declining so that they are becoming a politically sensitive yet nonetheless heavy drag on the financial system. Because price reform is limited, materials can be wasted without cost to the management. Yet, reforming a large state sector, much of which does not produce goods in demand or at competitive prices will be very difficult. Efforts to reform industries and close down the worst were met by rioting and industrial conflict in 1997 in Sichuan province. Hence, given the isolation from encroaching world market forces, the reform of these industries promises to be conflictual and difficult for employees and administrators. Nevertheless, in late 1997 the Chinese authorities promised a large-scale privatization of these industries.

Media attention has centred on multinational foreign investment in the coastal provinces of China. Here the brand names of Motorola, Chrysler, and Sony are produced in special economic zones like Shenzhen, north of Hong Kong. It is important to realize that at least 60 to 70 per cent of all investment on the mainland comes from overseas Chinese, especially Hong Kong and Taiwan. Given this level of Chinese investment, one should not be surprised that a subcontracting system has emerged.

Under the label of township and village enterprises (TVEs), a network of manufacturing enterprises has arisen, one co-ordinated with foreign trade corporations and by local officials. From 1984 to 1990 the value of TVE exports rose from US\$2.38

billion to US\$12.5 billion, or 20 per cent of total exports. By 1989, the number of TVEs rose to 18.6 million employing over 93 million workers. Of course, this transformation illustrates the internationalization of eastern China's rural economy.

Winners and losers since the reform

Those who have benefited most from the reforms constitute a new class of entrepreneur or owner operators who do not aspire to gain political office and show a reluctance to join the CCP. The Hong Kong cultural style is hegemonic for this emerging elite. Goodman's description is valuable:

The most visible of the new economic elites are the owner operators, private entrepreneurs who have developed their own businesses. These owner operators have been a new and key feature of the reform era. Originally they were regarded by the Chinese Communist Party (CCP) as small scale entrepreneurs who could be mobilized to meet the demands not easily met by the planned economy... As the state withdrew from direct economic management of enterprises and market reforms were introduced owner operators of larger enterprises emerged in a wider range of activities, notably light industry.

(Goodman, 1995, p. 134)



Losers? Workers from China's poorer provinces hoping to find work in Shanghai are living on the streets

If the owner operators are the winners, even though most are now small scale, who are the losers in the transformation of the Chinese economy? To be sure, the most significant cleavage is regional in that the eastern coastal provinces now have opportunities to participate in the new Pacific-Asia economy by obtaining training, capital and technology from their oversea Chinese contacts. Workers and peasants from the western, poorer provinces are at a great disadvantage. Currently, in response to this regional inequality, an estimated 100 to 120 million workers have become a floating population, most of whom are seeking work in the special economic zones and the eastern provinces in general. For despite the planned involvement of the village and town communities regulating and taxing the Pacific-Asian enterprises, significant inequities have arisen that mark the winners from the losers.

### 11.7 The role of gender in the distribution of winners and losers

Until now we have evaluated winners and losers in terms of income inequality, poverty, access to economic opportunity and ethnic differences. Important as these factors are, gender is critical for understanding how inequality and opportunity work themselves out in the Asia-Pacific (see also Chapter 5). Tables 11.4 and 11.5 provide a comparative analysis of how gender impacts life chances, education, employment and income. Women's participation in the rise of the Asia-Pacific economy is central to understanding the social structure of the labour force, migration patterns and differences among states. Of course, the costs and benefits of women's participation in the labour force and their role in the economy in general has created a vast and complex literature, well beyond the space limitations of this short chapter.

When one examines these tables for the mature NIEs, the relative deprivation experienced by South Korean women stands out. Their wage rates, participation rates and higher education rates lag their sisters in the other mature NIEs and even those from some of the almost NIEs. The literature suggests that Confucian beliefs combined with the harshness of South Korea's centralized authoritarian industrialization explain a large part of this difference. A Confucian tenet is said to require three obediences from a woman: to the father when a youth, to her husband when married and to her son when aged. It is also remarkable that South Korean women have the world's highest rate of non-marriage. Occupational segregation, discouragement from seeking leadership roles and marginalization from non-traditional participation are typical experiences of South Korean women.

Among the Taiwanese, the situation is different, precisely because the production system employs a network of subcontractors that use the discipline of family ties to exploit women workers (Chapter 5). An example of this is what has been described as a government sponsored 'living room factory' programme which was designed to solve the labour shortage problem by mobilizing surplus labour in the community/family to engage in production. The government provided special loans for families to purchase machines to do homework. Workshops were conducted to train housewives to apply themselves to productive work. Soon a second programme, 'mothers' workshops' was promoted to draw more domestic-based women into production. Each industrial form of production, therefore, articulates with family and gender roles so as to recruit women as a subordinate labour category.

In the electronics factories of South-East Asia, a preference for women in audio and circuit assembly has resulted in the feminization of the manufacturing workforce. The intensity of the work combined with the cultural socialization of the young women toward obedience meant that conventional forms of protest were resisted. Accordingly, in the 1980s a number of observers in Malaysia recorded hysterical outbursts as stressed-

		Literacy		Female	enrolment a males	% Female enrolment		
	Adult literacy rate (%) Female 1990	Adult literacy rate (%) Male 1990	Adult female literacy as % of male 1992	Overall Female enrolment % (6–23 yrs) 1990	Secondary school age (12– 17 yrs) 1990	Tertiary school age (18–23 yrs) 1990	Secondary school age % (12–17 yrs) 1990	Tertiary school age % (18–23 yrs) 1990
Mature NIE	ls -							
South Korea	95.80	99.10	97	69.10	97	53	81.80	24.00
Taiwan	$84.00^{1}$	95.50 <sup>1</sup>	88 <sup>1</sup>	NA	105 <sup>1</sup>	98 <sup>1</sup>	97.31 <sup>1</sup>	33.90 <sup>1</sup>
Singapore	84.30	95.40	88	67.00	101	86	87.80	19.90
Emerging NIEs								
Malaysia	75.40	87.80	86	58.80	106	99	66.70	7.00
Thailand	91.40	95.80	95	43.90	88	86	34.60	16.20
Indonesia	76.40	88.80	86	56.20	92	56	57.70	13.50
Philippines	93.70	94.40	99	64.50	103	113	76.30	27.60
China	70.00	88.20	79	40.00	78	75	36.30	12.30
Pacific Latin American NIEs								
Chile	94.30	94.80	99	65.70	100	93	84.40	28.50
Mexico	86.00	90.70	95	61.50	98	76	59.50	16.50
Peru	80.90	93.60	86	69.70	81	61	66.80	26.20

Table 11.4 Gender and education

<sup>1</sup> Data from *Statistical Yearbook of the Republic of China 1996*, Executive Yuan, Directorate-General of Budget, Accounting and Statistics, Republic of China, 1996. Source: United Nations (1996)

	Life chances			Employment				Occupations			
-	Total fert ility rate 1992	Ma ter nal mort ality rate (per 100,000) live births 1989– 951	Cha nge in mate rnal mor tality rate (per l00k live births) from 1980 to 1989– 951	Wom en's non- agricul tural wage as % of men's 1992	Wo men's share of adult lab our force (age IS and above) 1994	Female econ omic activity as a % of male 1994	Cha nge from 1970 to 1994 in fem ale eco nomic activity as a % of male	Ad minis trative and mana gerial. Female as % of male 1990	Prof ess ional and techn ical. Female as % of male 1990	Cler ical and sales. Fem ale as % of male 1990	Ser vices, Female as % of male 1990
Mature NIEs											
South Korea	1.70	30	-4	53.50	34	52	5	4	74	79	156
Taiwan	NA	NA	NA	63.90 <sup>2</sup>	38 <sup>2</sup>	63 <sup>2</sup>	18	NA	NA	NA	NA
Sing apore	1.70	10	-1	71.10	36	58	22	19	68	NA	69
Emerging	g NIEs										
Mala ysia	3.60	34	-25	NA	36	55	10	NA	NA	NA	NA
Tha iland	2.10	155	-115	68.20	44	77	-9	29	111	134	128
Indo nesia	2.90	450	-350	NA	31	44	3	7	69	79	135
Philip pines	3.90	208	128	60.80	31	44	-5	38	172	168	138
China	2.00	115	71	59.40	43	81	7	13	82	65	107
Pacific NIEs Latin-American											
Chile	2.50	40	-15	60.50	29	39	12	24	108	86	263
Mexico	3.20	$200^{3}$	108	75.00	28	37	16	24	76	71	82
Peru	3.40	165 <sup>3</sup>	-145	NA	24	32	7	28	69	109	60

 Table 11.5 Gender: Life chances and work

<sup>1</sup> Data from *World Development Report 1996*, World Bank, Oxford University Press, New York, 1996.

<sup>2</sup> Data from *Statistical Yearbook of the Republic of China 1996*, Executive Yuan, Directorate-General of Budget, Accounting and Statistics, Republic of China, 1996.

<sup>3</sup> 1988 data from *World Development Report 1994*, World Bank, Oxford University Press, New York, 1994. NA=Data not available

Source: United Nations (1996)

		% of popula	ttion below th line	Number of poor (millions)			
	Years	First year	Last year	Change	First year	Last year	Change %
Emerging NIEs							
Malaysia	1973– 87	37	14	-23	4.1	2.2	-46
Thailand	1962– 86	59	26	-30	16.7	13.6	-18
Indonesia	1972– 82	58	17	-41	67.9	30	-56
Singapore	1972– 82	51	10	-21	0.7	0.2	-71
Others							
Brazil	1960– 80	50	21	-29	36.1	25.4	-29.6
India	1972– 83	54	43	-11	311.4	315	1
Pakistan	1962– 84	54	23	-31	26.5	21.3	-19
Sri Lanka	1963– 82	37	27	-10	3.9	4.1	5

### Table 11.6 Change in poverty

Source: World Bank (1993)

out women went into trances. So disruptive of production were these culturally specific expressions of resistance, some international firms recruited local Malay spiritualists to rid the factory of the ghosts that were thought to be disturbing their workforce.

A major theme in the literature, especially in newly reformed China, the Philippines and Thailand, focuses on the emergence of prostitution or sex work among women both married and unmarried. One of the demands in the expanding Asia-Pacific economy is for sexual services and entertainment. Great concern surrounds this trade both for health reasons and for the protection of children. Young women are commonly forced into this trade especially in Thailand (see Chapter 5).

### **11.8** Conclusion

In conclusion, whilst it is clear that the East Asian economies have been particularly attentive to the distributional considerations in their respective development strategies, as with any transformation process, there are always going to be winners and losers from the changes it initiates. This chapter has concentrated on first documenting these distributional issues along a number of different dimensions, and then on offering explanations for the different experiences characterizing each set of economies delineated. In doing this a contrast with the Latin American Asia-Pacific experience has also played a part, though for illustrative purposes only.

The chapter has stressed the key role of social class, ethnic group and gender category as explanatory categories for analysing the varying experiences in the East of the Asia-Pacific. Whilst not totally unique, the specific combinations of these within particular countries offers a rigorous way of characterizing the 'winners and losers'.

### References

- Amsden, A.H. (1985) 'The State and Taiwan's economic development' in Evans, P.B., Rueschmeyer, D. and Skocpol, T. (eds) *Bringing the State Back In*, New York, Cambridge University Press.
- Bello, W. and Rosenfeld, S. (1990) *Dragons in Distress—Asia's Miracle Economies in Crisis*, San Francisco, The Institute for Food and Development Policy.
- Cotton, J. and Hyung-a van Leest, K. (1996) 'The new rich and the new middle class in South Korea: the rise and fall of the "golf republic" in Robison, R. and Goodman, D.S.G. (eds).
- Chu, J.J. (1996) 'Taiwan: a fragmented "middle" class in the making' in Robison, R. and Goodman, D.S.G. (eds).
- Goodman, D.S.G. (1995) 'New economic elites' in Benewick, R. and Wingrove, P. (eds) *China in the 1990s*, Vancouver, UBC Press.
- McGrew, A. and Brook, C. (eds) (1998) *Asia-Pacific in the New World Order*, London, Routledge in association with The Open University.
- Page J. (1994) 'Special section: The World Bank's *The Asian Miracle:* an introduction', *World Development*, vol. 22, no. 4, pp. 618–23.
- Robison, R. and Goodman, D.S.G. (eds) (1996) *The New Rich in Asia-Mobile phones, McDonald's and Middle Class Revolution*, London, Routledge.
- United Nations (1996) Human Development Report 1995, New York, Oxford University Press.
- Wade, R. (1990) Governing the Market—Economic Theory and the Role of Government in East Asian Industrialization, New Jersey, Princeton University Press.
- World Bank (1993) The East Asian Miracle, New York, Oxford University Press.
- Yoshihara, K. (1988) *The Rise of Ersatz Capitalism in South-East Asia*, New York, Oxford University Press.

### CHAPTER 12 The environment, traditional production and population

Elspeth Young, Colin Hunt and R Gerard Ward

### **12.1 Introduction**

As the previous chapter has shown, economic growth in the Asia-Pacific region has generated costs as well as benefits. While significant numbers of people can be described as 'winners', others, including small farmers, women, minority ethnic groups and informal sector producers, may currently be seen as 'losers'. This chapter moves from this theme to emphasize environmental aspects of economic growth through the adaptation of traditional production systems. Societies and population groups which still strongly sustain themselves through traditional production are well represented in the Asia-Pacific region, particularly in the Pacific micro-states. They are not necessarily, as the preceding discussion might imply, the 'losers'. Through modern forms of traditional production, in which human populations sustain themselves by combining subsistence use of natural resources in a non-cash economy with cash-earning activities compatible with subsistence, it is possible to be a 'winner'. Historically traditional production has been remarkably resilient and innovative in its reaction to wide sweeping economic and political changes and we anticipate that, provided its vital role is still recognized and supported, such adaptations can continue into the twenty-first century.

Traditional production is a holistic system in which economic activity, social and political structures and environmental resources are interdependent. Change in any of these prime components has major implications for the others. The introduction of cash cropping in traditional production systems, for example, may lead to vegetation clearance and soil erosion. It may also cause social disruption by widening the gap between the richer families of cash-cropping farmers and their poorer neighbours who have not been able to take advantage of these opportunities. Recognition of these interlinkages underlies the philosophy of sustainable development. In contemporary and future Asia-Pacific population development will only be sustainable if the socio-cultural and political dimensions of traditional production systems are acknowledged to be as significant as their economic roles. Traditional production, through linking people with their immediate natural environment, provides the basis for group and individual identity. This function needs to be recognized in its contemporary transformation to modern, dynamic systems designed to cope with people's changing aspirations and goals, and with their incorporation into regional and global economies. Control of population growth, enhancement of equity in resource access and distribution of benefits from resource use are important principles in modern forms of traditional production. If Asia-Pacific

development is to be sustainable competition must be tempered with integration and cooperation.

One important question concerns the inherent sustainability of traditional production systems. It is commonly assumed that, because of their holistic nature, traditional production systems are more 'environmentally friendly' and sustainable than more technologically complex methods. These beliefs, which promote the superiority of indigenous knowledge in environmental management, have made valuable contributions to the recognition of the particular rights and needs of indigenous groups and have helped to increase their political empowerment. They have also, however, been interpreted somewhat romantically as furnishing a development model which industrialized peoples would do well to adopt. Such an assumption requires some qualification. As we show here for the Asia-Pacific natural resources such as the rain forest have been depleted under traditional production systems little modified through contact with modern industrialized society. While subsequent contact may have accelerated the pace of change the necessary processes were already in place. Thus, in accepting that traditional production systems are essentially more sustainable we also stress that they have long been subjected to pressures arising from population growth and economic and political aspirations. Contemporary Asia-Pacific traditional production and its environmental impacts needs to be seen in pragmatic rather than emotional terms.

The chapter outline is as follows. After briefly pointing out the main elements of diversity and inequality which are important to our understanding of the environment and traditional methods of production in the Asia-Pacific we explore two major themes. *First,* and most important, we examine the realities of traditional production methods and resource tenure and its environmental effects in the Asia-Pacific of today. *Second,* we consider some population issues, including population mobility, urban growth and present and future environmental consequences of these processes. *Third,* we comment, with particular reference to large-scale mining operations, on the environmental effects of both externally and internally driven pressures on natural resource use. Examples refer primarily to what, in global terms, might be termed the 'minor players'—the Pacific Island micro-states and some of the indigenous minorities of industrialized Asia-Pacific countries. It is among such groups that contemporary relationships between the environment and changes in traditional production are most clearly demonstrated (see **van Fossen, 1998**). Despite this particular focus, many of the issues raised are relevant to rapidly industrializing Asia-Pacific states such as China, Malaysia and Indonesia.

### 12.2 Key issues concerning traditional production in the Pacific

Traditional production—the derivation of human sustenance through subsistence agriculture, subsistence use of coastal and marine resources, hunting and foraging for vegetable and animal foods and other products in forested and arid environments—is still of prime importance in many Asia-Pacific societies. It operates in conjunction with resource tenure systems which, for many people, are still dominated by customary resource ownership, through which community control of land and marine resources is set within frameworks of customary law and tradition. This does not necessarily mean, as is sometimes assumed, that traditional production is static, reflecting the conditions of the

era preceding colonization of the Asia-Pacific by industrializing and industrial nations. In the past traditional production has continually been adapted to accommodate changing relationships between people and resources. These have usually resulted from population growth or decline and the adoption of new technology which has enabled resource exploitation to spread. Today traditional production must arguably be even more dynamic, changing to meet the pressures exerted by external development forces stemming from the incorporation of Asia-Pacific societies into regional and global economies. Increasingly it co-exists and intermingles with industrialized forms of production founded on more individualistic forms of resource ownership, with the latter becoming more and more prominent. While many societies have themselves accepted these changes as a vital element for their future survival external influences are also playing a very significant role. Structural adjustment programmes established to raise the economic performance of debt-ridden Pacific Island countries, such as that currently operating in Papua New Guinea (PNG) under World Bank and International Monetary Fund (IMF) aegis, have overtly promoted the abandonment of earlier forms of communal resource ownership as a necessary condition for 'development'. Structured adjustment programmes-the preferred contemporary policy instrument of the World Bank and the IMF-require countries that are to receive aid and financial assistance from these organizations to reform their domestic economies along 'market-friendly' lines. This usually involves a package of measures including: increasing labour market flexibilities; reducing budget deficits; introducing privatization programmes; liberalizing internal capital and other markets; adjusting exchange rates and reducing trade barriers so as to stimulate trade.



Building work in progress on the Three Gorges Dam on the Yangtze River in China

A key element in these contemporary adaptations of traditional production is the effect they have on the environment and its natural resources. For example, the harnessing of a mighty river, such as the Yangtze or Mekong, for hydro-electricity generation could potentially destroy the intensive irrigated agriculture, both above the dam through inundation and downstream through altered water conditions, of millions of people who remain dependent on this method of sustenance. Massive clear felling of tropical rain forest not only has a direct effect on global climate change and biodiversity but also leads to soil erosion and loss of agricultural productivity throughout other parts of the water catchment area. Agricultural intensification through the increased use of machinery, chemical fertilizers and the introduction of new crops and cropping systems not only forces people to be increasingly dependent on capital inputs but may also in the long term result in soil degradation and the disturbance of fragile population resource balances. Developments such as these are frequently accompanied by radical changes in definitions of traditional resource ownership-sale of common interests in land and water within a more individualistic framework which benefits some people more than others; and individual definition of resource ownership so that desired improvements can be carried out without interference from one's neighbours.

Diversity, both between and within Asia-Pacific nations, strongly affects both the adaptation of traditional production and the environmental consequences of these changes. Asia-Pacific countries vary enormously in geographical size, population and population distribution (Hull, 1998); in topography, climate, soils and vegetation; in natural resources; in socio-cultural characteristics; and in colonial experience (Herr, 1998). These dimensions underlie any understanding of issues affecting the Asia-Pacific at the end of the twentieth century. Because of diversity, benefits from the economic dynamism of the 'new' Asia-Pacific are unevenly spread. Rapid economic growth in newly industrializing countries such as Malaysia and the Republic of Korea is occurring at considerable environmental and human cost not only to their own regions and populations but also to other countries, whose resources are being exploited to fuel such growth (Maddock, 1998). Rapid and unplanned urban growth is affecting both larger and smaller Asia-Pacific countries and environmental problems such as air pollution, inadequate waste disposal and water supply contamination are apparent not only in vast cities such as Beijing and Manila but also in relatively small cities and towns like Port Moresby (Papua New Guinea) and Funafuti (Tuvalu). And some Asia-Pacific countries, particularly some of the micro-states, not only suffer from environmental problems of their own making but are also facing threats from changes such as sea-level rise, to which industrially-driven global transformation is contributing (see van Fossen, 1998).

While changes in traditional production and associated environmental consequences are occurring in many parts of the Asia-Pacific, their effects are particularly obvious in less industrialized states such as the Polynesian and Micronesian micro-states, in Melanesia and in countries with high rural populations such as Indonesia and Philippines. Indigenous minorities in some industrialized countries such as Australia, Canada and New Zealand are being affected by similar transformations. All of these population groups have experienced rapid and recent social and economic change and the reactions and challenges which face those affected are thus clearly exposed. These peoples are the main focus of the discussion which follows.



Figure 12.1

### 12.3 Change and adaptation in traditional systems of resource access and use

### Western (mis)conceptions of traditional tenure systems

It is sometimes thought that in traditional systems land and other resources are, or were, held in common with open access to all. Such ideas were in vogue amongst Western European thinkers in the late eighteenth and the nineteenth centuries when it was commonly thought that human society evolved through a series of stages towards a 'more civilized' state akin to that then existing in their own Western European countries. In one philosophical stance it was argued that if people did not farm or graze domesticated animals on the land they could not have any system of ownership of the land. It was assumed that, in a later stage, when people practised shifting cultivation, along with hunting and gathering, land tenure would be communal with a high degree of sharing of resources and relatively open access to land or sea for all. When European administrators came to the Pacific Islands to establish colonial administrations, they usually lacked detailed knowledge of the customs and practices of the indigenous people and not unnaturally tended to apply their own interpretations to these practices which they understood imperfectly. They often assumed that land was held by the occupants in some form of communal tenure approaching that of 'commons' within which all members of a group would have open and relatively equal access. In fact, in the Pacific Islands, there were few, if any, commons with open access for all. Even in coastal waters, and certainly in the lagoons and on the reefs, each area was normally the recognized preserve of one group or family. This was in marked contrast to the European concept of open or common access to marine resources. Perhaps only the air and the more distant open ocean were held in common within the Pacific Islands.

The tenure and resource use systems of hunting and gathering peoples, who did not practice subsistence farming, were even more blatantly ignored by Europeans. The Asia-Pacific region contained many such groups-the Australian Aborigines, the Eskimo/Inuit peoples of North America and forest dwelling groups in many parts of the island chain stretching through modern-day Philippines to Indonesia. Today these peoples have survived as indigenous minorities who continue to struggle for recognition of their tenure rights to land and marine resources and their right to use and manage these resources as they wish. Australia, for example, was described from its early colonial period as 'terra nullius', a land not owned or ostensibly used by its 'first peoples', the Aborigines and Torres Strait Islanders (Taylor, 1998). This myth, based on a complete failure to identify the complexities of social, cultural and economic organization of these hunting and foraging peoples, justified the seizure of land and resources by the incomers and led to widespread dispossession and displacement of many of the indigenous peoples from their customary territories (Young, 1992). Only in 1992, with the Australian High Court ruling in favour of recognition of prior ownership of Australian land before colonial settlement, was this myth finally exploded. But the conflicts emerging in the continuing debate over the granting of Native Title rights to indigenous Australians demonstrate that for many people the myth remains a reality.

Other indigenous groups for whom hunting and foraging remain significant have suffered in similar ways. These include rain-forest dwellers in Malaysia and the Philippines, the marine harvesting Ainu in Hokkaido in northern Japan and both Eskimo/Inuit and Indian groups in Canada and the USA. Even when official treaties between incoming colonial powers and the indigenous residents were negotiated, as in Canada and the USA, the level of compensation offered could in no way recompense the ensuing loss of control over natural resources, loss of social identity and loss of political power.

Subsistence farmers, who clearly marked the landscape not only by cultivating crops but also through activities such as composted garden beds, and complex terracing and irrigation systems on steep slopes, fared somewhat better. Europeans recognized the visible results of their farming as indicators of long-term relationships with land and interpreted these as equivalent to their own concepts of 'ownership'. Treaties such as those negotiated at Waitangi in 1840 between the British Crown and the Maori people in New Zealand (Eccleston *et al.*, 1998) at least set in place a legal framework for negotiation over resource use, albeit one which proved easy to set aside. Here also it is only in recent decades that some negotiations have achieved greater equity in resource allocation.

### Pacific land tenure and resource use

Land tenure systems in the Pacific Islands prior to colonial intervention had certain common features (see Box 12.1). A key point is that in any area a number of individuals or groups might have rights to particular forms of use and occupation and such rights would normally be long term provided they continued to be exercised. Such tenure systems normally provided considerable security for community members and had the advantage of being flexible enough to meet the subsistence needs of groups whose relative sizes changed by the normal vagaries of demographic processes.

### Box 12.1 Customary land tenure in the Pacific Islands

A community of several clans would claim an area of land as its territory. Within that territory, each clan would be acknowledged as the controller or owner of particular sections. The whole was not the common property of the community for more intensive uses, despite the fact that, by custom, all residents might have been free to gather forest products from anywhere in the uncultivated parts of the territory. Although some limited rights might have approached commonality within this forest or uncultivated land, individual trees or products would be recognized as the property of individuals and control of hunting and gathering would rest with

particular people or sub-groups. Within the land of one clan, members would not all have equal rights to clear and cultivate any part because specific individuals, nuclear, or extended families would hold residual and relatively exclusive rights to occupy, stemming from the last period of cultivation of the particular piece. The land of a house site might be very specifically under the control of a particular nuclear or extended family. Specific resource sites, such as a spring or a source of clay, although within the boundaries of a clan's land, might be controlled by specific members of that or another clan, with relatively free access being allowed to almost all, but under 'grace and favour'. A superficial examination of this type of system, Influenced by ideas stemming from nineteenth century sociology, might well conclude that as all members appeared to be able to hunt, gather, or collect water from any area or site within the community's broader territory, and that people cultivated gardens scattered throughout the territory, that the land was common property.

Source: Ward (1997)

Customary tenure systems of the type described in Box 12.1 matched production systems based on cultivation of gardens of root crops, such as taro or yams, or grains such as rice, grown in dry land plots cleared from forest. Fertility was maintained by systems of shifting (swidden) cultivation in which short periods of cultivation were followed by long periods of fallow during which the former gardens reverted to woody secondary growth or eventually to forest. Such systems, managed with simple tools such as stone or metal axes and adzes and wooden digging sticks, and operating within tenure arrangements which allowed farmers to shift their plots within the group territory, could be very stable and relatively protective of the environment. However, if population densities became too high, or the fallow period was unduly shortened, soils might not regain their fertility and a downward spiral of productivity could begin. Thus, in some parts of the region, considerable areas are now reduced to a cover of sparse grassland on thin soils lacking in many essential plant nutrients.

#### Sustainability and environmental impacts

Under stable conditions, the traditional shifting cultivation production systems can be environmentally sustainable. However, when the relationships between people and their environment change rapidly that environment may well deteriorate dramatically. There have been two eras when such effects have been most dramatic. The earlier resulted from the initial introduction of agriculture into different parts of the region. When this occurred, forest clearance and the use of fire resulted in rapid and extensive erosion in many areas with lower-lying land often being overlaid by extensive deposits of material washed down from the hillsides. This tended to be most marked in areas of low or seasonal rainfall where fire could spread most easily and regeneration might be delayed by seasonal climatic conditions. Once a degree of balance was attained between the new agriculturalists, their technology and local environmental conditions, the rate of environmental degradation was reduced and a new, generally sustainable, level of use established. Thus the drier parts of many Pacific Islands and the hill country of northern South-East Asia and southern China now have large areas of anthropogenic savannah or grassland which, with the mosaic of swidden gardens, secondary growth and remnant forest patches, record centuries of use by indigenous farmers under changing conditions of population pressures and technological change. At times these systems have been sustainable; at times not.

The second period of rapid environmental degradation is our current era. As technological capacity to change environments has dramatically increased, population growth and the demands for higher living standards have placed growing pressure on local resources. Almost simultaneously in terms of the long stretch of human occupation, the globalization of the world economy has resulted in rapid exploitation of many forest areas and mineral deposits in places far distant from the locations where the timber or minerals will ultimately be consumed. This has brought people who, for long relatively stable periods, have used traditional agricultural, tenurial and social systems, into direct contact with those wielding great technological and political power. While some impacts of the extractive industries component of this recent conjunction are discussed below we must note that changes in political and economic systems stemming from European expansionism of the nineteenth century and the new forms of colonial agriculture also had major environmental and social impacts.

#### The facts of communal tenure systems

In countries such as China and Japan, communal tenure systems and related agricultural systems did exist at one time, but were gradually replaced over the last millennium by systems of private and state ownership in accord with changes in technology and socio-political and economic systems. In parts of South-East Asia and the Pacific Islands which were less influenced by commercialism and centralized governments, communal systems persisted in some areas and continue to be used today. However, the establishment of states, either indigenous or colonial, often led to the imposition of new forms of tenure which modified or ignored and overrode local communal tenure. These new tenure systems were usually linked to agricultural systems which used permanent fields (rather

than shifting cultivation) and in many parts of the region were dependent on irrigation. Permanent tree crops planted on foreign-owned plantations or by indigenous smallholders for export production have been another major feature. In parts of South-East Asia and the Pacific Island countries, colonial governments decreed that land which was 'waste and vacant' belonged to the state, leaving land then used in the hands of the traditional owners. The definition of 'waste and vacant' would not necessarily be accepted by the local indigenous population. Furthermore, colonial settlers and others frequently purchased land from traditional communal owners, or took land by force or trickery, thus establishing other non-traditional forms of tenure which applied in specific areas. In Indonesia and Malaysia the post-independence states have assumed the ultimate control of almost all land with the result that legal ownership on the part of the occupants derives from specific allocations made by the state. Thus in many countries in the region indigenous owners lost control of a considerable area, and immigrants of different social background, from overseas or from elsewhere in the state, have moved in bringing new agricultural and economic systems, new settlement patterns, and new power structures. Different forms of tenure now exist side by side, some land remaining under customary tenure (see Box 12.1) while some is held by individuals under permanent or limited term rights derived from the state.

Settlers from Europe and elsewhere sought to establish themselves on land held under tenure systems which were familiar to them from their homelands, and which gave them long-term security of occupation and the right to sell. Thus it is common for the land once used for large-scale commercial agriculture in the nineteenth century plantations, to be allocated as freehold or leasehold land by governments which have taken the ultimate control of the country's land into their own hands. In some cases the land allocated in this way is part of that which was declared to be 'waste and vacant' in an earlier era, and thus to be the property of the state. Some land was purchased by the state from customary owners. The new administration and urban centres also required land over which the incoming occupiers had secure tenure and thus governments sought to establish the towns on land which was not held under customary tenure. In recent decades governmentplanned agricultural settlement schemes in Malaysia, Indonesia, Papua New Guinea, and other countries have been established on land which has been deemed not to be customary tenure or has been purchased or leased from customary owners. Underlying such schemes is a common belief amongst planners that communal forms of tenure do not provide a suitable basis for commercial farming, and that forms of tenure which allow land to be alienated or transferred are necessary to encourage individual effort and the use of land for security against the provision of credit to farmers. Those who wish to invest capital in urban or rural commercial credit ventures may be reluctant to do so when they are not sure of the likely duration of their rights of occupation and of the security needed to raise funds from financial institutions.

### New agricultural systems and their consequences

The systems of plantation agriculture or small-holder commercial farming carried out on the lands held under non-traditional forms of tenure also used wage labour rather than family or clan labour; were oriented towards production for export; and created new patterns of settlement through much of the region. Farm workers were often recruited from distant areas, or from other countries. They created new minority groups in the countries or colonies, and they lived in plantation quarters or, later, on small independent farms in dispersed settlement patterns quite different in form and social organization to those of the indigenous peoples who commonly lived in nucleated village communities surrounded by their communal land.

The new agricultural systems brought new environmental impacts. Permanent cropping requires different techniques and inputs (such as fertilizers) to those appropriate for shifting cultivation. Long fallow periods must be replaced by use of crop rotation, fertilizer or pesticides if soil fertility and crop security are to be maintained. Mechanization and frequent working with heavy equipment may alter soil structures and increase erosion. Permanent tree crops may reduce soil fertility. Thus the new colonial agriculture ushered in a new period of environmental instability with parallels to that of the earliest period of agriculture in many parts of the region. Indigenous farmers using traditional techniques under customary tenure were often pushed off their land. New states of sustainability needed to be established. This occurred in some places, but not in others.

### Changes in the Pacific Island states

Apart from those who farmed on non-customary land, people who still occupy land under customary tenure are also experiencing new needs in relation to their rights over land, and as a result customary land tenure practices are changing quite rapidly in many Pacific Island states (see, for example, Ward and Kingdon, 1995). Most Pacific Island farmers have chosen to be engaged in systems of mixed cash-crop and subsistence farming. While customary tenure systems were well suited for the latter, especially under shifting cultivation with short cropping periods and long fallows during which secondary forest regrew, commercial cash-cropping farmers are often growing crops such as coconuts or coffee, or using land for pasture. This means that land stays in use for periods as long as several decades. Because customary systems allow a farmer security of holding as long as he or she keeps the land in use, the possibility of land re-allocation is now reduced. Furthermore, commercial farmers are likely to plant much larger areas than if their needs were restricted to subsistence requirements. People, usually men, of influence in the community may establish control over large plots by the customary practice of clearing forest and planting crops and then maintain it under their own control for so long that they come to consider themselves the owners of the plot, rather than simply the owners of the crops. Some seek to pass the plot on to their direct descendants rather than seeing it eventually return to the control of the wider group. Parallel with this trend, and also as a result of commercialism with its different timetables and labour requirements from those of subsistence agriculture, is the tendency for farmers to employ labour for wages rather than depend on kinsfolk mobilized through the principles of reciprocity. Thus one of the mechanisms which binds a traditional Pacific Island community together, reciprocity, is weakening at the same time as communal land is gradually being privatized. In the past all members of the community might expect to have access to some land for subsistence needs. Today this is not always the case. Since access to land is more limited and because bigger farmers are controlling much of the customary land for long-term commercial use, not all have an adequate share for subsistence. Landlessness is beginning to appear. In practice, tenure is changing even if the rhetoric of governments, which often stresses the need to maintain customary ways, is not yet ready to acknowledge the changes and deal with the implications of the possible emergence of landless groups for whom the older customary communal ways no longer provide a safety net.

The tendency for only some members of communities to benefit from the rights which customary land tenure should provide is also seen in areas where large-scale logging by foreign or local companies is occurring. In Melanesian countries such as Papua New Guinea and the Solomon Islands the rights of landowners to a share in the returns from logging on their community's land is generally acknowledged. But there are instances in which those who are appointed or are self-selected to negotiate with the logging companies are able to retain most of the benefits for themselves while the majority of the community see little benefit. Similar risks occur in negotiations with mining companies.

Changes in resource tenure and management pose difficult questions for contemporary Pacific societies, particularly in terms of the maintenance of a sustainable population/resource balance into the next century. They also have clear environmental implications. Box 12.2 presents a case-study of what Allen (1996) termed Papua New Guinea's dilemma in land management.

### Box 12.2 Land management adaptation in the Papua New Guinea Highlands

At current rates Papua New Guinea's population of around 4 million will probably exceed 6.5 million by 2015. With the relatively slow growth of the monetary economy and low level of industrialization most of the population will remain dependent on subsistence farming, practised under customary land holding systems which today still account for over 95 per cent of the country's land ownership. While the details of Papua New Guinean farming systems vary widely, from coast to mountains and from steeper to more gently sloping land, they are all essentially based on shifting cultivation. Agricultural intensification is essential if people are to survive. The Highlands, which support almost 40 per cent of the country's population and for the most part have the most sophisticated and productive subsistence agricultural systems, lie at the 'cutting edge'. But intensification must accord not only with customary tenure systems and new commercial needs but also with the environmental fragility which characterizes this.

What choices will a Highlands subsistence farmer, faced with an ever-increasing number of people in his family and with very limited opportunities to extend the amount of cropland available to him, be able to make? How can he produce more food? He can use the land more intensively by shortening the length of the fallow period. But this will not

give the land that vital resting period to enable it to be rejuvenated so that conditions similar to those occurring under virgin forest are recreated. Unless new soil management techniques are adopted this approach will eventually result in the replacement of forest by grassland and, where the soil's nutrient balances are never fully restored, will lead to soil erosion and complete degradation. The extensive areas of grassland which cover many of Papua New Cuince's major upland valleys are now considered to be directly due to human intervention and hence show that such changes have been occurring for several generations, well before colonial contact was established.

A Highlands farmer could also make his land more productive by introducing new crops or using new technologies. These measures have also occurred in the past. Sweet potato, which has replaced the taro staple crop over the last 500 years, undoubtedly allowed larger numbers of people to be fed. Cash crops, such as coffee which is now part of most Highlands 'small-holders' activities, have generated some prized cash input which have helped to support families when commodity prices have been high. But all of these changes have required adaptation of the traditional production systems-including recognition of more individual forms of land ownership within the communal system (see above). And regardless of the adaptations accepted the population/resource question will also include another element—the fact that the products of the agricultural system are not designed only to feed the family; they also provide the commodities for wealth production through flamboyant exchange systems which cement kinship relationships so as to provide mutual protection within the clan structure. These are central components of village society-the very essence of social identity. Their destruction through inappropriate adaptation of traditional production would be culturally disastrous. Source: Allen (1996)

### Logging and the environment

An additional component of this land management dilemma is that changes such as those described above also lead to decline in forest cover. The environmental consequences of tropical rain forest loss, not only on the local scale as described above but also on regional and global scales, are now well recognized. Forest use in Pacific Island societies has included, and still does include, the harvesting and exploitation of both timber and non-timber products for a wide diversity of purposes. Traditional resource tenure has been a vital element in such forest use, determining not only who is responsible for specific forest areas but also who may legitimately harvest the forest's timber, nuts, fruits and the wildlife which it harbours. As with agriculture, population growth coupled with improved technology, physical access and demands for higher material living standards have placed increasing strain on Pacific forest resources. Along with these internal pressures has come increased demand for tropical timber from both the industrialized world and also rapidly industrializing Pacific neighbours such as Malaysia. The shortterm monetary benefits to be gained from permitting foreign logging companies to operate on customary land have, for many islanders, been perceived to be more important than concerns with long-term depletion and loss of sustainability. Commercial logging is generally believed to be the prime cause of heavy forest loss in countries such as the Solomon Islands (where, at the present rate of exploitation, the forest will last for only nine more years). But commercial logging is not the only reason for forest depletion. Forest has also been lost under traditional production alone (see above) and, as Ward (1995) documents for Western Samoa, it seems that the relationships between modernization, population change and resource use are much more complex than often assumed. Ward's longitudinal study shows that over the last four decades Western Samoa's forest has significantly declined (by approximately 50 per cent between 1954 and 1990) not principally because of commercial logging but because of progressive adaptation of traditional land tenure systems with increasing individualization of property ownership. Younger families, who have as yet not been able to exert widely recognized individual rights to land, have turned to the clearance of previously inaccessible uncleared forest land as a more attractive alternative for increasing their subsistence production. Thus parts of the islands now exhibit unexpected land use anomalies—low production from land in accessible locations and held under former customary practices; and more production from land on forest margins in remoter parts of the country.

### Contemporary use of marine resources and environmental impacts (based on Hunt, 1997)

Pacific Islands people also use traditional forms of production to harvest their marine resources. Most coastal-dwelling people in Pacific Island countries still make substantial use of inshore marine resources for subsistence purposes. Despite the increase in fishing for cash, a large proportion of the catch is still consumed by the fishers, or shared, and does not enter markets. The fact that subsistence fishing is important, or even dominant, means that the true importance to Pacific Island states of their marine resources is not adequately reflected in budget estimates or national accounts.

By showing estimates of the value of fish taken in the coastal waters of selected Pacific Island countries, Figure 12.2 provides a more comprehensive appreciation than national accounts of the importance of commercial and subsistence fishing and also of the relative size of the coastal fisheries by country. It is clear that the value of subsistence fishing is significant in most countries and that in several countries it has a monetary value of many millions of US dollars.



### Figure 12.2

Note: Data sources and methods for estimating the value of subsistence fishing are not clearly defined. We assume that coastal waters are, like territorial seas, waters within 12 nautical miles of the coast; and that the relationships shown in the figure should be used only as a guide.

Source: Dalzell et al. (1995, p. 149)

Customary marine tenure and inshore resources management

In the inshore waters of Pacific Island countries, the most widespread and important measure for the conservation of marine resources was regulation of access through customary marine tenure arrangements in which rights to fish were controlled by a clan, chief or family. In contrast to common practice in Western societies, no single group owned all rights. Instead access and use rights to the fisheries were carefully allocated through negotiation in group decision making. In customary marine tenure, as in land tenure (see above) social boundaries, delineating who can use which resources, are as important as physical boundaries. Marine boundary definition in general, however, is harder for people outside the group, and particularly for non-Pacific Islanders, to understand and define (Crocombe, 1994).

One of the customary marine tenure measures commonly adopted to conserve stocks was seasonal closure of reefs during critical times in fish life-cycles. By restraining people's activities customary holders of rights gained substantial benefits because fish stocks were preserved (Johannes, 1978). This was not necessarily deliberate. Hyndman (1993), while acknowledging the management implications of restricting access, doubts whether the conservation value of such measures was actually understood. He suggested that they were instead designed to ensure that marine resources played a key role in kinship modes of production, whereby resources are shared out within extended family networks; and that they could be used in the exchange of gifts. As Carrier (1987, p. 164) succinctly put it, '[customary marine tenure] made it possible to be generous'.

Customary law is the basis for decision making concerning access to resources and determining the sanctions that might apply when rules were disregarded. Nevertheless, customary law is flexible in that there are constant negotiations at the community level regarding access and use. And the underlying principles of customary marine tenure themselves are subject to continuous reinterpretation and transformation.

Transformation of traditional customary marine tenure has occurred largely because of coastal population decreases early in colonization, subsequent population increases (about 3.5 million people now live on the coast in the South Pacific), the adoption of more complex technology (fishing power and mobility), the intrusion of the cash economy and a breakdown of chiefly authority (Johannes, 1978; Crocombe, 1994).

### Competition under increasing scarcity: examples of the environmental implications of contemporary marine resource use

Today inshore marine resources have to satisfy both subsistence needs and increasing desires for cash by both coastal communities and Asia-Pacific national governments. While, because of problems of conducting rigorous scientific surveys of Asia-Pacific fish stocks, accurate data are not available it does seem that inshore marine resources are becoming scarcer. This has resulted in competition, sometimes fierce, between local populations and industrial fisheries for the use of inshore marine resources. A key bone of contention is the use of small fish for bait. In Papua New Guinea and the Solomon Islands, for example, bait fish resources are exploited to support industrial tuna fisheries. In Papua New Guinea, the distribution of bait fish royalties, in funds allocated to the local groups, trust funds and Provincial governments has caused much conflict. Local business groups formed to receive these royalties were in some cases not accepted because they did not represent those families holding customary marine resource rights. As a result industrial pole-and-line tuna fisheries were closed.

In the Solomon Islands, the jurisdiction of local communities over fisheries is enshrined in legislation. As *The Provincial Government Act, 1987*, Section 3 (7), says, 'Nothing in this section shall be construed as affecting traditional rights, privileges and usages in respect of land and fisheries in any part of the Solomon Islands'. As Hviding (1996) comments, some local communities have refused to sign bait contracts with their local tuna cannery because they fear that conflicts could arise over the distribution of royalty incomes which do not always find their way back to the resource owners. Resource owners also fear possible adverse impact on their subsistence fishing because if too much bait is taken by tuna boats, there will not be enough small fish to feed the bigger fish on which people largely depend. They are also concerned about the possibilities of oil pollution from bait-fishing vessels operating in the lagoons; and, as they are well aware, the presence of commercial bait fishers in the villages can lead to social problems. Despite these concerns, however, bait fishing royalties constitute one of the largest expenditure categories of the Solomon Taiyo tuna cannery.

Competition can also lead to more widespread depletion of marine resources. For example, in the group of islands which form the Republic of Kiribati, where fish is the prime natural resource, people in the outer islands are developing their commercial fishing to supply the urban area on South Tarawa. This is partly because subsistence fish stocks in the Tarawa Lagoon have been depleted by over fishing and the effects of pollution; and partly because outer island communities need cash incomes and employment. But this commercial fishing is in competition with the subsistence fishers in the island inshore waters. With no central control over fishing activities risks of resource depletion can only be resolved locally, either through Western-type controls in the case of peri-urban areas of South Tarawa or, in the remoter islands, through reinforcement of customary marine tenure.

Kiribati has also experienced depletion in stocks of larger, deep-water fish. Before 1994 the large foreign tuna fleets, catching and freezing huge quantities of fish for the canneries in the United States and Thailand, were allowed to fish up to 12 nautical miles off the coast. But local Kiribati fishers complained that the foreign fleets were affecting their own tuna catches and the Kiribati government has since banned them from waters within 60 nautical miles off the coast.

### Alternative management systems

The scarcity of inshore resources, and the pressure on them, has led to alternative management models being proposed (Dalzell et al., 1995). These alternative models strengthen rather than weaken customary marine tenure. Legislative support for local tenure arrangements allows the reintroduction of effective traditional methods, such as temporary closure of fishing. In these alternative models, local knowledge substitutes for, or complements, scientific data, while local planning substitutes for, or complements, fisheries department planning. The need for inshore management plans is most pressing in some of the smaller Pacific Islands and atolls where the impact of greatly increased fishing effort on fish stocks is jeopardizing the supply of essential protein to rapidly growing populations (Dalzell et al., 1995). The harvesting of commercial invertebrates such as trochus shell and bêche-de-mer (which in many cases have been severely depleted) can also come under local management arrangements. However it is important that marine tenure and access rights are clearly defined as otherwise open access, and over-use of fish, might prevail (Hunt, 1996). Altogether, as with land, the adaptation of customary use and management of marine resources to accommodate commercial exploitation inevitably places pressures on the Asia-Pacific environment.

### 12.4 Population, traditional production and the Pacific Island environment

As the previous section has implied, traditional production, from both land and sea environments in the Asia-Pacific, has undergone continual transformation but the rate of change has significantly increased during recent decades. Such changes reflect increased exposure to and involvement in the cash economy and the advance of new technologies throughout the region. They also reflect another factor—population numbers and high rates of population growth. Many Pacific Island countries have high rates of population growth, stemming from high fertility and declining mortality. For example, by 2025 both Papua New Guinea and the Solomon Islands will, on present growth rates, have doubled their current populations (UNDP, 1994).

Population, along with lifestyle aspirations which reflect socio-cultural beliefs, is a major component in achieving the necessary balance between the economic exploitation of natural resources and the impact of such exploitation on the environment. Together these elements determine whether Asia-Pacific development can be more sustainable. The following environmental impacts from high population growth (Table 12.1), some of which have already been discussed, have been identified by Thistlethwaite and Davis (1996) for Melanesia. These are widely applicable in much of the Asia-Pacific region.

# Table 12.1 Environmental impacts of highpopulation growth

- soil erosion, declining soil fertility and siltation, and fallow reduction in agricultural lands
- · coral reef degradation from on-shore activity
- · erosion and soil/water degradation from forest logging
- · biodiversity reduction from agricultural and urban expansion
- · fish stock reduction through over-fishing
- · mangrove destruction for housing, fuel wood and other developments
- · water pollution from industrial developments and urban expansion
- · air pollution, especially from poorly maintained motor vehicles in town and cities
- · the destruction of ecologically fragile lands through settlement and cropping spread

Source: Thistlethwaite and Davis (1996)

### Population pressures on sustainability

In terms of traditional production a key question is whether such systems can adapt to increasing population; or will they inevitably disappear? How can subsistence producers, many of whom are still largely bound by customary resource ownership and traditional means of production, cope with such pressures? An obvious response is the limiting of population growth, by family planning or other means. However there is no guarantee that, in the short or medium term at least, such an approach will make a significant impact. Thistlethwaite and Davis (1996, pp. 104–6) for example, discussing

environmental sustainability for the Melanesian countries, point out that large families were not only a response to formerly high levels of mortality but could also be seen as essential when there is no state social security system. Cultural values, such as the status derived from large families in many Pacific Island countries, or the traditional belief in the need for male children which formerly prevailed in China and other Asia-Pacific countries, may also counteract family planning programmes. While significant declines in contemporary population growth rates are certainly occurring in many Pacific countries it also appears that these are unevenly spread and are predominantly concentrated in the educated elite, mostly urban-based families. Although limiting population growth through family planning is clearly vital it must therefore be seen as providing long-term rather than short-term solutions.

Alternative approaches to maintaining a balance between population and natural resources include making resources more productive either by extending the areas under cultivation or being harvested for marine products, or by intensification of resource use. Such an approach, as earlier discussions have demonstrated, would not be new. It was an inherent part of traditional production systems, partly in response to increasing population and also, as Allen (1996) stresses, because such systems also had to provide for the overall wealth of the people through producing items to enhance status and generate individual and local political power. The other alternative is reduction of population through migration beyond the region which provides the natural resource base. People might be encouraged to resettle, either voluntarily or involuntarily, in other more sparsely populated rural areas or in towns and cities where alternative means of family sustenance through the cash economy are supposedly available. Such movement often must accommodate traditional resource ownership demands and pressures, and has both positive and negative environmental consequences.

### Environment, traditional production and rural-rural population movement

Because of the prevalence of shifting cultivation within Pacific traditional production, population movement within the rural sector has always occurred but, until population growth placed pressure on resources, the impact on the environment was limited. Resettlement under these conditions could be described as voluntary, and a natural response to the need to extend cropping areas was to move to new and formerly unused lands and thus maintain the population-resource balance.

The introduction of involuntary rural resettlement adds another element in the equation. Large-scale movements associated with Indonesia's programme to resettle rural families from densely populated Java to more sparsely populated parts of the archipelago such as Irian Jaya, or Vietnam's programme to resettle the thousands of people displaced through the construction of massive hydro-electric schemes such as the Hoa Binh dam on the Red River are well documented. China's contemporary massive development of dams across the Yangtze gorges will have similar consequences. Major issues arising in such cases include the fact that resettlement often takes place on the customary land of others, and that customary land-holders do not feel that their interests have been adequately recognized; and that the settlers have often come from rural societies with traditional production practices different from those appropriate to the areas into which they have moved. For example, the introduction of irrigated rice farming into the Indonesian

province of Irian Jaya, where the traditional production systems, like those in neighbouring Papua New Guinea, are based on swidden cultivation of sweet potato and taro provides a particularly graphic example of the problems involved.

In the Pacific Island countries rural population resettlement has occurred on a smaller scale. But its environmental effects have been similar. Throughout island Melanesia contemporary coastal villages commonly contain two distinct groups of people-the local group whose customary resource tenure includes both land and sea; and an incomer group, usually from inland parts of the island, with traditional rights to more distant land and with no traditional marine rights. Many of those groups were involuntary migrants, resettled in the early years of colonial administration as a device to improve the level of control which external powers such as Britain, France and, prior to 1914, Germany could wield over the indigenous labour force which they required to develop their relatively vast coconut and rubber plantations. Traditional production systems of both of these groups were inevitably affected and, with contemporary high rates of population growth, pressures on the environment have increased. Today the depletion of resources in the vicinity of these large coastal villages and escalating disputes over access to both land and sea resources are only two symptoms of the consequences of such movement. In general, while enforced resettlement may often be inevitable, it is dangerous to view it as a process which will peacefully solve imbalances in population and resources. The strength of traditional resource tenure systems in the Pacific will always be a vital part of the equation.

#### Rural-urban population mobility, environment and traditional production

Rural-urban population movement has led to very rapid urban growth in the Asia-Pacific in recent decades. Port Moresby, Papua New Guinea's capital, has expanded from 75,000 to over 250,000 since 1971 and Port Vila (Vanuatu) (Figure 12.1) has been growing at over 10 per cent per annum in the last decade; contemporary population densities in South Tarawa (Kiribati) and on Majuro (Marshall Islands) have been estimated at 1,354 and 2,025 persons per square kilometre respectively (UNDP, 1994; Connell and Lea, 1993). Smaller towns are also experiencing such changes. Economic and social problems such as rising urban crime rates and unemployment are now entrenched. Environmental problems have received less attention but provide clear warnings for the future. These include residential overcrowding, atmospheric and water pollution, inadequate waste management and sanitation, traffic congestion and, as a result of these, rising incidence of environmental health and safety problems. It is the migrants from the rural areas who are generally the poorest urban dwellers and most at risk.

A distinct characteristic of many Pacific Island towns and cities is that traditional production, often coupled with customary land tenure, has continued to provide part of the economic base. This is a prime element in the informal economy, whereby many town residents support themselves by producing fresh food or catching fish to trade at street stalls; or by selling logs for fuel wood. Although much of the recent expansion of Pacific cities and towns has occurred within areas already designated for urban development, the rapidity of growth coupled with lack of designated urban land has also meant that urban settlement has encroached, often with complete lack of planning, on

land still held under traditional resource ownership. Box 12.3 outlines examples of these problems in contemporary Port Moresby.

Box 12.3 Port Moresby: population growth, land and the environment

Port Moresby, the capital of Papua New Guinea and largest city of the Pacific micro-states, combines both city and country lifestyles with resultant problems both of defining land ownership and of establishing a means for survival. These problems have clear environmental implications. Port Moresby includes both coastal villages which predate the growth of the town and are located on customary land, and also 'squatter settlements', inhabited by migrants from all parts of Papua New Guinea who have established themselves on vacant areas of public or customary land. In the latter case they may or may not have entered into formal agreement with the customary land-holders to whom those areas belong. Not surprisingly many tensions exist and customary land-holders, as yet unable to receive what they perceive to be fair compensation for the loss of their land, have increasingly seen themselves as 'second-class citizens' (Connell and Lea, 1993, Chapter 7). Both they and the migrant inhabitants of the squatter settlements also suffer from inadequate urban services—poor

garbage collection, lack of adequate water supplies and increasingly difficult access to vitally important supplies of the fuel wood on which many people still depend for cooking. The environmental impacts, demonstrated, for example, by the destruction of coastal mangroves cut green for fuel wood all around Port Moresby, are obvious. Such problems are unlikely to disappear as such cities become older. The descendants of migrants, born in the city, may not have rights to return to their forebears' rural lands, yet may still not be accepted as having rights in the town area. Effectively they will be landless, forced to continue squatting on vacant areas left open to them.

Such characteristics are not restricted to smaller cities such as Port Moresby. Huge metropolitan centres, such as Bangkok, also incorporate agricultural land remarkably close to the city's heart. Much of the food which feeds the city's population is still produced through traditional production systems in these areas and provides a livelihood for families who still lead a predominantly rural way of life.

Some Pacific Island towns and cities suffer from environmental problems which are partly site specific and partly human induced. These include water quality and management; waste management; and air pollution. Water and waste problems are particularly obvious on Pacific Island atolls where, in parts of Kiribati and Tuvalu for example, urban densities are now extremely high (well over 1,200 people per km sq., and locally over 5,000 per km sq.). Major problems include disposal of household wastes and sewage which contaminate lagoons and beaches; and difficulties arising from water scarcity.



Land and water pollution caused by a nickel smelter on New Caledonia

### Box 12.4 Fresh water—a scarce resource on atolls

A characteristic of many atolls, including those which make up Tuvalu and Kiribati, is the presence of a fresh water lens (deposit) just below ground level on the larger islands. The lenses are a renewable resource in that they are recharged by rainfall. In the absence of rivers or fresh water lakes, the only other source of fresh water is collected rainwater.

In Tuvalu, where the rainfall is higher and generally more reliable than in Kiribati, the majority of households have been fitted, under a United Nations Development Programme (UNDP) project, with rainwater tanks, metal roofing, plastic guttering and down pipes. Such substitution of polluted ground-water supplies by rainwater is one of the keys to the prevention of water-borne disease and to escape from health threats posed by lens contamination by bacteria, viruses, and dissolved nitrogen.

While the Kiribati government can declare any area a water reserve, in practice the lenses constitute a common property resource subject to open access through wells. As the density of the human population above a lens increases, the continuous supply of potable water can be jeopardized both by drawing down water at a rate faster than replenishment, which encourages salt water intrusion, and by pollution through the very permeable overlying soils.

The fresh water lenses on South Tarawa (Kiribati) presently provide an extraction rate of 1,250 cubic metres per day to some 3,500 connections (out of about 4,000 households in total) but the supply is only available for one to two hours for three periods each day. The main source of water on South Tarawa for 40 per

cent of households is wells that directly tap the lenses, while rainwater tanks are a main source for 35 per cent of households.

The supply of fresh water to the increasing population of South Tarawa, and to islands with growing populations, such as Kiritimati, is a major development issue. While untapped lenses can be utilized lens supply is limited. It appears as though heavier supplementation by collected rainwater will eventually be necessary. However, the collection of rainwater will probably be an expensive option on South Tarawa given its relatively low and erratic rainfall. And collecting rainwater may reduce the rate of lens replenishment.

As yet air pollution is a much more severe problem in Asia-Pacific mega-cities such as Beijing, Bangkok and Manila than in smaller urban conglomerations. While major contributors are factory and vehicle emissions, particularly obvious because of the prevalence of cheaper, less energy efficient technology, physical location can also contribute. In Beijing, for example, the frequent occurrence of temperature inversion allows emissions to persist and dirty fog, reminiscent of the 'smog' which affected cities like London in the 1950s, is very common.

### Pacific international migration-relief of resource stresses?

As Ward (1996) has recently summarized, many Pacific peoples were not only highly mobile in the past but have in recent decades participated in a new migration story. This has created a contemporary diaspora not only of increasing numbers of Asian peoples to the United States, Canada and Australia but also of Pacific Islanders to many parts of the Pacific Rim. While it is difficult to obtain accurate estimates of the exact numbers of Pacific Islanders living in the Pacific Rim it seems that over 400,000 people who could claim to be Pacific Islanders by birth or descent are now living in countries such as the USA, Australia, New Zealand and Canada and even further afield in metropolitan France. It has been estimated that 37 per cent of the total population of the Polynesian countries now lives elsewhere. The significance of these international movements suggests that this process must relieve pressures on the limited resources in Pacific Island countries and also allow necessary adjustments in traditional production systems to accommodate demands for rising living standards. Is this actually the case? The evidence is conflicting. Unlike earlier migrations these relocated groups can and often do retain strong contact with their kin in the 'home' country and many also retain their traditional resource rights.

The international migration of Pacific Islanders has certainly slowed overall rates of population growth in some micro-states and this must therefore relieve pressure on resources used for subsistence purposes. More Cook Islanders and Niueans now live overseas than at home. Families living and working in New Zealand or Sydney also send significant amounts of their earnings back home, not only to support their families but also to finance important local social institutions, including church groups; and for local business investment. In fact, as Ward (1996) stresses, to discuss Pacific sustainable development in terms of the population-resources of the islands alone is no longer relevant. The equation is much more broadly based. Resources flowing from overseas

within these kinship networks can do much both to enhance technological advancement and raise living standards at home.

The relationship between international migration, lower rates of population growth and pressures on local resource use is, however, more complex than it appears. One important component of that complexity is the persistence of traditional production systems. As many commentators have highlighted (e.g. Ward, 1996) Pacific population mobility has shown a strong and persistent element of 'circularity', a process through which migrants maintain their roots with home villages and kin, and may frequently return to live there for significant amounts of time. Circulation occurs within both internal and international migration. Thus Cook Islanders or Western Samoans living in Auckland may not only send money to their families in Rarotonga or Apia but may also seek to hold on to their customary land and marine tenure rights, thereby inhibiting processes which might otherwise see these rights redistributed to meet the needs of poorer villagers who have stayed at home.

### 12.5 Environmental impact of large-scale resource use and traditional production

Large-scale resource utilization stems not only from the development priorities adopted by Pacific Island countries themselves, seeking to create formal sector jobs and to increase foreign exchange earnings, but also from the search for resources by highly industrialized countries of the 'Western' world, many of which are located in, or have colonial ties in, the Asia-Pacific; or by rapidly industrializing Asia-Pacific nations themselves. Increased competition for favourable access to these resources is a major feature. It is reflected in the demand for land for the establishment of plantations of palm oil, rubber and coffee; in tropical forests where logging has progressively extended from mainland South-East Asia (e.g. Peninsular Malaysia), to island South-East Asia (e.g. Indonesia), to Melanesia (e.g. Papua New Guinea, Solomon Islands); in the use of marine resources, where commercial tuna harvesting is primarily conducted by fleets from the USA,



### Mining phosphates on Nauru

Japan, Taiwan and the Republic of Korea; and in mining for gold, copper and diamonds by companies from industrialized nations such as Australia, Canada, the USA and France. In most cases Pacific Island countries, and particularly the micro-states, lack the capital and technological resources to enter this competition and can only derive benefit primarily through royalty and rental payments. Significant direct environmental, economic and social impacts are, however, inevitable, particularly for the local people whose land and other resources are coveted.

Forests, fish, crops and even minerals have all been used under traditional production systems and customary resource tenure regulated their use. While this type of small-scale, labour intensive activity is of little relevance to the operations of modern mineral and logging companies the rights to the resources themselves still have to be recognized. Thus large-scale resource development generally involves negotiation with traditional owners, both for access to the resource and for the establishment of rental or royalty-type agreements. An additional step, increasingly sought by contemporary traditional owners, is for joint venture arrangements whereby the developer guarantees local employment and training, business enterprise support and possibly better service provision in exchange for agreement that the two groups will work closely together. Here we briefly examine some issues relating to the Asia-Pacific mining sector.

### Mining, traditional land and the environment

Minerals are of great economic importance in the Asia-Pacific region. In industrialized resource-rich countries like Canada and Australia, which derive a high proportion of their export earnings from minerals, the mining industry wields considerable economic and political power both at national and state/provincial levels. In less-developed, mineral-rich Asia-Pacific countries, such as the Philippines, Indonesia and Papua New Guinea, high earnings from mineral exploitation may be the only element which can counter national financial debt. Under those circumstances it is not surprising that the pressures on land-holders to allow mining to take place with as few restrictions as possible are very strong. Where such land-holders still operate within traditional production systems these pressures, like those influencing the adaptation of agricultural systems or the exploitation of fisheries and forests, often lead to conflicts concerning the economic, social and environmental effects of mining and undermine the sustainability of the regions within which mining occurs. Conflicts occur both between the external operators—the state and the mining company—and the local land-holding community; and within that local community.



**Figure 12.3** *Papua New Guinea: regions and mining sites* 

A prime example of both types of conflict can be seen in the long-standing saga of Bougainville Copper (Papua New Guinea), which, over the last 25 years, has seen negotiation and re-negotiation of royalty, compensation and other agreements between the Australian-based CRA company, the Australian Federal Government, Papua New Guinea's national government, the North Solomons (Bougainville) provincial government and the local land-holders—over the mine at Panguna (see Figure 12.3). Major problems have stemmed from the failure of governments and developers to recognize the complexities of traditional land tenure systems. These have included conflicts between local groups over the distribution of benefits and compensation from mining; and the destruction of traditional forms of production because of environmental degradation and land loss. Complete breakdown since 1989 has led to the closure of the mine, with immediate severe impact on Papua New Guinea's balance of payments; the eruption of civil war, with disastrous economic and social consequences for the people of the island and extremely damaging political consequences for the country as a whole; and the cessation of any attempt to deal with local environmental detriment, such as loss of land or downstream river pollution, acknowledged as stemming directly from the mine. Recent exposure of the national government's plans to use foreign mercenaries to end the war has again escalated the political conflict, particularly since there are suspicions that
this move is related to arrangements to re-open the copper mine without necessarily obtaining the agreement of the local people. At present the North Solomons people are undoubtedly the losers—they have lost cropland, their rivers are contaminated, a whole generation of children are growing up with no schooling, infant mortality has risen and families have been ripped apart by the civil war.

The Bougainville story may be extreme. But it has provided lessons for newer mining ventures such as Ok Tedi copper, Porgera gold and Kutubu oil on Papua New Guinea's mainland (see Figure 12.3). Recent production from these developments has again boosted mining earnings to over 60 per cent of Papua New Guinea's export revenues. Problems similar to those at Bougainville have occurred—notably massive ecological loss downstream in the Ok Tedi and Fly rivers into which copper tailings are washed; and serious arguments at Porgera over entitlements to shares of compensation moneys, some of which excluded downstream land-holders who have been subsequently affected by the mine. While Lihir Gold (a consortium of RTZ, the Papua New Guinea government and other share holding interests), due to begin production in 1997, has apparently made determined efforts to anticipate and ameliorate the local economic, environmental and social impacts of the mine, uncertainties remain (see Figure 12.3). Here the location of the mine, on a small island in the north-east of Papua New Guinea, and proposals to deal with waste at least partly by ocean disposal raises concern over the impact on marine resources.

The environmental effects of large-scale mining on traditional production systems are not restricted to less-developed Pacific Island countries like Papua New Guinea. They have also been a major issue in Australia, where mineral rights are generally controlled by the state. In remote areas large mining companies operate on land held either under unalienable freehold tenure by Aboriginal groups, or which is alienated, but over which, from their own perspective, Aboriginal people still hold customary rights (Young, 1995). In the former case, exemplified by the Ranger Uranium development in Arnhem Land (see Figure 12.1), both government and developer were obliged to negotiate with traditional owners to ensure a fair return of revenue for compensation and the support of local Aboriginal development, and had to establish a monitoring process to gauge the environmental effects of waste disposal, particularly vital in the case of possible contamination from uranium extraction sites in a tropical wetland area. The social impact of the mine, although the subject of independent review, received much less attention. These issues are once more contentious. A recent Federal government decision to open up an additional uranium mine at nearby Jabiluka, the subject of a development agreement between a mining company and Aboriginal land-owning groups in the early 1980s, is now being firmly opposed by traditional owners. They say that the situation has changed-they are now much more aware of the social consequences of mining, such as high alcohol consumption, violence and family breakdown; and they are no longer convinced that the uranium wastes do not, in the long term, threaten the wetland environments on which their traditional production systems depend. Royalties, while they have given the people valuable opportunities for economic advancement, cannot necessarily compensate for such losses.

Where mining has occurred on alienated land Aboriginal benefits are even harder to perceive. Western Australia's Argyle Diamond mine, under CRA's Ashton Joint Venture has occurred on such land in a state which has not recognized indigenous land rights legislation (Figure 12.1). Here the Aboriginal traditional owners have received a pittance in terms of compensation moneys, the distribution of such moneys has, in their eyes, been quite inequitable, and the environmental destruction—of a site of great cultural significance to the traditional owners—enormous and irreparable (Dixon and Dillon, 1990).

The creation of joint ventures, whereby local land-holders benefit through direct participation and profit-sharing in the mining project, is another approach which may decrease some of these obvious inequalities. As yet joint ventures in mining, normally involving investment through a development corporation, are relatively uncommon. At present Canada, through negotiations under its comprehensive claims legislation, probably offers the best opportunities for traditional resource owners to establish their stakes in the mining industry. Small joint venture agreements, for example between aboriginal Canadian peoples and Chevron Oil in the Mackenzie River valley (Young, 1995), may provide pointers for other such agreements in parts of the Canadian Arctic.

Conflicts between traditional resource owners and mining must ultimately be resolved through careful negotiation processes within which all stake-holders have fair representation. As O'Faircheallaigh (1996) has mentioned traditional land owners can only achieve this with appropriate supportive legislation, financial resources to provide them with essential expertise, accurate information and sound advice, and genuine goodwill both on the part of the developers and the state. The establishment of 'buffer organizations', exemplified by Australia's Aboriginal land councils, is a key element in reaching this goal (Kalit and Young, 1997).

#### **12.6** Conclusion

The relationship between Asia-Pacific traditional production and the environment is not straightforward. Evidence provided through studies of contemporary Pacific Island communities shows that traditional production systems, whether utilizing land or marine resources, are not static but have continually adapted to the needs of growing populations and to the demands of new cropping systems and new technologies. While such adaptations have arguably led to increasing individualization of resource ownership, it does not appear that traditional forms of common property holding are completely incompatible with the demands of the modern cash economy, and therefore the need to destroy these customary systems in the name of 'progress' must be carefully considered. People whose societies still value customary practices are not necessarily all 'losers'.

Nevertheless there are some very important issues to be considered, particularly in relation to population and the environment. Many Asia-Pacific populations are growing much faster than in the past and, while population mobility may alleviate some of the pressures on resource use its effects cause inequalities and are not necessarily clear-cut. In particular the retention of traditional rights by absentees may undermine the potential for Asia-Pacific peoples to cope with the threats posed by increased pressure on their own resources. Environmental consequences, both within the rapidly expanding towns and also in rural areas where both subsistence and commercial farming are practised, are potentially severe.

Another key issue concerns the demand for Pacific Island resources by external powers, both richer neighbours within the Asia-Pacific region and industrialized nations elsewhere. Whether one is considering mining, tropical forest logging or fishing, the lack of equity in terms of finance, technological know-how and expertise and political clout between the foreign development companies and the traditional resource owners is obvious. In the resulting conflicts it is usually the environmental consequences that are discounted by both sides. The former want the resources for their own economic benefit, and, unless forced, may pay little heed to the long-term environmental effects of their activities. The latter want the money, to raise their living standards, provide themselves with services otherwise not available, and generally participate in 'modernization'. Longer term considerations such as the compatibility of such forms of development with overall future resource sustainability may well be pushed out of sight. Here potential conflicts of interest between large and small Asia-Pacific nations, exemplified by China and the island micro-states, are bound to occur. Resolution of these problems is the key to the future of the Asia-Pacific environment.

#### References

- Allen, B. (1996) 'Land management: Papua New Guinea's dilemma', *The Asia-Pacific Magazine*, vol. 1, no. 1, pp. 36–42.
- Carrier, J. (1987) 'Marine tenure and conservation in PNG' in McCay, B. and Acheson, J. (eds) The Question of the Commons: the Culture and Ecology of Communal Resources, Tucson, University of Arizona Press.
- Connell, J. and Lea, J.P. (1993) 'Planning the future: Melanesian cities in 2010', Pacific 2010 Monograph, Canberra, Australian National University.
- Crocombe, R. (1994) 'Overview' in South, R., Goulet, D., Tuquiri, S and Church, M. (eds) Proceedings of the International Workshop on Traditional Marine Tenure and Sustainable Management of Marine Resources in Asia and the Pacific, Suva, Fiji, University of the South Pacific.
- Dalzell, P., Adams, T. and Polunin, N. (1995) 'Coastal fisheries in the South Pacific', paper delivered to the joint FFA/SPC workshop, *Management of South Pacific Inshore Fisheries*, Noumea, SPC.
- Dixon, R. and Dillon, M. (eds) (1990), *Aborigines and Diamond Mining*, Perth, University of Western Australia Press.
- Herr, R. (1998) 'Restructuring foreign and defence policy: Pacific Islands' in McGrew, A. and Brook, C. (eds) *Asia-Pacific in the New World Order*, London, Routledge in association with The Open University.
- Hull, T. (1998) 'Population' in Maidment, R. and Mackerras, C. (eds).
- Hunt, C. (1996) 'Property rights and environmental management on Pacific atolls', *International Journal of Social Economics*, 23, pp. 221–34.
- Hunt, C. (1997) 'Cooperative approaches to marine resource management in the South Pacific' in Larmour, P. (ed.).
- Hviding, E. (1996) Guardians of Marovo Lagoon: Practice, Place, and Politics in Maritime Melanesia, Honolulu, University of Hawaii Press.
- Hyndman, D. (1993) 'Sea tenure and the management of living marine resources in PNG', *Pacific Studies*, 16, pp. 99–114.
- Johannes, R. (1978) 'Traditional marine conservation methods in Oceania and their demise', *Annual Review of Ecological Systems*, 9, pp. 349–64.

- Kalit, K. and Young, E. (1997) 'Common property conflict and resolution: aboriginal Australia and Papua New Guines' in Larmour, P. (ed.).
- Larmour, P. (ed.) *The Governance of Common Property in the Pacific Region*, Canberra, Research School of Pacific and Asian Studies, Australian National University.
- Maddock, R. (1998) 'Environmental issues and their governance' in Maidment, R., Goldblatt, D. and Mitchell, J. (eds) *Governance in the Asia-Pacific*, London, Routledge in association with The Open University.
- Maidment, R. and Mackerras, C. (eds) (1998) *Culture and Society in the Asia-Pacific*, London, Routledge in association with The Open University.
- O'Faircheallaigh, C. (1996) 'Negotiating with resource companies: issues and constraints for Aboriginal communities in Australia' in Howitt, R. Connell, J. and Hirsch, P. (eds) *Resources, Nations and Indigenous Peoples*, Melbourne, Oxford University Press.
- Taylor, S. (1998) 'Pacific images' in Maidment, R. and Mackerras, C. (eds).
- Thistlethwaite, B. and Davis, D. (1996) 'A sustainable future for Melanesia?: natural resources, population and development', *Pacific 2010 Monograph*, Canberra, Australian National University.
- United Nations Development Program (1994) Pacific Human Development Report: Putting People First, Suva, Fiji, United Nations Development Program.
- van Fossen, A. (1998) 'Pacific Islands', 'Melanesia', 'Micronesia' and 'Polynesia' in Ecclestone, B., Dawson, M. and McNamara, D. (eds) *The Asia-Pacific Profile*, London, Routledge in association with The Open University.
- Ward, R.G. (1995) 'Deforestation in Western Samoa', Pacific Viewpoint, vol. 36, no. 1, pp. 76–93.
- Ward, R.G. (1996) 'Expanding worlds of Oceania: implications of migration', *Contemporary Migration in Oceania, JCAS Symposium Series 3*, Osaka, National Museum of Ethnology.
- Ward, R.G. (1997) 'Changing forms of communal tenure' in Larmour, P. (ed.).
- Ward, R.G. and Kingdon, E.B. (eds) (1995) Land, Custom and Practice in the South Pacific, Melbourne, Cambridge University Press.
- Young, E. (1992) 'Aboriginal land rights in Australia: expectations, achievements and implications', *Applied Geography*, vol. 12, no. 2, pp. 146–61.
- Young, E. (1995) *Third World in the First: Development and Indigenous Peoples*, London, Routledge.

### CHAPTER 13 The Pacific and beyond: APEC, ASEM and regional economic management

Richard Higgott

#### **13.1 Introduction**

The agenda of international economic co-operation has become increasingly important in the globalizing world of the last several decades. This trend has become especially pronounced since the end of the Cold War and can be as strongly identified in the Asia-Pacific region as in any other part of the world. As most of the chapters in this book demonstrate, East Asia is becoming a more significant economic player in the global economic order, beginning to compare favourably in brute economic terms with Europe and North America. As a consequence the global economic balance of power is becoming more regionally triangular in shape. While North America, Europe and the Asia-Pacific do not make three distinct 'trade blocs', three distinct powerful regional economies are, nevertheless, in the process of consolidation. These developments have considerable consequences for the management of global economic relations.

To date, the development of regionalism has not been at the expense of the multilateral system (see Chapter 7). For the happy coincidence of regional openness and global openness to continue the system needs to be managed. For that to happen, the institutional development of all three regions needs to have some symmetry. As yet, the level of institutional organization of the regional economies of the Asia-Pacific do not match the levels of institutional complexity to be found in North America and Europe. Yet the Asia-Pacific is a veritable laboratory for experimentation in regional economic co-operation. Thus the closing decade of the twentieth century represents a new departure in international economic relations for those East Asian states that—for most of the post-Second World War era—conducted their international relations in general and their international economic relations in particular within the context of the disciplines imposed on them by the bi-polar structures of the Cold War era.

Enhanced economic co-operation within the Asia-Pacific region via the development of the Asia-Pacific Economic Co-operation Forum (APEC) and the attempts to develop a new political and economic dialogue between East Asia and Europe via the Asia-Europe Meeting (the ASEM process) inaugurated in Bangkok in March 1996, offer the opportunity to break new ground in both the theory and practice of multilateral cooperation at both the intra-regional and inter-regional levels (**Smith**, **1998**). At their simplest, APEC and the ASEM represent attempts by the state policy-making elites of East Asia to consolidate the channels of economic and political communication with the other two developed regions of the globe—North America and Western Europe. This is not just regionally important, it is important for the overall management of the contemporary international economic order. One of the aims of this chapter is to demonstrate the importance of the Asia-Pacific's intra-regional and inter-regional economic management for the way in which the global economic order develops in the next century.

The final years of the twentieth century have been an important and innovative time in regional economic co-operation in the Asia-Pacific. However, there are limits to the pace and scope of enhanced institutional co-operation in the region that are often lost within the rhetoric of summitry and the bonhomie that is generated when Heads of Government and Senior Ministers meet. The first four APEC summits-held in Seattle (1993), Bogor (1994), Osaka (1995) and Manila (1996) scaled the rhetorical peaks. Similarly, the first ASEM meeting in Bangkok generated a substantial store of goodwill in most quarters. The Asian states represented at this meeting were the seven member states of the Association of South-East Asian Nations (ASEAN)-Brunei, Indonesia, Malaysia, the Philippines, Singapore, Thailand and Vietnam-and the North-East Asian states of Japan, South Korea and the People's Republic of China. (This group is also sometimes called the Asia 10'.) Also present were the member states of the European Union (EU). It might not have been, as European Commissioner Jorge Pina (1996) suggested, '...the most important event since Alexander the Great's journey to India'!! But it was '...adjudged a surprising success... those Heads of Government who found an excuse to stay at home must now be regretting their absence' (Nuttall, 1996.)

In examining APEC and ASEM, this chapter will offer some insight into the range of opportunities for further institutional co-operation within the Asia-Pacific and between East Asia and Europe. It will argue that institutional co-operation has a serious and non-substitutable role to play in intra-regional and inter-regional co-operation. It will also be argued that the development of these processes are the inevitable outcome of the growing global significance of Asia-Pacific dynamism. However, the chapter also attempts to offer a range of cautionary remarks concerning the limits to institutional co-operation. Its argument will be that we must never assume that institutional co-operation is the motor of change at either the regional or inter-regional level. Rather it plays a largely secondary, facilitative, consultative, trust enhancing and mediating role that accompanies real, material, integrative development at the level of economic production and exchange. *Market driven* economic integration drives change. *State driven* institutional co-operation provides a framework—with varying degrees of success—for the management of change. The chapter focuses on this framework.

The chapter is divided into four parts. Part one looks briefly at the theory and practice of regionalism in the Asia-Pacific region over the last several decades. It focuses on *de facto* structural economic co-integration. Part two makes some general comments about the evolution of APEC—in contrast to *de facto* economic integration, this is identified as *de jure* institutional economic co-operation. Part three comments on the state of interregional relations between Europe and East Asia as contextualized by the ASEM initiative. Part four compares the roles and institutional strengths and weaknesses of APEC and ASEM as actors in contemporary international relations.

The major theme implicit throughout the chapter is that both APEC and ASEM can play an important role in keeping the wider multilateral economic regime open in a post-USA hegemonic era. In a period when the USA seems to have less capacity and, perhaps more importantly, less will to underwrite the global economic system in the manner that it did for much of the post-Second World War era, alternative sources of strength are required to guarantee what economists would describe as the collective benefits of an open international economic system. APEC and ASEM cannot substitute for US leadership. The international order would not function efficiently without the USA playing a role commensurate with its economic and politico/military strength; but the USA alone can no longer unilaterally manage the system and other actors have important roles to play. APEC and ASEM, if managed and nurtured properly, can become significant vehicles for 'social learning' and the consolidation of the norms of multilateral citizenship. If not managed properly, they leave a large section of the international economic dialogue lacking in modes of intra-regional and inter-regional governance. They are thus not just regionally, but potentially globally, important developments.

There is an inevitable normative argument to this chapter—APEC needs to play a continued role in regional bonding and ASEM should be a positive force for continued multilateral openness. Currently the trans-Atlantic dialogue between the USA and Europe is strong, if not always harmonious. Similarly, via the USA's presence in APEC, there is now a vehicle for the trans-Pacific economic dialogue. The development of ASEM provides a third leg to the triangle of inter-regional economic intercourse, as illustrated in Figure 13.1. However, neither APEC nor ASEM can be 'force fed'. This will only lead to disappointment and possible rupture.



Note: Australasia + = Australia, New Zealand Canada, Mexico and Chile



## 13.2 *De facto* structural economic regionalization in East Asia and the Asia-Pacific

Defining region, co-operation or integration with reference to the Asia-Pacific is difficult. By many of the yardsticks of understanding-based on one or more of the criteria of ethnicity, race, language, religion, culture, history, economic or political cohesiveness-the states of East Asia lack a record of regional consciousness (Buzan, 1998). And while it is true that regional economic interdependence (seen through the lenses of deepening trade interaction and growing economic complementarity-see Chapter 2) may be developing this does not *ipso facto* imply development of a greater sense of region. There is no self-evident definition of region and the yardsticks of 'regionness' vary according to the policy issues or questions present and above all by what the dominant actors in a given group of countries at a given time see as their political priorities. While there may be structural incentives to regional co-operation (dramatic technological and economic growth and enhanced interaction flowing from it) regions are socially and politically constructed—that is, they need positive social and political action to advance them. APEC can be seen as an Asia-Pacific region-wide exercise in economic dialogue with serious aspirations to become the vehicle for setting regional economic policy direction for the twenty-first century. APEC is not, however, the only voice of region in the Asia-Pacific. The East Asian Economic Caucus (EAEC) consisting of the states that make up the 'Asia 10' is an East Asian response to APEC that rests largely on rhetoric and state-centred understandings of political interest to motivate it. Smaller exercises in regional economic co-operation such as the ASEAN Free Trade Area (ASEAN FTA)—an attempt by the ASEAN countries to develop a genuine free trade area amongst themselves-and some of the emerging natural economic territories (NETS) or growth triangles, as they are more commonly called (see Chapter 7), and that intersect state boundaries of several countries, can be thought of as yet a third level of regional economic interaction that may give rise to newer forms of geo-governance.

In order to appreciate these proliferating and multiple level understandings of region we need an analytical device. It is useful to distinguish between *de facto structural economic* **regionalization** (regional integration) and *de jure institutional economic* **cooperation** which is at the core of any serious understanding of the events in train in the Asia-Pacific (Yoshida *et al.*, 1994; Oman, 1995). Material explanations, familiar to economists, are central here. But ideational explanations of regionalization—that is, those that stress the importance of ideas and emerging questions of regional identity—are also important. At the heart of the material-ideational dichotomy is the relationship between *private sector economic power* and *public sector political authority* and the different levels of institutional policy competence—both national and international—that are, or are not, emerging to deal with the tensions inherent in this relationship.

#### De facto regionalization (regional integration)

Regionalization in East Asia is driven by growing intra-regional trade and foreign direct investment (see Chapters 2, 4, 7, 8 and 10). For this reason it can also be seen as structural. It is principally an undirected phenomena of the corporate world/private world. It is not necessarily policy driven by governments. The most important driving force is

the globalization of production networks—uniting a myriad of production units for the provision of components, materials and management for particular product assembly in numerous countries and thus fostering a continued momentum towards the further integration of economies within and across regions based on a web of production, sourcing and distribution, that is likely to accelerate (see Chapters 2, 10 and 14 in particular). This is a *structural* or *de facto* explanation of regionalization.

In effect, what we are seeing is the addition of a horizontal dimension to the regional division of labour that will ensure the emergence of new players and sectors. The globalization of technology, finance and production networks, in which Japan has had a central role, are the key to understanding this process (see Chapter 2; Barnard and Ravenhill, 1995; Barnard, 1996). Many of the states of South-East Asia exhibit a growing dependence on Japanese technology and finance (see Chapters 4 and 10). At the same time, the Japanese market continues to absorb a growing share of East Asia's exports. This gives Japan an increasing structural significance in the region; identified by Hatch and Yamamura (1996) as the 'Japanese embrace of Asia' that is consolidating a regional production alliance—if alliance is what it is since 80 per cent of Japanese exports and 50 per cent of Japanese imports are represented by intra-company trade (Encarnation, 1994).

Notwithstanding this growth, the USA still remains the largest market for most East Asian states, including Japan itself, and is likely to remain so for the foreseeable future. This process allows us to identify two distinct but quite contrary characteristics of *de facto* regionalization. On the one hand, we see the emergence of an increasingly distinct region of production identified as East Asia with Japan at its hub and coinciding with those states that make up the East Asian Economic Caucus, or the Asia '10', that were present at the first ASEM meeting. On the other hand, we see the consolidation of a pan-Pacific trading region, roughly delineated by APEC membership, and in which the USA remains central to the region (see Chapter 2). These evolving configurations have implications not only for the trans-Pacific relationship, but also for the future of the APEC-ASEM relationship discussed in a later part of this chapter.

While it might be the changing nature of both the global *and* the regional economy that provides the structural context within which regional interaction takes place, it is governments that are the principal *agents* of any regional *institutional* co-operation.

#### De jure co-operation

Governmental policy responses emerge from within a policy-making community consisting also of actors from the corporate sector and the wider intellectual communities of the Asia-Pacific. This tripartite policy community—governments, companies and policy advisors and research brokers—is the locus from which comes pressure on governments to engage in enhanced policy co-ordination and harmonization in areas such as competition rules, product, environmental and labour standards, and even erstwhile sacred areas of government sovereignty such as public procurement, so salient to the ability of firms to compete in the markets of the Asia-Pacific.

For sure, governments do not always act exactly as the wider policy community (especially the corporate sector) would like, but it is quite clear that bodies such as the Pacific Economic Co-operation Council (PECC), the Pacific Basin Economic Council (PBEC) and the Pacific Basin Forum (PBF) have been influential over time in the

advocacy and practice of regional economic co-operation in the Asia-Pacific. This is *de jure* co-operation. It can take a number of forms—loosely agreed, or institutionally sanctioned trade commitments between states to enhance co-operation in a range of areas at the soft end of the spectrum (to be found in the APEC process) or at the strong end of the spectrum in a range of activities of a more binding nature with a common external preferential tariff (CEPT) as the ASEAN FTA is trying to develop.

The development of stronger regional trade arrangements in the Asia-Pacific is not discrete from the other dimensions of 'regionness' that are developing. State decision makers are capable of playing on more than one chess board at the same time. There are dynamics of regionalization internal to a given region, but these are also consolidated and operationalized in a global context in which regions identify and locate themselves *vis-à-vis* other parts of the economic and political order. As important as trade questions might be, a concentration on the political economy of trade has had the effect of causing us, until recently, not to focus on the other main game in the global economy—namely the political economy of capital mobility. The dramatic deregulation of the international financial system in the 1980s has meant that the competition to attract foreign direct investment (FDI) has become more intense.

But to explain efforts to reduce barriers to the continued development of intra-regional trade we also need to look to other factors. Welfare enhancement and the evolution of strong patterns of intra-regional trade may make good understanding in economic terms. But such static models do not help us understand the wider historico-structural context of agency (state) induced change. These points may be as, if not more, salient than trade flows and any 'natural' understanding of region. The spur to the liberalization process in the ASEAN, for example, has been a growing recognition amongst the membership of the ASEAN policy community that if the continued flow of international investment funds is to continue, ASEAN states have to maintain their competitiveness *vis-à-vis* other investment-hungry areas. ASEAN FTA, for example, is premised on an assumption that its members can offer a range of *collective* inducements to foreign investors and a *political* endeavour to enhance collective action not present at the state level.

## 13.3 *De jure* institutional regional co-operation—the prospects and limits of APEC

The dominant status of *de facto* processes and the supposed limited role of the 'political' dynamic in regionalization can be overstated. Nowhere is this better exemplified than in Kenichi Ohmae's exhortation to *The End of the Nation State and The Rise of Regional Economies* (Ohmae, 1995). Ohmae over emphasizes the picture of a world in which states have lost 'their role as critical participants in the global economy'. Processes of regionalization of terpresent a state-directed enterprise and indeed, the globalization of corporate behaviour, industrial production, capital mobility and communication have undermined much of the power once held by the nation-state. Yet Ohmae underestimates the complexity of this process in East Asia where a creative tension exists between the inter-state system, the regional and globalized networks of production and regional and global markets. The structure of the regional political economy is determined by the interplay of these political and economic factors. Greater

regional economic integration is not structurally determined. State policy elites are not passive actors. They have interests. In most cases we should assume that the intervention of state elites, to enhance the processes, is in part in their own interest.

#### Globalization, regionalization and the state

Because globalization weakens the efficacy of national policy instruments, a collective action approach to problem solving, with regard to issues demanding transnational management solutions, is probably easier at the regional level—or in theory this should be the case, given an expectation of at least some shared regional political, economic and socio-cultural understandings (see **Brook, 1998**). It is *within* a region that things seem more politically manageable. The governing elites of East Asia's major states, given their past history of conflictual interaction, might not have thought of each other as natural partners, but one effect of globalization is the evolution of a strong regional dialogue about economic co-operation. At over-lapping regional/state levels, the evolution of a strong policy community of the principal players (both public and private) has become increasingly wedded to the principles of liberalization. Entering into agreements with regional neighbours provides policy discipline in the face of domestic pressures the aim of which is to make up collectively at the regional level for the policy autonomy that is lost at the national level.

Members of governmental policy-making communities drawn into the trans-regional discourse on economic liberalization through bodies such as PECC, and PBEC and the ASEAN business groups have strongly supported both state-based and regionally-based coalitions for deregulation and liberalization. As a consequence, and at different levels, ASEAN, the EAEC and APEC should be seen as compromise products of the competing views of different groups of important regional policy actors. Some of the non-Asian members of APEC (e.g. Australia, Canada, New Zealand, USA) tend to prefer a more legalistic/formalized approach towards co-operation than the Asians who prefer a more consensus, trust-oriented approach. At first this was largely a rhetorical debate over styles and speed, but as APEC's agenda has begun to firm up it has taken a more concrete form. Since the Seattle summit there has been an attempt to speed up the development of APEC, especially via the proposals of the Eminent Persons Group, the Bogor Declaration's commitment, rhetorical at least, to full liberalization by 2020 and the Osaka Action Agenda (see Chapter 7). But if the aim of APEC has been to share information, enhance transparency, and build trust via regular interaction where it has not previously existed, this is not as unproblematic as it sounds. Most Asian members of APEC resist it becoming a formal negotiating body rather than one that simply affirms broad principles and develops modus operandi. If this is to be done in an explicitly Asian way, then it is going to take a much greater level of intellectual and practical leadership from the region.

#### Inter-state co-operation and 'Asian' identity

APEC and other regional inter-state co-operation bodies such as ASEAN and its ancillaries (ASEAN FTA and the ASEAN Regional Forum (ARF)) are *statist*. They are seen as a way to enhance regime legitimacy. In contrast to the common East Asian perception of the European Union (EU), East Asian regional organizations are geared to

sovereignty enhancement not sovereignty pooling. APEC is determined not to replicate the institutional structures of the EU (Higgott, 1995). Moreover, open regionalism—the progressive liberalization of trade within the Asia-Pacific region via concerted unilateral liberalization but which is extended to non APEC members on a most favoured nation (MFN) basis—is invariably contrasted with the institutional-cum-discriminatory EU model (MFN refers to the way that a favourable trade rule extended to one country should be extended to all trading partners).

The institutional development of APEC to-date has not been as slight as some of its detractors would suggest. In addition to its many working groups, regular meetings of senior officials and a functioning secretariat in Singapore, the summits have forced leaders to make commitments. That these may not be honoured in the short term is less important at this stage than that they appear in the official record of the organization. Moreover, advocates of APEC ignore the dynamics of the domestic politics of its member states. Many Asian governments, the public rhetoric of APEC summitry notwithstanding, are not fully convinced of the virtues of either open regionalism or concerted unilateral liberalization.

This having been said, there is a process of 'enmeshment' taking place that alters the dynamics of inter-state relations in the region. There is a growing desire on the part of a wide range of policy actors in the Asia-Pacific and East Asia to establish a greater sense of regional cohesion in order that the given region (APEC, the EAEC, or ASEAN depending on the level) might play a more significant role in the conduct of inter-state relations within the region and between the region and other international actors in a range of different issue areas. Questions of regional identity are becoming important in Asia-Pacific regionalism. An 'Asian' identity is viewed in part as a reaction to the way in which the USA is perceived to treat the region (Mahbubani, 1994) and as a way of stemming the intrusion of Western cultural and moral value systems without rejecting the dynamic aspects of Western economic and technological modernization. Importantly for the future theory and practice of regionalism in the Asia-Pacific, the issue of an 'Asian' identity is beginning to emerge in Japan as well as in the ASEAN states. It is the role of Japan in Asia, rather than the relationship of Japan with Asia, that will determine the nature of twenty-first century co-operation. Equally significant, although it cannot be dealt with in this chapter, will be the role of China as it continues to develop its economic potential (see Chapter 14).

In the Asia-Pacific context the growth of a notion of an 'Asian' identity—no matter how loose—has significance to the extent that outsiders might come to feel increasingly obliged to define their policies to individual states in regionalist terms. Thus to the extent that the notion of an 'Asian way' to diplomacy has any meaning, it is more in how it is received and responded to by actors *exogenous* to the region rather than to the degree that it affects the behaviour of actors within the region. 'Asian identity' arguments have shifted the international discourse on a range of issues, such as human rights or economic relations, into a mode more amenable to East Asian elite interest.

While East Asian policy elites attempt to stress their differences from the ordered rationalism of European and North American regional integration, their economic policies, especially trade policies, already exhibit greater institutional constraint than is often acknowledged. Institutions—defined here as organized rules, codes of conduct and structures that make gains from co-operation possible over time by solving collective

action problems—despite uncertainties, are important to East Asian policy makers as a kind of socio-political cement that mitigates self interest and opportunism. Interaction within the context of these institutional settings creates a move in a particular direction and constrains vested interest in these settings and arrangements where priority is attached to process rather than to outcome. Co-operative social learning through iteration is both theoretically possible and is occurring in practice. There is evidence in Asia-Pacific that adjustment to the principles of free trade and open regionalism in an era of regional growth is taking place in important quarters of the economic policy community (Higgott, 1994a; Harris, 1996). But this is not an inevitable process. Domestic political pressures mean that Asian-Pacific policy elite responses to regional co-operation can be selective and tactical rather than cognitive and universalist.

The development of Asia-Pacific economic institutional structures and symbolic notions of identity are part of the re-ordering process accompanying the end of the Cold War bi-polar political structure and the post-Second World War economic structures once underwritten by US hegemony. Institutional arrangements are now less coherent than in the era of these disciplines. Moreover, institutions are likely to have overlapping competencies in a globalized era. The world economy reflects the combined influences of twentieth century technology, a nineteenth century free trade ideology and the reemergence of a polycentric alternative to the modern state system. Explaining this reordering process is not an exercise in economics as much as one in international politics. International relations in Asia-Pacific would appear to be adapting accordingly.

#### Summary

We do not have to accept the cartography of the 'borderless world' (Ohmae, 1990) to recognize that we are entering an era of diminished state autonomy and sovereignty. The implications of these theoretical insights for the practice of the global trading system we are only just beginning to imagine. In an increasingly inter-linked and globalized world the distinc-tion between domestic and foreign economic policy is losing meaning and the presence of multiple identities, loyalties and conflicting sovereignties are more common. A globalized economy run by overlapping and inter-connected networks of state and non state actors in both public and private domains is mitigating the significance of space and territory. Symbolic understandings of space now exist side by side with geographical understandings of space. Such insights are not mere abstractions. They help unpack the multiple understandings of region in East Asia and the Pacific. They help us to explain why it is rational at one and the same time to talk of East Asia, the Asia-Pacific plus a range of sub regional domains such as ASEAN and ASEAN FTA, and the growth triangles such as those of Southern China and the Sea of Japan zone. They also help explain the growing interest in developing new relationships between the states of East Asia and Europe via ASEM.

#### 13.4 ASEM: closing the global inter-regional triangle

ASEM as an inter-regional dialogue process is the product of conscious decision making on the part of significant regional leaders in East Asia and Europe (see Camroux and Lechervy, 1996). This has to be the case because no formal institutional mechanism has previously existed. This act of political will, however, is built on a recognition of global structural changes that make such a dialogue necessary. What these changes imply might not seem initially self evident since the notions of 'Europe' and 'East Asia' are both arbitrary constructs and both areas are extremely heterogeneous. Moreover, while 'the idea of Europe' is at least European in origin, 'the idea of East Asia' is effectively an idea non-Asian in origin. So why are the closing years of the twentieth century witnessing attempts to institutionalize such a dialogue?

Notwithstanding its current problems and occasional setbacks, the direction of the European Union since the Treaty of Rome is not contested. It is much more institutionally advanced than East Asia (*qua* Asia 10). It has secured a single integrated market (SIM) and is struggling towards a single currency, a central bank, a common foreign and security policy (CFSP) and extended membership in both the medium and longer term (Wallace and Wallace, 1996). East Asia's path to greater institutional co-operation is much less predictable. Several strong aspects of regionalization—the idea of the region(s) of East Asia as shared economic space, the need for a regionalization of the security dialogue (through the ARF) and even the notion of a collective, albeit instrumentally limited, use of the notion of a common East Asian identity for responding to some extra regional questions in international relations—are emerging.

There are, however, several seriously contestable security issues in the region that have the capability of tearing it apart. It is not necessary to be a pessimistic realist expecting conflict to happen to recognize what could evolve in worst case scenarios. Inter-regional dialogue, across the security and economic domains, is thus important. At the most basic level therefore ASEM, for its East Asian members, is one more pillar in an emerging regional architecture that helps consolidate other useful emerging tendencies towards dialogue and co-operation between them. In this regard, even the symbolic and practical utility of co-ordinating positions prior to an ASEM meeting is not unimportant for East Asian states beginning to secure an understanding of their collective regional importance in world affairs—as was the case at the initial Bangkok summit in 1996.

#### East Asia and the EU

The growth in importance of East Asian states in the global economy requires that they play a greater role, both individually and collectively, in the management of the global order—at both the regional and the global level—than previously. In order to do so a range of mechanisms of dialogue and policy co-ordination are required. ASEM, along with ASEAN, the ASEAN FTA and APEC, constitute that growing basket of fora for enhanced regional policy intercourse and co-ordination. ASEM has given a form of legitimacy to Dr Mahathir's vision of the EAEC—which is of a free trade area *just* for East Asian nations (see Sopiee, 1996). The development of a regional identity in East Asia does not imply duplication of the EU model. But if the argument that *de facto* regionalization is taking place in the space defined as East Asia—as outlined in part one of this chapter—is accepted, then the concomitant development of a *de jure* institutionalized political dialogue is hardly surprising.

There are, of course, other substantive reasons why ASEM has emerged at this particular time, the most basic of which is to counter the under-developed state of

economic relations between East Asia and Europe. East Asian states have important markets in Europe and see the need for the EU to remain open. Similarly, ASEM holds an attraction for Europeans given that they are excluded from the APEC process. East Asia is the EU's largest trading partner. In 1994, it took 23 per cent of the EU's foreign trade compared with the 20 per cent which went to the rest of Europe and the 17 per cent which went to the USA. More importantly, the EU's trade with East Asia is growing faster than its trans-Atlantic trade. It trebled between 1985 and 1992. If present growth rates continue, it will be more than 50 per cent larger than the EU's trade with the USA in the coming decades.

The trade relationship between the two areas—primarily the product of unregulated private sector activity—is significant, but could be much greater in an era of strong growth and globalization. Thus this is a time of both considerable opportunity and risk for the inter-regional relationship. The relationship in these early stages is dynamic but brittle. The Bangkok ASEM generated expectations, but not without residual suspicion and tensions between individual participants—for example, Indonesia and Portugal over East Timor. As the EU Commission's policy paper *Towards a New Asia Strategy* (1994) points out, a European failure to develop a dialogue with Asia will give succour to those in Europe who see Asia as a threat (especially to employment levels) rather than as a partner.

But given that the market place is generally thought, in both parts of the world, to be the best place to develop trade and financial relations we need to consider the subsidiarity question—what can governments do in this context that is not done by markets? The answer lies in the area of information and networks. Markets function best with perfect information and understanding. It is clear that knowledge about the two parts of the world is not as substantial as is often assumed. Networks are not as strong between East Asia and Europe as they are either across the Atlantic or across the Pacific. Thus building inter-regional networks and information conduits is considered strategically important by the policy elites of both regions. The explanation for this lies in the wider context of the multilateral arena and especially in their relationship with the USA, as the third arm of the triangle, and in support of the multilateral trading system embodied in the World Trade Organization (WTO).

#### ASEM and the USA

The USA, is a central player within ASEM even in its absence. For both the Europeans and the East Asians, ASEM offers an opportunity to balance the love-hate relationship that both groups have with the USA. A relationship that neither group can do without, but one in which they feel that they are often treated as the junior partner subservient to US will. In the post-Cold War era, the thinking in policy circles of both Europe and East Asia is that it is time to balance out their relationship with the USA more than was the case when they were subject to Cold War disciplines. As is now well articulated in policy circles, ASEM offers the opportunity to close the third side of the North America-Europe-East Asia triangle, or to confirm the tripolar relationship.

This is not, nor should it be thought of as, an opportunity for ASEM members to act in a negative or obstructive way towards US policy initiatives or practices. But more as an opportunity to enhance relationships with the USA and particularly to make sure that East Asian and EU voices can be used to reinforce continued liberalization in the multilateral trading system. This will be necessary in the face of recurrent bouts of economic nationalism by the USA. There is strong evidence that the USA, while not closing its markets, is moving away from a diffuse, to a more specific, form of reciprocity in its international trading relations (Higgott, 1996; Bhagwati, 1996). Nowhere is this better illustrated than in its attitude towards the General Agreement on Trade in Services (GATS)—where MFN at US insistence does not apply—and its attempt to spread the use of the 'extra-territoriality' principles (the imposition of US trade policy initiatives on third party countries—see below). This is not to suggest that European and Asian states are always going to be the new found champions of free trade. This is known not to be the case, rather, it is to suggest that we are seeing a change in the attitudes of the USA and East Asia to the traditional script that governed the post-war international trade regime. This change is twofold:

- 1 The USA is becoming a 'normal', as opposed to a hegemonic, country. Normalcy here can be taken to mean the USA is developing all the usual vices of states seeking to optimize, by fair means or foul, their competitive advantage in what is for them now a much more fiercely competitive global market than at any time in the last one hundred years.
- 2 In contrast with (1) there is a growing recognition on the part of the majority of the East Asian economic policy-making community that their competitive advantage may be best served by an open, liberal international trade regime underwritten by adherence to a set of norms, principles and codified rules arbitrated by a multilateral body—the WTO. These disciplines may protect them from the behaviour of the USA on the one hand, especially its desire to make bilateral deals in which the means are justified by the ends. On the other hand, WTO disciplines can also be tools with which to curtail rent-seeking groups in the domestic economy.

In this context, it may become common to see Europeans and East Asians acting jointly to confront the USA with some of its more outrageous transgressions of multilateralism. For example, the Helms-Burton Act and the d'Amato Bill to punish third parties trading with Cuba, Libya and Iran—or any other states to which the USA might temporarily take a dislike (so called 'extra-territoriality'). As Gerry Segal (1997) comments, a role for ASEM is to keep the Americans 'honestly committed to multilateralism' and at a more general level, collectively oppose any other attempts at what Bhagwati and Patrick (1990) call 'aggressive unilateralism'.

If this is done properly and constructively—that is if it appears to be underwritten by a commitment to open multilateralism and not posited in zero-sum (we win, they lose) terms for American observers—then ASEM will represent a useful addition to international trade diplomacy. Theoretically US foreign policy makers should not object to ASEM. It was quite clear that APEC, and especially the calling of the 1993 Seattle summit, was used as a negotiating stick with which to beat the Europeans when the Uruguay Round of multilateral trade relations became bogged down in a USA-EU stand-off in its closing stages. If the APEC strategy succeeded in securing the necessary concessions from the EU to close the round—and the anecdotal folklore says that it did—then logically it makes sense to repeat the formula in other contexts. ASEM thus becomes the balancing leg on a tripod of inter-regional negotiation. Unlike the Europeans, who can

respond via the EU, the East Asians have no serious vehicle for a co-ordinated response to the more overt bouts of US economic nationalism. History tells us that Europeans do not take to bullying by the USA as easily as individual East Asian states, for obvious historical reasons of weakness, did for most of the Cold War era. Throwing in their lot with the Europeans on some of these issues may improve East Asian negotiating positions.

More generally, ASEM, and APEC, are bodies in which East Asians interact with 'the West'. In so doing, such bodies mitigate the chances of Samuel Huntington's rather unfortunate proposition of a clash of civilizations having any long-term credibility (Huntington, 1996). The existence of APEC and ASEM dissagregates the notion of the East. This is important from European, American and East Asian perspectives. ASEM is especially important from an East Asian perspective in this regard since when many East Asians think of the West they are, in effect, thinking of the USA as opposed to that disparate group of countries that make up Europe. While much of the debate about competing capitalism has been posited as three systems-American laissez faire liberalism, European Social Market Capitalism and the East Asian 'developmentalist' or 'developmental state' model-this has merely disguised the fact that there may be more in common in the approaches towards economic development of Germany and Japan than Germany and the UK or Japan and Malaysia. ASEM may help break down these residual stereotypes. For this process to be consolidated, changes need to come about in the structure of knowledge in both sets of societies. If ASEM is successful then the issues and agenda items of the inter-regional relationship will be incorporated much more comprehensively into the 'core thinking' of the respective tripartite-government, business, research-policy communities of the two regions.

The importance of ASEM is to provide a template within which relationships between East Asia and Europe can develop anew. Interdependence between the two areas is growing and there are real opportunities for reciprocity and co-operation that enhances the interests of both areas. At an economic level, closer relationships provide growing markets in both areas and the prospect of technology swaps. This is not a one way street. Not only do some East Asian states have higher GNPs per capita than some European states (contrast South Korea, Taiwan, Hong Kong, Singapore with Ireland, Spain, Greece and Portugal) they are also technologically and industrially more sophisticated. Notwithstanding European reliance on certain East Asian technologies such as semiconductors, technology transfer is still primarily from Europe to East Asia at the moment. However, a greater symmetry in this relationship can be expected to develop over time.

Since Europe and East Asia, in contrast to the more homogenous USA, are the world's most culturally diverse market places, both regions have a strong vested interest in an open multilateral system. If one of the important things about the gradual institutionalization of APEC is the facility it offers for social learning amongst the policy communities of the region, then one of the potential benefits of a range of enhanced contacts through the ASEM process, and its ancillary activities—at the secondary level such as the creation of the Council for Europe Asia Co-operation (CAEC), the Asia-Europe Foundation and European representation within the Council for Security Co-operation in the Asia-Pacific (CSCAP)—is the opportunity it offers for mutual learning between Europeans and East Asians. Differences and misperceptions exist in a range of

important areas. Rhetorical assertions that East Asia and Europe need to deepen mutual understanding are more important than is often the case in the language of diplomacy.

But just as APEC cannot drive the agenda of intra-regional economic development in the Asia-Pacific, then neither is ASEM in a position to drive the inter-regional economic relationship. In both cases it is private sector market power, not state sponsored institutional direction that is the determinant factor. ASEM's role should be one of providing infrastmetural assistance to the relationship—such as data gathering and assistance in the creation of networks and inter-regional policy learning. This role is not dissimilar to the role that APEC has played in trying to strengthen levels of understanding across the Pacific.

The very existence of ASEM, while it does not suggest a readiness to share sovereignty, implicitly acknowledges that there are limits to state power and that collective discussion and successful policy co-ordination require some voluntary curtailment of the pursuit of pure state interest. Whether social learning will take place within ASEM depends on a number of factors. Perhaps none more pressing than the role that China plays in the organization. China's policies in the politico-strategic domain currently reflect a clear preference for traditional realist understandings of international politics, but in its attempts to join the WTO, there is clear evidence that it has undergone a socialization process of sorts (see Harris, 1996). There is thus no logical reason why it should not do so in other fora.

#### 13.5 The limits to institutionalization: APEC and ASEM compared

Both APEC and ASEM have their dewy-eyed devotees and their myopic critics. Both groups often miss the real significance of these organizations. Notwithstanding their inevitable youthful institutional weaknesses, there are three significant points about them:

- 1 Along with the ARE, they represent the first real attempts at region-wide multilateral co-operation in issue-specific contexts in the Asia-Pacific.
- 2 They both represent indigenous—*East Asian*—attempts to manage various aspects of their regional and international relations. Unlike many other structures that have come and gone in East Asia over the recent centuries, they have not been imposed from outside but developed from within.
- 3 They are not only historically significant they also throw up innovative theoretical challenges to traditional understandings of the theory and practice of regional organization as it has developed in the northern hemisphere.

Traditional frameworks—be they realist or liberal—are too static to be useful templates for understanding developments in the inter-regional and intra-regional relations of East Asia. They fail to cope with the evolving processes that accompany dynamic economic growth, political change and identity building of the kind we are currently witnessing in the region. APEC and ASEM are East Asian influenced arrangements for the late twentieth century. They reflect East Asian views and East Asian interests. Yet they are not simple replications of European, formal legal institutional approaches. This is too simple a dichotomy. They represent a new form of regional organization—crudely, an amalgam of aspects of East Asian and European approaches that have developed from the dialectical relationship of regionalization with globalization.

Globalization and regionalization are at times complementary, at other times in conflict. The important thing about regionalization is the role it plays as a nascent mesolevel of authority between the global economy and the sovereign state. APEC and ASEM, to greater or lesser extents, are manifestations of nascent *de jure* co-operation that may, or may not as the case might be, over time enhance *de facto* integration and inter-regional co-operation.

APEC is more strongly institutionalized with a much more defined agenda than the ASEM processes. But APEC is at a crossroads. At one level it is the victim of its own surprising success—especially since the Bogor Declaration established its goal as the achievement of regional free trade by 2020. But as APEC passes the 'talk fest stage' and moves to a concrete agenda, fundamental differences amongst members that were initially submerged are thrown into sharp relief. The further APEC travels down the road to greater trade liberalization, the greater the pressures on it, and the more defined the dilemmas and differences between its members become. The notable issue here is the consolidation of the 'Asian way' towards liberalization at the expense of the institutional approach and what that does to the prospects of achieving the economic goals set at Bogor (see Chapter 7).

The essence of the East Asian approach to liberalization is the emphasis on concerted unilateral liberalization. Most East Asian members of APEC are more comfortable with this approach—supposed to shepherd all members to either the 2010 or 2020 deadline for full liberalization, but at their own speed and in their own way (a voluntary way as Dr Mahathir of Malaysia is always keen to point out). Non-East Asian members—especially the USA and Australia—have reluctantly gone along with this approach so far, but would prefer a fixed timetable with comprehensive and binding commitments. Developed members of APEC are concerned that they might fulfil their 2010 liberalization programmes only to see many of the East Asian members 'free-ride' to 2020 and beyond. If member state plans for liberalization, as a meaningful concept, will come under strain. The disappointing Manila Summit, at which the divide between a group of 'fast track' states led by the USA and a 'slower track' group of developing countries, supports this analysis. In spite of the difficulties in getting APEC to where it is today, it is possible to argue that the difficult times still lie in the future.

#### The future of the ASEM and APEC processes

The resolution of these difficulties will not be helped by the still undeveloped institutional structure of the organization. But the divide that exists over the pace and structure of liberalization also exists on the 'more-versus-less' institutionalization issue. Developed members—led by Australia and the USA and the APEC Eminent Persons Group between 1993–96—wish to see APEC strengthened as an institution whereas East Asian members, and especially the ASEAN states, prefer to see APEC more as a *process* than an *institution* (see Kahler, 1994; Higgott, 1994b). The East Asian position is likely to triumph. For the time being, progress in APEC will be via its summit meetings with the host nation taking the lead. Between 1993–96, summit hosts (the USA, Indonesia,

Japan and the Philippines) were what Gallant and Stubbs (1996, p. 14) call 'APEC boosters' that did much to ensure that APEC developed strongly. Over the next few years summit hosts will include Malaysia and the PRC. Both are currently reluctant participants in the APEC process. If some of the newer members of the organization also host future summits, it could be a long time—after 2010—before APEC boosters again have control of its agenda.

ASEM is clearly different in scope and aspirations to APEC. As yet, it has only a long wish list of items for discussion. There is no strongly structured, let alone institutionalized agenda for ASEM. Yet, as has been suggested, there is potentially an important role for ASEM but the consolidation of its initial agenda and a modest degree of institutionalization must come, and come quickly, if it is to fulfil its role as a supporter of the APEC agenda and as a facilitator for both the multilateral and inter-regional economic dialogue in a tri-polar world.

For example if the Europeans and the Asians were willing to use ASEM as a vehicle for addressing both the theory and practice of anti-dumping rules, rules of origin and investment codes—all of which have been exacerbated as potential areas of conflict by the evolution of preferential arrangements hiding under a loose interpretation of article XXIV of the GATT (which allows exceptions to the rules for freer trade)—then ASEM could become a major force in the dialogue over the development of the global trading regime under the WTO. Similarly, ASEM could play a role in the advancement of an inter-regional agreement on investment. While this is also a global priority—attested to by OECD and WTO work in this area—ASEM could help this process along. Cross investment from one region to the other is large and growing. Regulation is needed to make sure that it is managed in a way that minimizes friction between foreign investors and the host countries.

#### Liberalization and the domestic political agenda

Keeping regionalism open is about keeping the world trading system open. A successful APEC and ASEM would remain consistent with the WTO principles of multilaterism. Moreover, it is an avowedly normative neoliberal project. The advocacy of global liberalization is not simply an international economic process it also has serious domestic political implications. To be successful it will not be sufficient for just the regional transnational policy community in East Asia to be committed (more or less) to liberalization. While it is certainly a prerequisite, their position will need legitimating within the domestic political communities of their own civil societies.

The states of East Asia may be occupying a more central position in late twentieth century international relations and its policy elites may be generating greater regional senses of awareness and identity in the development of institutions such as APEC and ASEM, but these policy-making elites need to address—in the short to medium term—the question of how to generate wide domestic political interest in and support for their multilateral diplomatic agendas. The 'official' status of these organizations needs to become embedded within the emerging civil societies of the member states. To-date, these nascent institutions have paid nothing other than lip service to this question. While there may be a substantial second track involvement of the corporate community (via the business foundations) and the research communities (operating primarily out of the think

tanks and to a lesser extent the universities) involved in the activities of APEC and ASEM, this involvement does not extend down into the further reaches of society. A common feature of regional summits nowadays are the parallel conferences convened by NGOs to coincide with the summits. For APEC and ASEM, and indeed similar transnational bodies, to develop legitimacy they have to engage with what Camroux and Lechervy (1996, p.448) call 'non-official Asia'. Without this rapprochement between the rulers and the ruled in East Asia the role of APEC/ASEM style organizations in what Noordin Sopiee (1995, p.191) sees as the 'empowerment of Asia...on the world stage' will always be constrained.

#### **13.6** Conclusion

The aim of this chapter has been to highlight the way in which the processes of regionalization and intra-regionalism are developing in the Asia-Pacific in the closing years of the twentieth century. It has argued that there are two identifiable, but inseparable processes in train at the regional level. These have been called *de facto integration* and *de jure co-operation*. The idea of region has also been shown to exist at multiple levels. Implicit in the chapter is a recognition of the emergence of East Asia as a significant regional and global actor with a role to play in the agenda of global economic management. In this context international empowerment for East Asian states and regional organizations is both necessary and desirable in the management of the modern world order.

As economic growth continues, East Asian states both individually and collectively need to play a role in international relations commensurate with their economic size. To date, East Asian states are not as active as their North American and European counterparts as international citizens in world affairs. This asymmetry of behaviour may have been understandable, and indeed acceptable, in the immediate decades after the Second World War. Then, the states of the region were emerging from a colonial past and/ or consolidating their position as international actors in the face of Cold War contests and an evolving international system. This is no longer the case. Events in the region, discussed in this chapter, have demonstrated a growing awareness of this in a number of ways.

One consequence of the growing symmetry in East Asia's relationship with Europe and North America is that a greater equality of rights and opportunities in international relations must perforce call forth a greater set of obligations and responsibilities. APEC has been an important vehicle for social learning about regional and multilateral diplomacy in the Asia-Pacific over the last 6–7 years. ASEM could be a similarly important route to social learning and citizenship not only in the issue specific context of Asian-European relations, but also in those other areas of global economic and political management that touch upon the common interests of the East Asian members of ASEM. Both organizations, young as they may be, represent a recognition of the need for the states of East Asia to look beyond the Pacific.

#### References

- Barnard, M. (1996) 'Regions in the global political economy: beyond the global local divide in the formation of the East Asian Region', *New Political Economy*, vol. 1, no. 3, pp. 355–74.
- Barnard, M. and Ravenhill, J. (1995) 'Beyond product cycles and flying geese: regionalization, hierarchy and industrialization in East Asia', *World Politics*, vol. 47, no. 2, pp. 171–209.
- Bhagwati, J. (1996) 'The US-Japan car dispute', International Affairs, vol. 72, no. 2, pp. 261-79.
- Bhagwati, J. and Patrick, H. (eds) (1990) Aggressive Unilateralism: America's 301 Trade Policy
- and the World Trading System, Ann Arbor, Michigan University Press.
- Brook, C. (1998) 'Regionalism and globalism' in McGrew, A. and Brook, C. (eds). Buzan, B. (1998) 'The Asia-Pacific: what sort of region in what sort of world?' in McGrew, A.
  - and Brook, C. (eds).
- Camroux, D. and Lechervy, C. (1996) 'Encounter of a third kind: the inaugural Asia-Euro meeting of March 1996', *The Pacific Review*, vol. 9, no. 3, pp. 441–52.
- Commission of the European Communities (1994) *Towards a New Asia Strategy*, Brussels, Commission of the European Communities.
- Encarnation, D. (1994) *The Regional Evolution of Japanese Multinationals in East Asia: A Comparative Study*, Cambridge MA, Massachusetts Institute of Technology.
- Gallant, N. and Stubbs, R. (1996) 'APEC's dilemmas: institution building around the Pacific rim', *Pacific Affairs*, vol. 70, no. 2, pp. 203–18.
- Harris, S. (1996) The WTO and APEC: What Role for China?, San Diego, IISS Conference.
- Hatch, W. and Yamamura, K. (1996) *Asia in Japan's Embrace: Building a Regional Production Alliance*, Cambridge, Cambridge University Press.
- Higgott, R.A. (1994a) 'Ideas, identity and the evolution of policy co-ordination in the Asia-Pacific', *The Pacific Review*, vol. 7, no. 4, pp. 367–79.
- Higgott, R.A. (1994b) 'APEC: a sceptical view' in Mack, A. and Ravenhill, J. (eds).
- Higgott, R.A. (1995) 'Economic co-operation in the Asia-Pacific: a theoretical comparison with the European Union', *The European Journal of Public Policy*, vol. 2, no. 3, pp. 361–83
- Higgott, R.A. (1996) 'Beyond embedded liberalism: governing the international trade regime in an era of economic nationalism' in Gummett, P. (ed.) *Nationalism and Internationalism in Public Policy*, Aldershot, Edward Elgar.
- Huntingdon, S.E. (1996) *The Clash of Civilizations and the Remaking of World Order*, New York, Simon and Schuster.
- Kahler, M. (1994) 'Institution building in the Pacific' in Mack, A. and Ravenhill, J. (eds).
- Mack, A. and Ravenhill, J. (eds) (1994) Building Economic and Security Regimes in the Asia-Pacific, Sydney, Allen and Unwin.
- McGrew, A. and Brook, C. (1998) *The Asia-Pacific in the New World Order*, London, Routledge in association with The Open University.
- Mahbubani, K. (1994) 'The Pacific way', Foreign Affairs, vol. 74, no. 1, pp. 100-11.
- Nuttall, S. (1996) *Report of the Council for Asia Europe Co-operation Task Force: Strengthening the Informal Dialogue*, Paris, First Plenary Meeting of CAEC, November, mimeo, pp. 1–12.
- Ohmae, K. (1990) The Borderless World, London, Collins.
- Ohmae, K. (1995) *The End of the Nation State and the Rise of Regional Economies*, London, HarperCollins.
- Oman, C. (1995) The Policy Challenges of Globalization and Regionalization, Paris, OECD Development Centre, Policy Brief.
- Pina, J. (1966) 'Europe Asia economic summit of historic importance', Inter Press Service English News Wire, 24 February.

- Segal, G. (1997) 'Thinking strategically about ASEM: the subsidiarity question', *The Pacific Review*, vol. 10, no. 1, pp. 124–34.
- Smith, M. (1998) "The EU and the Asia-Pacific' in McGrew, A. and Brook, C. (eds).
- Sopiee, N. (1995) 'The development of an East Asian consciousness' in Sheriden, G. (ed.) *Living with Dragons: Australia Confronts its Asian Destiny*, Sydney, Allen and Unwin.
- Sopiee, N. (1996) 'The start of a world without the US', Yomiuri Shimbun, 14 April.
- Wallace, H. and Wallace, W. (eds) (1996) Policy Making in the European Union, Oxford, Oxford University Press.
- WTO (1995) *Regionalism and the World Trading System*, Geneva, World Trade Organization Secretariat.
- Yoshida, M., Akimune, I., Nohara, M. and Sato, K. (1994) 'Regional economic integration in East Asia: special features and policy implications' in Cable, V. and Henderson, D. (eds) *Trade Blocs? The Future of Regional Integration*, London, Royal Institute of International Affairs.

### CHAPTER 14 Danger and opportunity in the Asia-Pacific

#### Jeffrey Henderson

#### **14.1 Introduction**

In Chinese the term for 'crisis' combines two symbols: the first, *wei*, meaning 'danger' and the second, *ji*, opportunity.<sup>1</sup> This dialectical conception of crisis, coming as it does from Asia's most important indigenous language, helps us to grasp the nature of the current circumstances of the Asia-Pacific region. Now, and into the early decades of the next century, the Asia-Pacific region and the world economy and inter-state system more generally are indeed in crisis in the Chinese sense. Economic development has been profoundly uneven in the region, compounding pre-existing tendencies towards sociopolitical fragmentation. It has also brought with it untold ecological damage, of which travellers to even one of the region's most advanced territories—Hong Kong—become immediately aware. Additionally, and attempts to develop regional institutions for economic governance notwithstanding (Chapters 7 and 13), the processes now afoot could well lead not only to cut-throat economic competition in the region, but to geopolitical conflict as well. In these senses and more, then, the immediate future of the Asia-Pacific region is one fraught with danger.

But, at one and the same time, this is a period of opportunity. Already, in little more than a generation, hundreds of millions of people have been lifted out of abject poverty, and many of these are now well on their way to enjoying the sort of prosperity that has been known in North America and Western Europe for some time (Chapter 1). Similar processes to those that have underlain such achievements continue to transform the lives of hundreds of millions more in the region. While some groups have been obliged to bear the negative consequences of development far more than others (see Chapters 5 and 11), it seems clear that the development project

<sup>1</sup> I am grateful to Suet Ying Ho for alerting me to this conception of crisis in the Chinese language. In addition I wish to thank Ray Bush and his colleagues in the Department of Politics, University of Leeds, for inviting me to join them in spring 1997. It was during my period there that this chapter was written.

in the Asia-Pacific region holds out the promise of a scale of generalized prosperity unknown in human history. With rising prosperity will come not only improved educational provision, literacy, enlightenment and more, but *possibly* the rise of social egalitarianism, the evolution of democratic state forms, and a measure of geo-political harmony.

It is this interface between danger and opportunity that sets the parameters for development in the region into the early decades of the next century. Partly by recuperating the arguments advanced in earlier chapters, but partly by extrapolating from them and supplementing them in important ways, this concluding chapter seeks to explore the economic possibilities inherent in the 'Pacific Century', identify their limitations and evaluate some of the implications of development in the region for other parts of the world. In so doing, this chapter will help to link discussions presented in this book to those developed in other books in the series (McGrew and Brook, 1998; Maidment and Mackerras, 1998; Maidment *et al.*, 1998).

The chapter begins by assessing the significance of the Asia-Pacific 'miracle' as an intellectual construct. It then turns to a discussion of the major forms of capitalism that compose the region's economies and to some of the processes of globalization that connect them to the world economy more generally. It then assesses the prospects for economic development in the region and the broader international implications, before concluding with a brief comment on the challenge which the Asia-Pacific represents.

#### 14.2 Intellectual significance of the Asia-Pacific region

A consideration of the significance of the East Asian 'models' of economic progress for growth and development elsewhere in the world takes us straight to the heart of the central issue of how the economic 'miracle' there came about. It does so because it throws into high relief seven determinants of economic development and performance that have been popularized, in various combinations, in academic and policy debate:

- the historical contexts and conjunctures out of which the transformative processes in the Asia-Pacific region emerged;
- the role of foreign aid (military and civil, and mainly from the USA) in the context of the Cold War;
- the subsequent benefits of foreign direct investment in terms of transfers of technology, skills and information;
- the significance of a regionally unique 'economic culture' rooted in Confucianism;
- the existence of repressive labour systems that have ensured supplies of cheap labour (cheap, that is, in previous decades in South Korea, Taiwan, Singapore, etc., but still centrally important in China, Vietnam, etc. today);
- the importance of free markets;
- the role of state policy and influence.

As other volumes in this series, and other chapters in this volume, deal extensively with many of these—and in any case I have interrogated their significance elsewhere (Henderson and Appelbaum, 1992)—I do not propose to traverse that same ground here. Rather, the focus in this section is on the last two determinants—markets and states—for it is these that form the major intellectual legacy of the Asia-Pacific region for countries in the throes of economic transformation, from South Africa to Russia and from Britain to Brazil.

#### Free markets

Transformation and rapid economic growth in the Asia-Pacific region, already evident by the 1960s, had come as a surprise to many development analysts schooled in the counterposed 'modernization' and 'dependency' traditions. Modernization theory, assuming that cultural and institutional forms akin to Anglo-American norms were particularly propitious for development in the Third World, infamously had questioned (following Max Weber) the capacity of Confucianism to support economic growth in East Asia. Dependency theory, on the other hand—particularly the form associated with Andre Gunder Frank—had implicitly denied the possibility of development in East Asia because of its imbrication (along with Latin America, Africa, etc.) in the neo-imperialist structure of the world economy and inter-state system. The destiny of these countries, it was agreed, was one of continued underdevelopment as the surpluses they produced were siphoned off from the 'periphery' into the 'centre' of the world economy.

However, economic growth and development in the Asia-Pacific region came as less of a surprise to economists influenced by the neoclassical paradigm. By the mid 1950s many orthodox economists had become convinced that blockages to economic growth and development in what had become to be known as the Third World' were largely self imposed. They were seen as the products of distortions in markets for labour, capital, natural resources, technology, etc., that arose from endogenous cultural forms and social practices, but particularly from the interference (and often rent seeking) of the national state. The reason why first Japan, and then a number of the other countries of the Asia-Pacific region, had been able to move decisively in the direction of dynamic, high productivity, high wage economies was because they alone had been relatively free, supposedly, of distortions in capital, labour and the other markets. In particular, and in contradistinction to most developing and many developed economies (in the latter case such as France or Britain at that time), their respective states, it was argued, had either not intervened to distort markets or, if they had intervened, had done so to correct preexisting distortions.

In the more sophisticated versions of the neoclassical account, what made Japan and the newly industrialized economies (NIEs) of the region 'different' was the fact that they had enjoyed 'market-enhancing' political regimes that had allowed the efficiencies derived from market-based resource allocations to be maximized. Thus, after initial periods of constraint (in the 1950s and early 1960s) their economies had been liberalized relatively quickly, export-oriented strategies had been pursued, and, as a consequence of these, domestic prices for capital, labour, etc. had soon become a reflection of international conditions. Releasing market forces in these ways supposedly formed the bedrock of the double-digit growth rates that were to come.

The experience of economic transformation and development in the Asia-Pacific region, then, seemed to vindicate the premises of neoclassical economic theory generally. From the 1970s through to the present day, Japan and the East Asian NIEs came to be seen—together with Thatcherite Britain, and to a lesser extent Chile (subsequent to the military coup of 1973)—as the world's most significant empirical supports for the truth claims of neoclassical theorizing. In such circumstances it was not long before this particular version of economic transformation in East Asia became enshrined as the

principal ideological support for the 'structural adjustment' policies of the International Monetary Fund (IMF), the World Bank and similar international economic agencies, as they attempted to influence the state-economy relation in the developing world, and more recently in the former state-socialist societies of Eastern Europe. By 'structural adjustment' the IMF and the World Bank refer to the belief that the reform of economic and political institutions, together with the liberalization of macroeconomic policy, are the preconditions for market-driven economic growth (the only form of growth they conceive as possible). Such reforms and policy changes have increasingly become preconditions for IMF and World Bank loans. In many developing countries and in Eastern Europe these have involved, *inter alia*, the abandonment of market protection and import-substituting industrialization (ISI) strategies, wholesale privatization of state assets and dramatic reductions in government expenditures on welfare, etc., with their inevitable human consequences.

The first attempt to use East Asian experience to legitimate neoclassical propositions came in the 1960s, in a series of OECD-sponsored studies of developing economies. Taiwan was the East Asian 'representative' in this research programme, and in the book summarizing the results of the work (Little *et al.*, 1970), it was used as a benchmark of what could be achieved if autarkic development strategies associated with import-substitution industrialization (the norm in the developing world at that time) were abandoned in favour of liberalized, relatively open domestic economies whose growth was predicated on a full engagement in international trade via export-oriented strategies.

In addition to projecting the Taiwanese experience as a model for the rest of the developing world, interestingly this work set the tone as to how that experience (and those of the East Asian NIEs generally) came to be explained in the more sophisticated neoclassical pronouncements from the 1970s onwards. While insisting that Taiwan's export success was the result of a multiplicity of factors, Little et al. acknowledge that 'among these the economic policies followed by the Government were certainly important' (1970, p. 256). Unfortunately, the only Government policies they discuss are the ones directed towards the liberalization of the domestic economy, the enhancement of markets and export promotion. Prescriptive economic planning, interventionist policies such as the selective protection of domestic markets, extensive state ownership of upstream industries and banks-all evident in Taiwan (see Wade, 1990)-are buried under the words 'various other measures were adopted as well' (Little et al., 1970, p. 255). Twenty years later, it seemed that the position of orthodox economics on East Asia had barely progressed. In spite of volumes of non-neoclassical scholarship to the contrary, the eminent Johns Hopkins University/World Bank economist, Bela Balassa, still felt able to comment that 'It would appear...that the success of the Taiwanese economy cannot be attributed to planning' (Balassa, 1991, p. 103).

Drawing on the work of Little, Balassa and others, the influence of neoclassical theorizing as the principal underpinning for the policy prescriptions of the major international economic agencies proceeded apace. Particularly important in this sense were developments in the World Bank, which by 1981, with the publication of the 'Berg Report' on Sub-Saharan Africa (World Bank, 1981), had clearly become 'captured' by the market-enhancing/minimalist state conception of the route to efficient, dynamic economies. Two years later, the Bank's *World Development Report* (World Bank, 1983) embodied what had become its prevailing wisdom: namely that 'the *universal* application

of neoclassical economic theory holds the key to economic efficiency' (former World Bank official quoted by Hayter and Watson, 1985, p. 140—my emphasis).

While there were other important intellectual monuments along the World Bank's 'road to Damascus' that need not detain us here, that same year (1983), one of the more ideologically committed of the neoclassical economists, Deepak Lal, underlined the significance of his view of the East Asian experience. In the context of an attack on the very idea that developing economies require forms of theorization distinct from other economies (and hence their own specialization, 'development economies'), he contrasted the structure and performance of the South Korean and Taiwanese economies with that of India. While the Indian Government since independence (1947) had pursued an autarkic development strategy based on ISI and domestic market protection, which he considered responsible for India's economic stagnation, South Korea had flourished, he suggested, as a consequence of Government policies that had ensured 'virtually free markets' (Lal, 1983, pp. 87-90). Unfortunately for Lal, 'virtually free' hid such a vast armoury of market constraining policies relating to technology access, corporate finance, speculative investment, foreign investment and many more (see among others, Chapter 3 in this book) that Nigel Harris felt able to comment that South Korean economic success has been 'just as much a triumph of state capitalism as [were] the achievements of the first Five Year Plans in the Soviet Union or the People's Republic of China' (Harris, 1987, p. 145).

Although Lal's was possibly an extreme position (as was Harris's from a different point on the ideological spectrum), it was symptomatic of the way the East Asian experience, reconstructed to fit neoclassical preconceptions, had become an ideological battering ram to wield against the defences of governments around the world who thought that a modicum of economic planning, and (sometimes) a commitment to redistribution and a welfare state, were the best ways of pursuing growth (with equity). In the case of the World Bank and the IMF, 'structural adjustment' policies based on the supposedly universal principles of neoclassical reasoning have continued to be foisted on the developing world, including, more recently, the former state-socialist societies of Eastern Europe. This has continued in spite of the fact that, as a consequence of Japanese pressure, the World Bank has been forced to reconsider the role of the state in economic growth and development.

By the late 1980s, the Japanese Government had become the second largest contributor (after the USA) to World Bank funds, and was becoming increasingly irritated that its own development experience—so much associated with the state's 'administrative guidance'—was effectively being denied in the policy advice (backed by leverage) which the Bank was offering to governments of the developing world. The consequence of Japanese pressure was the Bank's reassessment of the East Asian route to industrial capitalism (the study was also financed by the Japanese Government) which culminated in its report, *The East Asian Miracle* (World Bank, 1993). As Wade (1996) shows, the report is a contradictory mishmash negotiated out of the political need of the Bank to accommodate the Japanese Government's legitimate interests, on the one hand, and its firm conviction that it should hold to the neoclassical 'structural adjustment' line on the other. Even so, the publication of *The East Asian Miracle* was a source of surprise, if not excitement, in academic and policy circles, though as yet there is little evidence that it has had much effect on World Bank—or IMF—advice and funding decisions.

Before moving on, it is worth noting that whatever the credibility of the various orthodox accounts of economic growth and performance in East Asia, the spectacular rise of China as an economic powerhouse seems to have posed certain explanatory difficulties for those economists and international agencies who continue to be wedded to the neoclassical paradigm. Indeed, with the exception of a few contributions which are staggering in their ideologically audacious 'economy with the truth' (see, for instance, Cheung, 1986, pp. 65–79) and a number of more measured interventions (for example, Sachs and Woo, 1994), neoclassical commentary on the recent development of the Chinese economy has been somewhat muted. This is perhaps not surprising given that in the lexicon of neoclassical theory it is supposed to be impossible for an economy to grow at the double-digit rates of the Chinese economy when the state still controls its 'commanding heights', engages in (an albeit modified form of) central planning, where markets are only partly responsible for resource allocation, where there remains considerable domestic market protection and where there is still no effective development of property rights and the rule of law (see Chapter 3). In short, the state—in both national and local guises—is absolutely central to recent economic progress in China, and it is impossible to conceive of a plausible and empirically adequate explanation of Chinese development without recognizing this fact.

#### States

If the perceived role of free markets in East Asian development has been appropriated by academic, political and policy interests for their own ends, then alternative explanations of economic expansion in the region have been used by others as a basis for policies to improve prospects in the developing world and to assist the transformation of the former state-socialist societies of Eastern Europe. Various conceptions of the role of the state in East Asian development and transformation were reviewed in Chapters 3 and 6, and there is no need to repeat these debates here, but it is important to note that whatever position is adopted the assertion of the *necessity* of a proactive state for development purposes in the region is *at least* of enormous symbolic significance when viewed from the perspective of other countries and regions of the world seeking pathways to prosperity.

As I have indicated earlier, many parts of the developing world, and more recently the former state-socialist societies of Europe also, have been under considerable economic and political pressure to liberalize and deregulate their economies, reduce welfare expenditure, etc.—in effect, to allow the state to 'wither away' as a subject of economic management—while countries such as Britain, Australia or the USA have for years been under the sway of political regimes for whom 'free markets' have become deities. It is in such places that the theory and practice of the economically proactive state, informed by East Asian experiences, comes as an intellectual breath of fresh air.

In many cases, however, analysts have gone much further than mere assertion of an economic role for the state and have attempted to specify the circumstances under which particular institutional frameworks and policy initiatives drawn from Asia-Pacific experiences may be worthy of attention in other contexts. I briefly review some of these arguments below.

In spite of a rising tide of 'state friendly' analyses of East Asian development from 1977 onwards (see Henderson, 1993, for a review), Robert Wade, in his now classic

book, *Governing the Market* (Wade, 1990), was the first to begin to think through the implications of his version of this perspective for developing countries in other parts of the world. While this is not the place to repeat his observations, it is worth noting that in addition to calling, for instance, for a central role for manufacturing industry in state policy thinking, and its need to be privileged in an investment structure which ensures tight regulation of finance capital, Wade considers a centralized authoritarian state to have the best prospects of developing the institutions necessary for 'governing the market' in the initial stages of industrialization. For him these institutions need to be in place before political democratization ensues, otherwise the tensions and conflicts typical of democracies are likely to engulf the economic bureaucracy, destroying the autonomy which Wade sees as necessary to effective economic management in the context of rapid industrialization.

Other contributions have sought to use various 'state friendly' analyses of Asia-Pacific development to argue for a reorientation of transformation strategies in Eastern Europe. These have generally been of two sorts. The first has extrapolated from the assumption that the principal points of commonality between the two regions have been in the institutional legacies of state socialism. Consequently, it has been argued that the most appropriate comparisons are to be made between Eastern Europe (particularly the territories of the former Soviet Union) and China (and to a lesser extent Vietnam). Bolstered by arguments about the decisive role of central and local state institutions in recent Chinese industrialization (see, for instance, Qian and Weingast, 1997; Nee and Su, 1998), such contributions generally have emphasized that the state orchestrated form of industrialization pursued by China (itself influenced by Japanese, South Korean and Taiwanese experience) has produced far superior economic performance, with greater generalized prosperity, than the more market-centred forms typical of Russia and elsewhere in Eastern Europe (see, for instance, Nolan, 1995; Lo and Radice, 1998).

Other contributions, however, have insisted that aspects of the experiences of capitalist East Asia may well be relevant to Eastern European circumstances. Judith Thornton (1995), for instance, has recommended that Eastern European countries adopt some of the institutional arrangements and management techniques of the Japanese economic bureaucracy, though without using a distorted version of its practices as an excuse to avoid dismantling state enterprises and heavy-handed state direction of industry. Vartiainen (1995), on the other hand, has argued that the Eastern European countries have much that is positive to learn from the experience of state-guided economic development, not only of South Korea and Taiwan, but also of Austria and Finland.

The most developed arguments in this vein, however, have come from Alice Amsden and her colleagues (1994) and some of the contributors to the volume edited by Henderson (1998). In the former case, Amsden *et al.* argue that 'all successful industrialization experiences have rested on state intervention' (1994, p. 49), and use the experiences of South Korea and Taiwan in particular to argue that the former statesocialist economies of Eastern Europe will require state co-ordination in order to be transformed into effective, high performing capitalist ones. In the latter case, a number of the contributors argue that while East Asian experiences of industrialization and transformation do not constitute panaceas for Eastern Europe, they contain pointers—for instance in the state-business relation, the nature of the regulatory regimes and the institutional capacity of the economic bureaucracy—as to what types of initiative might be useful for building economic dynamism in the region.

While I have outlined some of the ways in which differing conceptions of economic growth and transformation in the Asia-Pacific have impacted on intellectual and policy debate elsewhere in the world, three further points need emphasis. First, the conception which locates free markets and market-friendly government policies as the keys to growth in the region continues to hold sway, in more and less sophisticated versions, in orthodox economic circles. Indeed, the US doyen of the political economy of Japan, Chalmers Johnson, has argued that junior faculty and graduate students in major economics departments in the USA are dissuaded by senior colleagues from studying the empirics of the Japanese or other economies in case their investigations should lead them to question the received wisdom of orthodox theory, and so damage their careers (Johnson, 1988). Second, given the fact that this version continues to have IMF/World Bank support, its ideological power for the rest of the world continues to be very significant. Third, though developed predominantly by political economists, sociologists and economic geographers, conceptions which see East Asian development as having been a product of symbiotic relations between states and markets (albeit emerging from different histories and circumscribed by different institutional contexts) now carry at least the 'balance of presumption' (Wade, 1990, p. 342) given the methodological difficulties of adjudicating between competing theories on the basis of 'the facts'.

Whatever the relation of the state to economic performance in the past, it seems likely that this relation is now changing at least in the NIEs. However, evidence is hard to come by, and what there is suggests, for instance, that even with the onset of democratization, the South Korean Government at any rate still retains substantial leverage over the *chaebol*. This is not an issue, however, that needs to be pursued further at this point. Rather we need to turn to other, more general questions which have influenced the economic past and will continue to influence the future of the region.

#### 14.3 Globalization and forms of capitalism

Raising the issue of the relation of states and markets to economic change in the Asia-Pacific region leads directly to a number of other issues. These include: the extent to which the economies of the Asia-Pacific region embody particular forms of capitalism which are distinct from those evident in Europe and elsewhere, and which have differing implications for economic performance; the extent to which these distinctions are being eroded as a consequence of the integration of the region's economies into the wider world economy; and the nature of the competitive challenge posed by the region's economies and their constituent business organizations. In what follows in this section I assess the nature and significance of these issues, both in their own right and as a prelude to the penultimate section. Here, and in that section, I return to the dialectic of danger and opportunity—the dimensions of crisis—with which I began this chapter.

#### Competing capitalisms

Competition in the world economy involves not merely the drive of firms for greater market share, or the struggle of workers across the globe for jobs and better standards of living. Competition today also involves a clash of different—in some cases radically different—versions of capitalism. With the advantage of hindsight, and over a hundred years of mature (internationally competitive) industrial capitalism behind us, we now know that the common core of capitalist economic systems is more restricted than earlier thought. Indeed, for the eminent social theorist, Gary Runciman, it amounts to no more than 'a mode of production in which formally free labour is recruited for regular employment by ongoing enterprises competing in the market for profit' (1995, p. 33). Around that core has been constructed a variety of institutionally variant forms of capitalism which incorporate sometimes vastly different forms of corporate and economic governance, different investment priorities and relations between finance and industrial capital, different capacities and orientations to state economic planning, different welfare regimes, etc.

Even within the Asia-Pacific itself, as Richard Whitley (Chapter 9) has shown, the institutional bases for different national capitalisms vary dramatically, as does the nature of business organization and managerial priorities and styles, as say between the large 'collectivist' Japanese corporation which seeks to incorporate many employees into decision-making structures, and the autocratic, family-controlled companies typical of Taiwan, Hong Kong and Southern China, for whom employees are mere labour power, to be used, exploited, and discarded as necessary. In order to develop some sense of the significance of these various forms of capitalism in the region, and of their international implications, we begin by advancing a provisional typology of these and other significant forms elsewhere in the world economy.

Form	Representative economy
Shareholder capitalism	Britain
Managerial capitalism	USA
Co-ordinated managerial capitalism	Germany
Collective capitalism	Japan
State-directed capitalism	South Korea
Co-ordinated proprietorial capitalism	Taiwan
Market Stalinism	China

# Table 14.1 Major forms of capitalism in the latetwentieth century

While this is not the place to unpack these categories in any detail, a few brief comments need to be made in order to be able to comprehend their methodological bases and to distinguish Asia-Pacific regional forms from each other and from those of other regions.

Two general points are needed in order to explain the distinctions made in Table 14.1. First, the terms used to designate the various forms are meant to emphasize the primary sources of institutional power responsible for economic and corporate governance. Thus, in the case of 'shareholder capitalism' (Britain), economic sovereignty rests almost exclusively in property owners who are disengaged from the firms they own and trade their assets regularly on the stock market in the interests of maximizing short-term profits from their investments. Firms raise their funds largely through stock market flotations, are managed in the interests of short-term profitability rather than market growth and adding value, and the economic structure evolves in an *ad hoc* fashion, with little (or at best irregular) input from state policy.

The notion of 'managerial capitalism' (USA), on the other hand, highlights the significance of managerial co-ordination in the context-unlike Britain-of firms that have been very large vertically and horizontally integrated organizations. While corporate finance is again largely capital-market generated, regulatory mechanisms have traditionally ensured that shareholders have been more engaged (committed, loyal) to the companies concerned than their British counterparts. In the case of 'co-ordinated managerial capitalism' (Germany), professionalized manage-ment in the context of large, integrated companies has also been significant, but with three major differences. Networks of small and medium-sized firms (mittelstand) subcontracting for the large firms have been central to the form of capitalism that has evolved since 1945; corporate finance is largely through bank credit in arrangements whereby the banks become committed to the long-term growth (rather than short-term profitability) of the companies concerned; and the state-central and local-plays a more proactive role in economic coordination and in encouraging socially integrative institutions which influence workermanagement relations, and society at large. Germany and other continental European economies, together with Japan, represent what might additionally be termed 'stakeholder capitalism'.

Japan, though often seen as the prototypical form of East Asian capitalism, is, as Whitley shows (Chapter 9), very different from the others (but note the role of the state— Chapter 6 and below). In its case, the notion of 'collective capitalism' addresses the significance of social and institutional integration to economic and corporate governance. Employee integration and systematic consultation is achieved within the large corporation (but hardly exists in the important small and medium-sized enterprise sector), in ways that Whitley has explained, while inter-firm integration is achieved across the *keiretsu* network. The 'collective' nature of Japanese capitalism is enhanced by the committed involvement of banks via a system where corporate finance is raised largely through bank loans rather than the stock market, and importantly by a proactive state which has sought to orchestrate the structure of the economy and the trajectory of development.

Drawing heavily on Japanese practices prior to the Second World War, 'state-directed capitalism' has emerged in South Korea since the early 1960s. Central to this form has been a partnership between an authoritarian—and until the late 1980s, militarized—state, and big business (the *chaebol*), wherein the state, for much of the period, has been the senior partner. Once again, corporate finance has been largely provided by bank loans, but under circumstances where the state has tightly controlled access (Chapter 4).

Additionally, the state has been heavily involved in technology acquisition, market protection and economic planning more generally.

Similarly influenced by Japanese practices, but also by the economic planning techniques of the Nationalist Chinese Government in the 1930s, Taiwan has also developed a form of capitalism since the late 1940s where economic co-ordination is in part a state function. Depicted as 'Leninist' by Taiwan's most eminent sociologist, H.H. Michael Hsiao (1995), the Taiwanese Government has engaged in many of the 'market-leading' activities typical of the Japanese economy and NIEs of East Asia (see Chapters 3 and 6), but additionally it has encouraged a substantial state (or ruling political party—the Kuomintang) owned industrial sector in upstream processing industries, banking, aerospace, etc. Unlike Japan and South Korea, but like Hong Kong, Taiwanese capitalism is also characterized by a large, small and medium-sized enterprise sector (of which the largest and best known companies include Tatung, Acer, Mitac and Evergreen) which is responsible for export performance. Such companies are generally family owned (hence 'proprietorial'), organizationally bounded by kinship networks (Chapter 9) and embedded in complex subcontracting networks nationally and internationally (in the latter case particularly with US and Chinese firms).

The term 'Market Stalinism' refers to an emergent form of capitalism of which the People's Republic of China is the principal representative. As a set of institutional arrangements that are (presumably) transitional between state socialism and capitalism in the East Asian context, the defining characteristic is the continuing decisive economic role of an authoritarian, repressive, 'Stalinist' state in a context where resource allocation has been partially marketized, and the economy partially connected to the world economy, primarily as a result of insertion into overseas Chinese business networks.

The second general point concerns the historical limits to these various forms of capitalism. The institutions and social arrangements that constitute these capitalisms have in some cases evolved over considerable periods of time (particularly in Britain, but also in the USA), but often have been the product of deliberate 'institutional engineering' on the part of respective states subsequent to historical ruptures such as defeat in war (Germany and Japan), military coups or acquisitions (South Korea and Taiwan), or changes in political élites (China). Whatever the historical sources of these capitalisms, their characteristic institutional arrangements have supported improvements in economic performance and prosperity for varying periods of time (though some better than others: for example, USA *vis-à-vis* Britain).

As with all products of human consciousness and action, however, there are inevitably limits to the effectivity of these arrangements. In the case of Britain, it has been argued that the long-term decline of the economy can be traced to the inappropriateness of the country's institutions subsequent to the intensification of international competition from the late nineteenth century onwards. As for the USA, while 'managerial capitalism' has been responsible for the highest level of productivity, and one of the best records of generalized prosperity (together with Germany, Scandinavia, Switzerland, etc.) the world has known, the relative decline in its international competitiveness in recent decades has been traced to the nature of its economic institutions. Similarly, the Japanese recession in recent years has been traced in part to the rigidities of the *keiretsu* and 'permanent employment' systems (of which more later), which some regard as being inappropriate in

a changed competitive environment where all high cost producers, including Japan, are under increased pressure from low cost producers elsewhere in Asia.

Whatever the nature and significance of the historical limits of the different forms of capitalism, changes are clearly under way, including in the Asia-Pacific region. The sources of change are in some cases endogenous, as in the case of China where the nature of uneven development is producing serious social dislocation (see below), but in all cases are partly exogenous and associated with various dimensions of globalization. It is to that issue and its implications for the Asia-Pacific region that I now turn.

#### Economic fusion

This is not the place to present an overview and critique of the various meanings of globalization. That has been done effectively enough by Paul Hirst and Grahame Thompson (1996) and has been the focus of an extraordinary empirical and analytic synthesis by Manuel Castells (1996, 1997a,b). Additionally, one of the key vehicles of economic integration in the region on the one hand, and the internationalization of its productive activities on the other-multinational corporations (MNCs) in manufacturing and financial services-have been analysed elsewhere in this book (Chapters 2, 8 and 10). What I propose to do here is to pick up on Castells' (1996) insistence that the 'glue', if not the defining characteristic, of globalization is increasingly the multifaceted nature of social, economic and political networks and the dynamics and implications of the processes of networking. It is clear that internationally organized business networks are decisive for the economic present and future of the Asia-Pacific region. To anticipate my argument, however, these networks are not necessarily entirely positive in their implications for that region, or indeed for elsewhere in the world. As we shall see, their logic and dynamics are responsible for compounding the processes of uneven development—and hence the economic basis for fragmentation, and possibly conflict—in the Asia-Pacific region, as they are elsewhere.

A useful conceptual device which helps us gain analytic purchase on most, if not all, national and international business networks (including those established by MNCs) is the notion of 'global commodity chains' which has been developed most extensively by the US sociologist, Gary Gereffi (see, for instance, Gereffi and Korzeniewicz, 1994). In his work, commodity chains are conceived as networks of business units involved in producer, supplier, subcontracting, exporting, purchasing, distribution and marketing activities. The business units may be subsidiaries of MNCs or independent companies of varying sizes. In the former case the commodity chain which links and organizes the various business units is conceived as being 'driven' by the MNC or 'producer' company, whereas in the latter the dominant corporate power 'driving' the chain is the 'buyer' responsible for marketing the product, usually under its own brand name.

Commodity chains link different sets of raw materials, labour, skills, technology and managerial inputs across geographic space. In the case of buyer-driven chains they link different types of firms embedded in different social and institutional contexts: that is, different forms of capitalism (though Gereffi himself has yet to engage with this point, but see Gereffi, 1996). While the framework devalues the way state action can influence economic outcomes at given locations in the chain, and while in general it tends towards an over emphasis on purely economic links, it does have the advantage of emphasizing

the fact that the value-adding capacities of companies at each nodal point in the chain tend to vary and in any case are constrained by the way the chain is organized and by the nature of corporate power within it (see Figures 14.1a and 14.1b).



# Figure 14.1a Global commodity chains: producer driven

The consequence of this is that the ability of MNC subsidiaries, or independent firms at particular nodes in buyer-driven chains, to contribute to the development and prosperity of economies of which they are a part is profoundly asymmetrical (Henderson, 1994). In the case of the production



flows of capital, expertise, designs, materials etc. where relevant

Figure 14. 1b Global commodity chains: buyer driven
and marketing of manufactured commodities organized by buyer-driven chains (such as garments, footwear, personal computers, some consumer electronics and metal products such as bicycles), the bulk of the value is added—and captured—not in the manufacturing stages (now performed increasingly in the developing world, including in East Asia), but at the stages of 'branding' (the company owning the brand name) and marketing, activities controlled overwhelmingly by firms from the advanced industrial economies of the USA, Japan and the European Union.

Many have argued that economic dynamism in the Asia-Pacific has been based on the national and international activities of domestically-owned firms, and overwhelmingly in manufacturing and related services. It has been estimated, for instance, that while foreign-owned firms are responsible for only about one per cent of gross domestic product (GDP) in Japan, this rises to between 12 and 15 per cent in Hong Kong and South Korea, and some 25 per cent in Taiwan (see Chapter 8). While Japanese companies led the waves of internationalization (to the USA, Europe and elsewhere in Asia) and by so doing established MNC-type 'producer-driven' chains, they have been followed in recent years by companies from South Korea and Taiwan (Chapter 2). In the latter cases MNC subsidiaries have also been established in Europe and North America, but increasingly South Korean and Taiwanese companies are dispersing their activities to less developed economies in the Asia-Pacific region, largely in search of lower labour costs. Thus, to give two examples, Taiwanese companies-particularly in electronics products-are now the most important source of new manufacturing investment in Malaysia, while South Korea's largest (and the world's second largest) steel producer, POSCO, is one of the leading foreign investors in heavy industry in China.

Though the economic integration in the region that is being forged by East Asian and other trans-Pacific (particularly US) MNCs is important, not least because of the developmental consequences of the regional divisions



### Figure 14.2 Trans-Pacific buyerdriven chains

of labour that are being established, it is the simultaneously integrative and fractious consequences of buyer-driven commodity chains that may prove more significant in the longer term.

Figure 14.2 provides a simplified visual indication of how the commodity chain for garments sold in the USA under US company brand names is organized. Designs are typically developed under the auspices of the 'buyer' company in the USA. These are transmitted to companies in Hong Kong who organize production by companies in Guangdong Province, China (adjacent to Hong Kong), in which the Hong Kong company will typically be an investor. In both Hong Kong and China parts of the garment may be subcontracted to various other companies, or indeed women working in their own homes (see Chapter 5), depending in part on its value and complexity. Once the garment has been fully assembled, the buyer's brand label is attached and the Hong Kong company which has the contract with the US buyer takes responsibility for organizing shipment to the US or other major markets (see also Chapter 10).

A number of things need to be emphasized about this system of international production.

- Largely in the case of garments only, the location for basic assembly is shifted, often at short notice, from country to country (for instance, to the Portuguese administered Chinese enclave of Macao, Thailand, Vietnam, Cambodia, in addition to China), utilizing overseas Chinese business networks in these countries, in order to beat nationally based quota restrictions for garment exports to the USA and in search of even cheaper supplies of labour.
- A vast range of commodities vital to the export trade (electronic goods, etc., as indicated above) are produced in East Asia under these sorts of arrangements. One difference, however, is that the vast majority of personal computers, for instance, whose basic production is done by Chinese companies (largely in Fujian Province, adjacent to Taiwan), is organized not by Hong Kong, but by Taiwanese companies. Even major Taiwanese companies such as Acer and Mitac are involved in organizing computer production in this way for buyers such as IBM, Compaq, etc.
- This way of organizing commodity production is known as 'original equipment manufacturing' (OEM), and while it provides benefits for the companies involved it does not deliver the substantial economic benefits derived from being able to sell the product under one's own brand name.
- In addition to the Hong Kong/Taiwan-China connection in commodity chain production in the Asia-Pacific basin, South Korean companies—including major ones such as Hyundai and Daewoo—continue to produce a last proportion (probably still the majority) of their output of computers, TVs and other electronic goods on an OEM basis for Japanese, US and European companies.

As I have signalled above, the developmental consequences—both economic and social—of these global production and marketing systems vary significantly from one node in the chain to another and tend to compound pre-existing tendencies to uneven development in the Asia-Pacific region. While these issues demand more detailed exposition, I shall deal with them later in the chapter. For now we need to return to a question flagged earlier: the nature and significance of the interface between the region's various forms of capitalism and the processes of globalization.

### Economic fissures

As I have hinted above, globalization connects not only firms, markets and populations internationally, it also connects the differing organizational structures and economic, social and political priorities that constitute the different forms of capitalism. In connecting these different forms, global business is able to exploit the variations in wages, labour-management relations and state welfare and regulatory regimes that the various capitalisms embody. To what extent, then, do the various forms of capitalism in East Asia that have been connected and projected outwards to other parts of the world economy by the processes of globalization constitute challenges to the more developed economies both of the Asia-Pacific basin and beyond?

The principal basis of the export economies of the Asia-Pacific NIEs, almost-NIEs and China is manufactured commodities. With the exception of garments and toys, which are currently the preserve of companies in Hong Kong, the coastal provinces of China, Vietnam, Thailand and other ASEAN countries, these manufactures largely consist of consumer electronic products, microcomputers, semiconductors, automobiles and auto components. As such, this export profile brings Korean, Taiwanese and Chinese firms into competition not so much with US or EU firms, but rather with their Japanese counterparts who are more involved in many of these products areas and/or are more export dependent.

The competitive threat of some of the East Asian economies for Japan, however, is based on more than the nature of the products that the various economies manufacture and export. Underlying their product foci are virulent forms of economic nationalism. While for much of the last three decades the principal object of this economic nationalism has been to 'catch up with and overtake' the USA, there is some evidence that Japan also has now become an object for competitive attention. Fuelled as it is by animosity that goes back at least to the dawn of Japan's colonial expansion (from 1895) and exacerbated by experiences of Japanese domination in the Second World War, the competitive threat posed by 'Greater China' and Korea to the Japanese economies on Japanese technology notwithstanding. Given the difficulties of developing effective international institutions for economic co-operation and governance in East Asia (see Chapter 13), the competitive dynamics emerging within the region could become a major cause for concern in the next few years.

Compared with any other form of capitalism in East Asia—and indeed with many of those evident in Western Europe and North America—the Japanese form has been exemplary. While the negative features of the Japanese form are well known (for example, rigidly gender-segmented and discriminatory labour markets, high stress levels generated by the nature of work and life, the relative underdevelopment of the welfare regime, etc.), the productive capacities embodied in its corporate structures and the socially integrative nature of its corporate governance systems (see Chapter 9) are among the best in the world. While the future of the Japanese form of capitalism may be threatened by those of some of its regional neighbours, it is also threatened by the dynamics of globalization more generally and by some of the rigidities of its business system in the context of increasing pressure (from the US Government, the WTO and the EU) to liberalize its markets.

The contraction of world markets for manufactured commodities during the 1980s and early 1990s, coupled with increased competition, the rising value of the yen and increasing labour costs have placed severe strain on the Japanese form of capitalism. Specifically, the 'permanent employment' contract covering around 30 per cent of the workforce in the large companies has had the effect of transforming labour costs—which in Western companies are flexible costs—into fixed costs. While in previous recessions the flexibility for cost cutting by major companies has been found by pushing the consequences of recession out into the subcontracting chain of the *keiretsu* (that is on to smaller firms whose workers are often female and do not have the protection afforded to their counterparts in the major companies), this has not been possible to the same extent in recent times. Additionally, while the institutional relationships between firms in a given *keirestu* have been advantageous to innovation and product quality, they have tended also to become rigidities in periods of recession where they have restricted the capacities of major companies to force down the cost of supplies.

There is now evidence that the response of companies such as Nissan and Hitachi to such rigidities has been to chip away at the 'permanent employment' contract and to disengage from their long-term relations with even some of their primary subcontractors, replacing these with forms of interaction that approximate more strictly market transactions (see Chapter 9 for contextual commentary). Indeed, some commentators have gone so far as to suggest that the processes of globalization will ultimately require the institutional reconstruction of the Japanese economy along Anglo-American lines (as they will other economies, such as that of Germany, which deviate significantly from the more strictly market co-ordinated Anglo-American forms). Be that as it may, it is now clear that the internationalization of Japanese investment has increased exponentially (see Chapter 2) and that this in part may be a response of companies wishing to escape the obligations and costs associated with the social and institutional organization of the Japanese economy.

In addition to the competitive threat that some of the NIEs and almost-NIEs of East Asia are beginning to pose for the Japanese economy, they are posing problems for the other economic giant of the Pacific Basin, the USA. While the USA has been running a substantial negative trade balance with Japan for many years (around US\$50 billion in 1996), more recently it has developed significant imbalances with other Asian economies, and in particular with China where the imbalance in 1996 (thus excluding the imbalance with Hong Kong) was around US\$40 billion (Bernstein and Munro, 1997, Chapter 5). What is interesting about the Chinese case is that its trade surplus with the USA has been built up in little more than ten years to a level that the Japanese economy achieved only after about twenty-five years. When one takes into account the fact that the productive potential of the Chinese economy (particularly now that it includes Hong Kong) is far greater than that of Japan and that there is evidence that state strategies have already begun to move it into higher value-added operations (though note the limitations of this that result from the nature of the international production system—see previous section), then it becomes clear that China may well loom as the major economic problem for the USA in the early decades of the next century. Some of the implications of these developments will be broached in the next section.

### **14.4 Prospects**

While the early decades of the twenty-first century undoubtedly promise significant opportunities for further economic growth, social development and prosperity in the Asia-Pacific region, they also threaten to see the maturation of a number of problems for some societies there, whose significance can already be discerned. While these problems certainly include impending ecological disaster in some parts of the region (see Chapter 12), it is some of the more strictly economic problems of development, and their possible geo-political consequences, that I wish to address here. Specifically I wish to engage with two issues: the first concerns the question of whether there are structural limitations on the prospects for future development in the region; the second broaches the nature of Chinese development and its consequences for China itself and the region more generally.

### Any more 'Japans'?

For some years now, the idea that the next century will see the global economic dominance of the Pacific region (meaning in this case East Asia specifically) has been hyped *ad nauseam* by management consultants, assorted pundits and by the business press and media more generally. While economic growth, market expansion and human progress have undoubtedly proceeded apace in the region, the question must be asked as to whether these developments are inexorable, or whether there are, in fact, structural limitations which may mean that, contrary to the hype, Japan will eventually be seen as not just the first, but possibly the last (at least for the bulk of the next century) fully developed industrial economy in the Asia-Pacific region. While I am aware that Singapore is now close to achieving first rank status as an economy, its small size and the peculiarities of its development suggest that it is a 'deviant case'. As such it is incapable of becoming one of the world's major economies and is peripheral to discussions of development prospects in the region generally.

For those who consider East Asia to be the locus of the world's next economic hegemons, their explanation of how that region will get from where it is now to a position of economic dominance is often associated (sometimes explicitly, but usually implicitly) with what has come to be called the 'flying geese' model of development. While there is some variation in the content of the model—depending on the commentator—the fact that it has been utilized by a number of contributors to this volume (Chapters 2 and 10), suggests that there is no need to delineate the model in detail. What is necessary, however, is to raise doubts about both the empirical and predictive capacities of the model as a prelude to examining whether there may be limits to the development project in East Asia.

First proposed by the Japanese economist Akamatsu Kaname in the 1930s to explain the alternating periods of free trade and protectionism that seemed to characterize the trajectories of economic development, the 'flying geese' analogy has been used in recent decades to suggest that the developing economies of East Asia are likely to follow, one after the other, Japan's lead. More specifically, it is suggested that just as Japan first combined cheap but relatively skilled labour with foreign technologies to produce low value-added commodities (garments, cheap electronic products, etc.) for export, and then moved on to produce more capital-intensive, higher value-added products which were associated with a deepening of indigenous technological and innovative capabilities, shedding the former in the process, so the other economies in the region in wave after wave (first South Korea, Taiwan, Hong Kong, Singapore; then Malaysia and Thailand; next China and Vietnam) will be able to replicate this process. The implication of the model is that while Japanese industry may well stay at the front of the flight of geese by means of its capacity to move into ever more innovative and higher value-added processes and products, the other economies in its wake will still be able to achieve the first rank as economies.

In order to reach—and maintain—'big league' status as an economic power, a society must be able to generate industries whose leading companies can move beyond factor-led competitiveness (for example, low labour costs) to a position where they can institutionalize innovation. Additionally, however, the value-added associated with innovative processes and products needs to be captured within the domestic economy if it is to have a significant effect on development and generalized prosperity. From both these perspectives it is not clear that decisive progress towards the economic first rank has been made by the East Asian NIEs or almost-NIEs, or that it is likely to be in the foreseeable future.

In a widely cited article, Bernard and Ravenfield (1995) deliver what appears to be a crushing blow to the 'flying geese' model. Arguing against the state-centric notion of development that underpins the model, they suggest that the evolution of a regional division of labour in manufacturing industries in the Asia-Pacific region has effectively locked the NIEs and almost-NIEs into an intermediate role from which they show little sign of escaping. They argue that even the lead firms in economies such as South Korea, Taiwan and Malaysia remain overwhelmingly dependent on technological inputs and innovation from Japanese companies on the one hand, and access to US markets on the other. When one adds to this the fact that even some of the globally prominent Korean companies, such as Hyundai and LG (formerly Lucky Goldstar), continue to produce 50 per cent or more of their output on an OEM basis, then the structural limitations on their development become clear.

More specifically, and in spite of optimistic work to the contrary (for instance, Chapter 10; Hobday, 1995), innovation in indigenous companies (as opposed to MNC subsidiaries) in the NIEs and almost-NIEs continues to be limited. Additionally, the continuing dependence on OEM strategies, associated as they are with the commodity chain form of international production, severely restricts their ability to generate and capture the highest value-added (see previous sections and Henderson, 1994). While these issues are problems enough for the Korean *chaebol*, they are even more problematic for Taiwanese and other overseas Chinese companies. While the Korean companies are sufficiently cash rich to undertake the huge investments necessary to establish, maintain and enhance their own brand names (hence dramatically enhancing their value generating and capturing capabilities), overseas Chinese companies are not. As Whitley (Chapter 9) has explained, the nature of the overseas Chinese business system circumscribes managerial inclinations to expand companies beyond a size that can be controlled from within kin and guanxi networks. As a consequence such companies may well be highly profitable, but they are unlikely to generate the huge investment surpluses necessary for them to become 'own brand manufacturers' (OEMs).

While Bernard and Ravenfield (1995) may have underestimated the capacity of states in the region to continue to perform developmental roles (hence helping to propel indigenous companies up the learning/innovation curve), the continuing dependence of NIE and almost-NIE firms on US (and to a lesser extent EU) markets—the growth of Asian markets notwithstanding—exposes them to protectionist impulses in those markets that could yet be a consequence of the trade imbalances discussed earlier.

When it comes to assessing the development prospects for the NIEs and almost-NIEs of East Asia, then, it does appear that structural limitations, both endogenous (domestic) and exogenous (international) in origin, are an important part of the picture. For the moment, the most optimistic position that can reasonably be taken is that 'the jury is still out'. For China, however, the story may or may not turn out to be very different. It is to that and related issues that I now turn.

#### The China conundrum

When China awakes, it will shake the world. (attributed to Napoleon Bonaparte)

With around 1.3 billion people, China is the world's most populous country. Over the last decade or so, the Chinese economy as a whole has achieved annual growth rates in excess of thirteen per cent per annum. Were these growth rates to continue—and its recent reabsorption of Hong Kong is relevant here—China would displace the USA as the world's largest economy sometime in the next 20 years (see Chapter 1).

Such statistics, however, obscure the fact that while China's most dynamic coastal provinces (such as Guangdong, Fujian, Jiangsu and the Shanghai region) have been racking up growth rates of over 20 per cent per annum, some of the interior and western provinces have economic and social conditions that have barely changed in a century. China, in other words, could well become the dominant economic power of the twenty-first century and by so doing release hundreds of millions of people into prosperity and consumerism. But if it does, it will be via a process of economic development that has probably been the most uneven the world has ever known.

Herein lies a conundrum composed of at least three related parts. Will China indeed become the greatest of the world's economic powerhouses? If it does, will that be associated with political liberalization and with China's development as a positive force for progress and prosperity globally? Or will China, under the strains of uneven development, implode as a unified state, as it has done on occasions in its past? While it is beyond the scope of this chapter and book to explore these questions in detail, a number of points need to be made.



Advertisement for property development, Beijing

The contemporary Chinese version of capitalism, or what we might term 'Market Stalinism', is a contradictory mishmash of elements, probably unlike any other in the world. Its economy can be subdivided into seven components: the state sector; the military sector; the local state sector; the foreign-invested joint-venture sector; the indigenous private sector; peasant agriculture; and the informal sector. While these are related to each other in important ways (not least through transfers of value), they embody their own logics of development, priorities and vested interests. As a consequence, while currently locked into a dynamic whole, there are within and between them sources of tension and potential conflict that under certain circumstances could disrupt the national economy as a functioning system.

The *state sector* consists largely of central and provincial state-owned enterprises involved in steel, chemicals, heavy engineering, shipbuilding, textiles, banking, etc. Additionally, state trading companies operate internationally, and, together with overseas divisions of the Bank of China, are involved in currency and other forms of financial speculation. While privatization of state assets was announced at the 1997 Congress of the Communist Party, the forms it might take or its extent are currently unclear. In the past privatization has been resisted (partly because of its negative implications for employment) and, in any case, even without privatization there is evidence that in some state companies (such as in steel) productivity has been improved and reasonable levels of profitability may be at hand. Taken as a whole, the state remains responsible for the largest contribution to non-agricultural GDP, and is likely to continue to be so for the foreseeable future.

Though formally responsible to the political apparatus of the central Party and state, the *military sector*—the economic arm of the People's Liberation Army (PLA)— constitutes an arena of economic power in its own right. While military involvement in the production of weapons, uniforms, etc., is not unknown in authoritarian Third World countries, what is unusual about the economic operations of the PLA is that it has its own

subsidiaries involved in the manufacture and export of an array of products, from toys to textile machinery, medical equipment, foodstuffs (frozen fish, for instance), etc. Additionally, one of its companies, Poly Technologies, was involved in the recent (1996) case of the illegal import of assault rifles into the USA, and there is evidence that the military has been implicated in the organization of prostitution rings. The PLA also has companies involved in hotels and retailing, international commercial trading, technology acquisition, and international financial transactions.

While some of these operations are designed to acquire the resources and technology necessary to modernize the military, many of them seem to be operated purely for profit. What is more, the PLA does not operate as a unified command when it comes to its economic operations. Rather, provincial command structures seem to be important—with their own distinct economic interests and 'officer-capitalists'—thus raising the possibility of intra-PLA economic rivalry. So committed to commercial operations has the PLA become that some observers have wondered whether it still constitutes an effective fighting force.

An element of China's economic transformation that sets it apart from the former state-socialist societies of Eastern Europe (not to mention helping to improve its economic performance relative to theirs) is the presence of a large *local state sector* consisting of 'township and village enterprises' (TVEs). As the name suggests, these enterprises are largely a rural phenomenon and were developed to help alleviate rural underemployment, improve prosperity and as a consequence help anchor the peasantry on the land. Invested in by local governments, individuals and groups, they are a form of local corporatism, and now constitute one of the most dynamic sectors of the Chinese economy. They are a major source of subcontracted manufacturing operations which link them to firms in Hong Kong and Taiwan, and via these 'overseas Chinese' business networks to companies in the USA and Europe.

Another significant contributor to China's economic dynamism has been the growth of *joint-venture* companies in which foreign and domestic interests have collaborated in manufacturing (for example, garments, electronic products, Pharmaceuticals, cars, steel, glass, aerospace) and services (for example, hotels, retailing, real estate). While these joint-ventures have sometimes been between major multinationals on the one hand (for example, Volkswagen, Chrysler, Hewlett Packard, Unilever) and state or private Chinese interests on the other, more often they have been between overseas Chinese companies (particularly from Taiwan and Hong Kong, though the population of the latter is now no longer 'overseas'). Indeed, it is estimated that Hong Kong companies remain responsible for around 60 per cent of all 'foreign' direct investment in China. These joint-venture companies, embedded in global commodity chains described in an earlier section, are one of the principle conduits for China's growing integration with the world economy.

Paralleling the growth of TVEs and joint-venture operations has been an explosion of *domestically-owned private* companies. Probably numbering in their millions, and with many of their entrepeneurs disgorged from the ranks of the PLA, these companies are often small, family-run businesses engaged in such activities as manufacturing, construction, retailing, etc. Some of them, however, have grown to a significant size and have become the economic basis for a burgeoning capitalist class. While many of these companies are oriented towards the domestic market, significant numbers of them are involved in subcontracting relations with joint-ventures (in garments, toys, consumer

electronics, for instance) and as a consequence are integrated into the world economy via the particular commodity chains constituted in those industries.

Agriculture continues to be the primary basis for the livelihood of about 60 per cent of China's population. Though the productivity of the land adjacent to the major urban centres has increased dramatically as a result of the relaxation of state controls, agricultural production in more remote areas has stagnated. As a consequence, rural poverty continues to be widespread and this has set in train waves of rural-urban migration which in size are probably historically unprecedented. Given the state's increasing inability to control internal population movements, with migration has come an explosion of urban underemployment. Numbering tens of millions—possibly approaching 100 million—these migrants live on the margins of society. They are employed on a casual basis in the most menial, low paid jobs; many of these are in the informal sector and sometimes merge with criminal activity. Additionally, as welfare benefits are tied to official place of residence or to the employer (where the latter is a state-owned enterprise), the new migrants often find themselves among the most degraded and exploited of Chinese society.

The *informal sector* (or 'grey' economy) is an increasingly important feature of China's Market Stalinism. Although it is impossible to estimate the number of people who are active in the informal sector or the size of the sector's contribution to GDP, it seems likely that proportionally, in both senses, China's informal sector is amongst the largest in the world. Informal sector operations range from 'skin trade' activities such as begging to homeworking (for instance, women sewing garments in their own homes), to small manufacturing or construction businesses, to petty crime, to organized criminal gangs ('triads') engaged in running prostitution, drugs, smuggling and money laundering. In the latter case, the triads have globally-organized networks and are believed to be responsible for handling a significant proportion of the world's illegal drug trafficking, now estimated to be worth US\$4 billion a day.

As can be inferred from these comments on the nature of Market Stalinism, the Chinese economy is cut through with multiple contradictions and has generated very high levels of income inequality and burgeoning crime and corruption. Added to this is the fact that some of the provinces (such as the coastal ones referred to above) are now 'out of control' as far as the central government is concerned. Some of them resist the transfer of taxation revenues to the central authority (thus reducing the state's capacity to redistribute income), while others have established internal tariff barriers to restrict the flow of competitive goods produced in other provinces. The government of Shanghai, for instance, passed a law in 1996 that restricts the size and capacity of cars that can be used as taxis. It so happens that the only car built in China that conforms to the new regulations is the Volkswagen model produced by a joint-venture company in Shanghai!

As a consequence of all these elements and more, Market Stalinism is clearly an unstable, transitional form of capitalism. The question, of course, is what type of economy and polity is China in transition to? There seem to be three general scenarios of relevance here.

• The first is that the tensions of uneven development will tear the central state apart. This has happened periodically in Chinese history, particularly in the context of weakened dynastic rule, and has resulted in the degeneration of society and state into control by provincially/ locally-based warlords. Were a similar phenomenon to occur in the

twenty-first century, then presumably much of the interior of the country would collapse into a form of 'bandit capitalism', similar to some of the territories of the former Soviet Union. Those provinces which already have high degrees of economic and *de facto* political autonomy, and are thoroughly imbricated in overseas Chinese networks and the world economy, however (Guangdong, Hong Kong, Fujian, Shanghai, etc.), would presumably pursue that autonomy to its logical conclusion and become significant powers in their own right (see Goldstone, 1995, and Huang, 1995, for opposing arguments on the likelihood of this scenario).

- The second suggests that the unity of the central state will remain intact and, with further economic reforms, compromises and repression, the spectre of political reform will be held at bay. As a consequence, Market Stalinism will mutate into a more thoroughly capitalist, but still authoritarian, economic and political form, approximating that of South Korea between the early 1960s and mid 1980s, or Indonesia today. Under these circumstances China will continue on its path to becoming the world's dominant economic power. As it does, and particularly if China insists on the absorption of Taiwan against the wishes of the Taiwanese people, then economic competition in the Pacific (particularly with Japan and the USA) could rapidly translate into political conflict. China, in other words, would emerge as the world's number one geo-political problem.
- The third scenario suggests again that the unity of the central state will remain intact but that the deepening of capitalist social relations and the class forces that it will release will result in a successful struggle for political reform. China, in other words, will take a similar path to political modernization and 'democracy' as have South Korea and Taiwan in recent years. Were this to happen, then China could still take its place as the dominant economic power, but in this case it would become a positive force for global prosperity and progress.

As the social sciences, for methodological reasons, cannot predict the future, we are unable to say which of these scenarios is likely to come to fruition. What we can say, however, is that the resolution of the question of Market Stalinism is fundamental not only to the future of China, but to the Asia-Pacific region generally, if not the globe.

### 14.5 Conclusion: opportunity and danger

As the end of the twentieth century rapidly approaches, global society has reached a 'hinge of history': one of those critical moments in social evolution where the possibilities for the future are delicately poised. Hinges by their nature open up new vistas, while closing down others. But the opening and closing may be only temporary, always bearing the possibility of partial reversal. Until the new vistas, the new possibilities, are rendered relatively permanent, they are inherently unstable; they reflect periods of simultaneous opportunity and danger.

As it is with the world, so it is with the Asia-Pacific region. The societies and economies that compose the region have come a very long way in a much shorter time than their European and North American counterparts. The bulk of that journey, at least since the end of the Second World War, has been overwhelmingly positive for humanity and prosperity, the horrors of the Korean and Vietnamese Wars and other conflagrations, enforced starvations, etc. notwithstanding. But the Asia-Pacific region is now, very much, at a critical turning point. Even if one rejects the hype of the 'Pacific Century', that region may well prove to be the key to what the twenty-first century holds for us all. On the one hand it is quite possible that the human ingenuity that the Asia-Pacific region possesses in abundance will be further released and channelled in ways that will benefit humankind globally. On the other, the contradictions of its economic and political development, and the international resistance that its economic prowess and geo-political muscle may engender, could well thrust the globe into a new age of barbarism.

The essays in this book, though written from different theoretical positions and with different concerns, implicitly address one or more parameters of the current crisis (in the Chinese sense) of the Asia-Pacific region. In this concluding chapter, I have tried to make explicit some of what is implicit. Inevitably, however, I have been selective in the issues with which I have engaged, and have given them my own spin. We live in exciting times, with economic and social opportunities undreamed of even a generation ago. It is up to us all to see that we do not live in dangerous ones.

### References

- Amsden, A.M., Kochanowicz, J. and Taylor L. (1994) The Market Meets its Match: Restructuring the Economies of Eastern Europe, Cambridge, MA, Harvard University Press.
- Balassa, B. (1991) Economic Policies in the Pacific Area Developing Countries, London, Macmillan.
- Bernard, M. and Ravenfield, J. (1995) 'Beyond product cycles and flying geese: regionalization, hierarchy and the industrialization of East Asia', World Politics, vol. 47, no. 2, pp. 171–209.
- Bernstein, R. and Munro, R.H. (1997) The Coming Conflict with China, New York, Knopf.
- Castells, M. (1996) The Rise of the Network Society, Oxford, Blackwell.
- Castells, M. (1997a) The Power of Identity, Oxford, Blackwell.
- Castells, M. (1997b) End of Millennium, Oxford, Blackwell.
- Cheung, S.N.S. (1986) Will China Go 'Capitalist'?, London, Institute of Economic Affairs.
- Gereffi, G. (1996) 'Global commodity chains: new forms of co-ordination and control among nations and firms in international industries', *Competition and Change: The Journal of Global Business and Political Economy*, vol. 1, no. 4, pp. 427–39.
- Gereffi, G. and Korzeniewicz, M. (eds) (1994) *Commodity Chains and Global Capitalism*, New York, Praeger.
- Goldstone, J.A. (1995) 'The coming Chinese collapse', Foreign Policy, vol. 96, pp. 35–52.
- Harris, N. (1987) The End of the Third World, Harmondsworth, Penguin.
- Hayter, T. and Watson, C. (1985) Aid: Rhetoric and Reality, London, Pluto Press.
- Henderson, J. (1993) 'Against the economic orthodoxy: on the making of the East Asian miracle', *Economy and Society*, vol. 22, no. 2, pp. 200–17.
- Henderson, J. (1994) 'Electronics industries and the developing world: uneven contributions and uncertain prospects' in Sklair, L. (ed.) *Capitalism and Development*, London, Routledge.
- Henderson, J. (ed.) (1998) Industrial Transformation in Eastern Europe in the Light of the East Asian Experience, London, Macmillan.
- Henderson, J. and Appelbaum, R.P. (1992) 'Situating the state in the East Asian development process' in Appelbaum, R.P. and Henderson, J. (eds) *States and Development in the Asian Pacific Rim*, Newbury Park, Sage Publications.
- Hirst, P. and Thompson, G. (1996) Globalisation in Question, Cambridge, Polity Press.
- Hobday, M. (1995) Innovation in East Asia, Aldershot, Edward Elgar.

- Hsiao, H.H.M. (1995) 'The state and business relations in Taiwan', *Journal of Far Eastern Business* vol. 1, no. 3, pp. 76–97
- Huang, Y. (1995) 'Why China will not collapse', Foreign Policy, vol. 96, pp. 54-68.
- Johnson, C. (1988) 'The Japanese political economy: a crisis in theory', *Ethics and International Affairs*, vol. 2, pp. 79–97.
- Lal, D. (1983) The Poverty of' Development Economies', London, Institute of Economic Affairs.
- Little, I., Scitovsky, T. and Scott, M. (1970) *Industry and Trade in Some Developing Countries*, Oxford, Oxford University Press.
- Lo, D. and Radice, H. (1998) 'Institutional change and economic development in East-Central Europe and China: contrasts in the light of the "East Asian model"' in Henderson, J. (ed.) Industrial Transformation in Eastern Europe in the Light of the East Asia Experience, London, Macmillan.
- Maidment, R., Goldblatt, D. and Mitchell, J. (eds) (1998) *Governance in the Asia-Pacific*, London, Routledge in association with The Open University.
- Maidment, R. and Mackerras, C. (eds) (1998) *Culture and Society in the Asia Pacific*, London, Routledge in association with The Open University.
- McGrew, T. and Brook, C. (eds) (1998) Asia-Pacific in the New World Order, London, Routledge in association with The Open University.
- Nee, V. and Su, S. (1998) 'Institutional foundations of robust economic performance: public sector industrial growth in China' in Henderson, J. (ed.) *Industrial Transformation in Eastern Europe in the Light of the East Asia Experience*, London, Macmillan.
- Nolan, P. (1995) China's Rise, Russia's Fall: Politics, Economics and Planning in the Transition from Stalinism, London, Macmillan.
- Qian, Y. and Weingast, B.R. (1997) 'Institutions, state activism and economic development: a comparison of state-owned and township-village enterprises in China' in Aoki, M., Kim, H.K. and Okuno-Fujiwara, M. (eds) *The Role of Government in East Asian Economic Development*, Oxford, Clarendon Press.
- Runciman, W.G. (1995) 'The "triumph" of capitalism as a topic in the theory of social selection', New Left Review, no. 210, pp. 33–47.
- Sachs, J.D. and Woo, W.T. (1994) 'Structural factors in the economic reforms of China, Eastern Europe and the former Soviet Union', *Economic Policy*, no. 18, pp. 101–31.
- Thornton, J. (1995) 'The Japanese bureaucracy and economic development: are there lessons for Russia and the reforming socialist economies?' in Kim, H.K., Muramatsu, M., Kempel, T.J. and Yamamura, K. (eds) *The Japanese Civil Service and Economic Development*,, Oxford, Clarendon Press.
- Vartiainen, J. (1995) 'The state and structural change: what can be learned from the successful late industrializers?' in Chang, H.J. and Rowthorn, R. (eds), *The Role of the State in Economic Change*, Oxford, Clarendon Press.
- Wade, R. (1990) *Governing the Market: Economic Theory and the Role of Government in East Asian Industrialisation*, Princeton, Princeton University Press.
- Wade, R. (1996) 'Japan, the World Bank and the art of paradigm maintenance: the East Asian miracle in political perspective', *New Left Review*, no. 217, pp. 3–36.
- World Bank (1981) Accelerated Development in Sub-Saharan Africa, Washington DC, The World Bank.
- World Bank (1983) World Development Report 1983, Washington DC, The World Bank.
- World Bank (1993) *The East Asian Miracle*, New York, Oxford University Press and The World Bank.

# Acknowledgements

Grateful acknowledgement is made to the following sources for permission to reproduce material in this book:

### Figures

Figures 1.1 and 11.1: World Bank 1993, *The Asian Miracle: Economic Growth and Public Policy*, Oxford University Press, Inc., New York; Figure 1.3: WTO Overseas Trade Statistics 1997, *Financial Times*, 11 April 1997; Figures 4.2 and 4.3: *Pacific Economic Development Report 1995: Advancing Regional Integration, 1994*, Pacific Economic Co-operation Council, Point Publishing; Figures 4.4 and 4.5: Kwan, C.H. 1996, 'A Yen Bloc in Asia: An integrative approach', *Journal of the Asia-Pacific Economy*, **1**(1), Routledge; Figures 4.6 and 4.7: 'East Asian stock prices and currency falls 1997' and 'Real GDP growth rates 1989–1996', *The Observer*, 31 August 1997; Figure 12.2: Dalzell, A., Adams, T. and Polunim, N. 1995, *Management of South Pacific Inshore Fisheries:* Paper to the FFA/SPC Workshop, 26 June–7 July 1995, South Pacific Commission.

### Tables

*Table 1.1:* Petri, P.A. 1994, 'The East Asian trading bloc: an analytical history' in Garnaut, R. and Drysdale, P. (eds) *Asia Pacific Regionalism*, HarperCollins Publishers Australia; *Table 1.2:* 'GDP growth rates annual average', *The Economist*, 1 March 1997, © The Economist, London 1997; *Table 2.5:* Kwan, C.H. 1994, *Economic Interdependence in the Asia-Pacific Region*, Routledge; *Table 3.5:* Reprinted from *European Economic Review*, **38**(3/4), April, Young, 'Lessons from the East Asian Nics: A contrarian view', p. 970, Copyright 1994, with kind permission from Elsevier Science Ltd, The Boulevard, Langford Lane, Kidlington, OX5 1GB, UK.

### Illustrations

Cover: Hulton Getty; p. 15: Rhodri Jones/Panos Pictures; p. 29: *The Economist*, 24 April 1993, © The Economist, London 1993; p. 47: Börje Tobiasson/Panos Pictures; p. 56: Sean Sprague/Panos Pictures; p. 76: Panos Pictures; p. 94: Chris Stowers/Panos Pictures; p. 113: Mark Edwards/Still Pictures; p. 122: Andrew Gunnartz/Panos Pictures; p. 132: Heldur Netocny/Panos Pictures; p.164: *Foreign Affairs*, May/June 1996, p. 115; p.264: Oliver Garcia/Panos Pictures; p. 306: Trygve Blstad/ Panos Pictures; p. 324: Don

Hinrichson/Still Pictures; p. 327: Andy Crump/Still Pictures; p. 378: Chris Stowers/Panos Pictures.

# List of contributors

**Brendan Barker** is located at the British Council in Tokyo, Japan. He works on issues in technology and innovation. Prior to moving to Tokyo he was Research Fellow at the Policy Research in Engineering, Science and Technology (PREST) centre, University of Manchester and a visiting scholar at the National Institute of Science and Technology Policy (NISTEP), Tokyo.

**Lucie Cheng** is Professor of Sociology and of Urban Planning, University of California, Los Angeles, USA. She served as the Director of the University's Asian American Studies Centre, and as the Founding Director of the Centre for Pacific Rim Studies. On leave from UCLA, she is the Founding Dean of the Graduate School for Social Transformation Studies, Shih-Hsin University, Taipei. Prof. Cheng publishes in the fields of international migration, and gender and development.

**Stephen W.K.Chiu** is Associate Professor of Sociology at The Chinese University of Hong Kong. He is the co-author of *East Asia and the World Economy* (1995) and *City-States in the Global Economy* (1997). He has published articles and book chapters on labour unions, industrial restructuring, and East Asian development.

**Partha Gangopadhyay** is Lecturer of Economics at the University of Western Sydney, NSW, Australia. He is the author of a number of articles on development economics and decision theory. He was previously Visiting Fellow, Delhi School of Economics, a Consultant at the National Institute of Public Finance and Policy, and a lecturer at Griffith University, Brisbane, Queensland, Australia.

**Akira Goto** is Professor at the Institute of Innovation Research, Hitotsubashi University, Tokyo. He is also the Director of Research, National Institute of Science and Technology Policy (NISTEP), Tokyo, Japan.

**Jeffrey Henderson** is Professor of International Economic Sociology at the Manchester Business School, Manchester University, UK. Between 1981 and 1991 he worked at the University of Hong Kong. His most recent books are *The Globalization of High Technology Production* (Routledge, 1989) and (with Richard Appelbaum) States and Development in the Asia Pacific Rim (Sage, 1992) and Industrial Transformation in Eastern Europe in the Light of the East Asian Experience (Macmillan, 1998). He is coeditor of the journal Competition and Change.

**Richard Higgott** is Professor of International Political Economy and Director of the ESRC Centre for the Study of Globalisation and Regionalisation at the University of Warwick. He is the editor of *The Pacific Review*.

**Ping-Chun Hsiung** is Associate Professor of Sociology, University of Toronto, Scarborough, Canada. She is the author of *Living Rooms as Factories: Class, Gender, and the Satellite Factory System in Taiwan.* 

**Colin Hunt** lectures in environmental and resources economics at the National Centre for Development Studies at the Australian National University. He has also taught at the University of Papua New Guinea. Over the last 20 years, he has worked on development projects in the Asia-Pacific region. From 1991 to 1994 he represented the Australian conservation movement on the planning team that prepared the 25-year plan for the Great Barrier Reef World Heritage Area.

**Paul Lubeck** is Professor of Sociology and Director of the Center for Global, International and Regional Studies at the University of California, Santa Cruz, where he teaches sociology, political economy and development studies. He has published extensively on industrializing states, labour, Islamic social movements and regional strategies for economic development. His book, *Islam and Urban Labour in Northern Nigeria*, was awarded the Herskovits Prize in 1987.

**Tai-lok Lui** is Associate Professor of Sociology at The Chinese University of Hong Kong. He is the author of *Waged Work at Home: The Social Organization of Industrial Outwork in Hong Kong* (1994) and the co-author of *City-States in the Global Economy* (1997).

**Eric D.Ramstetter** specializes in the study of multinational corporations in the Asia-Pacific region. Between 1988 and 1998 he was Professor of Economic Policy, Faculty of Economics, Kansai University, Osaka, Japan. In April 1998 he became Research Professor at the International Centre for the Study of East Asian Development, Kitakyushu, Japan.

**Ajit Singh** has been teaching economics at Cambridge University since 1965. He is currently Professor of Economics at the University and Senior Fellow of Queens' College, Cambridge. He has been a senior economic adviser to the governments of Mexico and Tanzania and is a consultant to various UN developmental organizations. He is the author of *Takeovers: Their Relevance to the Stockmarket and the Theory of the Firm* and co-author of *Growth, Profitability and Valuation,* both published by Cambridge University Press. His most recent books are the monograph *Corporate Financial Patterns in Industrialising Countries* Technical Paper No.2, World Bank/IFC, Washington DC, 1995, and the edited volume (with A.Dutt, and K.Kim), *The State, Markets And Development* (Edward Elgar, 1994).

**Grahame Thompson** is Senior Lecturer in Economics, the Open University, UK. His latest book (with Paul Hirst) is *Globalization in Question* (Polity Press, 1996). He is also managing editor of the journal *Economy and Society*.

**R.Gerard Ward** is Professor of Human Geography, Research School of Pacific and Asian Studies, Australian National University, Canberra, Australia. He was Director of the School from 1980 to 1993. He was Foundation Professor of Geography in the University of Papua New Guinea, 1967–91. He has been conducting research in the Pacific Islands since 1956. His current work includes an extensive study of the consequences of distance and isolation within the Pacific Islands region. Amongst his most recent books are *New Directions in the South Pacific* (Academy of the Social Sciences in Australia, Canberra, 1988), and *Land, Custom and Practice in the South Pacific* (Cambridge University Press, Melbourne, 1995).

**Richard Whitley** is Professor of Organisational Sociology at the Manchester Business School, University of Manchester, UK. He is also Co-Director of the European Science Foundation's Programme on European Management and Organisations in Transition and Chair-elect of the European Group for Organisational Studies. Among his recent books are *Governance at Work: the Social Regulation of Economic Relations* (Oxford University Press, 1997), *The Changing European firm* (Routledge, 1996), and *Business Systems in East Asia* (Sage, 1992).

**Ippei Yamazawa** is Professor of economics, Hitotsubashi University, Tokyo, Japan. He has published books and journal articles on international trade and investment, trade policy, and economic development. Since 1968 he has been working on the issue of Asia-Pacific economic co-operation. He was the Japanese member of APEC Eminent Persons Group (1993–95). Since then he has been Coordinator, APEC Study Centre Japan Consortium.

**Elspeth Young** is a Reader and Director of Studies in Graduate Studies in Environmental Management and Development, Research School of Pacific and Asian Studies, Australian National University, Canberra, Australia. She has formerly held academic posts within other departments at the Australian National University, at the University of Papua New Guinea and at the University of New South Wales. Her major books include *Third World in the First: Development and Indigenous Peoples* (Routledge, 1995); and *Aborigines, Land and Society* (Longmans, Melbourne, 1993).

## Index

Note: page numbers in **bold** type indicate box references

Acer 246, 367, 372 AFTA 51, 168, 270, 339, 340, 343, 346 agriculture 23-4, 175, 312-15 in China 380 new systems 312-15 subsistence 305, 309, 313 swidden cultivation 310, 322 traditional see traditional production Aichi Steel 221 aid official development assistance 97 US post-war aid 66, 96-7, 261-2, 277, 357 'air display' technological progress 269 air pollution 325 Amsden, A. 138, 144, 146-7, 152, 230, 244, 363 Anglo-American business systems 217-20, 217-18, 238-40 Anglo-American economies 8 Anglo-American financial systems 87, 90, 92, 95 ANZCER 168 APEC 10, 163-84, 336-7, 341-5, 347-54 APEC Work Projects 172 Bogor Declaration 164, 171, 173, 177-8, 342, 351 economic co-operation 165-70, 172, 336-7, 341-4 Eminent Persons Group 16, 342, 352 influencing state policy 340-1, 353 membership 164, 167-70 'open regionalism' 16, 181-2, 343 Osaka Action Plan 165, 171-2, 179, 342 regional integration 164-7, 179-81 relations with EU 182-4 see also ASEM structure and institutional development 164, 343, 352-3 trade liberalization 164-5, 171-81, 351 concerted unilateral 171-2, 173, 178-9, 351

facilitation measures 178-9 most favoured nation (MFN) basis 180, 182, 343 trade barriers 173-8 Working Party on Industrial Science and Technology (WGIST) 270-1 ARF (ASEAN Regional Forum) 343, 345, 350 ASEAN 51, 100-1, 336, 343-4, 346 Asia-Pacific trade 28-30, 31-2, 35, 38, 103-5 and economic integration 51, 165-6, 343 technology and innovation 258-9, 260-1, 270 trade liberalization 341, 342 ASEAN Free Trade see AFTA ASEAN Regional Forum see ARF ASEM (Asia-Europe Meeting) 183-4, 336-7, 345-54 compared with APEC 350-1 and global inter-regional triangle 345-6, 347 influencing state policy 340-1, 353 and USA 347-50 Asia-Europe Foundation 349 Asia-Europe Meeting see ASEM Asia-Pacific Economic Co-operation see APEC 'Asian' identity 343-4, 346 Association of South-East Asian Nations see ASEAN Australia 3, 8, 20, 22, 49, 93 indigenous minorities 307, 308-9, 330-1 mining activities 328, 329-31 technology and innovation 255, 256, 266 trade and FDI 35, 36, 46, 109 unions 219 Australia-New Zealand Closer Economic Relations Agreement see ANZCER Austria 363 Bae, K. 229, 231 Baht economic zone 168-70 balance of payments cycle 47-8 Balassa. B. 138. 139-41. 180. 360 bandit capitalism 381 Bangkok 323, 325 banks and banking Anglo-American 92, 215, 218 development 92-4 Islamic 86 Japanese 88-9, 92, 150-1, 235, 242, 366 Korean 25, 88-9, 92, 228, 242 oligopolistic 88–9 retail 84 state controlled 85-6, 88-9, 228, 233, 243, 360 Taiwanese 233, 235 see also financial systems Beijing 325 Bernard, M. and Ravenhill, J. 269-70, 376-7 Bhagwati, J. 29, 51

Bogor Declaration, APEC 164, 171, 173, 177-8, 342, 351 bond markets 84, 92, 97 bonds, government 84, 98 Brazil 14 Brunei 127 Buddhism 290 budget deficits ASEAN 46 US 32, 41-2 bureaucracy 145-6, 147-8, 152, 155, 158-9, 279 Burma/Myanmar 8, 20, 84 business systems 10, 213-47 Anglo-American system 217-20, 217-18, 238-40 changes in 240–7 Chinese family businesses (CFBs) 233-8, 246-7, 367 components of 214-17, 216-17 economic agents 214-15, 218-19, 220-2, 227-8, 233-4, 239 Japanese 220-6, 226, 238-40, 241-2 Korean 227-32, 232, 238-40, 243-5 market organization 215, 219, 222-3, 235-6, 238-9 overall comparisons 238-40, 239 Taiwanese 233-40, 238, 245-7 work co-ordination and control 216, 219-20, 223-6, 229-32, 236-8, 239 business-state relations 10, 26-7, 72-3, 80, 147-8 developmental states 147-8, 149-54 interdependence 149-54, 241 Cambodia 20, 50 Canada 8, 20, 22, 49 foreign MNC presence 192-3, 207-8 indigenous minorities 307, 308, 309, 331 mining activities 328, 331 R&D, technology and innovation 255, 256 trade and FDI 31, 36, 46, 192-3 Canada-USA Free Trade Agreement see CUSFTA capital adequacy constraints 85 capital, financial and real 83-4 capital flows 85, 94, 98-9, 105 capital markets 140-1, 366 capitalism ersatz 290 forms of 7-9, 365-8 globalization and 364-74 Catholicism 290 Central Provident Fund (CPF), Singapore 89 centralization, decision making 216, 219, 230 chaebol, South Korea 89, 142-3, 151, 156, 227-32, 243-5, 364 Chakravarty, S. and Singh, A. 73-4 The Challenge of Development, 1991 World Development Report 61, 66-7 Cheng, Tun-Jen 155 child prostitution 131-3, 301

Chile 3, 88, 278 poverty and inequality 285-6, 299-300 technology and innovation 256, 257 China (PRC) 20, 23, 74-6, 92, 94, 278, 295-7, 377-82 agriculture 380 chaotic capitalism/market Stalinism 9, 365, 367, 378-82 command economy/state-led industrialization 7, 24, 25, 74-6, 361-2, 363 corruption 380 economic growth 11-14, 55, 58, 74-6, 79, 377 economic reform and private enterprise 75-6, 79, 260, 295-7, 378, 380 education 120, 278 foreign MNCs in 205-8, 296 future for 11, 13-14, 343, 350, 381-2 gender issues 114-19, 120-1, 129, 130, 301 global commodity chains 370, 371, 372, 379-81 guanxi and kin networks 376-7, 379 inward FDI 32-3, 44-5, 97, 205-7, 296, 379-80 joint ventures 379-80 labour markets 50, 75, 114-19, 120-1, 301 liberalization 295-6, 350 migration 129, 130, 289-90, 379-80 rural-urban 322, 380 PLA and military sector 378-9, 380 population 278, 377 portfolio investment 46 poverty and inequality 60, 260, 285-6, 296-7, 380-1 property rights 14, 361-2 responsibility system 296 savings 64, 65 state sector 296, 378, 379 stock market 93 technology and innovation 255, 256, 259-60, 262, 264, 266, 376-7 township and village enterprises (TVEs) 75, 296, 379 trade 350. 374 Chinese family businesses (CFBs) 233-8, 246-7, 367 Chrysler 379 citizenship 276-81 Clark, R. 221, 224 co-operation institutional 336-7, 339, 341-4 technical 172 see also economic co-operation; regional integration co-ordinated managerial capitalism 365, 366 co-ordinated proprietorial capitalism 365, 367 co-ordination role of state 73, 80-1 work 216-17, 219-20, 223-6, 229-32, 236-8, 239 Collective Action Plans (CAPs), APEC 178 collective capitalism 365, 366 colonialism European 289-90, 354

Japanese 262, 282-4 and land tenure systems 308-9, 311-12 command economies 7-8 financial systems in 85-6 commercial fisheries 317, 327-8 Committee on Science and Technology, ASEAN 270 Common Agricultural Policy (CAP), EU 180 comparative advantage 35-8, 36, 39, 51 competition China and 381-2 intra-regional 356, 372-4 for use of resources 318-19, 327-8 competitiveness 99-100, 104-5 complementarity 39, 42, 170 Confucianism 282, 357, 358 contagion, market 107, 108 Cook Islands 317, 326 corporate bond market 92 corruption 287, 293-4, 380 Council for Europe Asia Co-operation (CAEC) 349 Council for Security Co-operation in the Asia-Pacific (CSCAP) 349 cronyism 293-4 curb markets 91-2 currency East Asian single 102 and exchange rate regimes 99-105 current accounts and capital inflows 98-9 deficits and surpluses 108-9 CUSFTA 181 Daewoo 142-3, 372 Davis, D. see Thistlethwaite, B. and Davis, D. debt, in Asia-Pacific 30, 47-8 decentralization, China 76, 79, 260 decision making, centralization 216, 219, 230 deforestation 306 dependency theory 358 deregulation, financial 95-6, 109 derivatives trading 98 developing economies 20, 23, 46, 84, 359, 362-3 development and agriculture 23-4 export-led 26-8, 139-40, 155, 287-9, 359-60 flying geese model 10, 36-8, 37, 51, 267-70, 375-6 and gender equality 113-20 import-substituting 10, 24-5, 139, 359 market-led model 138-43, 277, 358-62 stages of 20-3 state-led 23-8, 39-40, 80-1, 142-9, 154-9, 362-4 total factor productivity (TFP) approach 61-9, 79-80

and trade surpluses 28-9 development banks 92-4 developmental state 137-8, 143-9, 277-9 bureaucracy 145-6, 147-8, 152, 279 business-state relationship 147-8, 149-54 capitalist 144-8 features 145 limitations 148-9 Devo, F. 144 distribution of income Gini coefficients 284. 286 in NIEs 23, 60, 279-82, 284-9 diversification Japanese firms 220-1 Korean 227-8, 243 Taiwanese 234, 246 diversity 306-7 division of labour ethnic 292-3 regional 339 domestic work 127, 128-30 EAEC (East Asian Economic Caucus) 338, 340, 342, 343, 346 East Asian miracle 68-9, 141-3, 148-9, 277-9, 357-64 developmental statist theories 148-9, 277-9, 362-4 doubts on sustainability 68-9 free market neo-classical approach 60-1, 141-3, 358-62, 364 and gender issues 112-36 geopolitical explanation 277 intellectual significance 357-64 strong state model 148-9 The East Asian Miracle, World Bank report 5-6, 60-1, 61, 72, 279-80, 361 Eastern Europe 359, 361, 362, 363 economic agents, in business systems 214-15, 239 Anglo-American 218-19 Japanese 220-2 Korean 227-8 Taiwanese 233-4 economic co-operation APEC 165-70, 172, 336-7 ASEM inter-regional 183-4 de jure 340-4 free trade areas 168 interstate 343-4 sub-regional informal economic zones 168-70, 169-70 see also economic integration; regional integration; regionalization economic growth 1-7, 11-14, 20-3, 26, 55-81, 374-7 authoritarian systems and 26, 277-8 business-government links and 26-7, 72-3, 80

East Asia 55, 56-74, 139-41 and education 68-9, 120, 121-3 through export-led development 26, 80, 139-40, 155, 359-60 and external stability 30-1, 52 future prospects 374-7 HPAEs 1-2, 279-80 and interest rates 90-1 and investment 47-8, 63-6, 68-9, 72-3, 79, 87-9, 91 neo-classical free market approach 4-7, 138-43, 358-62, 364 and 'openness' 4-5, 63, 66-7, 71-2, 73-4 production function approach 62-3 role of state in 23-8, 39-40, 80-1, 139, 142-9, 154-9, 362-4 slowing 12-13, 96, 108-9 and social equity 279-82 sources of and reasons for 26, 27-8, 60-74, 277-81 technology and 69, 72, 79-80, 250-3, 290 total factor productivity (TFP) approach 61-9. 78-90 and trade liberalization 27-8 World Bank approach 4-6, 60-7, 70-2, 78-80, 138-9, 359-61, 364 see also development economic integration 4, 17, 163-84, 337-44 ASEAN and 51, 165-6, 343 de facto structural 337, 338-9 de jure institutional 337, 341-4 in Europe 163, 167 'strategic' or 'close' 73-4, 76, 79 economic policies see government intervention; macroeconomic policies; 'market-friendly' policies education 68-9, 120-3, 255, 258, 299 and gender equality 120-3, 299 Eminent Persons Group, APEC 342, 352 employment in Anglo-American system 219-20 in Chinese family businesses 236-8 gender issues 112-36, 224-5, 231, 237 Japanese 'permament' 224-6, 242, 368, 373-4 Korean firms 230-2 entrepôts 155, 197 entrepreneurial familism 125-6, 234 environment, and changing production systems 306-7, 312-16, 320-5, 327-31 impact of mining 328-31 and large-scale resource use 327-31 logging 315-16 and population growth 314-15, 320-5, 323-4 equity market 92 ethnicity 51, 52, 289, 291-3 ethnic division of labour 292-3 Indonesia 289, 293, 295 Malaysia 291-3 Taiwan 289 Europe, East Asian FDI in 46, 271

European Union (EU) 5, 163, 167, 178, 181-4, 345 Common Agricultural Policy (CAP) 180 relations with APEC 182-4, 346-7 Towards a New Asia Strategy 346-7 trade with East Asia 12, 346-7 see also ASEM Evans, P. 149, 152, 153, 158 Evergreen 367 exchange controls 85 exchange rate regimes 99-105 and competitiveness 99-100, 104-5 and trade patterns 101-5 expatriates 266-7 export finance 92 Export Processing Zones (EPZs) 124 export-led development 27-8, 80, 139-40, 155, 287-9, 359-60 economic growth through 26, 80, 139-40, 155, 359-60 gender implications 124-5 and government intervention 27-8, 139-40 winners and losers 287-9 externalities (R&D) 251-2, 251 'extra-territoriality' principles 347, 348 facilitation measures, trade liberalization 178-9 family firms Chinese family businesses (CFBs) 233-8, 246-7, 367 finance 89. 91-2 role of women 125-6 family, and working women 124-5 Fanuc 221 Fiji 317 financial integration 105-9 financial intermediation 84-5 financial repression 85, 87, 94, 107 financial systems 83-110, 142 Anglo-American type 87, 90, 92 capital markets 92-3 contagion in 107, 108 curb/informal markets 91-2 development 83-94 development banks 92-4 export finance 92 government intervention 85-7, 110, 142 internationalization 96-9 long-termism 72-3 'market-friendly' policies 85, 90, 96 monetarization 84 regulation and deregulation 89, 95-6, 109 two-tier 91-2 Finland 363 firms

ownership and management of 218-19, 221, 228, 234 relationships between see market organization see also business systems; economic agents; family firms; multinational corporations; small and medium sized enterprises fiscal policy 99 fisheries see marine resources 'flying geese' model of development 10, 36-8, 37, 51, 375-6 and innovation 267-70, 376 Ford 264 foreign direct investment (FDI) 17, 32-3, 43-5, 46-7, 97, 186-7 by Japan 32-3, 40, 44, 52, 262-3 by NIEs 43-4, 46, 97, 187, 243, 263, 290 in Europe 46, 271 indicating MNC activities 186-7, 190, 207-9 intra-regional 32-3, 43-5, 46-7, 97 restrictions on 27, 72, 175, 178 and technology flows 259, 262-3 Fortress Europe 181 France 14, 328 Fraser, Malcolm 166 free trade 141-2 see also 'openness'; protectionism; trade barriers; trade liberalization free trade areas (FTAs) 163, 168 see also AFTA: ANZCER: NAFTA Friedman, D. 153-4, 220 Fujitsu 221 Funo 150 GATT trade liberalization 4-5, 352 Uruguay Round 177-8, 181-2, 348 gender equality 112-36 and economic development 113-20 education 120-3, 299 labour force participation 114-15, 119-20, 298, 300 dual roles for working women 124-5 employment by industry 114, 116, 117 employment by occupation 117, 118, 300 role of state 123-5 wages 117, 119-20, 231 see also women in employment General Agreement on Tariffs and Trade in Services (GATS) 347 Gereffi, G. 369

Germany 14, 58, 365, 366 international role of DM 99-105 global commodity chains 241, 369-72, 376 global inter-regional triangle 338, 345-6, 347 globalization 240-7, 260, 341-2, 344-5, 364-74 and economic fissures 372-4 and economic fusion 368-72 effect on East Asian business systems 240-2, 260 and forms of capitalism 364-8 and networks 368-72 and regionalization 339, 342, 351 and state 341-2, 344-5 Goldstar see LG Group 'governed interdependence theory' 149-54 government failure 25-6 government intervention in capital flows 85, 94 East Asian success 80 and export-led development 27-8, 139-40 financial systems 85-6, 110 interest rates 89-91 in investment 26, 85, 88-9, 94, 139 and 'late industrialization' 144, 146 and market distortion 25, 31, 72, 140, 358-9 and market failure 25-6 in 'market-friendly' approach 66-7, 70 neo-liberal pressure on 95-6 private sector 24-5 in savings 72-3 see also macroeconomic policies; state-led industrialization Greater East Asia Co-Prosperity Sphere 4 growth triangle (Singapore-Johore-Riau) 168, 169 guanxi and kin networks 376-7, 379 Haggard, S. 10, 155 Helms-Burton Act 348 Hewlett Packard 379 high-performing Asian economies (HPAEs) 1-2, 279-80 Hitachi 223, 263, 374 Honda 221 Hong Kong 'chaotic' capitalism 9 economic growth 1, 11-12 FDI inward 33, 196-7 outward 44, 186, 187, 379-80 foreign MNC presence 185, 196-7, 207-8 free market development 141, 148 global commodity chains 370, 371-2 labour markets 49, 50

gender issues 114-19, 121, 124, 127, 134 women migrants 127, 128-9, 130 as NIE 20, 22-3 role of state in development 27, 143, 148, 154, 157 stock market 92-3.98 technology and innovation 256, 266-7 total factor productivity 67, 68 trade 35 entrepôt 155, 197 'hot money' 98-9 household-based entrepreneurship 125-6 housing 87-8 hunting and gathering 305, 308-10 Hyundai 142-3, 228, 229, 231, 372, 376 identity 'Asian' 343-4. 346 and traditional production 303-4 Ikkan 150 IMF 4-5, 138, 305, 359, 361, 364 import-substituting industrialization 10, 24-5, 139, 359 India 14, 64, 65, 97, 360 indigenous minorities, and use of resources 307, 308-9, 330-1 Individual Action Plans, APEC 173, 178, 179 Indonesia 14, 20, 84, 94, 99, 293-5 agriculture and land ownership 295, 307, 312, 322 312, 322 authoritarian capitalism 8, 24, 25, 39-40, 293-4 basic indicators 278 corruption and cronyism 293-4 debt 30, 295 economic growth 1, 12, 25, 28, 39-40 emigration 50, 127, 128, 129-30 ethnicity 289, 293, 295 gender issues 114-16, 118-19, 121, 127, 299-300 investment 64, 65 inward FDI and MNCs 33, 43-5, 74, 97, 201-2, 207-8 Islam 86, 295 mining activities 328 poverty and inequality 48, 50, 60, 285-6, 294-5, 301 savings 64, 65, 88 stock market 93 technology and innovation 256, 259, 266 trade 35, 108-9 unemployment 50 Industrial Bank of Japan 223 industrial giants 21-2 industrial policies 70-1, 142-3, 257-8 and citizenship 276-81 industrial relations Anglo-American 219-20 Japan 223-6

Korea 229-32 Taiwan 236-8 industrial targeting strategy 26-7 industrialization import-substituting 10, 24-5, 139, 359 late 144, 146, 252-3 near-NIEs 289-90 neo-classical approach 141-3, 358-62 statist perspective 143-4 see also state-led industrialization infant industry protection 175 infant mortality 285 inflation 59, 89 informal markets 91-2 infrastructure 94, 139 innovation, systems of 252-60, 267-71, 376 'air display' 269 'flying geese' model 267-70, 376 global 271 national 253-60 regional 260, 267-71 see also R&D; technology institutional co-operation 336-7, 339, 341-4, 350-1, 352-3 integration financial 105-9 technological 265 see also economic integration; regional integration intellectual property rights 172, 175 inter-industry trade 39 interdependence business-state 149-54, 240-1, 367 of Chinese family businesses 233-4, 235 economic 51 interest rates 85, 87, 89-91, 94, 98-9, 106-7 internationalization 370 financial systems 96-9 R&D 258 intra-industry trade 38 intra-regional FDI 32-3, 43-5, 46-7, 97 intra-regional trade 31, 43, 52, 101-2, 103, 240-1 and regionalization 339, 341 investment and economic growth 47-8, 63-6, 68-9, 72-3, 79, 87-9 government intervention in 26, 85, 94 and interest rates 91 see also foreign direct investment (FDI); portfolio investment; savings Irian Jaya 322 Islam 86, 290, 295

Islamic banking 86 Janelli, R.L. 228-30 Japan 31-44, 220-6, 309, 372-4 banks and banking 88-9, 92, 150-1, 235, 242, 366 business systems 220-6, 226, 238-40, 241-2, 260 business-state interdependence 150-3 colonial past 4, 262, 282-4, 373 comparative advantage 36, 37 economic dominance 1, 14, 20, 22, 28-9 economic growth 56-7, 58, 280-1 financial systems 73, 92-3, 96 and 'flying geese' model 37, 268-9, 375-6 foreign and domestic MNCs 51, 185, 193-4, 207-8 global commodity chains 370 income distribution and living standards 48, 49, 280-1 industrial policy 70-1, 257-8 industrial relations 223-6 international role 71, 80, 343 of yen 99-105 investment foreign portfolio 46-7, 241-2 inward FDI 72. 97. 193-4. 242 outward FDI 32-3, 43-5, 46, 52, 97, 242, 262-3 state guidance in 26-8, 39, 72-3, 145-6 US post-war aid 97, 261–2 keiretsu 89, 150-1, 223, 242, 366, 368, 373-4 labour markets gender issues 114-19, 121, 127, 224-5 'permament' employment 224-6, 242, 368, 373-4 wages 50 migration and investment 51 women 127, 130-1, 132 MITI 73, 80–1, **145–6,** 150–1 negotiated/collective capitalism 8-9, 365, 366, 373-4 neo-classical account of 358-9 'openness' 31-2, 71, 73-4, 94 R&D and technology 225, 256, 258, 260-3, 266-7, 271, 376 ringi system 224 savings 72-3, 88 sex industry 130-1, 132 small and medium-sized enterprises 221, 366 state-led industrialization 39, 73, 80-1, 139, 141, 150-2 structural rigidities 22, 368, 373-4 The East Asian Miracle financed by 5-6, 279, 361 trade 12-13, 28-9, 31-44, 339-40 exports 12-13, 35, 42-3, 340 imports 32, 35, 40 intra-industry 38 intra-regional 32-3, 34, 40-3, 339-40

trade surpluses intra-regional 31-2, 34, 99, 105 with USA 28-9, 32, 52 unionization 225-6 zaibatsu 150-1, 221-2, 242 Japan Sea economic zone 168-70 Java 295, 322 Johnson, C. 138, 144-6, 147, 154, 364 kaisha 220-3 Keidanren 151 keiretsu 89, 150-1, 223, 242, 366, 368, 373-4 kigyo shudan 223 Kim, E.M. 229, 230, 231, 243 Kiribati 317, 319, 323, 324-5 Komatsu 221 Korea see North Korea: South Korea Krugman, P. 68–9, 79

Kuomintang 152, 245, 246, 367 Kuwait 127

labour markets 49-51, 52-3, 59, 112-36, 140 in Anglo-American business system 219-20 gender issues 112-36, 224-5, 231, 237 employment by industry 114, 116, 117 employment by occupation 117, 118, 300 labour force participation 114-15, 119-20 wage inequality 117, 119-20, 231 government policies 50-1 labour shortages 49-51, 59 segmentation 52-3 see also migration Lal, D. 360–1 land management, PNG 314-15 land reform 283-4, 289 land tenure 308–15. 309–10 communal 311-12 impact of colonialism 308-9, 311-12 in Indonesia 295, 312 and new systems of agriculture 312-15 under traditional production 308-10, 309-10 land-rich economies 22 Laos 20, 48, 84 'late industrialization' 144, 146, 252-3 Latin America 12, 58, 139-41 poverty levels 48, 60, 285 savings 64, 65, 88 LG Group 57, 142, 263, 376 Little, I. 138, 139, 359-60 Living Rooms as Factories programme 124

logging 311, 314, 315-16, 327 long-termism 72-3 'Look East' policies 28, 291 Lucky Goldstar see LG Group Macau 286 macroeconomic policies industrial 70-1, 142-3, 257-8 influence of institutions 340-1, 353 and innovation 252, 257-8 labour market 50-1, 128, 129, 130 monetary 99 see also 'market-friendly' policies Mahathir bin Mohamed 291, 346 Malaysia 93, 94, 99, 291-3, 309 authoritarian capitalism 8, 25 basic indicators 278 economic growth 1, 12, 28, 108 education 278, 293 ethnicity 291-3 foreign MNC presence 202-4, 207-8 foreign workers 49-50, 127, 128, 129, 291 gender issues 114-19, 121, 298-301 government intervention 25, 291-3 import-substituting industrialization 25 inward FDI 33, 43-5, 74, 97, 108, 202-4, 263 Islamic banking 86 labour markets 49-50, 59 land ownership 312 logging 327 as near-NIE 20, 37, 289-90 New Economic Policy (NEP) 291-3 poverty and inequality 60, 285-6, 291-3, 301 savings 64, 65, 88 SMEs 291, 293 technology and innovation 256, 259, 262, 266, 291, 293 trade 35, 108-9 managerial capitalism 365, 366 Manila 325 Manila Action Plans for APEC (MAPA) 165, 173, 178, 179, 183 marine resources 305, 308, 316-20, 327-8 commercial fisheries 317, 327-8 subsistence fisheries 316-17 traditional systems 305, 308, 316-18 market distortion 25, 31, 72, 140, 358-9 market failure 25-6 'market-friendly' policies 5, 61, 66-7, 70-1, 277 and financial systems 85, 90, 96, 110 market organization, business systems 215, 219, 222-3, 235-6, 238-9 market Stalinism 365, 367, 378-82 market-led development 4-7, 138-43, 277, 358-62, 364

Marshall Islands 317, 323 Matsushita 223 Melanesia 307, 320, 321, 322, 327 Mexico 3, 25, 88, 256, 276-7 basic indicators 278 poverty and inequality 285-6, 299-300 migration 49-53, 127-31, 289-90, 321-6 Chinese 289-90, 379-80 feminization of 127-31 government regulation 50-1, 128, 129, 130 and labour shortages 49-51, 52-3, 59 Pacific Islanders 325-6 rural-urban 321-5 skilled labour flows 266-7 and wages 119, 129 minerals and mining 311, 328-31 Ministry of International Trade and Industry (MITI), Japan 73, 80-1, 145-6, 150-1 Mitac 246, 367, 372 Mitsubishi 150, 222, 223 Mitsui 150, 222, 223 modernization theory 358 monetarization 84 monetary policy 99 most favoured nation (MFN) 180, 182, 343 Mothers' Workshops programme 124-5 Multifibre Agreement (MFA) 178 Multilateral Agreement on Investment (OECD) 178 multinational corporations (MNCs) 17, 27, 185-209, 263-7, 290, 369-72 agents of regional integration 185-6, 368, 370-1 in Canada 192-3, 207-8 in China 205-7. 296 as classification 187-8 and global commodity chains 369-72 in Hong Kong 185, 196-7, 207-8 in Indonesia 201-2, 207-8 in Japan 185, 193-4, 207-8 in Malaysia 202-4, 207-8 measurement of 185-209 by foreign MNCs' share of GDP 186-7, 191-2, 207-9 by inward FDI as ratio of GDP 186-7, 190, 191-2, 207-9 methodology 191-2 problems 186-7 OLI advantages 188-9 in Singapore 185, 198-200, 207-8, 290 in S. Korea 185, 197-8, 207-8 in Taiwan 185, 200-1, 207-8 and technology flows 253, 258, 263-7 in Thailand 204-5, 207-8 trade propensities 189-90 in USA 194-6, 207-8

NAFTA 5, 51, 163, 164, 168, 171 Nanya Plastics 233 nationalism 289 US economic 347-8 natural resources changing use of 308-20 large scale 327-31 use in traditional production 305-6, 308-10 Nauru 317 near-NIEs 20, 23, 37, 84 economic growth 58-9 industrialization 289-90 non-monetarized sectors 84 see also China: Indonesia; Malaysia; Thailand Nelson, R.R. 253-4, 261 neo-classical economics 4-7, 6 and economic growth 4-7, 138-43, 358-62, 364 trade liberalization 141-2, 181-2 neo-liberalism 95 networks global commodity chains 241, 369-72 globalization and 368-72 guanxi and kin networks 376-7, 379 institutional 150-2 Japanese interfirm 220, 222-3 Taiwanese family/personal 235-6 New Caledonia 317 New Zealand 3, 8, 20, 22, 49, 307 inward FDI 46 technology and innovation 255, 256, 266 trade 35, 36 newly industrialized economies (NIEs) 20, 22-3 economic growth 58-9, 139-41 and 'flying geese model' 36-8, 37, 267-70, 375-6 income distribution 279-82, 284-9 inward FDI 32-3, 97, 196-201 neo-classical account of development 70-1, 277, 358-9 outward FDI 43-4, 46, 97, 187, 243, 263, 290 role of education 120-3 standards of living 48, 49, 60 technology flows 260-1, 270 total factor productivity 67-8, 78-9 trade 34-45, 52, 99, 103 intra-regional 43, 103 with Japan 34, 99 trade surplus 40-1 with USA 34, 40-1, 42-3, 52 see also East Asian miracle; Hong Kong;

near-NIEs: Singapore; South Korea; Taiwan Nike 265 Nissan 223, 225-6, 374 non-monetarized sectors 84 North American Free Trade Area see NAFTA North Korea 7, 24, 25 OECD 4-5, 71, 80, 178, 352 official development assistance 97 Ohira Masayoshi 165-6 Onis. Z. 147-8 Open Economic Associations (OEAs) 179-81, 182 'open regionalism' 16, 182, 343 'openness' 4-5, 63, 66-7, 71-2, 73-4 see also free trade; protectionism: trade barriers: trade liberalization original equipment manufacturing (OEM) 246, 372, 376 Osaka Action Plan 165, 171-2, 179, 342 outsourcing 265 own brand manufacturing (OEM) 377 ownership, company 215, 218-19, 228, 234 ownership-locational-internalization (OLI) paradigm, MNCs 188-9 Pacific Basin Economic Council see PBEC Pacific Basin Forum (PBF) 340 Pacific Islands changing use of resources 305-7, 312-15 forest use and logging 315-16 fresh water supply 323-5, 325 land tenure 308-15, 309-10 migration 322, 325-6 population growth 323-5 use of marine resources 316-20 PAFTAD 165 Pakistan 64, 65 Papua New Guinea (PNG) 9, 305, 307, 314-15, 327 fisheries 317, 318 land tenure and management 312, 314-15 mining activities 328, 329-30 population growth 314-15, 320, 323, 323-4 Park Chung Hee 156 Partners for Progress (PFP) 172 PBEC 165, 340, 342 PECC 166-7, 173, 270, 340, 342 People's Liberation Army (PLA), China 378-9, 380 Peru 276-7, 278, 285-6, 299-300

Philippines 9, 24-5, 276-7, 278, 307, 309 economic growth 1, 12, 28 financial development 84, 93 gender issues 114-19, 127, 299-300, 301 inward FDI 33, 43-4, 45 labour markets 50 migration of women from 127-8, 129-31 mining activities 328 poverty 48, 285-6 savings 64, 65, 88 technology and innovation 256, 259, 266 trade 35, 108-9 Plaza Accord 32, 44, 52, 104, 262 Pohang Iron and Steel Company 244 Poly Technologies 379 Polynesia 307 Poongsan Corporation 230, 231 population growth 305, 314-15, 320-6 and environment 314-15, 320-5, 323-4 Port Moresby 307, 323-4 portfolio investment 46-7, 97-9, 215 POSCO 370 Post-Cold War Trade and Development see PAFTAD postal savings institutions 88 poverty 23-4, 48-9, 59-60, 301 near-NIEs 291-3, 294-5 NIEs 282-9 price distortion 25, 31, 72 private sector 24-5, 141, 144-8 in China 75-6. 79. 294-7. 378. 380 Japan 73 savings and investment 73, 79 Taiwan 233-8 technology in 254, 255 production function approach 62-3 property rights China 14, 361-2 intellectual 172, 175 prostitution child 131-3, 301 migrant workers 130-1 protectionism 29-30, 141-2 infant industry protection 175 see also 'openness'; trade barriers; trade liberalization: VERs public/state sector China 296, 378, 379 and economic growth 26, 141 R&D 254. 255 Taiwan 26, 233, 238

R&D 250-3, 259, 262-3 chaebol non-co-operative 229 and economic growth 250-3, 256 externalities and spillover effects 251-2, 251, 263 and FDI 259, 262-3 and higher education 255, 258 see also innovation; technology Ravenhill, B. 262 see also Bernard, M. and Ravenhill, J. recruitment 121-3, 219, 229 regional integration APEC model 164-7, 179-81 concept of 'region' 338-9, 353 EU model 163, 164, 183, 344 five stages of 180 MNCs as agents of 185-6, 368, 370-1 and multilateral liberalization 181-2 see also economic co-operation regionalism and 'Asian identity' 343-4 and multilateral liberalization 181-2, 335, 337 'open' 16, 182, 343 and trade 43, 340-1 regionalization 2-5, 338-41, 342 de facto 338-41 and globalization 339, 342, 351 of innovation systems 260, 270-1 regulation, financial 89, 95-6 reverse engineering 251, 258 ringi system 224 Samsung 57, 142-3, 263 Sanwa 150 Saudi Arabia 127 savings 23-4, 63-6, 72-3, 87-9 Science and Technology Task Force, PECC 270 segmentation, labour market 52-3 service trade 175, 176, 178 sex industry 128, 130-3, 301 shareholders 218-19, 221, 242-3, 366 short-termism 218–19 Silin, R.H. 236, 237 Singapore 26, 27, 99, 276-7, 278 authoritarian capitalism 8 basic indicators 278 economic growth 1, 11-12, 276-7, 278, 375 entrepôt trade 155 FDI inward 33, 45, 97, 157, 198-200

outward 44, 97, 187, 290 financial development 93, 98 foreign MNC presence 185, 198-200, 207-8, 290 gender issues 114-19, 121, 124, 134, 299-300 labour markets 49, 59, 143, 157 as NIE 20. 22-3 poverty and inequality 280-1, 285-6, 289, 301 public and private savings 88, 89 state-led industrialization 141, 143, 148-9, 154, 157 technology and innovation 256, 258, 266-7 total factor productivity 67, 68 trade 35, 108-9 women migrants 127, 128-9, 130, 131 Singapore-Johore-Riau growth triangle 168, 169 Singh, A. see Chakravarty, S. and Singh, A. small and medium sized enterprises (SMEs) 88-9, 91-2, 218-19 Japan 221, 366 Malaysia 291, 293 Taiwan 149, 152, 156, 233-8, 246, 258, 288 social equity, and economic growth 279-81 sogo shosha 151, 235 Solomon Islands 320, 329-30 fisheries 317, 318-19 logging 314, 316, 327 Sony Corporation 150 South China economic zone 168-70 South Korea 141-4, 227-32, 238-40, 276-7, 287-90 authoritarian/state-directed capitalism 8, 142, 365, 366-7 banks and banking 25, 88-9, 92, 228, 242 basic indicators 278 business systems 227-32, 232, 238-40, 243-5 business-state interdependence 150, 151, 152-3, 367 chaebol 89, 142-3, 151, 156, 227-32, 243-5, 364 economic growth 1, 11-12, 14, 30, 57 education 120, 231, 244, 278, 284 financial development 73, 84, 92, 93, 142 foreign MNC presence 185, 197-8, 207-8 gender issues 114-19, 120-1, 124, 133, 231, 298, 299-300 global commodity chains 370, 372, 376 import-substituting industrialization 30 industrial policy 70-1, 142-3, 257-8 industrial relations 229-32, 244-5, 287-8 inflation 30 investment inward FDI 33, 72, 97, 197-8 outward FDI 44, 97, 187, 243, 263, 290 portfolio 46-7 state direction of 25, 26-8, 72-3, 88-9, 156, 228 US post-war aid 66, 97 Japanese colonial past 262, 282-4, 373 labour markets 49, 50, 59, 132 land reform 283-4

'late industrialization' 146-7 as NIE 20, 22-3, 37 poverty and social equity 60, 280, 284-8 protectionism or liberalization 27-8, 30, 71, 73-4, 94-5, 141-2 savings 42, 64-6, 72-3 sex industry 132 state-led industrialization 73, 139, 142-3, 146-7, 148, 152-3, 155-6 technology and innovation 256, 258, 261-3, 266-7 total factor productivity 67, 68, 69 trade 35, 36, 42-3 export-led development 27-8, 30, 155, 156, 283-4, 287-90 trade unions 244, 287-8 wages 50, 231, 244-5 South Tarawa 323, 325 sovereignty, state and globalization 341, 344-5 and inter-state co-operation 343-4 specialization Japanese firms 221-2 Korean firms 231–2 Taiwanese firms 234, 246 work 216 spillover effects, R&D 251-2, 263 Sri Lanka 64, 65, 127, 129-30 stability, external 30-1, 52 standards of living 23, 48, 49 see also distribution of income; poverty state co-ordination role 73, 80-1 and globalization 341–2 levels of involvement 24-6 and regionalization 342, 353 role in gender equality 123-5 strong state model 148-9 see also business-state relations; developmental state; government intervention state-directed capitalism 365, 366-7 state-led industrialization 23-8, 39-40, 80-1, 139, 142-9, 154-9, 362-4 China (PRC) 74-6, 361-2, 363 diversity in 154-9 Japan 39, 73, 80-1, 139, 141, 150-2 Singapore 141, 143, 148-9, 154, 157 South Korea 73, 139, 142-3, 146-7, 148, 152-3, 155-6 Taiwan 39, 139, 141, 149, 152-3, 156 'sterilization', capital inflows 99 stock market 92-3, 97, 107-8, 366 structural adjustment policies 279, 305, 359, 361, 364 sub-regional informal economic zones (SREZs) 168-70, 169-70 Sub-Saharan Africa 58, 60 subcontracting

Japanese firms 220-2, 223 Korean chaebol 228 overseas Chinese networks 379-80 Taiwan 234, 235, 288 subsistence agriculture 305, 309, 313, 314-15 subsistence fisheries 316-17 Suharto, General and family 293-4, 294, 295 Sukarno, President 293, 295 Sumitomo 150, 222 sustainability East Asian miracle 68-9 and population growth 326 of traditional production systems 304, 310-11, 321 swidden cultivation 310, 322 systems see business systems; financial systems Taiwan 99, 233-40, 245-7, 276-7, 278, 282-6, 287-90 banks and banking 233, 235 basic indicators 278 business systems 233-40, 245-7 Chinese family businesses (CFBs) 126, 233-8, 246-7, 367 private sector 233-8 public sector 26, 233, 238 SMEs 149, 152, 156, 233–8, 246, 258, 288 business-state interdependence 150, 151, 152-3 capital outflow controls 94 cultural and language factors 288 economic growth 1, 11-12, 39, 57 education 278, 284 financial development 84, 92, 93, 142 foreign MNC presence 185, 200-1, 207-8 gender issues 114-19, 121, 124-5, 126, 133-4, 237, 298, 299-300 global commodity chains 370, 371-2, 376 industrial policy 70-1, 142-3 industrial relations 236-8 investment inward FDI 33, 45, 97, 200-1 outward FDI 44, 97, 187, 263, 290 state 26 US post-war aid 97 Japanese colonial past 262, 282-4 Kuomintang 152, 245, 246, 367 labour market 49, 50, 59, 245-6 land reform 283-4 Living Rooms as Factories programme 124 Mothers' Workshops programme 124-5 negotiated/co-ordinated proprietorial capitalism 8-9, 365, 367 as NIE 20, 22-3, 37 'openness' 73, 141 population data 236, 278

poverty and equity 280-1, 284-6, 288 savings 42, 64, 65 sex industry 132, 133 state-led industrialization 39, 139, 141, 149, 152-3, 156 technology and innovation 256, 258, 261-3, 266-7 total factor productivity 67, 68 trade 35, 36, 42-3 export-led development 155, 287-90, 359-60 women migrants 127, 128-9, 130, 132 Taiwan Electrical Appliances Manfacturer Association 151 Taiwan Textile Federation 151 tariffs commitment to reduce 177-8 levels in APEC 173-4 Tatung 367 technology APEC co-operation 172 and economic growth 69, 72, 79-80, 250-3, 290 emerging regional structures 270-1 and flows of skilled labour 266-7 interfirm relationships and transfer of 263-6 and 'late industrialization' 252-3 licensing 72 market- or state-driven 252, 255 and MNCs 253, 258, 263-6 national systems of innovation ASEAN Asian states 258-9 'Asian' Asian states 257-8 in China 259-60 'European' Asian states 255-7 reverse engineering 251, 258 technology flows 260-7, 270 see also innovation; R&D Thailand 39, 94 basic indicators 278 chaotic capitalism 9 economic growth 1, 12, 14, 28, 108 FDI inward 33, 43-5, 74, 97, 108, 204-5, 263 outward 97 female emigration 127, 128, 130-1 foreign MNC presence 204-5, 207-8 gender issues 114-16, 118-19, 127, 299-300 import-substituting industrialization 25 labour market 50 as near-NIE 20, 37 poverty and equity 285-6, 301 savings 64, 65, 88 sex industry 133, 301 stock market 93 technology and innovation 256, 259, 262, 266

trade 35, 108-9 Thistlethwaite, B. and Davis, D. 320, 321 Thurow, L. 77 Tokai Bank 223 Tonga 317 Toshiba 223, 263 total factor productivity (TFP) theory 61-9, 78-80 township and village enterprises (TVEs) 75, 296, 379 Toyota 44-5, 221, 223 trade 28-46, 51-3, 101-5 ASEAN states 28-30, 31-2, 35, 38, 103-5 comparative advantage 35-8 complementarity 39, 170 East Asia-Europe 101-4, 346 and exchange rate regimes 101-5 inter-industry 39 intra-APEC 170 intra-industry 38 intra-regional 31, 43, 52, 101-2, 103, 240-1, 339 Japan as potential problem 31-2 'managed' 29-30, 31 and regionalization 340-1 trans-Pacific 31-2, 42-3 and USA 40-3, 101-4, 339-40 trade barriers 173-9 in APEC members 173-7 facilitation measures 178-9 FDI restrictions 175, 178 intellectual property rights 172, 175 Japan 31–2 non-tariff barriers 174, 178 reduction 177-8, 341 service trade 175, 176, 178 tariff levels 173-4 see also protectionism; trade liberalization trade imbalances 28-30, 31-2, 33-4, 40-1, 108-9 ASEAN 28-30, 31-2 Chinese 374 current accounts 108-9 Japanese surpluses 28-9, 31-2, 34, 52, 99, 105 NIEs surpluses 28-9, 40-1 US deficits 28-9, 32, 33-4, 52, 374 trade intensity 4, 35, 103 trade liberalization 4-5, 141-2, 164-5, 171-8, 347-53 agriculture 175 APEC 164-5, 171-81, 351 ASEAN 341, 342 concerted unilateral 171-2, 173, 178-9, 351 and economic growth 27-8 global 181–2 most favoured nation (MFN) basis 180, 182, 343

multilateral 5, 181-2, 335, 337, 347-50, 352-3 in neo-classical economics 141-2, 181-2 regional integration and 181-2 in South Korea 27-8, 30, 95 see also free trade: GATT: 'openness'; trade barriers; World Bank traditional production systems 303-32 adaptation to changing use of resources 305, 308-20, 314-15 and environment 306-7, 321-5 hunting and gathering 305, 308-10 and identity 303-4 impact of mining 311, 328-31 land tenure and resource use 308-10, 309-10 and large-scale resource use 327-31 and logging 311, 314, 315-16, 327 marine resources in 305, 308, 316-18 and population changes 305, 321-5 sustainability of 304, 310-11, 321 swidden cultivation 310, 322 trans-Pacific trade 31-2, 42-3 Tuvalu 307, 317, 323, 324-5 unemployment 49-50 Unilever 379 Unions in Anglo-American firms 219-20 Japanese 225-6 Korean 244, 287-8 Taiwanese 245-6 United Kingdom, shareholder capitalism 365, 366, 368 United Malay Nationalist Organization (UMNO) 291, 292 United States (USA) 80, 194-6, 207-8, 347-50 and ASEM 347-50 budget deficits 32, 41-2 dominance of 14, 20, 21-2 economic growth 55-6, 58, 77-8 economic nationalism 347-8 foreign MNC presence 194-6, 207-8 international role of US dollar 99-105 inward FDI 46, 194-6 managerial capitalism 365, 366, 369 mining activities 328 in post-hegemonic era 337, 344, 348 post-war aid in East Asia 96-7, 261-2, 277, 357 R&D and technology 255, 256, 260-2, 271 savings 32 standard of living 49 total factor productivity 68, 80

trade 2-3, 28-9, 31-4, 35, 40-3, 101-4 comparative advantage 36 EU-USA-Asia 'triangle' 338, 345-6, 347 'extra-territoriality' principles 347, 348 protectionism and VERs 29-30 trade deficits 28-9, 32, 33-4, 52, 374 unemployment 77 unionization 219 wages 77-8 see also Anglo-American business systems; Anglo-American economies; Anglo-American financial systems Uruguay Round, GATT 177-8, 181-2, 348 Vanuatu 211, 212, 218, 317 VERs 29-30, 31 vertical integration 215, 227, 234 Vietnam 20, 48, 50, 259, 322 command economy 7, 24, 25 inward FDI 97, 263 Volkswagen 379, 381 Wade, R. 138, 141-2, 233, 361, 362-3 Wade, R. and White, G. 143-4 wage labour, agricultural 312, 313-14 wages gender inequality 117, 119-20, 231 government policies 50-1 and migrant workers 119, 129 South Korea 50. 231. 244–5 United States 77-8 Ward, R.G. 316, 325-6 water supply 285, 294, 323-5, 325 wei ji 356, 382 Weiss, L. 149-51, 152-3 welfare and social security 87, 89 Western Samoa 316, 317, 326 White, G. see Wade, R. and White, G. women in employment 114-15, 119-20 by industry 114, 116, 117 by occupation 117, 118 in entertainment and sex industries 128, 130-1 and family roles 124-5 foreign domestic workers 127, 128-30 in household-based entrepreneurship 125-6 in Japan 224-5 as migrant workers 127-31 resistance to exploitation 133-5 see also gender equality work co-ordination and control 216, 239 Anglo-American business systems 219-20

and business systems 216-17 Japanese business systems 223-6 Korean 229–32 Taiwanese 236-8 working conditions Korea 229-32, 244-5, 287-8 migrant domestic labour 129-30 Working Party on Industrial Science and Technology (WGIST) 270-1 World Bank 137, 138 Berg Report (1981) 360 call for 'market-friendly' approach 5, 61, 66-7, 70-1, 277, 364 and government intervention 70-1, 137, 138-9 growth theories and East Asia 4-6, 60-9, 70-2, 78-80, 138-9, 359-61, 364 structural adjustment policies 279, 305, 359, 361, 364 The Challenge of Development (1991) 61, 66-7 The East Asian Miracle (1993) 5-6, 60-1, 61, 72, 279-80, 361 trade liberalization 4-5, 139, 360 World Development Report (1983) 360 WTO (World Trade Organization) 4-5, 183, 347, 348, 350, 352-3 and APEC 164, 175

Yellow Sea economic zone 168-70

zaibatsu 150-1, 221-2, 242